Table of Contents

First Revised Sheet No. 0	17
Original Sheet No. 1	18
Original Sheet No. 2	19
Original Sheet No. 3	20
Original Sheet No. 4	21
Original Sheet No. 5	22
Original Sheet No. 6	23
Original Sheet No. 7	24
First Revised Sheet No. 8	25
Original Sheet No. 8A	27
Original Sheet No. 8B	29
Original Sheet No. 9	31
First Revised Sheet No. 10	33
Original Sheet No. 10A	35
Original Sheet No. 10B	37
Original Sheet No. 11	39
First Revised Sheet No. 12	41
Original Sheet No. 12A	43
Original Sheet No. 12B	45
First Revised Sheet No. 13	47
Original Sheet No. 13A	49
Original Sheet No. 13B	51
Original Sheet No. 14	53
Original Sheet No. 15	55
First Revised Sheet No. 16	57
Original Sheet No. 16A	58
Original Sheet No. 16B	59
Original Sheet No. 17	60
First Revised Sheet No. 18	61
Original Sheet No. 18A	63
Original Sheet No. 18B	65
Original Sheet No. 19	67
First Revised Sheet No. 20	69
Original Sheet No. 20A	71
Original Sheet No. 20B	73
Original Sheet No. 21	75
First Revised Sheet No. 22	77
Original Sheet No. 22A	78
Original Sheet No. 22B	79
First Revised Sheet No. 23	80
Original Sheet No. 23A	81
Original Sheet No. 23B	82
Original Sheet No. 24	83
First Revised Sheet No. 25	84
Original Sheet No. 25A	85

First Revised Sheet No. 26	86
Original Sheet No. 26A	87
Second Revised Sheet No. 27	88
Original Sheet No. 27A	89
Original Sheet No. 27B	90
First Revised Sheet No. 28	91
Original Sheet No. 28A	92
First Revised Sheet No. 29	93
Original Sheet No. 29A	94
Original Sheet No. 30	95
Original Sheet No. 31	96
Second Revised Sheet No. 32	97
First Revised Sheet No. 33	98
Original Sheet No. 33A	99
Original Sheet No. 33B	100
First Revised Sheet No. 33C	101
First Revised Sheet No. 33D	102
Original Sheet No. 33E	103
Original Sheet No. 33F	104
First Revised Sheet No. 33G	105
First Revised Sheet No. 33H	106
First Revised Sheet No. 33I	108
First Revised Sheet No. 33J	109
Original Sheet No. 33K	110
Original Sheet No. 33L	111
Original Sheet No. 33N	112
First Revised Sheet No. 33N	113
First Revised Sheet No. 330	114
Original Sheet No. 33P	115
First Revised Sheet No. 33C.01	116
Third Revised Sheet No. 33D.01	118
Second Revised Sheet No. 33G.01	119
First Revised Sheet No. 33H.01	120
First Revised Sheet No. 331.01	121
First Revised Sheet No. 33L.01	122
	123
Original Sheet No. 33N.01	
Original Sheet No. 33P.01	124
First Revised Sheet No. 33C.02	125
Original Sheet No. 33G.02	128
First Revised Sheet No. 33H.02	130
First Revised Sheet No. 33J.02	131
Original Sheet No. 33L.02	135
Original Sheet No. 33N.02	136
First Revised Sheet No. 33C.03	137
Original Sheet No. 33G.03	138
Original Sheet No. 33N.03	139

0 101 (N. 000.04	4.40
Original Sheet No. 33G.04	140
First Revised Sheet No. 33N.04	141
Second Revised Sheet No. 33N.04	142
Original Sheet No. 33G.05	143
Second Revised Sheet No. 33G.06	144
Second Revised Sheet No. 34	146
Original Sheet No. 34A	147
Original Sheet No. 34B	148
Third Revised Sheet No. 34C	149
Original Sheet No. 34D	150
First Revised Sheet No. 34E	151
First Revised Sheet No. 34F	152
Second Revised Sheet No. 34G	153
Original Sheet No. 34H	154
Original Sheet No. 34I	155
Original Sheet No. 34J	156
Original Sheet No. 34K	158
First Revised Sheet No. 34L	159
Original Sheet No. 34M	160
Second Revised Sheet No. 34C.01	161
Original Sheet No. 34D.01	162
Original Sheet No. 34E.01	163
Original Sheet No. 34H.01	169
Original Sheet No. 34J.01	170
Original Sheet No. 34L.01	171
Original Sheet No. 34C.02	172
Original Sheet No. 34D.02	173
Original Sheet No. 34E.02	178
Original Sheet No. 34L.02	179
Original Sheet No. 34C.03	180
Original Sheet No. 34D.03	181
Original Sheet No. 34E.03	186
First Revised Sheet No. 34E.03	187
Original Sheet No. 34L.03	188
Original Sheet No. 34C.04	189
Original Sheet No. 34E.04	190
Original Sheet No. 34C.05	191
First Revised Sheet No. 34E.05	192
Original Sheet No. 34C.06	193
Original Sheet No. 35	194
Original Sheet No. 35A	195
Original Sheet No. 35B	196
Third Revised Sheet No. 35C	197
Original Sheet No. 35A.01	198
Second Revised Sheet No. 35B.01	199
Fourth Revised Sheet No. 35C.01	200

First Revised Sheet No. 35B.02	201
Third Revised Sheet No. 35C.02	203
First Revised Sheet No. 35B.03	204
Third Revised Sheet No. 35C.03	205
First Revised Sheet No. 35B.04	206
Third Revised Sheet No. 35C.04	207
First Revised Sheet No. 35B.05	208
Second Revised Sheet No. 35C.05	209
Original Sheet No. 35B.06	210
Original Sheet No. 35C.06	211
First Revised Sheet No. 35B.07	212
Original Sheet No. 35C.07	214
First Revised Sheet No. 35B.08	215
First Revised Sheet No. 35C.08	216
First Revised Sheet No. 35B.09	217
Original Sheet No. 35C.09	218
Original Sheet No. 35C.10	219
Original Sheet No. 35C.11	220
First Revised Sheet No. 35C.12	221
Original Sheet No. 35C.13	222
Original Sheet No. 35C.14	223
Original Sheet No. 35C.15	224
Original Sheet No. 35C.16	225
Original Sheet No. 35C.17	226
Original Sheet No. 35C.18	227
Original Sheet No. 35C.19	228
Original Sheet No. 35C.20	229
Original Sheet No. 36	230
Original Sheet No. 37	232
Sheet Nos. 38 - 39	233
Original Sheet No. 40	234
Original Sheet No. 41	235
Original Sheet No. 42	236
Original Sheet No. 43	237
Original Sheet No. 44	238
Original Sheet No. 45	239
Original Sheet No. 46	240
Original Sheet No. 47	241
Original Sheet No. 48	242
Original Sheet No. 49	243
Original Sheet No. 50	244
Original Sheet No. 51	245
Original Sheet No. 52	246
Original Sheet No. 53	247
Original Sheet No. 54	248
Original Sheet No. 55	249
Chighter Officer (10. 00	270

Original Chaot No. EG	250
Original Sheet No. 56	250
Original Sheet No. 57	251
Original Sheet No. 58	252
Original Sheet No. 59	253
Original Sheet No. 60	254
Original Sheet No. 61	255
Original Sheet No. 62	256
Original Sheet No. 63	257
Original Sheet No. 64	258
Original Sheet No. 65	259
Original Sheet No. 66	260
Original Sheet No. 67	261
Original Sheet No. 68	262
Original Sheet No. 69	263
Original Sheet No. 70	264
Original Sheet No. 71	265
Original Sheet No. 72	266
Original Sheet No. 73	267
Original Sheet No. 74	268
Original Sheet No. 75	269
Original Sheet No. 76	270
Original Sheet No. 77	271
Original Sheet No. 78	272
Original Sheet No. 79	273
•	273
Original Sheet No. 80	274
Original Sheet No. 81	
Original Sheet No. 82	276
Original Sheet No. 83	277
Original Sheet No. 84	278
Original Sheet No. 85	279
Original Sheet No. 86	280
Original Sheet No. 87	281
Original Sheet No. 88	282
Original Sheet No. 89	283
Original Sheet No. 90	284
Original Sheet No. 91	285
Original Sheet No. 92	286
Original Sheet No. 93	287
Original Sheet No. 94	288
Original Sheet No. 95	289
Original Sheet No. 96	290
Original Sheet No. 97	291
Original Sheet No. 98	292
Original Sheet No. 99	293
Original Sheet No. 100	294
Original Sheet No. 101	295
•	

Original Shoot No. 102	296
Original Sheet No. 102	297
Original Sheet No. 103	
Original Sheet No. 104	298
Original Sheet No. 105	299
Original Sheet No. 106	300
Original Sheet No. 107	301
Original Sheet No. 108	302
Original Sheet No. 109	303
Original Sheet No. 110	304
Original Sheet No. 111	305
Original Sheet No. 112	306
Original Sheet No. 113	307
Original Sheet No. 114	308
Original Sheet No. 115	309
Original Sheet No. 116	310
Original Sheet No. 117	311
Original Sheet No. 118	312
Original Sheet No. 119	313
Original Sheet No. 120	314
Original Sheet No. 121	315
Original Sheet No. 122	316
Original Sheet No. 123	317
Original Sheet No. 124	318
Original Sheet No. 125	319
Original Sheet No. 126	320
Original Sheet No. 127	321
Original Sheet No. 128	322
Original Sheet No. 129	323
Original Sheet No. 129 Original Sheet No. 130	324
Original Sheet No. 131	325
•	
Original Sheet No. 132	326
Original Sheet No. 133	327
Original Sheet No. 134	328
Original Sheet No. 135	329
Original Sheet No. 136	330
Original Sheet No. 137	331
Original Sheet No. 138	332
Original Sheet No. 139	333
Original Sheet No. 140	334
Original Sheet No. 141	335
Original Sheet No. 142	336
Original Sheet No. 143	337
Original Sheet No. 144	338
Original Sheet No. 145	339
Original Sheet No. 146	340
Original Sheet No. 147	341

Original Sheet No. 148	342
Original Sheet No. 149	343
Original Sheet No. 150	344
Original Sheet No. 151	345
<u> </u>	
Original Sheet No. 152	346
Original Sheet No. 153	347
Original Sheet No. 154	348
Original Sheet No. 155	349
Original Sheet No. 156	350
Original Sheet No. 157	351
Original Sheet No. 158	352
Original Sheet No. 159	353
<u> </u>	
Original Sheet No. 160	354
Original Sheet No. 161	355
Original Sheet No. 162	356
Original Sheet No. 163	357
Original Sheet No. 164	358
Original Sheet No. 165	359
Original Sheet No. 166	360
Original Sheet No. 167	361
Original Sheet No. 168	362
<u> </u>	
Original Sheet No. 169	363
Original Sheet No. 170	364
Original Sheet No. 171	365
Original Sheet No. 172	366
Original Sheet No. 173	367
Original Sheet No. 174	368
Original Sheet No. 175	369
Original Sheet No. 176	370
Original Sheet No. 177	371
Original Sheet No. 178	372
Original Sheet No. 179	373
Original Sheet No. 180	374
Original Sheet No. 181	375
Original Sheet No. 182	376
Original Sheet No. 183	377
Original Sheet No. 184	378
Original Sheet No. 185	379
Original Sheet No. 186	380
Original Sheet No. 187	381
<u> </u>	
Original Sheet No. 188	382
Original Sheet No. 189	383
Original Sheet No. 190	384
Original Sheet No. 191	385
Original Sheet No. 192	386
Original Sheet No. 193	387
•	

Original Sheet No. 194 Original Sheet No. 195	388 389
Original Sheet No. 196	390
Original Sheet No. 197	391
Original Sheet No. 198	392
Original Sheet No. 199	393
Original Sheet No. 200	394
First Revised Sheet No. 201	395
Original Sheet No. 201A	396
Original Sheet No. 202	397
Original Sheet No. 203	398
Original Sheet No. 204	399
Original Sheet No. 205	400
Original Sheet No. 206	401
Original Sheet No. 207	402
Original Sheet No. 208	403
Original Sheet No. 209	404
Original Sheet No. 210	405
Original Sheet No. 211	406
Original Sheet No. 212	407
Original Sheet No. 213	408
Original Sheet No. 214	409
Original Sheet No. 215	410
Original Sheet No. 216	411
Original Sheet No. 217	412
Original Sheet No. 218	413
Original Sheet No. 219	414
Original Sheet No. 220	415
Original Sheet No. 221	416
Original Sheet No. 222	417
Original Sheet No. 223	418
Original Sheet No. 224	419
Original Sheet No. 225	420
Original Sheet No. 226	421
Original Sheet No. 227	422
Original Sheet No. 228	423
Original Sheet No. 229	424
Original Sheet No. 230	425 426
Original Sheet No. 231	420
Original Sheet No. 232	427
Original Sheet No. 233 Sheet Nos. 234 - 274	429
Original Sheet No. 275	430
Original Sheet No. 276	430
Original Sheet No. 277	432
Original Sheet No. 277 Original Sheet No. 278	432
Original Oricet No. 270	400

First Davised Cheet No. 270	424
First Revised Sheet No. 279	434
Original Sheet No. 280	435
Original Sheet No. 281	436
Original Sheet No. 281A	437
Sheet No. 282	438
Sheet No. 283	439
Original Sheet No. 284	440
Original Sheet No. 285	441
Original Sheet No. 286	442
Original Sheet No. 287	443
Original Sheet No. 288	444
Original Sheet No. 289	445
Original Sheet No. 290	446
<u> </u>	
Original Sheet No. 291	447
Original Sheet No. 292	448
Original Sheet No. 293	449
Original Sheet No. 294	450
Original Sheet No. 295	451
Original Sheet No. 296	452
Original Sheet No. 297	453
Original Sheet No. 298	454
Original Sheet No. 299	455
Original Sheet No. 300	456
Original Sheet No. 301	457
Original Sheet No. 302	458
Original Sheet No. 303	459
Original Sheet No. 304	460
Original Sheet No. 305	461
First Revised Sheet No. 306	462
Original Sheet No. 306A	463
First Revised Sheet No. 307	464
Original Sheet No. 308	465
Original Sheet No. 309	466
Original Sheet No. 310	467
Original Sheet No. 311	468
Original Sheet No. 312	469
Original Sheet No. 313	470
Original Sheet No. 314	471
Original Sheet No. 315	472
Original Sheet No. 316	473
Original Sheet No. 317	474
Original Sheet No. 318	475
Original Sheet No. 319	476
Original Sheet No. 320	477
Original Sheet No. 321	478
Original Sheet No. 322	479
-	

Original Sheet No. 323	480
Original Sheet No. 324	481
Original Sheet No. 325	482
<u> </u>	483
Original Sheet No. 326	
Original Sheet No. 327	484
Original Sheet No. 328	485
Original Sheet No. 329	486
Original Sheet No. 330	487
Original Sheet No. 331	488
Original Sheet No. 332	489
<u> </u>	
Original Sheet No. 333	490
Original Sheet No. 334	491
Original Sheet No. 335	492
Original Sheet No. 336	493
Original Sheet No. 337	494
Original Sheet No. 338	495
Original Sheet No. 339	496
· ·	
Original Sheet No. 340	497
Original Sheet No. 341	498
Original Sheet No. 342	499
Original Sheet No. 343	500
Original Sheet No. 344	501
Original Sheet No. 345	502
Original Sheet No. 346	503
Original Sheet No. 347	504
Original Sheet No. 348	505
<u> </u>	506
Original Sheet No. 349	
Original Sheet No. 350	507
Original Sheet No. 351	508
Original Sheet No. 352	509
Original Sheet No. 353	510
Original Sheet No. 354	511
Original Sheet No. 355	512
Original Sheet No. 356	513
Original Sheet No. 357	514
Original Sheet No. 358	515
<u> </u>	
Original Sheet No. 359	516
Original Sheet No. 360	517
Original Sheet No. 361	518
Original Sheet No. 362	519
Original Sheet No. 363	520
Original Sheet No. 364	521
Original Sheet No. 365	522
Original Sheet No. 366	523
Original Sheet No. 367	524
<u> </u>	
Original Sheet No. 368	525

Original Shoot No. 260	F26
Original Sheet No. 369	526
Original Sheet No. 370	527
Original Sheet No. 371	528
Original Sheet No. 372	529
Original Sheet No. 373	530
Original Sheet No. 374	531
Original Sheet No. 375	532
Original Sheet No. 376	533
Original Sheet No. 377	534
Original Sheet No. 378	535
Original Sheet No. 379	536
Original Sheet No. 380	537
Original Sheet No. 381	538
<u> </u>	
Original Sheet No. 382	539
Original Sheet No. 383	540
Original Sheet No. 384	541
Original Sheet No. 385	542
Original Sheet No. 386	543
Original Sheet No. 387	544
Original Sheet No. 388	545
Original Sheet No. 389	546
Original Sheet No. 390	547
Original Sheet No. 391	548
Original Sheet No. 392	549
Original Sheet No. 393	550
Original Sheet No. 394	551
Original Sheet No. 395	552
Original Sheet No. 396	553
Original Sheet No. 397	554
Original Sheet No. 398	555
	556
Original Sheet No. 399	
Original Sheet No. 400	557
Original Sheet No. 401	558
Original Sheet No. 402	559
Original Sheet No. 403	560
Original Sheet No. 404	561
Original Sheet No. 405	562
First Revised Sheet No. 405	563
Original Sheet No. 405A	564
First Revised Sheet No. 406	565
Original Sheet No. 407	566
Original Sheet No. 408	567
Original Sheet No. 409	568
Original Sheet No. 410	569
Original Sheet No. 411	570
Sheet No. 412	571
Officet No. 712	57 1

Original Sheet No. 413	572
First Revised Sheet No. 413	573
Original Sheet No. 414	574
First Revised Sheet No. 414	575
Original Sheet No. 415	576
Original Sheet No. 416	577
First Revised Sheet No. 417	578
Original Sheet No. 417A	579
First Revised Sheet No. 418	580
First Revised Sheet No. 419	581
Second Revised Sheet No. 419	582
Original Sheet No. 419A	583
First Revised Sheet No. 420	584
Original Sheet No. 421	585
Substitute First Revised Sheet No. 422	586
Second Revised Sheet No. 422	587
Original Sheet No. 422A	588
First Revised Sheet No. 422A	589
First Revised Sheet No. 423	590
Second Revised Sheet No. 423	591
Original Sheet No. 424	592
First Revised Sheet No. 425	593
Original Sheet No. 425A	594
First Revised Sheet No. 426	595
Original Sheet No. 427	596
First Revised Sheet No. 428	597
Original Sheet No. 429	598
Original Sheet No. 430	599
Original Sheet No. 431	600
Original Sheet No. 432	601
Original Sheet No. 433	602
First Revised Sheet No. 434	603
First Revised Sheet No. 435	604
First Revised Sheet No. 436	605
Original Sheet No. 436A	606
Original Sheet No. 437	607
Original Sheet No. 438	608
Original Sheet No. 439	609
Original Sheet No. 440	610
Original Sheet No. 441	611
Original Sheet No. 442	612
Original Sheet No. 443	613
	614
Original Sheet No. 444	615
Original Sheet No. 445	
Original Sheet No. 446	616
Original Sheet No. 447	617

Original Sheet No. 448	618
Original Sheet No. 449	619
Original Sheet No. 450	620
Original Sheet No. 451	621
Original Sheet No. 452	622
•	623
Original Sheet No. 453	
Original Sheet No. 454	624
Original Sheet No. 455	625
First Revised Sheet No. 456	626
Original Sheet No. 457	627
Original Sheet No. 458	628
Original Sheet No. 459	629
Original Sheet No. 460	630
Original Sheet No. 461	631
Original Sheet No. 462	632
Original Sheet No. 463	633
Original Sheet No. 464	634
Original Sheet No. 465	635
Original Sheet No. 466	636
Original Sheet No. 467	637
Original Sheet No. 468	638
Original Sheet No. 469	639
Original Sheet No. 470	640
Original Sheet No. 471	641
Original Sheet No. 472	642
Original Sheet No. 473	643
Original Sheet No. 474	644
Original Sheet No. 475	645
Original Sheet No. 476	646
First Revised Sheet No. 477	647
	648
Original Sheet No. 477	
Original Sheet No. 477A	649
Original Sheet No. 478	650
Original Sheet No. 479	651
Original Sheet No. 480	652
Original Sheet No. 481	653
Original Sheet No. 482	654
Original Sheet No. 483	655
Original Sheet No. 484	656
Original Sheet No. 485	657
Original Sheet No. 486	658
Original Sheet No. 487	659
Original Sheet No. 488	660
Original Sheet No. 489	661
· ·	662
Original Sheet No. 490	
First Revised Sheet No. 491	663

Sheet No. 492	665
Original Sheet No. 493	666
<u> </u>	667
Original Sheet No. 494	668
Original Sheet No. 495	
Original Sheet No. 496	669
Original Sheet No. 497	670
Original Sheet No. 498	671
Original Sheet No. 499	672
Original Sheet No. 500	673
Original Sheet No. 501	674
Original Sheet No. 502	675
Original Sheet No. 503	676
Original Sheet No. 504	677
Original Sheet No. 505	678
Original Sheet No. 506	679
Original Sheet No. 507	680
Original Sheet No. 508	681
Original Sheet No. 509	682
Original Sheet No. 510	683
Original Sheet No. 511	684
Original Sheet No. 512	685
Original Sheet No. 513	686
Original Sheet No. 514	687
Original Sheet No. 515	688
Original Sheet No. 516	689
Original Sheet No. 517	690
Original Sheet No. 518	691
Original Sheet No. 519	692
Sheet No. 520	693
Original Sheet No. 521	694
	695
Original Sheet No. 522	696
Original Sheet No. 523 Sheet No. 524	
	697
Original Sheet No. 525	698
Original Sheet No. 526	699
Sheet No. 527	700
Original Sheet No. 528	701
Sheet No. 529	702
First Revised Sheet No. 530	703
Second Revised Sheet No. 530	704
Second Revised Sheet No. 531	705
Third Revised Sheet No. 531	706
First Revised Sheet No. 532	707
Second Revised Sheet No. 532	708
Original Sheet No. 533	709
Original Sheet No. 534	710

October Object No. 505	744
Original Sheet No. 535	711
Sixth Revised Sheet No. 536	712
First Revised Sheet No. 537	713
Fourth Revised Sheet No. 538	714
Second Revised Sheet No. 539	715
Fourth Revised Sheet No. 539	716
Fifth Revised Sheet No. 540	717
First Revised Sheet No. 540A	718
First Revised Sheet No. 540B	719
Original Sheet No. 540D	720
Original Sheet No. 540E	721
Original Sheet No. 540F	722
Original Sheet No. 541	723
Sheet Nos. 542 - 599	724
Substitute First Revised Sheet No. 600	725
Substitute First Revised Sheet No. 601	727
Substitute First Revised Sheet No. 602	727 729
Substitute First Revised Sheet No. 603	730
Substitute First Revised Sheet No. 604	732
2nd Sub First Revised Sheet No. 605	733
Substitute First Revised Sheet No. 606	734
Substitute First Revised Sheet No. 607	735
Substitute Original Sheet No. 607A	736
Original Sheet No. 607B	738
Original Sheet No. 607C	739
Original Sheet No. 607D	741
Original Sheet No. 607E	742
Original Sheet No. 607F	743
Original Sheet No. 607G	744
Original Sheet No. 607H	745
Original Sheet No. 607I	746
Original Sheet No. 607J	747
Original Sheet No. 607K	748
Original Sheet No. 607L	749
Original Sheet No. 607M	750
Original Sheet No. 607N	752
Original Sheet No. 607O	754
Substitute First Revised Sheet No. 608	755
Substitute First Revised Sheet No. 609	757
Substitute First Revised Sheet No. 610	758
Substitute First Revised Sheet No. 611	759
Substitute Original Sheet No. 611A	761
Original Sheet No. 611B	762
Original Sheet No. 611C	763
Original Sheet No. 611D	764
Substitute First Revised Sheet No. 612	765

Substitute First Revised Sheet No. 613	767
Substitute First Revised Sheet No. 614	769
Substitute First Revised Sheet No. 615	770
Substitute First Revised Sheet No. 616	771
Substitute Original Sheet No. 616A	772
Original Sheet No. 616B	774
Original Sheet No. 616C	775
Original Sheet No. 616D	776
Original Sheet No. 616E	777
Original Sheet No. 616F	779
Original Sheet No. 616G	781
Original Sheet No. 616H	782
Original Sheet No. 616I	783
Original Sheet No. 616J	784
Original Sheet No. 617	785
Substitute First Revised Sheet No. 618	787
Substitute First Revised Sheet No. 619	789
Original Sheet No. 620	790
Original Sheet No. 621	792
Substitute First Revised Sheet No. 622	794
Substitute First Revised Sheet No. 623	796
Original Sheet No. 624	797
Sheet Nos. 625 - 699	799

FERC Docket: RP08-385-000

First Revised Sheet No. 0 Natural Gas Pipeline Company of America LLC: Seventh Revised Volume No. 1

First Revised Sheet No. 0

Superseding: Original Sheet No. 0

FERC GAS TARIFF

SEVENTH REVISED VOLUME NO. 1

(Supersedes Sixth Revised Volume No. 1)

of

NATURAL GAS PIPELINE COMPANY OF AMERICA LLC

Filed with the

FEDERAL ENERGY REGULATORY COMMISSION

Communications Concerning this Tariff Should be Addressed to:

Bruce H. Newsome, Vice President Natural Gas Pipeline Company of America LLC 3250 Lacey Road, 7th Floor Downers Grove, Illinois 60515-7918 Telephone: (630) 725-3070 Facsimile: (630) 725-3108

Original Sheet No. 1 original Sheet No. 1: Pending

Seventh Revised Volume No.																				eet 	No
Preliminary Statement																					6
System Map																				•	7
Currently Effective Rates Rate Schedule FTS Peak																					
Rate Schedule FTS-G Peak																					
Rate Schedule ITS Peak																					
Rate Schedule FRSS					•	•		•		•										.1	6
Rate Schedule IBS																				.1	7
Rate Schedule FFTS Peak																					
Gathering Rates			•																	. 2	2
Offshore Zone Charges																				. 2	3
Moraine Lateral Incremen	nta	1	Ch	ar	ge	•		•		•										. 2	4
Transportation Fuel Rete	ent	io	n	Fa	cto	ors	3.													. 2	5
Rate Schedule BESS Rate Schedule DSS Rate Schedule NSS Rate Schedule PALS Rate Schedule IPS		•	•		•	•	•	•	•	•	•	•	:	· ·	· ·	· ·	:	:	•	.2	7 7 9

Original Sheet No. 2 Original Sheet No. 2: Pending

Seventh Revised Volume No. 1										
Currently Effective Rates (Continued) Section 37 Lateral Line Charge										
Rate Schedule F-1		31								
Annual Charges Adjustment (ACA) Charge Section 49 Negotiated Rate		32								
Statement of Negotiated Rate Transactions		33								
Footnotes (Applicable to Sheet Nos. 8 through 25)		36								

Original Sheet No. 3 Original Sheet No. 3: Pending

Seventh Revised Volume No. 1										
Rate Scl										
FFTS	Flexible Firm Transportation Service									
FTS	Firm Transportation Service			•	65					
FRSS	Firm Reverse Storage Service									
FTS-G	Firm Transportation Service for Former G-1 Bu	yers.			111					
ITS	Interruptible Transportation Service				116					
IBS	Interruptible Balancing Service				128					
BESS	Best Efforts Storage Service				147					
DSS	Delivered Firm Storage Service									
NSS	Nominated Firm Storage Service				187					
PALS	Park and Loan Service				207					
LPS	Line Pack Service									
General	Terms and Conditions									
1.	Definitions				275					
2.	Reserved for Future Use				282					
3.	Reserved for Future Use									
4.	Reserved for Future Use									
5.	Priority of Service									
6.	Receipt Points									
7.	Delivery of Transportation Gas for the Account			•						
, •	of Shipper				337					
8.	Nomination/Reporting and Balancing									
9.	Pooling and Storage Points									
10.	Determination of Daily Receipts									
11.	Determination of Deliveries									
12.	Balancing Service and Overrun Charges									
13.										
14.	Imbalances									
	Unauthorized Delivery of Gas	• •		•	391					
15.	Statements, Billing, Payment, Penalties and				204					
1.6	Discounting Policy									
16.	Evaluation of Credit									
17.	Interactive Website									
18.	Reserved for Future Use									
19.	Capacity Release by Firm Shippers									
20.	Advertisement and Marketing Fees									
21.	Tracking of Certain Account No. 858 Costs				444					
22.	Pre-Granted Abandonment, Contract Rollovers and									
	Right of First Refusal				450					
23.	Operational Control									
24.	Method of Measurement and Measurement Equipment.									
25.	Pressure and Delivery Conditions									
26.	Quality									

Original Sheet No. 4 Original Sheet No. 4 : Pending

	Revised Volume No. 1	eet No
General	Terms and Conditions (contd.)	
27.	Force Majeure	479
28.	Possession of Gas, Title and Responsibility	
29.	Notification	
30.	Facilities/Obligation to Carry Out Agreement/Filings .	
31.	Indemnification	
32.	Successors and Assigns	
33.	Regulation	
34.	Sales by Natural's Merchant Division	
35.	Shipper Complaint Procedure	
36.	Reserved for Future Use	
37.	Lateral Line Charge	
38.	Recovery of Order No. 636 Gas Supply Realignment Costs	493
30.	and Upstream Supplier Transition Costs	101
39.	Reserved for Future Use	
40.		
40.	Annual Charges Adjustment Charge	
41.	Research, Development and Demonstration Adjustment Reserved For Future Use	
42.		
	Reports	
44.	Waivers	
45.	Reserved for Future Use	
46.	Gathering Affiliates	
47.	Reserved for Future Use	
48.	Compliance with 18 C.F.R., Section 284.10	
49.	Negotiated Rates	
50.	Non-Conforming Agreements	
51.	Acquired Capacity	 541
_		
	f Service Agreement	
	Schedules FTS, FFTS, FTS-G and ITS	
	Schedule IBS	
	Schedules DSS, NSS, FRSS and BESS	
	Schedule PALS	
Rate :	Schedule LPS	 621

FERC Docket: RP08-319-000

Original Sheet No. 5 Original Sheet No. 5: Pending

	me No. 2 Schedule 		Sheet No.
X-25 X-26 X-32 X-38	Exchange Agreement Exchange Agreement Exchange Agreement Exchange Agreement Transportation Agreement Lease of Gas Properties	- Phillips Petroleum - Phillips - Michigan Wisconsin & Phillips - Phillips Petroleum - Roger C. Hanks - Stingray Pipeline	2 8 17 26 29 32

FERC Docket: RP08-319-000

Original Sheet No. 6 original Sheet No. 6: Pending

PRELIMINARY STATEMENT

Natural Gas Pipeline Company of America LLC (Natural) is a natural gas transmission company primarily engaged in the transportation and storage of gas in interstate commerce. transmission system consists of the Amarillo Line and the Gulf Coast Line, as well as the A/G Line connecting the two mainlines. The Amarillo Line extends from the gas producing areas in north central Texas, southwest Texas, southeast Oklahoma, Kansas, Nebraska, Iowa and Illinois, terminating at points in and near the Chicago metropolitan area. The Gulf Coast Line extends from the offshore and onshore gas producing areas of south Louisiana and the Gulf Coast of Texas through the states of Louisiana, Texas, Arkansas, Missouri and Illinois, terminating at points in and near the Chicago metropolitan area which are in common with the terminal points of the Amarillo Line. The A/G Line, an interconnection between the Amarillo Line and the Gulf Coast Line, is located in Carter, Murray, Johnston and Bryan Counties, Oklahoma, and Lamar, Red River, Franklin, Titus, Morris and Cass Counties, Texas. Natural also operates a total of thirteen storage reservoirs located in the states of Illinois, Iowa, Oklahoma and Texas.

The Currently Effective Rates, Rate Schedules, General Terms and Conditions, and Forms of Service Agreement applicable to the transportation and storage services performed by Natural are contained herein.

Original Sheet No. 7 original Sheet No. 7: Pending

SYSTEM MAP

FERC Docket: RP10-147-004

First Revised Sheet No. 8 First Revised Sheet No. 8

Superseding: Original Sheet No. 8

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS

PEAK PERIOD MAXIMUM RATES 3/, 4/, 5/

	DELIVERY ZONE 2/								
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas				
Iowa-Illinois Reservation 8.7900	\$ 4.1600	\$ 7.2800	\$ 9.2100	\$10.7200	\$10.1900	\$			
Commodity 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$			
Amarillo Reservation 7.1100	\$ 6.4000	\$ 4.3200	\$11.2200	\$10.5100	\$11.1300	\$			
Commodity 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$			
Gulf Coast Mainline Reservation 8.1600	\$ 6.5900	\$ 7.0100	\$ 5.0200	\$ 5.9900	\$ 5.7100	\$			
8.1600 Commodity 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$			
Midcontinent Reservation 5.7800	\$ 9.6800	\$ 4.2700	\$ 6.4700	\$ 7.4400	\$ 7.8700	\$			
Commodity 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$			
Texok Reservation 7.9100	\$10.4000	\$ 6.1400	\$ 3.5500	\$ 5.9500	\$ 6.9600	\$			
Commodity 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$			
Louisiana Reservation	\$11.6800	\$ 8.0700	\$ 3.8600	\$ 3.4900	\$ 4.0200	\$			
9.5800 Commodity 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$			
South Texas Reservation	\$12.0600	\$ 8.8100	\$ 4.6900	\$ 5.7300	\$ 3.6800				
\$10.3100 Commodity 0.0049	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$			
Permian Reservation	\$13.6500	\$ 4.7800	\$ 8.8900	\$10.2700	\$11.0900	\$			
3.2500 Commodity 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$			

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be

increased

by \$0.1261.

determined

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be

in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently

and

effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of peak rates, see Footnote 1/.

FERC Docket: RP10-147-004

Original Sheet No. 8A Original Sheet No. 8A

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS

PEAK PERIOD MAXIMUM RATES 3/, 4/, 5/

			DELIVERY	ZONE 2/		
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Reservation	\$ 4.0800	\$ 7.1300	\$ 9.0200	\$10.5000	\$ 9.9800	\$
8.6100 Commodity 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Reservation	\$ 6.2700	\$ 4.2300	\$10.9900	\$10.3000	\$10.9000	\$
6.9600 Commodity 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Reservation	\$ 6.4500	\$ 6.8700	\$ 4.9200	\$ 5.8700	\$ 5.6000	\$
7.9900 Commodity 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Reservation 5.6600	\$ 9.4800	\$ 4.1800	\$ 6.3400	\$ 7.2900	\$ 7.7000	\$
Commodity 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Reservation 7.7400	\$10.1800	\$ 6.0100	\$ 3.4800	\$ 5.8200	\$ 6.8200	\$
Commodity 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Reservation 9.3900	\$11.4400	\$ 7.9000	\$ 3.7800	\$ 3.4200	\$ 3.9300	\$
Commodity 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Reservation	\$11.8100	\$ 8.6300	\$ 4.6000	\$ 5.6100	\$ 3.6000	
\$10.1000 Commodity 0.0049	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Reservation	\$13.3700	\$ 4.6800	\$ 8.7100	\$10.0600	\$10.8600	\$
3.1800 Commodity 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be increased

by \$0.1235.

and

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be determined

in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of peak rates, see Footnote 1/.

FERC Docket: RP10-147-004

Original Sheet No. 8B Original Sheet No. 8B

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS

PEAK PERIOD MAXIMUM RATES 3/, 4/, 5/

RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Reservation 8.3400	\$ 3.9500	\$ 6.9000	\$ 8.7300	\$10.1700	\$ 9.6600	\$
Commodity 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Reservation	\$ 6.0700	\$ 4.0900	\$10.6400	\$ 9.9700	\$10.5500	\$
6.7400 Commodity 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Reservation	\$ 6.2500	\$ 6.6500	\$ 4.7700	\$ 5.6900	\$ 5.4200	\$
7.7400 Commodity 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Reservation	\$ 9.1800	\$ 4.0500	\$ 6.1400	\$ 7.0600	\$ 7.4600	\$
5.4800 Commodity 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Reservation	\$ 9.8600	\$ 5.8200	\$ 3.3700	\$ 5.6400	\$ 6.6100	\$
7.5000 Commodity 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Reservation	\$11.0800	\$ 7.6500	\$ 3.6600	\$ 3.3100	\$ 3.8100	\$
9.0900 Commodity 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Reservation	\$11.4400	\$ 8.3500	\$ 4.4500	\$ 5.4400	\$ 3.4900	\$
9.7800 Commodity 0.0049	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Reservation	\$12.9400	\$ 4.5400	\$ 8.4400	\$ 9.7400	\$10.5200	\$
3.0800 Commodity 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be increased

by \$0.1196.

and

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be determined

in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of peak rates, see Footnote 1/.

FERC Docket: RP08-319-000

Original Sheet No. 9 original Sheet No. 9: Pending

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS

PEAK PERIOD MINIMUM RATES 3/, 4/, 5/

RECEIPT ZONE 2/ Permian			DELIVERY	ZONE 2/							
	Market	Midcontinent	Texok	Louisiana	South Texas						
Iowa-Illinois Reservation 0.0000			\$ 0.0000	\$ 0.0000		\$					
Commodity 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	Ş					
Amarillo Reservation 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$					
Commodity 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$					
Gulf Coast Mainline Reservation 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$					
Commodity 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$					
Midcontinent Reservation 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$					
Commodity 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$					
Texok Reservation	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$					
Commodity 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$					
Louisiana Reservation 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$					
Commodity 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$					
South Texas Reservation	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$					
0.0000 Commodity 0.0049	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$					
Permian Reservation	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$ 0.0000	\$					
0.0000 Commodity 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$					

NOTES: The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

and

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of peak rates, see Footnote 1/.

FERC Docket: RP10-147-004

First Revised Sheet No. 10 First Revised Sheet No. 10 Superseding: Original Sheet No. 10

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS

OFF-PEAK PERIOD MAXIMUM RATES 3/, 4/, 5/

	O F. F. – F			AXIMUM	RATES	3/, 4/, 5/		
		DELIVERY ZONE 2/						
RECEIPT ZONE 2/	Market	Midcontinent	Texok	Louisiana	South Texas			
Iowa-Illinois Reservation 8.7900	\$ 3.9000	\$ 7.2800	\$ 9.2100	\$10.7200	\$10.1900	\$		
Commodity 0.0039	\$ 0.0005	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$		
Amarillo Reservation 7.1100	\$ 5.7300	\$ 4.3200	\$11.2200	\$10.5100	\$11.1300	\$		
Commodity 0.0028	\$ 0.0020	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$		
Gulf Coast Mainline Reservation 8.1600	\$ 5.7800	\$ 7.0100	\$ 5.0200	\$ 5.9900	\$ 5.7100	\$		
Commodity 0.0035	\$ 0.0023	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$		
Midcontinent Reservation 5.7800	\$ 8.1500	\$ 4.2700	\$ 6.4700	\$ 7.4400	\$ 7.8700	\$		
Commodity 0.0019	\$ 0.0034	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$		
Texok Reservation 7.9100	\$ 8.7200	\$ 6.1400	\$ 3.5500	\$ 5.9500	\$ 6.9600	\$		
Commodity 0.0033	\$ 0.0037	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$		
Louisiana Reservation 9.5800	\$ 9.7200	\$ 8.0700	\$ 3.8600	\$ 3.4900	\$ 4.0200	\$		
Commodity	\$ 0.0045	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$		
South Texas Reservation	\$10.0300	\$ 8.8100	\$ 4.6900	\$ 5.7300	\$ 3.6800			
\$10.3100 Commodity 0.0049	\$ 0.0048	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$		
Permian Reservation	\$11.5800	\$ 4.7800	\$ 8.8900	\$10.2700	\$11.0900	\$		
3.2500 Commodity 0.0002	\$ 0.0048	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$		

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be increased by \$0.1261.

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be determined in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS.

and

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of off-peak rates, see Footnote 1/.

FERC Docket: RP10-147-004

Original Sheet No. 10A Original Sheet No. 10A

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS

OFF-PEAK PERIOD MAXIMUM RATES 3/, 4/, 5/

		DELIVERY ZONE 2/				
RECEIPT ZONE 2/ Permian		Midcontinent			South Texas	
Iowa-Illinois Reservation 3.6100	\$ 3.8200	\$ 7.1300	\$ 9.0200	\$10.5000	\$ 9.9800	
Commodity	\$ 0.0005	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	
Amarillo Reservation 6.9600	\$ 5.6100	\$ 4.2300	\$10.9900	\$10.3000	\$10.9000	
Commodity	\$ 0.0020	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	
Gulf Coast Mainline Reservation 7.9900 Commodity 0.0035	\$ 5.6600	\$ 6.8700	\$ 4.9200	\$ 5.8700	\$ 5.6000	
	\$ 0.0023	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	
Midcontinent Reservation	\$ 7.9800	\$ 4.1800	\$ 6.3400	\$ 7.2900	\$ 7.7000	
5.6600 Commodity 0.0019	\$ 0.0034	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	
Texok Reservation 7.7400 Commodity 0.0033	\$ 8.5400	\$ 6.0100	\$ 3.4800	\$ 5.8200	\$ 6.8200	
	\$ 0.0037	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	
Louisiana Reservation 9.3900 Commodity 0.0044	\$ 9.5200	\$ 7.9000	\$ 3.7800	\$ 3.4200	\$ 3.9300	
	\$ 0.0045	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	
South Texas Reservation	\$ 9.8200	\$ 8.6300	\$ 4.6000	\$ 5.6100	\$ 3.6000	
10.1000 Commodity 0.0049	\$ 0.0048	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	
Permian Reservation	\$11.3400	\$ 4.6800	\$ 8.7100	\$10.0600	\$10.8600	
.1800 Commodity .0002	\$ 0.0048	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be

increased

by \$0.1235.

determined

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be !

in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS.

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section

and

12

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of off-peak rates, see Footnote 1/.

FERC Docket: RP10-147-004

Original Sheet No. 10B Original Sheet No. 10B

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS

OFF-PEAK PERIOD MAXIMUM RATES 3/, 4/, 5/

			DELIVERY	ZONE 2/		
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Reservation	\$ 3.7000	\$ 6.9000	\$ 8.7300	\$10.1700	\$ 9.6600	
8.3400 Commodity 0.0039	\$ 0.0005	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	
Amarillo Reservation	\$ 5.4400	\$ 4.0900	\$10.6400	\$ 9.9700	\$10.5500	
6.7400 Commodity 0.0028	\$ 0.0020	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	
Gulf Coast Mainline Reservation	\$ 5.4800	\$ 6.6500	\$ 4.7700	\$ 5.6900	\$ 5.4200	
7.7400 Commodity 0.0035	\$ 0.0023	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	
Midcontinent Reservation	\$ 7.7300	\$ 4.0500	\$ 6.1400	\$ 7.0600	\$ 7.4600	5
5.4800 Commodity).0019	\$ 0.0034	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	5
Texok Reservation	\$ 8.2700	\$ 5.8200	\$ 3.3700	\$ 5.6400	\$ 6.6100	Ş
7.5000 Commodity 0.0033	\$ 0.0037	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	Ş
Louisiana Reservation	\$ 9.2200	\$ 7.6500	\$ 3.6600	\$ 3.3100	\$ 3.8100	5
0.0900 Commodity 0.0044	\$ 0.0045	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	
South Texas Reservation	\$ 9.5100	\$ 8.3500	\$ 4.4500	\$ 5.4400	\$ 3.4900	5
9.7800 Commodity).0049	\$ 0.0048	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	5
Permian Reservation	\$10.9800	\$ 4.5400	\$ 8.4400	\$ 9.7400	\$10.5200	Ş
3.0800 Commodity 0.0002	\$ 0.0048	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be

increased

by \$0.1196.

determined

and

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be

in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

FERC Docket: RP08-319-000

Original Sheet No. 11 original Sheet No. 11: Pending

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS

OFF-PEAK PERIOD MINIMUM RATES 3/, 4/, 5/

DELIVERY ZONE 2/ ______ RECEIPT ZONE 2/ Market Midcontinent Louisiana South Texas Texok Permian -------------------------Iowa-Illinois \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Ś Reservation 0.0000 Commodity \$ 0.0005 \$ 0.0029 \$ 0.0042 \$ 0.0052 \$ 0.0049 \$ 0.0039 Amarillo Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Commodity \$ 0.0020 \$ 0.0010 \$ 0.0056 \$ 0.0054 \$ 0.0055 \$ 0.0028 Gulf Coast Mainline \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ Reservation 0.0000 \$ 0.0014 Commodity \$ 0.0023 \$ 0.0027 \$ 0.0020 \$ 0.0018 \$ 0.0035 Midcontinent \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Reservation \$ 0.0000 Ś 0.0000 Commodity \$ 0.0034 \$ 0.0010 \$ 0.0023 \$ 0.0030 \$ 0.0037 \$ 0.0019 Texok Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Commodity \$ 0.0037 \$ 0.0021 \$ 0.0002 \$ 0.0015 \$ 0.0031 0.0033 Louisiana \$ 0.0000 \$ 0.0000 \$ 0.0000 Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 Commodity \$ 0.0045 \$ 0.0034 \$ 0.0006 \$ 0.0003 \$ 0.0007 Ś 0.0044 South Texas \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Ś Reservation 0.0000 Commodity \$ 0.0048 \$ 0.0039 \$ 0.0011 \$ 0.0018 \$ 0.0006 \$ 0.0049 Permian Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0012 \$ 0.0040 \$ 0.0049 Commodity \$ 0.0048 \$ 0.0055 \$ 0.0002

NOTES: The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS

and

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

FERC Docket: RP10-147-004

First Revised Sheet No. 12 First Revised Sheet No. 12 Superseding: Original Sheet No. 12

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS-G

PEAK PERIOD COMMODITY RATES 3/, 4/

			DELIVERY	ZONE 2/		
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Maximum	\$ 0.2776	\$ 0.4812	\$ 0.6094	\$ 0.7098	\$ 0.6745	\$
0.5816 Minimum 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Maximum 0.4702	\$ 0.4313	\$ 0.2848	\$ 0.7434	\$ 0.6966	\$ 0.7369	\$
Minimum 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Maximum 0.5398	\$ 0.4451	\$ 0.4638	\$ 0.3317	\$ 0.3961	\$ 0.3774	\$
Minimum 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Maximum 0.3820	\$ 0.6553	\$ 0.2816	\$ 0.4276	\$ 0.4921	\$ 0.5209	\$
Minimum 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Maximum 0.5230	\$ 0.7041	\$ 0.4058	\$ 0.2336	\$ 0.3924	\$ 0.4610	\$
Minimum 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Maximum 0.6344	\$ 0.7922	\$ 0.5340	\$ 0.2544	\$ 0.2299	\$ 0.2647	\$
Minimum 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Maximum 0.6827	\$ 0.8180	\$ 0.5829	\$ 0.3098	\$ 0.3787	\$ 0.2423	\$
0.0049 Minimum	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Maximum	\$ 0.9238	\$ 0.3156	\$ 0.5888	\$ 0.6802	\$ 0.7343	\$
0.2139 Minimum 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Commodity Rate shall be increased

by \$0.0082.

For Shippers electing the SW option, the applicable maximum Commodity Rate shall be determined in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS at a 50% load factor.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS-G and the General Terms and Conditions will be added or deducted on the monthly bills of

affected

12

Shippers when applicable.

FERC Docket: RP10-147-004

Original Sheet No. 12A Original Sheet No. 12A

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS-G

PEAK PERIOD COMMODITY RATES 3/, 4/

			DELIVERY	ZONE 2/		
RECEIPT ZONE 2/	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Maximum	\$ 0.2719	\$ 0.4712	\$ 0.5968	\$ 0.6952	\$ 0.6605	Ś
0.5696 Minimum	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
0.0039 Amarillo						
Maximum 0.4605	\$ 0.4225	\$ 0.2789	\$ 0.7280	\$ 0.6823	\$ 0.7217	\$
Minimum 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Maximum	\$ 0.4359	\$ 0.4542	\$ 0.3249	\$ 0.3879	\$ 0.3696	\$
0.5287 Minimum 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Maximum	\$ 0.6418	\$ 0.2758	\$ 0.4188	\$ 0.4820	\$ 0.5101	Ś
0.3741 Minimum 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok						
Maximum 0.5122	\$ 0.6895	\$ 0.3974	\$ 0.2288	\$ 0.3843	\$ 0.4514	\$
Minimum 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Maximum	\$ 0.7758	\$ 0.5229	\$ 0.2492	\$ 0.2252	\$ 0.2593	\$
0.6213 Minimum 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Maximum	\$ 0.8011	\$ 0.5709	\$ 0.3034	\$ 0.3709	\$ 0.2373	\$
0.6687 Minimum	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
0.0049						
Maximum 0.2095	\$ 0.9048	\$ 0.3091	\$ 0.5766	\$ 0.6662	\$ 0.7192	\$
Minimum 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Commodity Rate shall be increased by \$0.0081.

For Shippers electing the SW option, the applicable maximum Commodity Rate shall be determined in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS at a 50% load factor.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS-G and the General Terms and Conditions will be added or deducted on the monthly bills of

affected

12

Shippers when applicable.

FERC Docket: RP10-147-004

Original Sheet No. 12B Original Sheet No. 12B

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS-G

PEAK PERIOD COMMODITY RATES 3/, 4/

			DELIVERY	ZONE 2/		
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Maximum 0.5517	\$ 0.2633	\$ 0.4564	\$ 0.5779	\$ 0.6732	\$ 0.6397	\$
Minimum 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Maximum 0.4460	\$ 0.4091	\$ 0.2701	\$ 0.7051	\$ 0.6607	\$ 0.6989	\$
Minimum 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Maximum 0.5120	\$ 0.4222	\$ 0.4398	\$ 0.3146	\$ 0.3757	\$ 0.3580	\$
Minimum 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Maximum 0.3623	\$ 0.6215	\$ 0.2671	\$ 0.4056	\$ 0.4667	\$ 0.4940	\$
Minimum 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Maximum	\$ 0.6678	\$ 0.3849	\$ 0.2216	\$ 0.3722	\$ 0.4372	\$
0.4961 Minimum 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Maximum	\$ 0.7513	\$ 0.5064	\$ 0.2413	\$ 0.2181	\$ 0.2511	\$
0.6017 Minimum 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Maximum	\$ 0.7758	\$ 0.5529	\$ 0.2938	\$ 0.3592	\$ 0.2298	\$
0.6476 Minimum 0.0049	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Maximum	\$ 0.8762	\$ 0.2993	\$ 0.5584	\$ 0.6451	\$ 0.6965	\$
0.2028 Minimum 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Commodity Rate shall be increased by \$0.0078.

For Shippers electing the SW option, the applicable maximum Commodity Rate shall be determined in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS at a 50% load factor.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS-G and the General Terms and Conditions will be added or deducted on the monthly bills of

affected

12

Shippers when applicable.

FERC Docket: RP10-147-004

First Revised Sheet No. 13 First Revised Sheet No. 13 Superseding: Original Sheet No. 13 $\,$

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS-G

OFF-PEAK PERIOD COMMODITY RATES 3/, 4/

			DELIVERY	ZONE 2/		
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Maximum	\$ 0.2556	\$ 0.4812	\$ 0.6094	\$ 0.7098	\$ 0.6745	\$
0.5816 Minimum 0.0039	\$ 0.0005	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Maximum 0.4702 Minimum	\$ 0.3770 \$ 0.0020	\$ 0.2848 \$ 0.0010	\$ 0.7434 \$ 0.0056	\$ 0.6966 \$ 0.0054	\$ 0.7369 \$ 0.0055	\$
0.0028	¥ 0.0020	7 0.0010	7 0.0000	y 0.0031	¥ 0.0000	٣
Gulf Coast Mainline Maximum 0.5398	\$ 0.3804	\$ 0.4638	\$ 0.3317	\$ 0.3961	\$ 0.3774	\$
Minimum 0.0035	\$ 0.0023	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Maximum 0.3820	\$ 0.5363	\$ 0.2816	\$ 0.4276	\$ 0.4921	\$ 0.5209	\$
Minimum 0.0019	\$ 0.0034	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Maximum 0.5230	\$ 0.5741	\$ 0.4058	\$ 0.2336	\$ 0.3924	\$ 0.4610	\$
Minimum 0.0033	\$ 0.0037	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Maximum 0.6344	\$ 0.6402	\$ 0.5340	\$ 0.2544	\$ 0.2299	\$ 0.2647	\$
Minimum 0.0044	\$ 0.0045	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Maximum 0.6827	\$ 0.6608	\$ 0.5829	\$ 0.3098	\$ 0.3787	\$ 0.2423	\$
0.0827 Minimum 0.0049	\$ 0.0048	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Maximum	\$ 0.7623	\$ 0.3156	\$ 0.5888	\$ 0.6802	\$ 0.7343	\$
0.2139 Minimum 0.0002	\$ 0.0048	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Commodity Rate shall be increased

by \$0.0082.

For Shippers electing the SW option, the applicable maximum Commodity Rate shall be determined in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS at a 50% load factor.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS-G and the General Terms and Conditions will be added or deducted on the monthly bills of

affected

12

Shippers when applicable.

FERC Docket: RP10-147-004

Original Sheet No. 13A Original Sheet No. 13A

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS-G

DELIVERY ZONE 2/

OFF-PEAK PERIOD COMMODITY RATES 3/, 4/

RECEIPT ZONE 2/ Market Midcontinent Permian Iowa-Illinois Maximum \$ 0.2503 \$ 0.4712 0.5696 Minimum \$ 0.0005 \$ 0.0029 0.0039	Texok \$ 0.5968 \$ 0.0042 \$ 0.7280 \$ 0.0056	\$ 0.6952 \$ 0.0052 \$ 0.6823 \$ 0.0054	\$ 0.6605 \$ 0.0049 \$ 0.7217 \$ 0.0055	\$
Iowa-Illinois Maximum \$ 0.2503 \$ 0.4712 0.5696 Minimum \$ 0.0005 \$ 0.0029	\$ 0.5968 \$ 0.0042 \$ 0.7280 \$ 0.0056	\$ 0.6952 \$ 0.0052 \$ 0.6823	\$ 0.6605 \$ 0.0049 \$ 0.7217	\$
Maximum \$ 0.2503 \$ 0.4712 .5696 Minimum \$ 0.0005 \$ 0.0029	\$ 0.0042 \$ 0.7280 \$ 0.0056	\$ 0.0052	\$ 0.0049	\$
	\$ 0.7280 \$ 0.0056	\$ 0.6823	\$ 0.7217	•
	\$ 0.0056			Ş
Amarillo Maximum \$ 0.3692 \$ 0.2789	,	\$ 0.0054	\$ 0.0055	
Minimum \$ 0.0020 \$ 0.0010				Ş
Gulf Coast Mainline	\$ 0.3249	\$ 0.3879	\$ 0.3696	Ş
Minimum \$ 0.0023 \$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	Ş
Midcontinent	\$ 0.4188 \$ 0.0023	\$ 0.4820 \$ 0.0030	\$ 0.5101 \$ 0.0037	Ş
.0019	+ 0.0020	4 0.0000	4 0.0007	7
Texok Maximum \$ 0.5622 \$ 0.3974 5122	\$ 0.2288	\$ 0.3843	\$ 0.4514	Ş
Minimum \$ 0.0037 \$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	Ş
Louisiana Maximum \$ 0.6270 \$ 0.5229 6213	\$ 0.2492	\$ 0.2252	\$ 0.2593	Ş
Minimum \$ 0.0045 \$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	Ş
South Texas Maximum \$ 0.6472 \$ 0.5709 6687	\$ 0.3034	\$ 0.3709	\$ 0.2373	Ş
Minimum \$ 0.0048 \$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	Ş
Permian				
Maximum \$ 0.7466 \$ 0.3091 2095	\$ 0.5766	\$ 0.6662	\$ 0.7192	Ş
Minimum \$ 0.0048 \$ 0.0012 .0002	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Commodity Rate shall be increased by \$0.0081.

For Shippers electing the SW option, the applicable maximum Commodity Rate shall be determined in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS at a 50% load factor.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS-G and the General Terms and Conditions will be added or deducted on the monthly bills of

affected

12

Shippers when applicable.

FERC Docket: RP10-147-004

Original Sheet No. 13B Original Sheet No. 13B

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FTS-G

OFF-PEAK PERIOD COMMODITY RATES 3/, 4/

		DELIVERY ZONE 2			2/		
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas		
Iowa-Illinois Maximum 0.5517	\$ 0.2424	\$ 0.4564	\$ 0.5779	\$ 0.6732	\$ 0.6397	Ş	
Minimum 0.0039	\$ 0.0005	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	Ş	
Amarillo Maximum	\$ 0.3575	\$ 0.2701	\$ 0.7051	\$ 0.6607	\$ 0.6989	Ş	
0.4460 Minimum 0.0028	\$ 0.0020	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	Ş	
Gulf Coast Mainline Maximum).5120	\$ 0.3608	\$ 0.4398	\$ 0.3146	\$ 0.3757	\$ 0.3580	\$	
Minimum 0.0035	\$ 0.0023	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	Ş	
Midcontinent Maximum	\$ 0.5087	\$ 0.2671	\$ 0.4056	\$ 0.4667	\$ 0.4940	Ş	
0.3623 Minimum 0.0019	\$ 0.0034	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	Ş	
Texok Maximum	\$ 0.5445	\$ 0.3849	\$ 0.2216	\$ 0.3722	\$ 0.4372	\$	
0.4961 Minimum 0.0033	\$ 0.0037	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	Ş	
Louisiana Maximum	\$ 0.6072	\$ 0.5064	\$ 0.2413	\$ 0.2181	\$ 0.2511	Ş	
0.6017 Minimum 0.0044	\$ 0.0045	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	Ş	
South Texas Maximum	\$ 0.6267	\$ 0.5529	\$ 0.2938	\$ 0.3592	\$ 0.2298	Ş	
).6476 Minimum).0049	\$ 0.0048	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$	
Permian Maximum	\$ 0.7230	\$ 0.2993	\$ 0.5584	\$ 0.6451	\$ 0.6965	\$	
).2028 Minimum).0002	\$ 0.0048	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$	

NOTES: For Shippers electing the LN option, the applicable maximum Commodity Rate shall be increased by \$0.0078.

For Shippers electing the SW option, the applicable maximum Commodity Rate shall be determined in accordance with the formula set out in Section 6.2(a)(3) of Rate Schedule FTS at a 50% load factor.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FTS-G and the General Terms and Conditions will be added or deducted on the monthly bills of

affected

12

Shippers when applicable.

FERC Docket: RP08-319-000

Louisiana

South Texas

0.0044

0.3544

0.0049

0.1103

0.0002

Permian

Maximum

Minimum

Maximum

Minimum

Maximum

Minimum

Original Sheet No. 14 Original Sheet No. 14: Pending

\$ 0.0193

\$ 0.4317

\$ 0.0201

\$ 0.4865

\$ 0.0206

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE ITS

PEAK PERIOD COMMODITY AND AOR RATES 3/, 4/

DELIVERY ZONE 2/

RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois						
Maximum 0.3018	\$ 0.1442	\$ 0.2495	\$ 0.3162	\$ 0.3685	\$ 0.3501	\$
Minimum 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Maximum	\$ 0.2261	\$ 0.1473	\$ 0.3860	\$ 0.3618	\$ 0.3826	\$
0.2438 Minimum 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Maximum	\$ 0.2340	\$ 0.2404	\$ 0.1717	\$ 0.2052	\$ 0.1954	\$
0.2800		,	,	,		
Minimum 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent						
Maximum 0.1978	\$ 0.3451	\$ 0.1457	\$ 0.2216	\$ 0.2552	\$ 0.2703	\$
Minimum 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok						
Maximum 0.2712	\$ 0.3709	\$ 0.2102	\$ 0.1205	\$ 0.2030	\$ 0.2392	\$
Minimum 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$

\$ 0.0034

\$ 0.0039

\$ 0.1633

\$ 0.0012

NOTES: The Balancing Service Charges are set out in Section 12 of the General Terms and Conditions. The Unauthorized Overrun Charge is set out in Section 8 of the Rate Schedule.

\$ 0.3024 \$ 0.1602

\$ 0.4180 \$ 0.2769 \$ 0.1314 \$ 0.1187 \$ 0.1368 \$

\$ 0.0003

\$ 0.1961

\$ 0.0018

\$ 0.0049

\$ 0.0007

\$ 0.1252 \$

\$ 0.3813 \$

\$ 0.0055 \$

\$ 0.0006

Ś

\$ 0.0006

\$ 0.0011

\$ 0.0040

\$ 0.3055 \$ 0.3531

Pursuant to Sections 21 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule ITS

and

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

FERC Docket: RP08-319-000

Original Sheet No. 15 original Sheet No. 15: Pending

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE ITS

OFF-PEAK PERIOD COMMODITY AND AOR RATES 3/, 4/

DELIVERY ZONE 2/ ______ RECEIPT ZONE 2/ Market Midcontinent Louisiana South Texas Texok Permian -------------------------Iowa-Illinois \$ 0.2495 \$ 0.3162 \$ 0.3685 \$ 0.1320 \$ 0.3501 Ś Maximum 0.3018 Minimum \$ 0.0005 \$ 0.0029 \$ 0.0042 \$ 0.0052 \$ 0.0049 \$ 0.0039 Amarillo \$ 0.1953 \$ 0.1473 \$ 0.3860 \$ 0.3618 \$ 0.3826 \$ Maximum 0.2438 \$ 0.0020 \$ 0.0010 \$ 0.0056 \$ 0.0054 \$ 0.0055 \$ Minimum 0.0028 Gulf Coast Mainline \$ 0.1973 \$ 0.2404 \$ 0.1717 \$ 0.2052 \$ 0.1954 \$ Maximum 0.2800 \$ 0.0023 \$ 0.0014 \$ 0.0020 Minimum \$ 0.0027 \$ 0.0018 \$ 0.0035 Midcontinent \$ 0.2782 \$ 0.1457 \$ 0.2216 \$ 0.2552 Maximum \$ 0.2703 Ś 0.1978 Minimum \$ 0.0034 \$ 0.0010 \$ 0.0023 \$ 0.0030 \$ 0.0037 \$ 0.0019 Texok Maximum \$ 0.2978 \$ 0.2102 \$ 0.1205 \$ 0.2030 \$ 0.2392 \$ Minimum \$ 0.0037 \$ 0.0021 \$ 0.0002 \$ 0.0015 \$ 0.0031 0.0033 Louisiana \$ 0.3323 \$ 0.2769 \$ 0.1314 \$ 0.1187 Maximum \$ 0.1368 \$ 0.3292 \$ 0.0034 Minimum \$ 0.0045 \$ 0.0006 \$ 0.0003 \$ 0.0007 Ś 0.0044 South Texas \$ 0.3430 \$ 0.3024 \$ 0.1602 \$ 0.1961 \$ 0.1252 Ś Maximum 0.3544 Minimum \$ 0.0048 \$ 0.0039 \$ 0.0011 \$ 0.0018 \$ 0.0006 \$ 0.0049 Permian Maximum \$ 0.3954 \$ 0.1633 \$ 0.3055 \$ 0.3531 \$ 0.3813 Ś 0.1103 \$ 0.0048 \$ 0.0012 \$ 0.0040 \$ 0.0049 Minimum \$ 0.0055 0.0002

NOTES: The Balancing Service Charges are set out in Section 12 of the General Terms and Conditions.

The Unauthorized Overrun Charge is set out in Section 8 of the Rate Schedule.

Pursuant to Sections 3, 21 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule ITS

and

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

FERC Docket: RP10-147-004

First Revised Sheet No. 16 First Revised Sheet No. 16

Superseding: Original Sheet No. 16

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FRSS

	Rates	per Dth
	Maximum 	Minimum
Delivered Service Reservation Rate 10-Day Service 20-Day Service	\$3.4400 \$3.7100	\$0.0000 \$0.0000
Authorized Overrun 10-Day Service 20-Day Service	Rate \$0.1131 \$0.1220	\$0.0000 \$0.0000
Storage-Only Option Reservation Rate 10-Day Service 20-Day Service	\$1.7800 \$2.0500	\$0.0000 \$0.0000
Authorized Overrun 10-Day Service 20-Day Service	Rate \$0.0585 \$0.0674	\$0.0000 \$0.0000

NOTE: The Unauthorized Overrun Charge is set out in Section 9 of the Rate Schedule.

Pursuant to Section 6.3(c) of Rate Schedule FRSS, Natural shall retain a Storage Fuel Charge of up to 1.19% of volumes tendered for injection. In addition, any applicable Transportation Fuel Charge will be assessed under Section 6.3(b) of Rate Schedule FRSS. The Fuel Retention Factors will apply unless Natural and Shipper mutually agree on monetary reimbursement.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rates.

FERC Docket: RP10-147-004

Original Sheet No. 16A Original Sheet No. 16A

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FRSS

	Rates per	Dth
	Maximum	Minimum
Delivered Service Reservation Rate		
10-Day Service	\$3.3400	\$0.0000
20-Day Service	\$3.6000	\$0.0000
Authorized Overrun		
10-Day Service	\$0.1131	\$0.0000
20-Day Service	\$0.1220	\$0.0000
Storage-Only Option		
Reservation Rate		
10-Day Service	\$1.7300	\$0.0000
20-Day Service	\$1.9900	\$0.0000
Authorized Overrun	Pa+o	
10-Day Service	\$0.0585	\$0.0000
20-Day Service	\$0.0674	\$0.0000
70 pay pervice	Q0.0074	70.0000

NOTE: The Unauthorized Overrun Charge is set out in Section 9 of the Rate Schedule.

Pursuant to Section 6.3(c) of Rate Schedule FRSS, Natural shall retain a Storage Fuel Charge of up to 1.19% of volumes tendered for injection. In addition, any applicable Transportation Fuel Charge will be assessed under Section 6.3(b) of Rate Schedule FRSS. The Fuel Retention Factors will apply unless Natural and Shipper mutually agree on monetary reimbursement.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rates.

FERC Docket: RP10-147-004

Original Sheet No. 16B Original Sheet No. 16B

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FRSS

	Rates pe	r Dth
	Maximum	Minimum
Delivered Service		
Reservation Rate		
10-Day Service	\$3.3400	\$0.0000
20-Day Service	\$3.6000	\$0.0000
Authorized Overrun	Rate	
10-Day Service	\$0.1131	\$0.0000
20-Day Service	\$0.1220	\$0.0000
Storage-Only Option		
Reservation Rate		
10-Day Service	\$1.7300	\$0.0000
20-Day Service	\$1.9900	\$0.0000
Authorized Overrun	Rate	
10-Day Service	\$0.0585	\$0.0000
20-Day Service	\$0.0674	\$0.0000
- 2		

NOTE: The Unauthorized Overrun Charge is set out in Section 9 of the Rate Schedule.

Pursuant to Section 6.3(c) of Rate Schedule FRSS, Natural shall retain a Storage Fuel Charge of up to 0.94% of volumes tendered for injection. In addition, any applicable Transportation Fuel Charge will be assessed under Section 6.3(b) of Rate Schedule FRSS. The Fuel Retention Factors will apply unless Natural and Shipper mutually agree on monetary reimbursement.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rates.

FERC Docket: RP08-319-000

 $\textbf{Original Sheet No. 17} \; \texttt{Original Sheet No. 17} \; : \; \; \texttt{Pending}$

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE IBS

	Rates per Dth		
	Maximum	Minimum	
Daily Access Rate	\$0.1473	\$0.0016	
Authorized Overrun Daily Access Rate	\$0.1473	\$0.0016	

NOTE: The Unauthorized Overrun Charge is set out in Section 8 of the Rate Schedule.

FERC Docket: RP10-147-004

First Revised Sheet No. 18 First Revised Sheet No. 18 Superseding: Original Sheet No. 18 $\,$

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FFTS

PEAK PERIOD MAXIMUM RATES 3/, 4/

	DELIVERY ZONE 2/					
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Reservation	\$ 0.1378	\$ 0.2392	\$ 0.3026	\$ 0.3524	\$ 0.3348	\$
0.2890 Commodity 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Reservation 0.2338	\$ 0.2119 \$ 0.0076	\$ 0.1419 \$ 0.0010	\$ 0.3690 \$ 0.0056	\$ 0.3457 \$ 0.0054	\$ 0.3658 \$ 0.0055	\$
Commodity 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	Ş
Gulf Coast Mainline Reservation 0.2682	\$ 0.2181	\$ 0.2306	\$ 0.1652	\$ 0.1971	\$ 0.1878	\$
Commodity 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Reservation 0.1900	\$ 0.3206	\$ 0.1404	\$ 0.2127	\$ 0.2446	\$ 0.2586	\$
Commodity 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Reservation 0.2599	\$ 0.3444	\$ 0.2019	\$ 0.1167	\$ 0.1955	\$ 0.2290	\$
Commodity 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Reservation 0.3151	\$ 0.3867	\$ 0.2653	\$ 0.1269	\$ 0.1148	\$ 0.1320	\$
Commodity 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Reservation 0.3390	\$ 0.3993	\$ 0.2895	\$ 0.1543	\$ 0.1885	\$ 0.1209	\$
Commodity 0.0049	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Reservation	\$ 0.4519	\$ 0.1572	\$ 0.2925	\$ 0.3378	\$ 0.3645	\$
0.1068 Commodity 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be increased by \$0.0042.

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be determined in accordance with the formula set out in Section 6.3(a)(3) of Rate Schedule FFTS.

and

In accordance with the formula set out in Section 6.3(a)(3) of Rate Schedule FF15.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FFTS

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

FERC Docket: RP10-147-004

Original Sheet No. 18A Original Sheet No. 18A

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FFTS

PEAK PERIOD MAXIMUM RATES 3/, 4/

	DELIVERY ZONE 2/					
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Reservation	\$ 0.1350	\$ 0.2343	\$ 0.2964	\$ 0.3451	\$ 0.3279	\$
0.2830 Commodity 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Reservation	\$ 0.2076	\$ 0.1390	\$ 0.3614	\$ 0.3386	\$ 0.3582	\$
0.2290 Commodity 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Reservation	\$ 0.2136	\$ 0.2258	\$ 0.1618	\$ 0.1930	\$ 0.1839	\$
0.2627 Commodity 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Reservation 0.1861	\$ 0.3140	\$ 0.1375	\$ 0.2083	\$ 0.2396	\$ 0.2533	\$
Commodity 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Reservation	\$ 0.3373	\$ 0.1977	\$ 0.1143	\$ 0.1914	\$ 0.2243	\$
0.2545 Commodity 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Reservation	\$ 0.3788	\$ 0.2598	\$ 0.1243	\$ 0.1125	\$ 0.1293	\$
0.3086 Commodity 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Reservation	\$ 0.3910	\$ 0.2836	\$ 0.1511	\$ 0.1846	\$ 0.1184	\$
0.3320 Commodity 0.0049	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Reservation	\$ 0.4426	\$ 0.1540	\$ 0.2864	\$ 0.3308	\$ 0.3570	\$
0.1046 Commodity 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be

increased

by \$0.0041.

determined

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be

in accordance with the formula set out in Section 6.3(a)(3) of Rate Schedule FFTS.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FFTS

and

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

FERC Docket: RP10-147-004

Original Sheet No. 18B Original Sheet No. 18B

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FFTS

PEAK PERIOD MAXIMUM RATES 3/, 4/

	DELIVERY ZONE 2/					
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Reservation	\$ 0.1307	\$ 0.2269	\$ 0.2870	\$ 0.3342	\$ 0.3176	\$
0.2741 Commodity 0.0039	\$ 0.0021	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Reservation	\$ 0.2010	\$ 0.1346	\$ 0.3500	\$ 0.3279	\$ 0.3469	\$
0.2217 Commodity 0.0028	\$ 0.0076	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Reservation 0.2544	\$ 0.2068	\$ 0.2187	\$ 0.1567	\$ 0.1869	\$ 0.1781	\$
Commodity 0.0035	\$ 0.0092	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Reservation 0.1802	\$ 0.3041	\$ 0.1331	\$ 0.2018	\$ 0.2320	\$ 0.2453	\$
Commodity 0.0019	\$ 0.0146	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Reservation 0.2465	\$ 0.3266	\$ 0.1915	\$ 0.1107	\$ 0.1854	\$ 0.2172	\$
Commodity 0.0033	\$ 0.0159	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Reservation	\$ 0.3668	\$ 0.2516	\$ 0.1203	\$ 0.1089	\$ 0.1252	\$
0.2988 Commodity 0.0044	\$ 0.0193	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Reservation	\$ 0.3787	\$ 0.2746	\$ 0.1464	\$ 0.1788	\$ 0.1146	\$
0.3215 Commodity 0.0049	\$ 0.0201	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Reservation	\$ 0.4286	\$ 0.1491	\$ 0.2774	\$ 0.3203	\$ 0.3457	\$
0.1013 Commodity 0.0002	\$ 0.0206	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be

increased

by \$0.0040.

determined

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be

in accordance with the formula set out in Section 6.3(a)(3) of Rate Schedule FFTS.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FFTS

and

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

FERC Docket: RP08-319-000

Original Sheet No. 19 original Sheet No. 19: Pending

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FFTS

PEAK PERIOD MINIMUM RATES 3/, 4/

DELIVERY ZONE 2/ ______ RECEIPT ZONE 2/ Market Midcontinent Louisiana South Texas Texok Permian -------------------------Iowa-Illinois \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Ś Reservation 0.0000 Commodity \$ 0.0021 \$ 0.0029 \$ 0.0042 \$ 0.0052 \$ 0.0049 \$ 0.0039 Amarillo Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Commodity \$ 0.0076 \$ 0.0010 \$ 0.0056 \$ 0.0054 \$ 0.0055 \$ 0.0028 Gulf Coast Mainline \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ Reservation 0.0000 \$ 0.0092 \$ 0.0014 Commodity \$ 0.0027 \$ 0.0020 \$ 0.0018 \$ 0.0035 Midcontinent \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Reservation \$ 0.0000 Ś 0.0000 Commodity \$ 0.0146 \$ 0.0010 \$ 0.0023 \$ 0.0030 \$ 0.0037 \$ 0.0019 Texok Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Commodity \$ 0.0159 \$ 0.0021 \$ 0.0002 \$ 0.0015 \$ 0.0031 \$ 0.0033 Louisiana \$ 0.0000 \$ 0.0000 \$ 0.0000 Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 Commodity \$ 0.0193 \$ 0.0034 \$ 0.0006 \$ 0.0003 \$ 0.0007 Ś 0.0044 South Texas \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Ś Reservation 0.0000 Commodity \$ 0.0201 \$ 0.0039 \$ 0.0011 \$ 0.0018 \$ 0.0006 \$ 0.0049 Permian Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0206 \$ 0.0012 \$ 0.0040 \$ 0.0049 Commodity \$ 0.0055 \$ 0.0002

NOTES: The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FFTS

and

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

FERC Docket: RP10-147-004

First Revised Sheet No. 20 First Revised Sheet No. 20 Superseding: Original Sheet No. 20

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FFTS

OFF-PEAK PERIOD MAXIMUM RATES 3/, 4/

			DELIVERY	ZONE 2/		
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Reservation 0.2890	\$ 0.1276	\$ 0.2392	\$ 0.3026	\$ 0.3524	\$ 0.3348	\$
Commodity 0.0039	\$ 0.0005	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Reservation 0.2338	\$ 0.1875	\$ 0.1419	\$ 0.3690	\$ 0.3457	\$ 0.3658	\$
Commodity 0.0028	\$ 0.0020	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Reservation 0.2682	\$ 0.1892	\$ 0.2306	\$ 0.1652	\$ 0.1971	\$ 0.1878	\$
Commodity 0.0035	\$ 0.0023	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Reservation	\$ 0.2666	\$ 0.1404	\$ 0.2127	\$ 0.2446	\$ 0.2586	\$
0.1900 Commodity 0.0019	\$ 0.0034	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Reservation	\$ 0.2853	\$ 0.2019	\$ 0.1167	\$ 0.1955	\$ 0.2290	\$
0.2599 Commodity 0.0033	\$ 0.0037	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Reservation	\$ 0.3180	\$ 0.2653	\$ 0.1269	\$ 0.1148	\$ 0.1320	\$
0.3151 Commodity 0.0044	\$ 0.0045	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Reservation	\$ 0.3281	\$ 0.2895	\$ 0.1543	\$ 0.1885	\$ 0.1209	\$
0.3390 Commodity 0.0049	\$ 0.0048	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Reservation	\$ 0.3789	\$ 0.1572	\$ 0.2925	\$ 0.3378	\$ 0.3645	\$
0.1068 Commodity 0.0002	\$ 0.0048	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be increased by \$0.0042.

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be determined in accordance with the formula set out in Section 6.3(a)(3) of Rate Schedule FFTS.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FFTS

and
the General Terms and Conditions will be added or deducted on the monthly bills of affected
Shippers when applicable.

12

FERC Docket: RP10-147-004

Original Sheet No. 20A Original Sheet No. 20A

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FFTS

OFF-PEAK PERIOD MAXIMUM RATES 3/, 4/

			DELIVERY	ZONE 2/		
RECEIPT ZONE 2/	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Reservation 0.2830	\$ 0.1249	\$ 0.2343	\$ 0.2964	\$ 0.3451	\$ 0.3279	\$
Commodity 0.0039	\$ 0.0005	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Reservation 0.2290	\$ 0.1836	\$ 0.1390	\$ 0.3614	\$ 0.3386	\$ 0.3582	\$
Commodity 0.0028	\$ 0.0020	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Reservation 0.2627	\$ 0.1853	\$ 0.2258	\$ 0.1618	\$ 0.1930	\$ 0.1839	\$
Commodity 0.0035	\$ 0.0023	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Reservation 0.1861	\$ 0.2611	\$ 0.1375	\$ 0.2083	\$ 0.2396	\$ 0.2533	\$
Commodity 0.0019	\$ 0.0034	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Reservation 0.2545	\$ 0.2794	\$ 0.1977	\$ 0.1143	\$ 0.1914	\$ 0.2243	\$
Commodity 0.0033	\$ 0.0037	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Reservation 0.3086	\$ 0.3114	\$ 0.2598	\$ 0.1243	\$ 0.1125	\$ 0.1293	\$
Commodity 0.0044	\$ 0.0045	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Reservation 0.3320	\$ 0.3213	\$ 0.2836	\$ 0.1511	\$ 0.1846	\$ 0.1184	\$
Commodity 0.0049	\$ 0.0048	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Reservation	\$ 0.3711	\$ 0.1540	\$ 0.2864	\$ 0.3308	\$ 0.3570	\$
0.1046 Commodity 0.0002	\$ 0.0048	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be

increased

by \$0.0041.

determined

and

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be

in accordance with the formula set out in Section 6.3(a)(3) of Rate Schedule FFTS.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FFTS

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

FERC Docket: RP10-147-004

Original Sheet No. 20B Original Sheet No. 20B

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FFTS

OFF-PEAK PERIOD MAXIMUM RATES 3/, 4/

	DELIVERY ZONE 2/					
RECEIPT ZONE 2/ Permian	Market	Midcontinent	Texok	Louisiana	South Texas	
Iowa-Illinois Reservation	\$ 0.1210	\$ 0.2269	\$ 0.2870	\$ 0.3342	\$ 0.3176	\$
0.2741 Commodity 0.0039	\$ 0.0005	\$ 0.0029	\$ 0.0042	\$ 0.0052	\$ 0.0049	\$
Amarillo Reservation 0.2217	\$ 0.1778	\$ 0.1346	\$ 0.3500	\$ 0.3279	\$ 0.3469	\$
Commodity	\$ 0.0020	\$ 0.0010	\$ 0.0056	\$ 0.0054	\$ 0.0055	\$
Gulf Coast Mainline Reservation 0.2544	\$ 0.1794	\$ 0.2187	\$ 0.1567	\$ 0.1869	\$ 0.1781	\$
Commodity 0.0035	\$ 0.0023	\$ 0.0027	\$ 0.0014	\$ 0.0020	\$ 0.0018	\$
Midcontinent Reservation 0.1802	\$ 0.2528	\$ 0.1331	\$ 0.2018	\$ 0.2320	\$ 0.2453	\$
Commodity 0.0019	\$ 0.0034	\$ 0.0010	\$ 0.0023	\$ 0.0030	\$ 0.0037	\$
Texok Reservation 0.2465	\$ 0.2706	\$ 0.1915	\$ 0.1107	\$ 0.1854	\$ 0.2172	\$
Commodity 0.0033	\$ 0.0037	\$ 0.0021	\$ 0.0002	\$ 0.0015	\$ 0.0031	\$
Louisiana Reservation 0.2988	\$ 0.3016	\$ 0.2516	\$ 0.1203	\$ 0.1089	\$ 0.1252	\$
Commodity 0.0044	\$ 0.0045	\$ 0.0034	\$ 0.0006	\$ 0.0003	\$ 0.0007	\$
South Texas Reservation	\$ 0.3111	\$ 0.2746	\$ 0.1464	\$ 0.1788	\$ 0.1146	\$
0.3215 Commodity 0.0049	\$ 0.0048	\$ 0.0039	\$ 0.0011	\$ 0.0018	\$ 0.0006	\$
Permian Reservation	\$ 0.3594	\$ 0.1491	\$ 0.2774	\$ 0.3203	\$ 0.3457	\$
0.1013 Commodity 0.0002	\$ 0.0048	\$ 0.0012	\$ 0.0040	\$ 0.0049	\$ 0.0055	\$

NOTES: For Shippers electing the LN option, the applicable maximum Reservation Rate shall be

increased

by \$0.0040.

determined

For Shippers electing the SW option, the applicable maximum Reservation Rate shall be

in accordance with the formula set out in Section 6.3(a)(3) of Rate Schedule FFTS.

The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12

of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FFTS

and

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of off-peak rates, see Footnote 1/.

FERC Docket: RP08-319-000

Original Sheet No. 21 original Sheet No. 21: Pending

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE FFTS

OFF-PEAK PERIOD MINIMUM RATES 3/, 4/

DELIVERY ZONE 2/ ______ RECEIPT ZONE 2/ Market Midcontinent Louisiana South Texas Texok Permian -------------------------Iowa-Illinois \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Ś Reservation 0.0000 Commodity \$ 0.0005 \$ 0.0029 \$ 0.0042 \$ 0.0052 \$ 0.0049 \$ 0.0039 Amarillo Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Commodity \$ 0.0020 \$ 0.0010 \$ 0.0056 \$ 0.0054 \$ 0.0055 \$ 0.0028 Gulf Coast Mainline \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ Reservation 0.0000 \$ 0.0023 \$ 0.0014 Commodity \$ 0.0027 \$ 0.0020 \$ 0.0018 \$ 0.0035 Midcontinent \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Reservation \$ 0.0000 Ś 0.0000 Commodity \$ 0.0034 \$ 0.0010 \$ 0.0023 \$ 0.0030 \$ 0.0037 \$ 0.0019 Texok Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Commodity \$ 0.0037 \$ 0.0021 \$ 0.0002 \$ 0.0015 \$ 0.0031 \$ 0.0033 Louisiana \$ 0.0000 \$ 0.0000 \$ 0.0000 Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 Commodity \$ 0.0045 \$ 0.0034 \$ 0.0006 \$ 0.0003 \$ 0.0007 Ś 0.0044 South Texas \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 Ś Reservation 0.0000 Commodity \$ 0.0048 \$ 0.0039 \$ 0.0011 \$ 0.0018 \$ 0.0006 \$ 0.0049 Permian Reservation \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0000 \$ 0.0012 \$ 0.0040 \$ 0.0049 Commodity \$ 0.0048 \$ 0.0055 \$ 0.0002

NOTES: The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

and

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in Rate Schedule FFTS

the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of off-peak rates, see Footnote 1/.

FERC Docket: RP10-147-004

First Revised Sheet No. 22 First Revised Sheet No. 22 Superseding: Original Sheet No. 22

GATHERING RATES -----

	Rates per Dth		
	Maximum	Minimum	
Firm			
Demand	\$ 8.5900	\$0.0000	
Commodity	\$ 0.0363	\$0.0363	
Interruptible Commodity	\$ 0.3276	\$0.0363	
Rate Schedule FFTS Demand	\$ 0.2826	\$0.0000	
Commodity	\$ 0.0363	\$0.0363	
G Customers Commodity	\$ 0.6003	\$0.0363	

NOTES:

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will be added to the commodity rates when applicable. See Sheet No. 32 for the currently effective rate.

Firm gathering rates are assessed on any point (and on the associated MDQ) which is a point on Natural's gathering system and which is also a Primary Point under any firm Agreement. In all other situations, the interruptible rate is applicable.

FERC Docket: RP10-147-004

Original Sheet No. 22A Original Sheet No. 22A

GATHERING RATES

	Rates per	Rates per Dth		
	Maximum	Minimum		
Firm				
Demand	\$ 8.4200	\$0.0000		
Commodity	\$ 0.0363	\$0.0363		
Interruptible Commodity	\$ 0.3276	\$0.0363		
Rate Schedule FFTS Demand	\$ 0.2767	\$0.0000		
Commodity	\$ 0.0363	\$0.0363		
G Customers Commodity	\$ 0.5880	\$0.0363		

NOTES:

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will be added to the commodity rates when applicable. See Sheet No. 32 for the currently effective rate.

Firm gathering rates are assessed on any point (and on the associated MDQ) which is a point on Natural's gathering system and which is also a Primary Point under any firm Agreement. In all other situations, the interruptible rate is applicable.

FERC Docket: RP10-147-004

Original Sheet No. 22B Original Sheet No. 22B

GATHERING RATES

	Rates per	Rates per Dth		
	Maximum	Minimum		
Firm				
Demand	\$ 8.1500	\$0.0000		
Commodity	\$ 0.0363	\$0.0363		
Interruptible Commodity	\$ 0.3276	\$0.0363		
Rate Schedule FFTS Demand	\$ 0.2680	\$0.0000		
Commodity	\$ 0.0363	\$0.0363		
G Customers Commodity	\$ 0.5694	\$0.0363		

NOTES:

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will be added to the commodity rates when applicable. See Sheet No. 32 for the currently effective rate.

Firm gathering rates are assessed on any point (and on the associated MDQ) which is a point on Natural's gathering system and which is also a Primary Point under any firm Agreement. In all other situations, the interruptible rate is applicable.

FERC Docket: RP10-147-004

First Revised Sheet No. 23 First Revised Sheet No. 23 Superseding: Original Sheet No. 23 $\,$

OFFSHORE ZONE CHARGES 2/ _____

	Peak and Off-Peak Rates		
	Maximum	Minimum	
Firm Rate Schedules Demand	\$2.0700	\$0.0000	
Interruptible Rate Schedules Commodity	\$0.0700	\$0.0000	
Rate Schedule FFTS Demand	\$0.0679	\$0.0000	
G Customers Commodity	\$0.1358	\$0.0000	

NOTES: The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

> Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when ${\tt applicable.}$

For footnotes, see Sheet Nos. 36 and 37. For definition of peak and off-peak rates, see Footnote 1/.

FERC Docket: RP10-147-004

Original Sheet No. 23A Original Sheet No. 23A

OFFSHORE ZONE CHARGES 2/

	Peak and Off-Peak Rat		
	Maximum	Minimum	
Firm Rate Schedules Demand	\$2.0200	\$0.0000	
Interruptible Rate Schedules Commodity	\$0.0700	\$0.0000	
Rate Schedule FFTS Demand	\$0.0665	\$0.0000	
G Customers Commodity	\$0.1330	\$0.0000	

NOTES: The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of peak and off-peak rates, see Footnote 1/.

FERC Docket: RP10-147-004

Original Sheet No. 23B Original Sheet No. 23B

OFFSHORE ZONE CHARGES 2/

	Peak and Off-Peak Rat	
	Maximum	Minimum
Firm Rate Schedules Demand	\$1.9600	\$0.0000
Interruptible Rate Schedules Commodity	\$0.0700	\$0.0000
Rate Schedule FFTS Demand	\$0.0644	\$0.0000
G Customers Commodity	\$0.1288	\$0.0000

NOTES: The Balancing Service and Authorized and Unauthorized Overrun Charges are set out in Section 12 of the General Terms and Conditions.

Pursuant to Sections 21, 38 and 40 of the General Terms and Conditions, various surcharge adjustments are applied to the Base Rates if applicable. See Sheet No. 32 for the currently effective rates. In addition, other charges or credits as provided for in the General Terms and Conditions will be added or deducted on the monthly bills of affected Shippers when applicable.

For footnotes, see Sheet Nos. 36 and 37. For definition of peak and off-peak rates, see Footnote 1/.

FERC Docket: RP08-319-000

Original Sheet No. 24 Original Sheet No. 24: Pending

CURRENTLY EFFECTIVE INCREMENTAL CHARGE APPLICABLE TO THE MORAINE LATERAL 2/

Charge Applicable to Firm and Interruptible Rate Schedules

Peak and Off-Peak Rates

Commodity Charge
Maximum
Minimum

\$.0401 \$.0028

See Sheet No. 36 for footnotes.

FERC Docket: RP10-147-004

First Revised Sheet No. 25 First Revised Sheet No. 25 Superseding: Original Sheet No. 25 $\,$

CURRENTLY EFFECTIVE TRANSPORTATION FUEL RETENTION FACTORS 1/ ______

DELIVERY ZONE 2/

RECEIPT ZONE 2/	Market	Midcontinent	Texok	Louisiana	South Texas	Permian
Iowa-Illinois	1.27%	0.60%	0.60%	0.60%	0.60%	0.60%
Amarillo	2.23%	0.60%	0.87%	1.09%	1.20%	0.60%
Gulf Coast Mainline	2.12%	0.60%	0.60%	1.06%	1.13%	0.60%
Midcontinent	3.16%	1.48%	1.68%	2.42%	2.49%	1.48%
Texok	2.66%	2.00%	1.12%	1.37%	1.64%	1.43%
Louisiana	2.28%	1.10%	0.73%	0.77%	0.76%	0.60%
South Texas	3.12%	1.98%	1.61%	1.58%	1.13%	1.41%
Permian	4.40%	1.89%	1.82%	2.56%	2.63%	1.33%

^{1/} Fuel Gas and Gas Lost and Unaccounted For (GLU) retention factors applicable to service provided under Rate Schedules FTS, FFTS, FTS-G and ITS. The Fuel Retention Factors stated above include a GLU

factor of 0.60%. The Fuel Retention Factors apply unless Natural and Shipper mutually agree on monetary reimbursement.

^{2/} See Footnote 2/ on Sheet No. 36 for definition of transportation zones.

FERC Docket: RP10-147-004

Original Sheet No. 25A Original Sheet No. 25A

CURRENTLY EFFECTIVE TRANSPORTATION FUEL RETENTION FACTORS 1/

DELIVERY ZONE 2/

RECEIPT ZONE 2/		Midcontinent			South Texas	Permian
Iowa-Illinois	1.00%	0.47%	0.47%	0.47%	0.47%	0.47%
Amarillo	1.75%	0.47%	0.68%	0.86%	0.94%	0.47%
Gulf Coast Mainline	1.67%	0.47%	0.47%	0.84%	0.89%	0.47%
Midcontinent	2.49%	1.17%	1.32%	1.90%	1.95%	1.17%
Texok	2.09%	1.57%	0.88%	1.08%	1.29%	1.12%
Louisiana	1.79%	0.86%	0.57%	0.61%	0.60%	0.47%
South Texas	2.45%	1.56%	1.27%	1.24%	0.89%	1.11%
Permian	3.46%	1.49%	1.43%	2.01%	2.06%	1.05%

^{1/} Fuel Gas and Gas Lost and Unaccounted For (GLU) retention factors applicable to service provided under Rate Schedules FTS, FFTS, FTS-G and ITS. The Fuel Retention Factors stated above include a GLU factor of 0.47%. The Fuel Retention Factors apply unless Natural and Shipper mutually agree on monetary reimbursement.

^{2/} See Footnote 2/ on Sheet No. 36 for definition of transportation zones.

FERC Docket: RP10-147-004

First Revised Sheet No. 26 First Revised Sheet No. 26 Superseding: Original Sheet No. 26 $\,$

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE BESS

	Rates per	Dth
	Maximum	Minimum
Capacity and AOR	\$0.0700	\$0.0000

NOTES: Pursuant to Section 6.1(a)(2) of Rate Schedule BESS, Natural shall retain up to 1.19% of volumes tendered for injection and 0% of volumes tendered for withdrawal unless Natural and Shipper mutually agree on monetary reimbursement.

> The Balancing Service Charges are set out in Section 12 of the General Terms and Conditions. The Unauthorized $\,$ Overrun Charge is set out in Section 10 of the Rate Schedule.

> Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will be added to the volumes withdrawn when applicable. See Sheet No. 32 for the currently effective rate.

Capacity and AOR

FERC Docket: RP10-147-004

Original Sheet No. 26A Original Sheet No. 26A

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE BESS

	Rates	per	Dth
Maxim	um 		Minimum
\$0.07	00		\$0.0000

NOTES: Pursuant to Section 6.1(a)(2) of Rate Schedule BESS, Natural shall retain up to 0.94% of volumes tendered for injection and 0% of volumes tendered for withdrawal unless Natural and Shipper mutually agree on monetary reimbursement.

The Balancing Service Charges are set out in Section 12 of the General Terms and Conditions. The Unauthorized Overrun Charge is set out in Section 10 of the Rate Schedule.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will be added to the volumes withdrawn when applicable. See Sheet No. 32 for the currently effective rate.

FERC Docket: RP10-147-004

Second Revised Sheet No. 27 Second Revised Sheet No. 27

Superseding: First Revised Sheet No. 27

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULES DSS AND NSS

		=== Maximum	Minimum
RATE SCHEDULE DS	SS		
Section 6.2(a)	Reservation Charge	\$ 5.70	\$0.0000
Section 6.2(b)	1 1 9 9		
	and Cycling Option Charge	0.8400	0.0000
	Unauthorized Inventory Cycling Charge	10.0000	0.0000
RATE SCHEDULE NS			
Section 6.2(a)	Reservation Charge	\$ 3.07	\$0.0000
	Reservation Charge		
	(Storage Expansion 2003)	\$ 3.96	
	(Sayre Storage Project)	\$ 4.12	
	(North Lansing NSS Project)	\$ 7.23	
	(Herscher Galesville NSS Project)	\$ 6.8750	
			chedule
		DSS	NSS
Section 5.3	Sourcing Charge	\$0.3710	\$0.3710
Section 9	Authorized Overrun Charges	, , , , , , , ,	, , , , , , , ,
	Injection/Withdrawal	0.1874	0.1009
	Storage Expansion 2003	0.0000	0.1302
	Sayre Storage Project	0.0000	0.1355
	North Lansing NSS Project	0.0000	0.2377
	Herscher Galesville NSS Project	0.0000	0.2260
	Maximum Storage Volume	0.0700	0.0700
	Minimum Withdrawal Requirement	0.0700	N/A
	MDQ for Transportation	0.1442	N/A
	Minimum Sourcing and Overrun Charge	0.0000	0.0000

NOTES: Pursuant to Section 9 of Rate Schedules DSS and NSS, unauthorized volumes will be charged the Overrun Charges which are set out in Section 12 of the General Terms and Conditions.

Pursuant to Section 6.2(c) of Rate Schedule DSS and Section 6.2(b) of Rate Schedule NSS, Natural shall retain a Storage Fuel Charge of up to 1.19% of volumes tendered for injection. In addition, a Transportation Fuel Charge will apply to Rate Schedule DSS Shippers equal to the Transportation Fuel Retention Factor for the Iowallinois Receipt Zone to the Market Delivery Zone times the volumes tendered for injection or gas transported under nominated firm transportation. The Fuel Retention Factors will apply unless Natural and Shipper mutually agree on monetary reimbursement.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rate.

FERC Docket: RP10-147-004

Original Sheet No. 27A Original Sheet No. 27A

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULES DSS AND NSS

		Maximum	Minimum
RATE SCHEDULE DS	S		
	Reservation Charge Authorized Inventory Cycling Charge	\$ 5.53	\$0.0000
	and Cycling Option Charge Unauthorized Inventory Cycling Charge	0.8400	
RATE SCHEDULE NS		10.0000	0.0000
Section 6.2(a)	Reservation Charge Reservation Charge	\$ 2.98	\$0.0000
	(Storage Expansion 2003) (Sayre Storage Project) (North Lansing NSS Project) (Herscher Galesville NSS Project)		\$0.0000 \$0.0000
Section 5.3 Section 9	Sourcing Charge Authorized Overrun Charges Injection/Withdrawal Storage Expansion 2003 Sayre Storage Project North Lansing NSS Project Herscher Galesville NSS Project Maximum Storage Volume Minimum Withdrawal Requirement MDQ for Transportation Minimum Sourcing and Overrun Charge	\$0.3710 0.1874 0.0000 0.0000 0.0000 0.0000 0.0700 0.0700 0.1442 0.0000	0.1009 0.1302 0.1355 0.2377 0.2260 0.0700 N/A N/A

NOTES: Pursuant to Section 9 of Rate Schedules DSS and NSS, unauthorized volumes will be charged the Overrun Charges which are set out in Section 12 of the General Terms and Conditions.

Pursuant to Section 6.2(c) of Rate Schedule DSS and Section 6.2(b) of Rate Schedule NSS, Natural shall retain a Storage Fuel Charge of up to 1.19% of volumes tendered for injection. In addition, a Transportation Fuel Charge will apply to Rate Schedule DSS Shippers equal to the Transportation Fuel Retention Factor for the Iowa-Illinois Receipt Zone to the Market Delivery Zone times the volumes tendered for injection or gas transported under nominated firm transportation. The Fuel Retention Factors will apply unless Natural and Shipper mutually agree on monetary reimbursement.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rate.

FERC Docket: RP10-147-004

Original Sheet No. 27B Original Sheet No. 27B

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULES DSS AND NSS

		Maximum	Minimum
RATE SCHEDULE DS	S		
	Reservation Charge Authorized Inventory Cycling Charge	\$ 5.53	\$0.0000
	and Cycling Option Charge Unauthorized Inventory Cycling Charge	0.8400	
RATE SCHEDULE NS		10.0000	0.0000
Section 6.2(a)	Reservation Charge Reservation Charge	\$ 2.98	\$0.0000
	(Storage Expansion 2003) (Sayre Storage Project) (North Lansing NSS Project) (Herscher Galesville NSS Project)		\$0.0000 \$0.0000
Section 5.3 Section 9	Sourcing Charge Authorized Overrun Charges Injection/Withdrawal Storage Expansion 2003 Sayre Storage Project North Lansing NSS Project Herscher Galesville NSS Project Maximum Storage Volume Minimum Withdrawal Requirement MDQ for Transportation Minimum Sourcing and Overrun Charge	\$0.3710 0.1874 0.0000 0.0000 0.0000 0.0000 0.0700 0.0700 0.1442 0.0000	0.1009 0.1302 0.1355 0.2377 0.2260 0.0700 N/A N/A

NOTES: Pursuant to Section 9 of Rate Schedules DSS and NSS, unauthorized volumes will be charged the Overrun Charges which are set out in Section 12 of the General Terms and Conditions.

Pursuant to Section 6.2(c) of Rate Schedule DSS and Section 6.2(b) of Rate Schedule NSS, Natural shall retain a Storage Fuel Charge of up to 0.94% of volumes tendered for injection. In addition, a Transportation Fuel Charge will apply to Rate Schedule DSS Shippers equal to the Transportation Fuel Retention Factor for the Iowa-Illinois Receipt Zone to the Market Delivery Zone times the volumes tendered for injection or gas transported under nominated firm transportation. The Fuel Retention Factors will apply unless Natural and Shipper mutually agree on monetary reimbursement.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rate.

FERC Docket: RP10-147-004

First Revised Sheet No. 28 First Revised Sheet No. 28

Superseding: Original Sheet No. 28

FOOTNOTES APPLICABLE TO SHEET NO. 27

NOTES: Pursuant to Section 9 of Rate Schedules DSS and NSS, unauthorized volumes will be charged the Overrun Charges which are set out in Section 12 of the General Terms and Conditions.

Pursuant to Section 6.2(c) of Rate Schedule DSS and Section 6.2(b) of Rate Schedule NSS, Natural shall retain a Storage Fuel Charge of up to 1.19% of volumes tendered for injection. In addition, a Transportation Fuel Charge will apply to Rate Schedule DSS Shippers equal to the Transportation Fuel Retention Factor for the Iowa-Illinois Receipt Zone to the Market Delivery Zone times the volumes tendered for injection or gas transported under nominated firm transportation. The Fuel Retention Factors will apply unless Natural and Shipper mutually agree on monetary reimbursement.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rates.

FERC Docket: RP10-147-004

Original Sheet No. 28A Original Sheet No. 28A

FOOTNOTES APPLICABLE TO SHEET NO. 27

NOTES: Pursuant to Section 9 of Rate Schedules DSS and NSS, unauthorized volumes will be charged the Overrun Charges which are set out in Section 12 of the General Terms and Conditions.

Pursuant to Section 6.2(c) of Rate Schedule DSS and Section 6.2(b) of Rate Schedule NSS, Natural shall retain a Storage Fuel Charge of up to 0.94% of volumes tendered for injection. In addition, a Transportation Fuel Charge will apply to Rate Schedule DSS Shippers equal to the Transportation Fuel Retention Factor for the Iowa-Illinois Receipt Zone to the Market Delivery Zone times the volumes tendered for injection or gas transported under nominated firm transportation. The Fuel Retention Factors will apply unless Natural and Shipper mutually agree on monetary reimbursement.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rates.

FERC Docket: RP10-147-004

First Revised Sheet No. 29 First Revised Sheet No. 29 Superseding: Original Sheet No. 29

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE PALS ______

		Rates p	Rates per Dth	
		Maximum	Minimum	
Section 6.1(a)	Usage Charge	\$0.1092	\$0.0000	

Pursuant to Section 6.1(b) of Rate Schedule PALS, Natural shall retain up to 1.19% of volumes tendered for injection and 0% of volumes tendered for withdrawal unless Natural and Shipper mutually agree on monetary reimbursement.

> Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rates.

FERC Docket: RP10-147-004

Original Sheet No. 29A Original Sheet No. 29A

CURRENTLY EFFECTIVE BASE RATES
APPLICABLE TO RATE SCHEDULE PALS

		Rates	Rates per Dth	
		Maximum	Minimum	
Section 6.1(a)	Usage Charge	\$0.1092	\$0.0000	

NOTES: Pursuant to Section 6.1(b) of Rate Schedule PALS, Natural shall retain up to 0.94% of volumes tendered for injection and 0% of volumes tendered for withdrawal unless Natural and Shipper mutually agree on monetary reimbursement.

Pursuant to Section 40 of the General Terms and Conditions, the ACA surcharge will apply when applicable. See Sheet No. 32 for the currently effective rates.

FERC Docket: RP08-319-000

Original Sheet No. 30 Original Sheet No. 30: Pending

CURRENTLY EFFECTIVE BASE RATES APPLICABLE TO RATE SCHEDULE LPS

	Rates per Dth*	
	Maximum	Minimum
Initial Rate	\$0.2532	\$0.0000
Park/Loan Balance Rate	\$0.1266	\$0.0000
Completion Rate	\$0.2532	\$0.0000
Authorized Overrun Initial Rate Park/Loan Balance Rate Completion Rate	\$0.2532 \$0.1266 \$0.2532	\$0.0000 \$0.0000 \$0.0000

NOTE: The Unauthorized Overrun Charge is set out in Section 6.1(b) of the Rate Schedule.

^{*} These rates apply to transactions under Rate Schedule LPS entered into on or after January 1, 2002.

FERC Docket: RP08-319-000

Original Sheet No. 31 Original Sheet No. 31: Pending

SECTION 37 LATERAL LINE CHARGE CURRENTLY EFFECTIVE RATES APPLICABLE TO RATE SCHEDULE F-1

Lateral Line Charge pursuant to Section 37 of the General Terms and Conditions of this Tariff. Each Buyer shall pay Natural a Monthly Facility Charge of one-twelfth (1/12) of the applicable Annual Facility Charge shown below:

Buyer	Annual Facility Charge Effective 1/1/95	Annual Facility Charge Reduction Until Fully Depreciated
Northern Indiana Public Service Company North Hayden Meter	\$113,988	\$2,980
Iowa-Illinois Gas and Electric Company Davenport Meter	\$ 25,697	\$ 591

FERC Docket: RP09-877-000

Second Revised Sheet No. 32 Second Revised Sheet No. 32 Superseding: First Revised Sheet No. 32

ACA Surcharge

SECTION 40 SURCHARGE ANNUAL CHARGES ADJUSTMENT (ACA) CHARGE ______

Applied to Rate Schedules FTS, FFTS, FTS-G, ITS, BESS, DSS, NSS and FRSS when applicable pursuant to Section 154.402 of the Commission's Regulations and Section 40 of this Tariff.

Rate per Dth
----Maximum Minimum \$.0019 \$.0019

FERC Docket: RP99-176-186

First Revised Sheet No. 33 First Revised Sheet No. 33

Superseding: Original Sheet No. 33

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity Primary Receipt	Rate Primar	Term of cy Delivery	Volume	Reservation	
Shipper Name Charge(s) Point(s) / PIN No(s).	Schedule	Contract / PIN No(s).	(Dth/d)	Charge(s)	
			-		
Suez Energy Marketing NA, Inc. $1/6/$	NSS 7/	2/	53,333 3/	\$3.07 4/, 6/	5/,
Occidental Energy Marketing, Inc. 1/6/7/	NSS 7/	2/	40,000 3/	\$3.07 4/, 6/	5/,

- 1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ Negotiated Rate Term. The negotiated rates set forth herein shall be in force during a term commencing on the latter of: (i) April 1, 2006; or (ii) the date capacity to provide the service hereunder is available on Natural's System; through March 31, 2011.
- 3/ Storage Agreement. The Negotiated Rates set forth herein shall apply to storage service provided by Natural to Shipper under Rate Schedule NSS with a
 Maximum Storage Volume (MSV) of Three Million (3,000,000) Dth and an MDQ of Forty Thousand (40,000) Dth/d on the Amarillo Leg 100% (the "Storage Agreement"),
- except for Suez Energy Marketing NA, Inc., which has a Maximum Storage Volume (MSV) of Three Million Nine Hundred Ninety Nine Thousand Nine Hundred Seventy
- Five (3,999,975) Dth and an MDQ of Fifty Three Thousand Three Hundred Thirty Three (53,333) Dth/day on the Amarillo Leg 100%.
- 4/ Negotiated Monthly Reservation Rate. The Negotiated Monthly Reservation Rate per Dth of MDQ (Peak and Off-Peak periods) shall apply only to service provided by Natural to Shipper under the Rate Schedule NSS Agreement, subject to all other terms and conditions set forth herein.
- 5/ Commodity Rate. All applicable maximum commodity rates, charges and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time, shall apply to storage service provided by Natural to Shipper under the Rate Schedule NSS Agreement, unless otherwise expressly provided herein.
 - 6/ Negotiated Rates.
- (a) Authorized Overrun Service. All Authorized Overrun Service provided to Shipper under the Storage Agreement shall be billed at the applicable maximum Authorized Overrun Rate as set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

FERC Docket: RP08-319-000

Original Sheet No. 33A Original Sheet No. 33A: Pending

FOOTNOTES APPLICABLE TO SHEET NO. 33

6/ Negotiated Rates. (con't)

- (b) Storage Fuel. In addition to the Negotiated Rates, Shipper shall be charged for Storage Fuel in accordance with the applicable provisions of Natural's FERC Gas Tariff, as may be revised from time to time.
- (c) Discountable Third Party Surcharges. The Negotiated Rates reflect an initial discount of all surcharges which contemplate ultimate payment to third

parties to support industry-wide research and which are discountable by Natural or similar reservation and commodity surcharges ("Discountable Third Party

Surcharges") otherwise payable by Shipper; provided, however, that in the event Natural is required to collect such Discountable Third Party Surcharges

from Shipper and to pay such Discountable Third Party Surcharges to the FERC or to any third party, then Shipper shall pay Natural any such amounts billed by Natural in addition to the Negotiated Rates.

- (d) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein, all applicable maximum rates, charges, surcharges and
- penalties of any nature set forth in Natural's FERC Gas Tariff (including any unauthorized overrun charges which may be applicable) shall apply to service provided to Shipper under the Storage Agreement.
- 7/ Receipt and Delivery Points. The Receipt Points for injection and the Delivery Points for withdrawal under the Rate Schedule NSS Agreement shall be the Storage

Points as defined in Section 9.3 of the General Terms and Conditions set forth in Natural's FERC Gas

FERC Docket: RP08-319-000

 $\textbf{Original Sheet No. 33B} \; \texttt{Original Sheet No. 33B} \; : \; \; \texttt{Pending}$

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

Commodity Primary Receipt Shipper Name Charge(s) Point(s) / PIN No(s).	Schedul	Term of mary Delivery e Contract (s) / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)	
Devon Gas Services, L.P.	FTS 2	1/01/2007 1/ / through 12/31/2016	- 59,000	2/	2/
Devon Gas Services, L.P. 2/	FTS 2	1/01/2007 1/ / through 12/31/2016	15,000	2/	2/
Crosstex Gulf Coast Marketing Ltd 2/	. FTS 2	1/01/2007 1/ / through 12/31/2016	51,000	2/	2/
J-W Gathering Company 2/	FTS 2	1/01/2007 1/ / through 12/31/2016	11,000	2/	2/
J-W Gathering Company 2/	FTS 2	1/01/2007 1/ / through 12/31/2016	4,000	2/	2/

PURSUANT TO SECTION 49

^{1/} Negotiated Rate Term. The Negotiated Rate Term shall commence on the date Segment 17 Expansion capacity to provide this service is available ("Effective Date"), which is anticipated to be January 1, 2007. The Negotiated Rate Term shall end ten (10) years from the Effective Date.

^{2/} This information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-389-000

First Revised Sheet No. 33C First Revised Sheet No. 33C

Superseding: Original Sheet No. 33C

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s)	Rate Primary De Schedule Point(s) / I	elivery Contract	Volume (Dth/d)		
	Phillips Company 1/ evon Bridgeport Plant	FTS Gulf South Goodri	2/ (a) .ch	10,000	\$6.6700	3/
(I	PIN No. 901850) 4/	(PIN No. 6297) 5/	′			
	Phillips Company 1/ evon Bridgeport Plant	FTS Kansas Gas Servic	2/ (b)	5,000	\$1.5708	3/
(E	PIN No. 901850) 4/	(PIN No. 3063) 5/	′			
Occider Express	ntal Energy Marketing, In	c. 1/ FTS-SW KMTP/NGPL	2/ (c)	41,000	\$6.0833 3/ 6/	3/ Pony
(E	PIN No. 11165) 4/	(PIN No. 5579) 5/	6/			

- 1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ Negotiated Rate Term. The Negotiated Rate Term shall commence on the date the Segment One Expansion and the Amarillo to Gulf Coast Expansion ("Expansion") is

ready to receive firm service ("Effective Date") which is anticipated to be April 15, 2006. For purposes of this section, the Expansion shall be considered

ready to receive firm service when the pipeline and compression are ready for firm service. This negotiated rate term shall end:

- (a) five (5) years and six (6) months from the Effective Date;
- (b) five (5) years and one month from the Effective Date; and (c) five (5) years from the Effective Date.
- 3/ Negotiated Rates.
- (a) Eligible Transportation Quantity. The Negotiated Rate shall apply to a total aggregate maximum daily transportation quantity of Dth/day as shown above,

for all quantities transported under the FTS Agreement and any associated capacity release replacement agreements (the "Eligible Transportation

Quantity"). Any quantities transported in excess of the Eligible Transportation Quantity shall be billed at the greater of the Negotiated Rate or the

applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff.

FERC Docket: RP99-176-172

First Revised Sheet No. 33D First Revised Sheet No. 33D

Superseding: Original Sheet No. 33D

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity Primary Receipt	Rate Primar	Term of Ty Delivery	Volume	Reservation
Shipper Name Charge(s) Point(s) / PIN No(:	Schedule	Contract b) / PIN No(s).	(Dth/d)	Charge(s)
			 _	
MidAmerican Energy Company 1/2/2/2/	FTS	12/01/2008	2/	2/
		through 11/30/2011		

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} This information is set out in the executed Firm Transportation Negotiated Rate Agreement which is currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 33E Original Sheet No. 33E : Pending

TRANSACTIONS	STATEMENT OF NEGOTIATED RATE
TIGHOROT TONO	PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate	Term of imary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract) / PIN No(s).	(Dth/d)	Charge(s)
Occide 1/	ental Energy Marketing, Inc. 1/	FTS	4/01/2008 1/ through 3/31/2011	10,000	1/

^{1/} This information is set out in the executed Transportation Rate Schedule FTS Agreement with corresponding Negotiated Rate Exhibit, which is currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 33F original Sheet No. 33F : Pending

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Eligible Primary Receipt	Rate Eligible	Term of Primary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract s) / PIN No(s).	(Dth/d)	Charge(s)
MidAmerio	can Energy Company 1/	DSS	12/01/2007 1/ through 4/30/2012	200,000	1/

⁻⁻⁻⁻⁻

^{1/} This information is set out in the executed Storage Rate Schedule DSS Agreement, as amended by Amendment No. 8 with corresponding Negotiated Rate Exhibit, which is currently on file with the FERC.

FERC Docket: RP10-674-000

First Revised Sheet No. 33G First Revised Sheet No. 33G

Superseding: Original Sheet No. 33G

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

.____

Commodity	Primary Receipt Shipper Name	Rate Primary Schedule	Term of y Delivery Contract	Volume (Dth/d)	Reservation Charge(s)	
Charge(s)	Point(s) / PIN No(s).		/ PIN No(s).	(DCII/ d)		
Interstate 6/,	Power and Light Company 1/7/, 9/	8/, 9/	5/01/2007 through	3/	4/, 6/, 10/	5/,
Interstate 10/	Power and Light Company 1/7/, 9/		11/02/2007 through 4/30/2013	42,314 3/	4/, 6/, 10/	6/,

- 1/ The Service Agreements with Shipper do not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ Negotiated Rate Terms. The negotiated rates set forth herein shall be effective for the respective term of each of the corresponding Service Agreements. For each Service Agreement, the applicable negotiated rate term is set forth in Footnote 3 below.
- 3/ Eligible Service Agreements. The negotiated rates set forth herein shall apply to firm transportation and DSS storage service provided by Natural to Shipper under the corresponding service agreements set forth below ("Service Agreements").
- (a) Transportation Agreements. The Transportation Agreements under which Natural provides service to Shipper shall have the respective terms and respective contract Maximum Daily Quantities (MDQs) set forth below:

Term

4/1	/2000 1 - 11	/1 /00	1 - 4/1/1/		5/1/07 to	11/1/07 to	11/2/07 to	4/1/08 to	11/1/08
to 4/1/ 3/31/2009	/2009 to 11, 9 10/31/2					11/01/2007	3/31/2008	10/31/2008	
			 FTS Agreer	nent No.	MDQ (D	th/day)			
1	N/A	N/A	1031	 76 /a	3,000	3,000	N/A	N/A	N/A
-	,	-11/	1065		12,000	13,455	13,455	12,000	
13,455	12,000		13,455 1103		4,918	4,918	4,918	4,918	
4,918	4,918		4,918						
6 , 700	4,700		11703 6,700		5,000	7,000	6 , 700	4,700	
0,700	4,700		11799			5,545	5,545		
5 , 545			5,545		27./2	10.000	10.000	10 000	
12,000	12,000		12770 12,000		N/A	12,000	12,000	12,000	
42,618	33,618		Total FTS 42,618	~	24,918	45,918	42,618	33,618	

FERC Docket: RP99-176-183

First Revised Sheet No. 33H First Revised Sheet No. 33H $\,$

Superseding: Original Sheet No. 33H

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt		Term of nary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point	Contract (s) / PIN No(s).	(Dth/d)	Charge(s)
	n Illinois Gas Company Nicor Gas 4/	FTS	1/	2/	3/
N/A	N/A	NSS	4/01/2009 N/A through 3/31/2013	219,262	\$3.07
N/A	N/A	DSS	4/01/2009 N/A through 3/31/2010	488,006	\$5.2084
N/A	N/A	DSS	4/01/2010 N/A through 3/31/2013	488,006	\$5.2084

^{1/} Negotiated Rate Term. The Negotiated Base Rates set forth herein shall apply to transportation service provided by Natural to Shipper under the service agreements

2/ Negotiated Rate Volume (MDQ/day)

2			MDQ (Dth/d) Off-Peak	MDQ (Dth/d)	MDQ (Dth/d)
MDQ (Dth/d)					
Peak Period			Period	Peak Period	Peak Period
March 1 -			April 1 -	November 1 -	December 1 -
	Rate	Agreement	through	through	through
through	Schedule	No.	October 31	November 30	February 28/29
March 31					
		100050	100.000	400.000	4.00.000
100,000	FTS	103253	100,000	100,000	100,000
145,368	FTS-SW	110208	115,368	145,368	145,368
•	FTS	110209	123,589	123,589	123,589
123,589	FTS	110212	15,142	15,142	15,142
15,142	FTS	110213	64,620	64,620	64,620
64,620			·	•	•
219,262	FTS-NB	110214	219,262	219,262	219,262
129,884	FTS	110215	19,884	129,884	129,884
123,004	FTS	132346			70,000

identified in footnote 2, which shall each have a primary term from April 1, 2009 through March 31, 2013, with the exception of FTS Agreement No. 132346, which shall have a primary term from March 1, 2009 through February 28, 2013, and the respective Maximum

Daily Quantities (MDQs) set forth in footnote 2 below.

Total FTS MDQ 657,865 797,865 867,865

FERC Docket: RP10-533-000

First Revised Sheet No. 331 First Revised Sheet No. 331 Superseding: Original Sheet No. 331

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Police Provide	Rate	Term of	Volume	Reservation	Commodity
Shipper Name	imary Delivery Schedule (s) / PIN No(s).	Contract	(Dth/d)	Charge(s)	Charge(s)
Enbridge Marketing (U.S.) L.P. 1/	NSS 1/	4/01/2007	24,067	1/	1/
		through 3/31/2013			
Enbridge Marketing (U.S.) L.P. 1/	NSS 1/	4/01/2007	10,000	1/	1/
		through 3/31/2013			
Occidental Energy Marketing, Inc. 1/	NSS 1/	4/01/2007	26 , 667	1/	1/
1/	1/	through 3/31/2013			

^{1/} This information is set out in the executed Rate Schedule NSS Agreement and corresponding Negotiated Rate Agreement, which are currently on file with the FERC.

FERC Docket: RP10-535-000

First Revised Sheet No. 33J First Revised Sheet No. 33J Superseding: Original Sheet No. 33J

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primar	Term of v Delivery	Volume	Reservation
-	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).		
Nicor Ga	as Company	FTS	4/01/2010	75 , 926	1/
1/	1/	1/	through 3/31/2014		

^{1/} This information is set out in the executed Transportation Rate Schedule FTS Agreement, and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-319-000

 $\textbf{Original Sheet No. 33K} \; \texttt{Original Sheet No. 33K} \; : \; \; \texttt{Pending}$

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt hipper Name Point(s) / PIN No(s).	Schedule	Term of mary Delivery Contract (s) / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)	
Occidenta	l Energy Marketing, Inc. 2/	NSS 2	4/01/2007 1/ / through 3/31/2012	26,667	2/	2/
Kinder Mo 2/	rgan Tejas Pipeline, L.P. 2/	NSS	4/01/2007 1/ 2/ through 3/31/2012	27,332	2/	
Kinder Mo 2/	rgan Tejas Pipeline, L.P. 2/	NSS	4/01/2007 1/ 2/ through 3/31/2012	70,000	2/	
Enserco E 2/	nergy Inc. 2/	NSS	4/01/2007 1/ 2/ through 3/31/2012	13,334	2/	

⁻⁻⁻⁻⁻

^{2/} This information is set out in the executed Storage Rate Schedule NSS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the

FERC Docket: RP08-319-000

Original Sheet No. 33L original Sheet No. 33L : Pending

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt Shipper Name		Term of ary Delivery Contract	Volume (Dth/d)	Reservation Charge(s)	
Charge(s)				(- , - ,	5-(-,	
				-		
CenterPoint	Energy Minnesota Gas	FTS-NB 2/	5/01/2007	270,000	2/	2/
		,	through 4/30/2019			
CenterPoint	Energy Minnesota Gas 1/	NSS 3/	5/01/2007	47,268	\$3.07 3/	3/
			through 4/30/2019			
CenterPoint	Energy Minnesota Gas 1/	NSS 3/	5/01/2007	47,268	\$3.07 3/	3/
			through 4/30/2019			
CenterPoint	Energy Minnesota Gas 1/	NSS 3/	5/01/2007	47,268	\$3.07 3/	3/
			through 4/30/2019			
CenterPoint	Energy Minnesota Gas 1/	NSS 3/	5/01/2007	29,098	\$3.07 3/	3/
	G,	σ,	through 4/30/2019			
CenterPoint	Energy Minnesota Gas 1/	NSS 3/	5/01/2007	29,098	\$3.07 3/	3/
			through 4/30/2019			

^{1/} The Storage Rate Schedule NSS Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas
Tariff.

^{2/} This information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Firm Transportation and Storage Negotiated Rate

Agreement, which are currently on file with the FERC. In addition to the Negotiated Rates, Shippers shall also pay Natural, pursuant to Section 6.4(b)(ii) of

Rate Schedule FTS, a monthly facilities charge of \$180,000/month beginning with the later of: (1) December 1, 2007, or (2) the date on which the constructed

facilities have been completed, tested and are available for service, and continuing over the term of the Agreement.

^{3/} This information is set out in the executed Firm Transportation and Storage Negotiated Rate Agreement, which is currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 33N Original Sheet No. 33N : Pending

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation
Commodity	Eligible Primary Receipt		rimary Delivery	(D) 1- (-1)	G1 (-)
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(s	Contract) / PIN No(s).	(Dth/d)	Charge(s)
		 		-	
FPLE For	ney, L.P. 1/	FTS (SW)	2/	300,000 9/	4/,5/,6/,7/
7/,8/			PIN No. 39772)	, , , , , , , , , , , , , , , , , , , ,	,, ., , , , ,

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

 $^{^{2/}}$ The term commences on August 1, 2003 (the "Effective Date") and to continue in effect for twenty (20) years; subject to Shipper's right to extend the term

of the FTS (SW) Agreement by an additional five (5) years effective as of the date falling twenty (20) years after the Effective Date and on the date

falling twenty five (25) years after the effective date, in either case upon at least twelve (12) months prior written notice.

^{3/} The Eligible Primary Receipt Points and MDQs are: Alliance/NGPL Grundy (PIN No. 37207), 20,000 Dth/d; Bridgeport Plant (PIN No. 901850), 27,739 Dth/d; Oneok/NGPL Custer (PIN No. 5388), 25,356 Dth/d; Enogex/NGPL (PIN No. 25686), 21,905 Dth/d;

Oneok/NGPL Custer (PIN No. 5388), 25,356 Dth/d; Enogex/NGPL (PIN No. 25686), 21,905 Dth/d; Enogex/NGPL (PIN No. 903474), 5,000 Dth/d; Stingray/NGPL (PIN No.

^{5433), 50,000} Dth/d; Enron/NGPL Refugio (PIN No. 6490); 75,000 Dth/d; KMTP/NGPL Katy (PIN No. 905231), 25,000 Dth/d; BP Prod/NGPL Red Oak (PIN No. 904567),

 $^{25,000\ \}text{Dth/d};\ \text{KMIGT}$ Pony Express (PIN No. 11165), $15,000\ \text{Dth/d};\ \text{Continent/NGPL}$ Panola (PIN No. 4904), $10,000\ \text{Dth/d}.$

^{4/} The Base Negotiated Reservation Rate shall be \$0.0883 per Dth of MDQ per day and shall be applicable to receipts at Eligible Primary Receipt Points with

deliveries to the Eligible Primary Delivery Point. The Base Negotiated Reservation Rate will be increased each contract year following the first contract

year, including any extensions of the term of the FTS (SW) Agreement on the anniversary of the Effective Date by a factor of 2.5%.

Effective Date: 11/01/2008 Status: Pending

FERC Docket: RP99-176-170

First Revised Sheet No. 33N First Revised Sheet No. 33N

Superseding: Original Sheet No. 33N

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Eligible Primary Receipt	Rate Eligible P	Term of rimary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract) / PIN No(s).	(Dth/d)	Charge(s)
	FOIR(S) / FIN NO(S).) / FIN NO(5).		
				· -	
FPLE For	ney, LLC 1/	FTS (SW)	2/	300,000 9/	4/,5/,6/,7/
7/,8/	3/	NGPL/KMNTPL (PIN No. 39772)		

The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The term commences on August 1, 2003 (the "Effective Date") and to continue in effect for twenty (20) years; subject to Shipper's right to extend the term

of the FTS (SW) Agreement by an additional five (5) years effective as of the date falling twenty (20) years after the Effective Date and on the date

falling twenty five (25) years after the effective date, in either case upon at least twelve (12) months prior written notice.

The Eligible Primary Receipt Points and MDQs are: Alliance/NGPL Grundy (PIN No. 37207), 20,000

Dth/d; Bridgeport Plant (PIN No. 901850), 27,739 Dth/d;
Oneok/NGPL Custer (PIN No. 5388), 25,356 Dth/d; Enogex/NGPL (PIN No. 25686), 21,905 Dth/d;
Enogex/NGPL (PIN No. 903474), 5,000 Dth/d; Stingray/NGPL (PIN No.

^{5433), 50,000} Dth/d; Enron/NGPL Refugio (PIN No. 6490); 75,000 Dth/d; KMTP/NGPL Katy (PIN No. 905231), 25,000 Dth/d; BP Prod/NGPL Red Oak (PIN No. 904567),

^{25,000} Dth/d; KMIGT Pony Express (PIN No. 11165), 15,000 Dth/d; Continent/NGPL Panola (PIN No. 4904), 10,000 Dth/d.

The Base Negotiated Reservation Rate shall be \$0.0883 per Dth of MDQ per day and shall be applicable to receipts at Eligible Primary Receipt Points with

deliveries to the Eligible Primary Delivery Point. The Base Negotiated Reservation Rate will be increased each contract year following the first contract

year, including any extensions of the term of the FTS (SW) Agreement on the anniversary of the Effective Date by a factor of 2.5%.

FERC Docket: RP99-176-191

First Revised Sheet No. 330 First Revised Sheet No. 330

Superseding: Original Sheet No. 330

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity Primary Receipt	Rate Prima	Term of ry Delivery	Volume	Reservation	
Shipper Name Charge(s) Point(s) / PIN No(s).		Contract) / PIN No(s).	(Dth/d)	Charge(s)	
Black Hills Utility Holdings, In 2/	c. DSS	11/01/2008	1,000	2/	2/
(formerly Aquila, Inc., d/b/a Aquila Networks) 1/		through 10/31/2013			
Black Hills Utility Holdings, In 2/	c. FTS 2/	11/01/2008	22,500	2/	2/
(formerly Aquila, Inc., d/b/a Aquila Networks) 1/		through 10/31/2013			
Black Hills Utility Holdings, In	c. FTS 2/	11/01/2008	12,000	2/	2/
(formerly Aquila, Inc., d/b/a Aquila Networks)		through 10/31/2013			
Black Hills Utility Holdings, In	c. FTS	4/01/2009	50,000 3/	2/	2/
(formerly Aquila, Inc., d/b/a Aquila Networks) 1/	_,	through 10/31/2013			

⁻⁻⁻⁻⁻

^{1/} The Storage Rate Schedule DSS Agreement and Transportation Rate Schedule FTS Agreement, as amended, with Shipper do not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} This information is set out in the executed Storage Rate Schedule DSS Agreement or Transportation Rate Schedule FTS Agreement, as amended, with corresponding Negotiated Rate exhibit, which are currently on file with the FERC.

^{3/} Eligible Firm Transportation Quantity: Fifty thousand (50,000) Dth/day for the period November 1 through March 31 of each contract year, and zero (0) for the period April 1 through October 31 of each contract year.

FERC Docket: RP99-176-181

Original Sheet No. 33P Original Sheet No. 33P

STATEMENT OF NEGOTIATED RATE TRANSACTIONS
PURSUANT TO SECTION 49

Commodity Primary Receipt Shipper Name Charge(s) Point(s) / PIN No(s).	Rate Term of Primary Delivery Schedule Contract Point(s) / PIN No(s)	Volume (Dth/d)	Reservation Charge(s)	
Iberdrola Renewables, Inc. (formerly 5/	NSS 1/	47,666 2/	\$5.9375	4/,
PPM Energy, Inc.)	•,		3/, 5/	
CIMA Energy, Ltd.	NSS 1/	6,667 2/	\$6.5600	4/,
5/ 6/	6/		3/, 5/	
TOTAL Gas & Power North America, Inc.		6,667 2/	\$6.8750	4/,
5/ 6/	6/		3/, 5/	

1/ Negotiated Rate Term. The negotiated rates set forth herein shall be in force during a term commencing on the date the 2009 Storage Expansion Project,

authorized in Docket No. CP08-32, is ready to receive firm service ("Effective Date") which is anticipated to be March 1, 2009. The Negotiated Rate Term shall

end five (5) years and one (1) month from the Effective Date, except for Iberdrola Renewables, Inc. (formerly PPM Energy, Inc.), which shall end ten (10) years and one (1) month from the Effective Date.

2/ Storage Agreement. The Negotiated Rates set forth herein shall apply to storage service provided by Natural to Shipper under Rate Schedule NSS with a Maximum

Storage Volume (MSV) of Five Hundred Thousand Twenty Five (500,025) Dth and an MDQ of Six Thousand Six Hundred Sixty Seven (6,667) Dth/day on the Gulf Coast

Leg 100% (the "Storage Agreement"), except for Iberdrola Renewables, Inc. (formerly PPM Energy, Inc.), which has a MSV of Three Million Five Hundred Seventy

Four Thousand Nine Hundred Fifty (3,574,950) Dth and an MDQ of Forth Seven Thousand Six Hundred Sixty Six (47,666) Dth/day on the Gulf Coast Leg 100%.

- 3/ Negotiated Monthly Base Reservation Rate. The Negotiated Monthly Reservation Rate per Dth of MDQ (Peak and Off-Peak periods) shall apply only to service
- provided by Natural to Shipper under the Rate Schedule NSS Agreement, subject to all other terms and conditions set forth herein.
- 4/ Negotiated Base Commodity Rate. Shipper shall not be charged any base commodity rate for service provided under the Storage Agreement.
 - 5/ Negotiated Rates.
- (a) General Negotiated Rate Limitations. The Negotiated Monthly Base Reservation Rate shall apply only to service provided to Shipper by Natural under the
- Storage Agreement during the Negotiated Rate Term, up to a maximum aggregate firm quantity equal to the respective Eligible Contract MDQ and the Eligible
- Contract MSV set forth herein, for all quantities stored on a firm basis under the Storage Agreement and any associated capacity release replacement
- agreements. For any aggregate quantities stored on a firm basis for Shipper and any associated capacity release replacement shippers which are in excess
- of the Eligible Contract MDQ or the Eligible Contract MSV, as applicable, Shipper shall be charged the greater of: (a) the 100% load factor equivalent of
- the Negotiated Monthly Base Reservation Rate; and (b) the 100% load factor equivalent of the applicable maximum monthly base reservation rate set forth in

Natural's FERC Gas Tariff, as may be revised from time to time.

FERC Docket: RP08-389-000

First Revised Sheet No. 33C.01 First Revised Sheet No. 33C.01

Superseding: Original Sheet No. 33C.01

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s)	Schedule	Contract	Volum (Dth/d		Reserva		_
	llips Company 1/ n Bridgeport Plant	FTS Gulf South Goodr	2/ (a)	10,000		\$6.6700	3/	
ConocoPhi	No. 901850) 4/ llips Company 1/ on Bridgeport Plant	(PIN No. 6297) 5 FTS Kansas Gas Servi	2/ (b)	5,000		\$1.5708	3/	
Occidenta	No. 901850) 4/	(PIN No. 3063) 5 nc. 1/ FTS-SW KMTP/NGPL	5/ 2/ (c)	41,000	\$6.0833	3/ 6/	3/	Pony
Express (PIN	No. 11165) 4/	(PIN No. 5579) 5	5/ 6/					

- 1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ Negotiated Rate Term. The Negotiated Rate Term shall commence on the date the Segment One Expansion and the Amarillo to Gulf Coast Expansion ("Expansion") is

ready to receive firm service ("Effective Date") which is anticipated to be April 15, 2006. For purposes of this section, the Expansion shall be considered

ready to receive firm service when the pipeline and compression are ready for firm service. This negotiated rate term shall end:

- (a) five (5) years and six (6) months from the Effective Date;
- (b) five (5) years and one month from the Effective Date; and (c) five (5) years from the Effective Date.
- Negotiated Rates.
- (a) Eligible Transportation Quantity. The Negotiated Rate shall apply to a total aggregate maximum daily transportation quantity of Dth/day as shown above,

for all quantities transported under the FTS Agreement and any associated capacity release replacement agreements (the "Eligible Transportation

Quantity"). Any quantities transported in excess of the Eligible Transportation Quantity shall be billed at the greater of the Negotiated Rate or the

applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff. FOOTNOTES APPLICABLE TO SHEET NO. 33C

3/ Negotiated Rates. (con't)

- The Negotiated Rate shall apply only to service provided to Shipper by Natural under the FTS Agreement from the Eligible Receipt Points to the Eligible
- Delivery Points during the Negotiated Rate Term. The Negotiated Rate shall apply only to a maximum daily firm transportation quantity equal to the

Eligible Transportation Quantity set forth above, for all quantities transported on a firm basis under the Transportation Agreement and any associated

capacity release replacement agreements. Any quantities transported on a firm basis in excess of the Eligible Transportation Quantity shall be billed at

the greater of the Negotiated Rate or all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised

from time to time.

- (c) Monthly Base Negotiated Reservation Rate ("Negotiated Rate"). The Negotiated Rate set forth above per Dth of MDQ for peak and off-peak periods is effective for and through the Negotiated Rate Term. The Negotiated Rate shall be fixed for the term of the FTS Agreement.
- (d) General Negotiated Rate Limitations. The Negotiated Rate shall apply only to: (i) service provided to Shipper by Natural under the Transportation

Agreement from the Eligible Receipt Points to the Eligible Delivery Points during the Negotiated Rate Term; and (ii) a maximum daily firm transportation

quantity equal to the Eligible Firm Transportation Quantity set forth in footnote 3(a) above, for all quantities transported on a firm basis under the

Transportation Agreement and any associated capacity release replacement agreements. Shipper shall be charged the greater of the Negotiated Rate and all

applicable rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time, for any aggregate quantities

transported on a firm basis for Shipper and any associated capacity release replacement shippers which: (i) are in excess of the Eligible Firm

- (e) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in
- Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payments to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

- (f) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein or agreed to in writing by Natural, the greater of the
- Negotiated Rate and all applicable maximum rates, charges, surcharges and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be

revised from time to time, shall apply to service provided to Shipper under the FTS Agreement, including without limitation all applicable: (i) Fuel and

Gas Lost and Unaccounted For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and

surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

4/ Eligible Receipt Points. The Negotiated Rates shall apply to transportation service provided from the following receipt point(s).

Primary Receipt Point. ConocoPhillips Company - Devon Bridgeport Plant, PIN No. 901850 (10,000 MDQ) and (5,000 MDQ), respectively;
Occidental Energy Marketing, Inc. - Pony Express, PIN No. 11165 (41,000 MDQ).

Secondary Receipt Points. The Negotiated Rates shall apply to service provided from all receipt points located in the following zones: ConocoPhillips (10,000

MDQ) - Texok and Midcontinent zones; ConocoPhillips (5,000 MDQ) - Midcontinent zone; Occidental - Permian, Midcontinent, Amarillo Mainline, and Texok zones.

FERC Docket: RP10-529-000

Third Revised Sheet No. 33D.01 Third Revised Sheet No. 33D.01

Superseding: Second Revised Sheet No. 33D.01

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

TRANSACTIONS

Commodity	Eligible Primary Receipt	Rate Term Eligible Primary De		Reservation
Charge(s)	Shipper Name	Schedule Contrac Point(s) / PIN N	ct (Dth/d)	Charge(s)
Oneok En 2/	ergy Services Company, L.P. 2/	1/ FTS 4/01/2 2/	35,000	2/
		throi 3/31/2	2	
Oneok En	ergy Services Company, L.P. 2/	1/ NSS 4/01/2	40,000 3/	2/
27	2/	throi 3/31/2		
Oneok En	ergy Services Company, L.P.	NSS 4/01/2	2010 62,569 4/	2/
۷/	2/	throi 3/31/2		

Sixty Nine (62,569) Dth/d on the Amarillo Leg 100% (the "Storage Agreement").

^{1/} The Transportation Rate Schedule FTS Agreement and Storage Rate Schedule NSS Agreement, as amended, with Shipper do not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} This information is set out in the executed Transportation Rate Schedule FTS Agreement or Storage Rate Schedule NSS Agreement, as amended, with corresponding Negotiated Rate Exhibit which are currently on file with the FERC.

^{3/} The Negotiated Rates set forth herein shall apply only to service provided to Shipper by Natural under the Rate Schedule NSS Agreement during the Negotiated Rate Term, up to a Maximum Storage Volume (MSV) of Three Million (3,000,000) Dth and a MDQ of Forty Thousand (40,000) Dth/d on the Amarillo Leg 100% (the "Storage Agreement").

^{4/} The Negotiated Rates set forth herein shall apply only to service provided to Shipper by Natural under the Rate Schedule NSS Agreement during the Negotiated

Rate Term, up to a MSV of Four Million Six Hundred Ninety Two Thousand Six Hundred Seventy Five (4,692,675) Dth and a MDQ of Sixty Two Thousand Five Hundred

FERC Docket: RP10-674-000

Second Revised Sheet No. 33G.01 Second Revised Sheet No. 33G.01

Superseding: First Revised Sheet No. 33G.01

FOOTNOTES APPLICABLE TO SHEET NO. 33G

3/ (con't)

(b) DSS Storage Agreements.

DSS Agreement No.	MDQ (Dth/day)	MSV (Dth)	Term
110349	8,300	415,000	11/2/07 - 4/30/2013
110357	34,014	1,700,700	5/1/07 - 4/30/2013
Total DSS	42,314	2,115,700	

4/ Negotiated Fixed Monthly Base Reservation Rates. For service provided under each of the Service Agreements, Shipper shall pay Natural the corresponding

Negotiated Fixed Monthly Base Reservation Rate set forth below:

Monthly Reservation Rate/Dth of MDQ

4/01/0	11 /01 /00	Service		11/02/07 to	5/01/08 to	7/01/08 to	11/01/08
to 4/01/0	9 to 11/01/09	to 4/01/10 Agreement No.		4/30/2008	6/30/2008	10/31/2008	
3/31/2009			10/31/2010	4/30/2008	0/30/2000	10/31/2000	
3/31/2009	10/31/2009	3/31/2010	10/31/2010	4/30/2013			
(a)	Transportation						
		103176	\$7.8744	N/A	N/A	N/A	N/A
N/A	N/A	N/A					
		106570	\$7.8744	\$7.8744	\$8.1092	\$8.4000	\$8.9899
\$8.4000	\$8.9899	\$8.400					
		110350	\$7.8744	\$7.8744	\$8.1092	\$8.4000	\$8.9899
\$8.4000	\$8.9899	\$8.400		45 0544	* 0 1000	** ***	**
20 4000	***	117037	\$7.8744	\$7.8744	\$8.1092	\$8.4000	\$8.9899
\$8.4000	\$8.9899	\$8.400		67 0744	60 1000	60 4000	¢0.0000
60 4000	\$8.9899	117996	\$7.8744	\$7.8744	\$8.1092	\$8.4000	\$8.9899
\$8.4000	\$8.9899	\$8.400		67 0744	ć0 1000	¢0 4000	\$8.9899
\$8.4000	\$8.9899	127768 \$8.400		\$7.8744	\$8.1092	\$8.4000	\$8.9899
20.4000	70.9099	20.400	00				
(b)	DSS Storage	110349	\$5.70	\$5.70	\$5.70	\$5.70	N/A
			\$5.70				
		110357	\$5.70	\$5.70	\$5.70	\$5.70	N/A
			\$5.70				

^{5/} Negotiated Fixed Base Commodity Rate. For service provided under each of the Transportation Agreements, Shipper shall pay Natural a Negotiated Fixed Base

Commodity Rate equal to the applicable maximum commodity rate set forth in Natural's FERC Gas Tariff for service under the relevant Transportation Agreement,

as such rate is in effect as of October 20, 2006.

FERC Docket: RP99-176-183

First Revised Sheet No. 33H.01 First Revised Sheet No. 33H.01

Superseding: Original Sheet No. 33H.01

FOOTNOTES APPLICABLE TO SHEET NO.

33H

3/ Detailed information on the level and applicability of the negotiated rates is set out in the executed Firm Transportation and Storage Negotiated Rate Agreement which is currently on file with the FERC.

4/ This information is set out in the amended Exhibits A and B for each of the associated service agreements which are currently on file with the FERC.

FERC Docket: RP99-176-165

First Revised Sheet No. 33J.01 First Revised Sheet No. 33J.01 Superseding: Original Sheet No. 33J.01

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate	Term of mary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract s) / PIN No(s).	(Dth/d)	Charge(s)
Tylex, I	nc. 1/	FTS	11/01/2008 1/ through 10/31/2018	2/	1/

2/ MDQ is as follows:

MDQ	Month	Year
3,500 Dth per day 4,500 Dth per day 5,000 Dth per day 4,000 Dth per day 3,000 Dth per day 2,000 Dth per day 3,000 Dth per day	November 1 through November 30 December 1 through December 31 January 1 through February 29 March 1 through March 31 April 1 through April 30 May 1 through September 30 October 1 through October 31	2008 through 2017 2008 through 2017 2009 through 2018 2009 through 2018 2009 through 2018 2009 through 2018 2009 through 2018

^{1/} This information is set out in the executed Transportation Rate Schedule FTS Agreement, and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP99-176-175

First Revised Sheet No. 33L.01 First Revised Sheet No. 33L.01

Superseding: Original Sheet No. 33L.01

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s)		Schedule	Term of y Delivery Contract / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)	
Centerpoint	Energy Services Inc.	1/	FTS 2/	12/01/2008 through 11/30/2011	2/	2/	2/

^{1/} The Service Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

 $^{2/\ {\}rm This}$ information is set out in the executed Firm Transportation Negotiated Agreement which is currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 33N.01 Original Sheet No. 33N.01: Pending

FOOTNOTES APPLICABLE TO SHEET NO. 33N

5/ In addition to the Base Negotiated Reservation Rate, Incremental Daily Volumetric Reservation Rates will apply to all quantities transported on any day

wherein gas is received at eligible primary and secondary receipt points and/or delivered at eligible primary and secondary delivery points, based on the following matrices.

Secondary	Primary
Receipt Point Zone	Delivery Point
Permian Zone	\$0.07
Amarillo Mainline Zone	\$0.04
Gulf Coast Mainline Zone	\$0.04

Secondary Delivery Point Zone (Including

Pools but Excluding Industrial End Users)

Amarillo Iowa-Illinois Iowa-Illinois Gulf

7 IIII G L L L L	10 10wa 111111015	iowa ililinois	5 041	- 1		
	ne Nicor, Peoples & NIPSCO Only (Peak) NIPSC	Midcontinent	Permian	Texok	Amarillo Mainline	IES Only
				(Excluding Lone	(Excluding IES @	Base +
	Base + Incr. Rates Primary Receipt Point Zone			Star @ \$0.04)	\$0.13 for Off-Peak) Rates
1	Midcontinent (Other than Segment 1)	\$0.05	\$0.09	\$0.02	\$0.07	\$0.35
ŗ	\$0.30 Texok - Segment 16		\$0.07 \$0.09	\$0.02	\$0.07	\$0.35
Ş	\$0.30 South Texas \$0.30	\$0.08 \$0.05 \$0.08	\$0.07 \$0.09 \$0.07	\$0.02	\$0.07	\$0.35
I	Alliance (Iowa-Illinois) \$0.30	\$0.08 \$0.05 \$0.08	\$0.07 \$0.09 \$0.07	\$0.02	\$0.07	\$0.35
I	Midcontinent (Segment 1) \$0.30	\$0.05 \$0.08	\$0.09 \$0.07	\$0.02	\$0.07	\$0.35
I	Louisiana \$0.30	\$0.05 \$0.08	\$0.09 \$0.07	\$0.02	\$0.07	\$0.35

FERC Docket: RP99-176-181

Original Sheet No. 33P.01 Original Sheet No. 33P.01

FOOTNOTES FOR SHEET NO. 363P

5/ Negotiated Rates. (CON'T)

(b) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in

Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to

Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payment to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

(c) Additional Rates, Charges and Surcharges. In addition to the Negotiated Monthly Base Reservation Rate, and unless otherwise expressly provided herein or

agreed to in writing by Natural, Shipper shall also pay Natural all other applicable maximum rates, charges, surcharges and penalties of any nature set

forth in Natural's FERC Gas Tariff, as may be revised from time to time, including without limitation all applicable maximum: (i) Storage Fuel chargers;

(ii) authorized and unauthorized overrun charges; and (iii) sourcing charges. Notwithstanding the foregoing sentence, to the extent that any such

additional maximum rate, charge, surcharge, or penalty is derived from the applicable base reservation rate, then Natural shall derive such additional

rate, charge, surcharge, or penalty utilizing the greater of: (i) the Negotiated Monthly Base Reservation Rate; and (ii) the applicable maximum base

reservation rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

6/ Receipt and Delivery Points. The Receipt Points for injection and the Delivery Points for withdrawal under the Rate Schedule NSS Agreement shall be the

Storage Points as defined in Section 9.3 of the General Terms and Conditions set forth in Natural's FERC Gas Tariff.

FERC Docket: RP08-389-000

First Revised Sheet No. 33C.02 First Revised Sheet No. 33C.02

Superseding: Original Sheet No. 33C.02

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt Shipper Name	Rate Primary D Schedule	Contract	Volume (Dth/d)		
Charge(s)	Point(s) / PIN No(s)	. Point(s) /	PIN NO(S).			
	llips Company 1/ n Bridgeport Plant	FTS Gulf South Goodr	2/ (a)	10,000	\$6.6700	3/
(PIN	No. 901850) 4/	(PIN No. 6297) 5	/			
	llips Company 1/ n Bridgeport Plant	FTS Kansas Gas Servi	2/ (b)	5,000	\$1.5708	3/
(PIN	No. 901850) 4/	(PIN No. 3063) 5	/			
Occidenta Express	l Energy Marketing, Ind	c. 1/ FTS-SW KMTP/NGPL	2/ (c)	41,000	\$6.0833 3/ 6/	3/ Pony
(PIN	No. 11165) 4/	(PIN No. 5579) 5	/ 6/			

- 1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ Negotiated Rate Term. The Negotiated Rate Term shall commence on the date the Segment One Expansion and the Amarillo to Gulf Coast Expansion ("Expansion") is

ready to receive firm service ("Effective Date") which is anticipated to be April 15, 2006. For purposes of this section, the Expansion shall be considered

ready to receive firm service when the pipeline and compression are ready for firm service. This negotiated rate term shall end:

- (a) five (5) years and six (6) months from the Effective Date;
- (b) five (5) years and one month from the Effective Date; and (c) five (5) years from the Effective Date.
- Negotiated Rates.
- (a) Eligible Transportation Quantity. The Negotiated Rate shall apply to a total aggregate maximum daily transportation quantity of Dth/day as shown above,

for all quantities transported under the FTS Agreement and any associated capacity release replacement agreements (the "Eligible Transportation

Quantity"). Any quantities transported in excess of the Eligible Transportation Quantity shall be billed at the greater of the Negotiated Rate or the

> applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff. FOOTNOTES APPLICABLE TO SHEET NO. 33C

- 3/ Negotiated Rates. (con't)
- The Negotiated Rate shall apply only to service provided to Shipper by Natural under the FTS Agreement from the Eligible Receipt Points to the Eligible

Delivery Points during the Negotiated Rate Term. The Negotiated Rate shall apply only to a $\mbox{{\tt maximum}}$ daily firm transportation quantity equal to the

Eligible Transportation Quantity set forth above, for all quantities transported on a firm basis under the Transportation Agreement and any associated

capacity release replacement agreements. Any quantities transported on a firm basis in excess of the Eligible Transportation Quantity shall be billed at

the greater of the Negotiated Rate or all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised

from time to time.

(c) Monthly Base Negotiated Reservation Rate ("Negotiated Rate"). The Negotiated Rate set forth above per Dth of MDQ for peak and off-peak periods is effective for and through the Negotiated Rate Term. The Negotiated Rate shall be fixed for the term of the FTS Agreement.

(d) General Negotiated Rate Limitations. The Negotiated Rate shall apply only to: (i) service provided to Shipper by Natural under the Transportation

 $\hbox{Agreement from the Eligible Receipt Points to the Eligible Delivery Points during the Negotiated Rate Term; and (ii) a maximum daily firm transportation }$

quantity equal to the Eligible Firm Transportation Quantity set forth in footnote 3(a) above, for all quantities transported on a firm basis under the

Transportation Agreement and any associated capacity release replacement agreements. Shipper shall be charged the greater of the Negotiated Rate and all

applicable rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time, for any aggregate quantities

transported on a firm basis for Shipper and any associated capacity release replacement shippers which: (i) are in excess of the Eligible Firm

(e) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in

Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payments to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

(f) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein or agreed to in writing by Natural, the greater of the

Negotiated Rate and all applicable maximum rates, charges, surcharges and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be

revised from time to time, shall apply to service provided to Shipper under the FTS Agreement, including without limitation all applicable: (i) Fuel and

Gas Lost and Unaccounted For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and

surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

4/ Eligible Receipt Points. The Negotiated Rates shall apply to transportation service provided from the following receipt point(s).

Primary Receipt Point. ConocoPhillips Company - Devon Bridgeport Plant, PIN No. 901850 (10,000 MDQ) and (5,000 MDQ), respectively;
Occidental Energy Marketing, Inc. - Pony Express, PIN No. 11165 (41,000 MDQ).

Secondary Receipt Points. The Negotiated Rates shall apply to service provided from all receipt points located in the following zones: ConocoPhillips (10,000

MDQ) - Texok and Midcontinent zones; ConocoPhillips (5,000 MDQ) - Midcontinent zone; Occidental - Permian, Midcontinent, Amarillo Mainline, and Texok zones.

FOOTNOTES APPLICABLE TO SHEET NO. 33C

5/ Eligible Delivery Points. The Negotiated Rates shall apply to transportation service provided to the following delivery point(s):

Primary Delivery Points: ConocoPhillips - Gulf South Goodrich, PIN 6297 (10,000 MDQ); ConocoPhillips - Kansas Gas Service, PIN 3063 (5,000 MDQ); Occidental - KMTP/NGPL, PIN 5579 (41,000 MDQ).

Secondary Delivery Points: The Negotiated Rates shall apply to service provided to the following secondary delivery points:

ConocoPhillips - (10,000 MDQ) - All pooling, storage, Hinshaw and Intrastate
Pipeline Interconnects and Interstate Interconnect points
located within the zones traversed by the primary path of the Transportation
Agreement, excluding the Lone Star Pipeline Interconnect located in
the Midcontinent Zone and the Kinder Morgan Texas Pipeline Interconnect located in
the Texok Zone;

ConocoPhillips - (5,000 MDQ) - All pooling, storage, Hinshaw and Intrastate Pipeline Interconnects and Interstate Interconnect points located

within the zones traversed by the primary path of the Transportation Agreement, excluding the Lone Star Pipeline Interconnect located in the Midcontinent Zone;

Occidental - Subject to the provisions in footnote 6 below regarding incremental rate secondary delivery points, the Negotiated Rate shall apply to all delivery points located in the Midcontinent and Texok zones.

6/ Incremental Rate Secondary Delivery Points. In addition to the Negotiated Rate, Shipper shall pay Natural an incremental volumetric daily reservation rate for

all quantities transported on a firm basis on any day to the Incremental Rate Secondary Delivery Points set forth herein for service to such point.

The following incremental volumetric daily reservation rates will apply to firm service provided to the secondary delivery points set forth below under the corresponding Point Tier:

Tier 1: Rate: \$0.030/Dth; Secondary Delivery Points: All points located in the LA zone.

Tier 2: Rate: \$0.035/Dth (Off-Peak) and \$0.0625/Dth (Peak); Secondary Delivery Points: Nicor (PIN No. 9258), NIPSCO (PIN No. 909260), PGL&C (PIN No. 909285),

MidAmerican (PIN No. 10568), and all existing interstate pipeline interconnects in the Market Delivery Zone, excluding MRT, Illini Carrier, Guardian and ANR Joliet.

Service provided on a firm basis to all other secondary delivery points not specifically set forth in this footnote 6 or in footnote 5, above, regarding

Secondary Delivery Points, will be provided at the greater of the Negotiated Rate or the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may

FERC Docket: RP08-319-000

Original Sheet No. 33G.02 original Sheet No. 33G.02 : Pending

FOOTNOTE APPLICABLE TO SHEET NO. 33G

6/ (a) General Negotiated Rate Limitations. The Negotiated Rates shall apply only to: (i) service provided to Shipper by Natural under any given Service

Agreement from the applicable Eligible and DSS Storage Receipt Points to the applicable Eligible Delivery Points during the applicable Negotiated Rate

Term; and (ii) a maximum daily firm transportation quantity equal to the applicable Eligible Firm Transportation and DSS Storage Quantity set forth in

footnote 3 above, for all quantities stored and/or transported on a firm basis under the relevant Service Agreement and any associated capacity release

replacement agreements. For any aggregate quantities stored and/or transported on a firm basis on any day for Shipper and any associated capacity release $\frac{1}{2}$

replacement shippers which: (i) are in excess of the Eligible Firm Transportation and DSS Storage Quantity, or (ii) involve any receipt or delivery

points which are not Eligible Receipt or Delivery Points under the relevant Service Agreement, Shipper shall pay Natural base reservation and commodity

rates which are equal to the greater of: (a) the 100% load factor equivalent of the relevant Negotiated Fixed Monthly Base Reservation Rate and

Negotiated Fixed Base Commodity Rate; and (b) the 100% load factor equivalent of the applicable maximum base reservation and commodity rates set forth in Natural's Gas Tariff, as may be revised from time to time.

(b) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in

Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payment to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

(c) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein or agreed to in writing by Natural, in addition to the

Negotiated Rates, Shipper shall pay Natural all applicable maximum rates, charges, surcharges and penalties of any nature set forth in Natural's FERC Gas

Tariff, as may be revised from time to time, for service provided to Shipper under the Service Agreement, including without limitation all applicable:

- (i) Fuel charges (including a Storage Fuel Charge and/or a Transportation Fuel Charge comprised of a Fuel Gas Charge and a Gas Lost and Unaccounted For
- Charge; (ii) authorized and unauthorized overrun charges and/or inventory cycling charges; (iii) sourcing charges; (iv) balancing service charges; (v)

reservation surcharges; (vi) commodity rates, charges and surcharges; (vii) ACA surcharges; and (viii) gathering charges, offshore charges and lateral

line charges. Notwithstanding the foregoing sentence, to the extent that any such additional maximum rate, charge, surcharge or penalty is derived from

the Negotiated Fixed Monthly Base Reservation Rate and/or the Negotiated Fixed Monthly Base Commodity Rate applicable to service provided under the

relevant Service Agreement, then Natural shall derive such additional rate, charge, surcharge, or penalty utilizing the greater of: (a) the maximum base

reservation rate or maximum base commodity rate set forth in Natural's FERC Gas Tariff, as revised from time to time, which is applicable to the relevant

Service Agreement, and (b) the applicable Negotiated Fixed Monthly Base Reservation Rate or Negotiated Fixed Monthly Base Commodity Rate which is

applicable to the relevant Service Agreement.

- 7/ Eligible Receipt Points.
- (a) Shipper's initial eligible primary receipt points and respective initial associated point MDQs shall be as set forth below in Footnote 9 (for each

Transportation Agreement, the applicable "Eligible Primary Receipt Points"). The Negotiated Rates shall only apply to service provided under each of the

Transportation Agreements from the applicable Eligible Primary Receipt Points, subject to the respective associated primary receipt point MDQ limitations

(the "Eligible Receipt Point MDQs"), and from the secondary receipt points permitted under (b) below (the "Eligible Secondary Receipt Points"), with such

eligible primary and secondary receipt points collectively referred to as the "Eligible Receipt Points".

FERC Docket: RP99-176-192

First Revised Sheet No. 33H.02 First Revised Sheet No. 33H.02 Superseding: Original Sheet No. 33H.02

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

	Rate	Term of	Volume	Reservation	Commodity
Primary Receipt Shipper Name Point(s) / PIN No(s).	Primary Delivery Schedule Point(s) / PIN No(s).	Contract	(Dth/d)	Charge(s)	Charge(s)
Nicor Gas Company	FTS	4/01/2009	1/	1/	1/
		through 3/31/2013			

^{1/} This information is set out in the executed Rate Schedule FTS Agreement, and corresponding Negotiated Rate Exhibit, as amended, which are currently on file with the FERC.

FERC Docket: RP08-389-000

First Revised Sheet No. 33J.02 First Revised Sheet No. 33J.02

Superseding: Original Sheet No. 33J.02

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s)	Schedule	Contract	Volum (Dth/d		Reserva		_
	llips Company 1/ n Bridgeport Plant	FTS Gulf South Goodr	2/ (a)	10,000		\$6.6700	3/	
ConocoPhi	No. 901850) 4/ llips Company 1/ on Bridgeport Plant	(PIN No. 6297) 5 FTS Kansas Gas Servi	2/ (b)	5,000		\$1.5708	3/	
Occidenta	No. 901850) 4/	(PIN No. 3063) 5 nc. 1/ FTS-SW KMTP/NGPL	5/ 2/ (c)	41,000	\$6.0833	3/ 6/	3/	Pony
Express (PIN	No. 11165) 4/	(PIN No. 5579) 5	5/ 6/					

- 1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ Negotiated Rate Term. The Negotiated Rate Term shall commence on the date the Segment One Expansion and the Amarillo to Gulf Coast Expansion ("Expansion") is

ready to receive firm service ("Effective Date") which is anticipated to be April 15, 2006. For purposes of this section, the Expansion shall be considered

ready to receive firm service when the pipeline and compression are ready for firm service. This negotiated rate term shall end:

- (a) five (5) years and six (6) months from the Effective Date;
- (b) five (5) years and one month from the Effective Date; and (c) five (5) years from the Effective Date.
- Negotiated Rates.
- (a) Eligible Transportation Quantity. The Negotiated Rate shall apply to a total aggregate maximum daily transportation quantity of Dth/day as shown above,

for all quantities transported under the FTS Agreement and any associated capacity release replacement agreements (the "Eligible Transportation

Quantity"). Any quantities transported in excess of the Eligible Transportation Quantity shall be billed at the greater of the Negotiated Rate or the

> applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff. FOOTNOTES APPLICABLE TO SHEET NO. 33C

- 3/ Negotiated Rates. (con't)
- The Negotiated Rate shall apply only to service provided to Shipper by Natural under the FTS Agreement from the Eligible Receipt Points to the Eligible

Delivery Points during the Negotiated Rate Term. The Negotiated Rate shall apply only to a $\mbox{{\tt maximum}}$ daily firm transportation quantity equal to the

Eligible Transportation Quantity set forth above, for all quantities transported on a firm basis under the Transportation Agreement and any associated

capacity release replacement agreements. Any quantities transported on a firm basis in excess of the Eligible Transportation Quantity shall be billed at

the greater of the Negotiated Rate or all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised

from time to time.

(c) Monthly Base Negotiated Reservation Rate ("Negotiated Rate"). The Negotiated Rate set forth above per Dth of MDQ for peak and off-peak periods is effective for and through the Negotiated Rate Term. The Negotiated Rate shall be fixed for the term of the FTS Agreement.

(d) General Negotiated Rate Limitations. The Negotiated Rate shall apply only to: (i) service provided to Shipper by Natural under the Transportation

Agreement from the Eligible Receipt Points to the Eligible Delivery Points during the Negotiated Rate Term; and (ii) a maximum daily firm transportation

quantity equal to the Eligible Firm Transportation Quantity set forth in footnote 3(a) above, for all quantities transported on a firm basis under the

Transportation Agreement and any associated capacity release replacement agreements. Shipper shall be charged the greater of the Negotiated Rate and all

applicable rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time, for any aggregate quantities

transported on a firm basis for Shipper and any associated capacity release replacement shippers which: (i) are in excess of the Eligible Firm

(e) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in

Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payments to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

(f) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein or agreed to in writing by Natural, the greater of the

Negotiated Rate and all applicable maximum rates, charges, surcharges and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be

revised from time to time, shall apply to service provided to Shipper under the FTS Agreement, including without limitation all applicable: (i) Fuel and

Gas Lost and Unaccounted For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and

surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

4/ Eligible Receipt Points. The Negotiated Rates shall apply to transportation service provided from the following receipt point(s).

Primary Receipt Point. ConocoPhillips Company - Devon Bridgeport Plant, PIN No. 901850 (10,000 MDQ) and (5,000 MDQ), respectively;
Occidental Energy Marketing, Inc. - Pony Express, PIN No. 11165 (41,000 MDQ).

Secondary Receipt Points. The Negotiated Rates shall apply to service provided from all receipt points located in the following zones: ConocoPhillips (10,000

MDQ) - Texok and Midcontinent zones; ConocoPhillips (5,000 MDQ) - Midcontinent zone; Occidental - Permian, Midcontinent, Amarillo Mainline, and Texok zones.

FOOTNOTES APPLICABLE TO SHEET NO. 33C

5/ Eligible Delivery Points. The Negotiated Rates shall apply to transportation service provided to the following delivery point(s):

Primary Delivery Points: ConocoPhillips - Gulf South Goodrich, PIN 6297 (10,000 MDQ); ConocoPhillips - Kansas Gas Service, PIN 3063 (5,000 MDQ); Occidental - KMTP/NGPL, PIN 5579 (41,000 MDQ).

Secondary Delivery Points: The Negotiated Rates shall apply to service provided to the following secondary delivery points:

ConocoPhillips - (10,000 MDQ) - All pooling, storage, Hinshaw and Intrastate
Pipeline Interconnects and Interstate Interconnect points
located within the zones traversed by the primary path of the Transportation
Agreement, excluding the Lone Star Pipeline Interconnect located in
the Midcontinent Zone and the Kinder Morgan Texas Pipeline Interconnect located in
the Texok Zone;

 ${\tt ConocoPhillips - (5,000 MDQ) - All pooling, storage, Hinshaw and Intrastate Pipeline Interconnects and Interstate Interconnect points located}$

within the zones traversed by the primary path of the Transportation Agreement, excluding the Lone Star Pipeline Interconnect located in the Midcontinent Zone;

Occidental - Subject to the provisions in footnote 6 below regarding incremental rate secondary delivery points, the Negotiated Rate shall apply to all delivery points located in the Midcontinent and Texok zones.

6/ Incremental Rate Secondary Delivery Points. In addition to the Negotiated Rate, Shipper shall pay Natural an incremental volumetric daily reservation rate for

all quantities transported on a firm basis on any day to the Incremental Rate Secondary Delivery Points set forth herein for service to such point.

The following incremental volumetric daily reservation rates will apply to firm service provided to the secondary delivery points set forth below under the corresponding Point Tier:

Tier 1: Rate: \$0.030/Dth; Secondary Delivery Points: All points located in the LA zone.

Tier 2: Rate: \$0.035/Dth (Off-Peak) and \$0.0625/Dth (Peak); Secondary Delivery Points: Nicor (PIN No. 9258), NIPSCO (PIN No. 909260), PGL&C (PIN No. 909285),

MidAmerican (PIN No. 10568), and all existing interstate pipeline interconnects in the Market Delivery Zone, excluding MRT, Illini Carrier,

Guardian and ANR Joliet.

Service provided on a firm basis to all other secondary delivery points not specifically set forth in this footnote 6 or in footnote 5, above, regarding

Secondary Delivery Points, will be provided at the greater of the Negotiated Rate or the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may

be revised from time to time, for service to such points.

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Prima	Term of ry Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract) / PIN No(s).	(Dth/d)	Charge(s)
Nicor Gas 2/	Company 1/ 2/	FTS 2	•	40,000	2/
			through 3/31/2011		
Nicor Gas	Company 1/	FTS 2	4/01/2008	20,000	2/
			through 3/31/2011		

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the

applicable Form of Service Agreement in Natural's FERC Gas Tariff.

2/ This information is set out in the executed Transportation Rate Schedule FTS Agreement, and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP99-176-173

Original Sheet No. 33L.02 Original Sheet No. 33L.02

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Prima	Term of ry Delivery	Volume	Reservation	
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(s	Contract) / PIN No(s).	(Dth/d)	Charge(s)	
Cargill En	ergy Services 1/	FTS	12/01/2008	4,900	2/	2/
	2/	2/	through 11/30/2011			
Cargill En	ergy Services 1/	FTS 2/	12/01/2008	9,025	2/	2/
			through 11/30/2011			

^{1/} The Service Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

 $^{2/\ \}mbox{This}$ information is set out in the executed Firm Transportation Negotiated Agreement which is currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 33N.02 Original Sheet No. 33N.02 : Pending

FOOTNOTES APPLICABLE TO SHEET NO. 33N

5/ Con't

Secondary Delivery

		-	 	 •	_	\sim		_					
_	 		 _	 _	_	_	_	_	_	_	_	_	_

Iowa-Illinois	Iowa-Illinois Midcontinent	Gulf Permian	Texok	Amarillo Mainline
Nicor, Peoples &	Nicor, Peoples &	Coast		
,	Pool	Pool	Pool	Pool
NIPSCO Only (Peak)	NIPSCO Only (Off Peak)	Mainline		
Base + Incr. Rates Secondary Receipt Po	int Zone			
Permian	\$0.05	\$0.05	\$0.05	\$0.05
\$0.30	\$0.08	\$0.05		
Amarillo Mainline	\$0.05	\$0.05	\$0.05	\$0.05
\$0.30	\$0.08	\$0.05		
Gulf Coast Mainline	\$0.05	\$0.05	\$0.05	\$0.05
\$0.30	\$0.08	\$0.05		

The Incremental Volumetric Reservation Rates will be increased each contract year following the first contract year, including any extensions of the term of the

FTS (SW) Agreement on the anniversary of the Effective Date by a factor of 2.5%.

6/ In the event that the physical flow of gas on Shipper's path from Alliance/NGPL Grundy (PIN No. 37207) to the point of intersection of Natural's Gulf Coast

Mainline and Natural's A-G Line continues in a southerly direction for sixty (60) consecutive days (a "Temporary Forward Haul"), then commencing on the thirty-

first day following Shipper's receipt of written notice by Natural, which may be given only after a Temporary Forward Haul has been established, the rates

applicable to such service shall be increased by \$0.07 per Dth. If the physical flow subsequently reverts to a northerly direction for more than thirty (30)

consecutive days, then commencing on the thirty-first day following the flow reversal, the \$0.07 per Dth increase shall be rescinded, until such time as Natural

again notifies Shipper that a Temporary Forward Haul has been established.

In the event that any Temporary Forward Haul continues for a total of at least two hundred seventy (270) consecutive days (a "Long Term Forward Haul") then

commencing on the thirty-first day following Shipper's receipt of written notice by Natural, which may be given only after a Long Term Forward Haul has been

established, the Negotiated Base Reservation Rate shall be increased by \$0.7985 per Dth of MDQ per month. If the physical flow subsequently reverts to a

northerly direction for more than two hundred seventy (270) consecutive days, then commencing on the thirty-first day following such flow reversal, the \$0.7985

per Dth of MDQ per month increase in the Base Negotiated Reservation Rate shall be rescinded.

7/ All current and any future reservation and commodity surcharges would be additive. GRI reservation and commodity surcharges are initially discounted to zero;

provided, however, that in the event that Natural is required to collect such GRI surcharges from Shipper or to pay such GRI surcharges to the FERC or to any

third party, then Shipper shall pay Natural any such amounts which are required to be so collected or paid.

FERC Docket: RP99-176-179

First Revised Sheet No. 33C.03 First Revised Sheet No. 33C.03

Superseding: Original Sheet No. 33C.03

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity Charge(s)	Eligible Primary Rece Shipper Name Point(s) / PIN No(s	Schedule	Term of Primary Delivery Contract () / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)	
N Boro	hillips Company 1/ der/NGPL Harper Keokuk N No. 908090)	FTS Nicor Gas/NGPL N	Manteno through	4,900	2/	2/
N Boro	hillips Company 1/ der/NGPL Harper Keokuk N No. 908090)	FTS Nicor Gas/NGPL Livingston (PIN	through	7,855	2/	2/

^{1/} The Transportation Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Agreement with corresponding Negotiated Rate Exhibit, which is currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 33G.03 original Sheet No. 33G.03 : Pending

FOOTNOTE APPLICABLE TO SHEET NO. 33G

- 7/ Eligible Receipt Points. (con't)
 - (b) Eligible Secondary Receipt Points.
- (1) Transportation Agreements. All secondary receipt points permitted under the relevant Transportation Agreement, excluding secondary receipt points

 which may be available in receipt zones on the Amarillo Mainline which are associated with the "opposite leg rights" provision of Section 5.5(a)(2)

 of the General Terms and Conditions of Natural's FERC Gas Tariff, as may be revised from time to time.
- (2) DSS Storage Agreements. For nominated firm transportation service, the Negotiated Fixed Monthly Base Reservation Rate shall apply to service provided from all secondary receipt points available under the DSS Agreements as set forth in Natural's FERC Gas Tariff, as may be revised from time to time.
 - 8/ Eligible Delivery Points.
- (a) Shipper's initial eligible primary delivery points and respective initial associated point MDQ shall be as set forth below in Footnote 9 (for each
 Transportation Agreement, the applicable "Eligible Primary Receipt Points"). The Negotiated Rates shall only apply to service provided under each of the
 Transportation Agreements from the applicable Eligible Primary Delivery Points, subject to the respective associated primary delivery point MDO

the respective associated primary delivery point MDQ limitations (the "Eligible Delivery Point MDQs"), and to the secondary delivery points permitted under the applicable Service Agreement set forth in (b)

below (the "Eligible Secondary Delivery Points"), with such eligible primary and secondary delivery points collectively referred to as the "Eligible Delivery Points".

- (b) Eligible Secondary Delivery Points.
- (1) Transportation Agreements. The Negotiated Fixed Monthly Base Reservation Rate shall only apply to transportation service provided under each
 Transportation Agreement to the following secondary delivery points: NICOR (9258),
 PGLC (909285), NIPSCO (909260), IES Utilities (25250),
 Interstate Power (901028), all Market Area interstate pipeline interconnects existing as of October 20, 2006, except MRT (900169), Illini (903987)
 and Horizon (39755/39855) and all storage and pooling points within the zones traversed by the Primary Path of the relevant Transportation
 Agreement.
- (2) DSS Storage Agreements. For nominated firm transportation service, the Negotiated Fixed Monthly Base Reservation Rate shall apply to service provided to all secondary delivery points located within the Market Delivery Zone as well as to the following secondary delivery points:

	PIN No.	DSS Storage Agreement No.	Name	
	FIN NO.	 		
CDP	10550	110349/110357	Iowa Illinois Gulf Coast Storage Point	
	25656	110349/110357	Iowa Illinois Amarillo Storage Point	
	901028	110349	Any existing and future interconnect	
			contained within the Interstate Power	
	25250	110357	Any existing and future interconnect	
	25250		contained within the IES Utilities CDP	

FERC Docket: RP08-319-000

Original Sheet No. 33N.03 original Sheet No. 33N.03 : Pending

FOOTNOTES APPLICABLE TO SHEET NO. 33N

8/ The Commodity Rate is the applicable maximum tariff rate pursuant to Natural's FERC Gas Tariff as of the Effective Date. The Commodity Rate shall remain fixed thereafter for the entire term of the FTS (SW) Agreement, including any extensions of such term.

9/ The Negotiated Rates set forth herein shall apply to transportation service provided by Natural to Shipper under the FTS (SW) Agreement with an MDQ of 300,000

Dth per day for the period August 1, 2003 to April 30, 2023, and an MDQ of 250,000 Dth per day for the period May 1, 2023 to July 31, 2023.

Other Miscellaneous Provisions

Eligible Secondary Receipt Points shall include all points available under the FTS (SW) Agreement in accordance with Natural's FERC Gas Tariff and will be eligible for the rates as stated herein.

Secondary Delivery Points shall include all existing pipeline interconnects and all pooling and storage points as permitted by the FTS (SW) Agreement, but not

including direct connects with electric generating stations and industrial plants whether through facilities owned by the direct customer or through facilities

owned by third parties constructed primarily to serve such stations or plants. All service to Eligible Secondary Delivery Points will be provided at the rates

described herein. Service to other secondary delivery points will be provided at the applicable maximum rate pursuant to Natural's FERC Gas Tariff as may be in effect from time to time.

Unless otherwise provided in the agreement, all applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff

shall apply to service hereunder, including without limitation all applicable gathering charges, offshore charges, lateral line charges, and ACA.

All Authorized Overrun Service provided to Shipper under the FTS (SW) Agreement shall be billed at the applicable maximum Authorized Overrun Rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

All Fuel/GLU shall be in accordance with Natural's FERC Gas Tariff, based on actual path nominated.

The Negotiated Rates shall apply only to a total aggregate maximum daily transportation quantity of three hundred thousand (300,000) Dth per day for all

quantities transported under the FTS (SW) Agreement and any associated capacity release replacement agreements. Any quantities transported in excess of the

maximum daily transportation quantity shall be billed at all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff.

The Base Negotiated Reservation Rate and Incremental Volumetric Reservation Rates, as such rates shall be increased as stated above, shall remain in effect for

the entire term of the FTS (SW) Agreement, including any extensions of the term of the FTS (SW) Agreement, notwithstanding any change in the maximum applicable tariff rate.

FERC Docket: RP08-319-000

Original Sheet No. 33G.04 original Sheet No. 33G.04 : Pending

FOOTNOTE APPLICABLE TO SHEET NO. 33G

8/ Eligible Delivery Points. (con't)

(c) Incremental Rate Secondary Delivery Points. In addition to the Negotiated Fixed Monthly Base Reservation Rate, Shipper shall pay Natural a Negotiated

Incremental Volumetric Daily Reservation Rate for all quantities transported on a firm basis on any day under each of the Transportation Agreements to the

Incremental Rate Secondary Delivery Points set forth below. The following Negotiated Incremental Volumetric Daily Reservation Rates will apply to firm

service provided to the secondary delivery points set forth below under the corresponding

point Tier: Tier 1 Rate: \$0.0200/Dth; Tier 2 Rate:

\$0.0500/Dth; Tier 3 Rate: \$0.1000/Dth. Service provided on a firm basis to all other secondary delivery points not specifically set forth in this
footnote section (c) or in (b) above will be provided at the greater of: (a) the 100% load

factor equivalent of the relevant Negotiated Fixed Monthly

Base Reservation Rate and Negotiated Fixed Base Commodity Rate; and (b) the 100% load factor equivalent of the weighted average maximum base reservation

and base commodity rates applicable for each contract set forth in Natural's FERC Gas Tariff, as may be revised from time to time for service to such points.

Secondary Delivery Point PIN No.

_____ _____

	25500 41074/909406	Tier 1 Secondary Delivery Points	Ameren/CIPS
			Aquila Networks
	909998		Ameren/CILCO
	909999		Illinois Power
	10568		MidAmerican MRT
	900169		MRT Illini
	903987		Horizon Pipeline
	909294/907112		·
	9254	Tier 2 Secondary Delivery Points	North Shore Gas
	909294/907112	Tier 3 Secondary Delivery Point	Wisconsin Electric
	N/A		Direct-connect end users
	N/A		Power generation facilities
Zone	N/A		All other Market Delivery
			Delivery

FERC Docket: RP99-176-167

First Revised Sheet No. 33N.04 First Revised Sheet No. 33N.04

Superseding: Original Sheet No. 33N.04

FOOTNOTES APPLICABLE TO SHEET NO. 33N

Incremental Rate for Selected Secondary Delivery Points

During the time period of May 1, 2008 through and including October 31, 2008, FPL and Natural agree to add the following:

For all quantities of gas transported on a firm basis on any day from all Receipt Points in the Texok Receipt Zone to all Secondary Delivery Points in the South

Texas Delivery Zone, FPL shall pay Natural the following charges: the Base Negotiated Reservation Rate (as may be discounted or otherwise amended from time to

time) plus an incremental daily volumetric reservation rate of 0.07/Dth. For all quantities of gas transported on a firm basis on any day from all Receipt

Points in the Texok Receipt Zone to all other Secondary Delivery Points not set forth in the Rate Agreement, FPL shall pay Natural the applicable maximum rate

set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

Effective Date: 11/01/2008 Status: Pending

FERC Docket: RP99-176-170

Second Revised Sheet No. 33N.04 Second Revised Sheet No. 33N.04

Superseding: First Revised Sheet No. 33N.04

FOOTNOTES APPLICABLE TO SHEET NO. 33N

Incremental Rate for Selected Secondary Delivery Points

During the time period of November 1, 2008 through and including October 31, 2009, FPLE and Natural agree to add the following:

For all quantities of gas transported on a firm basis on any day from all Receipt Points in the Texok Receipt Zone to all Secondary Delivery Points in the South

Texas Delivery Zone, FPLE shall pay Natural the following charges: the Base Negotiated Reservation Rate plus an incremental daily volumetric reservation rate of

0.14/Dth (from November 1, 2008 through March 31, 2009) and 0.07/Dth (from April 1, 2009 through October 31, 2009). For all quantities of gas transported on a

firm basis on any day from all Receipt Points in the Texok Receipt Zone to all other Secondary Delivery Points not set forth in the Rate Agreement, FPLE shall

pay Natural the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

FERC Docket: RP08-319-000

Original Sheet No. 33G.05 Original Sheet No. 33G.05 : Pending

FOOTNOTE APPLICABLE TO SHEET NO. 33G

9/ Eligible Primary Point MDQ.					
	(a) Effective Period				
		Primary Receipt Points (MDQ/PIN No.)			
5556),	103176 5/1/2007 11/01/2007 3,000 1,500 (PIN No. 901028-A), 1,500 (PIN No. 901028-G)	1,250 (PIN No. 3024), 250 (PIN No. 1,500 (PIN No. 5433)			
3592)	106570 5/1/2007 10/31/2007 12,000 4,000 (PIN No. 901028-A), 4,000 (PIN No. 25250),	4,000 (PIN No. 901850), 8,000 (PIN No.			
	4,000 (PIN No. 908090)				
3592)	11/1/2007 3/31/2008 13,455 4,000 (PIN No. 901028-A), 5,455 (PIN No. 25250),	4,000 (PIN No. 901850), 9,455 (PIN No.			
	4,000 (PIN No. 908090)				
3592)	4/1/2008 10/31/2008 12,000 4,000 (PIN No. 901028-A), 4,000 (PIN No. 25250),	4,000 (PIN No. 901850), 8,000 (PIN No.			
	4,000 (PIN No. 908090)				
3592)	11/1/2008 3/31/2009 13,455 4,000 (PIN No. 901028-A), 5,455 (PIN No. 25250),	4,000 (PIN No. 901850), 9,455 (PIN No.			
	4,000 (PIN No. 908090)				
3592)	4/1/2009 10/31/2009 12,000 4,000 (PIN No. 901028-A), 4,000 (PIN No. 25250),	4,000 (PIN No. 901850), 8,000 (PIN No.			
	4,000 (PIN No. 908090)				
3592)	11/1/2009 3/31/2010 13,455 4,000 (PIN No. 901028-A), 5,455 (PIN No. 25250),	4,000 (PIN No. 901850), 9,455 (PIN No.			
	4,000 (PIN No. 908090)				
3592)	4/1/2010 10/31/2010 12,000 4,000 (PIN No. 901028-A), 4,000 (PIN No. 25250),	4,000 (PIN No. 901850), 8,000 (PIN No.			
	4,000 (PIN No. 908090)				

FERC Docket: RP10-674-000

Second Revised Sheet No. 33G.06 Second Revised Sheet No. 33G.06

Superseding: First Revised Sheet No. 33G.06

FOOTNOTE APPLICABLE TO SHEET NO. 33G

9/ Eligible Primary Point MDQ. (con't)					
(a)	Effective Period				
	FTS Agreement No. From To Primary Delivery Points (MDQ/PIN No.		Primary Receipt Points (MDQ/PIN No.)		
900439), 2	110350 5/1/2007 10/31/2010 2,459 (PIN No. 901028-A), 2,459 (PIN No. 90	•	2,459 (PIN No. 901501), 1,229 (PIN No. 1,230 (PIN No. 900546)		
901501) 5	117037 5/1/2007 10/31/200° 5,000 (PIN No. 901028-A)	,	2,500 (PIN No. 903072), 2,500 (PIN No.		
901501)	11/1/2007 11/01/200° 7,000 (PIN No. 901028-A) 11/2/2007 3/31/2009	,	2,500 (PIN No. 903072), 4,500 (PIN No. 2,200 (PIN No. 903072), 4,500 (PIN No.		
901501)	6,700 (PIN No. 901028-A) 4/1/2008 10/31/2008		2,200 (PIN No. 903072), 4,300 (PIN No.		
	4,700 (PIN No. 901028-A) 11/1/2008 3/31/200	,	2,200 (PIN No. 903072), 4,500 (PIN No.		
	6,700 (PIN No. 901028-A) 4/1/2009 7/31/2009	9 4,700	2,200 (PIN No. 903072), 2,500 (PIN No.		
	4,700 (PIN No. 901028-A) 8/1/2009 10/31/2009 4,700 (PIN No. 901028-A)	9 4,700	2,200 (PIN No. 901477), 2,500 (PIN No.		
	11/1/2009 3/31/2010 6,700 (PIN No. 901028-A)	6,700	2,200 (PIN No. 901477), 4,500 (PIN No.		
901501)	4/1/2010 10/31/2010 4,700 (PIN No. 901028-A)	0 4,700	2,200 (PIN No. 901477), 2,500 (PIN No.		
5,5	117996 5/1/2007 10/31/2007 11/1/2007 3/31/2007 545 (PIN No. 25250)		545 (PIN No. 3592), 545 (PIN No. 5558)		
906125)		_	1,590 (PIN No. 905572), 2,865 (PIN No.		
5,5	4/1/2008 10/31/2008 11/1/2008 3/31/2009 545 (PIN No. 25250)		545 (PIN No. 3592), 545 (PIN No. 5558)		
906125)	4/1/2009 10/31/2009	n.	1,590 (PIN No. 905572), 2,865 (PIN No.		
5,5	11/1/2009 10/31/2001 11/1/2009 3/31/2011 545 (PIN No. 25250)		545 (PIN No. 3592), 545 (PIN No. 5558)		
906125)	4/1/2010 10/31/201	0	1,590 (PIN No. 905572), 2,865 (PIN No.		
F (127768 11/1/2007 10/31/201	12,000	12,000 (PIN No. 905207)		
5,000 (PIN No. 9258), 7,000 (PIN No. 25250) (b) DSS Agreement No.					
, , ,	110349 5/1/2007 11/01/200	7 10,000	10,000 (PIN No. 10550), 10,000 (PIN No.		
	0,000 (PIN No. 901028) 11/2/2007 4/30/201	3 8,300	8,300 (PIN No. 10550), 8,300 (PIN No.		
	8,300 (PIN No. 11054) 110357 5/1/2007 4/30/201: 4,014 (PIN No. 25250)	3 34,014	34,014 (PIN No. 10550), 34,014 (PIN No.		

^{10/} Miscellaneous Provisions

Conditions Precedent. Shipper and Natural understand and agree that this Agreement shall be

subject to any and all applicable conditions precedent under

Natural's FERC Gas Tariff and the regulations and policies of the FERC, including without limitation any requirements for Natural to file for and receive FERC approval of the Negotiated Rates.

FERC Docket: RP99-176-177

Second Revised Sheet No. 34 Second Revised Sheet No. 34

Superseding: First Revised Sheet No. 34

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Eligible Primary Receipt	Rate Term of Eligible Primary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Contract Point(s) / PIN No(s).	(Dth/d)	Charge(s)
Apache (Corporation 1/	FTS (SW) 12/22/2008 2/	34,489	2/
		through 3/31/2012		

^{1/} The Transportation Agreement, as amended, with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Agreement with corresponding Negotiated Rate Exhibit, which is currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 34A original Sheet No. 34A : Pending

	STATEMENT OF NEGOTIATED RATE
TRANSACTIONS	
	PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s).	Schedule	Term of ry Delivery Contract) / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)	
Laclede Er	nergy Resources, Inc.	FTS 1/	5/01/2007 through 4/30/2010	350	1/	1/

⁻⁻⁻⁻⁻

^{1/} The information is set out in the executed transportation Rate Schedule FTS Agreement with corresponding Negotiated Rate Exhibit, which is currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 34B Original Sheet No. 34B: Pending

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s).	Schedule	Term of Delivery Contract / PIN No(s).	Volume (Dth/d)	Reservation Charges(s)	
4/	4/	4/		-	2/, 3/, 4/	3/,
	ty Power Company, LP liance/NGPL-Joliet Grundy	FTS (SW) Tractebel I	1/ Fuels/NGPL Wise	30,000	\$1.58184	
(PIN	No. 37207)	(PIN No. 374	488)			

- 1/ The term commences on the earlier of: (i) November 1, 2003; or (ii) the date of initial commercial operation of Shipper's plant; but in no event prior to

 June 1, 2003, and continues in effect for ten (10) years.
- 2/ Shipper will make an advance payment of \$500,000 within thirty (30) days of being awarded firm transportation capacity, in addition to all other reservation and commodity charges.
- 3/ GRI reservation and commodity surcharges are initially discounted to zero; provided, however that in the event that Natural is required to collect such GRI surcharges from Shipper or to pay such GRI surcharges to the FERC or to any third party, then Shipper shall pay Natural any such amounts which are required to be so collected or paid.

Unless otherwise provided in the agreement, all applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas

Tariff shall apply to service hereunder including without limitation all applicable gathering

Tariff shall apply to service hereunder, including without limitation all applicable gathering charges, offshore charges, lateral line charges, ACA, and Account No. 858 surcharges.

To the extent that Natural is permitted to discount any new surcharges which may be included in Natural's FERC Gas Tariff after the date of this Agreement,

then Natural shall discount such surcharges for Shipper to the maximum extent permissible under

then Natural shall discount such surcharges for Shipper to the maximum extent permissible under Natural's tariff.

All authorized Overrun Service provided to Shipper under the FTS Agreement shall be billed at the applicable maximum Authorized Overrun Rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

4/ See the negotiated rate agreement and the FTS (SW) service agreement on file with the Commission for the details of how the negotiated rates apply to changes

in the primary receipt or delivery points and to the use of secondary receipt and delivery points.

FERC Docket: RP10-200-000

Third Revised Sheet No. 34C Third Revised Sheet No. 34C

Superseding: Second Revised Sheet No. 34C

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Prima	Term of ary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(Contract s) / PIN No(s).	(Dth/d)	Charges(s)
EDF Trad	ing North America, LLC 1/	FTS 2/	11/01/2009	50,000	2/
			through 10/31/2010		

^{1/} The Transportation Rate Schedule FTS Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 34D Original Sheet No. 34D: Pending

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s).	Schedule	Term of mary Delivery Contract (s) / PIN No(s).	Volume (Dth/d)	Reservation Charges(s)	
Tenaska 6/	Gas Storage, LLC 1/ 7/	NSS 7/	4/01/2008 through 3/31/2011 2/	9,193 3(a)/	\$3.07/Dth of MDQ 4/, 6/	5/,
Tenaska 6/	Gas Storage, LLC 1/7/	NSS 7/	4/01/2008 through 3/31/2011 2/	10,807 3(b)/	\$3.07/Dth of MDQ 4/, 6/	5/,

(a) The Negotiated Rates set forth herein shall apply to storage service provided by Natural to Shipper under Rate Schedule NSS with an Eligible Contract
Maximum Storage Volume ("MSV") of Six Hundred Eighty Nine Thousand Four Hundred Seventy
Five (689,475) Dth and an Eligible Contract MDQ of Nine
Thousand One Hundred Ninety Three (9,193) Dth per day, MDQ of 9,193 Dth per day and an MSV of 689,475 Dth on the Amarillo Leg, and an MDQ of 0 Dth per
day and a MSV of 0 Dth on the Gulf Coast Leg, with an Amarillo Percentage equal to one hundred percent (100%) and a Gulf Coast Percentage equal to zero (0%).

(b) The Negotiated Rates set forth herein shall apply to storage service provided by Natural to Shipper under Rate Schedule NSS with an Eligible Contract
Maximum Storage Volume ("MSV") of Eight Hundred Ten Thousand Five Hundred Twenty Five
(810,525) Dth and an Eligible Contract MDQ of Ten Thousand Eight
Hundred Seven (10,807) Dth per day. MDQ of 10,807 Dth per day and an MSV of 810,525 Dth on the Amarillo Leg, and an MDQ of 0 Dth per day and a MSV of
Dth on the Gulf Coast Leg, with an Amarillo Percentage equal to one hundred percent (100%) and a Gulf Coast Percentage equal to zero (0%).

4/ Fixed Negotiated Monthly Base Reservation Rate. The Negotiated Monthly Base Reservation Rate per Dth of MDQ (Peak and Off-Peak periods) shall apply only to service provided by Natural to Shipper under the Rate Schedule NSS Agreement, subject to all other terms and conditions set forth herein.

^{1/} The Negotiated Rate Exhibit with Shipper does not deviate in any material respect from the applicable Form of Service Agreement in Natural's FERC Gas Tariff.

^{2/} Negotiated Rate Term. The negotiated rates set forth herein shall be in force during a term commencing on April 1, 2008 through March 31, 2011.

^{3/} Eligible Firm Quantities.

FERC Docket: RP99-176-187

First Revised Sheet No. 34E First Revised Sheet No. 34E Superseding: Original Sheet No. 34E

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

TRANSACTIONS

Commodity	Primary Receipt	Rate Prima	Term of ary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(s)	Contract / PIN No(s).	(Dth/d)	Charge(s)
The Proc	ter and Gamble Paper			-	
	cts Company 1/	FTS	4/01/2009	16,900	2/
			through 3/31/2014		

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} This information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-656-000

First Revised Sheet No. 34F First Revised Sheet No. 34F Superseding: Original Sheet No. 34F

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt	Prima	ary Delivery		
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s	s) / PIN No(s).		-
Occident	al Energy Marketing, Inc.	NSS	5/01/2010	10,000	1/
1/	1/		1/		
			through		
			4/30/2013		

^{1/} This information is set out in the executed Storage Rate Schedule NSS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-912-000

Second Revised Sheet No. 34G Second Revised Sheet No. 34G

Superseding: First Revised Sheet No. 34G

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt	Prima	ry Delivery		
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s) / PIN No(s).		
				•	
Stand F	nergy Corporation 1/	FTS	7/01/2010	970 2/	2/
2/	2/	2	/	510 27	2/
27	2,	-	through		
			6/30/2011		

^{1/} The Agreement with Shipper does not deviate in any material respect from the applicable Form of Service Agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transporation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 34H Original Sheet No. 34H: Pending

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primarv	Term of Delivery	Volume	Reservation	
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract / PIN No(s).	(Dth/d)	Charge(s)	
	Illinois Gas Company, rizon/NGPL McHenry	FTS 12/, 13	4/01/2002	100,000	\$0.0305	4/,
	Nicor Gas 1/ PIN No. 39755) 11/, 13/	•	through		3/, 5/, 7/,	9/,
10/ / 11/	11N NO. 337337 117 , 137		3/31/2012 2/		8/, 9/, 10/, 14/	

- 1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ Negotiated Rate Term. The negotiated rates set forth herein shall be effective for a ten (10) year term, beginning the later of the in-service date of the

Horizon Pipeline and the in-service date of the Primary Receipt Point set forth above between Natural and the Horizon Pipeline.

- 3/ Negotiated Reservation Rate. The Negotiated Reservation Rate shall apply only to service provided to Shipper by Natural under the FTS Agreement from the
- Eligible Receipt Points to the Eligible Delivery Points, subject to all other applicable terms and conditions set forth herein.
- 4/ Negotiated Commodity Rate. For all quantities transported by Natural under the FTS Agreement, Shipper shall pay Natural the applicable minimum base commodity
- rate. Shipper shall not be responsible for payment of any other commodity charges or surcharges set forth in Natural's FERC Gas Tariff, as may be revised from $\frac{1}{2}$

time to time, other than the ACA surcharge (or any successor surcharge).

- 5/ Eligible Transportation Quantity. The negotiated rates shall apply only to a total aggregate maximum daily transportation quantity of 100,000 Dth/d, for all
- quantities transported under the FTS Agreement and any associated capacity release replacement agreements. Any quantities transported in excess of the

Eligible Transportation Quantity shall be billed at all applicable maximum rates, charges and surcharges set forth in Natural's FERC Gas Tariff.

- 6/ Authorized Overrun Service. All Authorized Overrun Service provided to Shipper under the FTS Agreement shall be billed at the applicable maximum Authorized
- Overrun Rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time, unless otherwise agreed to by the parties.
- 7/ Fuel Gas and Lost and Unaccounted For. To the extent Shipper's primary path represents a physical backhaul on Natural's system, shipper shall be charged only
- for Gas Lost and Unaccounted For, without any fuel gas component, for any quantities transported on any day from the Eligible Receipt Point to the Eligible
- Delivery Points which represent a physical backhaul, as permitted under Section 6.11 of Natural's FTS Rate Schedule. Fuel Gas and the Gas Lost and Unaccounted
- For component shall apply to all other quantities transported by Natural for Shipper under the FTS Agreement. All Fuel Gas and Gas Lost and Unaccounted For
- charges shall be assessed at the applicable rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.
- 8/ Reservation Surcharges Included. The negotiated rates are inclusive of all applicable reservation surcharges including, but not limited to, Account No. 858 reservation surcharges.
 - 9/ GRI Surcharges. Natural shall not bill Shipper for any GRI reservation or commodity surcharges.

FERC Docket: RP08-319-000

Original Sheet No. 34I original Sheet No. 34I : Pending

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Prima:	Term of ry Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract) / PIN No(s).	(Dth/d)	Charge(s)
Wisconsi 1/	in Electric Power Company 1/	FTS	4/01/2007	3,000	1/
			through 3/31/2012		

⁻⁻⁻⁻⁻

^{1/} This information is set out in the executed Transportation Rate Schedule FTS Agreement with corresponding Negotiated Rate Exhibit, which is currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 34J original Sheet No. 34J: Pending

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primary	Term of Delivery	Volume	Reservation	
<u> -</u>	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)	
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).			
Illinois	Power Company 1/	FTS	2/	90,000 3/	3/, 4/, 6/	3/,
5/, 6/ NGP	L/MRT Clinton (PIN No. NG	PL/ILL Power Me	etro East,	•		

900169) 7/ Madison (PIN No. 40330) 7/

- 1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ The Negotiated Rate Agreement shall be effective for a term of ten (10) years from the later of: (i) June 1, 2002; or (ii) the date of initial commercial operation of Natural's proposed lateral; provided, however, that in no event shall the term commence prior to June 1, 2002.
- 3/ Eligible Transportation Quantity. The Negotiated Reservation Rate and the Negotiated Commodity Rate shall apply only to total aggregate deliveries on any day
- of a quantity equal to the Rate Schedule FTS MDQ, for all quantities transported under the Rate Schedule FTS Agreement and any associated capacity release
- replacement agreements. Unless otherwise agreed in writing by Natural, any quantities transported in excess of the Eliqible Transportation Quantity shall be
- billed at all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff.
- 4/ The Negotiated Monthly Reservation Rate per Dth of MDQ = Peak \$3.2223/Dth and Off-Peak \$3.3109/Dth of MDQ.
 - 5/ The Negotiated Commodity Rate per Dth = Peak \$0.0021/Dth and Off-Peak \$0.0005/Dth.
 - 6/ Other Provisions:

Authorized Overrun Service. Unless Natural otherwise expressly agrees in writing, Authorized Overrun Service provided to Shipper under the Rate Schedule FTS

Agreement shall be billed at the applicable maximum Authorized Overrun Rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

GRI Surcharge. The Negotiated Rates reflect an initial discount of all GRI reservation and commodity surcharges otherwise payable by Shipper; provided, however,

that in the event that Natural is required to collect such GRI Surcharges from Shipper and to pay such GRI Surcharges to the FERC or to any third party, then

Shipper shall pay Natural any such amounts which are required to be so collected and paid.

Fuel Gas and Lost and Unaccounted For. In addition to the Negotiated Rates, Shipper shall be charged for Fuel Gas and Gas Lost and Unaccounted For in accordance

with the applicable provisions of Natural's FERC Gas Tariff, as may be revised from time to time.

Applicable Maximum Rates, Charges, and Surcharges. The Negotiated Reservation Rates are inclusive of all Account No. 858 surcharges set forth in Natural's FERC

Gas Tariff which are or may be applicable to service provided to Shipper under the Rate Schedule FTS Agreement. The Negotiated Monthly Inclusive Reservation

Rates shall be applicable at all times during the Negotiated Rate Term. Unless otherwise expressly provided herein, all applicable maximum rates, charges,

surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff shall apply to service provided to Shipper under the Rate Schedule FTS Agreement,

including without limitation all applicable gathering charges, offshore charges, lateral line charges, and ACA surcharges.

FERC Docket: RP08-319-000

Original Sheet No. 34K Original Sheet No. 34K: Pending

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation	
Commodity	Primary Receipt Shipper Name	Primary Schedule	Delivery Contract	(Dth/d)	Charge(s)	
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).			
Illinois	Power Company 1/	NSS 6/	2/	10,000 3/, 5/	\$3.0700 3/, 5/	3/,

- 1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ The Negotiated Rate Agreement shall be effective for a term of ten (10) years from the later of: (i) June 1, 2002; or (ii) the date of initial commercial operation of Natural's proposed lateral; provided, however, that in no event shall the term commence prior to June 1, 2002.
- 3/ Negotiated Storage Quantity. The Negotiated Rates shall apply only to a total aggregate maximum daily storage quantity of ten thousand (10,000) Dth per day on

the Gulf Coast Leg, for all quantities stored under the Rate Schedule NSS Agreement and any associated capacity release replacement agreements. Any quantities

stored in excess of the Negotiated Storage Quantity shall be billed at all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas

Tariff.

- 4/ All applicable maximum commodity rates, charges, and surcharges shall apply to storage service provided by Natural to Shipper under the Rate Schedule NSS

 Agreement, unless otherwise expressly provided herein.
 - 5/ Other Provisions:

Authorized Overrun Service. All Authorized Overrun Service provided to Shipper under the Rate Schedule NSS Agreement shall be billed at the applicable maximum

Authorized Overrun Rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

Storage Fuel. In addition to the Negotiated Rates, Shipper shall be charged for Storage Fuel in accordance with the applicable provisions of Natural's FERC

Gas Tariff, as may be revised from time to time.

Surcharges. The Negotiated Rates reflect an initial discount of all GRI reservation and commodity surcharges otherwise payable by Shipper; provided, however,

that in the event Natural is required to collect such GRI Surcharges from Shipper and to pay such GRI Surcharges to the FERC or to any third party, then

Shipper shall pay Natural any such amounts billed by Natural in addition to the Negotiated Rates.

Applicable Maximum Rates, Charges, and Surcharges. Unless otherwise expressly provided herein, all applicable maximum rates, charges, and penalties of any nature set forth in Natural's FERC Gas Tariff shall apply to service provided to Shipper under the

nature set forth in Natural's FERC Gas Tariff shall apply to service provided to Shipper under the Rate Schedule NSS Agreement, including without limitation all applicable ACA surcharges.

6/ The Receipt Points for injection and the Delivery Points for withdrawal under the NSS service agreement shall be the storage points as defined in the General Terms and Conditions of Natural's FERC Gas Tariff.

FERC Docket: RP10-527-000

First Revised Sheet No. 34L First Revised Sheet No. 34L Superseding: Original Sheet No. 34L

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49 _____

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt		y Delivery		
	hipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).		
Northern	n Indiana Public Service	FTS	4/01/2010	15,000	1/
1/	1/	1/			
Compa	nny		through		
-	_		3/31/2013		

^{1/} This information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Agreement, which are currently on file with the FERC.

FERC Docket: RP99-176-202

 $\textbf{Original Sheet No. 34M} \; \texttt{Original Sheet No. 34M} \\$

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt	Primary	Delivery		
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).		-
				· -	
mb - Deess	d of mountain of Hairranite.	FTS	11/01/2008	10 000	1 /
The Board	d of Trustees of University	1/	11/01/2008	10,000	1/
1/	1/	1/			
of Ill:	inois		through		
			6/30/2012		

^{1/} This information is set out in the amended Transportation Rate Schedule FTS Agreement with corresponding Negotiated Rate Exhibit, which is currently on file with the FERC.

FERC Docket: RP99-176-198

Second Revised Sheet No. 34C.01 Second Revised Sheet No. 34C.01

Superseding: First Revised Sheet No. 34C.01

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primar	Term of y Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(s)	Contract / PIN No(s).	(Dth/d)	Charges(s)
Marathon 2/	Oil Company 1/	NSS 2/	5/01/2009	20,000	2/
			through 4/30/2012		

^{1/} The firm Storage Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} This information is set out in the executed Storage Rate Schedule NSS Agreement, as amended, and the corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 34D.01 original Sheet No. 34D.01 : Pending

FOOTNOTES APPLICABLE TO SHEET NO. 34D

5/ Fixed Negotiated Base Commodity Rate. Shipper shall not be charged any base commodity rate for service provided under the Storage Agreement.

6/ Negotiated Rate Terms, Conditions and Additional Charges.

(a) General Negotiated Rate Limitations. The Negotiated Monthly Base Reservation Rate shall apply only to service provided to Shipper by Natural under the

NSS Agreement during the Negotiated Rate Term, up to a maximum aggregate firm quantity equal to the respective Eligible Contract MDQ and the Eligible

Contract MSV set forth in footnote three above, for all quantities stored on a firm basis under the NSS Agreement and any associated capacity release

replacement agreements. For any aggregate quantities stored on a firm basis for Shipper and any associated capacity release replacement Shippers which

are in excess of the Contract MDQ or the Contract MSV, as applicable, Shipper shall be charged the greater of: (a) the 100% load factor equivalent of

the Negotiated Monthly Base Reservation Rate; and (b) the 100% load factor equivalent of the applicable maximum monthly base reservation rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

(b) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in

Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular $\overline{\text{D}}$ is countable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payment to Natural of any portion of such Discountable Third Party Surcharge which Natural is not

permitted to discount and for which Natural bills Shipper.

(c) Additional Rates, Charges and Surcharges. In addition to the Negotiated Monthly Base Reservation Rate, and unless otherwise expressly provided herein or

agreed to in writing by Natural, Shipper shall also pay Natural all other applicable maximum rates, charges, surcharges and penalties of any nature set

forth in Natural's FERC Gas Tariff, as may be revised from time to time, including without limitation all applicable maximum: (i) Storage Fuel charges;

(ii) authorized and unauthorized overrun charges and (iii) sourcing charges.

Notwithstanding the foregoing sentence, to the extent that any such

additional maximum rate, charge, surcharge or penalty is derived from the applicable base reservation rate, then Natural shall derive such additional

rate, charge, surcharge or penalty utilizing the greater of: (i) the Negotiated Monthly Base Reservation Rate and (ii) the applicable maximum base

reservation rate set forth in Natural's FERC Gas Tariff, as may be revised from time to

7/ Receipt and Delivery Points. The Receipt Points for injection and the Delivery Points for withdrawal under the Rate Schedule NSS Agreement shall be the

Storage Points as defined in Section 9.3 of the General Terms and Conditions set forth in Natural's FERC Gas Tariff.

FERC Docket: RP08-389-000

Original Sheet No. 34E.01 Original Sheet No. 34E.01

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s)		Contract	Volume (Dth/d)		
	llips Company 1/ n Bridgeport Plant	FTS Gulf South Goodri	2/ (a) ch	10,000	\$6.6700 3	3/
(PIN	No. 901850) 4/	(PIN No. 6297) 5/				
	llips Company 1/ n Bridgeport Plant	FTS Kansas Gas Servic	2/ (b) e	5,000	\$1.5708 3	3/
(PIN	No. 901850) 4/	(PIN No. 3063) 5/				
Occidenta Express	l Energy Marketing, Ind	c. 1/ FTS-SW KMTP/NGPL	2/ (c)	41,000	\$6.0833 3/ 6/ 3	3/ Pony
(PIN	No. 11165) 4/	(PIN No. 5579) 5/	6/			

- The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- Negotiated Rate Term. The Negotiated Rate Term shall commence on the date the Segment One Expansion and the Amarillo to Gulf Coast Expansion ("Expansion") is

ready to receive firm service ("Effective Date") which is anticipated to be April 15, 2006. For purposes of this section, the Expansion shall be considered ready to receive firm service when the pipeline and compression are ready for firm service. This

negotiated rate term shall end:

- (a) five (5) years and six (6) months from the Effective Date;
- (b) five (5) years and one month from the Effective Date; and
- (c) five (5) years from the Effective Date.
- 3/ Negotiated Rates.
- (a) Eligible Transportation Quantity. The Negotiated Rate shall apply to a total aggregate maximum daily transportation quantity of Dth/day as shown above,

for all quantities transported under the FTS Agreement and any associated capacity release replacement agreements (the "Eligible Transportation

Quantity"). Any quantities transported in excess of the Eligible Transportation Quantity shall be billed at the greater of the Negotiated Rate or the

applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff. FOOTNOTES APPLICABLE TO SHEET NO. 33C _____

- 3/ Negotiated Rates. (con't)
- The Negotiated Rate shall apply only to service provided to Shipper by Natural under the FTS Agreement from the Eligible Receipt Points to the Eligible

Delivery Points during the Negotiated Rate Term. The Negotiated Rate shall apply only to a maximum daily firm transportation quantity equal to the

Eligible Transportation Quantity set forth above, for all quantities transported on a firm basis under the Transportation Agreement and any associated

capacity release replacement agreements. Any quantities transported on a firm basis in excess of the Eligible Transportation Quantity shall be billed at

the greater of the Negotiated Rate or all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

- (c) Monthly Base Negotiated Reservation Rate ("Negotiated Rate"). The Negotiated Rate set forth above per Dth of MDQ for peak and off-peak periods is effective for and through the Negotiated Rate Term. The Negotiated Rate shall be fixed for the term of the FTS Agreement.
- (d) General Negotiated Rate Limitations. The Negotiated Rate shall apply only to: (i) service provided to Shipper by Natural under the Transportation

Agreement from the Eligible Receipt Points to the Eligible Delivery Points during the Negotiated Rate Term; and (ii) a maximum daily firm transportation

quantity equal to the Eligible Firm Transportation Quantity set forth in footnote 3(a) above, for all quantities transported on a firm basis under the

Transportation Agreement and any associated capacity release replacement agreements. Shipper shall be charged the greater of the Negotiated Rate and all

applicable rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time, for any aggregate quantities

transported on a firm basis for Shipper and any associated capacity release replacement shippers which: (i) are in excess of the Eligible Firm

Transportation Quantity, or (ii) involve any receipt or delivery points which are not Eligible Receipt or Delivery Points.

- (e) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in
- Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payments to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

(f) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein or agreed to in writing by Natural, the greater of the Negotiated Rate and all applicable maximum rates, charges, surcharges and penalties of any

Negotiated Rate and all applicable maximum rates, charges, surcharges and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be

revised from time to time, shall apply to service provided to Shipper under the FTS Agreement, including without limitation all applicable: (i) Fuel and

Gas Lost and Unaccounted For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and

surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

4/ Eligible Receipt Points. The Negotiated Rates shall apply to transportation service provided from the following receipt point(s).

Primary Receipt Point. ConocoPhillips Company - Devon Bridgeport Plant, PIN No. 901850 (10,000 MDQ) and (5,000 MDQ), respectively;

Occidental Energy Marketing, Inc. - Pony Express, PIN No. 11165 (41,000 MDQ).

Secondary Receipt Points. The Negotiated Rates shall apply to service provided from all receipt points located in the following zones: ConocoPhillips (10,000

MDQ) - Texok and Midcontinent zones; ConocoPhillips (5,000 MDQ) - Midcontinent zone; Occidental - Permian, Midcontinent, Amarillo Mainline, and Texok zones.

FOOTNOTES APPLICABLE TO SHEET NO. 33C

5/ Eligible Delivery Points. The Negotiated Rates shall apply to transportation service provided to the following delivery point(s):

Primary Delivery Points: ConocoPhillips - Gulf South Goodrich, PIN 6297 (10,000 MDQ); ConocoPhillips - Kansas Gas Service, PIN 3063 (5,000 MDQ); Occidental - KMTP/NGPL, PIN 5579 (41,000 MDQ).

Secondary Delivery Points: The Negotiated Rates shall apply to service provided to the following secondary delivery points:

ConocoPhillips - (10,000 MDQ) - All pooling, storage, Hinshaw and Intrastate Pipeline Interconnects and Interstate Interconnect points located within the zones traversed by the primary path of the Transportation

Agreement, excluding the Lone Star Pipeline Interconnect located in
the Midcontinent Zone and the Kinder Morgan Texas Pipeline Interconnect l

the Midcontinent Zone and the Kinder Morgan Texas Pipeline Interconnect located in the Texok Zone;

 ${\tt ConocoPhillips - (5,000 MDQ) - All pooling, storage, Hinshaw and Intrastate Pipeline}$

Interconnects and Interstate Interconnect points located
within the zones traversed by the primary path of the Transportation Agreement,
excluding the Lone Star Pipeline Interconnect located in the
Midcontinent Zone;

Occidental - Subject to the provisions in footnote 6 below regarding incremental rate secondary delivery points, the Negotiated Rate shall apply to all delivery points located in the Midcontinent and Texok zones.

6/ Incremental Rate Secondary Delivery Points. In addition to the Negotiated Rate, Shipper shall pay Natural an incremental volumetric daily reservation rate for all quantities transported on a firm basis on any day to the Incremental Rate Secondary Delivery

The following incremental volumetric daily reservation rates will apply to firm service provided to the secondary delivery points set forth below under the

Tier 1: Rate: \$0.030/Dth; Secondary Delivery Points: All points located in the LA zone.

Tier 2: Rate: \$0.035/Dth (Off-Peak) and \$0.0625/Dth (Peak); Secondary Delivery Points: Nicor (PIN No. 9258), NIPSCO (PIN No. 909260), PGL&C (PIN No. 909285),

MidAmerican (PIN No. 10568), and all existing interstate pipeline interconnects in the Market Delivery Zone, excluding MRT, Illini Carrier,

Guardian and ANR Joliet.

Service provided on a firm basis to all other secondary delivery points not specifically set forth in this footnote 6 or in footnote 5, above, regarding

Secondary Delivery Points, will be provided at the greater of the Negotiated Rate or the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may

be revised from time to time, for service to such points.

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Points set forth herein for service to such point.

corresponding Point Tier:

Commodity	Primary Receipt	Rate Prima	Term of ary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(s	Contract s) / PIN No(s).	(Dth/d)	Charge(s)
Nicor Gas	s Company 1/ 2/	FTS 2	4/01/2008 2/ through 3/31/2011	40,000	2/
Nicor Gas 2/	s Company 1/ 2/	FTS 2	4/01/2008 2/ through 3/31/2011	20,000	2/

1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable Form of Service Agreement in Natural's FERC Gas Tariff.

2/ This information is set out in the executed Transportation Rate Schedule FTS Agreement, and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

use.

Sheet No. 34 has been reserved for future

future use.

TRANSACTIONS

Sheet No. 34C.01 has been reserved for

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Rate Pri	Term of mary Delivery	Volume	Reservation
Schedule Point(s)	Contract / PIN No(s).	(Dth/d)	Charge(s)
NSS 6/	4/01/2008	22,526 3/	\$3.07/Dth of
σ,	through 3/31/2011		MDQ 4/ 5/
	Pri Schedule Point(s)	Primary Delivery Schedule Contract Point(s) / PIN No(s). NSS 4/01/2008 6/ through	Primary Delivery Schedule Contract (Dth/d) Point(s) / PIN No(s). NSS 4/01/2008 22,526 3/ 6/ through 3/31/2011

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable pro forma service agreement in Natural's FERC Gas Tariff.

^{2/} Negotiated Rate Term. The negotiated rates set forth herein shall be in force during a term commencing on April 1, 2008 through March 31, 2011.

^{3/} NSS Agreement - Eligible Firm Quantities. The Negotiated Rates set forth herein shall apply to storage service provided by Natural to Shipper under Rate

Schedule $\overline{\text{NSS}}$ with a Contract Maximum Storage Volume ("MSV") of One Million Six Hundred Eighty Nine Thousand Four Hundred Fifty (1,689,450) Dth and a Contract

MDQ of Twenty Two Thousand Five Hundred Twenty Six (22,526) Dth per day. MDQ of 22,526 Dth per day and an MSV of 1,689,450 Dth on the Amarillo Leg, and an MDQ

of 0 Dth per day and a MSV of 0 Dth on the Gulf Coast Leg, with an Amarillo Percentage equal to one hundred percent (100%) and a Gulf Coast Percentage equal to zero (0%).

4/ Negotiated Monthly Base Reservation Rate. The Negotiated Monthly Base Reservation Rte per Dth of MDQ (Peak and Off-Peak periods) shall apply only to service

provided by Natural to Shipper under the Rate Schedule NSS Agreement, as amended, subject to all other terms and conditions set forth herein.

- 5/ Negotiated Rate Terms, Conditions and Other Charges.
- (a) General Negotiated Rate Limitations. The Negotiated Rates shall apply only to service provided to Shipper by Natural under the NSS Agreement during the

Negotiated Rate Term up to a maximum aggregate firm quantity equal to the respective Eligible Contract MDQ and the Eligible Contract MSV set forth in

footnote three above, for all quantities stored on a firm basis under the NSS Agreement and any associated capacity release replacement agreements.

Shipper shall be charged all applicable maximum rates, charges and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time,

for any aggregate quantities stored on a firm basis for Shipper and any associated capacity release replacement shippers which are in excess of the $\dot{}$

respective Eligible Contract MDQ or the Eligible Contract MSV, as applicable.

FOOTNOTES APPLICABLE TO SHEET NO.

34D.02

- 5/ Negotiated Rate Terms, Conditions and Other Charges (con't)
- (b) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in

Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payment to Natural of any portion of such

Tariff. Shipper shall only be responsible for payment to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

(c) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein or agreed to in writing by Natural, the greater of the Negotiated Rates and all applicable maximum rates, charges, surcharges and penalties of any

nature set forth in Natural's FERC Gas Tariff, as may be
revised from time to time, shall apply to service provided to Shipper under the NSS

Agreement including without limitation all applicable: (i) Storage

Fuel charges; (ii) authorized and unauthorized overrun charges; and (iii) sourcing charges.

6/ Receipt and Delivery Points. The Receipt Points for injection and the Delivery Points for withdrawal under the Rate Schedule NSS Agreement shall be the

Storage Points as defined in Section 9.3 of the General Terms and Conditions set forth in Natural's FERC Gas Tariff.

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Prima	Term of rv Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(s	Contract) / PIN No(s).	(Dth/d)	Charge(s)
Tenaska M	Marketing Ventures 1/ 2/	FTS 2	4/01/2008	25,000	2/
			through 3/31/2010		

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable pro forma service agreement in Natural's FERC Gas

Tariff.

2/ This information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 34H.01 original Sheet No. 34H.01 : Pending

ADDITIONAL FOOTNOTES APPLICABLE TO SHEET

NO. 34H

10/Applicable Maximum Rates, Charges and Surcharges. The applicable maximum ACA surcharge, or any successor to such surcharge and all applicable maximum

penalties and charges of any nature set forth in Natural's FERC Gas Tariff which are not base rates or surcharges, shall apply to service provided to Shipper under the FTS Agreement.

11/Eligible Receipt Points. The negotiated rates shall only apply to transportation service provided from the following receipt points:

Primary Receipt Points: Horizon/NGPL McHenry, PIN No. 39755 (100,000 Dth/d MDQ).

Secondary Receipt Points: The negotiated rates shall not apply to service provided from any secondary receipt points. Service from any secondary receipt

points shall be charged the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

12/Eligible Delivery Points: The negotiated rates shall only apply to transportation service provided to the following delivery points:

Primary Delivery Points: Nicor Gas/NGPL Rockford, PIN No. 900012 (85,000 Dth/d MDQ); Nicor Gas/NGPL Sycamore, PIN No. 900013 (10,000 Dth/d MDQ); and Nicor Gas/NGPL South Dixon, PIN No. 900019 (5,000 Dth/d MDQ).

Secondary Delivery Points. The negotiated rates shall only apply to transportation service provided to secondary delivery points within the zones traversed by $\frac{1}{2}$

the primary path which are physical points of interconnect within the Nicor Gas CDP (PIN No. 9258) located on the Illinois lateral downstream of NGPL

compressor station 110 and upstream of NGPL compressor station 116. Service to all other secondary delivery points shall be charged the applicable maximum

rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

13/ Aggregate Receipt/Delivery Quantity Limitations. The negotiated rates shall apply to all quantities received or delivered by Natural for Shipper's account in

any particular transportation zone from the Eligible Receipt Points to the Eligible Delivery Points, regardless of whether such quantities are transported on

a primary or secondary firm basis, to the extent that such aggregate firm receipt or delivery quantities do not exceed the aggregate amount of Shipper's

Eligible Point MDQs set forth in footnotes 11 and 12 above for receipts or deliveries, as applicable, in such zone.

14/ Applicable Rate Considerations. Unless Natural otherwise expressly agrees in writing. Shipper shall not be entitled to change any of its Primary Receipt

Points or Primary Delivery Points under the FTS Agreement if such change would result in Shipper paying a transportation rate to Natural which is lower than

the applicable negotiated rates.

FERC Docket: RP08-319-000

Original Sheet No. 34J.01 original Sheet No. 34J.01 : Pending

FOOTNOOTES APPLICABLE TO SHEET NO. 34J

7/ Eligible Receipt and Delivery Points. The Negotiated Rates shall only apply to transportation service provided from the following receipt points to the following delivery points:

Primary Receipt Points: NGPL/MRT Clinton, PIN No. 900169 (90,000 Dth MDQ) Secondary Receipt Point: None

Primary Delivery Points: NGPL/Ill Power Metro East Madison, PIN No. 40330 (90,000 Dth MDQ) Secondary Delivery Points: All points within the primary path of the Eligible Primary Pathing excluding storage and pooling points.

Eligible Primary Pathing. Shipper's primary path under the Rate Schedule FTS Agreement will be from MRT Clinton Co. (PIN No. 900169) to Natural's interconnect

with Illinois Power Metro East Madison Co., (PIN No. 40330) or as otherwise mutually agreed.

Aggregate Receipt/Delivery Quantity Limitation. The Negotiated Rates shall apply to all quantities received or delivered by Natural for Shipper's account in any

particular transportation zone which are transported on a primary or secondary basis, to the extent that such aggregate firm receipt or delivery quantities do

not exceed the aggregate amount of Shipper's Eligible Point MDQs set forth above for receipts or deliveries, as applicable, in such zone. Unless Natural

expressly agrees in writing, service provided in excess of the applicable zone quantity limitation shall be charged all applicable maximum rates, charges and surcharges set forth in Natural's FERC Gas Tariff.

Applicable Rate Considerations. Unless Natural otherwise expressly agrees in writing. Shipper shall not be entitled to change any of its Primary Receipt Points

or Primary Delivery Points under the Rate Schedule FTS Agreement if such change would result in Shipper paying a reservation rate to Natural which is lower than

the applicable Negotiated Reservation Rate. Any time during the term, Shipper shall be entitled to change, at the Negotiated Reservation Rate, the Primary

Receipt Point set forth above to any other receipt point which exists as of the date of commencement reflected in Footnote No. 2 above and is within Shipper's

then-existing primary path, subject to capacity availability and all applicable provisions of Natural's FERC Gas Tariff regarding receipt point changes, provided

that any such Primary Receipt Point change may not result in Shipper extending its then-existing primary path.

FERC Docket: RP99-176-193

Original Sheet No. 34L.01 Original Sheet No. 34L.01

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

Commodity	Eligible Primary Receipt	Rate Eligible I	Term of Primary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract s) / PIN No(s).	(Dth/d)	Charge(s)
4/, 5/	Indiana Public Service 5/	FTS	4/01/2009 5/	2/	3/, 5/
Company	y 1/		through 3/31/2013		

2/ Negotiated Rate Volume (Dth of MDQ/day):

Rate Schedule	Agreement No.	April 1 through October 31	November 1 through March 31
FTS	111887	75,000	75,000
FTS-SW	110492	77,883	77,883
FTS	110493	N/A	63,000

3/ Fixed Negotiated Monthly Base Reservation Rate (\$/Dth of MDQ):

Agreement No.	Off-Peak April 1 through October 31	Peak November 1 through March 31
111887	\$6.7257	\$ 6.6440
110492	\$7.1665	\$ 7.2053
110493	N/A	\$10.7210

PURSUANT TO SECTION 49

^{1/} The Firm Transportation Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

FERC Docket: RP99-176-161

Original Sheet No. 34C.02 Original Sheet No. 34C.02: Pending

STATEMENT OF NEGOTIATED RATE TRANSACTIONS
PURSUANT TO SECTION 49

Commodity Primary Receipt Shipper Name Charge(s) Point(s) / PIN No(s).	Prim Schedule	ary Delivery Contract		Reservation Charges(s)
Reliant Energy Services, Inc. 1/3(a)/, 6/4/	FTS	5/01/2008 5/ through 10/31/2010	40,000	2/, 6/
Reliant Energy Services, Inc. 1/3(b)/, 6/8/	NSS	5/01/2008 8/ through 3/31/2011	,	\$3.07/Dth of MDQ
Reliant Energy Services, Inc. 1/3(b)/, 6/8/	NSS	5/01/2008 8/ through 3/31/2011	,	\$3.96/Dth of MDQ
Reliant Energy Services, Inc. 1/3(b)/, 6/8/	NSS	5/01/2008 8/ through 3/31/2011	·	\$3.07/Dth of MDQ
Reliant Energy Services, Inc. 1/3(b)/, 6/ Iowa-Illinois Gulf Coast Storage (PIN No. 10550) (14(b)/	REA/NGPL	Aurora, LP through	,	\$0.50/Dth of MDQ 6/

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable pro forma service agreement in Natural's FERC Gas Tariff.

^{2/} The Fixed Negotiated Monthly Base Reservation Rate shall be \$10.84/Dth of MDQ for the Peak Period and \$8.45/Dth of MDQ for the Off-Peak Period.

^{3/} Fixed Negotiated Base Commodity Rate:

⁽a) shall be \$0.0054/Dth during the Peak Period and \$0.0020/Dth MDQ during the Off-Peak Period.

⁽b) Shipper shall not be charged any base commodity rate for service provided under the Storage Agreement.

FERC Docket: RP08-389-000

Original Sheet No. 34D.02 Original Sheet No. 34D.02

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity	Primary Receipt Shipper Name	-	Term of Delivery Contract	Volume (Dth/d)		
Charge(s)	Point(s) / PIN No(s)					-
	llips Company 1/ n Bridgeport Plant	FTS Gulf South Goo	2/ (a) drich	10,000	\$6.6700 3/	
(PIN	No. 901850) 4/	(PIN No. 6297)	5/			
	llips Company 1/ n Bridgeport Plant	FTS Kansas Gas Ser	, , , ,	5,000	\$1.5708 3/	
(PIN	No. 901850) 4/	(PIN No. 3063)	5/			
Occidenta Express	l Energy Marketing, In	c. 1/ FTS-SW KMTP/NGPL	2/ (c)	41,000	\$6.0833 3/ 6/ 3/	Pony
(PIN	No. 11165) 4/	(PIN No. 5579)	5/ 6/			

- 1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ Negotiated Rate Term. The Negotiated Rate Term shall commence on the date the Segment One Expansion and the Amarillo to Gulf Coast Expansion ("Expansion") is

ready to receive firm service ("Effective Date") which is anticipated to be April 15, 2006. For

purposes of this section, the Expansion shall be considered ready to receive firm service when the pipeline and compression are ready for firm service. This negotiated rate term shall end:

- (a) five (5) years and six (6) months from the Effective Date;
- (b) five (5) years and one month from the Effective Date; and
- (c) five (5) years from the Effective Date.
- 3/ Negotiated Rates.
- (a) Eligible Transportation Quantity. The Negotiated Rate shall apply to a total aggregate maximum daily transportation quantity of Dth/day as shown above,

for all quantities transported under the FTS Agreement and any associated capacity release replacement agreements (the "Eligible Transportation

Quantity"). Any quantities transported in excess of the Eligible Transportation Quantity shall be billed at the greater of the Negotiated Rate or the

applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff.

FOOTNOTES APPLICABLE TO SHEET NO. 33C

- 3/ Negotiated Rates. (con't)
- (b) The Negotiated Rate shall apply only to service provided to Shipper by Natural under the FTS Agreement from the Eligible Receipt Points to the Eligible

Delivery Points during the Negotiated Rate Term. The Negotiated Rate shall apply only to a maximum daily firm transportation quantity equal to the

Eligible Transportation Quantity set forth above, for all quantities transported on a firm basis under the Transportation Agreement and any associated

capacity release replacement agreements. Any quantities transported on a firm basis in excess of the Eligible Transportation Quantity shall be billed at

the greater of the Negotiated Rate or all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

- (c) Monthly Base Negotiated Reservation Rate ("Negotiated Rate"). The Negotiated Rate set forth above per Dth of MDQ for peak and off-peak periods is effective for and through the Negotiated Rate Term. The Negotiated Rate shall be fixed for the term of the FTS Agreement.
- (d) General Negotiated Rate Limitations. The Negotiated Rate shall apply only to: (i) service provided to Shipper by Natural under the Transportation

Agreement from the Eligible Receipt Points to the Eligible Delivery Points during the Negotiated Rate Term; and (ii) a maximum daily firm transportation

quantity equal to the Eligible Firm Transportation Quantity set forth in footnote 3(a) above, for all quantities transported on a firm basis under the

Transportation Agreement and any associated capacity release replacement agreements. Shipper shall be charged the greater of the Negotiated Rate and all

applicable rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time, for any aggregate quantities

transported on a firm basis for Shipper and any associated capacity release replacement shippers which: (i) are in excess of the Eligible Firm

Transportation Quantity, or (ii) involve any receipt or delivery points which are not Eligible Receipt or Delivery Points.

- (e) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in
- Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payments to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

(f) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein or agreed to in writing by Natural, the greater of the Negotiated Rate and all applicable maximum rates, charges, surcharges and penalties of any

Negotiated Rate and all applicable maximum rates, charges, surcharges and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be

revised from time to time, shall apply to service provided to Shipper under the FTS Agreement, including without limitation all applicable: (i) Fuel and

Gas Lost and Unaccounted For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and

surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

4/ Eligible Receipt Points. The Negotiated Rates shall apply to transportation service provided from the following receipt point(s).

Primary Receipt Point. ConocoPhillips Company - Devon Bridgeport Plant, PIN No. 901850 (10,000 MDQ) and (5,000 MDQ), respectively;

Occidental Energy Marketing, Inc. - Pony Express, PIN No. 11165 (41,000 MDQ).

Secondary Receipt Points. The Negotiated Rates shall apply to service provided from all receipt points located in the following zones: ConocoPhillips (10,000

MDQ) - Texok and Midcontinent zones; ConocoPhillips (5,000 MDQ) - Midcontinent zone; Occidental - Permian, Midcontinent, Amarillo Mainline, and Texok zones.

FOOTNOTES APPLICABLE TO SHEET NO. 33C

5/ Eligible Delivery Points. The Negotiated Rates shall apply to transportation service provided to the following delivery point(s):

Primary Delivery Points: ConocoPhillips - Gulf South Goodrich, PIN 6297 (10,000 MDQ); ConocoPhillips - Kansas Gas Service, PIN 3063 (5,000 MDQ); Occidental - KMTP/NGPL, PIN 5579 (41,000 MDQ).

Secondary Delivery Points: The Negotiated Rates shall apply to service provided to the following secondary delivery points:

ConocoPhillips - (10,000 MDQ) - All pooling, storage, Hinshaw and Intrastate Pipeline Interconnects and Interstate Interconnect points

located within the zones traversed by the primary path of the Transportation Agreement, excluding the Lone Star Pipeline Interconnect located in

the Midcontinent Zone and the Kinder Morgan Texas Pipeline Interconnect located in the Texok Zone;

 ${\tt ConocoPhillips - (5,000 MDQ) - All pooling, storage, Hinshaw and Intrastate Pipeline}$

Interconnects and Interstate Interconnect points located
within the zones traversed by the primary path of the Transportation Agreement,
excluding the Lone Star Pipeline Interconnect located in the
Midcontinent Zone;

Occidental - Subject to the provisions in footnote 6 below regarding incremental rate secondary delivery points, the Negotiated Rate shall apply to all delivery points located in the Midcontinent and Texok zones.

6/ Incremental Rate Secondary Delivery Points. In addition to the Negotiated Rate, Shipper shall pay Natural an incremental volumetric daily reservation rate for all quantities transported on a firm basis on any day to the Incremental Rate Secondary Delivery

Points set forth herein for service to such point.

The following incremental volumetric daily reservation rates will apply to firm service provided to the secondary delivery points set forth below under the corresponding Point Tier:

Tier 1: Rate: \$0.030/Dth; Secondary Delivery Points: All points located in the LA zone.

Tier 2: Rate: \$0.035/Dth (Off-Peak) and \$0.0625/Dth (Peak); Secondary Delivery Points: Nicor (PIN No. 9258), NIPSCO (PIN No. 909260), PGL&C (PIN No. 909285),

MidAmerican (PIN No. 10568), and all existing interstate pipeline interconnects in the Market Delivery Zone, excluding MRT, Illini Carrier,

Guardian and ANR Joliet.

Service provided on a firm basis to all other secondary delivery points not specifically set forth in this footnote 6 or in footnote 5, above, regarding

Secondary Delivery Points, will be provided at the greater of the Negotiated Rate or the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may

be revised from time to time, for service to such points.

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

TRANSACTIONS

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s).	Schedule	Term of ary Delivery Contract s) / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)
				-	
Nicor Gas 2/	s Company 1/ 2/	FTS 2	4/01/2008 2/ through 3/31/2011	40,000	2/
Nicor Gas 2/	s Company 1/ 2/	FTS 2	4/01/2008 2/ through 3/31/2011	20,000	2/

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable Form of Service Agreement in Natural's FERC Gas Tariff.

2/ This information is set out in the executed Transportation Rate Schedule FTS Agreement, and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

use.

Sheet No. 34 has been reserved for future

future use.

TRANSACTIONS

Sheet No. 34C.01 has been reserved for

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s).	Schedule	Term of mary Delivery Contract / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)
Merrill	Lynch Commodities, Inc. 1/6/	NSS 6/	4/01/2008 through 3/31/2011 2/	 22,526 3/	\$3.07/Dth of MDQ 4/ 5/

⁻⁻⁻⁻⁻

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable pro forma service agreement in Natural's FERC Gas Tariff.

^{2/} Negotiated Rate Term. The negotiated rates set forth herein shall be in force during a term commencing on April 1, 2008 through March 31, 2011.

^{3/} NSS Agreement - Eligible Firm Quantities. The Negotiated Rates set forth herein shall apply to storage service provided by Natural to Shipper under Rate

Schedule $\overline{\text{NSS}}$ with a Contract Maximum Storage Volume ("MSV") of One Million Six Hundred Eighty Nine Thousand Four Hundred Fifty (1,689,450) Dth and a Contract

MDQ of Twenty Two Thousand Five Hundred Twenty Six (22,526) Dth per day. MDQ of 22,526 Dth per day and an MSV of 1,689,450 Dth on the Amarillo Leg, and an MDQ

of 0 Dth per day and a MSV of 0 Dth on the Gulf Coast Leg, with an Amarillo Percentage equal to one hundred percent (100%) and a Gulf Coast Percentage equal to zero (0%).

- 4/ Negotiated Monthly Base Reservation Rate. The Negotiated Monthly Base Reservation Rte per Dth of MDQ (Peak and Off-Peak periods) shall apply only to service provided by Natural to Shipper under the Rate Schedule NSS Agreement, as amended, subject to all
- other terms and conditions set forth herein.
 - 5/ Negotiated Rate Terms, Conditions and Other Charges.
- (a) General Negotiated Rate Limitations. The Negotiated Rates shall apply only to service provided to Shipper by Natural under the NSS Agreement during the
- Negotiated Rate Term up to a maximum aggregate firm quantity equal to the respective Eligible Contract MDQ and the Eligible Contract MSV set forth in
- footnote three above, for all quantities stored on a firm basis under the NSS Agreement and any associated capacity release replacement agreements.
- Shipper shall be charged all applicable maximum rates, charges and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time,
- for any aggregate quantities stored on a firm basis for Shipper and any associated capacity release replacement shippers which are in excess of the
 - respective Eligible Contract MDQ or the Eligible Contract MSV, as applicable.

FERC Docket: RP99-176-190

Original Sheet No. 34E.02 Original Sheet No. 34E.02

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity	Eligible Primary Receipt	Rate	Term of Primary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract s) / PIN No(s).	(Dth/d)	Charge(s)
Niska Ga 2/	as Storage LLC 1/	NSS	4/01/2009	66,667	2/
			through 3/31/2012		

⁻⁻⁻⁻⁻

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} This information is set out in the executed Storage Rate Schedule NSS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP99-176-193

Original Sheet No. 34L.02 Original Sheet No. 34L.02

FOOTNOTES APPLICABLE TO SHEET NO. 34L.01

4/ Fixed Negotiated Base Commodity Rates (\$/Dth):

	Off-Peak	Peak
	April 1	November 1
Agreement	through	through
No.	October 31	March 31
111887	\$0.0043	\$0.0186
110492	\$0.0043	\$0.0181
110493	N/A	\$0.0201

^{5/} This information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Firm Transportation Negotiated Rate Agreement, which are currently on file with FERC.

FERC Docket: RP99-176-161

Original Sheet No. 34C.03 Original Sheet No. 34C.03 : Pending

FOOTNOTES APPLICABLE TO SHEET NO.

34C.02

4/ Eligible Receipt Points:

(a) Eligible Primary Receipt Points:

TPC/NGPL Gage (PIN 902900) 40,000 Dth/d of MDQ.

(b) Eligible Secondary Receipt Points:

All secondary receipt points permitted under the transportation service agreement.

FRSS: None.

5/ Eligible Delivery Points:

(a) Eligible Primary Delivery Points:

REA/NGPL Aurora (PIN 37435) 40,000 Dth/d of MDQ (each April 1 through October 31); Columbia Gulf Chalkley (PIN 11295) 40,000 Dth/d of MDQ (each November 1

through March 31)

(b) Eligible Secondary Delivery Points:

All secondary delivery points permitted under the transportation service agreement.

FRSS:

6/ Other Charges:

(a) General Negotiated Rate Limitations. The Negotiated Monthly Base Reservation Rate and Negotiated Base Commodity Rates, if applicable, shall apply only to:

(i) service provided to Shipper by Natural under the Transportation Agreement from the Eligible Receipt Points to the Eligible Delivery Points during

the Negotiated Rate Term; and (ii) an aggregate maximum daily firm transportation

quantity equal to the respective Eligible Firm Transportation

Quantity set forth herein, for all quantities transported on a firm basis under the Transportation Agreement and any associated capacity release

replacement agreements. For any aggregate quantities transported on a firm basis on any day for Shipper and any associated capacity release

replacement shipper which: (i) are in excess of the applicable Eligible Firm Transportation Quantity; or (ii) involve any receipt or delivery points

which are not Eligible Receipt or Delivery Points, Shipper shall be charged the greater of: (a) the 100% load factor daily rate equivalent of the

Negotiated Monthly Base Reservation Rate and the Negotiated Base Commodity Rate; and (b) the 100% load factor daily rate equivalent of the applicable

maximum base reservation and base commodity rates set forth in Natural's FERC Gas Tariff, as may be revised from time to time, as well as all

applicable Additional Rates, Charges and Surcharges described in (c) below.

NSS: Service provided to Shipper by Natural under the Storage Agreement during the Negotiated Rate Term, up to a maximum aggregate firm quantity equal to

the respective Eligible Contract MDQ and the Eligible Contract MSV set forth herein, for all quantities stored on a firm basis under the Storage

Agreement and any associated capacity release replacement agreements. For any aggregate quantities stored on a firm basis for Shipper and any

associated capacity release replacement shippers which are in excess of the Contract MDQ or the Contract MSV, as applicable, Shipper shall be charged

the greater of: (a) the 100% load factor equivalent of the Negotiated Monthly Base

Reservation Rate; and (b) the 100% load factor equivalent of the applicable maximum monthly base reservation rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

FERC Docket: RP08-389-000

Original Sheet No. 34D.03 Original Sheet No. 34D.03

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

._____

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s)	Primary Del Schedule (Contract	Volume (Dth/d)		n
	llips Company 1/ n Bridgeport Plant	FTS Gulf South Goodric	2/ (a) ch	10,000	\$6.6700 3/	
(PIN	No. 901850) 4/	(PIN No. 6297) 5/				
	llips Company 1/ n Bridgeport Plant	FTS Kansas Gas Service	2/ (b)	5,000	\$1.5708 3/	
(PIN	No. 901850) 4/	(PIN No. 3063) 5/				
Occidenta Express	l Energy Marketing, Ind	c. 1/ FTS-SW KMTP/NGPL	2/ (c)	41,000	\$6.0833 3/ 6/ 3/	Pony
(PIN	No. 11165) 4/	(PIN No. 5579) 5/	6/			

- The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- Negotiated Rate Term. The Negotiated Rate Term shall commence on the date the Segment One Expansion and the Amarillo to Gulf Coast Expansion ("Expansion") is

ready to receive firm service ("Effective Date") which is anticipated to be April 15, 2006. For purposes of this section, the Expansion shall be considered ready to receive firm service when the pipeline and compression are ready for firm service. This

negotiated rate term shall end:

- (a) five (5) years and six (6) months from the Effective Date;
- (b) five (5) years and one month from the Effective Date; and
- (c) five (5) years from the Effective Date.
- 3/ Negotiated Rates.
- (a) Eligible Transportation Quantity. The Negotiated Rate shall apply to a total aggregate maximum daily transportation quantity of Dth/day as shown above,

for all quantities transported under the FTS Agreement and any associated capacity release replacement agreements (the "Eligible Transportation

Quantity"). Any quantities transported in excess of the Eligible Transportation Quantity shall be billed at the greater of the Negotiated Rate or the

applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff. FOOTNOTES APPLICABLE TO SHEET NO. 33C _____

- 3/ Negotiated Rates. (con't)
- The Negotiated Rate shall apply only to service provided to Shipper by Natural under the FTS Agreement from the Eligible Receipt Points to the Eligible

Delivery Points during the Negotiated Rate Term. The Negotiated Rate shall apply only to a maximum daily firm transportation quantity equal to the

Eligible Transportation Quantity set forth above, for all quantities transported on a firm basis under the Transportation Agreement and any associated

capacity release replacement agreements. Any quantities transported on a firm basis in excess of the Eligible Transportation Quantity shall be billed at

the greater of the Negotiated Rate or all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

- (c) Monthly Base Negotiated Reservation Rate ("Negotiated Rate"). The Negotiated Rate set forth above per Dth of MDQ for peak and off-peak periods is effective for and through the Negotiated Rate Term. The Negotiated Rate shall be fixed for the term of the FTS Agreement.
- (d) General Negotiated Rate Limitations. The Negotiated Rate shall apply only to: (i) service provided to Shipper by Natural under the Transportation

Agreement from the Eligible Receipt Points to the Eligible Delivery Points during the Negotiated Rate Term; and (ii) a maximum daily firm transportation

quantity equal to the Eligible Firm Transportation Quantity set forth in footnote 3(a) above, for all quantities transported on a firm basis under the

Transportation Agreement and any associated capacity release replacement agreements. Shipper shall be charged the greater of the Negotiated Rate and all

applicable rates, charges, and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time, for any aggregate quantities

transported on a firm basis for Shipper and any associated capacity release replacement shippers which: (i) are in excess of the Eligible Firm

Transportation Quantity, or (ii) involve any receipt or delivery points which are not Eligible Receipt or Delivery Points.

- (e) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in
- Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payments to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

(f) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein or agreed to in writing by Natural, the greater of the Negotiated Rate and all applicable maximum rates, charges, surcharges and penalties of any

Negotiated Rate and all applicable maximum rates, charges, surcharges and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be

revised from time to time, shall apply to service provided to Shipper under the FTS Agreement, including without limitation all applicable: (i) Fuel and

Gas Lost and Unaccounted For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and

surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

4/ Eligible Receipt Points. The Negotiated Rates shall apply to transportation service provided from the following receipt point(s).

Primary Receipt Point. ConocoPhillips Company - Devon Bridgeport Plant, PIN No. 901850 (10,000 MDQ) and (5,000 MDQ), respectively;

Occidental Energy Marketing, Inc. - Pony Express, PIN No. 11165 (41,000 MDQ).

Secondary Receipt Points. The Negotiated Rates shall apply to service provided from all receipt points located in the following zones: ConocoPhillips (10,000

MDQ) - Texok and Midcontinent zones; ConocoPhillips (5,000 MDQ) - Midcontinent zone; Occidental - Permian, Midcontinent, Amarillo Mainline, and Texok zones.

FOOTNOTES APPLICABLE TO SHEET NO. 33C

5/ Eligible Delivery Points. The Negotiated Rates shall apply to transportation service provided to the following delivery point(s):

Primary Delivery Points: ConocoPhillips - Gulf South Goodrich, PIN 6297 (10,000 MDQ); ConocoPhillips - Kansas Gas Service, PIN 3063 (5,000 MDQ); Occidental - KMTP/NGPL, PIN 5579 (41,000 MDQ).

Secondary Delivery Points: The Negotiated Rates shall apply to service provided to the following secondary delivery points:

ConocoPhillips - (10,000 MDQ) - All pooling, storage, Hinshaw and Intrastate Pipeline Interconnects and Interstate Interconnect points located within the zones traversed by the primary path of the Transportation

Agreement, excluding the Lone Star Pipeline Interconnect located in the Midcontinent Zone and the Kinder Morgan Texas Pipeline Interconnect located in the Texok Zone;

ConocoPhillips - (5,000 MDQ) - All pooling, storage, Hinshaw and Intrastate Pipeline

Interconnects and Interstate Interconnect points located
within the zones traversed by the primary path of the Transportation Agreement,
excluding the Lone Star Pipeline Interconnect located in the
Midcontinent Zone;

Occidental - Subject to the provisions in footnote 6 below regarding incremental rate secondary delivery points, the Negotiated Rate shall apply to all delivery points located in the Midcontinent and Texok zones.

6/ Incremental Rate Secondary Delivery Points. In addition to the Negotiated Rate, Shipper shall pay Natural an incremental volumetric daily reservation rate for all quantities transported on a firm basis on any day to the Incremental Rate Secondary Delivery

Points set forth herein for service to such point.

The following incremental volumetric daily reservation rates will apply to firm service provided to the secondary delivery points set forth below under the

Tier 1: Rate: \$0.030/Dth; Secondary Delivery Points: All points located in the LA zone.

Tier 2: Rate: \$0.035/Dth (Off-Peak) and \$0.0625/Dth (Peak); Secondary Delivery Points: Nicor (PIN No. 9258), NIPSCO (PIN No. 909260), PGL&C (PIN No. 909285),
MidAmerican (PIN No. 10568), and all existing interstate pipeline interconnects in

the Market Delivery Zone, excluding MRT, Illini Carrier, Guardian and ANR Joliet.

Service provided on a firm basis to all other secondary delivery points not specifically set forth in this footnote 6 or in footnote 5, above, regarding

Secondary Delivery Points, will be provided at the greater of the Negotiated Rate or the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may

be revised from time to time, for service to such points.

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

corresponding Point Tier:

PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt		ary Delivery	(D) 1. (4)	Q1 (-)
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(s	Contract s) / PIN No(s).	(Dth/d)	Charge(s)
Nicor Gas	s Company 1/ 2/	FTS	4/01/2008	40,000	2/
			through 3/31/2011		
Nicor Gas	S Company 1/	FTS	4/01/2008	20,000	2/
,	·		through 3/31/2011		

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable Form of Service Agreement in Natural's FERC Gas Tariff.

2/ This information is set out in the executed Transportation Rate Schedule FTS Agreement, and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

 $\label{eq:Sheet No. 34 has been reserved for future} \\ \text{use.}$

future use.

TRANSACTIONS

Sheet No. 34C.01 has been reserved for

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s).	Schedule	Term of imary Delivery Contract / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)
Merrill :	Lynch Commodities, Inc. 1/	NSS 6,	4/01/2008 / through 3/31/2011 2/	22,526 3/	\$3.07/Dth of MDQ 4/ 5/

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable pro forma service agreement in Natural's FERC Gas Tariff.

^{2/} Negotiated Rate Term. The negotiated rates set forth herein shall be in force during a term commencing on April 1, 2008 through March 31, 2011.

^{3/} NSS Agreement - Eligible Firm Quantities. The Negotiated Rates set forth herein shall apply to storage service provided by Natural to Shipper under Rate

Schedule $\overline{\text{NSS}}$ with a Contract Maximum Storage Volume ("MSV") of One Million Six Hundred Eighty Nine Thousand Four Hundred Fifty (1,689,450) Dth and a Contract

MDQ of Twenty Two Thousand Five Hundred Twenty Six (22,526) Dth per day. MDQ of 22,526 Dth per day and an MSV of 1,689,450 Dth on the Amarillo Leg, and an MDQ

of 0 Dth per day and a MSV of 0 Dth on the Gulf Coast Leg, with an Amarillo Percentage equal to one hundred percent (100%) and a Gulf Coast Percentage equal to zero (0%).

4/ Negotiated Monthly Base Reservation Rate. The Negotiated Monthly Base Reservation Rte per Dth of MDQ (Peak and Off-Peak periods) shall apply only to service

provided by Natural to Shipper under the Rate Schedule NSS Agreement, as amended, subject to all other terms and conditions set forth herein.

- 5/ Negotiated Rate Terms, Conditions and Other Charges.
- (a) General Negotiated Rate Limitations. The Negotiated Rates shall apply only to service provided to Shipper by Natural under the NSS Agreement during the

Negotiated Rate Term up to a maximum aggregate firm quantity equal to the respective Eligible Contract MDQ and the Eligible Contract MSV set forth in

footnote three above, for all quantities stored on a firm basis under the NSS Agreement and any associated capacity release replacement agreements.

Shipper shall be charged all applicable maximum rates, charges and surcharges set forth in Natural's FERC Gas Tariff, as may be revised from time to time,

for any aggregate quantities stored on a firm basis for Shipper and any associated capacity release replacement shippers which are in excess of the $\frac{1}{2}$

respective Eligible Contract MDQ or the Eligible Contract MSV, as applicable.

FOOTNOTES APPLICABLE TO SHEET NO.

34D.02

- 5/ Negotiated Rate Terms, Conditions and Other Charges (con't)
- (b) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in

Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas

Tariff. Shipper shall only be responsible for payment to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

- (c) Applicable Maximum Rates, Charges and Surcharges. Unless otherwise expressly provided herein or agreed to in writing by Natural, the greater of the
- Negotiated Rates and all applicable maximum rates, charges, surcharges and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be

revised from time to time, shall apply to service provided to Shipper under the NSS Agreement including without limitation all applicable: (i) Storage

Fuel charges; (ii) authorized and unauthorized overrun charges; and (iii) sourcing charges.

6/ Receipt and Delivery Points. The Receipt Points for injection and the Delivery Points for withdrawal under the Rate Schedule NSS Agreement shall be the

Storage Points as defined in Section 9.3 of the General Terms and Conditions set forth in Natural's FERC Gas Tariff.

FERC Docket: RP99-176-189

Original Sheet No. 34E.03 Original Sheet No. 34E.03

TRANSACTIONS	STATEMENT OF NEGOTIATED RATE	
	PURSUANT TO SECTION 4	9

Commodity	Eligible Primary Receipt	Rate Eligible B	Term of Primary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract s) / PIN No(s).	(Dth/d)	Charge(s)
Masefiel 3/	d Natural Gas, Inc. 1/ 3/	FTS	4/01/2009 3/ through	2/	3/
			10/31/2010		

April 1, 2009 - June 30, 2009 13,500 Dth/d July 1, 2009 - October 31, 2009 6,130 Dth/d November 1, 2009 - March 31, 2010 0 Dth/d April 1, 2010 - June 30, 2010 13,500 Dth/d July 1, 2010 - October 31, 2010 6,120 Dth/d

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} Eligible Firm Transportation Quantity:

^{3/} This information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

Effective Date: 05/01/2009 Status: Pending

FERC Docket: RP99-176-197

First Revised Sheet No. 34E.03 First Revised Sheet No. 34E.03

Superseding: Original Sheet No. 34E.03

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

.____

Commodity	Eligible Primary Receipt	_	Term of Primary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(s	Contract s) / PIN No(s).	(Dth/d)	Charge(s)
Iberdrol 3/	a Renewables, Inc. 1/ 3/	FTS	5/01/2009 3/ through 10/31/2010	2/	3/
Iberdrol 3/	a Renewables, Inc. 1/ 3/	FTS	5/01/2009 3/ through 12/31/2012	10,000	3/

May 1, 2009 - June 30, 2009 13,500 Dth/d July 1, 2009 - October 31, 2009 6,130 Dth/d November 1, 2009 - March 31, 2010 0 Dth/d April 1, 2010 - June 30, 2010 13,500 Dth/d July 1, 2010 - October 31, 2010 6,120 Dth/d

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} Eligible Firm Transportation Quantity:

^{3/} This information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP99-176-193

Original Sheet No. 34L.03 Original Sheet No. 34L.03

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity	Eligible Primary Receipt	Rate	Term of rimary Delivery	Volume	Reservation
Commodity	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s) / PIN No(s).		
Northern	Indiana Public Service	DSS	4/01/2009	209,550	\$5.70/Dth
2/	2/		2/	,	
Compan	y 1/		Through 3/31/2013		of MDQ 2/

^{1/} The Storage Rate Schedule DSS Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} This information is set out in the executed Storage Rate Schedule DSS Agreement with a negotiated rate exhibit, which is currently on file with the FERC.

FERC Docket: RP99-176-161

Original Sheet No. 34C.04 original Sheet No. 34C.04 : Pending

FOOTNOTES APPLICABLE TO SHEET NO.

34C.02

._____

6/ Other Charges: (con't)

FRSS: (i)service provided to Shipper by Natural under the FRSS Agreement from the Eligible Receipt Points to the Eligible Delivery Points during the

Negotiated Rate Term; and (ii) maximum aggregate Eligible Firm Quantities equal to the respective Eligible Contract MDQ and the Eligible MSWQ set

forth herein, for all quantities stored (and/or transported, if the FRSS Agreement is for Delivered Service) on a firm basis under the FRSS $\,$

Agreement and any associated capacity release replacement agreements. Shipper shall be charged the greater of: (a) the 100% load factor equivalent

of the Negotiated Monthly Base Reservation Rate and (b) the 100% load factor equivalent of the applicable maximum base reservation rate set forth in

Natural's FERC Gas Tariff, as maybe revised from time to time, for any aggregate quantities stored (and/or transported on a firm basis for Shipper

if the FRSS Agreement is for Delivered Service) and any associated capacity release replacement shippers which: (i) are in excess of the Contract

 $\,$ MDQ or the Contract MSV, as applicable; or (ii) involve any receipt or delivery points which are not Eligible Receipt or Delivery Points.

(b) Discountable Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in

Natural's FERC Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to

discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges").

Natural shall discount any particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas Tariff.

Shipper shall only be responsible for payment to Natural of any portion of such Discountable Third Party Surcharge which Natural is not permitted to discount and for which Natural bills Shipper.

(c) Additional Rates, Charges and Surcharges. In addition to the Negotiated Monthly Base Reservation Rate and the Negotiated Base Commodity Rate, and unless

otherwise expressly provided herein or agreed to in writing by Natural, Shipper shall also pay Natural all other applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be

surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to time, including without limitation all applicable maximum:

FTS: (i) Fuel and Gas Lost and Unaccounted For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation charges and surcharges; (iv) commodity charges and surcharges; (v) ACA surcharges and (vi) gathering charges, offshore charges and lateral line charges.

NSS: (i) Storage Fuel Charges; (ii) authorized and unauthorized overrun charges and (iii) sourcing charges.

FRSS: (i) Storage Fuel Charges (and if the FRSS Agreement is for Delivered Service, a Transportation Fuel Charge comprised of a Fuel Gas Charge and a Gas α

Lost and Unaccounted For Charge); (ii) authorized and unauthorized overrun charges; and (iii) reservation surcharges; (iv) any applicable commodity rates, charges and surcharge; (v) Balancing Service Charges; and (vi) ACA surcharges.

races, charges and surcharge, (v) surancing service charges, and (vi) non surcharges.

Notwithstanding the foregoing sentence, to the extent that any such additional maximum rate, charge, surcharge, or penalty is derived from the applicable base

reservation rate or applicable base commodity rate set forth in Natural's FERC Gas Tariff, then Natural shall derive such additional rate, charge, surcharge or

penalty utilizing the greater of: (a) the Negotiated Monthly Base Reservation Rate or Negotiated Base Commodity Rate, if applicable, and (b) the applicable

maximum base reservation rate or applicable maximum base commodity rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

FERC Docket: RP99-176-188

Original Sheet No. 34E.04 Original Sheet No. 34E.04

TRANSACTIONS	STATEMENT OF NEGOTI	ATED RATE
	PURSUANT TO	SECTION 49

Commodity	Primary Receipt	Rate	Term of ary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract / PIN No(s).	(Dth/d)	Charge(s)
Kinder M 3/	Norgan Tejas Pipeline LLC 1/	NSS	4/01/2009	53,334 2/	3/
			through 3/31/2012		

^{1/} The Storage Rate Schedule NSS Agreement, as amended, with Shipper does not deviate in any material
respect from the applicable form of service agreement in
 Natural's FERC Gas Tariff.

^{2/} The Negotiated Rates set forth herein shall apply only to service provided to Shipper by Natural under the Rate Schedule NSS Agreement during the Negotiated
Rate Term, up to a Maximum Storage Volume (MSV) of Four Million Fifty (4,000,050) Dth and a MDQ of Fifty Three Thousand Three Hundred Thirty Four (53,334)
Dth/d on the Gulf Coast Leg 100% (the "Storage Agreement").

^{3/} This information is set out in the executed Storage Rate Schedule NSS Agreement, as amended, with corresponding Negotiated Rate Exhibit which are currently on file with the FERC.

FERC Docket: RP99-176-161

Original Sheet No. 34C.05 Original Sheet No. 34C.05 : Pending

FOOTNOTES APPLICABLE TO SHEET NO.

34C.02

7/ Eligible Firm Quantities:

(a) Contract MDQ of Thirty Thousand (30,000) Dth per day and a Contract Maximum Storage Volume

("MSV") of Two Million Two Hundred Fifty Thousand (2,250,000) Dth.

Leg Rights consist of: MDQ of 0 Dth per day and MSV of 0 Dth on the Amarillo Leg and MDQ of 30,000 Dth per day and MSV of 2,250,000 Dth on the Gulf Coast Leg,

with an Amarillo Percentage equal to zero (0%) and a Gulf Coast Percentage equal to one hundred percent (100%).

(b) Contract MDQ of Thirteen Thousand Three Hundred Thirty Three (13,333) Dth per day and a Contract Maximum Storage Volume ("MSV) of Nine Hundred Ninety Nine

Thousand Nine Hundred Seventy Five (999,975) Dth. Leg Rights consist of: MDQ of 0 Dth per day and MSV of 0 Dth on the Amarillo Leg and MDQ of 13,333 Dth per

day and MSV of 999,975 Dth on the Gulf Coast Leg, with an Amarillo Percentage equal to zero (0%) and a Gulf Coast Percentage equal to one hundred percent (100%).

(c)Contract MDQ of Two Thousand Six Hundred Sixty Seven (2,667) Dth per day and a Contract Maximum Storage Volume ("MSV) of Two Hundred Thousand Twenty Five

(200,025) Dth. Leg Rights consist of: MDQ of 0 Dth per day and MSV of 0 Dth on the Amarillo Leg and MDQ of 2,667 Dth per day and MSV of 200,025 Dth on the

Gulf Coast Leg, with an Amarillo Percentage equal to zero (0%) and a Gulf Coast Percentage equal to one hundred percent (100%).

(d) Contract MDQ of Twenty Two Thousand (22,000) Dth per day and Contract Maximum Storage Withdrawal Quantity ("MSWQ") of Four Hundred Forty Thousand (440,000)

8/ Receipt and Delivery Points. The Receipt Points for injection and the Delivery Points for withdrawal under the Rate Schedule NSS Agreement shall be the Storage

Points as defined in Section 9.3 of the General Terms and Conditions set forth in Natural's FERC Gas Tariff.

FERC Docket: RP10-532-000

First Revised Sheet No. 34E.05 First Revised Sheet No. 34E.05

Superseding: Original Sheet No. 34E.05

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Prima:	Term of ry Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Point(s)	Contract / PIN No(s).	(Dth/d)	Charge(s)
				-	
J.P. Mor 3/	rgan Ventures Energy 3/	NSS 3/	4/01/2009	74,667	2/ 3/
Corpor	ration 1/		through 3/31/2012		
J.P. Mor 3/	gan Ventures Energy 3/	NSS 3/	4/01/2010	58,304	4/ 3/
Corpor	ration		through 3/31/2013		

^{1/} The Storage Rate Schedule NSS Agreement, as amended, with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The Negotiated Rates set forth herein shall apply only to service provided to Shipper by Natural under the Rate Schedule NSS Agreement during the Negotiated Rate Term, up to a Maximum Storage Volume (MSV) of Five Million Six Hundred Thousand Twenty Five (5,600,025) Dth and a MDQ of Seventy Four Thousand Six Hundred

Sixty Seven (74,667) Dth/d on the Gulf Coast Leg 100% (the "Storage Agreement").

^{3/} This information is set out in the executed Storage Rate Schedule NSS Agreement, as amended, with corresponding Negotiated Rate Exhibit which are currently on file with the FERC.

^{4/} The Negotiated Rates set forth herein shall apply only to service provided to Shipper by Natural under the Rate Schedule NSS Agreement during the Negotiated Rate Term, up to a MSV of Four Million Three Hundred Seventy Two Thousand Eight Hundred

^(4,372,800) Dth and a MDQ of Fifty Eight Thousand Three Hundred Four

FERC Docket: RP10-536-000

Original Sheet No. 34C.06 Original Sheet No. 34C.06

MD 3 NO 3 CM T ON C	STATEMENT OF NEGOTIATED I	KATE
TRANSACTIONS	PURSUANT TO SECTION	าก 4.9
	IONOGINI IO BECII)IV 1J

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt	Primar	ry Delivery		
	Shipper Name	Schedule	Contract	(Dth/d)	Charges(s)
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).		
0	al Bassass Marshatinas Tar	MOO	4/01/2010	(2, (7,	1 /
1/	al Energy Marketing, Inc.	NSS 1/	4/01/2010	62 , 667	1/
1/	1/	1/	. 1		
			through		
			3/31/2013		

^{1/} This information is set out in the executed Storage Rate Schedule NSS Agreement, as amended, and the corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 35 Original Sheet No. 35: Pending

STATEMENT OF NEGOTIATED RATE TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Term of Primary Delivery	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule Contract Point(s) / PIN No(s).	(Dth/d)	Charge(s)
			-	
	Marketing and Trade 1/ 4/, 6/, 7/, 8/	FTS 12/01/2002 Illinois Power/Centralia through	20,000	2/, 3/, 5/
		(PIN No. 900170) 11/30/2006 5/, 6/, 7/, 8/		
	Marketing and Trade 1/ Sabine (PIN No. 3592)	FTS 12/01/2002 7/, 8/, 11/, 12/ through	20,000	3/, 9/, 11/
	7/, 8/, 10/, 12/	11/30/2006		

1/ The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

2/ Negotiated Reservation and Commodity Rates. The Negotiated Reservation Rate and Negotiated Commodity Rate set forth below shall apply only to service provided

to Shipper by Natural under the FTS Agreement from the Eligible Receipt Points to the Eligible Delivery Points set forth below, subject to all other applicable terms and conditions set forth herein.

Negotiated Reservation Rate: Peak 5.8616/Dth of MDQ (12/1/2002 through 3/31/2003); Peak 6.0021/Dth of MDQ (11/1/2003 through 11/30/2006); and Off-Peak 6.6858/Dth of MDQ (4/1/2003 through 11/30/2006).

Negotiated Commodity Rate: Peak \$0.0146/Dth and Off-Peak \$0.0034/Dth.

- 3/ Other Negotiated Rate Provisions:
- (i) Eligible Transportation Quantity. The Negotiated Reservation Rate and Negotiated Commodity Rate shall apply only to total aggregate deliveries on any day $\frac{1}{2}$

of a quantity equal to the FTS Agreement MDQ, for all quantities transported under the FTS Agreement and any associated capacity release replacement

agreements. Unless otherwise agreed in writing by Natural, any quantities transported in excess of the Eligible Transportation Quantity shall be billed at all applicable maximum rates, charges and surcharges set forth in Natural's FERC Gas

at all applicable maximum rates, charges and surcharges set forth in Natural's FERC Ga Tariff.

- (ii) Authorized Overrun Service. Unless Natural otherwise expressly agrees in writing, Authorized Overrun Service provided to Shipper under the FTS Agreement shall be billed at the applicable maximum Authorized Overrun Rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.
- (iii) Fuel Gas and Gas Lost and Unaccounted For. In addition to the Negotiated Rates, Shipper shall be charged for Fuel Gas and Gas Lost and Unaccounted For in accordance with the applicable provisions of Natural's FERC Gas Tariff, as may be revised from the contract to the contract of the contract of
- (iv) GRI Surcharges. The Negotiated Rates reflect an initial discount of all GRI reservation and commodity surcharges otherwise payable by Shipper; provided,
 however, that in the event Natural is required to collect such GRI Surcharges from Shipper and to pay such GRI Surcharges to the FERC or to any third
 party, then Shipper shall pay Natural any such amounts which are required to be so collected

and paid.

FERC Docket: RP08-319-000

Original Sheet No. 35A Original Sheet No. 35A: Pending

FOOTNOTES APPLICABLE TO SHEET NO. 35

3/ Con't

(v) Applicable Maximum Rates, Charges and Surcharges. In accordance with Section 6.3 of Natural's Rate Schedule FTS, the Negotiated Monthly Inclusive Reservation Rates are inclusive of all Account 858 surcharges set forth in Natural's

Reservation Rates are inclusive of all Account 858 surcharges set forth in Natural's FERC Gas Tariff which are or may be applicable to service

provided to Shipper under the FTS Agreement. The Negotiated Monthly Inclusive Reservation Rates shall be applicable at all times during the

Negotiated Rate Term. Unless otherwise expressly provided herein, all applicable maximum rates, charges, surcharges and penalties of any nature set

forth in Natural's FERC Gas Tariff shall apply to service provided to Shipper under the FTS Agreement, including without limitation all applicable gathering charges, offshore charges, lateral line charges and ACA surcharges.

4/ Eligible Receipt Points: The Negotiated Rates shall only apply to transportation service provided from the following receipt points and only up to the point MDQ indicated below for each point:

Primary Receipt Points: Red River No. 2, PIN NO. 25070 (2,325 Dth of MDQ 12/1/2002 - 3/31/2003); Chico Plt., PIN NO. 901501 (17,675 Dth of MDQ 12/1/2002 - 3/31/2003) and (20,000 Dth of MDQ 4/1/2003 - 11/30/2006).

Secondary Receipt Points: The Negotiated Rates shall apply to service provided from all secondary receipt points permitted under the FTS Agreement.

5/ Eligible Delivery Points: The Negotiated Rates shall only apply to transportation service provided to the following delivery points:

Primary Delivery Points: Illinois Power/Centralia, PIN No. 900170 (20,000 Dth of MDQ).

Secondary Delivery Points: The Negotiated Rates shall apply to NICOR (PIN No. 9258), NIPSCO (PIN No. 909260), PGL&C (PIN No. 909285), MidAmerican (PIN No.

10568), Illinois Power (PIN No. 909999), ANR-Joliet #2 (PIN No. 904758) and all storage and pooling points within the zones traversed by the primary path as permitted by the FTS Agreement.

Incremental Rate Secondary Delivery Points: Incremental Rate Secondary Delivery Points shall include Midwestern Kankakee (PIN Number 25400) with an

additional incremental daily volumetric reservation rate of \$.06 per Dth applied to all quantities transported on any day to Midwestern Kankakee (PIN No. 25400) for the period commencing February 21, 2003 and ending November 30, 2006.

- 6/ Eligible Primary Pathing. Shipper's primary path under the FTS Agreement will be from Natural's Midcontinent Zone northeast through Segment 14 and south to Natural's interconnect with Illinois Power in Segment 28.
- 7/ Aggregate Receipt/Delivery Quantity Limitation. The Negotiated Rates shall apply to all quantities received or delivered by Natural for Shipper's account

in any particular transportation zone which are transported on a primary or secondary firm basis, to the extent that such aggregate firm receipt or

delivery quantities do not exceed the aggregate amount of Shipper's Eligible Point MDQs set forth above for receipts or deliveries, as applicable, in such

zone. Unless Natural otherwise expressly agrees in writing, service provided in excess of the applicable zone quantity limitation shall be charged all

applicable maximum rates, charges and surcharges set forth in Natural's FERC Gas Tariff.

FERC Docket: RP08-319-000

TRANSACTIONS

Original Sheet No. 35B Original Sheet No. 35B : Pending

STATEMENT OF NEGOTIATED RATE

PURSUANT TO SECTION 49

Commodity Eligible Primary Receipt	Rate		Volume	Reservation
Shipper Name Charge(s) Point(s) / PIN No(s).	Schedule Point(s	Contract s) / PIN No(s).	(Dth/d)	Charge(s)
1/				
Kinder Morgan Tejas Pipeline, LP	NSS	4/01/2003 6/	53,334 2/	\$2.5000
4/, 5/ 6/		through 3/31/2006		3/, 5/
BP Energy Company 4/, 5/	NSS	4/01/2003 6/	38,333 2/	\$2.3438
4/, 3/		through 3/31/2007		3/, 5/
Occidental Energy Marketing, Inc. 4/, 5/	NSS	4/01/2003 6/	26,667 2/	\$2.5000
4/, 3/		through 3/31/2006		3/, 5/
Reliant Energy Aurora, LP	NSS	4/01/2003 6/	13,333 2/	\$2.5000
4/, 3/		through 3/31/2008		3/, 5/
Ameren Energy Generating Company 4/, 5/	NSS	4/01/2003 6/	10,000 2/	\$2.6500
4/, 5/ 6/		through 3/31/2008		3/, 5/
Union Electric Company dba Ameren Ul	E NSS	4/01/2003 6/	5,000 2/	\$2.7500
1/, 5/		through 3/31/2008		3/, 5/

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

 $^{2/\,}$ The Negotiated Rates set forth herein shall apply to storage service provided by Natural to Shipper under Rate Schedule NSS with the Maximum Storage Volume (MSV)

of Dth, set forth below, and the MDQ (Dth/day), set forth above, on the Gulf Coast Leg, equal to total Gulf Coast leg rights of 100% and total Amarillo Leg

rights equal to zero percent: (a) Kinder Morgan Tejas Pipeline: 4,000,050; (b) BP Energy Company: 2,874,975; (c) Occidental Energy Marketing: 2,000,025; (d)

Reliant Energy Aurora: 999,975; (e) Ameren Energy Gathering: 750,000; and (f) Union Electric Company dba Ameren UE: 375,000.

^{3/} Negotiated Monthly Reservation Rate. The Negotiated Monthly Reservation Rate per Dth of MDQ (Peak and Off-Peak periods) shall apply only to service provided to

Shipper by Natural under the NSS Agreement, subject to all other terms and conditions set forth herein.

FERC Docket: RP10-537-000

Third Revised Sheet No. 35C Third Revised Sheet No. 35C Superseding: Second Revised Sheet No. 35C

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primar	Term of v Delivery	Volume	Reservation	
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract / PIN No(s).	(Dth/d)	Charge(s)	
				-		
Wiscons	in Electric Power Company	DSS 1/	4/01/2010	24,133 1/	1/	1/
			through 3/31/2013			
Wiscons	in Electric Power Company	NSS 1/	4/01/2010	19,000 1/	1/	1/
			through 3/31/2013			

^{1/} The information is set out in the executed Storage Rate Schedule Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 35A.01 original Sheet No. 35A.01 : Pending

FOOTNOTES APPLICABLE TO SHEET NO.

35

8/ Applicable Rate Considerations:

Unless Natural otherwise expressly agrees in writing, Shipper shall not be entitled to change any of its Primary Receipt Points or Primary Delivery Points

under the FTS Agreement if such change would result in Shipper paying a reservation rate to Natural which is lower than the applicable Negotiated

Reservation Rate. At any time during the term herein, Shipper shall be entitled to change, at the Negotiated Reservation Rate, the Primary Receipt Point

set forth above to any other receipt point which exists as of the commencement herein and is within Shipper's then-existing primary path, subject to

capacity availability and all applicable provisions of Natural's FERC Gas Tariff regarding receipt point changes, provided that any such Primary Receipt

Point change may not result in Shipper extending its then-existing primary path.

9/ Negotiated Reservation and Commodity Rates. The Negotiated Reservation Rate and Negotiated Commodity Rate set forth below shall apply only to service

provided to Shipper by Natural under the FTS Agreement from the Eligible Receipt Points to the Eligible Delivery Points set forth below, subject to all

other applicable terms and conditions set forth herein.

Negotiated Reservation Rate: Peak \$3.9471/Dth of MDQ and Off-Peak \$4.4481/Dth of MDQ.

Negotiated Commodity Rate: Peak \$0.0193/Dth and Off-Peak \$0.0045/Dth.

10/ Eligible Receipt Points. The Negotiated Rates shall only apply to transportation service provided from the following receipt points:

Primary Receipt Points: Sabine, PIN No. 3592 (20,000 Dth of MDQ).

Secondary Receipt Points: The Negotiated Rates shall apply to service provided from all secondary receipt points permitted under the FTS Agreement.

11/ Eligible Delivery Points. The Negotiated Rates shall only apply to transportation service provided to the following delivery points:

Primary Delivery Points: Illinois Power/Mansfield, PIN No. 900147 (3,000 Dth of MDQ); Illinois Power/Deland Piatt, PIN No. 900148 (17,000 Dth of MDQ).

Secondary Delivery Points: The Negotiated Rates shall apply to NICOR (PIN No. 9258), NIPSCO (PIN No. 909260), PGL&C (PIN No. 909285), MidAmerican (PIN No.

10568), Illinois Power (PIN No. 909999) and all storage and pooling points within the zones traversed by the primary path as permitted by the FTS

Agreement.

Incremental Rate Secondary Delivery Points: Incremental Rate Secondary Delivery Points shall include Midwestern Kankakee (PIN Number 25400) with an

additional incremental daily volumetric reservation rate of 0.03 per Dth applied to all quantities transported on any day to Midwestern Kankakee (PIN No.

25400) for the period commencing February 21, 2003 and ending November 30, 2006.

12/ Eligible Primary Pathing. Shipper's primary path under the FTS Agreement will be from Natural's Louisiana Zone then north on Natural's Gulf Coast mainline to the interconnect with Illinois Power in Segment 28.

FERC Docket: RP99-176-200

Second Revised Sheet No. 35B.01 Second Revised Sheet No. 35B.01 Superseding: First Revised Sheet No. 35B.01

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primarv	Term of Delivery	Volume	Reservation
-	Shipper Name Point(s) / PIN No(s).	Schedule	Contract / PIN No(s).	(Dth/d)	Charge(s)
Charge(s)	POINT(S) / PIN NO(S).				
				-	
BP Canada 1/	Energy Marketing Corp.	FTS 1/	6/01/2009	150,000	1/
,	·	_,	through 4/30/2013		

^{1/} This information is set out in the amended Transportation Rate Schedule FTS Agreement with corresponding Negotiated Rate Exhibit, which is currently on file with the FERC.

FERC Docket: RP10-999-000

Fourth Revised Sheet No. 35C.01 Fourth Revised Sheet No. 35C.01

Superseding: Third Revised Sheet No. 35C.01

STATEMENT OF NEGOTIATED RATE TRANSACTIONS
PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primar	Term of Ty Delivery	Volume	Reservation	
Charge(s)	Shipper Name Point(s) / PIN No(s).		Contract / PIN No(s).	(Dth/d)	Charge(s)	
				-		
Tyson F	oods, Inc. 1/ 2/	FTS 2/	8/01/2010	2/	2/	2/
			through 7/31/2013			

^{1/} The Transportation Rate Schedule FTS Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP99-176-194

First Revised Sheet No. 35B.02 First Revised Sheet No. 35B.02

Superseding: Original Sheet No. 35B.02

STATEMENT OF NEGOTIATED RATE

TRANSACTIONS

PURSUANT TO SECTION 49

.____

		Rate	Term of	Volume	Reservation
Commodity	Eligible Primary Receipt	Eligible Pr	imary Delivery		
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).		
					0/ 0/ 5/
	al Energy Marketing, Inc. 1/		4/01/2009	80,000	2/, 3/, 5/
4/ , 5/	6/, 8/	7/,	- /		
			through		
			3/31/2012		

- 1/ The Negotiated Rate Agreement, as amended, with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.
- 2/ Monthly Reservation Charge Determination. Shipper's Monthly Reservation Charge shall consist of the relevant Negotiated Reservation Rate for each service

described in Footnote 3 below, multiplied by the relevant Service MDQ, multiplied by 30.42. This Monthly Reservation Charge calculation shall be adjusted as

necessary for any month for which Natural and Shipper have agreed to use the Forward Spread Rate Option described in footnote 3(6) below for some or all of the

relevant service MDQ. For billing purposes, Natural in any month may charge Shipper a single Blended Negotiated Reservation Rate (the "Blended Negotiated

Reservation Rate") on Shipper's entire FTS Agreement MDQ, to the extent that use of such Blended Negotiated Reservation Rate does not result in any change in

Shipper's Monthly Reservation Charge or in any other rates or charges hereunder. In no event shall Shipper's Monthly Reservation Charge at any time exceed the

maximum monthly reservation charge in Natural's FERC Gas Tariff applicable to service provided under the FTS Agreement.

- 3/ Negotiated Rates and Eligible Transportation Quantities.
 - (1) Gage to Stingray Service.

(a) Gage to Stingray Negotiated Reservation Rate. For firm transportation service provided to Shipper from the Gage Receipt Point to the Stingray Delivery

Point (the "Gage to Stingray Service"), the Gage to Stingray Negotiated Reservation Rate shall be the greater of: (a) \$0.1450/Dth (the "Gage to Stingray Floor

Rate"), or (b) the Gage to Stingray Index Rate, which shall equal: (i) the NGI Bidweek Survey Price for Chicago Citygates ("Chicago Index"); minus (ii) Inside

FERC's Gas Market Monthly Report for the NGPL Midcontinent Zone Index plus \$0.04/Dth ("NGPL Amarillo Index"); minus (iii) Minimum Commodity Rate (MCR)

applicable to the Amarillo Mainline to Market primary path; minus (iv) Fuel/GLU; minus (v) the ACA surcharge. "Fuel/GLU" in this calculation is defined as (a)

the NGPL Amarillo Index multiplied by (b) the reciprocal of 1 minus the Amarillo Mainline to Market fuel/GLU percentage, minus (c) the NGPL Amarillo Index.

- (b) Negotiated Reservation Rate Credit. In the event that the Gage to Stingray Negotiated Reservation Rate in any month exceeds 0.1450/Dth, then Shipper
- shall receive for that month a partial reservation rate credit from Natural for Shipper's five thousand (5,000) Dth/d of firm capacity from the Gage to Stingray

Delivery Point. The amount of such credit to Shipper shall be determined as follows:

- (i) To the extent that the relevant Gage to Stingray Negotiated Reservation Rate is greater than 0.1450/Dth and less than or equal to 0.1650/Dth,
- Shipper's credit shall be one hundred percent (100%) of the amount by which such rate exceeds \$0.1450/Dth and (ii) To the extent that the relevant Gage to
- Stingray Negotiated Reservation Rate exceeds \$0.1650/Dth, Shipper's credit shall be the amount of the credit determined pursuant to Footnote 3(1)(b)(i) above,

as well as thirty percent (30%) of the amount by which such rate exceeds \$0.165/Dth.

(c) Service MDQ. Shipper's MDQ for the Gage to Stingray Service shall be $5,000\ \mathrm{Dth/d.}$

FERC Docket: RP10-999-000

Third Revised Sheet No. 35C.02 Third Revised Sheet No. 35C.02 Superseding: Third Revised Sheet No. 35C.01

Sheet No. 35C.02 has been reserved for

future use.

FERC Docket: RP99-176-194

First Revised Sheet No. 35B.03 First Revised Sheet No. 35B.03

Superseding: Original Sheet No. 35B.03

FOOTNOTES APPLICABLE TO SHEET NO.

35B.02

- 3/ Negotiated Rates and Eligible Transportation Quantities. (Con't)
 - (2) Crosshaul Service.

(a)Crosshaul Negotiated Reservation Rate. For firm transportation service provided to Shipper from the Gage or El Paso Receipt Points to the Sabine

Delivery Point, (the "Crosshaul Service"), the Crosshaul Negotiated Reservation Rate shall be the greater of: (a) \$0.1050/Dth (the "Crosshaul Floor Rate"), or

(b) the Crosshaul Index Rate, which shall equal: (i) the NYMEX Settlement Price on expiration day for the flow month contract; minus (ii) \$0.05; minus (iii)

Inside FERC's Gas Market Monthly Report for the NGPL Midcontinent Zone Index (NGPL Midcontinent Index); minus (iv) the weighted average of the Minimum Commodity

Rate (MCR) applicable to the Amarillo Mainline to Louisiana and the Permian to Louisiana primary paths; minus (v) Fuel/GLU; minus (vi) the ACA Surcharge.

"Fuel/GLU" in this calculation is defined as (a) the NGPL Midcontinent Index, multiplied by (b) the reciprocal of 1 minus the Midcontinent to Louisiana fuel/GLU percentage, minus (c) the NGPL Midcontinent Index.

(b) Negotiated Reservation Rate Credit. In the event that the Crosshaul Negotiated Reservation Rate in any month exceeds 0.1050/Dth, then Shipper shall

receive for that month a partial reservation rate credit from Natural for Shipper's twenty-five thousand (25,000) Dth/d of firm capacity from the Gage and El

Paso Receipt Points to the Sabine Delivery Point. The amount of such credit to Shipper shall be determined as follows:

- (i) To the extent that the relevant Crosshaul Negotiated Reservation Rate is greater than \$0.1050 and less than or equal to \$0.1250, Shipper's credit shall be one hundred percent (100%) of the amount by which such rate exceeds \$0.1050/Dth.
- (ii) To the extent that the relevant Crosshaul Negotiated Reservation Rate exceeds \$0.1250/Dth, Shipper's credit shall be the amount of the credit

determined pursuant to Footnote 3(2)(b)(i) above, as well as thirty percent (30%) of the amount by which such rate exceeds \$0.1250/Dth.

- (c) Service MDQ. Shipper's MDQ for the Crosshaul Service shall be 25,000 Dth/d.
- (3) Texok Service.
- (a)Texok Negotiated Reservation Rate. For firm transportation service provided to Shipper from the Texok Receipt Point to the Sabine Delivery Point (the

"Texok Service"), the Texok Negotiated Reservation Rate shall be the greater of: (a) \$0.04/Dth (the "Texok Floor Rate"), or (b) the Texok Index Rate, which

shall equal: (i) the NYMEX Settlement Price on expiration day for the flow month contract; minus (ii) 0.05; minus (iii) Inside FERC's Gas Market Monthly Report

for the NGPL Texok Zone Index (NGPL Texok Index); minus (iv) Minimum Commodity Rate (MCR) applicable to the Texok to Louisiana primary path; minus (v) Fuel/GLU; minus (vi) the ACA Surcharge. "Fuel/GLU" in this calculation is defined as (a) the NGPL Texok

minus (vi) the ACA Surcharge. "Fuel/GLU" in this calculation is defined as (a) the NGPL Texok Index multiplied by (b) the reciprocal of 1 minus the Texok to

Louisiana fuel/GLU percentage, minus (c) the NGPL Texok Index.

FERC Docket: RP10-583-000

Third Revised Sheet No. 35C.03 Third Revised Sheet No. 35C.03

Superseding: Second Revised Sheet No. 35C.03

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Charge(s) Point(s) / PIN No(s).	Point(s) / PIN No(s).	Volume (Dth/d)	
Wisconsin Electric Power Company 1/2/2/	2/ (#110824) through	2/	2/
Wisconsin Electric Power Company 1/2/2/	3/31/2011 FTS 4/01/2008 2/ (#113833) through 3/31/2009	10,000	2/
Wisconsin Electric Power Company 1/2/2/		12,000	2/
Wisconsin Electric Power Company 1/2/2/		10,250	2/
Wisconsin Electric Power Company 1/2/2/2/		2/	2/
Wisconsin Electric Power Company 2/ 2/	FTS-NB 4/01/2010 2/ (#129979) through	2/	2/
Wisconsin Electric Power Company 1/2/2/2/	2/ (#129113) through	5,000 2/	\$3.07
Wisconsin Electric Power Company 1/2/2/2/	3/31/2011 NSS 4/01/2008 2/ (#129114) through 3/31/2011	5,000 2/	\$3.07

^{1/} The Negotiated Rate Agreement with Shipper does not deviate in any material respect from the applicable Form of Service Agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and the Storage Rate Schedule NSS Agreement with corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP99-176-194

First Revised Sheet No. 35B.04 First Revised Sheet No. 35B.04

Superseding: Original Sheet No. 35B.04

FOOTNOTES APPLICABLE TO SHEET NO.

35B.02

- 3/ Negotiated Rates and Eligible Transportation Quantities. (Con't)
 - (3) Texok Service (Con't)

(b) Negotiated Reservation Rate Credit. In the event that the Texok Negotiated Reservation Rate in any month exceeds \$0.04/Dth, then Shipper shall receive for that month a partial reservation rate credit from Natural for Shipper's twenty thousand

(20,000) Dth/d of firm capacity from the Texok Receipt Point to the

Sabine Delivery Point. The amount of such credit to Shipper shall be determined as follows:

- To the extent that the relevant Texok Negotiated Reservation Rate is greater than \$0.04 and less than or equal to \$0.06, Shipper's credit shall be one hundred percent (100%) of the amount by which such rate exceeds \$0.04/Dth.
- (ii) To the extent that the relevant Texok Negotiated Reservation Rate exceeds \$0.06/Dth, Shipper's credit shall be the amount of the credit determined pursuant to Footnote 3(3)(b)(i) above, as well as thirty percent (30%) of the amount by which such
 - (c) Service MDQ. Shipper's MDQ for the Texok Service shall be 20,000 Dth/d.
 - (4) South Texas Service.

rate exceeds \$0.06/Dth.

(a) South Texas Negotiated Reservation Rate. For firm transportation service provided to Shipper from either of the South Texas Receipt Points to either the

Florida Gas Delivery Point or the Sabine Delivery Point (the "South Texas Service"), the South Texas Negotiated Reservation Rate shall be the greater of: (a)

\$0.0617/Dth (the "South Texas Floor Rate"), or (b) the South Texas Index Rate, which shall equal: (i) the NYMEX Settlement Price on expiration day for the flow month contract; minus (ii) \$0.05; minus (iii) Inside FERC's Gas Market Monthly Report for the NGPL

South Texas Zone Index (NGPL South Texas Index); minus (iv)

Minimum Commodity Rate (MCR) applicable to the South Texas to Louisiana primary path; minus (v) Fuel/GLU; minus (vi) the ACA Surcharge. "Fuel/GLU" in this

calculation is defined as (a) the NGPL South Texas Index multiplied by (b) the reciprocal of 1 minus the South Texas to Louisiana fuel/GLU percentage, minus (c) the NGPL South Texas Index.

(b) Negotiated Reservation Rate Credit. In the event that the South Texas Negotiated Reservation Rate in any month exceeds \$0.0617/Dth, then Shipper shall

receive for that month a partial reservation rate credit from Natural for Shipper's thirty thousand (30,000) Dth/d of firm capacity from the South Texas Receipt

Points to the Florida Gas and Sabine Delivery Point. The amount of such credit to Shipper shall be determined as follows:

- To the extent that the relevant South Texas Negotiated Reservation Rate is greater than \$0.0617 and less than or equal to \$0.0817, Shipper's credit
 - shall be one hundred percent (100%) of the amount by which such rate exceeds \$0.0617/Dth.
- (ii) To the extent that the relevant South Texas Negotiated Reservation Rate exceeds \$0.0817/Dth, Shipper's credit shall be the amount of the credit

determined pursuant to Footnote 3(4)(b)(i) above, as well as thirty percent (30%) of the amount by which such rate exceeds \$0.0817/Dth.

(c) Service MDQ. Shipper's MDQ for the South Texas Service shall be 30,000 Dth/d.

FERC Docket: RP10-70-000

Third Revised Sheet No. 35C.04 Third Revised Sheet No. 35C.04

Superseding: Second Revised Sheet No. 35C.04

TRANSACTIONS

STATEMENT OF NEGOTIATED RATE
PURSUANT TO SECTION 49

._____

Commodity	Primary Receipt	Rate Prim	Term of ary Delivery	Volume	Reservation
±	Shipper Name	Schedule	Contract	(Dth/d)	Charges(s)
Charge(s)	Point(s) / PIN No(s).	Point(s) / PIN No(s).		
OGE Ener	gy Resources, Inc. 1/	FTS	11/01/2009	20,000	2/
2/	2/	2/			
			through		
			10/31/2012		

^{1/} The Transportation Rate Schedule FTS Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP99-176-194

First Revised Sheet No. 35B.05 First Revised Sheet No. 35B.05

Superseding: Original Sheet No. 35B.05

FOOTNOTES APPLICABLE TO SHEET NO. 35B.02

3/ Negotiated Rates and Eligible Transportation Quantities. (Con't)

(5) Alternate Index. If for any month one or more of the published gas price indices ("Original Index") referenced above is not published for any reason, then

subject to the provisions of footnote 3(6) below regarding the possible application of the Forward Spread Rate Option, the corresponding Alternate Index

designated below shall be used for purposes of calculating the applicable Negotiated Reservation Rate for such month:

Original Index Alternate

NGPL Midcontinent Index Arithmetic average of the following component indices, published in Inside FERC's Gas Market Monthly Report for: ANR - Oklahoma less \$0.02; and Panhandle Eastern

- Oklahoma less \$0.01.

NGPL South Texas Index Arithmetic average of the following component indices, published in Inside FERC's Gas Market Monthly Report for: Tennessee Zone O; Texas Eastern - South Texas Zone and Transco Zone 1.

For any month in which one or more of the component indices used to calculate an Alternate Index is not published for any reason, then the Alternate

Index shall be calculated using the arithmetic average of the remaining relevant component indices that are published for such month. For any month in

which none of the relevant component indices is published for the Original Index or the applicable Alternate Index, then Natural and Shipper shall agree

to use an appropriate alternate index for such month, which meets all applicable FERC requirements for permissible price indices.

FERC Docket: RP10-397-000

Second Revised Sheet No. 35C.05 Second Revised Sheet No. 35C.05

Superseding: Sheet No. 35C.05

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primarv	Term of Delivery	Volume	Reservation
<u> </u>	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).		-
Fasken (Dil and Ranch, Ltd. 1/	FTS 2/	3/01/2010 through 2/28/2013	1,000	2/

^{1/} The Transportation Rate Schedule FTS Agreement with Shipper does not deviate in any material respect from the applicable Form of Service Agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 35B.06 Original Sheet No. 35B.06 : Pending

FOOTNOTES APPLICABLE TO SHEET NO. 35B.02

- 3/ Negotiated Rates and Eligible Transportation Quantities. (Con't)
 - (6) Forward Spread Rate Option.

(a) Agreement of the Parties Required. With regard to any one or more of the respective Index Rate components in the Negotiated Reservation Rate formulas

set forth in Footnote 3 (1)-(4) above, Natural and Shipper may from time to time agree to utilize an alternate index rate component which is based upon

a locked-in forward basis spread (a "Locked-In Rate Component") in lieu of such otherwise applicable Index Rate component, in order to determine the

Negotiated Reservation Rate for the relevant Service. In order to utilize a Locked-In Rate Component for a particular Service, Natural and Shipper

must, at their respective sole discretion and without obligation, agree in writing (including but not limited to electronic email exchanges) to: (i)

the relevant delivery zone basis to be utilized ("Delivery Basis"); (ii) the relevant receipt zone basis to be utilized ("Receipt Basis"); (iii) the

amount of capacity from the relevant Service to which such rate component will apply ("Applicable Capacity"); and (iv) the term for which such rate $\dot{}$

component will be applicable for the relevant Service, which term may be one or more complete months within the then-remaining term of this Agreement

("Applicable Term"). In the event that Natural or Shipper (the "Triggering Party") desires to utilize the Forward Spread Rate Option, then such

Triggering Party shall notify the other party in writing of the desired Applicable Term and Applicable Capacity, as well as the best available market

quotes on the corresponding Delivery Basis and Receipt Basis. Natural and Shipper shall then have twenty-four (24) hours to agree in writing to the

Receipt Basis, Delivery Basis, Applicable Term and Applicable Capacity. If such agreement is timely reached, then the Forward Spread rate Option shall

apply in accordance with such agreement, regardless of whether the agreed spread is subsequently executed. If such agreement is not timely reached,

then the Forward Spread Rate Option shall not apply for the desired Applicable Term and Applicable Capacity.

(b)Locked-In Rate Component. The applicable Locked-In Rate Component will be determined by the following formula: (1) the Delivery Basis; minus (ii) the

Receipt Basis; minus (iii) Fuel/GLU; minus (iv) the applicable minimum gas commodity rate in Natural's tariff; minus (v) the Annual Charge Adjustment

surcharge (ACA). "Fuel/GLU" will be determined by the following formula: [((i))] the relevant NYMEX Henry Hub natural gas futures contract price for

the Applicable Term, as if hedged; plus (ii) the Receipt Basis]; times (iii) the result of [(1/(1-applicable Receipt Zone Fuel/GLU percentage stated in Natural's tariff))-1].

(c) Effect on Negotiated Reservation Rates. Any Locked-In Rate Component which may be agreed upon by Natural and Shipper shall apply in lieu of the

otherwise applicable Index Rate component in the formula used for determining the relevant Negotiated Reservation Rate set forth in Footnote 3(1)-(4)

above. All other provisions of such formula and of Footnote 3 (1)-(4) above shall continue to apply during the Applicable Term, including without

limitation the applicable stated minimum reservation rate and the applicable Negotiated Reservation Rate Credit. The Locked-In Rate Component shall

apply only to the Applicable Capacity and for the Applicable Term related to the relevant Service. If the Applicable Capacity is less than the

relevant Service MDQ set forth in Footnote 3 (1)-(4) above, then during the Applicable Term, the relevant Negotiated Reservation Rate formula shall

apply as written in Footnote 3 (1)-(4) above to the amount of Service MDQ capacity to which the Locked-In Rate Component does not apply. Natural and

Shipper may agree, at their respective sole discretion and without obligation, to utilize a Locked-In Rate Component for any one or more of the

Services described in Footnote 3 (1)-(4) above during the same time period or for overlapping time periods.

FERC Docket: RP10-181-000

Original Sheet No. 35C.06 Original Sheet No. 35C.06

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s).	Schedule	Term of ry Delivery Contract / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)	
Black H	ills Utility Holdings, Inc.	FTS 2/	12/01/2009	12,000	2/	2/
	ly Aquila, Inc., d/b/a a Networks) 1/		through 10/31/2013			
Black H	ills Utility Holdings, Inc.	FTS-G 2/	12/01/2009	1,402	2/	2/
	ly Aquila, Inc., d/b/a a Networks) 1/	27	through 10/31/2013			

⁻⁻⁻⁻⁻

^{1/} The Transportation Rate Schedules FTS and FTS-G Agreements with Shipper do not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} This information is set out in the executed Transportation Rate Schedules FTS and FTS-G Agreements with corresponding Negotiated Rate exhibits, which are currently on file with the FERC.

FERC Docket: RP99-176-194

First Revised Sheet No. 35B.07 First Revised Sheet No. 35B.07

Superseding: Original Sheet No. 35B.07

FOOTNOTES APPLICABLE TO SHEET NO.

35B.02

4/ Negotiated Commodity Rate. For all firm service provided under the FTS Agreement from the Eligible Receipt Points to the Eligible Delivery Points which does not

exceed the applicable discounted quantity limitation set forth herein, Shipper shall pay Natural a Negotiated Commodity Rate equal to the weighted average (based

on Primary Path MDQ) of the minimum commodity rates set forth in Natural's FERC Gas Tariff. For all other firm service provided under the FTS Agreement, Shipper

shall pay Natural the applicable maximum commodity rate set forth in Natural's FERC Gas Tariff, as such maximum rates may be revised from time to time.

5/ Eligible Firm Transportation Quantities. The Negotiated Reservation Rates applicable to the various services described in Footnotes 3(1) - 3(4) above shall apply

to all firm transportation service provided to Shipper from the respective Eligible Receipt Points to the respective Eligible Delivery Points, regardless of

Shipper's release of Shipper's firm capacity, to the extent that all such firm service provided on any day to Shipper and to any related capacity release

replacement Shippers does not exceed the relevant Service MDQ set forth in Footnotes 3, 4 and 5 and meets all requirements applicable to the Negotiated

Reservation Rates (for each service, the "Eligible Transportation Quantity"); provided, however, that the Eligible Transportation Quantity shall be 120,000

Dth/day for firm service provided on any day to Shipper and any related capacity release replacement Shippers from any Eligible Receipt Point located in Segments

23, 24, 25 or 26 to any Eligible Delivery Point located in Segments 23, 24, 25 or 26. Any quantities transported on a firm basis in excess of the applicable

Eligible Transportation Quantity shall be billed at the greater of the Blended Negotiated Reservation Rate or the applicable maximum reservation rate, as may be revised from time to time.

Eligible Daily Contract MDQ Overrun Quantity. Subject to the provisions of Footnote 5 below regarding the Northern Delivery Points AOR Rates, a Contract MDQ

Authorized Overrun Rate of \$0.1450/Dth shall apply to all authorized overrun deliveries under the FTS Agreement on any day on which total deliveries under the FTS

Agreement exceed 80,000/Dth but are less than or equal to 120,000/Dth (the "Discounted Contract MDQ Overrun Quantity"). Any overrun quantities delivered on any

day which are in excess of the Discounted Contract MDQ Overrun Quantity shall be billed at the greater of the Blended Negotiated Reservation Rate or the

applicable maximum authorized overrun rate set forth in Natural's FERC Gas Tariff, as my be revised from time to time. Determination of authorized overrun

quantities on any day shall take into consideration any increase in the Eligible Firm Transportation Quantity that may apply.

Northern Delivery Points AOR Rates. Notwithstanding the provisions of Paragraphs 1 and 2 of this Footnote 5 above, all authorized overrun quantities delivered on

any day from Eligible Receipt Points to delivery points located in Natural's Market Delivery Zone (the "Northern Delivery Points") which in the aggregate exceed

17,500 Dth but do not exceed 37,600 Dth (the "Discounted NDP AOR Quantity") shall be billed at a discounted authorized overrun rate of \$0.1450/Dth. If on any day

aggregate eligible authorized overrun deliveries to the Northern Delivery Points exceed 37,600 Dth (the "Maximum Discounted NDP AOR Quantity"), then all such

excess deliveries shall be billed at the greater of the Blended Negotiated Reservation Rate or the applicable maximum authorized overrun rate set forth in

Natural's FERC Gas Tariff, as may be revised from time to time.

Authorized Receipt Zone Overrun. Daily authorized overrun receipts in excess of Shipper's applicable zone quantity limitations ("Authorized Receipt Zone

Overrun") but within Shipper's contract MDQ will be billed an AOR rate of \$0.00/Dth. Authorized Receipt Zone Overrun in excess of Shipper's contract MDQ will be

billed at the greater of the Blended Negotiated Reservation Rate or the applicable maximum Tariff rate.

Fuel Gas and Gas Lost and Unaccounted For. In addition to the Negotiated Rates, Shipper shall be charged for Fuel Gas and Gas Lost and Unaccounted For ("Fuel and

 $\operatorname{GLU"}$) in accordance with the applicable provisions of Natural's FERC Gas Tariff, as may be revised from time to time.

Discounted Third Party Surcharges. From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC

Gas Tariff which Natural is: (i) required to collect from Shipper and remit to the FERC or to

another third party; and (ii) permitted to discount the amount of

such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). Natural shall discount any

particular Discountable Third Party Surcharge to the maximum extent permitted under the provisions of Natural's FERC Gas Tariff. Shipper shall only be responsible for payment to Natural of any portion of such Discountable Third Party Surcharge which

Natural is not permitted to discount and for which Natural bills Shipper.

FERC Docket: RP10-182-000

Original Sheet No. 35C.07 Original Sheet No. 35C.07

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Priman	Term of ry Delivery	Volume	Reservation	
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)	
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).			
MidAmer	cican Energy Company	FTS 1/	12/01/2009 through 11/30/2014	1/	1/	1/

^{1/} This information is set out in the executed Firm Transportation Negotiated Rate Agreement which is currently on file with the FERC.

FERC Docket: RP99-176-194

First Revised Sheet No. 35B.08 First Revised Sheet No. 35B.08

Superseding: Original Sheet No. 35B.08

FOOTNOTES APPLICABLE TO SHEET NO.

35B.02

5/ Con't

Applicable Maximum Rates, Charges and Surcharges. In addition to the Negotiated Reservation Rates, and unless otherwise expressly provided in this Agreement

or agreed to in writing by Natural, Shipper shall also pay Natural all other applicable maximum rates, charges, surcharges and penalties of any nature set
forth in Natural's FERC Gas Tariff, as may be revised from time to time, including without

limitation all applicable maximum: (i) Fuel and Gas Lost and

Unaccounted For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation charges and surcharges; (iv) ACA surcharges; and (vi) gathering charges, offshore charges and lateral line

charges. Notwithstanding the foregoing sentence, to the extent

that any such additional maximum rate, charge, surcharge or penalty is derived from the applicable base reservation rate set forth in Natural's FERC Gas

Tariff, then Natural shall derive such additional rate, charge, surcharge or penalty utilizing the greater of: (i) the applicable maximum base reservation

rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time; and (ii) the Negotiated Monthly Base Reservation Rate.

Reservation Charge True Up. If the Blended Negotiated Reservation Rate as defined in footnote 2 above, derived using the Index Rates calculated in footnotes

3(1), 3(2), 3(3) and 3(4) times the relevant path MDQs, results in a rate that is less than the weighted average of the Floor Rates defined in footnotes 3(1),

3(2), 3(3) and 3(4) (the "Blended Contract Floor Rate"), Natural will maintain an adjustment account in the amount of the difference between that Blended

Negotiated Reservation Rate based on the Index Rates and the Blended Contract Floor Rate that was billed that month. Such amount will be applied against the

Blended Negotiated Reservation Rate billed in future months during the same Contract Year but only up to the total amount of the adjustment account, and only

to the extent that the Blended Negotiated Reservation Rate exceeds the Maximum Tariff Rate in such future month. In the event that the Index Rate for any

path is less than \$0.00, the Index Rate for that path used to calculate that Blended Negotiated Rate will be \$0.00.

If the Blended Negotiated Reservation Rate, as defined in footnote 2 above, results in a rate that is greater than the applicable Maximum Tariff Rate, Natural

will maintain an adjustment account in the amount of the difference between the Maximum Tariff Rate and the Blended Negotiated Reservation Rate that was

charged for that month. Such amount will be applied against the Blended Negotiated Reservation Rate billed in future months during the same Contract Year but

only up to the total amount of the adjustment account, and only to the extent that the Blended Negotiated Reservation Rate for that future month is less than

the Blended Contract Floor Rate.

Capacity Release. In the event that Shipper releases any capacity under the FTS Agreement, the Negotiated Reservation Rates shall remain in effect for all

service provided to Shipper under the FTS Agreement, to the extent that service under the FTS Agreement and any related capacity release replacement

agreements complies with all of the requirements applicable to the Negotiated Rates which are set forth herein. Shipper shall be billed the greater of the

Blended Negotiated Reservation Rate or the applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas Tariff for all such service provided which does not comply with all requirements applicable to the Negotiated Reservation

Rates. All capacity releases by Shipper under the FTS Agreement

shall be subject to all applicable capacity release requirements set forth in Natural's FERC Gas Tariff, including any and all requirements regarding the

rates which Shipper may charge to Replacement Shippers.

FERC Docket: RP10-540-000

First Revised Sheet No. 35C.08 First Revised Sheet No. 35C.08

Superseding: Original Sheet No. 35C.08

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity Primary Receipt Shipper Name Charge(s) Point(s) / PIN No(s).	Schedule	Term of ry Delivery Contract / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)	
EOG Resources, Inc. 1/	FTS 2/	1/01/2010 through 12/31/2013	40,000	2/	2/
EOG Resources, Inc. 1/ 2/	FTS 2/	4/01/2010 through 3/31/2014	25,000	2/	2/

^{1/} The Transportation Rate Schedule FTS Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP99-176-194

First Revised Sheet No. 35B.09 First Revised Sheet No. 35B.09

Superseding: Original Sheet No. 35B.09

FOOTNOTES APPLICABLE TO SHEET NO.

35B.02

6/ Eligible Receipt Points. The Negotiated Rates shall only apply to transportation service provided from the following receipt points, and in accordance with

all other applicable provisions herein:

Primary Receipt Points: TPC/NGPL Gage ("Gage Receipt Point"), PIN 902900 (17,500 Dth MDQ);
Marathon Indian Basin Plt /NGPL Eddy ("El Paso Receipt Point").

Marathon Indian Basin Plt./NGPL Eddy ("El Paso Receipt Point"),
PIN 1753 (12,500 Dth MDQ); ETP/NGPL Thompsonville Jim Hogg (a "South Texas Receipt Point"), PIN
24001 (10,000 Dth MDQ); KM Tejas/NGPL Nueces (a "South Texas
Receipt Point"), PIN 3886 (20,000 Dth MDQ); and ETP/NGPL Inter #2 Tap Panola ("Texok Receipt

Point"), PIN 3352 (20,000 Dth MDQ).

Secondary Receipt Points. The Negotiated Rates shall apply to transportation service provided from all secondary receipt points permitted under the FTS Agreement.

7/ Eligible Delivery Points. The Negotiated Rates shall only apply to transportation service provided to the following delivery points, and in accordance with all other applicable provisions herein:

Primary Delivery Points: Sabine PL/NGPL Henry Plt Vermilion (Sabine Delivery Point), PIN 3592 (47,000 Dth MDQ 4/1 - 10/31; 56,500 Dth MDQ 11/1 - 3/31);
Florida Gas/NGPL Vermilion, PIN 7936 (28,000 Dth MDQ 4/1 - 10/31; 18,500 Dth MDQ 11/1 - 3/31) and Stingray/NGPL Cameron (Stingray Delivery Point), PIN 5433 (5,000 Dth MDQ).

Secondary Delivery Points. The Negotiated Rates shall apply to transportation service provided to the Duke-Lee/NGPL PIN 37329 and to all secondary delivery points permitted under the FTS Agreement other than delivery points to direct connect industrial end-users and power plants.

- 8/ Eligible Primary Pathing. In order for the Negotiated Rates to apply to service provided to Shipper on a primary firm basis, the following Primary Pathing must be utilized:
- (1) 5,000 Dth/d with a Primary Receipt Point of TPC/NGPL Gage (PIN 902900) will be pathed through Compressor Station 110 with a Primary Delivery Point of Stingray/NGPL Cameron (PIN 5433);
- (2) 12,500 Dth/d with a Primary Receipt Point of TPC/NGPL Gage (PIN 902900) will be pathed through Compressor Station 801 with a Primary Delivery Point of Sabine PL/NGPL Henry Plt Vermilion (PIN 3592);
- (3) 12,500 Dth/d with a Primary Receipt Point of Marathon Indian Basin Plt./NGPL Eddy (PIN 1753) will be pathed through Compressor Station 801 with a Primary Delivery Point of Sabine PL/NGPL Henry Plt Vermilion (PIN 3592);
- (4) 20,000 Dth/d with a Primary Receipt Point of ETP/NGPL Inter #2 Tap Panola (PIN 3352) and a Primary Delivery Point of Sabine PL/NGPL Henry Plt Vermilion (PIN 3592);
- (5) 10,000 Dth/d with a Primary Receipt Point of ETP/NGPL Thompsonville Jim Hogg (PIN 24001) and a Primary Delivery Point of Florida Gas/NGPL Vermilion (PIN 7936) and
- (6) 20,000 Dth/d with a Primary Receipt Point of KM Tejas/NGPL Nueces (PIN 3886) and a Primary Delivery Point of Sabine PL/NGPL Henry Plt Vermilion (PIN 3592) for: 2,000 Dth/d 4/1 10/31; 11,500 Dth/d 11/1 3/31; and a Primary Delivery Point of Florida Gas/NGPL Vermilion (PIN 7936) for: 18,000 Dth/d 4/1 10/31; 8,500 Dth/d 11/1 3/31.

FERC Docket: RP10-313-000

Original Sheet No. 35C.09 Original Sheet No. 35C.09

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity	Primary	Receipt	Rate Primar	Term of y Delivery	Volume	Reservation	
Charge(s)	Shipper Name Point(s) /	PIN No(s).	Schedule Point(s)	Contract / PIN No(s).	(Dth/d)	Charge(s)	
Fasken	Oil and Ranch,	Ltd. 1/	FTS 2/	2/01/2010	2,500	2/	2/
				through 1/31/2013			

^{1/} The Transportation Rate Schedule FTS Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-541-000

Original Sheet No. 35C.10 Original Sheet No. 35C.10

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt	Prima:	ry Delivery		
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s) / PIN No(s).		
				•	
G1-11 F	27	Maa	4 /01 /0010	06 660 1/	1 /
Snell Er	nergy North America	NSS	4/01/2010	26,668 1/	1/
1/	1/	1,	/		
(US),	L.P.		through		
			3/31/2013		

⁻⁻⁻⁻⁻

^{1/} The information is set out in the executed Storage Rate Schedule NSS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-531-000

Original Sheet No. 35C.11 Original Sheet No. 35C.11

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primar	Term of	Volume	Reservation
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract / PIN No(s).	(Dth/d)	Charge(s)
	to Energy Services Company 1/2/	NSS 2/	4/01/2010 through 3/31/2013	33,333 2/	2/

^{1/} The Agreement with Shipper does not deviate in any material respect from the applicable Form of Service Agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Storage Rate Schedule NSS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-970-000

First Revised Sheet No. 35C.12 First Revised Sheet No. 35C.12 Superseding: Original Sheet No. 35C.12

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity Charge(s)	Primary Receipt Shipper Name Point(s) / PIN No(s).	Schedule	Term of y Delivery Contract / PIN No(s).	Volume (Dth/d)	Reservation Charge(s)
				-	
BP Ener	rgy Company 1/	NSS 1/	4/01/2010 through 3/31/2013	38,333 1/	1/
BP Ener	gy Company 1/	NSS 1/	7/16/2010 through 7/15/2013	7,500 1/	1/

^{1/} The information is set out in the executed Storage Rate Schedule NSS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-528-000

Original Sheet No. 35C.13 Original Sheet No. 35C.13

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Primar	Term of y Delivery	Volume	Reservation
-	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).		
United	Energy Trading, LLC	FTS	4/01/2010	10,000	1/
1/	1/	1/			
			through		
			3/31/2015		

⁻⁻⁻⁻⁻

^{1/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-534-000

Original Sheet No. 35C.14 Original Sheet No. 35C.14

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Prima	Term of ry Delivery	Volume	Reservation
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s) / PIN No(s).		
Tate &	Lyle Ingredients	FTS	4/01/2010	2/	2/
2/	2/	2	/		
Ameri	cas, Inc. 1/		through 3/31/2013		

^{1/} The Transportation Rate Schedule FTS Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-675-000

Original Sheet No. 35C.15 Original Sheet No. 35C.15

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt	Prima	ry Delivery		
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s) / PIN No(s).		<u>-</u>
				-	
Minha C	C+ IIC 1/	NSS	5/01/2010	13,333 2/	2/
2/	as Storage LLC 1/	NSS	7/01/2010	13,333 2/	2/
4/	2/	4.	/		
			through		
			4/30/2013		

^{1/} The Agreement with Shipper does not deviate in any material respect from the applicable Form of Service Agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Storage Rate Schedule NSS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-676-000

Original Sheet No. 35C.16 Original Sheet No. 35C.16

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt	Prima	ry Delivery		
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s) / PIN No(s).		_
				-	
BP Cana	da Energy Marketing	NSS	5/01/2010	66,667 1/	1/
1/	1/	1	/		
Corpo	ration		through		
			4/30/2013		

⁻⁻⁻⁻⁻

^{1/} The information is set out in the executed Storage Rate Schedule NSS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-910-000

Original Sheet No. 35C.17 Original Sheet No. 35C.17

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

Commodity	Primary Receipt	Rate Prima	Term of ry Delivery	Volume	Reservation	
Charge(s)	Shipper Name Point(s) / PIN No(s).	Schedule	Contract) / PIN No(s).	(Dth/d)	Charge(s)	
EOG Reso	ources, Inc. 1/	FTS 2/	7/01/2010	2/	2/	2/
			through 11/30/2013			

^{1/} The Transportation Rate Schedule FTS Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Transportation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-909-000

Original Sheet No. 35C.18 Original Sheet No. 35C.18

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

		Rate	Term of	Volume	Reservation
Commodity	Primary Receipt	Prima	ry Delivery		
	Shipper Name	Schedule	Contract	(Dth/d)	Charge(s)
Charge(s)	Point(s) / PIN No(s).	Point(s) / PIN No(s).		
				=	
Green V	alley Chemical Corporation	FTS	7/01/2010	4,000	1/
1/	1/	1	/	,	
			through		
			6/30/2013		

⁻⁻⁻⁻⁻

^{1/} The information is set out in the executed Transporation Rate Schedule FTS Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-1225-000

Original Sheet No. 35C.19 Original Sheet No. 35C.19

STATEMENT OF NEGOTIATED RATE TRANSACTIONS PURSUANT TO SECTION 49

.____

	Primary Receipt Shipper Name Point(s) / PIN No(s).	Schedule	y Delivery Contract	Volume (Dth/d)	Reservation Charge(s)	
Village 2/	of Bethany, IL 1/	FTS 2/		300	2/	
			through 11/30/2013			
Village	of Bethany, IL 1/	FTS-G 2/	9/01/2010	1,200	2/	2/
			through 11/30/2013			
Town of	Corning, IA 1/	FTS 2/	9/01/2010	250	2/	2/
			through 11/30/2013			
Town of	Corning, IA 1/ 2/	FTS-G 2/	9/01/2010	1,800	2/	2/
			through 11/30/2013			
City of	Perryville, MO 1/	FTS 2/	9/01/2010	1,500	2/	2/
			through 11/30/2013			
City of	Perryville, MO 1/	FTS-G 2/	9/01/2010	4,975	2/	2/
			through 11/30/2013			

⁻⁻⁻⁻⁻

^{1/} The Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP10-1225-000

Original Sheet No. 35C.20 Original Sheet No. 35C.20

STATEMENT OF NEGOTIATED RATE TRANSACTIONS
PURSUANT TO SECTION 49

	Primary Receipt Shipper Name	Primar Schedule		Volume (Dth/d)		
Charge(s)	Point(s) / PIN No(s).	Point(s)	/ PIN No(s).			
City of	Pinckneyville, IL 1/	FTS 2/	9/01/2010	250	2/	
			through 11/30/2013			
City of	Pinckneyville, IL 1/	FTS-G 2/	9/01/2010	3,000	2/	2/
			through 11/30/2013			
City of	Sullivan, IL 1/	FTS 2/	9/01/2010	700	2/	2/
			through 11/30/2013			
City of	Sullivan, IL 1/	FTS-G 2/	9/01/2010	5,000	2/	2/
			through 11/30/2013			
City of	Salem, IL 1/	FTS 2/	9/01/2010	2,000	2/	2/
			through 3/31/2014			
City of	Salem, IL 1/	FTS-G 2/	9/01/2010	3,500	2/	2/
			through 11/30/2012			
City of	Salem, IL 1/	DSS 2/	9/01/2010	1,500	2/	2/
			through 11/30/2012			

^{1/} The Agreement with Shipper does not deviate in any material respect from the applicable form of service agreement in Natural's FERC Gas Tariff.

^{2/} The information is set out in the executed Agreement and corresponding Negotiated Rate Exhibit, which are currently on file with the FERC.

FERC Docket: RP08-319-000

Original Sheet No. 36 original Sheet No. 36: Pending

FOOTNOTES APPLICABLE TO SHEET NOS. 8 THROUGH 25

1/ Peak rates are effective November through March and off-peak rates are effective April through October.

- 2/ Natural's transportation zones are:
 - a. IOWA-ILLINOIS RECEIPT ZONE

Includes Station 109 on the Amarillo Line and extends from Station 109 northward on the Amarillo Line and Southward on the Gulf Coast Line to the Illinois-Missouri border.

b. AMARILLO MAINLINE RECEIPT ZONE

Includes Station 106 and extends northward on the Amarillo Line from Station 106 to a point just south of Station 109.

c. GULF COAST MAINLINE RECEIPT ZONE

Extends northward on the Gulf Coast Line from the interconnection of the A/G and Gulf Coast
Lines (south of Station 305) to the Illinois-Missouri border.

d. MIDCONTINENT RECEIPT/DELIVERY ZONE

Extends westward from a point just west of Station 801 on the A/G Line to Natural's "Triangle"

area and ends just south of Station 106 on the Amarillo Line. This zone includes laterals in

Wise County, Texas, in Marshall County, Oklahoma and in Love County, Oklahoma.

e. TEXOK RECEIPT/DELIVERY ZONE

Includes Station 801 and extends from Station 801 on Natural's A/G Line southeastward to Station 302 and then east to the Texas/Louisiana border. This zone includes any contiguous offshore laterals which interconnect with Natural's system in this zone.

f. LOUISIANA RECEIPT/DELIVERY ZONE

Includes all of Natural's Louisiana Lateral east of the Texas/Louisiana Border and any contiguous offshore laterals which interconnect with Natural's system in this zone.

g. SOUTH TEXAS RECEIPT/DELIVERY ZONE

Consists of all facilities south of Station 302 on the Gulf Coast Line.

h. PERMIAN RECEIPT/DELIVERY ZONE

Consists of all facilities south of Station 112 on the Amarillo Line.

i. MARKET DELIVERY ZONE

Encompasses the same area as the Iowa-Illinois, Amarillo Mainline and Gulf Coast

Mainline

Receipt Zones combined. A separate incremental charge is applicable to transportation
on the

Moraine Lateral and is set out on Sheet No. 17. The Moraine Lateral is the 17.8 mile
segment

of pipeline in Lake County, Illinois that was formerly owned and operated by Moraine
Pipeline

Company. The Commission approved the acquisition of these facilities by Natural by
Order

issued September 22, 1993 in Docket No. CP93-351-000.

j. OFFSHORE ZONE

Consists of all non-contiguous laterals in the Gulf of Mexico.

FERC Docket: RP08-319-000

maximum

Original Sheet No. 37 Original Sheet No. 37: Pending

FOOTNOTES APPLICABLE TO SHEET NOS. 8 THROUGH 25

- 3/ The rates exclude the applicable charge, if any, for the transportation of liquids.
- 4/ The charges of the indicated Rate Schedules include the following storage cost components:

	RATE SCHEDULE FTS Reservation Charge All Zones*	RATE SCHEDULE FFTS Reservation Charge All Zones*	RATE SCHEDULE FTS-G Commodity Charge All Zones*	RATE SCHEDULE ITS Commodity Charge All Zones*
PEAK Maximum	\$0.2400	\$0.0079	\$0.0158	\$0.0079
Minimum	\$0.0000	\$0.0000	\$0.0000	\$0.0000
OFF-PEAK				
Maximum	\$0.2400	\$0.0079	\$0.0158	\$0.0079
Minimum	\$0.0000	\$0.0000	\$0.0000	\$0.0000

^{5/} For any service under Rate Schedule FTS or Rate Schedule FFTS to which the inclusive rate option in Section 6.3 of Rate Schedule FTS or Section 6.4 of Rate Schedule FFTS applies, the applicable

Inclusive Reservation rate shall be the sum of the applicable maximum reservation rate and 858 surcharge rate and the applicable minimum rate shall be the sum of the applicable minimum reservation

rate and 858 surcharge rate.

^{*} Except the Offshore Zone. The Offshore Zone rates do not contain a storage cost component.

Effective Date: 04/18/2008 Status: Effective FERC Docket: RP08-319-000

Sheet Nos. 38 - 39 Sheet Nos. 38 - 39 : Pending

Sheet Nos. 38 and 39 have been reserved for future use.

FERC Docket: RP08-319-000

Original Sheet No. 40 original Sheet No. 40: Pending

RATE SCHEDULE FFTS
FLEXIBLE FIRM TRANSPORTATION SERVICE

AVATTABITATY

This Rate Schedule FFTS is available to any entity (hereinafter called Shipper) which: (i) submits to Natural Gas Pipeline Company of America LLC (hereinafter called Natural) a valid request as defined in Section 3 hereof which Natural has the firm capacity and operational capability to satisfy; and (ii) executes a Flexible Firm Transportation Service Agreement (FFTS Agreement) with Natural applicable to service under this Rate Schedule FFTS. The form of FFTS Agreement is contained in this Tariff. There is no limitation on the number of FFTS Agreements any one Shipper may have.

2. APPLICABILITY, CHARACTER AND PRIORITY OF SERVICE

- (a) This Rate Schedule FFTS defines Natural's flexible firm transportation service under which firm service is available for at least the minimum number of days over a defined time period specified in the FFTS Agreement, as provided in Sections 2(c) and 7(a) of this Rate Schedule. A Shipper entering into an FFTS Agreement may elect a Late Notice (LN) option, a System-Wide Service (SW) option and/or No-Notice Balancing (NB) option, as described in Section 5 of this Rate Schedule FFTS, subject to payment of the rate and compliance with the terms and conditions attendant to such service option(s). This Rate Schedule FFTS shall apply to all gas received by Natural from or on behalf of Shipper pursuant to an FFTS Agreement. As more fully set out in the General Terms and Conditions of this Tariff, Natural is not providing a supply service under this Rate Schedule FFTS.
- (b) Service hereunder shall consist of the acceptance by Natural of natural gas by or on behalf of Shipper for transportation on days when service hereunder is available under the FFTS Agreement (Available Days) at the primary Receipt Point(s) specified in the FFTS Agreement, or at secondary Receipt Point(s) applicable to the FFTS Agreement, the transportation of that natural gas through Natural's system, and the delivery of that natural gas, adjusted for Fuel Gas and Gas Lost and Unaccounted For as set out in the General Terms and Conditions of this Tariff, by Natural to Shipper or for Shipper's account at the primary Delivery Point(s) specified in the FFTS Agreement, or at secondary Delivery Point(s) applicable to such

FERC Docket: RP08-319-000

Original Sheet No. 41 original Sheet No. 41: Pending

RATE SCHEDULE FFTS

FFTS Agreement. Natural shall not be required: (1) to accept hereunder on any day gas tendered, or to deliver hereunder on any day gas requested, in excess of the Maximum Daily Quantity (MDQ) specified in the FFTS Agreement (or the reduced volume specified on a partial Available Day) for each primary Receipt or Delivery Point or for the aggregate of all primary Receipt or Delivery Points (except to the extent Natural has accepted confirmed nominations allowing authorized overrun); (2) to accept or deliver gas hereunder on any day which is not an Available Day or partial Available Day under the FFTS Agreement; (3) to accept or deliver hereunder on any day gas which is not properly nominated pursuant to the General Terms and Conditions of this Tariff (it being recognized that a limited nonotice service is available under the NB service option); or (4) to receive or deliver gas hereunder on any day at secondary points except as provided in the General Terms and Conditions of this Tariff (it being recognized that a Shipper electing the SW option will have system-wide secondary point rights). Service hereunder shall be firm except for service at secondary points, subject to Section 2(c) helow.

- (c) (1) Service hereunder shall be provided on a firm basis on each Available Day (including partial Available Days up to the reduced volume) under the FFTS Agreement. The minimum and maximum number of Available Days over a month or other defined time period, and whether the Shipper is willing to accept notification of a partial Available Day under Section 7.1(a) hereof (and any limitations on such agreement), shall be stated in the FFTS Agreement, based on mutual agreement between Natural and Shipper, as provided in Section 3.2(a) of this Rate Schedule FFTS. Also based on mutual agreement between Natural and Shipper, the FFTS Agreement shall specify if certain days will not be Available Days (e.g., the FFTS Agreement might specify that weekend days or certain holidays are not Available Days). Administration of Available Days is covered in Section 7(a) of this Rate Schedule FFTS.
- (2) Service hereunder shall be firm and shall have the same priority on Available Days (including partial Available Days up to the reduced volume) as service under Rate Schedule FTS. No service shall be performed hereunder on a day which is not an Available Day or partial Available Day as determined under Section 7(a) of this Rate Schedule FFTS. Allocation of capacity, curtailment, priorities of service and the addition and/or deletion of primary points are governed by the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 42 original Sheet No. 42: Pending

RATE SCHEDULE FFTS

- (d) Natural posts unsubscribed firm transportation capacity by pipeline segment on the Informational Postings portion of its Interactive Website pursuant to Section 5.1(e) of the General Terms and Conditions of this Tariff. Such capacity will be available to provide service under either Rate Schedule FTS or FFTS; provided, however, that Natural may elect to make such firm capacity, available only for service under Rate Schedule FTS. Natural will only utilize for service under Rate Schedule FFTS firm capacity which is posted as available (such postings to be on the same basis and following the same procedures as for Rate Schedule FTS) and which is in excess of Natural's existing firm commitments. Natural awards firm capacity under this Rate Schedule FFTS, it will adjust the firm capacity posting on the Interactive Website to reflect its obligations under the FFTS Agreement. The adjustment may consist of a reduction in the unsubscribed firm capacity, based on an engineering study. Given the nature of service under Rate Schedule FFTS (specifically the parameters relating to Available Days under the FFTS Agreement), Natural may not have to reduce posted firm capacity on a oneto-one basis in relation to the MDQ under the FFTS Agreement. Alternatively, Natural may continue to reflect the capacity dedicated to the FFTS Agreement as unsubscribed firm capacity, but subject to limits on the number of days available based on the Available Day and related parameters under the FFTS Agreement. For the purposes of this provision, the term "related parameters" shall include the minimum and maximum number of Available Days, the associated time period(s), whether Shipper will accept partial Available Days (and any limitations thereon) and any restrictions in the FFTS Agreement that certain days are not Available Days. The revised firm capacity posted as unsubscribed will be available for service under Rate Schedule FTS or Rate Schedule FFTS; provided, however, that Natural may elect to make such firm capacity available only for service under Rate Schedule FTS.
- (e) The service provided under this Rate Schedule FFTS shall be performed under Part 284 of the Commission's Regulations. Shipper shall only tender gas for transportation under this Rate Schedule FFTS to the extent such service would qualify under the applicable statutes, regulations, Commission orders and the blanket certificate authorizing service by Natural under this Rate Schedule. For service under Subpart B of Part 284 of the Commission's Regulations, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by the Commission's Regulations, Shipper shall (prior to tendering gas under an FFTS Agreement) cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit the necessary certification.

FERC Docket: RP08-319-000

Original Sheet No. 43 original Sheet No. 43: Pending

RATE SCHEDULE FFTS

(f) Shipper may release and segment capacity dedicated to service hereunder to the extent permitted by, and subject to the terms and conditions contained in, the General Terms and Conditions of this Tariff and subject to the conditions stated in this subsection and in Section 8(c) of this Rate Schedule FTTS. As under Rate Schedule FTTS, a Shipper electing the SW option may not segment a capacity release except within the primary path(s) under the FFTS Agreement and neither the releasing nor any replacement Shipper involved in a segmented release will have system-wide secondary point access under the FFTS-SW Agreement or any release Agreement originating there from after a

FERC Docket: RP08-319-000

Original Sheet No. 44 Original Sheet No. 44: Pending

RATE SCHEDULE FFTS

3. VALID REQUESTS

- 3.1 A request for service under this Rate Schedule FFTS shall be valid as of the date received if it complies with this Section and contains adequate information on all of the items specified in Section 3.2, subject to any necessary verification of such information and to the following:
- A request shall not be valid and Natural shall not (a) be required to grant any such request: (i) as to which Natural does not have the firm capacity and operational capability to effect receipt, transportation and/or delivery on a firm basis consistent with the terms and conditions of this Rate Schedule FFTS and the request for service hereunder, including the minimum number of Available Days and the associated defined time period(s) and any other requested parameters as to Available Days requested by Shipper; (ii) which, except as provided for in Section 3.1(b) hereof, would require the construction, modification, expansion, or acquisition of any facilities; provided, however, that Natural may agree in its reasonable discretion to construct, modify, expand, or acquire facilities to enable it to perform such services; (iii) unless and until Shipper has provided Natural with the information required in Section 3.2 hereof; (iv) if Natural determines, based on the credit analysis referenced in Section 3.2(h), that Shipper does not possess sufficient financial stability to make it reasonably likely the service provided hereunder will be paid for on a timely basis; (v) if the service requested would not comply with this Rate Schedule FFTS; or (vi) if the service requested is at less than the applicable maximum rate; provided, however, that Natural may agree to provide service hereunder at a discount consistent with this Rate Schedule FFTS. Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to this Section and while an FFTS Agreement is in effect. Such matters are governed by the General Terms and Conditions of this Tariff.
- (b) Natural agrees to construct minor meter and tap facilities to effectuate service requested under this Rate Schedule FFTS provided that: (i) the cost of such facilities does not exceed \$100,000; (ii) Natural shall be reimbursed prior to commencing construction; and (iii) the facilities will not adversely affect Natural's operations.

FERC Docket: RP08-319-000

Original Sheet No. 45 original Sheet No. 45: Pending

RATE SCHEDULE FFTS

- (c) In the event that Natural agrees to construct, modify, expand or acquire facilities to perform service under this Rate Schedule FFTS, Natural shall maintain a separate record of the nature and costs of such facilities.
- (d) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request, in whole or in part, due to lack of firm capacity or system capability or if the request is incomplete or does not comply with this Rate Schedule FFTS. Any request shall be null and void unless it is substantially complete and complies with this Rate Schedule. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the FFTS Agreement, after which Shipper shall have fifteen (15) days to provide the specified information. In the event such information is not received within fifteen (15) days, Shipper's request shall be null and void.
- (e) Natural shall tender an FFTS Agreement to Shipper for execution when Shipper's request for service is accepted. Unless waived by Natural, a request for service shall be invalid if Shipper fails to execute an FFTS Agreement hereunder within ten (10) days after an FFTS Agreement which accurately reflects the Shipper's request has been tendered by Natural for execution.
- 3.2 Requests for service hereunder shall be deemed valid only after the information specified in this Section is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department, One Allen Center, 500 Dallas Street, Houston, Texas 77002, or Telecopy Number (713) 369-9305. Qualifying bids in an auction shall be treated as a valid request provided that the bid is supplemented with all information required under this Section 3.2 as required by the General Terms and Conditions of this Tariff. The information required for a valid request shall be as follows:

FERC Docket: RP08-319-000

Original Sheet No. 46 Original Sheet No. 46: Pending

RATE SCHEDULE FFTS

(a) GAS OUANTITIES

The request shall specify in Dth the aggregate MDQ and the MDQ for each primary point, exclusive of applicable Fuel Gas and Gas Lost and Unaccounted For, provided that Natural shall not be obligated to accept requests for an aggregate MDQ of less than one hundred (100) Dth per day. If Shipper is willing to accept a lesser MDQ than it has requested, Shipper shall also specify the minimum acceptable MDQ. The request shall also specify the minimum and maximum acceptable number of Available Days; the associated defined time period(s); whether Shipper is willing to accept notification of partial Available Days under Section 7.1(a) hereof (and any limitations on such agreement); and whether Shipper requests that certain days not be Available Days. The minimum and maximum number of Available Days or other parameters as to Available Days may vary from one defined time period to another over the term of the FFTS Agreement and the request must specify any such variance desired by Shipper.

(b) RECEIPT POINT(S)

The request shall specify the primary point(s) at which Shipper desires Natural to receive gas. By utilizing the comment section, the request may also specify a pipeline segment if the Shipper is willing to accept any point in that segment or the request may specify points which Shipper is willing to accept, and/or which points Shipper is not willing to accept, within a pipeline segment.

(c) DELIVERY POINT(S)

The request shall specify the primary point(s) at which Shipper desires Natural to deliver gas. By utilizing the comment section, the request may also specify a pipeline segment if the Shipper is willing to accept any point in that segment or the request may specify points which Shipper is willing to accept, and/or which points Shipper is not willing to accept, within a pipeline segment.

(d) LIMITATION OF POINT(S)

(1) A Shipper may request any number of primary Receipt and primary Delivery Points so long as the summation of MDQs at all primary Receipt Points and at all primary Delivery Points equals the aggregate MDQ.

FERC Docket: RP08-319-000

Original Sheet No. 47 Original Sheet No. 47: Pending

RATE SCHEDULE FFTS

(2) The availability to Shipper of secondary points, and the related priorities and volumes, shall be governed by the General Terms and Conditions of this Tariff. System-wide service on a secondary point basis will be available hereunder to a Shipper which elects the SW service option.

(e) SERVICE OPTION

 $$\operatorname{\textsc{The}}$ request shall specify whether Shipper elects any or all of the LN, SW or NB service options.

(f) TERM OF SERVICE

The request shall specify:

(1) The date service is requested to commence;

and

(2) The date service is requested to terminate.

(g) REQUESTED RATE

The request shall state the rate at which the Shipper is willing to contract for the service requested, including the additional charges for service options where applicable. If no rate is stated, Shipper will be deemed to have agreed to the maximum rate.

(h) CREDIT

 $\label{eq:Acceptance} \text{Acceptance of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.}$

(i) COMPLIANCE WITH FFTS TARIFF

Submission of a request for service hereunder shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule FFTS, including the applicable General Terms and Conditions. This agreement extends to the terms and conditions applicable to any service option (NB, SW and/or LN) elected by a Shipper under subsection (e) of this Section 3.2 and to any service option elected during the term of the FFTS Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 48 Original Sheet No. 48: Pending

RATE SCHEDULE FFTS

(i) COMMISSION-REQUIRED FILING INFORMATION

The following information is to be provided at the time a request for service hereunder is submitted, if available, or when an initial nomination for service under an executed FFTS Agreement is submitted, and when any subsequent changes occur:

- (1) Affiliation of the Shipper with Natural; and
- (2) The identity of the Shipper, including whether it is a local distribution company, an interstate pipeline company, an intrastate pipeline company, an end user, a producer, or a marketer.

4. TERM

- (a) The term of service hereunder shall be set forth in the FFTS Agreement between Shipper and Natural.
- (b) The General Terms and Conditions of this Tariff shall govern the applicability of, and the terms and conditions relating to, rollovers and the right of firm refusal vis a vis an FFTS Agreement. Upon termination of any FFTS Agreement, and subject to any such rollover right or right of first refusal, service by Natural to Shipper thereunder shall be terminated and automatically abandoned.
- (c) Natural may terminate any FFTS Agreement if Natural is required by the FERC or some other agency or court to provide firm service for others utilizing the system capacity or capability required for service under such FFTS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the type covered by the FFTS Agreement; provided, however, that nothing herein shall be read to protect Natural from the consequences of its own negligence.

FERC Docket: RP08-319-000

Original Sheet No. 49 Original Sheet No. 49: Pending

RATE SCHEDULE FFTS

5. SERVICE OPTION

- 5.1 (a) A Shipper electing the LN option shall have the right to submit late nominations and Intra-day Nominations consistent with the rights under that service option specified in the General Terms and Conditions of this Tariff.
- (b) The LN option may be requested as part of the request for service under Section 3 of this Rate Schedule or may be requested at any time during the term of an FFTS Agreement; provided, however, that the election of the LN service option must be for a minimum of one year or the remaining term of the FFTS Agreement, whichever is less; and provided further that the LN option must be effective on the first day of a month unless otherwise mutually agreed.
- (c) The maximum Reservation Rate applicable to an FFTS Agreement with an LN option shall be increased by the LN Surcharge as set out in Section 6.3(a)(2) of this Rate Schedule FFTS.
- (d) Natural shall post on the Informational Postings portion of its Interactive Website each month the availability of system capability, if any, to render additional LN service in excess of then existing commitments.
- A Shipper electing the SW option shall have the right on a secondary point basis to nominate at all Receipt and Delivery Points on Natural's system; provided, however, that a Shipper under an FFTS Agreement paying the highest maximum Reservation Charge which could apply under Natural's effective rate matrix to any haul on Natural's system shall automatically have such rights. Except for points which are within a path defined by primary points under the FFTS Agreement, all such secondary points shall be treated as having an out-of-path priority for scheduling and curtailment. Such out-ofpath secondary points shall be on the same basis for scheduling and curtailment as out-of-path secondary points under any other FTS or FFTS Agreement, including those within a zone transversed by such Agreement and those arising out of opposite leg secondary point rights. For an FFTS Agreement with an SW option, the applicable maximum Reservation Rate shall be determined by the SW option formula set out in Section 6.3(a)(3) of this Rate Schedule FFTS.

FERC Docket: RP08-319-000

Original Sheet No. 50 original Sheet No. 50: Pending

RATE SCHEDULE FFTS

- 5.3 (a) Shippers selecting the NB option may, subject to the limitations in subsection (d), inject or withdraw gas on any Available Day under the FFTS Agreement on a no-notice basis (within the limits set out herein) under Rate Schedule NSS or from a Third Party Storage Point as defined in Section 9 of the General Terms and Conditions of this Tariff. The NB option shall be available on a non-discriminatory basis for use in conjunction with either Rate Schedule NSS or a Third Party Storage Point which meets the criteria set out in subsection (d)(3).
- (b) Injections may be made on a firm, no-notice basis on any Available Day to the extent Natural can effectuate such injections employing the capacity dedicated to the FFTS-NB Agreement not otherwise utilized on that day and the storage rights under the NSS Agreement(s) and/or third party storage associated with such FFTS-NB Agreement not otherwise utilized on that day. A Shipper will be able to utilize rights under the NB option on any Available Day to the extent it has adequate storage and/or transportation capacity, as applicable, under its FFTS-NB Agreement(s) and the related NSS Agreement (or third party storage) comprising that Shipper's NB option, which rights the Shipper is not otherwise utilizing on that day. It shall be Shipper's responsibility to assure that Natural is not called upon to perform no-notice service hereunder in excess of that level. Shipper may be subject to Overrun and Balancing Service Charges to the extent no-notice volumes exceed its available rights.
- (c) Withdrawals may be made on a firm, no-notice basis on any Available Day during the period of April 1 through October 31 to the extent Natural can effectuate such withdrawals employing the capacity dedicated to the FFTS-NB Agreement not otherwise utilized on that day and the storage rights under the NSS Agreement(s) and/or third party storage associated with such FFTS-NB Agreement not otherwise utilized on that day. It shall be Shipper's responsibility to assure that Natural is not called upon to perform no-notice service hereunder in excess of that level. Shipper may be subject to Overrun and Balancing Service Charges to the extent no-notice volumes exceed its available rights.
- (d) The applicable maximum rate for Shippers electing the NB option shall not be higher than for other Shippers under Rate Schedule FFTS, but the following special conditions shall apply:

FERC Docket: RP08-319-000

Original Sheet No. 51 Original Sheet No. 51: Pending

RATE SCHEDULE FFTS

(1) The NB option may be requested as part of the request for service under Section 3 of this Rate Schedule FFTS or by mutual agreement (Natural's agreement shall not be unreasonably withheld and any decision to withhold must be based on operating reasons) provided, however, that the election of the NB option must be for a minimum of one year or the remaining term of the FFTS Agreement, whichever is less; and provided further, that the NB option must be effective on the first day of a month unless otherwise mutually agreed. Natural shall provide a written explanation of the operational basis for rejecting any request to add the NB option to an existing Agreement.

- (2) The primary path under the FFTS Agreement must be such that the NSS storage point would be treated as in-the path under Section 5.6 of the General Terms and Conditions of this Tariff or, where a Third Party Storage Point is being utilized, such point must be within the primary path under the FFTS-NB Agreement.
- (3) Any Third Party Storage Point shall be available for service under this option only to the extent the operator of that point and Natural enter into a Storage Point Swing Service Operating (SPSSO) agreement which defines how such operator will accommodate Shipper swing volumes on a no-notice basis, how the operator is to make the corresponding operational physical changes, the limitations on the level of swings to be accommodated and the consequences if such levels are exceeded or operational changes are not made. It is the intent that such an agreement provide Natural with the ability to call upon such third party storage (within the limits defined in the agreement) reasonably comparable to that which Natural exercises with respect to its own storage fields, which may include short-term notice arrangements with the operator. The SPSSO agreement must specify a predetermined allocation methodology and shall specify the extent to which and the conditions under which Shippers shall be kept whole because the operator is agreeing to take the swing. Because the third party storage provider under an SPSSO agreement will be providing swing service, Natural shall not be responsible (as it might under an ordinary operational balancing agreement) for swing within the agreed limits applicable to such Third Party Storage Point. If Natural rejects any request that NB service be provided utilizing a Third Party Storage Point, Natural shall make such rejection in writing and set out the reasons for the rejection.

FERC Docket: RP08-319-000

Original Sheet No. 52 Original Sheet No. 52: Pending

RATE SCHEDULE FFTS

(4) The Shipper electing an NB option must designate both: (i) which NSS Agreement or Third Party Storage Point is to be associated with the FFTS-NB Agreement in providing no-notice service (third party storage requires the consent of the operator of such storage) and (ii) which points are to be covered by no-notice service under the FFTS-NB Agreement (such point designation shall require the concurrence of the point operator to the extent required by any allocation agreement at the point). The point designation in the prior sentence may be to a CDP (if it is to a point within a CDP, such CDP may need to be subdivided for administrative purposes) or to a specific point on which telemetry exists, or which is otherwise mutually agreed, covered by the FFTS-NB Agreement. Any number of FFTS and/or FTS Agreements may be associated with any NSS Agreement or Third Party Storage Point, but an FFTS Agreement may not be associated with more than one NSS Agreement or Third Party Storage Point; provided, however, that under a predetermination allocation agreement pursuant to Section 11 of the General Terms and Conditions of this Tariff, a Shipper or point operator can designate a linkage between an FFTS Agreement and multiple NSS Agreements and Third Party Storage Points so long as an administratively workable protocol for linking the transportation and storage components is specified. designations hereunder may be changed on reasonable prior notice to Natural.

- (5) The NB option is available for use only in conjunction with an NSS Agreement or comparable third party storage agreement at a Third Party Storage Point. The NB option shall terminate if neither of such elements is present.
- (e) No-notice activity under an FFTS-NB Agreement will be posted to the storage point in the receipt zone where the imbalance occurs or to the closest storage point where Shipper has adequate storage rights to accommodate such activity, consistent with Section 5.6(c)(4) of the General Terms and Conditions of this Tariff.
- 5.4 A Shipper under Rate Schedule FFTS is not obligated to choose any of the options described in this Section 5, but may select any one of them, any two of them or all three of them. Once a Shipper has elected an option and so long as it remains in effect, the Agreement for service to such Shipper may be referenced by placing the relevant letters after FFTS (FFTS-LN, FFTS-NB, FFTS-SW).

FERC Docket: RP08-319-000

Original Sheet No. 53 Original Sheet No. 53: Pending

RATE SCHEDULE FFTS

6. RATE

- 6.1 Shipper shall pay Natural each month for service under this Rate Schedule FFTS the sum of the following charges:
- (a) a Reservation Charge based on Shipper's MDQ and the number of Available Days under the FFTS Agreement that month;
- (b) a Commodity Charge for each unit of gas delivered hereunder by Natural during the billing month;
 - (c) Gathering and Offshore Charge(s);
- $\mbox{(d)} \qquad \mbox{all applicable types of transition cost charges;} \\ \mbox{and} \\$
- (e) any other applicable charges and penalties set out in this Rate Schedule FFTS or the General Terms and Conditions of this Tariff.
- 6.2 Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.
- $6.3\,$ The charges identified in items (a) (d) of Section 6.1 shall be assessed as follows:
- (a) (1) The maximum Reservation Charge in any month for service under this Rate Schedule FFTS shall be the applicable maximum unit Daily Reservation Rate(s) set out in this Tariff multiplied by Shipper's MDQ and multiplied by the number of Available Days during that month. The Reservation Charge payable by any Shipper shall be the maximum Reservation Charge unless a lower Reservation Charge is specified in the FFTS Agreement or a separate discount agreement or is assessed pursuant to Section 6.9 hereof,

FERC Docket: RP08-319-000

Original Sheet No. 54 Original Sheet No. 54: Pending

RATE SCHEDULE FFTS

which lower charge must be consistent with the minimum Reservation Rate(s) specified in this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma Agreement set out in this Tariff, discounts may be limited to specific volumes. Shipper shall be obligated to pay the Reservation Charge hereunder beginning as of the date service under the FFTS Agreement commences. Service shall be deemed to commence on the later of the date specified in the FFTS Agreement or the date Natural is able to render firm service hereunder to Shipper.

- (2) For Shippers electing the LN option, the applicable maximum reservation rate shall be increased by the LN Surcharge as set out in this Tariff.
- (3) For Shippers electing the SW option, the applicable maximum unit Daily Reservation Rate shall be determined in accordance with the following formula:

where:

- RR = applicable maximum unit Daily Reservation Rate.
- PPR = the applicable maximum unit Daily Reservation Rate
 for the weighted average of all primary paths under
 the Agreement.
- HRR = the highest maximum unit Daily Reservation Rate
 which could apply to any haul on Natural's system
 under Natural's effective Rate Schedule FFTS rate
 matrix.

If applicable, the LN Surcharge shall be added to the rate so determined. $\,$

(b) The maximum monthly Commodity Charge shall be the applicable maximum unit Commodity Rate(s) set out in this Tariff multiplied by the quantity of gas delivered by Natural to Shipper hereunder during the billing month plus an amount equal to the volumes attributable to Fuel Gas and Gas Lost and Unaccounted For multiplied by one-half of the applicable maximum unit Commodity Rate(s). The Commodity Charge payable by any Shipper shall be the maximum Commodity Charge unless a lower Commodity Charge is

FERC Docket: RP08-319-000

Original Sheet No. 55 Original Sheet No. 55: Pending

RATE SCHEDULE FFTS

specified in the FFTS Agreement or a separate discount agreement or is assessed pursuant to Section 6.9 hereof, which lower charge must be consistent with the minimum Commodity Rate(s) specified in this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma Agreement set out in the Tariff, discounts may be limited to specific volumes. For any volumes delivered in excess of Shipper's MDQ, which deliveries shall be subject to Section 9 of this Rate Schedule FFTS, Shipper shall pay the applicable Overrun Charge set out in Section 12 of the General Terms and Conditions of this Tariff plus the sum of the applicable transportation Commodity Rate(s) multiplied by the overrun volumes and an amount equal to the volumes attributable to Fuel Gas and Gas Lost and Unaccounted For multiplied by one-half of the applicable Commodity Rate(s).

- (c) The Reservation and Commodity Charges will be based on the primary Receipt and Delivery Points specified in the FFTS Agreement; provided, however, that for a Shipper electing the SW option utilizing a point outside of its primary path, the Commodity Charge shall be based on the actual path utilized. Unless Natural agrees to the contrary, the applicable Reservation Charges will not be reduced during the primary term of the FFTS Agreement due to a change in the primary points; provided, however, that Commodity Charges will be adjusted to reflect a change in a primary point which results in a shortened path as provided in Section 5.1(c)(5) of the General Terms and Conditions of this Tariff.
- (d) Separate Gathering Charge(s) shall apply to all volumes received by Natural under this Rate Schedule at any Receipt Point(s) on its gathering facilities. Similarly, separate Offshore Charges shall apply to all volumes received by Natural under this Rate Schedule at any Receipt Point(s) on any non-contiguous offshore lateral.
- (e) Transition cost charges shall be assessed in accordance with the General Terms and Conditions of this Tariff.
- $6.4\,$ OPTIONAL RATE FORM INCLUSIVE OF RESERVATION SURCHARGES. If a Shipper desires to have a reservation rate which is inclusive of reservation surcharges, the form of Inclusive Reservation rate included in the Agreement shall, in lieu of that set out in Sections 6.1 and 6.3 above, be as set out in this Section.
- (a) The Inclusive Reservation rate hereunder shall be a rate which includes all applicable base reservation rates and the reservation surcharges under Section 21 of the General Terms and Conditions (858 Surcharges). The Inclusive Reservation rate may not

FERC Docket: RP08-319-000

Original Sheet No. 56 original Sheet No. 56: Pending

RATE SCHEDULE FFTS

exceed the applicable maximum Inclusive Reservation rate and may not be less than the applicable minimum Inclusive Reservation rate under this provision; provided, however, that nothing herein shall preclude a negotiated rate at a different level under Section 49 of the General Terms and Conditions of this Tariff.

- (b) The applicable maximum Inclusive Reservation rate for any Agreement shall be the sum of the applicable maximum base reservation rate and 858 Surcharge rate. If a Shipper is exempt from 858 Surcharges, the applicable maximum Inclusive Reservation rate for that Shipper shall be reduced accordingly. If the applicable maximum rate for any of these components changes, the applicable maximum Inclusive Reservation rate shall change by recalculation of the sum of these components.
- (c) In the event that Shipper and Natural agree on a discounted Inclusive Reservation rate, the Inclusive Reservation rate shall be deemed to consist of the applicable maximum 858 Surcharge with the discounted base reservation rate being calculated from the Inclusive Reservation rate after subtraction of those components. If the Inclusive Reservation rate is less than the 858 Surcharge component, the discount shall be deemed to apply first to the reservation rate and then to the 858 Surcharge. If the level of the applicable maximum rate for any rate component changes, the level of the various discounted component(s) shall be recalculated. The discounted Inclusive Reservation rate may not be less than the sum of the applicable minimum Reservation rates for these rate components.
- (d) The Inclusive Reservation rate hereunder shall be billed as a monthly demand rate applied to the MDQ of the Agreement.
- (e) If an Agreement includes a rate form which conforms to this Section and does not contain any other negotiated rate elements, the filing requirements under Section 49 of the General Terms and Conditions of this Tariff shall not apply to such Agreement.
- (f) The Inclusive Reservation rate shall not include commodity rate(s), commodity surcharges, ACA, fuel or overrun charges, which are determined and assessed as applicable under other provisions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 57 Original Sheet No. 57: Pending

RATE SCHEDULE FFTS

6.5 (a) Shipper shall reimburse Natural within five (5) business days after costs have been incurred by Natural for all fees required by the FERC or any regulatory body including, but not limited to, filing, reporting, and application fees to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees (such as general rate case filing fees).

(b) If Natural constructs, acquires or modifies any facilities to perform service hereunder, then as specified in an agreement between the parties either:

FERC Docket: RP08-319-000

Original Sheet No. 58 Original Sheet No. 58: Pending

RATE SCHEDULE FFTS

(i) Shipper shall reimburse Natural for the cost of such facilities or facility modifications as described in the General Terms and Conditions of this Tariff; or

- (ii) Natural shall assess a monthly charge reflecting such facility costs.
- 6.6 The ACA and RD&D charges will be assessed, when applicable, as provided in the General Terms and Conditions of this Tariff, on volumes delivered by Natural to Shipper under this Rate Schedule FFTS. Shipper shall also reimburse Natural for Fuel Gas and for Gas Lost and Unaccounted For hereunder as provided by the General Terms and Conditions of this Tariff.
- 6.7 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in: (1) the rates and charges applicable under this Rate Schedule FFTS, including both the level and design of such rates and charges; or (2) the terms and conditions of this Rate Schedule FFTS. Natural agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.
- (b) If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the FFTS Agreement or a separate discount agreement, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such highest rate, Shipper shall execute or provide such documentation within fifteen (15) calendar days after a written request by Natural. If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for transportation service hereunder, the rate shall be decreased to such reduced rate.

FERC Docket: RP08-319-000

Original Sheet No. 59 Original Sheet No. 59: Pending

RATE SCHEDULE FFTS

- 6.8 Shipper may request that Natural process its gas in order that such gas will meet the quality specifications of the General Terms and Conditions of this Tariff. Natural may, but is not obligated to, provide such service. The providing of such service and the terms and conditions therefore shall be negotiated between Natural and Shipper and the terms and conditions agreed upon shall be specified in the FFTS Agreement or a separate treating/processing agreement. For any such service, Natural shall assess Shipper a charge reflecting all the costs of providing the service, including a Reservation as well as Commodity Charge where appropriate.
- Natural may from time to time and at any time, upon twenty-four (24) hours' verbal or written notice, subject to any provisions on discounting in the FFTS Agreement or in a separate discount agreement, charge any individual Shipper for service under this Rate Schedule FFTS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate may not be less than the applicable minimum rate for service under Rate Schedule FFTS set forth in this Tariff. Natural will confirm any verbal notice of the applicable charge in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Unless otherwise agreed in the FFTS Agreement or in a separate discount agreement, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the FFTS Agreement and/or any discount agreement) upon twenty-four (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with the Commission any and all reports as required by the Commission's Regulations with respect to the institution or discontinuance of any discount.
- 6.10 All revenues and gas in kind collected by Natural as a result of providing service under this Rate Schedule FFTS shall be retained by Natural unless Natural has otherwise explicitly agreed on a different disposition of such amounts. Where crediting or refund mechanisms apply under other provision(s) of this Tariff or pursuant to effective Commission orders or settlements, such mechanisms shall supersede this Section to the extent necessary to carry out such provision(s).

FERC Docket: RP08-319-000

Original Sheet No. 60 Original Sheet No. 60: Pending

RATE SCHEDULE FFTS

6.11 (a) Natural may, but is not obligated to, transport liquids for Shipper. Any transportation of liquids on behalf of Shipper shall be pursuant to a separate agreement, which agreement shall set out the negotiated rate for such transportation and shall supersede subsection (b) of this Section; provided, however, that if Natural does provide any such transportation in the absence of a separate agreement, subsection (b) of this Section shall apply. The credit evaluation provisions of the General Terms and Conditions of this Tariff shall be applicable to such transportation unless otherwise specified in a separate agreement.

- (b) (1) Absent a separate agreement (in which case the negotiated rate set out in such agreement shall govern), Natural shall charge Shipper and Shipper shall pay Natural for Natural's transportation and handling of Shipper's liquids a monthly fee equal to \$1.025 per barrel of liquids.
- If, at any time and from time to time, as a (2) result of any legislation or any order, opinion, or decision of the Commission or of any other governmental authority having jurisdiction over Natural's rates (including any order, opinion, or decision based in whole or in part on a contested or uncontested settlement agreement relative to Natural's rates) an amount is allocated or attributed to the transportation and handling of liquids, liquefiables, plant fuel, vaporous non-hydrocarbons and/or similar materials, and such amount is credited to Natural's cost of service or otherwise excluded in determining Natural's rates, Shipper shall reimburse Natural for the sum of money (including the interest reflected therein) Natural is required to refund or credit, or has refunded or credited to its customers which is attributed by Natural to such transportation and handling for Shipper under the FFTS Agreement and which has not been assessed and collected from Shipper for the time period to which the refund or credit applied. If such refunded or credited money is less than the sum of money assessed and collected from Shipper for the applicable time period, Natural shall remit the difference to Shipper within thirty (30) days of the applicable order, opinion, or decision. Thereafter, Shipper shall pay Natural monthly fees based on amounts which are attributable by Natural to such transportation and handling for Shipper under the FFTS Agreement as calculated and assessed by Natural hereunder. Such monthly fees shall be calculated so as to permit Natural to generate monthly revenues therefrom to equal the amounts expected to be credited to Natural's cost of service or excluded from the determination of Natural's rates.

FERC Docket: RP08-319-000

Original Sheet No. 61 original Sheet No. 61: Pending

RATE SCHEDULE FFTS

(3) The amounts attributed by Natural to such transportation and handling for Shipper under the FFTS Agreement shall be established from time to time above based upon the application of a methodology determined by Natural in good faith to be consistent with applicable orders, opinions, or decisions pertaining to such transportation and handling.

- (4) Any settlement, compromise, or other action of Natural in connection with any such order, opinion, or decision pertaining to transportation of liquids, liquefiables, plant fuel, vaporous non-hydrocarbons and/or other similar materials shall be conclusive and binding upon Shipper.
- 6.12 Natural shall perform exchanges and backhauls hereunder to the extent firm capacity is available. Exchanges and backhauls shall be subject to the maximum and minimum rates under this Rate Schedule except that no fuel need be charged.

7. NOMINATIONS, BALANCING SERVICE CHARGES AND IMBALANCES

- (a) (1) Natural shall notify Shipper no later than two and one-half hours prior to the deadline for timely nominations for a day if that day is not to be an Available Day. Notification shall be made by e-mail or other mutually agreed method. To the extent consistent with the FFTS Agreement, the notification may specify that the day is a partial Available Day, which means that the service available to Shipper that day under the FFTS Agreement is less than Shipper's full MDQ, in which case the notice must specify the volume of service which is available to Shipper that day. A partial Available Day shall not be accounted for as a full Available Day, but rather as a fraction of an Available Day in the ratio of the volume of service made available for the Shipper under its FFTS Agreement that day to that Shipper's MDQ under the FFTS Agreement.
- (2) If timely notice has not been provided under Section 7(a)(1), the day shall be deemed a full Available Day unless inconsistent with the FFTS Agreement or unless the maximum number of Available Days for the defined time period has already been reached. Once the maximum number of Available Days for a defined time period has been reached under a specific FFTS Agreement, any service provided under that FFTS Agreement on a subsequent day during that defined time period shall be overrun, which is governed by Section 9 of this Rate Schedule FFTS.

FERC Docket: RP08-319-000

Original Sheet No. 62 original Sheet No. 62: Pending

RATE SCHEDULE FFTS

No service shall be provided under an FFTS Agreement on a day which is not an Available Day or partial Available Day.

- separately for each FFTS Agreement, subject to the parameters defined in the FFTS Agreement, i.e., the minimum and maximum Available Days and the associated defined time period(s), any restrictions stated in the FFTS Agreement that certain days are not Available Days, and whether partial Available Days are permitted under the FFTS Agreement. To the extent that Natural can identify the conditions under which it will be unable to consider a particular day an Available Day, Natural shall specify such conditions in the FFTS Agreement; provided, however, that Natural will still be able to determine that a day is not an Available Day based on other conditions. The determination of which days are Available Days under an FFTS Agreement is to be based on operational considerations and/or on Natural's commitments under other firm Agreements, including the number of Available Days under other FFTS Agreements.
- (b) Shipper shall provide Natural with nominations of receipts and deliveries by Receipt and Delivery Point, including pooling points where applicable, in accordance with the General Terms and Conditions of this Tariff. Shippers electing the LN option may submit late and Intra-day Nominations consistent with their rights under that option as specified in the General Terms and Conditions of this Tariff. It shall be Shipper's responsibility to cause gas to be delivered to Natural at Receipt Point(s), and to cause gas to be taken from Natural at Delivery Point(s), in accordance with the information supplied to Natural. In the event that actual receipts and deliveries at any point fail to conform to Shipper's nominations, Balancing Service Charges will apply to the extent provided in the General Terms and Conditions of this Tariff.
- (c) It shall be Shipper's responsibility to keep receipts and deliveries in balance. Shipper must submit nominations to rectify an imbalance to the extent required by Section 8.4 of the General Terms and Conditions of this Tariff. Any imbalance between actual receipts and actual deliveries shall be eliminated by cashout on a monthly basis in accordance with the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 63 original Sheet No. 63: Pending

RATE SCHEDULE FFTS

(d) To the extent a Shipper elects the NB option, and within the limitations of its no-notice rights thereunder as defined in Section 5 of this Rate Schedule FFTS, Shipper shall not be obligated to submit nominations to effectuate injections or withdrawals (withdrawals are available on a no-notice basis only from April 1 through October 31) in conjunction with service under an FFTS Agreement.

- 8. RECEIPT AND DELIVERY POINTS; UPSTREAM AND DOWNSTREAM ARRANGEMENTS; RELEASES AND SEGMENTATION
 - The primary point(s) of receipt for all gas tendered to Natural for transportation hereunder and the primary point(s) of delivery for all gas delivered by Natural to Shipper (or to a thirdparty on behalf of Shipper) hereunder shall be specified in the FFTS Agreement. For each individual primary Receipt and primary Delivery Point, and for the aggregate of all such points, Natural's maximum obligation to accept and deliver gas on a firm basis shall be $% \left\{ \left(1\right) \right\} =\left\{ \left($ specified in Dth in the FFTS Agreement. The sum of the MDQs for primary Receipt Points and the sum of the MDQs for primary Delivery Points shall not exceed the aggregate MDQ. All stated MDQs shall be exclusive of applicable Fuel Gas and Gas Lost and Unaccounted For. The availability of secondary Receipt and Delivery Points, and the related volumes and priorities at such points, shall be governed by the General Terms and Conditions of this Tariff. Conditions of delivery at Receipt and Delivery Points are set out in the General Terms and Conditions of this Tariff.
 - (b) Shipper shall make all necessary arrangements with other parties, both: (1) at or upstream of the Receipt Point(s) where gas is tendered to Natural hereunder; and (2) at or downstream of the Delivery Point(s) where Natural delivers gas hereunder to or for the account of Shipper. Such arrangements must be consistent with this Rate Schedule FFTS and must be coordinated with Natural.
 - (c) If a Shipper releases its capacity hereunder, in whole or in part, any and all replacement Shippers or sub-replacement Shippers must accept the same provisions regarding and circumstances relating to Available Days as apply under the original Agreement, including the number of Available Days remaining in any applicable defined time period. Any required capacity release posting must so state and must describe the contract provisions relating to Available Days. Capacity on an Available Day or partial Available Day shall be prorated among the original and replacement

FERC Docket: RP08-319-000

Original Sheet No. 64 Original Sheet No. 64: Pending

RATE SCHEDULE FFTS

Shippers based on their respective MDQs effective on that day. Natural shall advise any replacement Shipper that a particular day is not an Available Day or is a partial Available Day. Once capacity reverts to the original Shipper, that Shipper is bound by the number of Available Days remaining in the relevant defined time period. If capacity is segmented by the Shipper itself or on release, the same contract provisions as to Available Days and the same determination by Natural of Available Days on any day shall apply to each segment.

9. OVERRUN SERVICE

Upon request of Shipper, Natural may (but is not obligated to) receive, transport, and deliver on any Available Day quantities of natural gas in excess of Shipper's MDQ under the FFTS Agreement when, in Natural's reasonable judgment, the capacity and operating capability of its system will permit such receipt, transportation and delivery without impairing the ability of Natural to meet its other obligations. On any day which would be an Available Day except that the maximum number of Available Days for the defined time period under the FFTS Agreement has been reached, any quantity of natural gas received, transported and delivered under such FFTS Agreement during the remainder of that defined time period shall be overrun and subject to this provision. In granting requests for Authorized Overrun Service, Natural shall act in a manner consistent with the overrun service priorities set out in the General Terms and Conditions of this Tariff. Charges applicable to either authorized or unauthorized overrun volumes shall be as set out in Section 12 of the General Terms and Conditions of this Tariff.

10. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule FFTS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Original Sheet No. 65 Original Sheet No. 65: Pending

RATE SCHEDULE FTS
FIRM TRANSPORTATION SERVICE

AVATLABILITY

This Rate Schedule FTS is available to any entity (hereinafter called Shipper) which: (i) submits to Natural Gas Pipeline Company of America LLC (hereinafter called Natural) a valid request as defined in Section 3 hereof which Natural has firm capacity available on all affected portions of its system and the firm operational capability to satisfy; and (ii) executes a Firm Transportation Service Agreement (FTS Agreement) with Natural applicable to service under this Rate Schedule FTS. The form of FTS Agreement is contained in this Tariff. There is no limitation on the number of FTS Agreements any one Shipper may have.

2. APPLICABILITY, CHARACTER AND PRIORITY OF SERVICE

- (a) This Rate Schedule FTS defines Natural's firm transportation service. A Shipper entering into an FTS Agreement may elect a Late Notice (LN) option, a System-wide Service (SW) option and/or a No-notice Balancing (NB) option, as described in Section 5 of this Rate Schedule FTS, subject to payment of the rate(s) and compliance with the terms and conditions attendant to such options. This Rate Schedule FTS shall apply to all gas received by Natural from or on behalf of Shipper pursuant to an FTS Agreement. As more fully set out in the General Terms and Conditions of this Tariff, Natural is not providing a supply service under this Rate Schedule FTS.
- (b) Service hereunder shall consist of the acceptance by Natural of natural gas by or on behalf of Shipper for transportation hereunder at the primary Receipt Point(s) specified in the FTS Agreement, or at secondary Receipt Points applicable to the FTS Agreement, the transportation of that natural gas through Natural's system, and the delivery of that natural gas, adjusted for Fuel Gas and Gas Lost and Unaccounted For as set out in the General Terms and Conditions of this Tariff, by Natural to Shipper or for Shipper's

FERC Docket: RP08-319-000

Original Sheet No. 66 Original Sheet No. 66: Pending

RATE SCHEDULE FTS

account at the primary Delivery Point(s) specified in the FTS Agreement, or at secondary Delivery Points applicable to such FTS Agreement. Natural shall not be required: (1) to accept on any day gas tendered, or to deliver on any day gas requested, in excess of the Maximum Daily Quantity (MDQ) specified in the FTS Agreement for each primary Receipt or Delivery Point or for the aggregate of all primary Receipt or Delivery Points (except to the extent Natural has accepted confirmed nominations allowing authorized overrun); (2) to accept or deliver hereunder on any day gas which is not properly nominated pursuant to and to the extent required by the General Terms and Conditions of this Tariff (it being recognized that a limited no-notice service is available under the NB service option); or (3) to receive or deliver gas hereunder on any day at secondary points except as provided in the General Terms and Conditions of this Tariff (it being recognized that a Shipper electing the SW service option will have system-wide secondary point rights). Service hereunder shall be firm except for service at secondary points.

- (c) The service provided under this Rate Schedule FTS shall be performed under Part 284 of the Commission's Regulations. Shipper shall only tender gas for transportation under this Rate Schedule FTS to the extent such service would qualify under the applicable statutes, regulations, Commission orders and the blanket certificate authorizing service by Natural under this Rate Schedule. For service under Subpart B of Part 284 of the Commission's Regulations, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by the Commission's Regulations, Shipper shall (prior to tendering gas under an FTS Agreement) cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit the necessary certification.
- (d) Allocation of capacity, curtailment, priorities of service for the purposes of scheduling and curtailment, and the addition and/or deletion of primary points are all governed by the General Terms and Conditions of this Tariff.
- (e) Shipper may release capacity dedicated to service hereunder pursuant to Natural's Capacity Release Program to the extent permitted by, and subject to the terms and conditions contained in, the General Terms and Conditions of this Tariff. A

FERC Docket: RP08-319-000

Original Sheet No. 67 Original Sheet No. 67: Pending

RATE SCHEDULE FTS

Shipper electing the SW option may not segment a capacity release except within the primary path(s) under the FTS Agreement and neither the releasing nor any replacement Shipper involved in a segmented release will have system-wide secondary point access under the FTS-SW Agreement or any release Agreement originating therefrom after such a release.

3. VALID REQUESTS

- 3.1 A request for service under this Rate Schedule FTS shall be valid as of the date received if it complies with this Section and contains adequate information on all of the items specified in Section 3.2, subject to any necessary verification of such information and to the following:
- (a) A request shall not be valid and Natural shall not be required to grant any such request: (i) for which adequate capacity is not available on any portion of Natural's transmission system (including primary Receipt and primary Delivery Points) necessary to provide such service; (ii) as to which Natural does not have the operational capability to effect receipt, transportation and/or delivery on a firm basis consistent with the terms and conditions of this Rate Schedule FTS; (iii) which, except as provided for in Section 3.1(b) hereof, would require the construction, modification, expansion, or acquisition of any facilities; provided, however, that Natural may agree in its reasonable discretion to construct, modify, expand, or acquire facilities to enable it to perform such services; (iv) unless and until Shipper has provided Natural with the information required in Section 3.2 hereof; (v) if Natural determines, based on the credit analysis referenced in Section 3.2(h), that Shipper does not possess sufficient financial stability to make it reasonably likely the service provided hereunder will be paid for on a timely basis; (vi) if the service requested would not comply with this Rate Schedule FTS; or (vii) if the service requested is at less than the applicable maximum rate; provided, however, that Natural may agree to provide service hereunder at a discount consistent with this Rate Schedule FTS. Natural may also reject any request, or may limit the term of the FTS Agreement granted to Shipper, in the event the service requested would utilize a portion of Natural's system such that a bottleneck would be created that would likely prevent Natural from utilizing upstream or downstream portions of its

FERC Docket: RP08-319-000

Original Sheet No. 68 Original Sheet No. 68: Pending

RATE SCHEDULE FTS

system and would likely result in Natural being unable to provide substantial firm services for others utilizing such upstream or downstream portions of the system. Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to this Section and while an FTS Agreement is in effect. Such matters are governed by the General Terms and Conditions of this Tariff.

- (b) Natural agrees to construct minor meter and tap facilities to effectuate service requested under this Rate Schedule FTS provided that: (i) the cost of such facilities does not exceed \$100,000.00; (ii) Natural shall be reimbursed prior to commencing construction; and (iii) the facilities will not adversely affect Natural's operations.
- (c) In the event that Natural agrees to construct, modify, expand or acquire facilities to perform service under this Rate Schedule FTS, Natural shall maintain a separate record of the nature and costs of such facilities.
- (d) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request, in whole or in part, due to lack of capacity or system capability or if the request is incomplete or does not comply with this Rate Schedule FTS. Any request shall be null and void unless it is substantially complete and complies with this Rate Schedule. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the FTS Agreement, after which Shipper shall have fifteen (15) days to provide the specified information. In the event such information is not received within fifteen (15) days, Shipper's request shall be null and void.
- (e) Natural shall tender an FTS Agreement to Shipper for execution when Shipper's request for service is accepted. Unless waived by Natural, a request for service shall be invalid if Shipper fails to execute an FTS Agreement hereunder within ten (10) days after an FTS Agreement which accurately reflects the Shipper's firm request has been tendered by Natural for execution.

FERC Docket: RP08-319-000

Original Sheet No. 69 Original Sheet No. 69: Pending

RATE SCHEDULE FTS

3.2 Requests for service hereunder shall be deemed valid only after the information specified in this Section is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department, One Allen Center, 500 Dallas Street, Houston, Texas 77002, or Telecopy Number (713) 369-9305. Qualifying bids in an auction shall be treated as a valid request provided that the bid is supplemented with all information required under this Section 3.2 as required by the General Terms and Conditions of this Tariff. The information required for a valid request shall be as follows:

(a) GAS QUANTITIES

The request shall specify in Dth the aggregate MDQ and the MDQ for each primary point, exclusive of applicable Fuel Gas and Gas Lost and Unaccounted For, provided that Natural shall not be obligated to accept requests for an aggregate MDQ of less than one hundred (100) Dth per day. If Shipper is willing to accept a lesser MDQ, Shipper shall also specify the minimum acceptable MDQ.

(b) RECEIPT POINT(S)

The request shall specify the primary point(s) at which Shipper desires Natural to receive gas. By utilizing the comment section, the request may also specify a pipeline segment if the Shipper is willing to accept any point in that segment or the request may specify points which Shipper is willing to accept, and/or which points Shipper is not willing to accept, within a pipeline segment.

(c) DELIVERY POINT(S)

The request shall specify the primary point(s) at which Shipper desires Natural to deliver gas. By utilizing the comment section, the request may also specify a pipeline segment if the Shipper is willing to accept any point in that segment or the request may specify points which Shipper is willing to accept, and/or which points Shipper is not willing to accept, within a pipeline segment.

(d) LIMITATION OF POINTS

(1) A Shipper may request any number of primary Receipt and primary Delivery Points so long as the summation of MDQs at all primary Receipt Points and at all primary Delivery Points equals the aggregate MDQ.

FERC Docket: RP08-319-000

Original Sheet No. 70 Original Sheet No. 70: Pending

RATE SCHEDULE FTS

(2) The availability to Shipper of secondary points, and the related priorities and volumes, shall be governed by the General Terms and Conditions of this Tariff. System-wide service on a secondary point basis will be available hereunder to a Shipper which elects the SW service option.

(e) SERVICE OPTIONS

 $$\operatorname{\textsc{The}}$ request shall specify whether Shipper elects any or all of the LN, SW or NB service options.

(f) TERM OF SERVICE

The request shall specify:

(1) The date service is requested to commence;

and

(2) The date service is requested to terminate.

(g) REQUESTED RATE

The request shall state the rate at which the Shipper is willing to contract for the service requested. If no rate is stated, Shipper will be deemed to have agreed to the maximum rate.

(h) CREDIT

 $\label{eq:continuous} Acceptance \ \mbox{of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.}$

(i) COMPLIANCE WITH FTS TARIFF

Submission of a request for service hereunder shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule FTS, including the applicable General Terms and Conditions. This agreement extends to the terms and conditions applicable to any service option (NB, SW and/or LN) elected by a Shipper under subsection (e) of this Section 3.2 and to an NB and/or LN service option elected during the term of the FTS Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 71 Original Sheet No. 71: Pending

RATE SCHEDULE FTS

(j) COMMISSION-REQUIRED FILING INFORMATION

The following information is to be provided at the time a request for service hereunder is submitted, if available, or when an initial nomination for service under an executed FTS Agreement is submitted, and when any subsequent changes occur:

- (1) Affiliation of the Shipper with Natural; and
- (2) The identity of the Shipper, including whether it is a local distribution company, an interstate pipeline company, an intrastate pipeline company, an end user, a producer, or a marketer.

4. TERM

- (a) The term of service hereunder shall be set forth in the FTS Agreement between Shipper and Natural.
- (b) The General Terms and Conditions of this Tariff shall govern the applicability of, and the terms and conditions relating to, rollovers and the right of first refusal vis a vis an FTS Agreement. Upon termination of any FTS Agreement, and subject to any such rollover or right of first refusal, service by Natural to Shipper thereunder shall be terminated and automatically abandoned.
- (c) Natural may terminate any FTS Agreement if Natural is required by the FERC or some other agency or court to provide firm service for others utilizing the system capacity or capability required for service under such FTS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the type covered by the FTS Agreement; provided, however, that nothing herein shall be read to protect Natural from the consequences of its own negligence.

FERC Docket: RP08-319-000

Original Sheet No. 72 original Sheet No. 72: Pending

RATE SCHEDULE FTS

5. SERVICE OPTIONS

- 5.1 (a) A Shipper electing the LN option shall have the right to submit late nominations and Intra-day Nominations consistent with the General Terms and Conditions of this Tariff.
- (b) The LN option may be requested as part of the request for service under Section 3 of this Rate Schedule or may be requested at any time during the term of an FTS Agreement; provided, however, that the election of the LN service option must be for a minimum of one year or the remaining term of the FTS Agreement, whichever is less; and provided further that the LN option must be effective on the first day of a month unless otherwise mutually agreed.
- (c) The maximum Reservation Rate applicable to an FTS Agreement with an LN option shall be increased by the LN Surcharge as set out in Section 6.2(a)(2) of this Rate Schedule FTS.
- (d) Natural shall post on the Informational Postings portion of its Interactive Website each month the availability of system capability, if any, to render additional LN service in excess of then existing commitments.
- $5.2\,$ A Shipper electing the SW option shall have the right on a secondary point basis to all Receipt and Delivery Points on Natural's system; provided, however, that a Shipper under an FTS Agreement paying the highest maximum Reservation Charge which could apply under Natural's effective rate matrix to any haul on Natural's system shall automatically have such rights. Except for points which are within a path defined by primary points under the FTS Agreement, all such secondary points shall be treated as having an out-of-path priority for scheduling and curtailment. Such out-of-path secondary points shall be on the same basis for scheduling and curtailment as out-ofpath secondary points under any other FTS Agreement, including those within a zone transversed by such Agreement and those arising out of opposite leg secondary point rights. The SW option may be requested as part of the request for service under Section 3 of this Rate Schedule FTS. For existing Agreements under Rate Schedule FTS which do not have the SW option, the Shipper may request the addition of the SW option at any time, provided that the rate for firm transportation

FERC Docket: RP08-319-000

Original Sheet No. 73 Original Sheet No. 73: Pending

RATE SCHEDULE FTS

service with the SW option shall be the applicable maximum rate specified for the SW option in Section 6.2(a)(3) of this Rate Schedule FTS unless Natural and the Shipper agree to a different rate at the time the option is requested and provided further that nothing shall be read to override or to modify in any way: (i) any obligation of Natural under the Natural Gas Act to treat similarly situated customers the same with respect to rates and discounting; (ii) any tariff provision relating to rates or discounting which results from Commission action on Natural's Order No. 637 compliance plan in Docket No. RP00-409; or (iii) any other provision of Natural's Tariff governing or relating to rates or discounts. For an FTS Agreement with an SW option, the applicable maximum Reservation Rate shall be determined by the SW option formula set out in Section 6.2(a)(3) of this Rate Schedule FTS.

5.3 (a) Shippers selecting the NB option may, subject to the limitations in subsection (d), inject or withdraw gas on a nonotice basis (within the limits set out herein) under Natural's

FERC Docket: RP08-319-000

Original Sheet No. 74 original Sheet No. 74: Pending

RATE SCHEDULE FTS

Rate Schedule NSS or from a Third Party Storage Point as defined in Section 9 of the General Terms and Conditions of this Tariff. The NB option shall be available on a non-discriminatory basis for use in conjunction with either Rate Schedule NSS or a Third Party Storage Point which meets the criteria set out in subsection (d)(3).

- (b) Injections may be made on a firm, no-notice basis on any day of the year to the extent Natural can effectuate such injections employing the capacity dedicated to the FTS-NB Agreement not otherwise utilized on that day and the storage rights under the NSS Agreement(s) and/or third party storage associated with such FTS-NB Agreement not otherwise utilized on that day. A Shipper will be able to utilize rights under the NB option on any day to the extent it has adequate storage and/or transportation capacity, as applicable, under its FTS-NB Agreement(s) and the related NSS Agreement (or third party storage) comprising that Shipper's NB option, which rights the Shipper is not otherwise utilizing on that day. It shall be Shipper's responsibility to assure that Natural is not called upon to perform no-notice service hereunder in excess of that level. Shipper may be subject to Overrun and Balancing Service Charges to the extent no-notice volumes exceed its available rights.
- (c) Withdrawals may be made on a firm, no-notice basis on any day during the period of April 1 through October 31 to the extent Natural can effectuate such withdrawals employing the capacity dedicated to the FTS-NB Agreement not otherwise utilized on that day and the storage rights under the NSS Agreement(s) and/or third party storage associated with such FTS-NB Agreement not otherwise utilized on that day. It shall be Shipper's responsibility to assure that Natural is not called upon to perform no-notice service hereunder in excess of that level. Shipper may be subject to Overrun and Balancing Service Charges to the extent no-notice volumes exceed its available rights.
- (d) The applicable maximum rate for Shippers electing the NB option shall not be higher than for other Shippers under Rate Schedule FTS, but the following special conditions shall apply:
- (1) The NB option may be requested as part of the request for service under Section 3 of this Rate Schedule FTS or by mutual agreement (Natural's agreement shall not be unreasonably withheld and any decision to withhold must be based on operating reasons); provided, however, that the election of the NB option must be for a minimum of one year or the remaining term of the FTS Agreement, whichever is less; and provided further, that the NB option must be effective on the first day of a month unless otherwise mutually agreed. Natural shall provide a written explanation of the operational basis for rejecting any request to add the NB option to an existing Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 75 original Sheet No. 75: Pending

RATE SCHEDULE FTS

(2) The primary path under the FTS Agreement must be such that the NSS storage point would be treated as in-the-path under Section 5.6 of the General Terms and Conditions of this Tariff or, where a Third Party Storage Point is being utilized, such point must be within the primary path under the FTS-NB Agreement.

- (3) Any Third Party Storage Point shall be available for service under this option only to the extent the operator of that point and Natural enter into a Storage Point Swing Service Operating (SPSSO) agreement which defines how such operator will accommodate Shipper swing volumes on a no-notice basis, how the operator is to make the corresponding operational physical changes, the limitations on the level of swings to be accommodated and the consequences if such levels are exceeded or operational changes are not made. It is the intent that such an agreement provide Natural with the ability to call upon such third party storage (within the limits defined in the agreement) reasonably comparable to that which Natural exercises with respect to its own storage fields, which may include short-term notice arrangements with the operator. The SPSSO agreement must specify a predetermined allocation methodology and shall specify the extent to which and the conditions under which Shippers shall be kept whole because the operator is agreeing to take the swing. Because the third party storage provider under an SPSSO agreement will be providing swing service, Natural shall not be responsible (as it might under an ordinary operational balancing agreement) for swing within the agreed limits applicable to such Third Party Storage Point. If Natural rejects any request that NB service be provided utilizing a Third Party Storage Point, Natural shall make such rejection in writing and set out the reasons for the rejection.
- (4) The Shipper electing an NB option must designate both (i) which NSS Agreement or Third Party Storage Point is to be associated with the FTS-NB Agreement in providing no-notice service (third party storage requires the consent of the operator of such storage) and (ii) which points are to be covered by no-notice service under the FTS-NB Agreement (such point designation shall require the concurrence of the point operator to the extent required by any allocation agreement at the point). The point designation in the prior sentence may be to a CDP (if it is to a point within a CDP, such CDP may need to be subdivided for administrative purposes) or to a specific point on which telemetry exists, or which is otherwise mutually agreed, covered by the FTS-NB Agreement. Any number of FTS Agreements and/or FFTS Agreements may be associated with any NSS Agreement or Third Party Storage Point, but an FTS Agreement may not be

FERC Docket: RP08-319-000

Original Sheet No. 76 original Sheet No. 76: Pending

RATE SCHEDULE FTS

associated with more than one NSS Agreement or Third Party Storage Point; provided, however, that under a predetermination allocation agreement pursuant to Section 11 of the General Terms and Conditions of this Tariff, a Shipper or point operator can designate a linkage between an FTS Agreement and multiple NSS Agreements and Third Party Storage Points so long as an administratively workable protocol for linking the transportation and storage components is specified. The designations hereunder may be changed on reasonable prior notice to Natural.

- $\,$ (5) The NB option is available for use only in conjunction with an NSS Agreement or comparable third party storage agreement at a Third Party Storage Point. The NB option shall terminate if neither of such elements is present.
- (e) No-notice activity under an FTS-NB Agreement will be posted to the storage point in the receipt zone where the imbalance occurs or to the closest storage point where Shipper has adequate storage rights to accommodate such activity, consistent with Section 5.6(c)(4) of the General Terms and Conditions of this Tariff.
- 5.4 A Shipper under Rate Schedule FTS is not obligated to choose any of the options described in this Section 5, but may select any one of them, any two of them or all three of them. Once a Shipper has elected an option and so long as it remains in effect, the Agreement for service to such Shipper may be referenced by placing the relevant letters after FTS.

6. RATE

- $6.1\,$ Shipper shall pay Natural each month for service under this Rate Schedule FTS the sum of the following charges:
 - (a) a Reservation Charge based on Shipper's MDQ;
- (b) a Commodity Charge for each unit of gas delivered hereunder by Natural during the billing month;
 - (c) Gathering and Offshore Charge(s);
 - (d) all applicable types of transition cost charges; and
- (e) any other applicable charges and penalties set out in this Rate Schedule FTS or the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 77 Original Sheet No. 77: Pending

RATE SCHEDULE FTS

Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.

FERC Docket: RP08-319-000

Original Sheet No. 78 original Sheet No. 78: Pending

RATE SCHEDULE FTS

- 6.2 The charges identified in items (a)-(d) of Section 6.1 shall be assessed as follows:
- (1) The maximum Reservation Charge for service under (a) this Rate Schedule FTS shall be the applicable maximum unit Reservation Rate(s) set out in this Tariff multiplied by Shipper's MDQ. The Reservation Charge payable by any Shipper shall be the maximum Reservation Charge unless a lower Reservation Charge is specified in the FTS Agreement or a separate discount agreement or is assessed pursuant to Section 6.8 hereof, which lower charge must be consistent with the minimum Reservation Rate(s) specified in this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma Agreement set out in this Tariff, discounts may be limited to specific volumes. Shipper shall be obligated to pay the Reservation Charge hereunder beginning as of the date FTS service commences. FTS service shall be deemed to commence on the later of the date specified in the FTS Agreement or the date Natural is able to render firm service hereunder to Shipper.
- (2) For Shippers electing the LN option, the applicable maximum Reservation Rate shall be increased by the LN Surcharge as set out in this Tariff.
- (3) For Shippers electing the SW option, the applicable maximum Reservation Rate shall be determined in accordance with the following formula:

where:

RR = applicable maximum Reservation Rate

PPR = the applicable maximum Reservation Rate for the weighted average of all primary paths under the Agreement

If applicable, the LN Surcharge shall be added to the rate so determined.

FERC Docket: RP08-319-000

Original Sheet No. 79 Original Sheet No. 79: Pending

RATE SCHEDULE FTS

- (b) The maximum monthly Commodity Charge shall be the applicable maximum unit Commodity Rate(s) set out in this Tariff multiplied by the quantity of gas delivered by Natural to Shipper hereunder during the billing month plus an amount equal to the volumes attributable to Fuel Gas and Gas Lost and Unaccounted For multiplied by one-half of the applicable maximum unit Commodity Rate(s). The Commodity Charge payable by any Shipper shall be the maximum Commodity Charge unless a lower Commodity Charge is specified in the FTS Agreement or a separate discount agreement or is assessed pursuant to Section 6.7 hereof, which lower charge must be consistent with the minimum Commodity Rate(s) specified in this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma Agreement set out in the Tariff, discounts may be limited to specific volumes. For any volumes delivered in excess of Shipper's MDQ, which deliveries shall be subject to Section 9 of this Rate Schedule FTS, Shipper shall pay the applicable Overrun Charge set out in Section 12 of the General Terms and Conditions of this Tariff plus the sum of the applicable transportation Commodity Rate(s) multiplied by the overrun volumes and an amount equal to the volumes attributable to Fuel Gas and Gas Lost and Unaccounted For multiplied by one-half of the applicable Commodity Rate(s).
- (c) The Reservation and Commodity Charges will be based on the primary Receipt and Delivery Points specified in the FTS Agreement; provided, however, that for a Shipper electing the SW option utilizing a point outside of its primary path, the Commodity Charge shall be based on the actual path utilized. Unless Natural agrees to the contrary, the applicable Reservation Charges will not be reduced during the primary term of the FTS Agreement due to a change in the primary points; provided, however, that Commodity Charges will be adjusted to reflect a change in a primary point which results in a shortened path as provided in Section 5.1(c)(5) of the General Terms and Conditions of this Tariff.
- (d) Separate Gathering Charge(s) shall apply to all volumes received by Natural under this Rate Schedule at any Receipt Point(s) on its gathering facilities. Similarly, separate Offshore Charges shall apply to all volumes received by Natural under this Rate Schedule at any Receipt Point(s) on any non-contiguous offshore lateral.
- (e) Transition cost charges shall be assessed in accordance with the General Terms and Conditions of this Tariff.
- 6.3 OPTIONAL RATE FORM INCLUSIVE OF RESERVATION SURCHARGES. If a Shipper desires to have a reservation rate which is inclusive of reservation surcharges, the form of Inclusive Reservation rate included in the Agreement shall, in lieu of that set out in Sections 6.1 and 6.2 above, be as set out in this Section.

FERC Docket: RP08-319-000

Original Sheet No. 80 Original Sheet No. 80: Pending

RATE SCHEDULE FTS

(a) The Inclusive Reservation rate hereunder shall be a rate which includes all applicable base reservation rates and the reservation surcharges under Section 21 of the General Terms and Conditions (858 Surcharges). The Inclusive Reservation rate may not exceed the applicable maximum Inclusive Reservation rate and may not be less than the applicable minimum Inclusive Reservation rate under this provision; provided, however, that nothing herein shall preclude a negotiated rate at a different level under Section 49 of the General Terms and Conditions of this Tariff.

- (b) The applicable maximum Inclusive Reservation rate for any Agreement shall be the sum of the applicable maximum base reservation rate and 858 Surcharge rate. If a Shipper is exempt from 858 Surcharges, the applicable maximum Inclusive Reservation rate for that Shipper shall be reduced accordingly. If the applicable maximum rate for any of these components changes, the applicable maximum Inclusive Reservation rate shall change by recalculation of the sum of these components.
- (c) In the event that Shipper and Natural agree on a discounted Inclusive Reservation rate, the Inclusive Reservation rate shall be deemed to consist of the applicable maximum 858 Surcharge with the discounted base reservation rate being calculated from the Inclusive Reservation rate after subtraction of those components. If the Inclusive Reservation rate is less than the applicable maximum rate for the 858 Surcharge component, the discount shall be deemed to apply first to the reservation rate and then to the 858 Surcharge. If the level of the applicable maximum rate for any rate component changes, the level of the various discounted component(s) shall be recalculated. The discounted Inclusive Reservation rate may not be less than the sum of the applicable minimum Reservation rates for these rate components.
- (d) The Inclusive Reservation rate hereunder shall be billed as a monthly demand rate applied to the MDQ of the Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 81 original Sheet No. 81: Pending

RATE SCHEDULE FTS

(e) If an Agreement includes a rate form which conforms to this Section and does not contain any other negotiated rate elements, then the filing requirements under Section 49 of the General Terms and Conditions of this Tariff shall not apply to such Agreement.

(f) The Inclusive Reservation rate shall not include commodity rate(s), commodity surcharges, ACA, fuel or overrun charges, which are determined and assessed as applicable under other provisions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 82 original Sheet No. 82: Pending

RATE SCHEDULE FTS

6.4 (a) Shipper shall reimburse Natural within five (5) business days after costs have been incurred by Natural for all fees required by the FERC or any regulatory body including, but not limited to, filing, reporting, and application fees to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees (such as general rate case filing fees).

- (b) If Natural constructs, acquires or modifies any facilities to perform service hereunder, then as specified in an agreement between the parties either:
- (i) Shipper shall reimburse Natural for the cost of such facilities or facility modifications as described in the General Terms and Conditions of this Tariff; or
- $\mbox{(ii)}$ Natural shall assess a monthly charge reflecting such facility costs.
- 6.5 The ACA and RD&D charges will be assessed, when applicable, as provided in the General Terms and Conditions of this Tariff, on volumes delivered by Natural to Shipper under this Rate Schedule FTS. Shipper shall also reimburse Natural for Fuel Gas and for Gas Lost and Unaccounted For hereunder as provided by the General Terms and Conditions of this Tariff.
- 6.6 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in (1) the rates and charges applicable under this Rate Schedule FTS, including both the level and design of such rates and charges; or (2) the terms and conditions of this Rate Schedule FTS. Natural agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.
- (b) If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the FTS Agreement or a separate discount agreement, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such highest rate, Shipper shall execute or provide such documentation within fifteen (15) calendar days after a written request by Natural. If, at any time and from time to time, the FERC or any

FERC Docket: RP08-319-000

Original Sheet No. 83 Original Sheet No. 83: Pending

RATE SCHEDULE FTS

other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for transportation service hereunder, the rate shall be decreased to such reduced rate.

- 6.7 Shipper may request that Natural process its gas in order that such gas will meet the quality specifications of the General Terms and Conditions of this Tariff. Natural may, but is not obligated to, provide such service. The providing of such service and the terms and conditions therefore shall be negotiated between Natural and Shipper and the terms and conditions agreed upon shall be specified in the FTS Agreement or a separate treating/processing agreement. For any such service, Natural shall assess Shipper a charge reflecting all the costs of providing the service, including a Reservation as well as Commodity Charge where appropriate.
- 6.8 Natural may from time to time and at any time, upon twenty-four (24) hours verbal or written notice, subject to any provisions on discounting in the FTS Agreement or in a separate discount agreement, charge any individual Shipper for service under this Rate Schedule FTS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate may not be less than the applicable minimum rate for service under Rate Schedule FTS set forth in this Tariff. Natural will confirm any verbal notice of the applicable charge in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Unless otherwise agreed in the FTS Agreement or in a separate discount agreement, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the FTS Agreement and/or any discount agreement) upon twenty-four (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with the Commission any and all reports as required by the Commission's Regulations with respect to the institution or discontinuance of any discount.
- 6.9 All revenues and gas in kind collected by Natural as a result of providing service under this Rate Schedule FTS shall be retained by Natural unless Natural has otherwise explicitly agreed on a different disposition of such amounts. Where crediting or

FERC Docket: RP08-319-000

Original Sheet No. 84 Original Sheet No. 84: Pending

RATE SCHEDULE FTS

refund mechanisms apply under other provision(s) of this Tariff or pursuant to effective Commission orders or settlements, such mechanisms shall supersede this Section to the extent necessary to carry out such provision(s).

- 6.10 (a) Natural may, but is not obligated to, transport liquids for Shipper. Any transportation of liquids on behalf of Shipper shall be pursuant to a separate agreement, which agreement shall set out the negotiated rate for such transportation and shall supersede subsection (b) of this Section; provided, however, that if Natural does provide any such transportation in the absence of a separate agreement, subsection (b) of this Section shall apply. The credit evaluation provisions of the General Terms and Conditions of this Tariff shall be applicable to such transportation unless otherwise specified in a separate agreement.
- (b) (1) Absent a separate agreement (in which case the negotiated rate set out in such agreement shall govern), Natural shall charge Shipper and Shipper shall pay Natural for Natural's transportation and handling of Shipper's liquids a monthly fee equal to \$1.025 per barrel of liquids.
- (2) If, at any time and from time to time, as a result of any legislation or any order, opinion, or decision of the Commission or of any other governmental authority having jurisdiction over Natural's rates (including any order, opinion, or decision based in whole or in part on a contested or uncontested settlement agreement relative to Natural's rates) an amount is allocated or attributed to the transportation and handling of liquids, liquefiables, plant fuel, vaporous nonhydrocarbons and/or similar materials, and such amount is credited to Natural's cost of service or otherwise excluded in determining Natural's rates, Shipper shall reimburse Natural for the sum of money (including the interest reflected therein) Natural is required to refund or credit, or has refunded or credited to its customers which is attributed by Natural to such transportation and handling for Shipper under the FTS Agreement and which has not been assessed and collected from Shipper for the time period to which the refund or credit applied. If such refunded or credited money is less than the sum of money assessed and collected from Shipper for the applicable time period, Natural shall remit the difference to Shipper within thirty (30) days of the applicable order, opinion, or decision. Thereafter, Shipper shall pay Natural monthly fees based on amounts which are

FERC Docket: RP08-319-000

Original Sheet No. 85 Original Sheet No. 85: Pending

RATE SCHEDULE FTS

attributable by Natural to such transportation and handling for Shipper under the FTS Agreement as calculated and assessed by Natural hereunder. Such monthly fees shall be calculated so as to permit Natural to generate monthly revenues therefrom to equal the amounts expected to be credited to Natural's cost of service or excluded from the determination of Natural's rates.

- (3) The amounts attributed by Natural to such transportation and handling for Shipper under the FTS Agreement shall be established from time to time above based upon the application of a methodology determined by Natural in good faith to be consistent with applicable orders, opinions, or decisions pertaining to such transportation and handling.
- (4) Any settlement, compromise, or other action of Natural in connection with any such order, opinion, or decision pertaining to transportation of liquids, liquefiables, plant fuel, vaporous non-hydrocarbons and/or other similar materials shall be conclusive and binding upon Shipper.
- 6.11 Natural shall perform exchanges and backhauls hereunder to the extent firm capacity is available. Exchanges and backhauls shall be subject to the maximum and minimum rates under this Rate Schedule except that no fuel need be charged.
- 7. NOMINATIONS, BALANCING SERVICE CHARGES AND IMBALANCES
 - (a) Except as set out in subsection (c), Shipper shall provide Natural with nominations of receipts and deliveries by Receipt and Delivery Point, including pooling points where applicable, in accordance with the General Terms and Conditions of this Tariff. Shippers electing the LN option may submit late and Intra-day Nominations consistent with the General Terms and Conditions of this Tariff. It shall be Shipper's responsibility to cause gas to be delivered to Natural at Receipt Point(s), and to cause gas to be taken from Natural at Delivery Point(s), in accordance with the information supplied to Natural. In the event that actual receipts or deliveries at any point fail to conform to Shipper's nominations, and Shipper has not elected the NB option and/or the discrepancy is greater than can be accommodated pursuant to Shipper's applicable no-notice rights, Balancing Service Charges will apply to the extent provided in the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 86 original Sheet No. 86: Pending

RATE SCHEDULE FTS

(b) It shall be Shipper's responsibility to keep receipts and deliveries in balance. Shipper must submit nominations to rectify an imbalance to the extent required by Section 8.4 of the General Terms and Conditions of this Tariff. Any imbalance between actual receipts and actual deliveries shall be eliminated by cashout on a monthly basis in accordance with the General Terms and Conditions of this Tariff.

- (c) To the extent a Shipper elects the NB option, and within the limitations of its no-notice rights thereunder as defined in Section 5 of this Rate Schedule FTS, Shipper shall not be obligated to submit nominations to effectuate injections or withdrawals (withdrawals are available on a no-notice basis only from April 1 through October 31) in conjunction with service under an FTS Agreement.
- 8. RECEIPT AND DELIVERY POINTS AND UPSTREAM AND DOWNSTREAM ARRANGEMENTS
 - The primary point(s) of receipt for all gas tendered to Natural for transportation hereunder and the primary point(s) of delivery for all gas delivered by Natural to Shipper (or to a third-party on behalf of Shipper) hereunder shall be specified in the FTS Agreement. For each individual primary Receipt and primary Delivery Point, and for the aggregate of all such points, Natural's maximum obligation to accept and deliver gas on a firm basis shall be specified in Dth in the FTS Agreement. The sum of the MDQs for primary Receipt Points and the sum of the MDQs for primary Delivery Points shall not exceed the aggregate MDQ. All stated MDQ shall be exclusive of applicable Fuel Gas and Gas Lost and Unaccounted For. The availability of secondary Receipt and Delivery Points, and the related volumes and priorities at such points, shall be governed by the General Terms and Conditions of this Tariff, it being recognized that a Shipper electing the SW service option shall have system-wide secondary point rights as set out in Section 5 of this Rate Schedule FTS.
 - (b) Conditions of delivery at Receipt and Delivery Points are set out in the General Terms and Conditions of this Tariff.
 - (c) Shipper shall make all necessary arrangements with other parties (1) at or upstream of the Receipt Point(s) where gas is tendered to Natural hereunder and (2) at or downstream of the Delivery Point(s) where Natural delivers gas hereunder to or for the account of Shipper. Such arrangements must be consistent with this Rate Schedule FTS and must be coordinated with Natural.

FERC Docket: RP08-319-000

Original Sheet No. 87 Original Sheet No. 87: Pending

RATE SCHEDULE FTS

9. OVERRUN SERVICE

Upon request of Shipper, Natural may (but is not obligated to) receive, transport, and deliver on any day quantities of natural gas in excess of Shipper's MDQ under the FTS Agreement when, in Natural's reasonable judgment, the capacity and operating capability of its system will permit such receipt, transportation and delivery without impairing the ability of Natural to meet its other obligations. In granting requests for Authorized Overrun Service, Natural shall act in a manner consistent with the overrun service priorities set out in the General Terms and Conditions of this Tariff. Charges applicable to either authorized or unauthorized overrun volumes shall be as set out in Section 12 of the General Terms and Conditions of this Tariff.

10. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule FTS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Original Sheet No. 88 Original Sheet No. 88: Pending

RATE SCHEDULE FRSS FIRM REVERSE STORAGE SERVICE

AVAILABILITY

- (a) Under this Rate Schedule FRSS, Natural provides firm reverse storage service to any entity (hereinafter called Shipper) which: (1) submits to Natural a valid request as defined in Section 3 hereof; and (2) executes a Firm Reverse Storage Service Agreement (FRSS Agreement) with Natural applicable to service under this Rate Schedule FRSS. The form of FRSS Agreement is contained in this Tariff. There is no limitation on the number of FRSS Agreements any one Shipper may have.
- (b) Shipper shall arrange separately for the transportation to inject gas into storage in connection with service hereunder, as further provided in Section $8\,(b)$ of this Rate Schedule FRSS.
- (c) A Shipper may elect delivered storage service or a storage-only option. A Shipper selecting the storage-only option hereunder shall arrange separately for transportation service upon withdrawal of gas from storage.

2. APPLICABILITY, CHARACTER AND PRIORITY OF SERVICE

This Rate Schedule FRSS defines a firm reverse storage service. Unless Shipper elects the storage-only option, service hereunder encompasses the delivery of gas withdrawn from storage to the Delivery Point(s) in the FRSS Agreement. The Shipper under an FRSS Agreement has firm summer withdrawal rights (during the Withdrawal Period defined in Section 5.1 hereof), followed by firm winter injection rights which must be used in satisfying Shipper's obligation to replace the gas withdrawn the prior summer (such firm injections to be made during the Injection Period defined in Section 5.2 of this Rate Schedule FRSS). Under this Rate Schedule FRSS, gas must be withdrawn before Shipper can tender injections. Firm withdrawal rights are available on a no-notice basis within the rights defined in Sections 5 and 7 of this Rate Schedule FRSS. Shipper must separately arrange for transportation to effectuate injections and must nominate injections. A Shipper selecting the storage-only option must separately arrange for transportation service to effectuate withdrawals, and must nominate such withdrawals except to the extent Shipper has no-notice rights under Section 7.5(b) of this Rate Schedule. This Rate Schedule FRSS shall apply to all gas withdrawn by Natural and delivered to or for the account of Shipper pursuant to an FRSS Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 89 Original Sheet No. 89: Pending

RATE SCHEDULE FRSS

(b) Service hereunder shall consist of the withdrawal of gas from storage by Natural and (unless Shipper selects the storage-only option hereunder) the delivery of such gas to Shipper or for Shipper's account during the Withdrawal Period at the primary Delivery Point(s) specified in the FRSS Agreement, or at secondary Delivery Point(s) applicable to such FRSS Agreement. If a Shipper elects the storage-only option, the paper storage point designated under Section 7.4(a)(2) hereof shall be both the Receipt Point and the Delivery Point under the FRSS Agreement. Service hereunder shall further consist of the acceptance by Natural during the Injection Period of gas from or for the account of Shipper for injection into storage to replace the gas withdrawn the prior Withdrawal Period. During the Injection Period, Shipper must replace all gas withdrawn over the prior Withdrawal Period. Fuel Gas, Gas Lost and Unaccounted For and Storage Fuel shall be provided by Shipper as set out in Sections 6.3(c), 6.3(d) and 7.2 of this Rate Schedule FRSS.

FERC Docket: RP08-319-000

Original Sheet No. 90 Original Sheet No. 90: Pending

RATE SCHEDULE FRSS

(c) Natural shall not be required, except to the extent it has accepted confirmed nominations allowing authorized overrun, to: (1) withdraw hereunder on any day gas requested in excess of Shipper's Maximum Daily Withdrawal Quantity (WQ) or in excess of Shipper's Maximum Seasonal Withdrawal Quantity (MSWQ); (2) accept on any day gas tendered for injection hereunder in excess of Shipper's Maximum Daily Injection Quantity (IQ); or (3) deliver gas at Delivery Points or accept gas at Receipt Points on a basis which is not consistent with Section 7 of this Rate Schedule FRSS and the related General Terms and Conditions of this Tariff. The terms "IQ", "WQ" and "MSWQ" are defined in Section 5 of this Rate Schedule FRSS. Service hereunder shall be firm within WQ, MSWQ and IQ. Gas may not be injected hereunder during the Withdrawal Period, nor may gas be withdrawn during the Injection Period, as those terms are defined in Section 5 of this Rate Schedule FRSS. Volumes of gas withdrawn during the Withdrawal Period must be replaced by equivalent volumes of gas injected by the end of the Injection Period.

- (d) The service provided under this Rate Schedule FRSS shall be performed under Part 284 of the Commission's Regulations. Shipper shall only take or tender gas under this Rate Schedule FRSS to the extent such service would qualify under the applicable statutes, regulations, Commission orders and the blanket certificate authorizing service by Natural under this Rate Schedule FRSS. For service under Subpart B of Part 284 of the Commission's Regulations, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by the Commission's Regulations, Shipper shall (prior to using service under a FRSS Agreement) cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit the necessary certification to Natural.
- (e) As more fully set out in the General Terms and Conditions of this Tariff, Natural is not providing a gas supply service under this Rate Schedule FRSS. This service will be provided through the management by Natural of its cushion gas and storage operations at all of its storage fields and related facilities.
- (f) Allocation of capacity, curtailment, and priorities of service for the purposes of scheduling and curtailment shall all be governed by Section 7 of this Rate Schedule FRSS and the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 91 Original Sheet No. 91: Pending

RATE SCHEDULE FRSS

- (g) (1) A Shipper may release all or a portion of its capacity under an FRSS Agreement pursuant to Natural's Capacity Release Program to the extent permitted by, and subject to the terms and conditions contained in, this subsection and Section 19 of the General Terms and Conditions of this Tariff. Storage and transportation rights under this Rate Schedule FRSS may be released separately (except for a Shipper selecting the storage-only option, which has no transportation service to release), provided that the separately released transportation capacity may not be segmented.
- (2) A Shipper under Rate Schedule FRSS may only release its remaining rights; i.e., its rights not previously utilized and not already released to others. Once capacity has been released, the remaining rights of the releasing Shipper shall be reduced by a volume equal to the volume of the rights released. No release may result in any Shipper or in Shippers in the aggregate (releasing Shipper together with all Replacement Shippers) having rights in excess of those available under this Tariff and the FRSS Agreement to the releasing Shipper prior to any release.
- (3) The FRSS Agreement shall specify a transportation path for purposes of the separate release of transportation (except for a Shipper selecting the storage-only option, which has no transportation service to release). The replacement Shipper must nominate such separately released transportation in the same manner as under Rate Schedule FTS (but such Shipper shall have only the rights under Rate Schedule FRSS and not any additional rights under Rate Schedule FTS). All separate releases of transportation rights in the aggregate are limited to the aggregate rights of the Shipper under the FRSS Agreement (not more than 10 or 20 times MDQ over any Withdrawal Period, as applicable) to which the releases relate. If transportation rights are released separately, the releasing Shipper's rights to have gas delivered hereunder shall be reduced accordingly.
- (4) The Shipper may release storage rights separately and, if transportation rights embedded in delivered storage service are released separately, the releasing Shipper can utilize or release the residual storage rights, but the transportation to take such gas away from the storage point on withdrawal (as well as transportation to the storage point on injection) must be separately contracted. All separate releases of storage rights in the aggregate are limited to the aggregate storage rights of the Shipper under the FRSS Agreement (not more than 10 or 20 times MDQ over any Withdrawal Period, as applicable) to which the releases relate. Storage released separately must be nominated in the same manner as under Rate Schedule NSS (but the replacement Shipper shall have only the rights under Rate Schedule FRSS and not any additional rights under

FERC Docket: RP08-319-000

Original Sheet No. 92 Original Sheet No. 92: Pending

RATE SCHEDULE FRSS

Rate Schedule NSS), with the priority of deliveries on withdrawal being contingent on that of the transportation Agreement used to take the gas away from storage. A Shipper releasing a portion of its storage rights under an FRSS Agreement must designate how any existing FRSS Balance, as defined in Section 5 of this Rate Schedule, is to be allocated among itself and the replacement Shipper(s).

(5) Unless Natural agrees otherwise, a replacement Shipper may only utilize the storage and delivery points specified in the FRSS Agreement. If the release terminates or the rights are recalled prior to the end of a Withdrawal Period, the rights reverting to the releasing Shipper shall be reduced by the volume of service actually utilized by the replacement Shipper pursuant to the release and, unless otherwise agreed by both the releasing Shipper and the replacement Shipper, the obligation to return any volume withdrawn prior to the recall or termination shall be on the releasing Shipper. Unless otherwise agreed by both the releasing Shipper and the replacement Shipper(s), or unless otherwise provided herein, the obligation of the releasing Shipper and any replacement Shipper to return gas during the Injection Period shall be based on each Shipper's October 1 FRSS Balance.

3. VALID REQUEST

- 3.1 A request for service under this Rate Schedule FRSS shall be valid as of the date received if it complies with this Section and contains adequate information on all of the items specified in Section 3.2, subject to any necessary verification of such information and to the following:
- (a) A request shall not be valid and Natural shall not be required to grant any such request:
- (1) For which adequate capacity is not available in Natural's storage system and related transmission or other facilities necessary to provide such service on a firm basis;
- (2) As to which Natural does not have the operational capability, including adequate flexibility as to management of its cushion gas and storage operations, to effectuate service hereunder on a firm basis consistent with the terms and conditions of this Rate Schedule FRSS;

FERC Docket: RP08-319-000

Original Sheet No. 93 Original Sheet No. 93: Pending

RATE SCHEDULE FRSS

(3) Which, except as provided in Section 3.1(b) hereof, would require the construction, modification, expansion, or acquisition of any facilities; provided, however, that Natural may agree in its reasonable discretion to construct, modify, expand, or acquire facilities to enable it to perform such service;

(4) Unless and until Shipper has provided Natural with the information required in Section 3.2;

FERC Docket: RP08-319-000

Original Sheet No. 94 Original Sheet No. 94: Pending

RATE SCHEDULE FRSS

(5) If Natural determines, based on the credit analysis referenced in Section 3.2(f), that Shipper does not possess sufficient financial stability to make it reasonably likely that the service provided hereunder will be paid for on a timely basis;

- $\mbox{(6)}$ If the service requested would not comply with this Rate Schedule FRSS; or
- (7) If the service requested is at less than the applicable maximum rate; provided, however, that Natural may agree to provide service hereunder at a discount consistent with this Rate Schedule FRSS.
- (b) Natural agrees to construct minor meter and tap facilities to effectuate service requested under this Rate Schedule FRSS, provided that: (1) the cost of such facilities does not exceed \$100,000.00; (2) Natural shall be reimbursed prior to commencing construction; and (3) the facilities will not adversely affect Natural's operations. In the event that Natural agrees to construct, modify, expand or acquire facilities to perform service under this Rate Schedule FRSS, Natural shall maintain a separate record of the nature and costs of such facilities.
- (c) Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to this Section and while a FRSS Agreement is in effect. Such matters are governed by other provisions of this Rate Schedule FRSS and by the General Terms and Conditions of this Tariff.
- (d) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request, in whole or in part, due to lack of firm capacity or firm system capability or if the request is incomplete or does not comply with this Rate Schedule FRSS. Any request shall be null and void upon notice by Natural to Shipper if Natural does not have adequate firm system capacity or capability to grant the request. The request shall also be null and void unless it is substantially complete and complies with this Rate Schedule FRSS. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the FRSS Agreement, after which Shipper shall have fifteen (15) calendar days to provide the specified information. In the event such information is not received by Natural within fifteen (15) days, Shipper's request shall be null and void.

FERC Docket: RP08-319-000

Original Sheet No. 95 Original Sheet No. 95: Pending

RATE SCHEDULE FRSS

(e) Natural shall tender an FRSS Agreement to Shipper for execution when Shipper's request for service is accepted. Unless waived by Natural, a request for service shall be invalid if Shipper fails to execute an FRSS Agreement hereunder within ten (10) days after an FRSS Agreement which accurately reflects the Shipper's request has been tendered by Natural for execution.

3.2 Requests for service hereunder shall be deemed valid only after the information specified in this Section 3.2 is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department, at One Allen Center, 500 Dallas Street, Houston, Texas 77002, or Telecopy Number (713) 369-9305. A qualifying bid in an auction shall be treated as a valid request, provided that the bid contains or is supplemented with all information required under this Section 3.2 as required by the General Terms and Conditions of this Tariff. The information required for a valid request shall be as follows:

(a) VOLUME

The request shall specify in Dth an aggregate MDQ, which shall be utilized to determine the WQ and the MSWQ as set out in Section 5 of this Rate Schedule FRSS. Shipper shall also specify an MSWQ of either ten (10) or twenty (20) times the MDQ. All stated quantities shall be exclusive of Fuel Gas, Gas Lost and Unaccounted For and Storage Fuel. Fuel Gas, Gas Lost and Unaccounted For and Storage Fuel shall be provided by Shipper as set out in Sections 6.3(c), 6.3(d) and 7.2 of this Rate Schedule FRSS. If Shipper is willing to accept a lesser MDQ in the event capacity is prorated, Shipper shall also specify the minimum acceptable MDQ.

(b) SERVICE OPTION

Shipper shall elect whether it wants delivered storage service or the storage-only option. Under the storage-only option, Shipper will receive only the storage injection and withdrawal rights hereunder and will not have delivered storage service rights or any transportation service rights under the FRSS Agreement. A Shipper electing the storage-only option may obtain firm no-notice delivery rights as provided in Section 7.5(b) of this Rate Schedule. Unless otherwise mutually agreed, the Shipper must request such firm no-notice rights at the time it submits its request for service under this Rate Schedule.

FERC Docket: RP08-319-000

Original Sheet No. 96 Original Sheet No. 96: Pending

RATE SCHEDULE FRSS

(c) POINTS

The request shall specify the primary Delivery Point(s) for service under this Rate Schedule FRSS and the MDQ for each such point. Such point requests must be consistent with Section 7 of this Rate Schedule FRSS and the General Terms and Conditions of this Tariff. A Shipper may request any number of primary Delivery Points so long as the summation of MDQs at all primary Delivery Points equals the aggregate MDQ. The Receipt Point(s) under any FRSS Agreement shall be as provided in Section 7.4 of this Rate Schedule FRSS. If the Shipper selects the storage-only option, the storage point designated under Section 7.4(a)(2) shall be both the Receipt Point and the Delivery Point for the FRSS Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 97 Original Sheet No. 97: Pending

RATE SCHEDULE FRSS

(d) TERM OF SERVICE

The request shall specify:

- (1) The date service is requested to commence; and
- (2) The date service is requested to terminate.
- (e) REQUESTED RATE

The request shall state the rate at which the Shipper is willing to contract for the service requested. If no rate is stated, Shipper will be deemed to have agreed to the applicable maximum rate.

(f) CREDIT

Acceptance of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.

(g) COMPLIANCE WITH FRSS TARIFF

Submission of a request shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule FRSS, including the applicable General Terms and Conditions.

(h) COMMISSION-REQUIRED FILING INFORMATION

Except to the extent such information has been or will be provided under another Agreement, the following information is to be provided at the time a request for service hereunder is submitted, if available, or when an initial nomination for service under an executed FRSS Agreement is submitted, and when any subsequent changes occur:

- (1) Affiliation of the Shipper with Natural; and
- (2) The identity of the Shipper, including whether it is a local distribution company, an interstate pipeline company, an intrastate pipeline company, an end user, a producer, or a marketer.

FERC Docket: RP08-319-000

Original Sheet No. 98 Original Sheet No. 98: Pending

RATE SCHEDULE FRSS

4. TERM

- (a) The term of service hereunder shall be set forth in the FRSS Agreement between Shipper and Natural.
- (b) The General Terms and Conditions of this Tariff shall govern the applicability of, and the terms and conditions relating to, rollovers and the right of first refusal vis a vis an FRSS Agreement. Upon termination of any FRSS Agreement, and subject to any such rollover or right of first refusal, service by Natural to Shipper thereunder shall be terminated and automatically abandoned.
- (c) Natural may terminate any FRSS Agreement if Natural is required by the Commission or some other agency or court to provide firm service for others utilizing the system capacity or capability required for service under such FRSS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the type covered by the FRSS Agreement; provided, however, that nothing herein shall be read to protect Natural from the consequences of its own negligence.

5. WITHDRAWAL AND INJECTION RIGHTS

- 5.1 (a) Under the terms of this Rate Schedule FRSS, gas must first be withdrawn by Shipper before Shipper can inject gas and the level of injections is limited to the volume previously withdrawn.
- (b) A Shipper's WQ defines the maximum quantity of gas which Shipper has a firm right to withdraw on any day pursuant to this Rate Schedule FRSS. A Shipper's WQ shall be 100% of its MDQ for the period of May 15 through September 30 (Withdrawal Period). A Shipper's WQ during all other periods is zero. Shipper may only withdraw gas during the Withdrawal Period.
- (c) A Shipper's MSWQ is the maximum aggregate volume which the Shipper has a firm right to withdraw over the Withdrawal Period, subject also to the daily firm withdrawal limits defined by Shipper's WQ. Shipper's MSWQ shall equal Shipper's MDQ multiplied by either ten (10) or twenty (20), as specified by Shipper in its request for service.

FERC Docket: RP08-319-000

Original Sheet No. 99 Original Sheet No. 99: Pending

RATE SCHEDULE FRSS

- (d) The aggregate withdrawals actually made as of any point in time (reduced by aggregate injections made as of that date) shall be Shipper's FRSS Balance.
- 5.2 A Shipper's IQ is the maximum amount of gas which Shipper can inject hereunder on a firm basis on any day. Firm injections hereunder may only be made during the period from December 1 through the last day of February (Injection Period). A Shipper's IQ during the period of December 1 through January 31 shall equal its FRSS Balance as of October 1 (October 1 FRSS Balance) divided by 30. A Shipper's IQ during February shall equal its October 1 FRSS Balance divided by 45. A Shipper's IQ during any other period shall be zero. A Shipper may only inject gas during the Injection Period. Over the Injection Period, Shipper cannot inject either more or less gas than its October 1 FRSS Balance. Shipper's remaining FRSS Balance under an FRSS Agreement on the last day of February must be zero. If the FRSS Balance on the last day of February is other than zero, any remaining balance shall be cashed out on the same basis as set out in the last sentence of Section 8(d) of this Rate Schedule FRSS.
- 5.3 If Shipper exceeds its rights hereunder, the volume by which Shipper exceeds its rights shall be subject to Overrun Charges under Section 9 of this Rate Schedule.

6. RATE

- $\,$ 6.1 $\,$ Shipper shall pay Natural each month for service under this Rate Schedule FRSS the sum of the following charges:
 - (a) A Reservation Charge based on Shipper's MDQ;
- (b) Unless Shipper has selected the storage-only option, a Transportation Fuel Charge, comprised of a Fuel Gas Charge and a Gas Lost and Unaccounted For Charge, on withdrawal volumes;
 - (c) A Storage Fuel Charge on injection volumes;
 - (d) Overrun Charges, where applicable; and

FERC Docket: RP08-319-000

Original Sheet No. 100 Original Sheet No. 100: Pending

RATE SCHEDULE FRSS

(e) Any other applicable charges and penalties set out in this Rate Schedule or the General Terms and Conditions of this Tariff (including but not limited to Balancing Service Charges); provided, however, that FRSS Agreements shall not be subject to charges under Section 21 (Account No. 858 surcharges) or Section 38 (Gas Supply Realignment Charges) of the General Terms and Conditions of this Tariff.

- 6.2 Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.
- 6.3 The charges identified in subsections (a) through (d) of Section 6.1 shall be assessed as follows:
- The maximum Reservation Charge for service under this (a) Rate Schedule FRSS shall consist of the applicable maximum unit Reservation Rate set out in this Tariff multiplied by Shipper's MDQ. Separate maximum Reservation Rates shall be stated for Shippers selecting an MSWQ of ten (10) times MDQ and those selecting an MSWQ of twenty (20) times MDQ. Separate maximum Reservation Rates are also stated for Shippers electing delivered storage service and for those electing the storage-only option. The Reservation Charge payable by any Shipper shall be the applicable maximum Reservation Charge unless a lower Reservation Charge is specified in the FRSS Agreement or a separate discount agreement, or in a notice consistent with Section 6.7 hereof, which lower charge must be consistent with the minimum rate(s) specified in this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma Agreement set out in this Tariff, discounts may be limited to specific volumes. The Reservation Charge shall constitute the minimum bill under this Rate Schedule FRSS. Shipper shall be obligated to pay the Reservation Charge hereunder as of the date service commences. Service shall be deemed to commence on the date specified in the FRSS Agreement.
- (b) Shipper shall provide gas in kind for the fuel required to transport gas from storage to Delivery Points in connection with service under this Rate Schedule FRSS; provided, however, that this charge shall not be applicable to Shippers electing the storage-only option. Such charge shall be assessed upon withdrawal, reducing the volumes delivered to Shipper. The Transportation Fuel Charge for delivered storage service shall equal the Fuel Gas and Gas Lost and Unaccounted For Rates applicable to transportation from the storage point to the Delivery Point multiplied by each Dth

FERC Docket: RP08-319-000

Original Sheet No. 101 Original Sheet No. 101: Pending

RATE SCHEDULE FRSS

of gas withdrawn for the account of Shipper. Upon the mutual agreement of Natural and Shipper, in lieu of Natural retaining gas in kind, Shipper shall reimburse Natural for fuel applicable to delivered storage service at a price established under Section 7.1 of the General Terms and Conditions of this Tariff multiplied by the Transportation Fuel Charge.

- (c) Shipper shall provide gas in kind for the fuel required to inject and withdraw gas from storage in connection with service under this Rate Schedule FRSS. Such charge shall be assessed upon injection, reducing the volumes deemed delivered to Natural for injection into Shipper's FRSS account, and thus fuel volumes shall not subsequently enter into the determination of Shipper's FRSS Balance. The Storage Fuel Charge shall equal the Storage Fuel Rate set out in this Tariff (stated as a percentage fuel rate) multiplied by each Dth of gas injected under the FRSS Agreement. Upon the mutual agreement of Natural and Shipper, in lieu of Natural retaining gas in kind, Shipper shall reimburse Natural for fuel at a price established under Section 7.1 of the General Terms and Conditions of this Tariff multiplied by the Storage Fuel Charge.
- (d) The Authorized Overrun or Unauthorized Overrun Charges, as applicable, are governed by Section 9 of this Rate Schedule FRSS.
- 6.4 (a) Shipper shall reimburse Natural within five (5) business days after costs have been incurred by Natural for all fees required by the Commission or any regulatory body including, but not limited to, filing, reporting, and application fees to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees (such as general rate case filing fees). Any costs collected hereunder shall be excluded from Natural's cost of service in determining generally applicable rates and charges.
- (b) If Natural constructs, acquires or modifies any facilities to perform service hereunder, then as specified in an agreement between the parties either:
- (1) Shipper shall reimburse Natural for the cost of such facilities or facility modifications as described in the General Terms and Conditions of this Tariff; or
- (2) Natural shall assess a monthly charge reflecting such facility costs.

FERC Docket: RP08-319-000

Original Sheet No. 102 Original Sheet No. 102: Pending

RATE SCHEDULE FRSS

- $6.5\,$ The ACA and RD&D charges will be assessed, when applicable, as provided in the General Terms and Conditions of this Tariff, on volumes delivered by Natural to Shipper under this Rate Schedule FRSS.
- 6.6 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in (1) the rates and charges applicable under this Rate Schedule FRSS, including both the level and design of such rates and charges; or (2) the terms and conditions of this Rate Schedule FRSS. Natural

FERC Docket: RP08-319-000

Original Sheet No. 103 Original Sheet No. 103: Pending

RATE SCHEDULE FRSS

agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.

- (b) If, at any time and from time to time, the Commission or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the FRSS Agreement or any separate discount agreement, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such highest rate, Shipper shall execute or provide such documentation within fifteen (15) calendar days after a written request by Natural. If, at any time and from time to time, the Commission or any other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for service hereunder, the rate shall be decreased to such reduced rate.
- 6.7 Natural may from time to time and at any time, upon twenty-four (24) hours' verbal or written notice, subject to any provisions on discounting in the FRSS Agreement or in a separate discount agreement, charge any individual Shipper for service under this Rate Schedule FRSS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate charged may not be less than the applicable minimum rate for service under this Rate Schedule FRSS set forth in this Tariff. Natural will confirm any verbal notice of the applicable rate in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Unless otherwise agreed in the FRSS Agreement or in a separate discount agreement, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the FRSS Agreement and/or any discount agreement) upon twenty-four (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with the Commission any and all reports as required by the Commission's Regulations with respect to the institution or discontinuance of any discount.

FERC Docket: RP08-319-000

Original Sheet No. 104 Original Sheet No. 104: Pending

RATE SCHEDULE FRSS

6.8 All revenues and gas in kind collected by Natural as a result of providing service under this Rate Schedule FRSS shall be retained by Natural unless Natural has otherwise explicitly agreed on a different disposition of such amounts. Where crediting or refund mechanisms apply under other provision(s) of this Tariff or pursuant to effective Commission orders or settlements, such mechanisms shall supersede this Section to the extent necessary to carry out such provision(s).

7. NO-NOTICE SERVICE, NOMINATIONS AND POINTS

- (a) Except for service to Shippers electing the storage-only option, which are covered by Section 7.5 hereof, service shall be provided hereunder on a no-notice basis during the Withdrawal Period only, subject to appropriate arrangements being made with the point operator at the Delivery Point(s) where the Shipper is not the point operator. Except as otherwise explicitly agreed, if actual deliveries pursuant to delivered storage service exceed nominated deliveries during the Withdrawal Period at any Delivery Point under an FRSS Agreement, such difference between confirmed nominations for deliveries and actual deliveries at Delivery Points eligible for service under this Rate Schedule FRSS shall be treated as a withdrawal from Shipper's FRSS storage account on a no-notice basis, within the limits of Shipper's firm storage rights hereunder, subject to the provisions of Section 7.3 as to points eligible for no-notice service and as to the conditions for no-notice service. It shall be Shipper's responsibility to keep any difference between nominated and actual deliveries under delivered storage service within a level which can be accommodated by use of such storage rights. If Shipper fails to do so, Overrun and/or Balancing Service Charges shall, except as provided in Section 7.1(b), apply to the extent specified in Section 9 of this Rate Schedule FRSS and the General Terms and Conditions of this Tariff.
- (b) If Shipper takes action based on inaccurate information provided by Natural regarding any parameter relevant to its operation under this Rate Schedule FRSS (such as Shipper's IQ, WQ, MSWQ or FRSS Balance), and such action results in Shipper exceeding its rights, Natural will waive (except as provided in the following sentence) any charges Shipper could reasonably have avoided had Natural provided accurate information. Charges hereunder shall not be waived to the extent that the inaccurate information provided by Natural resulted from Shipper's failure to provide Natural accurate information within its control on a timely basis.

FERC Docket: RP08-319-000

Original Sheet No. 105 Original Sheet No. 105: Pending

RATE SCHEDULE FRSS

(c) Shipper must nominate all injections and any Authorized Overrun service or service at secondary Delivery Points (except as otherwise provided in Section 7.3).

- 7.2 (a) Except for deliveries pursuant to the storage-only option (in which case the provisions of Section 7.4(a)(2) hereof shall apply), the primary Delivery Point(s) for all gas delivered by Natural to Shipper (or to a third party on behalf of Shipper) hereunder shall be specified in the FRSS Agreement. For a Shipper with delivered storage service, all points in the zone in which the Delivery Point is located may be used as secondary out-of-path Delivery Points hereunder, with an MDQ at each point and an aggregate MDQ at all points in the zone no greater than the sum of the MDQs for primary Delivery Points in that zone, consistent with the General Terms and Conditions of this Tariff, but such secondary points shall only be eligible for no-notice service to the extent provided in Section 7.3 of this Rate Schedule FRSS.
- (b) For each individual primary Delivery Point, and for the aggregate of all such points, Natural's maximum obligation to deliver gas on a firm basis (i.e., the MDQ) shall be specified in Dth in the FRSS Agreement. The sum of the MDQs at primary Delivery Points may not exceed the aggregate MDQ. All stated MDQs shall be exclusive of applicable Fuel Gas, Gas Lost and Unaccounted For, and Storage Fuel. Fuel Gas and Gas Lost and Unaccounted For shall be provided by Shipper for delivered firm storage service upon withdrawal by reducing the volumes deemed delivered to Shipper. Storage Fuel shall be provided by Shipper upon injection by reducing the volume deemed delivered to Natural for injection.
- 7.3 All primary Delivery Points under the FRSS Agreement, along with all secondary Delivery Points that together with one or more of the primary points comprise a Central Delivery Point, shall be eligible for no-notice service on withdrawal as part of delivered storage service under this Rate Schedule FRSS, subject to appropriate arrangements being made with the point operator where the Shipper is not the point operator at the Delivery Point. A Shipper may also request, by notice to Natural, to take no-notice delivered storage service at other secondary Delivery Points in lieu of points described in the prior written notice to Natural, and Natural shall promptly notify the Shipper whether such points are eligible for no-notice service.
- 7.4 (a) (1) Except as provided in Sections 7.4(a) (2) and 7.4(c) hereof, the Receipt Points for service hereunder shall be the storage point(s) defined in this Tariff which are in the same Receipt Zone as the Delivery Point(s).

FERC Docket: RP08-319-000

Original Sheet No. 106 original Sheet No. 106: Pending

RATE SCHEDULE FRSS

- (2) If the Shipper has elected the storage-only option, its Receipt Point and Delivery Point shall be the same and shall be a storage point designated by the Shipper. If the Shipper elects firm nonotice rights under Section 7.5(b)(2), the storage point must be in the same Receipt Zone as the point at which firm no-notice deliveries are to be made under the designated transportation Agreement. A Shipper electing the storage-only option shall not have any secondary point or other transportation service rights under the FRSS Agreement.
- (3) Where Shipper has selected delivered storage service and Shipper's Delivery Point is in the Market Delivery Zone, the Receipt Point shall be either the Amarillo or Gulf Coast storage point, whichever is consistent with the location of the Delivery Point. The location of the Receipt Point for delivered storage service shall be determined based on the location of the Delivery Point and the application of Sections 5.6 and 9.2 of the General Terms and Conditions of this Tariff.
- (4) Each FRSS Agreement shall have the storage point(s) consistent with this Section 7.4 as Receipt Point(s) and will be subject to the limitations and conditions (including volumes and priorities) specified in the General Terms and Conditions of this Tariff relating to such points. All gas shall be withdrawn from and injected into such storage point(s). Natural shall establish and maintain an accounting of all gas withdrawn by Delivery Point. Gas volumes to be injected must, subject to Section 7.4(c), be tendered at the zone storage point(s) which corresponds to the Delivery Point(s) at which gas was delivered by Natural to Shipper on withdrawal or in the case of the storage-only option, at the Receipt Point(s) determined under Section 7.4(a)(2) hereof, which are zone storage points. The volumes injected over the Injection Period at each such zone storage point must be equivalent to the volumes attributable to that zone storage point delivered over the Withdrawal Period.
- (b) While such storage point(s) are not physical points, the Receipt Point(s) hereunder shall be treated as primary points on withdrawal for scheduling, curtailment and similar purposes; provided, however, that if the Shipper has selected the storage-only option, Section 7.5 shall control as to withdrawals. For injection, such storage points shall be treated as primary Delivery Points (under the separate transportation Agreement used for injection) for scheduling and curtailment to the extent provided in Section 5.6 of the General Terms and Conditions of this Tariff. Any separate firm transportation Agreement held by Shipper under which it desires to make injections into its FRSS account may have the applicable storage point(s) as Delivery Points and such storage points shall be treated as primary Delivery Points (under the separate transportation Agreement used to inject gas) for scheduling and curtailment to the extent provided in Section 5.6 of the General

FERC Docket: RP08-319-000

Original Sheet No. 107 Original Sheet No. 107: Pending

RATE SCHEDULE FRSS

Terms and Conditions of this Tariff. Further, such transportation Agreement may specify that such storage point(s) are to be the Delivery Point(s) during the Injection Period, with the applicable priority, and that the Shipper's Delivery Point(s) are to be the primary Delivery Points during the Withdrawal Period.

- (c) By mutual agreement and on basis which is not unduly discriminatory, Natural may accept gas tendered by Shipper for injection at point(s) other than the Receipt Point(s) if accepting gas at such alternative point(s) is operationally beneficial to Natural's system.
- 7.5 (a) A Shipper electing the storage-only option shall have both its Receipt and Delivery Point under the FRSS Agreement at the designated storage point as provided under Section 7.4(a)(2) of this Rate Schedule. Such a Shipper must separately arrange for transportation of gas withdrawn and must nominate withdrawals and deliveries unless it has firm no-notice rights under Section 7.5(b).
- At the time a Shipper submits its request for service (b) (1) hereunder and elects the storage-only option (or at such later time as is mutually agreed), such Shipper may request firm no-notice deliveries to a specified point which is either a primary Delivery Point, or a secondary Delivery Point within the primary path, under a designated separate firm transportation Agreement. The designated firm transportation Agreement may be a new or existing Agreement, including a capacity Release Agreement. Shipper shall be eligible for such firm no-notice deliveries up to the lesser of its firm withdrawal rights under the FRSS Agreement or its available (unutilized and unnominated) firm transportation service rights under the firm transportation Agreement(s), subject to the remainder of this subsection. The path of the designated transportation Agreement must be such that deliveries to the specified Delivery Point can be supported operationally from an upstream or downstream physical storage field. On any day, Shipper's firm no-notice rights shall be limited to withdrawals within its rights under the FRSS Agreement, and within the unutilized and unnominated portion of its service rights under such designated firm transportation Agreement(s). If the designated firm transportation Agreement is an FFTS Agreement, firm no-notice service is available only on Available Days or partial Available Days, up to the unutilized and unnominated portion of the MDQ in effect on that day under the FFTS Agreement, and within Shipper's withdrawal rights under the FRSS Agreement.
- (2) Shipper must designate the firm transportation Agreement(s) to be utilized for such firm no-notice rights. Such designation may be revised by mutual agreement of Shipper and Natural.

FERC Docket: RP08-319-000

Original Sheet No. 108 Original Sheet No. 108: Pending

RATE SCHEDULE FRSS

Any such designation must be consistent with this Section 7.5 and with Section 7.4 (a) (2) of this Rate Schedule. In addition, a predetermined allocation agreement with the point operator must be in place which defines how such firm no-notice deliveries are determined.

- (3) Shipper's no-notice rights shall terminate if it does not have an appropriate and valid firm transportation Agreement designated under this Section 7.5(b); for example, if the originally designated firm transportation Agreement has expired or been released and no substitute Agreement has been designated.
- (4) The separate firm transportation Agreement used to support firm no-notice rights hereunder shall be designated as an FTS-FR, FFTS-FR or FTS-G/FR Agreement, as applicable.

8. OTHER OPERATING CONDITIONS

- (a) Shipper shall arrange separately for the transportation of gas to be injected in connection with service under this Rate Schedule FRSS, and for withdrawals if Shipper has selected the storage-only option. Shipper shall make all necessary arrangements separately upstream of the point where gas is tendered to Natural. Shipper is also responsible for all arrangements downstream of the Delivery Point(s). Such arrangements must be consistent with this Rate Schedule FRSS and must be coordinated with Natural, subject to the no-notice aspects of the service being provided under the FRSS Agreement. To be eligible for service hereunder, Shipper or its supplier or purchaser must comply with all applicable provisions of any separate transportation Agreement(s) and any applicable Rate Schedule(s) and the applicable General Terms and Conditions of this Tariff.
- (b) Eligibility for this service does not require that a Shipper have firm transportation for injection, but Shipper has an obligation to return by the end of the Injection Period all gas which it withdrew hereunder during the prior Withdrawal Period. That obligation is not excused if Shipper selects interruptible transportation and Shipper's gas is not scheduled for injection or if Shipper fails to nominate and/or tender sufficient gas for injection under any Agreement. Eligibility for service under the storage-only option does not require that a Shipper have firm transportation service for withdrawal, but the scheduling for withdrawals may be affected by the transportation service selected.
- (c) Natural will manage its cushion gas and storage operations to support withdrawals under Rate Schedule FRSS. Shippers will provide gas for injection by the end of the Injection Period sufficient to replace gas withdrawn during the prior Withdrawal Period, as set out in Section 5 of this Rate Schedule FRSS.

FERC Docket: RP08-319-000

Original Sheet No. 109 Original Sheet No. 109: Pending

RATE SCHEDULE FRSS

(d) Shipper must have an FRSS Balance of zero at the end of the term of a FRSS Agreement. If at any time service pursuant to this Rate Schedule FRSS and/or a FRSS Agreement is canceled or terminates prior to the end of the term specified in the FRSS Agreement, Shipper shall be required to eliminate any then-remaining FRSS Balance gas over a period of time determined by dividing: (1)the FRSS Balance, by (2) fifty percent (50%) of Shipper's MDQ; provided, however, that if service is being terminated pursuant to Section 19.19 of the General Terms and Conditions of this Tariff, the withdrawal period shall be equal in duration to the prior notice period required under said Section 19.19. If Natural is unable to accept any volume tendered by Shipper, up to the volume specified in the preceding sentence, the period shall be extended accordingly. If the gas is not injected within the specified time period, the remaining volumes shall be cashed out with no offsets, trading or netting and the proceeds retained by Natural, based on the greater than twenty percent (20%) tier for imbalances set out at Section 13.3 of the General Terms and Conditions of this Tariff.

9. OVERRUN SERVICE

During the Withdrawal Period, Natural may (but is not obligated to), upon request from Shipper, withdraw and deliver volumes in excess of a Shipper's WQ and/or MSWQ. Natural may (but is not obligated to), upon request from Shipper, receive for injection hereunder on any day during the Injection Period as authorized overrun quantities of natural gas in excess of Shipper's IQ. Such requests can be granted when in Natural's reasonable judgment, the capacity, operating capability and management of cushion gas and storage operations on its system will permit without impairing the ability of Natural to meet its other service obligations of higher priority; provided, however, that in no event will Shipper be allowed to inject by the end of the Injection Period a volume of gas which does not equal Shipper's October 1 FRSS Balance. In granting requests for Authorized Overrun Service, Natural shall act in a manner consistent with the overrun service priorities set out in the General Terms and Conditions of this Tariff. Maximum charges for Authorized Overrun shall be set out on the rate sheet for Rate Schedule FRSS; provided, however, that such Authorized Overrun charges may be waived or discounted. If Shipper exceeds its WQ, IQ or MSWQ without authorization from Natural, the charges for Unauthorized Overrun set out in Section 12.2(a)(1) of the General Terms and Conditions of this Tariff shall apply.

FERC Docket: RP08-319-000

Original Sheet No. 110 original Sheet No. 110: Pending

RATE SCHEDULE FRSS

10. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule FRSS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Original Sheet No. 111 Original Sheet No. 111: Pending

RATE SCHEDULE FTS-G FIRM TRANSPORTATION SERVICE FOR FORMER G-1 BUYERS

1. AVAILABILITY

This Rate Schedule FTS-G is available for firm transportation service by Natural Gas Pipeline Company of America LLC (hereinafter called Natural) only to Buyers under Natural's Rate Schedule G-1 (hereinafter called Shipper), or former sales customers which qualify for service under Rate Schedule G-1, which converted all or any portion of their sales contracts to service hereunder. No new service requests shall be accepted or granted subsequent to such conversion process. The form of FTS-G Agreement is contained in this Tariff. A Shipper requesting transportation under this Rate Schedule FTS-G must provide all data required by Section 3.2 of Rate Schedule FTS.

2. PRIORITY OF SERVICES

Shipper shall have the same priority for, and rights to use, transmission and storage capacity on Natural's system hereunder as it would under Rate Schedule FTS except as provided in Section 4 of this Rate Schedule FTS-G.

3. APPLICABILITY AND CHARACTER OF SERVICE

- (a) The service provided under this Rate Schedule FTS-G shall be performed under Subpart B or G of Part 284 of the Commission's Regulations. This Rate Schedule FTS-G shall apply to all gas received by Natural for Shipper pursuant to an FTS-E/G Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 112 Original Sheet No. 112: Pending

RATE SCHEDULE FTS-G

TERMS AND CONDITIONS OF SERVICE

The terms and conditions of service hereunder shall be identical to those set out in Rate Schedule FTS and the related General Terms and Conditions, with references to "FTS" replaced by "FTS-G," with the following exceptions:

- (a) The availability of service hereunder is limited as set out in Section 1 of this Rate Schedule FTS-G;
- (b) There shall be no reservation charge or monthly minimum bill for service under this Rate Schedule FTS-G. The rate provisions applicable to service hereunder are set out in Section 5 of this Rate Schedule FTS-G;
- (c) Shipper's rights to use other services are restricted as set out in Section 6 of this Rate Schedule FTS-G;
- (d) Shipper shall be subject to the Insufficient Supply Penalty set out in Section 8 of this Rate Schedule FTS-G;
- (e) Shipper's right to change primary Delivery Points is restricted to points within its traditional market area; and
- (f) Service hereunder is not eligible for Natural's Capacity Release Program.

5. RATE

- (a) Shipper shall pay Natural each month for service hereunder a one-part Volumetric Rate for each unit of gas delivered to it by Natural during that month, plus an amount equal to the volumes attributable to Fuel and Gas Lost and Unaccounted For multiplied by one-half (1/2) of such Volumetric Rate.
- (b) The maximum monthly Volumetric Charge shall be the applicable maximum unit Volumetric Rate(s) set out in this Tariff for service hereunder multiplied by the quantity of gas delivered by Natural during the billing month, plus an amount equal to the volumes attributable to Fuel and Gas Lost and Unaccounted For multiplied by one-half (1/2) of the applicable maximum unit Volumetric Rate(s). The maximum unit Volumetric Rate(s) shall be calculated as the derivative, at a fifty percent (50%) imputed load factor, of the corresponding maximum Reservation and Commodity Rates

FERC Docket: RP08-319-000

Original Sheet No. 113 original Sheet No. 113: Pending

RATE SCHEDULE FTS-G

which would be applicable to such transportation under Rate Schedule FTS. Neither Natural nor any party to Docket No. CP89-1281 may file to request a change in the rate design in the prior sentence prior to December 1, 1995 unless ordered to do so by the Commission. The Volumetric Rate shall be based on the primary Receipt and Delivery Points specified in the FTS-G Agreement and, unless Natural agrees otherwise, the Volumetric Charge shall not be reduced by a change in the primary Receipt and Delivery Points originally specified in the FTS-G Agreement. The charge payable by Shipper shall be such maximum Volumetric Charge unless a lower charge is agreed upon in the FTS-G Agreement or a separate discount agreement or is assessed consistent with Section 5.7 of Rate Schedule FTS. Such lower charge must be consistent with the minimum rate(s) specified in this Tariff. Shipper shall also pay such Volumetric Rate for any authorized transportation of volumes in excess of the Shipper's MDQ.

- (c) Shippers under this Rate Schedule shall be subject to other applicable charges and penalties set out in Rate Schedule FTS and the General Terms and Conditions of this Tariff except as otherwise provided in this Rate Schedule.
- (d) Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.

6. RESTRICTIONS ON OTHER SERVICES

A Shipper under this Rate Schedule FTS-G may not receive deliveries of gas in the service area covered by the FTS-G Agreement except under a DSS, FRSS, FFTS, FTS or FTS-G Agreement with Natural and may not utilize Natural's Capacity Release Program in connection with service hereunder or receive gas thereunder as a Replacement Shipper; provided, however, that Shipper may contract with others to supply markets not served by Natural when Shipper was a Buyer under Rate Schedule G-1 or to serve increased demand in any of its markets above the level served by Natural when Shipper was a Buyer under Rate Schedule G-1;

FERC Docket: RP08-319-000

Original Sheet No. 114 Original Sheet No. 114: Pending

RATE SCHEDULE FTS-G

and provided, further, that Natural may waive the restrictions on other services (but not the prohibition on capacity releases) contained herein to the extent such a waiver is necessary to retain load on its system.

7. CONVERSION RIGHTS

A Shipper under this Rate Schedule FTS-G may convert to service under Rate Schedule FTS at any time, on thirty (30) days' notice. Upon such a conversion, the Shipper maintains the same priority for capacity as it had under this Rate Schedule FTS-G.

FERC Docket: RP08-319-000

Original Sheet No. 115 Original Sheet No. 115: Pending

RATE SCHEDULE FTS-G

INSUFFICIENT SUPPLY PENALTY

- If Natural becomes aware that an Insufficient Supply Situation exists under which Shipper's supply at the Receipt Points applicable to Shipper's FTS-G Agreement is less than the volume of gas Shipper is taking at the Delivery Points applicable to Shipper's FTS-G Agreement, Natural shall promptly notify Shipper of such situation and inform Shipper that it may be subject to the Insufficient Supply Penalty if corrective action is not taken. Such notification will be given when Natural becomes aware that an Insufficient Supply Situation exists or will exist because it: is unable to confirm Shipper's nominations, or (b) otherwise becomes aware (i) that gas is not being delivered to Natural for Shipper's account or (ii) that Shipper is taking more gas at its Delivery Point(s) than has been nominated. Natural shall make such notification by telephone, telefax, or any other method mutually agreeable to Natural and Shipper. Such corrective action would be either a reduction in takes by Shipper at the Delivery Point(s) or an increase in gas tendered to Natural at the Receipt Point(s) available under the FTS-G Agreement.
- If Shipper fails to correct the Insufficient Supply Situation by the end of the next day after notice by Natural, Natural shall assess Shipper and Shipper shall pay Natural an Insufficient Supply Penalty for all gas taken thereafter in excess of the volumes received during the period of the temporary loss, interruption and/or shortage of supply at Shipper's Receipt $\operatorname{Point}(s)$. The Insufficient Supply Penalty shall be a unit rate multiplied by the discrepancy in volumes experienced during the Insufficient Supply Situation. The unit rate shall equal the sum of (1) the highest gas price actually assessed by Natural that month under the cashout provision of the General Terms and Conditions of this Tariff, plus (2) \$15 per Dth during a Critical Time or \$5.00per Dth at any other time. In addition, volumes taken shall be subject to all other charges applicable under this Rate Schedule FTS-G; provided that the same volumes taken shall not be subject to the other Balancing Service Charges and penalties under Rate Schedule FTS nor shall such volumes be considered when determining the monthly imbalance. There shall be a zero (0) tolerance level and no correction period in determining volumes subject to the Insufficient Supply Penalty.
- (c) The notice requirement for changes in nominations shall be waived when revised nominations are required to rectify the Insufficient Supply Situation.

FERC Docket: RP08-319-000

Original Sheet No. 116 original Sheet No. 116: Pending

RATE SCHEDULE ITS INTERRUPTIBLE TRANSPORTATION SERVICE

AVATLABILITY

This Rate Schedule ITS is available to any entity (hereinafter called Shipper) which: (i) submits to Natural Gas Pipeline Company of America LLC (hereinafter called Natural) a valid request as defined in Section 3 hereof; and (ii) executes an Interruptible Transportation Service Agreement (ITS Agreement) with Natural applicable to service under this Rate Schedule ITS. The form of ITS Agreement is contained in this Tariff. There is no limitation on the number of ITS Agreements any one Shipper may have.

2. APPLICABILITY, CHARACTER AND PRIORITY OF SERVICE

- (a) This Rate Schedule ITS defines an interruptible transportation service. This Rate Schedule ITS shall apply to all gas received by Natural for Shipper pursuant to an ITS Agreement. As more fully set out in the General Terms and Conditions of this Tariff, Natural is not providing a supply service under this Rate Schedule ITS.
- Service hereunder shall consist of the acceptance by Natural of natural gas from or for the account of Shipper at Receipt Point(s) under the ITS Agreement, the transportation of that natural gas through Natural's system, and the delivery of that natural gas, adjusted for Fuel Gas and Gas Lost and Unaccounted For as set out in the General Terms and Conditions, by Natural to Shipper or for Shipper's account at Delivery Point(s) under the ITS Agreement. Natural shall not be required: (1) to accept on any day gas tendered, or to deliver on any day gas requested, in excess of the Maximum Daily Quantity (MDQ) specified in the ITS Agreement; (2) to accept or deliver gas at any point in excess of the MDQ applicable to such point hereunder pursuant to the General Terms and Conditions of this Tariff; or (3) to accept or deliver on any day gas hereunder which is not properly nominated pursuant to and to the extent required by the General Terms and Conditions of this Tariff. In addition, service hereunder shall be fully interruptible as set out in the General Terms and Conditions of this Tariff.
- (c) The service provided under this Rate Schedule ITS shall be performed under Part 284 of the Commission's Regulations. Shipper shall only tender gas for transportation under this Rate

FERC Docket: RP08-319-000

Original Sheet No. 117 original Sheet No. 117: Pending

RATE SCHEDULE ITS

Schedule ITS to the extent such service would qualify under the applicable statutes, regulations, Commission orders and the blanket certificate authorizing service by Natural under this Rate Schedule. For service under Subpart B of Part 284 of the Commission's Regulations, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by the Commission's Regulations, Shipper shall (prior to tendering gas under an ITS Agreement) cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit the necessary certification.

(d) Service hereunder is provided on an interruptible basis. Curtailment and priorities of service for the purposes of scheduling and curtailment are governed by the General Terms and Conditions of this Tariff.

3. VALID REQUESTS

- 3.1 A request for service under this Rate Schedule ITS shall be valid as of the date received if it complies with this Section and contains adequate information on all of the items specified in Section 3.2, subject to any necessary verification of such information and to the following:
- (a) a request shall not be valid and Natural shall not be required to grant any such request: (i) which, except as provided for in Section 3.1(b) hereof, would require the construction, modification, expansion, or acquisition of any facilities; provided, however, that Natural may agree in its reasonable discretion to construct, modify, expand, or acquire facilities to enable it to perform such services; (ii) unless and until Shipper has provided Natural with the information required in Section 3.2 hereof; (iii) if Natural determines, based on the credit analysis referenced in Section 3.2(d), that Shipper does not possess sufficient financial stability to make it reasonably likely the service provided hereunder will be paid for on a timely basis; (iv) if the service requested would not comply with this Rate Schedule ITS; or (v) if the service requested is at less than the applicable maximum rate; provided, however, that Natural may agree to provide service hereunder at a discount consistent with this Rate Schedule ITS. Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to this Section and while an ITS Agreement is in effect. Such matters are governed by the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 118 Original Sheet No. 118: Pending

RATE SCHEDULE ITS

(b) Natural agrees to construct minor meter and tap facilities to effectuate service requested under this Rate Schedule ITS provided that: (i) the cost of such facilities does not exceed \$100,000.00; (ii) Natural shall be reimbursed prior to commencing construction; and (iii) the facilities will not adversely affect Natural's operations.

- (c) In the event that Natural agrees to construct, modify, expand or acquire facilities to perform service under this Rate Schedule ITS, Natural shall maintain a separate record of the nature and costs of such facilities.
- (d) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request because such request is incomplete or does not comply with this Rate Schedule ITS. Any request shall be null and void unless it is substantially complete and complies with this Rate Schedule. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the ITS Agreement, after which Shipper shall have fifteen (15) days to provide the specified information. In the event such information is not received within fifteen (15) days, Shipper's request shall be null and void.
- (e) Natural shall tender an ITS Agreement to Shipper for execution when Shipper's request for service is accepted. Unless waived by Natural, a request for service shall be invalid if Shipper fails to execute an ITS Agreement hereunder within thirty (30) days after an ITS Agreement has been tendered by Natural for execution.
- 3.2 Requests for service hereunder shall be deemed valid only after the following information is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department, One Allen Center, 500 Dallas Street, Houston, Texas 77002, or Telecopy Number (713) 369-9305:

(a) GAS QUANTITIES

The request shall specify in Dth the aggregate MDQ, exclusive of applicable Fuel Gas and Gas Lost and Unaccounted; provided, however, that Natural shall not be obligated to accept requests for an aggregate MDQ of less than one hundred (100) Dth per day.

FERC Docket: RP08-319-000

Original Sheet No. 119 Original Sheet No. 119: Pending

RATE SCHEDULE ITS

(b) AVAILABILITY OF POINTS

(1) A Shipper may utilize all available Receipt and Delivery Points on Natural's system under any ITS Agreement, as more fully set out in the General Terms and Conditions of this Tariff.

- $\mbox{\footnotemark}(2)$ The available volume and priorities at any point shall be governed by the General Terms and Conditions of this Tariff.
 - (c) TERM OF SERVICE

The request shall specify:

(1) The date service is requested to commence;

and

- (2) The date service is requested to terminate.
- (d) CREDIT

Acceptance of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.

(e) COMPLIANCE WITH ITS TARIFF

Submission of a request for service hereunder shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule ITS, including the applicable General Terms and Conditions.

(f) COMMISSION-REQUIRED FILING INFORMATION

The following information is to be provided at the time a request for service hereunder is submitted, if available, or when an initial nomination for transportation under an executed ITS Agreement is submitted, and when any subsequent changes occur:

(1) Affiliation of the Shipper with Natural; and

FERC Docket: RP08-319-000

Original Sheet No. 120 Original Sheet No. 120: Pending

RATE SCHEDULE ITS

(2) The identity of the Shipper, including whether it is a local distribution company, an interstate pipeline company, an intrastate pipeline company, an end user, a producer, or a marketer.

4. TERM

- (a) The term of service hereunder shall be set forth in the ITS Agreement between Shipper and Natural. Natural may terminate the ITS Agreement if Shipper fails to cause gas to be delivered during any twelve (12) consecutive calendar months when capacity is available, unless Shipper's failure to deliver gas was attributable to circumstances of Force Majeure.
- (b) The General Terms and Conditions of this Tariff shall govern the applicability of rollovers vis a vis an ITS Agreement. Upon termination of any ITS Agreement, and subject to such rollovers, service by Natural to Shipper thereunder shall be terminated and automatically abandoned.
- (c) Natural may terminate any ITS Agreement if Natural is required by the FERC or some other agency or court to provide service for others utilizing the interruptible system capacity or capability required for service under such ITS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the ITS type covered by the ITS Agreement; provided, however, that nothing herein shall be read to protect Natural from the consequences of its own negligence.

FERC Docket: RP08-319-000

Original Sheet No. 121 Original Sheet No. 121: Pending

RATE SCHEDULE ITS

5. RATE

- 5.1 Shipper shall pay Natural each month for service under this Rate Schedule ITS the sum of the following charges:
- (a) a Commodity Charge for each unit of gas delivered hereunder by Natural during the billing month;
 - (b) Gathering Charge(s);
- $% \left(c\right) =\left\{ c\right\} =\left\{ c\right\}$ all applicable types of transition cost charges; and
- (d) any other applicable charges and penalties set out in this Rate Schedule ITS or the General Terms and Conditions of this Tariff.

Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.

- 5.2 The charges identified in items (a)-(c) of Section 5.1 shall be assessed as follows:
- (a) The maximum monthly Commodity Charge shall be the applicable maximum unit Commodity Rate(s) set out in this Tariff multiplied by the quantity of gas delivered by Natural to Shipper hereunder during the billing month plus an amount equal to the volumes attributable to Fuel Gas and Gas Lost and Unaccounted For multiplied by one-half of the applicable maximum unit Commodity Rate(s). The Commodity Charge payable by any Shipper shall be the maximum Commodity Charge unless a lower Commodity Charge is specified in the ITS Agreement or a separate discount agreement or is assessed pursuant to Section 5.7 hereof, which lower charge must be consistent with the minimum Commodity Rate(s) specified in this Tariff. For any volumes delivered in excess of Shipper's MDQ, which deliveries shall be subject to Section 8 of this Rate Schedule ITS, Shipper shall pay as a Commodity Rate either the Authorized Overrun Rate or the Unauthorized Overrun Rate specified in this Tariff, as applicable, plus an amount equal to the volumes attributable to Fuel Gas and Gas Lost and Unaccounted For multiplied by one-half of the applicable Commodity Rate(s).

FERC Docket: RP08-319-000

Original Sheet No. 122 Original Sheet No. 122: Pending

RATE SCHEDULE ITS

(b) Separate Gathering Charge(s) shall apply to all volumes received by Natural under this Rate Schedule at any Receipt Point(s) on its gathering facilities.

 $\,$ (c) Transition cost charges shall be assessed, if applicable, in accordance with the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 123 Original Sheet No. 123: Pending

RATE SCHEDULE ITS

5.3 (a) Shipper shall reimburse Natural within five (5) days after costs have been incurred by Natural for all fees required by the FERC or any regulatory body including, but not limited to, filing, reporting, and application fees to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees (such as general rate case filing fees).

- (b) If Natural constructs, acquires or modifies any facilities to perform service hereunder, then as specified in an agreement between the parties either:
- (i) Shipper shall reimburse Natural for the cost of such facilities or facility modifications as described in the General Terms and Conditions of this Tariff; or
- $\mbox{(ii)}\mbox{\ Natural shall}$ assess a monthly charge reflecting such facility costs.
- 5.4 The ACA and RD&D charges will be assessed, when applicable, as provided in the General Terms and Conditions of this Tariff, on volumes delivered by Natural to Shipper under this Rate Schedule ITS. Shipper shall also reimburse Natural for Fuel Gas and for Gas Lost and Unaccounted For hereunder as provided in the General Terms and Conditions of this Tariff.
- 5.5 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in (1) the rates and charges applicable under this Rate Schedule ITS, including both the level and design of such rates and charges; or (2) the terms and conditions of this Rate Schedule ITS. Natural agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.
- (b) If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the ITS Agreement or a separate discount agreement, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such highest rate, Shipper shall execute or provide such documentation

FERC Docket: RP08-319-000

Original Sheet No. 124 Original Sheet No. 124: Pending

RATE SCHEDULE ITS

within fifteen (15) days after a written request by Natural. If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for transportation service hereunder, the rate shall be decreased to such reduced rate.

- 5.6 Shipper may request that Natural process its gas in order that such gas will meet the quality specifications of the General Terms and Conditions of this Tariff. Natural may, but is not obligated to provide such service. The providing of such service and the terms and conditions therefore shall be negotiated between Natural and Shipper and the terms and conditions agreed upon shall be specified in the ITS Agreement or a separate treating/processing agreement. For any such service, Natural shall assess Shipper a charge reflecting all the costs of providing the service.
- 5.7 Natural may from time to time and at any time, upon twenty-four (24) hours' verbal or written notice, subject to any provisions on discounting in the ITS Agreement or in a separate discount agreement, charge any individual Shipper for service under this Rate Schedule ITS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate charged may not be less than the applicable minimum rate for service under Rate Schedule ITS set forth in this Tariff. Natural will confirm any verbal notice of the applicable rate in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Unless otherwise agreed in the ITS Agreement or in a separate discount agreement, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the ITS Agreement and/or any discount agreement) upon twenty-four (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with the Commission any and all reports as required by the Commission's Regulations with respect to the institution or discontinuance of any discount.
- 5.8 All revenues and gas in kind collected by Natural as a result of providing service under this Rate Schedule ITS shall be retained by Natural unless Natural has otherwise explicitly agreed on a different disposition of such amounts. Where crediting or refund mechanisms apply under other provision(s) of this Tariff or

FERC Docket: RP08-319-000

Original Sheet No. 125 Original Sheet No. 125: Pending

RATE SCHEDULE ITS

pursuant to effective Commission orders or settlements, such mechanisms shall supersede this Section to the extent necessary to carry out such provision(s).

- 5.9 (a) Natural may, but is not obligated to, transport liquids for Shipper. Any transportation of liquids on behalf of Shipper shall be pursuant to a separate agreement, which agreement shall set out the negotiated rate for such transportation and shall supersede subsection (b) of this Section; provided, however, that if Natural does provide such transportation in the absence of a separate agreement, subsection (b) of this Section shall apply. The credit evaluation provisions of the General Terms and Conditions of this Tariff shall be applicable to such transportation unless otherwise specified in a separate agreement.
- (b) (1) Absent a separate agreement (in which case the negotiated rate set out in such agreement shall govern), Natural shall charge Shipper and Shipper shall pay Natural for Natural's transportation and handling of Shipper's liquids a monthly fee equal to \$1.025 per barrel of liquids.
- (2) If, at any time and from time to time, as a result of any legislation or any order, opinion, or decision of the Commission or of any other governmental authority having jurisdiction over Natural's rates including any order, opinion, or decision based in whole or in part on a contested or uncontested settlement agreement relative to Natural's rates, an amount is allocated or attributed to the transportation and handling of liquids, liquefiables, plant fuel, vaporous non-hydrocarbons and/or similar materials, and such amount is credited to Natural's cost of service or otherwise excluded in determining Natural's rates, Shipper shall reimburse Natural for the sum of money (including the interest reflected therein) Natural is required to refund or credit, or has refunded or credited to its customers which is attributed by Natural to such transportation and handling for Shipper under the ITS Agreement and which has not been assessed and collected from Shipper for the time period to which the refund or credit applied. If such refunded or credited money is less than the sum of money assessed and collected from Shipper for the applicable time period, Natural shall remit the difference to Shipper within thirty (30) days of the applicable order, opinion, or decision. Thereafter, Shipper shall pay Natural monthly fees based on amounts which are attributable by Natural to such transportation and handling for Shipper under the ITS Agreement as calculated and assessed by

FERC Docket: RP08-319-000

Original Sheet No. 126 original Sheet No. 126: Pending

RATE SCHEDULE ITS

Natural hereunder. Such monthly fees shall be calculated so as to permit Natural to generate monthly revenues therefrom to equal the amounts expected to be credited to Natural's cost of service or excluded from the determination of Natural's rates.

- (3) The amounts attributed by Natural to such transportation and handling for Shipper under the ITS Agreement shall be established from time to time above based upon the application of a methodology determined by Natural in good faith to be consistent with applicable orders, opinions, or decisions pertaining to such transportation and handling.
- (4) Any settlement, compromise, or other action of Natural in connection with any such order, opinion, or decision pertaining to transportation of liquids, liquefiables, plant fuel, vaporous non-hydrocarbons and/or other similar materials shall be conclusive and binding upon Shipper.
- 5.10 Natural shall perform exchanges and backhauls hereunder to the extent capacity is available. Exchanges and backhauls shall be subject to the maximum and minimum rates under this Rate Schedule except that no fuel need be charged.
- 6. NOMINATIONS, BALANCING SERVICE CHARGES, IMBALANCES AND OVERRUN CHARGES
 - (a) Shipper shall provide Natural with daily nominations of receipts and deliveries by Receipt and Delivery Point, including pooling points where applicable, in accordance with the General Terms and Conditions of this Tariff. It shall be Shipper's responsibility to cause gas to be delivered to Natural at Receipt Point(s), and to cause gas to be taken from Natural at Delivery Point(s), in accordance with the information supplied to Natural. In the event that actual receipts or deliveries at any point fail to conform to Shipper's nominations, Balancing Service Charges will apply to the extent provided in the General Terms and Conditions of this Tariff.
 - (b) It shall be Shipper's responsibility to keep receipts and deliveries in balance. Natural may curtail service hereunder to the extent necessary to bring receipts and deliveries into balance. Any imbalance between actual receipts and actual deliveries shall be eliminated by cashout on a monthly basis in accordance with the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 127 Original Sheet No. 127: Pending

RATE SCHEDULE ITS

7. RECEIPT AND DELIVERY POINTS AND UPSTREAM AND DOWNSTREAM ARRANGEMENTS

- (a) An ITS Agreement shall include all available Receipt and Delivery Points on Natural's system, as more fully set out in the General Terms and Conditions of this Tariff. Natural's aggregate maximum obligation to accept and deliver gas on an interruptible basis shall be specified in Dth in the ITS Agreement. All stated MDQs shall be exclusive of applicable Fuel Gas and Gas Lost and Unaccounted For. The volumes available at each Receipt and Delivery Point, and the related priorities, shall be governed by the General Terms and Conditions of this Tariff.
- (b) Conditions of delivery at Receipt and Delivery Points are set out in the General Terms and Conditions of this Tariff.
- (c) Shipper shall make all necessary arrangements with other parties (1) at or upstream of the Receipt Point(s) where gas is tendered to Natural hereunder and (2) at or downstream of the Delivery Point(s) where Natural delivers gas hereunder to or for the account of Shipper. Such arrangements must be consistent with this Rate Schedule ITS and must be coordinated with Natural.

8. OVERRUN SERVICE

Upon request of Shipper, Natural may (but is not obligated to) receive, transport, and deliver on any day quantities of natural gas in excess of Shipper's MDQ under the ITS Agreement when, in Natural's reasonable judgment, the capacity and operating capability of its system will permit such receipt, transportation and delivery without impairing the ability of Natural to meet its other obligations. In granting requests for Authorized Overrun Service, Natural shall act in a manner consistent with the overrun service priorities set out in the General Terms and Conditions of this Tariff. Except for Authorized Overrun Service, for which Shipper shall pay the applicable rate set out in the rate sheets of this Tariff, Shipper shall pay Natural the charges for Unauthorized Overrun set out in Section 12.2(a)(1) of the General Terms and Conditions of this Tariff if gas tendered to Natural or deliveries to Shipper under an ITS Agreement exceed the MDQ under such ITS Agreement.

9. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule ITS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Original Sheet No. 128 Original Sheet No. 128: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

AVAILABILITY

- (a) This Rate Schedule IBS is available to any entity (hereinafter called Shipper) which: (i) submits to Natural Gas Pipeline Company of America LLC (hereinafter called Natural) a valid request as defined in Section 3 hereof; (ii) executes an Interruptible Balancing Service Agreement (IBS Agreement) with Natural applicable to service under this Rate Schedule IBS; and (iii) meets the qualifications set out in subsection (b) of this Section.
- To be eligible for service under this Rate Schedule IBS, the Shipper must identify the service under its IBS Agreement as being for one specified end-use facility for each IBS Agreement; except that more than one facility may be specified to the extent limited aggregation of end-use facilities and points is permitted under Section 7(c) of this Rate Schedule IBS. Such end-use facilities need not be directly connected to Natural's system, but the conditions specified in Section 3.2(c) of this Rate Schedule IBS must be met. To be eligible for service hereunder, the Shipper must have and maintain in effect an FTS, FTS-G, or ITS Agreement to which the IBS Agreement is linked, consistent with Section 3.2(a) of this Rate Schedule IBS. The IBS Agreement must also specify a single Delivery Point for each end-use facility; provided, however, that limited aggregation of enduse facilities and points is permitted under Section 7(c) of this Rate Schedule IBS. The Delivery Point may not be a Central Delivery Point, except to the limited extent provided under Section 7(c) hereof, but may be any other primary or secondary Delivery Point available under the linked transportation Agreement; provided that the Delivery Point must be consistent with the location of the end-use facility. An IBS Agreement may only be linked to one transportation Agreement. A transportation Agreement may have more than one IBS Agreement linked to it, provided that the other requirements of this Rate Schedule IBS $\,$ are satisfied.
- (c) The form of the IBS Agreement is contained in this Tariff. There is no limitation on the number of IBS Agreements any one Shipper may have.

2. APPLICABILITY, CHARACTER AND PRIORITY OF SERVICE

(a) This Rate Schedule IBS defines an interruptible imbalance management service for specified end-use facilities. Service hereunder shall be provided pursuant to Access Requests under Section 6 of this Rate Schedule IBS. This Rate Schedule shall apply to all imbalance management services provided by Natural for Shipper pursuant to an IBS Agreement. This service will be provided based on line pack

FERC Docket: RP08-319-000

Original Sheet No. 129 Original Sheet No. 129: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

capability and on offsetting Access Requests under Section 6 of this Rate Schedule IBS. Natural is not providing a supply or storage service under this Rate Schedule IBS.

- (b) This service is designed to help a Shipper accommodate differences between nominations and physical gas flows to specified end-use facilities under the FTS, FTS-G or ITS Agreement to which the IBS Agreement is linked consistent with Section 3.2(a) of this Rate Schedule IBS.
- (c) Natural shall not be required: (1) to provide service hereunder in excess of the maximum daily access volume under the IBS Agreement as determined under Section 6(a) hereof or in excess of the maximum allowable cumulative IBS Balance as determined under Section 6(f) hereof; (2) to provide service hereunder on any day in excess of the MDQ of the IBS Agreement, reduced in the case of a swing up by the volumes nominated under the linked transportation Agreement for that day (and further reduced by releases of the linked transportation Agreement where applicable); or (3) to provide service hereunder which is not consistent with a confirmed Access Request pursuant to Section 6 of this Rate Schedule IBS.
- (d) The service provided under this Rate Schedule IBS shall be performed under Part 284 of the Commission's Regulations. Shipper shall only request service under this Rate Schedule IBS to the extent such service would qualify under the applicable statutes, regulations, Commission orders and the blanket certificate authorizing service by Natural under this Rate Schedule. For service under Subpart B of Part 284 of the Commission's Regulations, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by the Commission's Regulations, Shipper shall (prior to tendering gas under an IBS Agreement) cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit the necessary certification.
- (e) Service hereunder is provided on an interruptible basis. Priorities of service for the purposes of scheduling and curtailment are governed by the General Terms and Conditions of this Tariff. Natural will only provide service hereunder if the standard set out in Section 6(b) hereof is met.

FERC Docket: RP08-319-000

Original Sheet No. 130 original Sheet No. 130 : Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

3. VALID REQUESTS

- 3.1 A request for service under this Rate Schedule IBS shall be valid as of the date received if it complies with this Section and contains adequate information on all of the items specified in Section 3.2, subject to any necessary verification of such information and to the following:
- A request shall not be valid and Natural shall not (a) be required to grant any such request: (i) which would require the construction, modification, expansion, or acquisition of any facilities; provided, however, that Natural may agree in its reasonable discretion to construct, modify, expand, or acquire facilities to enable it to perform such services: (ii) unless and until Shipper has provided Natural with the information required in Section 3.2 hereof; (iii) if Natural determines, based on the credit analysis referenced in Section 3.2(e) hereof, that Shipper lacks the financial stability to make it reasonably likely the service provided hereunder will be paid for on a timely basis; (iv) if the service requested would not comply with this Rate Schedule IBS; or (v) if the service requested is at less than the applicable maximum rate; provided, however, that Natural may agree to provide service hereunder at a discount consistent with this Rate Schedule IBS. Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to this Section and while an IBS Agreement is in effect. Such matters are governed by Section 6(b) of this Rate Schedule IBS and the General Terms and Conditions of this Tariff.
- (b) In the event that Natural agrees to construct, modify, expand or acquire facilities to perform service under this Rate Schedule IBS, Natural shall maintain a separate record of the nature and costs of such facilities.
- (c) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request because such request is incomplete or does not comply with this Rate Schedule IBS. Any request shall be null and void unless it is substantially complete and complies with this Rate Schedule. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the IBS Agreement, after which Shipper shall have fifteen (15) days to provide the specified information. In the event such information is not received within fifteen (15) days, Shipper's request shall be null and void.

FERC Docket: RP08-319-000

Original Sheet No. 131 Original Sheet No. 131: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

(d) Natural shall tender an IBS Agreement to Shipper for execution when Shipper's request for service is accepted. Unless waived by Natural, a request for service shall be invalid if Shipper fails to execute an IBS Agreement hereunder within ten (10) days after an IBS Agreement which accurately reflects the requested service has been tendered by Natural for execution.

3.2 Requests for service hereunder shall be deemed valid only after the following information is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department at One Allen Center, 500 Dallas, Houston, Texas 77002, or Telecopy Number (713)369-9325:

(a) LINKED TRANSPORTATION AGREEMENT

Shipper shall identify in its request an FTS, FTS-G or ITS Agreement to which service hereunder will be linked. Such linkage must be consistent with the Rate Schedule IBS. The balancing service hereunder shall relate to the volumes delivered to the Delivery Point(s) on behalf of the end-use facility or facilities under the linked transportation Agreement. The effectiveness of the IBS Agreement is contingent upon Shipper having and continuing in effect the linked FTS, FTS-G or ITS Agreement, unless the parties agree on a substitute linked transportation Agreement. The linked transportation Agreement may be a released transportation Agreement but the Replacement Shipper must enter into a separate IBS Agreement; it cannot rely on the IBS Agreement of the Releasing Shipper.

(b) CONTRACT QUANTITY

The request must specify the MDQ under the IBS Agreement. The MDQ of the IBS Agreement may be equal to or less than the MDQ of the linked transportation Agreement. If the transportation Agreement is linked to more than one IBS Agreement, then the sum of the MDQs under IBS Agreements served by that linked transportation Agreement must be equal to or less than the transportation contract

FERC Docket: RP08-319-000

Original Sheet No. 132 Original Sheet No. 132: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

MDQ. The MDQ under an IBS Agreement must also be consistent with the point MDQ under the linked transportation Agreement. Subject to the above limitations, the MDQ under the IBS Agreement shall be as specified by Shipper in its request for service; provided, however, that the MDQ may not exceed the higher of: (i) the highest daily usage served by Natural at the end-use facility or facilities during the prior twelve (12) months; or (ii) the highest daily usage which will be served by Natural at such end-use facility or facilities during the ensuing twelve (12) months as reasonably projected by Shipper. The MDQ shall be redetermined at the request of either Shipper or Natural no sooner than six (6) months after the prior MDQ determination, or at any time by mutual agreement. In submitting nominations under the linked transportation Agreement, Shipper must separately identify the volumes attributed to the end-use facility or facilities covered by the IBS Agreement.

(c) END-USE FACILITY

Shipper shall specify the end-use facility or facilities for which imbalance management service is to be provided under the IBS Agreement. The IBS Agreement shall cover a single end-use facility except to the extent aggregation is permitted under Section 7(c) of this Rate Schedule IBS. Shipper shall also specify the Delivery Point for each such end-use facility. That Delivery Point must also be a point of delivery (either primary or secondary) available under the linked transportation Agreement, but may not be a Central Delivery Point, except to the extent permitted under Section 7(c) of this Rate Schedule IBS. The location of such Delivery Point(s) must be consistent with the location of the end-use facility or facilities. Electronic measurement equipment must be in place and the information must be accessible to Natural such that real time data on deliveries to the end-use facility or facilities are available to Natural.

(1) The request shall contain a statement either:
(i) that the service under the IBS Agreement will be used for the enduse facility or facilities identified in the request directly connected to Natural's system; or (ii) that service under the IBS Agreement will be used for the end-use facility or facilities identified in the request behind a Delivery Point or Points and the requirements of (2) are or will be met. If not directly connected to Natural, the end-use facility or facilities must be directly connected to facilities of the point operator which are interconnected with Natural's system facilities.

FERC Docket: RP08-319-000

Original Sheet No. 133 Original Sheet No. 133: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

connected to Natural's system, a separate IBS Monitoring Agreement must be agreed upon and executed among Natural, the Shipper, the operator of the end-use facility and the point operator and must provide (i) that the point operator will monitor the use of the service to assure that service is being provided hereunder to the end-use facility or facilities consistent with Rate Schedule IBS; and (ii) that, during periods when deliveries to the end-use facility are from more than one source, gas from Natural is deemed the first gas delivered to and used by the end-use facility unless another allocation is set out in the IBS Monitoring Agreement or is otherwise agreed upon by the parties. Natural will provide to the point operator relevant flow and nomination information related to the IBS Agreement. Service hereunder for any end-use facility not directly connected to Natural shall be contingent upon satisfaction on a continuing basis of these requirements.

(d) TERM OF SERVICE

The request shall specify:

(1) The date service is requested to commence;

and

- (2) The date service is requested to terminate.
- (e) CREDIT

 $\mbox{Acceptance of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.}$

(f) COMPLIANCE WITH IBS TARIFF

Submission of a request for service hereunder shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule IBS, including the applicable General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 134 Original Sheet No. 134: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

(g) COMMISSION-REQUIRED FILING INFORMATION

The following information is to be provided at the time a request for service hereunder is submitted, if available, or when the initial Access Request for balancing service under an executed IBS Agreement is submitted, and when any subsequent changes occur:

(1) Affiliation of the Shipper with Natural; and

(2) The identity of the Shipper, including whether it is a local distribution company, an interstate pipeline company, an intrastate pipeline company, an end user, a producer, or a marketer.

4. TERM

- (a) The term of service hereunder shall be set forth in the IBS Agreement between Shipper and Natural. The term provision must state that the IBS Agreement will terminate upon termination or expiration, and will be suspended upon release (if the release is partial, the suspension will be partial corresponding to the partial nature of the release), of the FTS, FTS-G or ITS Agreement to which the IBS Agreement is linked unless the parties mutually agree to substitute a new linked transportation Agreement. Where any specified end-use facility is not directly connected to Natural's system, the IBS Agreement will not apply to that end-use facility unless an IBS Monitoring Agreement covering that end-use facility is in effect.
- (b) Upon termination or expiration of any IBS Agreement, service by Natural to Shipper thereunder shall cease and shall be automatically abandoned.
- (c) Natural may terminate any IBS Agreement if Natural is required by the FERC or some other agency or court to provide service for others utilizing the interruptible system capacity or capability required for service under such IBS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the type covered by the IBS Agreement; provided, however, that nothing herein shall be read to protect Natural from the consequences of its own negligence.

FERC Docket: RP08-319-000

Original Sheet No. 135 Original Sheet No. 135: Pending

RATE SCHEDULE IBS
INTERRUPTIBLE BALANCING SERVICE

5. RATE

- 5.1 (a) Shipper shall pay Natural each month for service under this Rate Schedule IBS the sum of the following charges:
- $\qquad \qquad \text{(1)} \qquad \text{An Access Charge as provided in subsection} \\ \text{(b) of this Section; and} \\$
- (2) Any other applicable charges and penalties set out in this Rate Schedule IBS or the General Terms and Conditions of this Tariff (including but not limited to Balancing Service Charges under Section 12 of the General Terms and Conditions where applicable consistent with Section 6 this Rate Schedule IBS).
- (b) The Access Charge shall be the sum for the billing month of the then effective applicable daily Access Rate under the IBS Agreement multiplied each day by the absolute values of confirmed daily Access Requests for that Agreement to which that rate applies (Access Requests for swing up are not netted against Access Requests to swing down). Notwithstanding the foregoing, the Access Charge for any day shall be reduced to the extent that Natural curtails or interrupts service under Rate Schedule IBS that day, with the reduction to be calculated by multiplying the applicable daily Access Rate times that portion of the confirmed Access Request which is adversely affected by the curtailment or interruption. The maximum monthly Access Charge shall be the sum for that billing month of the applicable maximum unit daily Access Rate set out in this Tariff multiplied each day by the absolute values of confirmed daily Access Requests under the IBS Agreement to which that rate applies. The Access Charge payable by Shipper shall be the applicable maximum Access Charge unless a lower Access Charge is specified in the IBS Agreement or a separate discount agreement or is assessed under Section 5.5 of this Rate Schedule IBS, which lower charge must be consistent with the minimum rate specified in this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma Agreement set out in this Tariff, discounts may be limited to specific volumes.
- (c) For any volumes of service provided hereunder in excess of Shipper's maximum daily access volume as determined under Section 6(a) hereof, or any IBS Balance in excess of the maximum allowed cumulative IBS Balance under Section 6(f), Shipper shall pay either an authorized overrun rate, or an unauthorized overrun rate consistent with Section 8 of this Rate Schedule.

FERC Docket: RP08-319-000

Original Sheet No. 136 original Sheet No. 136: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

- (d) Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.
- 5.2 (a) Shipper shall reimburse Natural within five (5) days after costs have been incurred by Natural for all fees required by the Commission or any regulatory body including, but not limited to, filing, reporting, and application fees, to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees.
- (b) If Natural constructs, acquires or modifies any facilities to perform service hereunder, then as specified in an agreement between the parties either:
- (i) Shipper shall reimburse Natural for the cost of such facilities or facility modifications as described in the General Terms and Conditions of this Tariff; or
- $\hbox{(ii)}\quad \hbox{Natural shall assess a monthly charge}\\ \hbox{reflecting such facility costs.}$
- 5.3 Shipper shall pay any transportation charges associated with effectuating the receipt or delivery of gas hereunder pursuant to the FTS, FTS-G or ITS Agreement to which Shipper's IBS Agreement is linked, including any applicable Fuel Gas or Gas Lost and Unaccounted For, in accordance with the terms of that separate linked transportation Agreement. Shipper shall pay a transportation charge and related Fuel Gas and Gas Lost and Unaccounted For once, but only once, under the linked transportation Agreement associated with any IBS Balance as defined in Section 6(b) of this Rate Schedule IBS. Shipper shall not pay any charge associated with IBS Payback except to the extent transportation charges under the linked transportation Agreement are required to satisfy the prior sentence. Any ACA, RD&D, transition cost charges or other generally applicable surcharges shall also be assessed under such linked transportation Agreement and no additional such charges shall be assessed under Rate Schedule IBS.

FERC Docket: RP08-319-000

Original Sheet No. 137 Original Sheet No. 137: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

- 5.4 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in: (1) the rates and charges applicable under this Rate Schedule IBS, including both the level and design of such rates and charges; or (2) the terms and conditions of this Rate Schedule IBS. Natural agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.
- (b) If, at any time and from time to time, the Commission or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the IBS Agreement or a separate discount agreement, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such highest rate, Shipper shall execute or provide such documentation within fifteen (15) days after a written request by Natural. If, at any time and from time to time, the Commission or any other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for balancing service hereunder, the rate shall be decreased to such reduced rate.
- 5.5 Natural may from time to time and at any time, upon twenty-four (24) hours verbal or written notice, subject to any provisions on discounting in the IBS Agreement or in a separate discount agreement, charge any individual Shipper for service under this Rate Schedule IBS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate charged may not be less than the applicable minimum rate for service under Rate Schedule IBS set forth in this Tariff. Natural will confirm any verbal notice of the applicable rate in writing. Such notification shall specifically

FERC Docket: RP08-319-000

Original Sheet No. 138 Original Sheet No. 138: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

state the effective date of such rate change and the quantity of gas so affected. Unless otherwise agreed in the IBS Agreement or in a separate discount agreement, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the IBS Agreement and/or any discount agreement) upon twenty-four (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with the Commission any and all reports as required by the Commission's Regulations with respect to the institution or discontinuance of any discount.

- 6. ACCESS REQUESTS, PAYBACK NOMINATIONS, BALANCING SERVICE CHARGES, OVERRUN CHARGES, AND CASHOUT.
 - Subject to the remainder of this Section 6, on any (a) day, Shipper may submit an Access Request for service under its IBS Agreement up to its MDQ; provided, however, that unless Natural otherwise agrees, the net effect (taking into account the direction of the request) of such Access Request for any day may only result in service hereunder equal in volume to the difference between Shipper's then effective MDQ under the IBS Agreement (reduced by releases of the linked transportation Agreement as set out below) and Shipper's nominations to serve the specified end-use facility or facilities that day under the linked transportation Agreement (maximum daily access volume). If Shipper has released any part of a linked firm transportation Agreement, such release(s) shall be taken into account in determining the level of service available under the IBS Agreement if the release(s) reduce the effective MDQ available to Shipper for nomination under the linked transportation Agreement to the designated Delivery Point(s) below the MDQ under the IBS Agreement. Except for authorized overrun, an Access Request for a swing up may not exceed the unnominated portion of the MDQ under the linked transportation Agreement (reduced as applicable by any releases).

FERC Docket: RP08-319-000

Original Sheet No. 139 Original Sheet No. 139: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

- (1) The Access Request, which shall constitute Shipper's nomination under the IBS Agreement, shall specify the volume and direction of the balancing service and must reference the specific IBS Agreement, end-use facility or facilities and Delivery Point(s) to which the service relates.
- (2) For a Shipper to request service hereunder, that Shipper must have in effect a linked FTS, FTS-G or ITS Agreement consistent with Section 3.2(a) of this Rate Schedule IBS. The Access Request must reference the linked transportation Agreement.
- (3) Intraday Access Requests may be submitted hereunder which increase the level of service requested. Intraday Access Requests may also be submitted which decrease the level of service requested, but such a decrease shall not reduce the Access Charges based on the original Access Request confirmed for the decreasing Shipper that day unless Natural obtains additional revenue from another Shipper under Rate Schedule IBS as the result of an intraday nomination for an increase Natural can only confirm because of the decrease. The reduction in Access Charges for the decreasing Shipper shall be limited to the additional revenue which Natural obtains from the intraday nomination for an increase in service under Rate Schedule IBS confirmation of which is made possible by the decrease, but in no event shall the reduction result in negative Access Charges for the decreasing Shipper. Such intraday Access Requests may be submitted on the same schedule as other intraday nominations.
- (4) For each major pipeline segment, Natural will post on each day, at least two (2) hours before nominations are due for that day, whether service under Rate Schedule IBS is likely to be available on that pipeline segment.
- (b) Natural shall confirm Access Requests made under an IBS Agreement which otherwise are consistent with this Rate Schedule IBS if, in Natural's reasonable judgment, Natural has the line pack capability to provide such service, without detriment to satisfaction of Natural's firm obligations or any interruptible obligations which may be of higher priority or to Natural's system operational needs, it being understood that Natural is not providing a supply or storage service under this Rate Schedule IBS. Natural will not confirm any Access Request if granting such a request would be detrimental to Natural's ability to provide any firm service or any priority interruptible rights associated with a

FERC Docket: RP08-319-000

Original Sheet No. 140 original Sheet No. 140: Pending

RATE SCHEDULE IBS
INTERRUPTIBLE BALANCING SERVICE

firm service [the latter are identified as exceptions to interruptible priorities at Section 5.7(a) of the General Terms and Conditions of this Tariff]. Natural will also not grant any Access Request if doing so would materially increase the likelihood that Natural would declare a Critical Time or issue an Operational Flow Order affecting services identified in the prior sentence.

(1) Natural will attempt to offset Access Requests hereunder in one direction with Access Requests in the opposite direction, but Natural will only do so if the general

FERC Docket: RP08-319-000

Original Sheet No. 141 Original Sheet No. 141: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

standards of this subsection (b) can be met and Natural will not confirm Access Requests if the impact is detrimental to Natural's system.

- (2) A confirmed Access Request shall be deemed authorization, up to the volume specified in the Access Request, for Natural to provide imbalance management service hereunder related to the linked transportation Agreement without further notice. Shipper shall not, however, be obligated to use such balancing volumes; i.e., Access Requests are used to define the level of balancing service available to the Shipper that day to cover IBS Balance volumes allocated under subparagraph (b) (3). This service is subject to the curtailment provisions of the General Terms and Conditions of this Tariff.
- (3) Natural shall account separately under each IBS Agreement for all nominations to each specified end-use facility and for volumes actually utilized at each end-use facility; provided, however, that to the extent aggregation is permitted under Section 7(c) hereof, Natural shall account for all aggregated facilities on an aggregated basis. Any difference on any day between nominations and physical gas flow under the linked FTS, FTS-G or ITS Agreement to the specified end-use facility or facilities in the direction of swing specified in the Access Request for that day shall be allocated to the IBS Agreement unless otherwise expressly agreed. The resulting volume, which may be positive or negative, is the IBS Balance. Such differences in the opposite direction from the Access Request shall be allocated to the linked transportation Agreement or to other services contracted by Shipper pursuant to a predetermined allocation Agreement. Service hereunder allows Shipper to have daily imbalances or variances within the level of the confirmed Access Request. $\label{total conditions} \mbox{Volumes allocated to the IBS Agreement hereunder shall be treated as} \\$ unauthorized overrun if the resulting IBS Balance is not within the volume level consistent with Shipper's confirmed Access Request under this Section 6. Volumes allocated to the IBS Agreement hereunder in excess of the maximum daily access volume or the maximum allowable cumulative IBS Balance which are consistent with the confirmed Access Request shall be authorized overrun under Section 8 of this Rate Schedule IBS.
- (c) Natural shall provide to Shipper its best information on the IBS Balances under an IBS Agreement by at least one half hour prior to the time timely nominations are due for the next Gas Day. Natural shall identify both the IBS Balance for the day and the cumulative IBS Balance. In the event that an allocation under Section 6(b) results in Shipper having a positive or negative IBS

FERC Docket: RP08-319-000

Original Sheet No. 142 Original Sheet No. 142: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

Balance, Shipper shall be obligated to make a timely IBS Payback nomination under the linked transportation Agreement based on such timely information provided by Natural in the next daily nomination cycle up to its MDQ under the linked transportation $\mbox{\tt Agreement}$ to bring its IBS Balance for a day to zero (IBS Payback nomination). Such IBS Payback nominations may also be made at the time intraday nominations are due under the next daily nomination cycle. Natural may adjust such IBS Balance information in future periods and Shipper's future payback obligations shall be adjusted accordingly, but Shipper may rely on any information provided by Natural in making IBS Payback nominations. IBS Payback nomination shall be separately identified and shall be a volume which is in excess of Shipper's projected requirements for that day at the relevant end-use facility or facilities. The IBS Payback nomination may represent (in whole or in part) overrun under the linked transportation Agreement, in which case the overrun shall be evaluated by Natural for scheduling purposes on the same basis as other overrun requests.

(1) IBS Payback nominations must reference the IBS Agreement. IBS Payback must be made to deliver gas to or take gas away from Delivery Points, as applicable, under the linked transportation Agreement, or at other mutually agreed points if delivery to those points would be beneficial to Natural's system, to achieve IBS Payback in relation to the Delivery Point(s). IBS Payback nominations shall be evaluated for scheduling and curtailment priority purposes on the same basis as other nominations under the linked transportation Agreement or at the mutually agreed points. Transportation charges shall be assessed under the linked transportation Agreement only if movement of the gas to the Delivery Point(s) has not previously been subject to a transportation charge, consistent with Section 5.3 of this Rate Schedule IBS.

Payback. To the extent Natural does not confirm an IBS Payback nomination, or the nomination is limited by MDQ under the IBS Agreement, Shipper shall be obligated to make a timely IBS Payback nomination in each subsequent daily nomination period until the IBS Balance for that day has been eliminated; provided, however, that if Natural advises Shipper that no IBS Payback nominations will be accepted until a specified future date or will be limited in a specified way over a defined period, Shipper may forego or limit its IBS Payback nomination accordingly.

FERC Docket: RP08-319-000

Original Sheet No. 143 Original Sheet No. 143: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

- (d) If Shipper fails to make a timely and valid IBS Payback nomination to reduce its IBS Balance for a day to zero, then the Balancing Service Charges in Section 12 of the General Terms and Conditions of the Tariff shall apply, with no tolerance or exemption, based on the IBS Balance information provided to Shipper by Natural. Such charges will continue as applied to the remaining volume of the IBS Balance for that day until valid IBS Payback nominations adequate to eliminate the IBS Balance have been submitted under this Section 6. Such Balancing Service Charges shall also apply, with no tolerance or exemption, to the extent Shipper makes a timely and valid IBS Payback nomination which is confirmed by Natural, but IBS Payback is not actually effectuated by Shipper. The phrase "with no tolerance or exemption" as used herein shall mean the following: under Section 12.2(a)(2) of the General Terms and Conditions of this Tariff, the 0% to 5% tier does not apply and the lowest charge (\$0.10/Dth) applies to variances between 0% and 10%; under Section 12.3(a) of the General Terms and Conditions of this Tariff, the 0% to 3% tier does not apply and the lowest charge (greater of \$1.00 or 50% of AMIP) applies to variances between 0% and 10%; and the exemptions in Sections 12.6 and 23.6(g) of the General Terms and Conditions of this Tariff do not apply. Each day for which an IBS Balance accrues is accounted for separately for purposes of applying Balancing Service Charges and the other provision of this Section 6. IBS Payback nominations shall apply first to the oldest day's IBS Balance. No Balancing Service Charges shall apply for any day, however, if Shipper makes a valid, timely and adequate IBS Payback nomination for that day within the limit specified in subsection (c) above, but that IBS Payback nomination is not confirmed by Natural for reasons other than that the nomination was not proper. Intraday IBS Payback nominations will not avoid Balancing Service Charges as to volumes not timely nominated unless confirmed by Natural and unless adequate under (c) above.
- (e) A Shipper which is submitting an IBS Payback nomination on any day may also on that day submit an Access Request for additional service under this Rate Schedule IBS. Subject to subsection (f) below, such Access Requests will be processed under and are subject to the same requirements and limitations as are imposed on all Access Requests by Sections 6(a) and 6(b) of this Rate Schedule IBS.
- (f) The cumulative IBS Balance under an IBS Agreement may not exceed two (2) times Shipper's MDQ, absent agreement by Natural.
- (g) Cashout provisions of the Tariff shall only apply to IBS Balances to the extent provided in Section 7(b) of this Rate Schedule. All IBS Balances must be zeroed out through the Access Request and IBS Payback nomination process in this Section 6 except as otherwise provided in Section 7(b) of this Rate Schedule.

FERC Docket: RP08-319-000

Original Sheet No. 144 Original Sheet No. 144: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

- 7. UPSTREAM AND DOWNSTREAM ARRANGEMENTS, END-OF-TERM IMBALANCE AND AGGREGATION
 - (a) Shipper shall make all necessary arrangements with other parties at or downstream of the Delivery Points under the IBS Agreement. Such arrangements must be consistent with this Rate Schedule IBS and the linked FTS, FTS-G or ITS Agreement and must be coordinated with Natural.
 - (b) Shipper must have a zero IBS Balance at the end of the term of the IBS Agreement. If an IBS Balance remains at the end of the term, the remaining volumes shall be cashed out as set out at Section 13.3 of the General Terms and Conditions of this Tariff. Shipper and Natural may also mutually agree to cashout an IBS Balance at the end of any month under an IBS Agreement consistent with said Section 13.3.
 - (c) Except with prior agreement by Natural on a case-by-case basis, an IBS Agreement may cover only a single end-use facility and a single related Delivery Point.
 - (1) Natural will agree to limited aggregation of enduse facilities and Delivery Points under a single IBS Agreement pursuant to this subsection (c) if the following criteria are met:
 - (i) all the end-use facilities must have the same owner or operator (or a common agent consistent with this provision);
 - $\hspace{1.5cm} \mbox{(ii)} \hspace{0.3cm} \mbox{all the end-use facilities must be of the same type; } \\$
 - (iii) it must be operationally feasible for Natural to provide service on an aggregated basis to all such end-use facilities and points consistent with this Rate Schedule IBS; and
 - (iv) the number of end-use facilities to be aggregated may not exceed five (5) and the number of related points to be aggregated may also not exceed five (5).

FERC Docket: RP08-319-000

Original Sheet No. 145 original Sheet No. 145: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

(2) Natural will allow such aggregation with a common agent for the end-use facilities, provided that the other requirements of aggregation are satisfied. Such an agency relationship would be subject to satisfaction of the conditions set out in Section 8.9 of the General Terms and Conditions of this Tariff and the following more specific conditions: the agent must meet the same creditworthiness standards as a Shipper, Natural can rely upon representations by the agent until Natural receives written notice that the agency relationship has been terminated and aggregation will cease to be effective as of the date on which there is no longer a common agent.

(3) If the above criteria are met, Natural will permit the aggregation under a single IBS Agreement of up to five (5) end-use facilities at different points. In addition, if all the above criteria are met, Natural will aggregate up to five (5) end-use facilities behind a single Central Delivery Point under a single IBS Agreement. The aggregated end-use facilities shall be accounted for in the aggregate, by use of a paper point. A Shipper may not aggregate in a single IBS Agreement both directly connected end-use facilities and end-use facilities which are not directly connected to Natural or which are behind a Central Delivery Point.

8. OVERRUN SERVICE

Upon request of Shipper, Natural may (but is not obligated to) provide service hereunder in excess of Shipper's maximum daily access volume under the IBS Agreement as determined under Section 6(a) hereof or Shipper's maximum allowed cumulative IBS Balance as determined under Section 6(f) hereof when, in Natural's reasonable judgment, its line pack capability or offsetting Access Requests will permit such balancing receipt or delivery without impairing the ability of Natural to meet its firm or other higher priority interruptible obligations, it being understood that Natural is not providing a supply or storage service under this Rate Schedule IBS. In granting requests for authorized overruns, Natural shall act in a manner consistent with the overrun service priorities set out in the General Terms and Conditions of this Tariff. Except for authorized overruns as reflected in confirmation of an Access Request, for which Shipper shall pay the applicable rate set out in the rate sheets of this Tariff, Shipper shall pay Natural the charges for unauthorized overrun set out in Section 12.2(a)(1) of the General Terms and Conditions of this Tariff if balancing volumes in the direction of swing specified in the Access Request under an IBS Agreement exceed the volumes specified in the Access Request or,

FERC Docket: RP08-319-000

Original Sheet No. 146 Original Sheet No. 146: Pending

RATE SCHEDULE IBS INTERRUPTIBLE BALANCING SERVICE

unless consistent with the Access Request, exceed the maximum daily access volumes under such IBS Agreement or the maximum allowed cumulative IBS Balance. Natural may waive or discount either authorized or unauthorized overrun charges on a basis which is not unduly discriminatory. With respect to authorized overrun charges associated with exceeding the allowed cumulative IBS Balance, it will be Natural's general policy to waive such charges if the charge results from Natural's decision not to accept IBS Payback and if maintaining the IBS Balance is beneficial to Natural's system.

9. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule IBS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Original Sheet No. 147 Original Sheet No. 147: Pending

RATE SCHEDULE BESS
BEST EFFORTS STORAGE SERVICE

1. AVAILABILITY

- (a) This Rate Schedule BESS is available for interruptible storage service to any entity (hereinafter called Shipper) which:
 (i) submits to Natural Gas Pipeline Company of America LLC (hereinafter called Natural) a valid request as defined in Section 4 hereof; and (ii) executes a Best Efforts Storage Service Agreement (BESS Agreement) with Natural applicable to service under this Rate Schedule BESS. The form of BESS Agreement is contained in this Tariff. There is no limitation on the number of BESS Agreements any one Shipper may have.
- (b) Shipper shall arrange separately with Natural and others as necessary for any transportation attendant to the storage services provided hereunder and Shipper shall pay separately for such transportation service.

2. APPLICABILITY AND CHARACTER OF SERVICE

- (a) This Rate Schedule BESS, which defines an interruptible storage service, shall apply to all gas received by Natural from or for the account of Shipper pursuant to a BESS Agreement. As more fully set out in the General Terms and Conditions of this Tariff, Natural is not providing a supply service under this Rate Schedule BESS.
- (b) Service hereunder shall consist of the acceptance by Natural of natural gas tendered hereunder by or on behalf of Shipper for injection into storage as specified in the BESS Agreement and the delivery by Natural of the natural gas withdrawn from storage pursuant to this Rate Schedule BESS to Shipper or for Shipper's account. Natural shall not be required to accept or deliver on any day any gas tendered or requested in excess of the Maximum Daily Quantity (MDQ) or Maximum Available Capacity (MAC) specified in the BESS Agreement or to accept or deliver any gas which is not properly nominated pursuant to and to the extent required by the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 148 Original Sheet No. 148: Pending

RATE SCHEDULE BESS

(c) The service provided under this Rate Schedule BESS shall be performed under Part 284 of the Commission's Regulations. Shipper shall only tender gas under this Rate Schedule BESS to the extent service hereunder would qualify under the applicable statutes, regulations, Commission orders and the blanket certificate authorizing service by Natural under this Rate Schedule. For service under Subpart B of Part 284 of the Commission's Regulations, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by the Commission's Regulations, Shipper shall (prior to tendering gas for storage under a BESS Agreement) cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit the necessary certification.

3. PRIORITY OF SERVICE

- (a) Service hereunder is provided on an interruptible basis. Curtailment and priorities of service for the purposes of scheduling and curtailment shall be governed by the General Terms and Conditions of this Tariff.
- (b) Natural may at any time notify Shipper of a curtailment in accordance with the General Terms and Conditions of this Tariff, in which event Shipper must immediately cease deliveries to or receipt of gas from Natural, in whole or in part as directed by Natural hereunder, and may be required by Natural to withdraw all or any part of the gas being held by Natural for Shipper's account. Natural shall specify the quantities to be curtailed in its order. If Shipper is required by Natural to withdraw gas being held by Natural for its account, Shipper shall be required to accept such gas over a period of time determined by dividing (i) the volume of gas Shipper is required to withdraw by (ii) 50% of Shipper's MDQ. If Natural is unable to withdraw any volumes requested by Shipper within Shipper's MDQ, this period shall be extended accordingly.
- (c) If Shipper fails to comply with any curtailment order of Natural under this Section 3, Shipper shall pay Natural a penalty of \$5.00 per Dth for all tenders or takes which varies by more than 10% each day from the quantities specified in such curtailment order. In addition, if Shipper fails to withdraw gas during the period designated in subsection (b) hereof, any remaining balance as of the end of that period shall be forfeited to Natural.

FERC Docket: RP08-319-000

Original Sheet No. 149 Original Sheet No. 149: Pending

RATE SCHEDULE BESS

4. VALID REQUESTS

- 4.1 A request for service under this Rate Schedule BESS shall be valid as of the date received if it complies with this Section 4 and contains adequate information on all of the items specified in Section 4.2, subject to any necessary verification of such information and to the following:
- (a) a request shall not be valid and Natural shall not be required to grant any such request: (i) which could in Natural's judgment interfere with efficient operation of its system or with service to any firm Shipper; (ii) which would require the construction, modification, expansion, or acquisition of any facilities; provided, however, that Natural may agree in its reasonable discretion to construct, modify, expand, or acquire facilities to enable it to perform such services; (iii) unless and until Shipper has provided Natural with the information required in Section 4.2 hereof; (iv) if Natural determines, based on the credit analysis referenced in Section 4.2(d), that Shipper does not possess sufficient financial stability to make it reasonably likely the service provided hereunder will be paid for on a timely basis; (v) if the service requested would not comply with this Rate Schedule BESS; or (vi) if the service requested is at less than the applicable maximum rate; provided, however, that Natural may agree to provide service hereunder at a discount consistent with this Rate Schedule BESS. Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to Section 4 hereof and while a BESS Agreement is in effect. Such matters are governed by Section 3 of this Rate Schedule and the applicable General Terms and Conditions of this Tariff.
- (b) In the event that Natural agrees to construct, modify, expand or acquire facilities to perform service under this Rate Schedule BESS, Natural shall maintain a separate record of the nature and costs of such facilities.
- (c) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request, in whole or in part. Any request shall be null and void unless it is substantially complete and complies with this Rate Schedule. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the BESS Agreement, after which Shipper shall have fifteen (15) days to provide the specified information. In the event such information is not received within fifteen (15) days, Shipper's request shall be null and void.

FERC Docket: RP08-319-000

Original Sheet No. 150 original Sheet No. 150 : Pending

RATE SCHEDULE BESS

(d) Natural shall tender a BESS Agreement to Shipper for execution when Shipper's request for service is accepted. Unless waived by Natural, a request for service shall be invalid if Shipper fails to execute a BESS Agreement hereunder within ten (10) days after the BESS Agreement has been tendered by Natural for execution.

4.2 Requests for service hereunder shall be deemed valid only after the following information is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department, One Allen Center, 500 Dallas Street, Houston, Texas 77002, or Telecopy Number (713) 369-9305:

(a) GAS QUANTITIES

The request shall specify in Dth the aggregate MAC and the aggregate MDQ; provided, however, that Natural shall not be obligated to accept requests for an MDQ of less than one hundred (100) Dth per day or an MAC of less than five hundred (500) Dth. All stated MDQs or MACs shall be exclusive of Storage Fuel.

(b) POINTS

The receipt points for injection and the delivery points for withdrawal under a BESS Agreement shall be the storage points defined in the General Terms and Conditions of this Tariff.

(c) TERM OF SERVICE

The request shall specify:

(1) The date service is requested to commence;

and

(2) The date service is requested to terminate.

(d) CREDIT

 $\label{lem:Acceptance of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.$

FERC Docket: RP08-319-000

Original Sheet No. 151 Original Sheet No. 151: Pending

RATE SCHEDULE BESS

(e) COMPLIANCE WITH BESS TARIFF

Submission of a request for service hereunder shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule BESS, including the applicable General Terms and Conditions.

(f) COMMISSION-REQUIRED FILING INFORMATION

The following information is to be provided at the time a request for service hereunder is submitted, if available, or when an initial nomination for storage under an executed BESS Agreement is submitted, and when any subsequent changes occur:

- (1) Affiliation of the Shipper with Natural; and
- (2) The identity of the Shipper, including whether it is a local distribution company, an interstate pipeline company, an intrastate pipeline company, an end user, a producer, or a marketer.

5. TERM

- (a) The term of service hereunder shall be set forth in the BESS Agreement between Shipper and Natural.
- (b) The General Terms and Conditions of this Tariff shall govern the applicability of, and the terms and conditions relating to, rollovers of a BESS Agreement. Upon termination of any BESS Agreement, and subject to such rollover, service by Natural to Shipper thereunder shall be terminated and automatically abandoned.

FERC Docket: RP08-319-000

Original Sheet No. 152 Original Sheet No. 152: Pending

RATE SCHEDULE BESS

(c) Natural may terminate any BESS Agreement if Natural is required by the FERC or some other agency or court to provide service for others utilizing the interruptible system capacity or capabilities required for service under such BESS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the type covered by the BESS Agreement; provided, however, that nothing herein shall be read to protect Natural from the consequences of its own negligence.

6. RATE

 $6.1\,$ (a) For storage service rendered to Shipper under this Rate Schedule BESS, Shipper shall pay Natural each month the sum of the following charges:

(1) CAPACITY CHARGE

A monthly charge per Dth determined by multiplying: (i) the weighted average daily balance of gas held by Natural for Shipper's account under a BESS Agreement that month times (ii) the Capacity Rate.

(2) STORAGE FUEL CHARGE

Shipper shall provide gas in kind for the fuel required to provide injections and withdrawals under this Rate Schedule BESS. Such charge shall be determined by multiplying percentage fuel rates by the volume of gas injected or withdrawn. Specifically, the Storage Fuel Charge shall equal the sum of (i) the applicable unit Injection Storage Fuel Rate set out in this Tariff multiplied by each Dth of gas injected under the BESS Agreement and (ii) and the applicable unit Withdrawal Storage Fuel Rate set out in this Tariff multiplied by each Dth of gas withdrawn under the BESS Agreement. Upon the mutual agreement of Natural and Shipper, in lieu of Natural retaining gas in kind, Shipper shall reimburse Natural for fuel at a price established under Section 7.1 of the General Terms and Conditions of this Tariff multiplied by the Storage Fuel Charge.

FERC Docket: RP08-319-000

Original Sheet No. 153 original Sheet No. 153: Pending

RATE SCHEDULE BESS

(3) OTHER CHARGES

Other applicable charges and penalties set out in this Rate Schedule or the General Terms and Conditions of this Tariff.

- (b) Charges payable by any Shipper shall be based on the maximum rates set forth in this Tariff applicable to Rate Schedule BESS, which rates are hereby incorporated herein, unless a lower rate for any of the charges in subsection (a) is specified in the BESS Agreement or a separate discount agreement or is assessed pursuant to Section 6.4 of this Rate Schedule. Such lower rate must be consistent with the minimum rate specified in this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma Agreement set out in this Tariff, discounts may be limited to specific volumes. For any volumes delivered in excess of Shipper's MDQ or MAC, which deliveries shall be subject to Section 10 of this Rate Schedule BESS, Shipper shall pay either the Authorized Overrun Rate or the Unauthorized Overrun Rate specified in Section 12 of the General Terms and Conditions of this Tariff, as applicable.
- (c) Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.
- (d) The charges referenced herein cover only storage services. Any associated transportation provided by Natural and/or others shall be billed separately.
- 6.2 (a) Shipper shall reimburse Natural within five (5) days after costs have been incurred by Natural for all fees required by the FERC or any regulatory body including, but not limited to, filing, reporting, and application fees to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees (such as general rate case filing fees). Any costs recovered hereunder shall be excluded from Natural's cost of service in determining generally applicable rates.

FERC Docket: RP08-319-000

Original Sheet No. 154 Original Sheet No. 154: Pending

RATE SCHEDULE BESS

(i) Shipper shall reimburse Natural for the cost of such facilities or facility modifications as described in the General Terms and Conditions of this Tariff; or

 $\ensuremath{\mbox{(ii)}}$ Natural shall assess a monthly charge reflecting such facility costs.

FERC Docket: RP08-319-000

Original Sheet No. 155 Original Sheet No. 155: Pending

RATE SCHEDULE BESS

- 6.3 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in (1) the rates and charges applicable under this Rate Schedule BESS, including both the level and design of such rates and charges; or (2) the terms and conditions of this Rate Schedule BESS. Natural agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.
- (b) If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the BESS Agreement or a separate discount agreement, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such highest rate, Shipper shall execute or provide such documentation within fifteen (15) days after a written request by Natural. If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for transportation service hereunder, the rate shall be decreased to such reduced rate.
- 6.4 Natural may from time to time and at any time, upon twenty-four (24) hours' verbal or written notice, subject to any provisions on discounting in the BESS Agreement or in a separate discount agreement, charge any individual Shipper for service under this Rate Schedule BESS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate charged may not be less than the applicable minimum rate for service under Rate Schedule BESS set forth in this Tariff. Natural will confirm any verbal notice of the applicable charge in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Unless otherwise agreed in the BESS Agreement or in a separate discount agreement, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the BESS agreement and/or a discount agreement) upon twentyfour (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with the Commission any and all reports as required by the Commission's Regulations with respect to the institution or discontinuance of any discount.

FERC Docket: RP08-319-000

Original Sheet No. 156 original Sheet No. 156: Pending

RATE SCHEDULE BESS

6.5 All revenues and gas in kind collected by Natural as a result of providing service under this Rate Schedule BESS shall be retained by Natural unless Natural has otherwise explicitly agreed on a different disposition of such amounts. Where crediting or refund mechanisms apply under other provision(s) of this Tariff or pursuant to effective Commission orders or settlements, such mechanisms shall supersede this Section to the extent necessary to carry out such provision(s).

7. MAXIMUM DAILY QUANTITY

The BESS Agreement shall specify in Dth an aggregate MDQ and an MAC. All stated MDQs and MACs shall be exclusive of applicable Storage Fuel. The MDQ shall be the maximum volume Natural is obligated, on an interruptible basis, to receive from or deliver to Shipper hereunder on any day. The MAC shall be the maximum volume Natural is obligated to hold for the account of Shipper hereunder on an interruptible basis.

8. NOMINATIONS

Shipper shall provide Natural with daily nominations of injections and withdrawals in accordance with the General Terms and Conditions of this Tariff. It shall be Shipper's responsibility to cause gas to be delivered to Natural for injection into storage hereunder and to cause gas to be taken from Natural through storage withdrawals hereunder in accordance with the information supplied to Natural.

9. RECEIPT AND DELIVERY POINTS AND UPSTREAM AND DOWNSTREAM ARRANGEMENTS

- (a) The receipt points for gas injected hereunder and the delivery points for gas withdrawn hereunder shall be storage points defined in the General Terms and Conditions of this Tariff. Each BESS Agreement shall have as receipt and delivery points all such storage points and will be subject to the limitations and conditions (including volumes and priorities) specified in the General Terms and Conditions of this Tariff relating to such points. Volumes tendered for injection at any storage point must be withdrawn at the same storage point.
- (b) Natural may utilize any or all of its storage and related facilities to provide the service described in this Rate Schedule, and the capacity available to provide such service shall be based on the capacity and functioning of all such storage facilities.

FERC Docket: RP08-319-000

Original Sheet No. 157 Original Sheet No. 157: Pending

RATE SCHEDULE BESS

(c) Shipper shall arrange separately for the transportation of gas to and from storage and shall make all necessary arrangements with other parties at or upstream of the point(s) where gas is tendered to Natural hereunder for injection into storage and downstream of the point where Natural delivers gas withdrawn from storage hereunder by or on behalf of Shipper. Such arrangements must be consistent with this Rate Schedule BESS and must be coordinated with Natural. To be eligible for service hereunder, Shipper must comply with all applicable provisions of such separate transportation arrangements and any applicable Rate Schedule and General Terms and Conditions.

10. OVERRUN SERVICE

Upon request of Shipper, Natural may (but is not obligated to) receive for injection or withdraw for delivery hereunder on any day quantities of natural gas in excess of Shipper's MDQ and/or MAC under the BESS Agreement when, in Natural's reasonable judgment, the capacity and operating capability of its system will permit such receipt or delivery without impairing the ability of Natural to meet its other service obligations; provided, however, in no event will Shipper be allowed to withdraw more gas than it has in storage. In granting requests for ${\tt Authorized}$ ${\tt Overrun}$ Service, Natural shall act in a manner consistent with the overrun service priorities set out in the General Terms and Conditions of this Tariff. Except for Authorized Overrun Service, for which Shipper shall pay the applicable rate set out on the rate sheets of this Tariff, Shipper shall pay Natural the charges for Unauthorized Overrun set out in Section 12.2(a)(1) of the General Terms and Conditions of this Tariff if injections or withdrawals under a BESS Agreement exceed the MDQ or MAC under such BESS Agreement.

11. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule BESS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Original Sheet No. 158 Original Sheet No. 158: Pending

RATE SCHEDULE DSS DELIVERED FIRM STORAGE SERVICE

1. AVAILABILITY

- (a) This Rate Schedule DSS is available for delivered firm storage service and for nominated firm transportation service (the latter limited to October 15 through April 15) to any entity (hereinafter called Shipper) which: (1) submits to Natural Gas Pipeline Company of America LLC (hereinafter called Natural) a valid request as defined in Section 3 hereof which Natural has capacity available on all affected portions of its system and the operational capability to satisfy; and (2) executes a Delivered Firm Storage Service Agreement (DSS Agreement) with Natural applicable to service under this Rate Schedule DSS. The form of DSS Agreement is contained in this Tariff. There is no limitation on the number of DSS Agreements any one Shipper may have.
- (b) Shipper shall arrange separately with Natural or others as necessary for any transportation attendant to the delivery of gas to storage in connection with delivered firm storage service hereunder. Such delivered firm storage service encompasses the delivery of gas from storage to Shipper at Delivery Point(s). A Shipper may also utilize separately, during the period from October 15 through April 15, the firm transportation embedded in such delivered firm storage service on a nominated basis as set out in this Rate Schedule DSS.

2. APPLICABILITY, CHARACTER AND PRIORITY OF SERVICE

(a) This Rate Schedule DSS defines a delivered firm storage service, to be available on a no-notice basis within the rights defined herein, which includes transportation from storage to Shipper's Delivery Point(s). Firm transportation may also be used hereunder separately, during the period of October 15 through April 15, on a nominated basis. This Rate Schedule DSS shall apply to all gas received by Natural from or for the account of Shipper pursuant to a DSS Agreement. As more fully set out in the General Terms and Conditions of this Tariff, Natural is not providing a gas supply service under this Rate Schedule DSS.

FERC Docket: RP08-319-000

Original Sheet No. 159 original Sheet No. 159: Pending

RATE SCHEDULE DSS

- Delivered firm storage service hereunder shall consist of the acceptance by Natural of natural gas from or for the account of Shipper for injection into storage hereunder, the storage of such gas, and the delivery to Shipper or for Shipper's account of natural gas withdrawn from storage hereunder by Natural at the primary Delivery Point(s) specified in the DSS Agreement, or at secondary Delivery Points applicable to such DSS Agreement. Storage Fuel and Fuel Gas and Gas Lost and Unaccounted For shall be provided by Shipper in connection with such service as set out herein. Shipper may utilize separately all or part of the firm transportation $% \left(\left(1\right) \right) =\left(1\right) \left(\left(1\right) \right) \left(1\right) \left(1\right)$ component of service hereunder within the limitations set out in this Rate Schedule DSS. Nominated firm transportation service hereunder shall consist of the acceptance by Natural of gas properly nominated by Shipper (such service may only be nominated from October 15 through April 15) and tendered hereunder for transportation pursuant to confirmed nominations at the Receipt Point(s) applicable to such service, the transportation of such gas, and the delivery to Shipper or for Shipper's account of such gas, in accordance with Shipper's confirmed nomination, adjusted for Fuel Gas and Gas Lost and Unaccounted For as provided herein, at the primary Delivery Point(s) under the DSS Agreement or at any secondary Delivery Point(s) applicable to such DSS Agreement.
- With respect to the delivered firm storage service provided hereunder, Natural shall not be required, except to the extent it has accepted confirmed nominations allowing authorized overrun, to: (1) accept on any day gas tendered for injection hereunder in excess of Shipper's Maximum Daily Injection Quantity (IQ) or which would cause Shipper to exceed its Maximum Storage Volume (MSV), or withdraw for delivery hereunder on any day gas requested in excess of Shipper's Maximum Daily Withdrawal Quantity (WQ), as such parameters are defined in Section 5 of this Rate Schedule DSS; (2) accept or withdraw gas in excess of any other injection or withdrawal restriction in said Section 5; or (3) deliver gas at Delivery Points on a basis which is not consistent with Section 7 of this Rate Schedule DSS and the related General Terms and Conditions of this Tariff. Shipper shall not be allowed to withdraw more gas than it has in storage. In the case of nominated firm transportation service hereunder, Natural shall not be required to: (i) accept or deliver on any day gas tendered for transportation hereunder which has not been properly nominated and confirmed or which would cause deliveries hereunder on that day to exceed Shipper's Maximum Daily Quantity (MDQ); or (ii) to receive gas at Receipt Point(s) or deliver gas at Delivery Point(s) on a basis which is not consistent with Section 7 of this Rate Schedule

FERC Docket: RP08-319-000

Original Sheet No. 160 original Sheet No. 160 : Pending

RATE SCHEDULE DSS

DSS and the related General Terms and Conditions of this Tariff. Delivered firm storage service hereunder within MDQ and MSV shall be firm, subject to the constraints set out in Section 5 of this Rate Schedule with respect to firm injection and withdrawal rights. Nominated transportation service within MDQ shall be firm, subject to the limitations described in this Rate Schedule DSS and the related General Terms and Conditions. To the extent Shipper receives nominated firm transportation service hereunder on any day, its MDQ for delivered firm storage service shall be reduced by the same volume and the WQ recalculated as set out in Section 5.6 of this Rate Schedule DSS.

- (d) The service provided under this Rate Schedule DSS shall be performed under Part 284 of the Commission's Regulations. Shipper shall only tender gas under this Rate Schedule DSS to the extent such service would qualify under the applicable statutes, regulations, Commission orders and the blanket certificate authorizing service by Natural under this Rate Schedule DSS. For service under Subpart B of Part 284 of the Commission's Regulations, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by the Commission's Regulations, Shipper shall (prior to tendering gas under a DSS Agreement) cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit the necessary certification.
- (e) Natural may utilize any and all of its storage and related facilities to provide the delivered firm storage service described in this Rate Schedule DSS. The capacity available to provide such service shall be based on the capacity and functions of all such storage fields.
- (f) Shipper is not required to have firm transportation service for delivery of gas to storage in order to have delivered firm storage service under this Rate Schedule DSS. However, injections under this Rate Schedule DSS will be dependent on any necessary transportation being confirmed by Natural in accordance with the priority sequence set out in the General Terms and Conditions of this Tariff. Storage points may be Delivery Points under transportation Agreements used for injection hereunder and such points will be treated as primary points for scheduling, curtailment and related purposes to the extent provided in Section 7.5(a) of this Rate Schedule DSS.

FERC Docket: RP08-319-000

Original Sheet No. 161 original Sheet No. 161: Pending

RATE SCHEDULE DSS

(g) Allocation of capacity, curtailment, and priorities of service for the purposes of scheduling and curtailment shall all be governed by the General Terms and Conditions of this Tariff.

Shipper may release capacity dedicated to service hereunder pursuant to Natural's Capacity Release Program to the extent permitted by, and subject to the terms and conditions contained in, Section 19 of the General Terms and Conditions of this Tariff; provided, however, that no segmented releases of capacity hereunder are permitted. Shipper may release its rights under a DSS Agreement in whole or in part or may limit its release to all or part of either the delivered firm storage service or the nominated firm transportation service; provided, however, that to the extent a Shipper releases nominated firm transportation service, the MDQ of delivered firm storage service remaining for use by Shipper or for release to others shall be reduced by an equal volume and the corresponding WQ recalculated as set out in Section 5.6 of this Rate Schedule DSS. If either the storage or the transportation component of this Rate Schedule DSS (or both) is released separately, the storage portion of the service shall be provided under the terms and conditions of Rate Schedule NSS, except that the injection, withdrawal and inventory parameters set out in Section 5 of this Rate Schedule DSS shall apply, and the terms and conditions of Rate Schedule FTS shall apply to the transportation portion of the service, but such transportation service shall be subject in addition to the restrictions on nominated firm storage service under this Rate Schedule DSS, all Receipt Points will be out-of-path secondary in nature, all Delivery Points will be secondary in nature and no LN, SW or NB service options shall be available; provided, however, that if the Original and each replacement Shipper maintains the same primary Delivery Points as applied to the Original Shipper prior to the release, such Delivery Points shall have the same priority as they did for the Original Shipper prior to the release. A Shipper releasing a portion of its storage rights under a DSS Agreement may designate how existing inventory is to be allocated among itself and the replacement Shippers; provided, however, that the resulting inventory cannot, absent express agreement by Natural, exceed any Shipper's MSV under the resulting Agreements (i.e., the MSV calculated in reference to the MDQ under any replacement Agreement for any inventory designated to go to such replacement Agreement and the MSV calculated in reference to the MDQ not released under the original Agreement for any inventory designated to remain with the Original Shipper) nor may a release result in any Shipper or all Shippers in the aggregate (Original and replacement)

FERC Docket: RP08-319-000

Original Sheet No. 162 original Sheet No. 162: Pending

RATE SCHEDULE DSS

having rights in excess of those available prior to any release on a firm basis to the releasing Shipper under this Tariff and the DSS Agreement. In the event Shipper segments its primary point transportation path as a result of a capacity release under the separate transportation Agreement(s) being used to deliver gas to storage for injection hereunder, Shipper's rights may be affected as provided in Section 19 of the General Terms and Conditions of this Tariff.

3. VALID REQUEST

- 3.1 A request for service under this Rate Schedule DSS shall be valid as of the date received if it complies with this Section and contains adequate information on all of the items specified in Section 3.2, subject to any necessary verification of such information and to the following:
- (a) A request shall not be valid and Natural shall not be required to grant any such request:
- (1) For which adequate capacity or operational capability is not available in Natural's storage system and related transmission or other facilities necessary to provide such service;
- (2) As to which Natural does not have the operational capability to effect receipt, storage injections/withdrawals and/or delivery on a basis consistent with the terms and conditions of this Rate Schedule DSS;
- (3) Which, except as provided for in Section 3.1(b) hereof, would require the construction, modification, expansion, or acquisition of any facilities; provided, however, that Natural may agree in its reasonable discretion to construct, modify, expand, or acquire facilities to enable it to perform such services;
- $\qquad \qquad \text{(4)} \quad \text{Unless and until Shipper has provided Natural with the information required in Section 3.2;}$

FERC Docket: RP08-319-000

Original Sheet No. 163 original Sheet No. 163: Pending

RATE SCHEDULE DSS

(5) If Natural determines, based on the credit analysis referenced in Section 3.2(e), that Shipper does not possess sufficient financial stability to make it reasonably likely the service provided hereunder will be paid for on a timely basis;

- (6) If the service requested would not comply with this Rate Schedule DSS; or $\,$
- (7) If the service requested is at less than the applicable maximum rate; provided, however, that Natural may agree to provide service hereunder at a discount consistent with this Rate Schedule DSS.
- (b) Natural agrees to construct minor meter and tap facilities to effectuate service requested under this Rate Schedule DSS provided that: (1) the cost of such facilities does not exceed \$100,000.00; (2) Natural shall be reimbursed prior to commencing construction; and (3) the facilities will not adversely affect Natural's operations. In the event that Natural agrees to construct, modify, expand or acquire facilities to perform service under this Rate Schedule DSS, Natural shall maintain a separate record of the nature and costs of such facilities.
- (c) Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to this Section and while a DSS Agreement is in effect. Such matters are governed by other provisions of this Rate Schedule DSS and by the General Terms and Conditions of this Tariff.
- (d) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request, in whole or in part, due to lack of capacity or system capability or if the request is incomplete or does not comply with this Rate Schedule DSS. Any request shall be null and void unless it is substantially complete and complies with this Rate Schedule DSS. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the DSS Agreement, after which Shipper shall have fifteen (15) calendar days to provide the specified information. In the event such information is not received within fifteen (15) days, Shipper's request shall be null and void.

FERC Docket: RP08-319-000

Original Sheet No. 164 original Sheet No. 164: Pending

RATE SCHEDULE DSS

3.2 Requests for service hereunder shall be deemed valid only after the information specified in this Section 3.2 is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department, One Allen Center, 500 Dallas Street, Houston, Texas 77002, or Telecopy Number (713) 369-9305. Qualifying bids in an auction shall be treated as a valid request provided that the bid is supplemented with all information required under this Section 3.2 as required by the General Terms and Conditions of this Tariff. The information required for a valid request shall be as follows:

(a) VOLUME

The request shall specify in Dth an aggregate MDQ, which shall be utilized to determine the WQ, the MSV and the IQ as set out in Section 5 of this Rate Schedule DSS. All stated quantities shall be exclusive of Storage Fuel, Fuel Gas and Gas Lost and Unaccounted For. Storage Fuel, Fuel Gas and Gas Lost and Unaccounted For shall be provided by Shipper as set out in Sections 6.2(c) and 7.4 of this Rate Schedule DSS. If Shipper is willing to accept a lesser MDQ, Shipper shall also specify the minimum acceptable MDQ.

(b) POINTS

The request shall specify the primary Delivery Point(s) for service under this Rate Schedule DSS, which must be physical points on Natural's system, and the MDQ for each such point. Such point requests must be consistent with Section 7 of this Rate Schedule DSS and the General Terms and Conditions of this Tariff. A Shipper may request any number of primary Delivery Points so long as the summation of MDQs at all primary Delivery Points equals the aggregate MDQ. The Receipt Point(s) under any DSS Agreement shall be as specified in Section 7 of this Rate Schedule DSS and the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 165 Original Sheet No. 165: Pending

RATE SCHEDULE DSS

(c) TERM OF SERVICE

The request shall specify:

(1) The date service is requested to commence;

and

- (2) The date service is requested to terminate.
- (d) REQUESTED RATE

The request shall state the rate at which the Shipper is willing to contract for the service requested. If no rate is stated, Shipper will be deemed to have agreed to the maximum rate.

(e) CREDIT

Acceptance of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.

(f) COMPLIANCE WITH DSS TARIFF

Submission of a request shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule DSS, including the applicable General Terms and Conditions.

(g) COMMISSION-REQUIRED FILING INFORMATION

Except to the extent such information has been or will be provided under another Agreement, the following information is to be provided at the time a request for service hereunder is submitted, if available, or when an initial nomination for service under an executed DSS Agreement is submitted, and when any subsequent changes occur:

FERC Docket: RP08-319-000

Original Sheet No. 166 original Sheet No. 166: Pending

RATE SCHEDULE DSS

(1) Affiliation of the Shipper with Natural; and

(2) The identity of the Shipper, including whether it is a local distribution company, an interstate pipeline company, an intrastate pipeline company, an end user, a producer, or a marketer.

4. TERM

- (a) The term of service hereunder shall be set forth in the DSS Agreement between Shipper and Natural as provided in Section $3.2\,(c)$.
- (b) The General Terms and Conditions of this Tariff shall govern the applicability of, and the terms and conditions relating to, rollovers and the right of first refusal vis a vis a DSS Agreement. Upon termination of any DSS Agreement, and subject to such rollover or right of first refusal, service by Natural to Shipper thereunder shall be terminated and automatically abandoned.
- (c) Natural may terminate any DSS Agreement if Natural is required by the Commission or some other agency or court to provide firm service for others utilizing the system capacity or capability required for service under such DSS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the type covered by the DSS Agreement; provided, however, that nothing herein shall be read to protect Natural from the consequences of its own negligence.

FERC Docket: RP08-319-000

Original Sheet No. 167 Original Sheet No. 167: Pending

RATE SCHEDULE DSS

5. INJECTION/WITHDRAWAL AND TRANSPORTATION RIGHTS

5.1 (a) A Shipper's WQ defines the maximum quantity of gas which it has a firm right to withdraw and have delivered to it by Natural on any day under delivered firm storage service pursuant to this Rate Schedule DSS, subject to the limitations in subsections (b) and (c) of this Section 5.1 and in Section 5.6. The Shipper's WQ (in the aggregate and at any primary Delivery Point) shall be a percentage of its MDQ which varies by time period, as follows:

WQ

Period	Percentage of MDQ
November 1 through February 15	100%
February 16 through February 28 (or 29)	90%
March 1 through March 31	75%
April 1 through April 30	45%
May 1 through September 30	10%
October 1 through October 31	45%

(b) A Shipper's net withdrawals in any month under delivered firm storage service pursuant to this Rate Schedule DSS are subject to the following maximums and minimums, expressed as a percentage of Shipper's MSV:

NET WITHDRAWAL PARAMETERS (AS A % OF MSV)

	Maximum 	Minimum
November	20%	10%
December	30%	10%
January	30%	10%
February	30%	10%
March	20%	5%
April-October	N/A	N/A

FERC Docket: RP08-319-000

Original Sheet No. 168 original Sheet No. 168: Pending

RATE SCHEDULE DSS

- (c) A Shipper's firm right to withdraw gas under delivered firm storage service over any consecutive fifteen (15) day period may not exceed that Shipper's average WQ over that period multiplied by ten (10). A Shipper may request waiver of this limitation, but any withdrawals made pursuant to such waiver shall be provided on an interruptible basis with the same priority as for authorized overruns under Section 5.5(b) of this Rate Schedule DSS; i.e., with higher priority than other authorized overrun and interruptible provided neither Shipper's MDQ nor its MSV is exceeded. If the request is made at least forty-eight (48) hours but not more than seventy-two (72) hours before the Shipper would become subject to this limitation, Natural will advise such Shipper as to whether it can waive this restriction in whole or in part within five (5) business hours after receipt of the request. The waiver request and Natural's response shall specify precisely the waiver sought and granted and also how the waiver is to affect future fifteen (15) day periods and the monthly requirements under subsection (b) of this Section 5.1.
- (d) If Shipper exceeds its firm withdrawal rights hereunder, or fails to satisfy its minimum withdrawal obligation under subsection (b), the volume by which Shipper exceeds its rights and/or the volume by which Shipper fails to satisfy its minimum monthly withdrawal obligation shall be subject to Overrun Charges under Section 12 of the General Terms and Conditions of this Tariff.
- 5.2 A Shipper's MSV is the maximum volume of gas which it is entitled to have in storage under this Rate Schedule DSS on a firm basis at any time. A Shipper's MSV equals its MDQ under the DSS Agreement multiplied by fifty (50). If a Shipper exceeds its MSV, it shall be subject to Overrun Charges as provided in Section 12 of the General Terms and Conditions of this Tariff. In no event will a Shipper be allowed to withdraw more gas than it has in storage; i.e., a Shipper may not create a negative inventory.
- 5.3 A Shipper's IQ is the maximum amount of gas it can inject into delivered firm storage service on a firm basis on any day, subject to subsection (c) below. If a Shipper exceeds its firm injection rights hereunder, such excess injections shall be subject to Overrun Charges under Section 12 of the General Terms and Conditions of this Tariff.
- (a) For the period from November 1 through April 30, a Shipper's IQ varies by time period and is a percentage of its MDQ, as follows:

FERC Docket: RP08-319-000

Original Sheet No. 169 original Sheet No. 169: Pending

RATE SCHEDULE DSS

ΙQ

Time Period Percentage of MDQ

November 1 through March 31 10% April 1 through April 30 15%

(b) For the period of May 1 through October 31, a Shipper's IQ varies by time period and its April 30 DSS storage balance (the April 30 storage balance is assumed to be zero until Natural notifies Shipper of the balance) and is determined by the formula:

IQ = (MSV - April 30 storage balance) (MDQ) (X)

where X varies by time period as follows:

May 1 through July 31 30% August 1 through October 31 40%

Notwithstanding the foregoing, Natural will post on the Informational Postings portion of its Interactive Website no later than March 31 of a year (except that for the Year 2001 the posting will be made by June 4) if operational conditions on its system will permit X to be increased from 30% up to a higher level not to exceed 40% over all or any portion of the period from May 1 through July 31 of that year (June 4 through July 31 for 2001). The posting may specify different values of X for different time periods between May 1 and July 31 (June 4 and July 31 for 2001), within the 30% to 40% range.

(c) (1) Subject to subsection (2) below, a Shipper's firm injection rights on any day shall be further limited as follows: fifty percent (50%) of a Shipper's IQ shall be available for injection from sources on the Amarillo Line and fifty percent (50%) of a Shipper's IQ shall be available for injection from sources on the Gulf Coast Line. Gas which a Shipper seeks to have scheduled and Natural is able to schedule on either line in excess of the resulting daily firm injection volume, or in excess of the storage balance limits defined in subsection (3) below, shall be treated as authorized overrun, with the same priority as other authorized overrun injections under Section 5.5(b) of this Rate Schedule DSS.

FERC Docket: RP08-319-000

Original Sheet No. 170 original Sheet No. 170: Pending

RATE SCHEDULE DSS

(2) While Natural shall allocate injections by leg on a daily basis, overruns of daily rights by leg shall be evaluated on a monthly basis. Shipper shall only be subject to overrun charges with respect to its daily injection rights by leg if and to the extent its total injections by leg hereunder during any month exceed the sum of its daily injection rights for that month. If Natural accepts confirmed nominations for authorized overrun of injection rights on a leg on any day, the volume authorized in excess of Shipper's firm daily rights shall be added to the sum of

FERC Docket: RP08-319-000

Original Sheet No. 171 Original Sheet No. 171: Pending

RATE SCHEDULE DSS

its daily rights for that leg for the month but, unless otherwise agreed, an equal volume shall be subtracted from the sum of daily rights for the other leg for that month. Notwithstanding the foregoing, Natural shall continue to evaluate overrun of total IQ under any Agreement, and assess any corresponding Overrun Charges, on a daily basis.

(3) Shipper's storage balance shall be divided into Amarillo and Gulf Coast components, reflecting the sourcing of the supply. Separate accounting shall be maintained by Natural for each component on a continuous basis. Shipper's total storage balance may not exceed its MSV except as otherwise provided in this Tariff. Volumes in the two components of Shipper's storage balance are subject to the following limits: volumes from Amarillo Line sources residing in the Amarillo Storage Point included in such balance may not at any time exceed fifty percent (50%) of Shipper's MSV and volumes from Gulf Coast Line sources residing in the Gulf Coast Storage Point may not at any time exceed sixty percent (60%) of Shipper's MSV. A Shipper which breaches the limitations in this subsection (3) on any day shall pay a Sourcing Charge as set out on the rate sheets of this Tariff for that day multiplied by the excess volume (in Dth); provided, however, that no Sourcing Charge shall apply prior to April 1, 1996 (when trading rights are implemented). If a Shipper's inventory hereunder exceeds its aggregate MSV, no Sourcing Charge shall apply, but any applicable Overrun Charges under Section 12 of the General Terms and Conditions of this Tariff shall be assessed. Notwithstanding payment of a Sourcing Charge, volumes in each component of a Shipper's storage inventory shall remain in that component until withdrawn or until transferred as set out below. A Shipper's DSS storage account as of November 1 shall automatically be adjusted so that, of its total DSS inventory, fifty percent (50%) of the volume resides in the Amarillo Storage Point and fifty percent (50%) of the volume resides in the Gulf Coast Storage Point. On any day, a Shipper can request that a specified volume be transferred from the Amarillo Storage Point to the Gulf Coast Storage Point, or vice versa, within the limitations of Shipper's available rights to maintain storage inventory in the component of its storage balance to which the volume is being transferred, where such a transfer would avoid charges hereunder or alleviate an actual or potential breach of the limitations under this subsection (3). To implement such a transfer, Shipper must specifically request such a transfer as part of its timely nomination for that day and Shipper must make available capacity to effectuate the transfer under an FTS or FTS-G Agreement with a primary path which allows Natural to make the requested transfer.

FERC Docket: RP08-319-000

Original Sheet No. 172 Original Sheet No. 172: Pending

RATE SCHEDULE DSS

(4) Rules regarding the allocation of injections by leg to determine such sourcing are set out in Section 5.6(c) of the General Terms and Conditions of this Tariff. In applying this subsection (c), all receipts on Natural's system on the Amarillo Line and north of Station 311 on the Gulf Coast Line shall be deemed to be sourced on the Amarillo Line and injected into the Amarillo Storage Point in the Iowa-Illinois Receipt Zone and all other receipts shall be deemed to be sourced on the Gulf Coast Line and injected into the Gulf Coast Storage Point in the Iowa-Illinois Receipt Zone, except as set out in subsection (5) below.

- (5) Notwithstanding the foregoing, Natural shall post prior to the beginning of each month, at least twenty-four (24) hours before first-of-the-month nominations are due, the extent (if any) to which receipts between Station 110 and Station 311 will be treated as Gulf Coast receipts. Natural will designate such receipts as Gulf Coast receipts for that month to the extent reasonably feasible operationally. Such designation may be reduced only prospectively through issuance of an Operational Flow Order or in the subsequent month's posting. Natural shall also review such designations daily and may change the designation by a posting prior to the time nominations are due for that day; provided, however, that the volume designated as Gulf Coast receipts in a monthly posting may not be reduced during a month by such daily postings. Natural shall also post prior to each month, at least twenty-four (24) hours before first-of-the-month nominations are due for that month, whether the restrictions on Shipper's cumulative storage balance on each leg (as a percentage of MSV) can be relaxed for a specified time period. Any posting hereunder shall specify clearly which restrictions are being modified and the magnitude and duration of the modification.
- 5.4 (a) A Shipper must have in storage under this Rate Schedule DSS at least ninety-five percent (95%) of its MSV at some time during the period from October 15 through November 15. If Shipper fails to meet this requirement, its MSV for that year shall be deemed to be equal to the greatest volume in storage during that period. If the maximum volume in storage during that period is less than ninety-five percent (95%) of Shipper's MSV, but is equal to or greater than ninety percent (90%) of Shipper's MSV, its MDQ (in the aggregate and at each Delivery Point) shall be reduced to seventy-five percent (75%) of the MDQ in the DSS Agreement, solely for the purposes of determining its WQ during the ensuing withdrawal cycle. If the

FERC Docket: RP08-319-000

Original Sheet No. 173 original Sheet No. 173: Pending

RATE SCHEDULE DSS

maximum volume in storage during that period is less than ninety percent (90%) of Shipper's MSV, then its MDQ (in the aggregate and at each Delivery Point) shall be reduced to fifty percent (50%) of the MDQ in the DSS Agreement, solely for purposes of determining its WQ during the ensuing withdrawal cycle. At the beginning of the subsequent injection cycle, Shipper's MDQ shall be restored for all purposes to its stated level, subject to further adjustment under this Section. Notwithstanding the foregoing, a Shipper's rights shall not be reduced in the event that its failure to have the required inventory in storage is attributable to a change by Natural in operating parameters which is inconsistent with operating plan(s) posted previously by Natural. Natural will grant any waiver required to effectuate the prior sentence; provided, however, that such a waiver shall be granted only to the extent Shipper did not have a reasonable opportunity following the posting of the change in operating parameters to modify its injection patterns and achieve the required inventory on a timely basis. No Overrun Charges shall apply based on Shipper's failure to satisfy the inventory requirements of this Section 5.4(a).

Shipper's remaining storage inventory under a DSS (b) Agreement on each April 30 must be no more than fifty percent (50%) of its MSV. Rules regarding the allocation of withdrawals by \log under Rate Schedule DSS are set out at Section 5.6(c) of the General Terms and Conditions of the Tariff. Notwithstanding the foregoing, Shipper may request relief from this seasonal cycling requirement. Natural shall grant such requests on a nondiscriminatory basis if the physical cycling requirements of its storage reservoirs have already been met. In addition, Natural shall grant waiver of the cycling requirement on a nondiscriminatory basis to the extent: (i) Natural's physical cycling requirement could be met by requiring the withdrawal of BESS inventory within the applicable limits specified in Rate Schedule BESS [Natural shall utilize Sections 3(b) and 3(c) of Rate Schedule BESS as a mechanism to achieve cycling of BESS volumes in lieu of withdrawals hereunder but such provisions shall be a substitute for cycling hereunder only to the extent the waiver conditions of this subsection can be satisfied in conjunction with the use of such provisions]; and (ii) system capability is available for the withdrawal and transportation of such BESS inventory; provided, however, that a request for waiver which would require withdrawal of BESS inventory may not be submitted prior to February 20 and, if Natural notifies the Shipper that the waiver can be granted, will become effective one (1) week from the date on which Natural grants the waiver, in whole or in part. A request for waiver shall be

FERC Docket: RP08-319-000

Original Sheet No. 174 Original Sheet No. 174: Pending

RATE SCHEDULE DSS

deemed continuing for the remainder of the withdrawal season unless Shipper specified otherwise, but Shipper must submit a new waiver request for any subsequent withdrawal season. If Shipper's remaining balance on April 30 under a DSS Agreement is greater than that allowed above, Shipper shall not be subject to Overrun Charges on such excess balance (unless some other limitation is exceeded) but shall pay (as applicable):

- (1) A charge equal to the Authorized Inventory Cycling Rate multiplied by the volume representing that portion of the remaining inventory in excess of fifty percent (50%) of its MSV as to which Shipper has obtained cycling relief; and/or
- (2) A charge equal to the Unauthorized Inventory Cycling Rate multiplied by the volume representing that portion of the remaining inventory in excess of fifty percent (50%) of its MSV as to which Shipper has not obtained cycling relief from Natural.
- Under circumstances where a waiver of the cycling requirement would be granted under subsection (b), Natural may provide cycling on behalf of Shippers on a non-discriminatory basis to the extent operationally feasible. Under this Cycling Option, the Rate Schedule DSS account of a Shipper will be reduced prior to April 30, with equivalent volumes to be returned to Shipper's Rate Schedule DSS account on the ensuing May 1. Natural shall assess a Cycling Option Charge equal to the Cycling Option Rate multiplied by the volumes cycled. The Cycling Option Charge shall be equal to or greater than the charge at the minimum Cycling Option Rate, but no greater than the charge based on the maximum Cycling Option Rate. The minimum and maximum Cycling Option Rate shall equal the minimum and maximum Authorized Inventory Cycling Rate, respectively. Prior to granting any requests for the Cycling Option, Natural shall post on the Informational Postings portion of its Interactive Website the availability of the Cycling Option and the rate Natural will charge (within the applicable maximum and minimum).
- 5.5 (a) A Shipper under this Rate Schedule DSS may request, as part of its nomination, injections in excess of IQ or in excess of its firm rights under Section 5.3(c) hereto or injections which would cause it to exceed its MSV and may, as part of its nomination, request withdrawals in excess of WQ or of its firm rights under Section 5.1(c). Such withdrawals and injections are subject to Section 9 of this Rate Schedule DSS. Rights available under Section 5.5(c) are firm and available on a no-notice basis and do not require a nomination.

FERC Docket: RP08-319-000

Original Sheet No. 175 Original Sheet No. 175: Pending

RATE SCHEDULE DSS

(b) Except as set out in subsection (c), such injections or withdrawals shall be treated as authorized overrun for purposes of scheduling and curtailment under the General Terms and Conditions of this Tariff; provided, however, that injections and withdrawals within MDQ and MSV [including withdrawals within MDQ pursuant to a waiver under Section 5.1(c) or injections within MDQ and MSV in excess of firm rights under Section 5.3(c)] shall be treated as higher priority than all other authorized overrun or interruptible volumes (except corresponding authorized overrun requests under Rate Schedule NSS, which shall be of equal priority), consistent with the priority system set out in Section 5 of the General Terms and Conditions of this Tariff; and provided further that no authorized overrun charges apply unless MDQ and/or MSV is exceeded.

FERC Docket: RP08-319-000

Original Sheet No. 176 Original Sheet No. 176: Pending

RATE SCHEDULE DSS

(c) Forty-eight (48) hours prior to the beginning of each gas day, Natural shall post on the Informational Postings portion of its Interactive Website the availability of additional storage injection and withdrawal capability in excess of IQ and WQ, determined by taking into account both daily and seasonal storage parameters. Such availability shall be stated as percentage(s) of MDQ and may vary $% \left(1\right) =\left(1\right) \left(1\right) +\left(1\right) \left(1\right) \left(1\right) +\left(1\right) \left(1\right) \left$ by pipeline leg as to injections. Such posting shall state the specific firm rights granted as to IQ and WQ and how the additional volumes are to be allocated between pipeline legs as to injections. Upon such posting, the storage injection and withdrawal capability stated as available in the posting shall be firm (and available for no notice rights) for that day provided neither Shipper's MDQ nor its MSV is exceeded. If no posting is made for any day, the default volume for firm injections and withdrawal rights on that day shall be five percent (5%) of MDQ in excess of the otherwise applicable IQ and WQ provided neither Shipper's MDQ nor its MSV is exceeded.

5.6 In the event that a Shipper receives nominated firm transportation service hereunder on any day (which service is available during the period of October 15 through April 15), such Shipper's MDQ available for delivered firm storage service hereunder shall be reduced by an equal volume. Such Shipper's WQ shall be recalculated to equal the lesser of (i) the WQ based on the full MDQ under the Agreement (before reduction for nominated firm transportation service), or (ii) the reduced MDQ; it being understood that the sum of the nominated firm transportation and the WQ in effect on any day may never exceed the MDQ under the Agreement.

6. RATE

- 6.1 Shipper shall pay Natural each month for service under this Rate Schedule DSS the sum of the following charges:
 - (a) A Reservation Charge based on Shipper's MDQ;
- (b) An Authorized or an Unauthorized Inventory Cycling Charge and a Cycling Option Charge;
- (c) A Fuel Charge consisting of a Storage Fuel Charge and a Transportation Fuel Charge, the latter comprised of a Fuel Gas Charge and a Gas Lost and Unaccounted For Charge;
 - (d) Overrun Charges, where applicable; and

FERC Docket: RP08-319-000

Original Sheet No. 177 Original Sheet No. 177: Pending

RATE SCHEDULE DSS

(e) Any other applicable charges and penalties set out in this Rate Schedule or the General Terms and Conditions of this Tariff (including but not limited to Balancing Service Charges); provided, however, that DSS Agreements shall not be subject to charges under Section 21 (Account No. 858 surcharges) or Section 38 (Gas Supply Realignment Charges) of the General Terms and Conditions of this Tariff.

Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.

- $6.2\,$ The charges identified in subsections (a) through (d) of Section 6.1 shall be assessed as follows:
- (a) The maximum Reservation Charge for service under this Rate Schedule DSS shall consist of the applicable maximum unit Reservation Rate set out in this Tariff multiplied by Shipper's MDQ. The Reservation Charge payable by any Shipper shall be the applicable maximum Reservation Charge unless a lower Reservation Charge is specified in the DSS Agreement or a separate discount agreement or is assessed pursuant to Section 6.6 hereof, which lower charge must be consistent with the minimum rate(s) specified in this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma Agreement set out in this Tariff, discounts may be limited to specific volumes. The Reservation Charge shall constitute the minimum bill under this Rate Schedule DSS. Shipper shall be obligated to pay the Reservation Charge hereunder as of the date DSS service commences. DSS service shall be deemed to commence on the later of the date specified in the DSS Agreement or the date Natural is able to render firm service hereunder to Shipper.
- (b) Authorized and Unauthorized Inventory Cycling Charges shall be applied as set out in Section $5.4\,(b)$ of this Rate Schedule DSS and the Cycling Option Charge shall be applied as set out in Section $5.4\,(c)$ of this Rate Schedule DSS.

FERC Docket: RP08-319-000

Original Sheet No. 178 Original Sheet No. 178: Pending

RATE SCHEDULE DSS

(c) Shipper shall provide gas in kind for the fuel required to inject and withdraw gas and transport gas from storage to Delivery Points in connection with delivered firm storage service under this Rate Schedule DSS. Shipper shall also provide gas in kind for the fuel required to transport gas in connection with nominated firm transportation service under this Rate Schedule DSS. For delivered firm storage service, such charge shall be assessed upon injection, reducing the volumes deemed delivered into Shipper's DSS inventory and thus fuel volumes shall not subsequently enter into the determination of Shipper's inventory or withdrawal rights. The Storage Fuel Charge shall equal the Storage Fuel Rate set out in this Tariff (stated as a percentage fuel rate) multiplied by each Dth of gas injected under the DSS Agreement. For both gas injected

FERC Docket: RP08-319-000

Original Sheet No. 179 Original Sheet No. 179: Pending

RATE SCHEDULE DSS

under delivered firm storage service and gas transported under nominated firm transportation hereunder, the Transportation Fuel Charge shall equal the Fuel Gas and Gas Lost and Unaccounted For Rates applicable to transportation from the Iowa-Illinois Receipt Zone to the Market Delivery Zone multiplied by each Dth of gas injected for the account of Shipper or tendered by Shipper for nominated firm transportation service under this Rate Schedule DSS. Upon the mutual agreement of Natural and Shipper, in lieu of Natural retaining gas in kind, Shipper shall reimburse Natural for fuel at a price established under Section 7.1 of the General Terms and Conditions of this Tariff multiplied by the Storage Fuel Charge.

- (d) Except as otherwise provided herein, Shipper shall pay the Authorized Overrun Rate or Unauthorized Overrun Rate, as applicable, specified in Section 12 of the General Terms and Conditions of this Tariff on the volumes of any overrun of its firm rights under Section 5 of this Rate Schedule DSS or on the volume by which Shipper has failed to satisfy its minimum monthly withdrawal obligation under said Section 5.
- 6.3 (a) Shipper shall reimburse Natural within five (5) business days after costs have been incurred by Natural for all fees required by the FERC or any regulatory body including, but not limited to, filing, reporting, and application fees to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees (such as general rate case filing fees). Any costs collected hereunder shall be excluded from Natural's cost of service in determining generally applicable rates and charges.
- (b) If Natural constructs, acquires or modifies any facilities to perform service hereunder, then as specified in an agreement between the parties either:
- $\hbox{(1)} \quad \text{Shipper shall reimburse Natural for the cost} \\ \text{of such facilities or facility modifications as described in the} \\ \text{General Terms and Conditions of this Tariff; or} \\$
- $\hbox{\ensuremath{\mbox{(2)}} \ensuremath{\mbox{Natural shall assess a monthly charge}} \\ \ensuremath{\mbox{reflecting such facility costs.}} \\$
- 6.4 The ACA and RD&D charges will be assessed, when applicable, as provided in the General Terms and Conditions of this Tariff, on volumes delivered by Natural to Shipper under this Rate Schedule DSS.

FERC Docket: RP08-319-000

Original Sheet No. 180 Original Sheet No. 180: Pending

RATE SCHEDULE DSS

6.5 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in (1) the rates and charges applicable under this Rate Schedule DSS, including both the level and design of such rates and charges; or (2) the terms and conditions of this Rate Schedule DSS. Natural agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.

FERC Docket: RP08-319-000

Original Sheet No. 181 Original Sheet No. 181: Pending

RATE SCHEDULE DSS

(b) If, at any time and from time to time, the Commission or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the DSS Agreement or any separate discount agreement, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such highest rate, Shipper shall execute or provide such documentation within fifteen (15) calendar days after a written request by Natural. If, at any time and from time to time, the Commission or any other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for service hereunder, the rate shall be decreased to such reduced rate.

- Natural may from time to time and at any time, upon twenty-four (24) hours' verbal or written notice, subject to any provisions on discounting in the DSS Agreement or in a separate discount agreement, charge any individual Shipper for service under this Rate Schedule DSS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate charged may not be less than the applicable minimum rate for service under this Rate Schedule DSS set forth in this Tariff. Natural will confirm any verbal notice of the applicable rate in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Unless otherwise agreed in the DSS Agreement or in a separate discount agreement, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the DSS Agreement and/or any discount agreement) upon twenty-four (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with the Commission any and all reports as required by the Commission's Regulations with respect to the institution or discontinuance of any discount.
- 6.7 All revenues and gas in kind collected by Natural as a result of providing service under this Rate Schedule DSS shall be retained by Natural unless Natural has otherwise explicitly agreed on a different disposition of such amounts. Where crediting or refund mechanisms apply under other provision(s) of this Tariff or pursuant to effective Commission orders or settlements, such mechanisms shall supersede this Section to the extent necessary to carry out such provision(s).

FERC Docket: RP08-319-000

Original Sheet No. 182 Original Sheet No. 182: Pending

RATE SCHEDULE DSS

7. NOMINATIONS, NO-NOTICE SERVICE AND POINTS

- 7.1 (a) In accordance with the General Terms and Conditions of this Tariff, Shipper shall nominate to Natural the quantity of gas to be injected into or withdrawn from storage or to be received for transportation and redelivered for its account under its DSS Agreement on each day. With respect to delivered firm storage service, such nominated volumes shall not limit Shipper's firm injection or withdrawal rights specified in Section 5 of this Rate Schedule DSS. Firm injection and withdrawal rights for delivered firm storage service shall be available to Shipper on a no-notice basis unless otherwise specifically provided in this Rate Schedule DSS or the applicable General Terms and Conditions.
- (b) Nominated firm transportation available under this Rate Schedule DSS shall be provided only in accordance with Shipper's nominations to Natural and shall be limited to the volume properly nominated and confirmed. If Shipper nominates such service, Shipper's MDQ shall be adjusted and its WQ shall be recalculated as set out in Section 5.6 of this Rate Schedule DSS. Such service may only be nominated for day(s) within the period from October 15 through April 15.
- (c) Natural may permit additional injections or withdrawals or transportation in excess of applicable firm limits and may allow a Shipper to withdraw less than the required minimum monthly volume in accordance with Sections 5.5 and 9 of this Rate Schedule DSS. The quantity of gas withdrawn by Shipper on any day shall not be greater than the total quantity of gas held by Natural for the account of Shipper as of that date.
- 7.2 Any imbalance between confirmed nominations for deliveries and actual deliveries at Delivery Points eligible for service under this Rate Schedule DSS shall be treated as an injection into or withdrawal from Shipper's DSS storage inventory within the limits of Shipper's firm storage rights hereunder (as adjusted under Section 5.6 hereof for any nominated firm transportation being used by Shipper on that day). It shall be Shipper's responsibility to keep any difference between nominated and actual deliveries within a level which can be accommodated by use of such storage rights (the sum of confirmed nominations and any remaining no notice rights). If Shipper fails to do so, Overrun and/or Balancing Service Charges shall apply to the extent specified in Section 12 of the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 183 Original Sheet No. 183: Pending

RATE SCHEDULE DSS

Notwithstanding the foregoing, if Shipper takes action based on inaccurate information provided by Natural regarding any parameter relevant to its operation under this Rate Schedule DSS (such as Shipper's IQ, WQ or storage balance) and such action results in Shipper exceeding its rights, Natural will waive (except as provided in the following sentence) any charges Shipper could reasonably have avoided had Natural provided accurate information. Shipper may rely on the most recent information provided by Natural to Shipper regarding any such parameter, but Shipper has an obligation to provide Natural on a timely basis with relevant information within its control to be used in the determination of such parameters and charges hereunder shall not be waived to the extent that the inaccurate information provided by Natural resulted from Shipper's failure to provide Natural accurate information within its control on a timely basis. Any imbalance between deliveries and receipts, after adjustment for DSS storage injections and withdrawals, shall be eliminated by cashout on a monthly basis in accordance with Section 13.3 of the General Terms and Conditions of this Tariff.

- 7.3 All primary Delivery Points under the DSS Agreement, along with all secondary Delivery Points that together with one or more of the primary points comprise a Central Delivery Point, shall be eligible for no-notice service as part of the delivered firm storage service under this Rate Schedule DSS. A Shipper may also request, by notice to Natural, to take no-notice service at other secondary Delivery Points in lieu of points described in the prior sentence upon prior written notice to Natural, and Natural shall promptly notify the Shipper whether such points are eligible for no-notice service.
- 7.4 The primary Delivery Point(s) for all gas delivered by Natural to Shipper (or to a third-party on behalf of Shipper) hereunder shall be specified in the DSS Agreement and must be located within the Market Delivery Zone. For each individual primary Delivery Point, and for the aggregate of all such points, Natural's maximum obligation to deliver gas on a firm basis (i.e., its MDQ) shall be specified in Dth in the DSS Agreement. The sum of the MDQs at primary Delivery Points may not exceed the aggregate MDQ. All stated MDQs shall be exclusive of applicable Storage Fuel, Fuel Gas and Gas Lost and Unaccounted For. Storage Fuel, Fuel Gas and Gas Lost and Unaccounted For shall be provided by Shipper for delivered firm storage service upon injection by reducing the volumes deemed delivered into such Shipper's DSS inventory under the separate transportation Agreement(s) used for injection hereunder

FERC Docket: RP08-319-000

Original Sheet No. 184 Original Sheet No. 184: Pending

RATE SCHEDULE DSS

and such fuel volumes shall not subsequently be taken into account in determining Shipper's DSS inventory or withdrawal rights. Fuel Gas and Gas Lost and Unaccounted For shall be assessed on nominated firm transportation service under this Rate Schedule DSS at the time such service is provided. All points in the Market Delivery Zone may be used as secondary out-of-path Delivery Points hereunder, consistent with the General Terms and Conditions of this Tariff, but such points shall only be eligible for no-notice service to the extent provided in Section 7.3 of this Rate Schedule DSS.

- (a) With respect to delivered firm storage service, the Receipt Points shall be the storage point(s) in the Iowa-Illinois Receipt Zone defined in this Tariff. Each DSS Agreement shall have such storage points as Receipt Point(s) and will be subject to the limitations and conditions (including volumes and priorities) specified in the General Terms and Conditions of this Tariff relating to such points. While such storage point(s) are not physical points, the Receipt Point(s) hereunder shall be treated as primary points on withdrawal for scheduling, curtailment and similar purposes. For injection, such storage points shall be treated as primary Delivery Points for scheduling and curtailment to the extent provided in Section 5.6 of the General Terms and Conditions of this Tariff. Any separate firm transportation Agreement held by Shipper under which it desires to make injections into its DSS account may have the Iowa-Illinois Receipt Zone storage points as Delivery Points and such storage points shall be treated as primary Delivery Points for scheduling and curtailment to the extent provided in Section 5.6 of the General Terms and Conditions of this Tariff. Further, such transportation Agreement may specify that such storage points are to be the Delivery Points during the injection season, with the applicable priority, and that the Shipper's market area points are to be the primary Delivery Points during the withdrawal season.
- (b) For nominated firm transportation service, the Receipt Point(s) under any DSS Agreement shall include all points in the Iowa-Illinois Receipt Zone. Natural shall determine for each DSS Agreement, at the time the request for service is processed, which Receipt Points can be treated operationally as inthe-path secondary in nature. All other Receipt Point(s) shall be out-of-path secondary in nature. The MDQ at each Receipt Point shall equal the aggregate MDQ. The priority applicable to secondary Receipt Points shall be governed by the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 185 Original Sheet No. 185: Pending

RATE SCHEDULE DSS

8. OPERATING CONDITIONS

- (a) Operating conditions at Receipt Point(s) for nominated firm transportation service hereunder and operating conditions at Delivery Point(s) for any service hereunder shall be governed by Sections 5, 7 and 8(b) of this Rate Schedule DSS and the General Terms and Conditions of this Tariff. Operating conditions related to injections and withdrawals hereunder shall be governed by Sections 5, 7 and 8(b) of this Rate Schedule DSS and the General Terms and Conditions of this Tariff.
- Shipper shall arrange separately for the transportation of gas to Natural's Iowa-Illinois Receipt Zone storage point(s) in connection with delivered firm storage service under this Rate Schedule DSS. Shipper shall make all necessary arrangements separately with Natural or other parties upstream of the point where gas is tendered to Natural in the Iowa-Illinois Receipt Zone for $\begin{array}{c} -1 \\ \text{transportation} \end{array} \text{ of gas pursuant to nominated firm transportation}$ service hereunder. Shipper is also responsible for all arrangements downstream of the Delivery Point(s). Such arrangements must be consistent with this Rate Schedule DSS and must be coordinated with Natural, subject to the no-notice aspects of the delivered firm storage service being provided under the DSS Agreement. To be eligible for service hereunder, Shipper or its supplier or purchaser must comply with all applicable provisions of any separate transportation Agreement(s) and any applicable Rate Schedule(s) and the applicable General Terms and Conditions.
- (c) Natural will provide all base gas required to support delivered service under Rate Schedule DSS. Shippers will provide working gas for such service as set out in Section 5 of this Rate Schedule DSS.
- (d) Shipper must have all of its gas withdrawn by the end of the term of a DSS Agreement (or Agreement under a successor service) which is not rolled over or extended. Such withdrawals may be effectuated through an in-field transfer, subject to the then applicable rules governing such transfers. If at any time service pursuant to this Rate Schedule DSS (or successor service) and/or a DSS Agreement (or Agreement under a successor service) is cancelled or terminated prior to the end of the term specified in the DSS Agreement (or successor Agreement), Shipper shall be required to withdraw its gas over a period of time determined by dividing: (1) the volume of gas Shipper is required to withdraw, by (2) fifty percent (50%) of Shipper's MDQ; provided, however, that if service is being terminated pursuant to Section 19.19 of the General Terms and Conditions of this Tariff, the withdrawal period shall be equal in

FERC Docket: RP08-319-000

Original Sheet No. 186 original Sheet No. 186: Pending

RATE SCHEDULE DSS

duration to the prior notice period required under said Section 19.19. If Natural is unable to withdraw any volume requested by Shipper, up to the volume specified in the preceding sentence, the period shall be extended accordingly. If the gas is not withdrawn within the specified time period, the remaining volumes shall be cashed out with no offsets, trading or netting and the proceeds distributed to Shipper, based on the greater than twenty percent (20%) tier for imbalances set out at Section 13.3 of the General Terms and Conditions of this Tariff.

9. OVERRUN SERVICE

Upon request of Shipper, Natural may (but is not obligated to) receive for injection hereunder on any day as authorized overrun quantities of natural gas in excess of Shipper's IQ or MSV or other restrictions on injections under Section 5 of this Rate Schedule DSS, as applicable. Similarly, Natural may, upon request from Shipper, withdraw and/or transport volumes in excess of a Shipper's WQ or MDQ or other restrictions on withdrawals or transportation volumes under this Rate Schedule DSS, as applicable, or may allow Shipper to withdraw less than the minimum required monthly volume. Such requests can be granted when in Natural's reasonable judgment, the capacity and operating capability of its system will permit without impairing the ability of Natural to meet its other service obligations; provided, however, that in no event will Shipper be allowed to withdraw more gas than it has in storage. In evaluating such requests, Natural shall take into account the sum of the service nominated for nominated firm storage service and the available delivered firm storage service rights. In granting requests for Authorized Overrun Service, Natural shall act in a manner consistent with the overrun service priorities set out in the ${\tt General}\ {\tt Terms}\ {\tt and}$ Conditions of this Tariff. Charges applicable to either authorized or unauthorized overrun volumes shall be as set out in Section 12 of the General Terms and Conditions of this Tariff.

10. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule DSS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Original Sheet No. 187 Original Sheet No. 187: Pending

RATE SCHEDULE NSS
NOMINATED FIRM STORAGE SERVICE

1. AVAILABILITY

- (a) This Rate Schedule NSS is available for nominated firm storage service (with limited no-notice rights in conjunction with the NB service option under Rate Schedules FTS, FFTS or FTS-G) to any entity (hereinafter called Shipper) which: (1) submits to Natural Gas Pipeline Company of America LLC (hereinafter called Natural) a valid request as defined in Section 3 hereof which Natural has capacity available on all affected portions of its system and the operational capability to satisfy; and (2) executes a Nominated Firm Storage Service Agreement (NSS Agreement) with Natural applicable to service under this Rate Schedule NSS. The form of NSS Agreement is contained in this Tariff. There is no limitation on the number of NSS Agreements any one Shipper may have.
- (b) Shipper shall arrange separately with Natural or others as necessary for any transportation attendant to the storage service provided hereunder. Shipper shall pay separately for such transportation service.

2. APPLICABILITY, CHARACTER AND PRIORITY OF SERVICE

(a) This Rate Schedule NSS defines a nominated firm storage service, to be provided on a prior notice basis; provided, however, that injections and withdrawals hereunder shall be available on a no-notice basis to the limited extent permitted in conjunction with firm transportation if Shipper makes the appropriate election of the NB service option under Rate Schedule FTS, FFTS or FTS-G and subject to the limitations on such no-notice rights set out in Rate Schedule FTS, FFTS or FTS-G and the General Terms and Conditions of this Tariff. This Rate Schedule NSS shall apply to all gas received by Natural from or for the account of Shipper pursuant to a NSS Agreement. As more fully set out in the General Terms and Conditions of this Tariff, Natural is not providing a gas supply service under this Rate Schedule NSS.

FERC Docket: RP08-319-000

Original Sheet No. 188 Original Sheet No. 188: Pending

RATE SCHEDULE NSS

- Service hereunder shall consist of the acceptance by Natural of natural gas from or for the account of Shipper for injection into storage hereunder, the storage of gas hereunder, and the withdrawal of such gas from storage hereunder by Natural for Shipper or for Shipper's account. Shipper shall provide Storage $\,$ Fuel for injection and withdrawal of gas as provided in this Rate Schedule NSS. Natural shall not be required (except to the extent Natural has accepted confirmed nominations allowing authorized overrun) to: (1) accept on any day gas tendered for injection hereunder in excess of Shipper's Maximum Daily Injection Quantity (IQ) or which would cause Shipper's storage balance to exceed its Maximum Storage Volume (MSV), as such terms are defined in Section 5 hereof; (2) withdraw for delivery hereunder on any day gas requested in excess of Shipper's Maximum Daily Withdrawal Quantity (WQ), as defined in Section 5 hereof; (3) accept or withdraw gas in excess of any other injection or withdrawal restrictions in said Section 5; or (4) accept or deliver gas hereunder which is not properly nominated and confirmed pursuant to the General Terms and Conditions of this Tariff, except within Shipper's no-notice rights hereunder exercised pursuant to the NB service option properly elected by it in accordance with Rate Schedule FTS or FTS-G and the General Terms and Conditions of this Tariff. Shipper shall not be allowed to withdraw hereunder more gas than it has in storage.
- (c) The service provided under this Rate Schedule NSS shall be performed under Part 284 of the Commission's Regulations. Shipper shall only tender gas under this Rate Schedule NSS to the extent such service would qualify under the applicable statutes, regulations, Commission orders and the blanket certificate authorizing service by Natural under this Rate Schedule NSS. For service under Subpart B of Part 284 of the Commission's Regulations, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by the Commission's Regulations, Shipper shall (prior to tendering gas under a NSS Agreement) cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit the necessary certification.

FERC Docket: RP08-319-000

Original Sheet No. 189 Original Sheet No. 189: Pending

RATE SCHEDULE NSS

(d) Natural may utilize any and all of its storage and related facilities to provide the service described in this Rate Schedule NSS. The capacity available to provide such service shall be based on the capacity and functions of all such storage fields. A Shipper hereunder may request the percentage of its service to be provided on the Amarillo Line and on the Gulf Coast Line, provided that the two percentages add to 100%. A Shipper may alternatively request separate contracts to cover separately service on the Gulf Coast Line and on the Amarillo Line. For the purpose of evaluating any new requests for firm service hereunder received on or after July 1, 1998, or the awarding of service hereunder pursuant to a unilateral rollover right, a contractual rollover right or a right of first refusal received pursuant to Section 22 of the General Terms and Conditions of this Tariff after July 1, 1998, Natural shall assess separately for the Gulf Coast or for the Amarillo Line, its ability to provide the service requested on that line. If MDQs are separately stated by pipeline leg in separate contracts, then NSS service shall be separately administered by pipeline leg.

FERC Docket: RP08-319-000

Original Sheet No. 190 Original Sheet No. 190: Pending

RATE SCHEDULE NSS

- (e) Shipper is not required to have firm transportation service for delivery of gas to storage or the transportation of gas away from storage in order to have firm storage service under this Rate Schedule NSS. However, injections and withdrawals under this Rate Schedule NSS will be dependent on any necessary transportation being confirmed by Natural and shall be subject to the scheduling and curtailment of such transportation pursuant to the provisions of this Tariff. Storage points hereunder shall be treated as inthe-path to the extent provided in Section 7(b) of this Rate Schedule NSS and Section 5.6 of the General Terms and Conditions of this Tariff.
- (f) Allocation of capacity, curtailment, and priorities of service for the purposes of scheduling and curtailment shall all be governed by the General Terms and Conditions of this Tariff.
- Shipper may release capacity dedicated to service hereunder pursuant to Natural's Capacity Release Program to the extent permitted by, and subject to the terms and conditions contained in, the General Terms and Conditions of this Tariff. Shipper releasing a portion of its rights under an NSS Agreement may designate how inventory is to be allocated among itself and the replacement Shippers; provided, however, that the resulting inventory cannot, absent express agreement by Natural, exceed any Shipper's MSV under the resulting Agreements (i.e., the MSV calculated in reference to the MDQ under any replacement Agreement for any inventory designated to go to such replacement Agreement and the MSV calculated in reference to the MDQ not released under the original Agreement for any inventory designated to remain with the Original Shipper) nor may a release result in any Shipper or all Shippers in the aggregate (Original and replacement) having rights in excess of those available prior to any release on a firm basis to the releasing Shipper under this Tariff and the NSS Agreement.

VALID REQUEST

- 3.1 A request for service under this Rate Schedule NSS shall be valid as of the date received if it complies with this Section and contains adequate information on all of the items specified in Section 3.2, subject to any necessary verification of such information and to the following:

FERC Docket: RP08-319-000

Original Sheet No. 191 Original Sheet No. 191: Pending

RATE SCHEDULE NSS

- (1) For which adequate capacity or operational capability is not available in Natural's storage system and related facilities necessary to provide such service;
- (2) As to which Natural does not have the operational capability to effect storage injections and/or withdrawals on a basis consistent with the terms and conditions of this Rate Schedule NSS;
- (3) Unless and until Shipper has provided Natural with the information required in Section 3.2;
- (4) If Natural determines, based on the credit analysis referenced in Section 3.2(e), that Shipper does not possess sufficient financial stability to make it reasonably likely the service provided hereunder will be paid for on a timely basis;
- $\mbox{(5)}$ If the service requested would not comply with this Rate Schedule NSS; or
- (6) If the service requested is at less than the applicable maximum rate; provided, however, that Natural may agree to provide service hereunder at a discount consistent with this Rate Schedule NSS.
- (b) Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to this Section and while a NSS Agreement is in effect. Such matters are governed by the General Terms and Conditions of this Tariff.
- (c) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request, in whole or in part, due to lack of capacity or system capability or if the request is incomplete or does not comply with this Rate Schedule NSS. Any request shall be null and void unless it is substantially complete and complies with this Rate Schedule NSS. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the NSS Agreement, after which Shipper shall have fifteen (15) calendar days to provide the specified information. In the event such information is not received within fifteen (15) days, Shipper's request shall be null and void.

FERC Docket: RP08-319-000

Original Sheet No. 192 Original Sheet No. 192: Pending

RATE SCHEDULE NSS

3.2 Requests for service hereunder shall be deemed valid only after the information specified in this Section 3.2 is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department, One Allen Center, 500 Dallas Street, Houston, Texas 77002, or Telecopy Number (713) 369-9305. Qualifying bids in an auction shall be treated as a valid request provided that the bid is supplemented with all information required under this Section 3.2 as required by the General Terms and Conditions of this Tariff. The information required for a valid request shall be as follows:

(a) VOLUME

The request shall specify in Dth an aggregate Maximum Daily Quantity (MDQ), which shall be utilized to determine the WQ, the MSV and the IQ as set out in Section 5 of this Rate Schedule. Consistent with Sections 2(d) and 5.3(b)(4) hereof, the request shall also specify the percentage of MDQ by pipeline leg (Amarillo or Gulf Coast). All stated quantities shall be exclusive of Storage Fuel. Storage Fuel shall be provided by Shipper as set out in Sections 6.2(b) and 7(c) of this Rate Schedule NSS. If Shipper is willing to accept a lesser MDQ, Shipper shall also specify the minimum acceptable MDQ, in total and by pipeline leg.

(b) POINTS

The Receipt Points for injection and the Delivery Points for withdrawal under an NSS Agreement shall be the storage points as defined in the General Terms and Conditions of this Tariff. Gas injected at any storage point must be withdrawn from the same storage point.

(c) TERM OF SERVICE

The request shall specify:

(1) The date service is requested to commence;

and

(2) The date service is requested to terminate.

FERC Docket: RP08-319-000

Original Sheet No. 193 Original Sheet No. 193: Pending

RATE SCHEDULE NSS

(d) REQUESTED RATE

The request shall state the rate at which the Shipper is willing to contract for the service requested. If no rate is stated, Shipper will be deemed to have agreed to the maximum rate.

(e) CREDIT

Acceptance of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.

(f) COMPLIANCE WITH NSS TARIFF

Submission of a request shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule NSS, including the applicable General Terms and Conditions.

(q) COMMISSION-REQUIRED FILING INFORMATION

The following information is to be provided at the time a request for service hereunder is submitted, if available, or when an initial nomination for storage under an executed NSS Agreement is submitted, and when any subsequent changes occur:

- (1) Affiliation of the Shipper with Natural; and
- (2) The identity of the Shipper, including whether it is a local distribution company, an interstate pipeline company, an intrastate pipeline company, an end user, a producer, or a marketer.

4. TERM

(a) The term of service hereunder shall be set forth in the NSS Agreement between Shipper and Natural.

FERC Docket: RP08-319-000

Original Sheet No. 194 Original Sheet No. 194: Pending

RATE SCHEDULE NSS

(b) The General Terms and Conditions of this Tariff shall govern the applicability of, and the terms and conditions relating to, rollovers and the right of first refusal vis a vis a NSS Agreement. Upon termination of any NSS Agreement, and subject to such rollover or right of first refusal, service by Natural to Shipper thereunder shall be terminated and automatically abandoned.

(c) Natural may terminate any NSS Agreement if Natural is required by the Commission or some other agency or court to provide firm service for others utilizing the system capacity or capability required for service under such NSS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the type covered by the NSS Agreement; provided, however, that nothing herein shall be read to protect Natural from the consequences of its own negligence.

5. INJECTION/WITHDRAWAL PARAMETERS

- 5.1 A Shipper's WQ defines the maximum quantity of gas which it has a firm right to withdraw on any day under this Rate Schedule NSS. If Shipper exceeds its firm withdrawal rights, it shall be subject to Overrun Charges as provided in Section 12 of the General Terms and Conditions of this Tariff. Except as provided in Section 5.7 and subject to Section 5.3(b), a Shipper's WQ shall vary with Shipper's inventory level in its NSS account as follows:
- (a) When Shipper's inventory level in its storage account under the NSS Agreement exceeds fifty percent (50%) of its MSV under the NSS Agreement, its WQ shall equal one hundred percent (100%) of its MDQ; and
- (b) When Shipper's inventory level in its storage account under the NSS Agreement is equal to or less than fifty percent (50%) of its MSV under the NSS Agreement, its WQ shall equal the lesser of seventy percent (70%) of its MDQ or the volume of gas remaining in storage for its account under the NSS Agreement.
- 5.2 A Shipper's MSV is the maximum volume of gas which it is entitled to have in storage under an NSS Agreement on a firm basis. A Shipper's MSV equals its MDQ under the NSS Agreement multiplied by seventy-five (75). If Shipper exceeds its MSV, Overrun Charges shall apply as set out in Section 12 of the General Terms and Conditions. In no event will a Shipper be allowed to withdraw more gas than it has in storage in the aggregate; i.e., Shipper may not create a negative storage inventory.

FERC Docket: RP08-319-000

Original Sheet No. 195 Original Sheet No. 195: Pending

RATE SCHEDULE NSS

- 5.3 (a) A Shipper's IQ shall be the maximum volume of gas such Shipper has the right to inject on any day under the NSS Agreement on a firm basis [subject to Section 5.7 and subsection (b) below]. If a Shipper exceeds its firm injection rights hereunder, such excess injections shall be subject to Overrun Charges under Section 12 of the General Terms and Conditions of this Tariff. A Shipper's IQ varies based on its inventory level, as follows:
- (1) When a Shipper's inventory level in its storage account under the NSS Agreement is less than sixty-seven percent (67%) of its MSV under the NSS Agreement, its IQ shall equal fifty percent (50%) of its MDQ; and
- (2) When a Shipper's inventory level in its storage account under the NSS Agreement is equal to or greater than sixty-seven percent (67%) of its MSV under the NSS Agreement, its IQ shall equal thirty-three percent (33%) of its MDQ.
- (b) (1) Subject to subsection (2) below, a Shipper's firm injection and withdrawal rights on any day shall be further limited as set out in this subsection. Each contract shall specify the percentage of service to which Shipper is entitled on the Amarillo Line (Amarillo Percentage) and the percentage of service to which Shipper is entitled on the Gulf Coast Line (Gulf Coast Percentage). The sum of such percentages shall be 100%. The Amarillo Percentage times a Shipper's IQ shall be available for injections from sources on the Amarillo Line and the Gulf Coast Percentage times a Shipper's IQ shall be available for injections from sources on the Gulf Coast Line. Similarly, the Amarillo Percentage times a Shipper's WQ shall be available for withdrawal from sources on the Amarillo Line and the Gulf Coast Percentage times a Shipper's WQ shall be available for withdrawal from sources on the Gulf Coast Line. Volumes which a Shipper seeks to have scheduled and Natural is able to schedule for either injection or withdrawal on either line in excess of the resulting daily firm volume hereunder or in excess of the restrictions in subsection (3) shall be treated as authorized overrun, with the same priority as other authorized overrun injections and withdrawals under Section 5.5(b) of this Rate Schedule NSS.

FERC Docket: RP08-319-000

Original Sheet No. 196 original Sheet No. 196: Pending

RATE SCHEDULE NSS

withdrawals by leg on a daily basis, overruns of daily rights by leg shall be evaluated on a monthly basis. Shipper shall only be subject to overrun charges with respect to such daily rights by leg if and to the extent its total injections or withdrawals by leg hereunder during any month exceed the sum of its daily injection or withdrawal rights, as applicable, for that month. If Natural accepts confirmed nominations for authorized overrun of such rights on a leg on any day, the volume authorized in excess of Shipper's firm daily rights shall be added to the sum of its daily rights for that leg for the month but, unless otherwise agreed, an equal volume shall be subtracted from the sum of daily rights for the other leg for that month. Notwithstanding the foregoing, Natural shall continue to evaluate overrun of total IQ and WQ under any Agreement, and assess any corresponding Overrun Charges, on a daily basis.

(3) Shipper's storage balance shall be divided into Amarillo and Gulf Coast components. Separate accounting shall be maintained by Natural of each component on a continuous basis. Shipper's total balance may not exceed its MSV except as otherwise provided in this Tariff. Volumes in the two components of Shipper's storage balance hereunder are subject to limits set out in the remainder of this subsection. Each contract shall specify the percentage of service to which Shipper is entitled on the Amarillo Line (Amarillo Percentage) and the percentage of service to which Shipper is entitled on the Gulf Coast Line (Gulf Coast Percentage). The sum of such percentage shall be 100%. Volumes from Amarillo Line sources residing in Amarillo Line storage points included in such balance may not at any time exceed Shipper's Amarillo Percentage times Shipper's MSV and volumes from Gulf Coast Line sources residing in Gulf Coast Line storage points included in such balance may not at any time exceed 107% of Shipper's Gulf Coast Percentage (up to 100%) times Shipper's MSV. A Shipper which breaches the limitations in this subsection (3) on any day shall pay a Sourcing Charge as set out on the rate sheets of this Tariff for that day multiplied by the excess volume (in Dth); provided, however, that no Sourcing Charge shall apply prior to April 1, 1996 (when trading rights are implemented). Payment of a Sourcing Charge hereunder shall not affect the location of any storage volumes nor shall any volumes be deemed transferred to storage points on the other leg as a result of Shipper having paid such a charge. If a Shipper's total storage balance hereunder exceeds its MSV, no Sourcing Charge shall apply, but ordinary Authorized or Unauthorized Overrun Charges under Section 12 of the General Terms and Conditions of this Tariff shall be assessed.

FERC Docket: RP08-319-000

Original Sheet No. 197 Original Sheet No. 197: Pending

RATE SCHEDULE NSS

(4) Shipper's NSS Agreement shall specify the percentage of service to which Shipper is entitled on the Amarillo Line (Amarillo Percentage) and the percentage of service to which Shipper is entitled on the Gulf Coast Line (Gulf Coast Percentage). The sum of the Amarillo Percentage and the Gulf Coast Percentage shall equal 100%. If Shipper requests separate MDQs for each pipeline leg, the request shall at Shipper's option be treated either as a request for separate contracts or as a request for separate MDQs per leg within a single contract, consistent with Section 2(d) of this Rate Schedule NSS. In either case, MDQs shall be separately applied by pipeline leg. Because of operational constraints on Natural's system, rights on the Amarillo Line in the aggregate for all NSS Shippers must be approximately 25% and rights on the Gulf Coast Line in the aggregate for all NSS Shippers must be approximately 75%, assuming Natural has sold all the NSS service its system is capable of providing.

(5) Rules regarding the allocation of injections and withdrawals by leg are set out in Section 5.6(c) of the General Terms and Conditions of this Tariff. In applying this subsection (b), all receipts on Natural's system on the Amarillo Line and north of Station 311 on the Gulf Coast Line shall be deemed to be sourced on the Amarillo Line and injected into Amarillo Line storage points and all other receipts shall be deemed to be sourced on the Gulf Coast Line and injected into Gulf Coast Line storage points, except as set out in subsection (6) below.

FERC Docket: RP08-319-000

Original Sheet No. 198 Original Sheet No. 198: Pending

RATE SCHEDULE NSS

(6) Notwithstanding the foregoing, Natural shall post prior to the beginning of each month, at least twenty-four (24) hours before first-of-the-month nominations are due, the extent (if any) to which receipts between Station 110 and Station 311 will be treated as Gulf Coast receipts. Natural will designate such receipts as Gulf Coast receipts for that month to the extent reasonably feasible operationally. Such designation may be reduced only prospectively through issuance of an Operational Flow Order or in the subsequent month's posting. Natural shall also review such designations daily and may change the designation by a posting prior to the time nominations are due for that day; provided, however, that the volume designated as Gulf Coast receipts in a monthly posting may not be reduced during a month by such daily postings. Natural shall also post prior to each month, at least twenty-four (24) hours before first-of-the-month nominations are due for that month, whether the restrictions on Shipper's cumulative storage balance on each leg (as a percentage of MSV) can be relaxed for a specified time period. In the event that Natural grants a waiver of restrictions on cumulative inventory by leg, Shipper's allocation of firm withdrawal rights by leg shall be modified to correspond to the increase in allowable inventory on the Amarillo Line. Any posting hereunder shall specify clearly which restrictions are being modified and the magnitude and duration of the modification.

- (7) This subsection shall govern the transfer of inventory. On any day, a Shipper can request that a specified volume be transferred from the Amarillo to the Gulf Coast components on its storage balance or vice versa within the limitations of Shipper's available rights to maintain storage inventory in the component of its storage balance to which the volume is being transferred, where such a transfer would avoid charges hereunder or alleviate an actual or potential breach of the limitations under subsection (3). To implement such a transfer, Shipper must specifically request such a transfer as part of its timely nomination for that day and Shipper must make available capacity to effectuate the transfer under an FTS or FTS-G Agreement with a primary path which allows Natural to make the requested transfer.
- 5.4 Service hereunder shall not be subject to any cycling requirement nor is Shipper required to inject any minimum inventory level.

FERC Docket: RP08-319-000

Original Sheet No. 199 Original Sheet No. 199: Pending

RATE SCHEDULE NSS

- 5.5 (a) A Shipper under this Rate Schedule NSS may request, as part of its nomination, injections in excess of IQ or in excess of its firm rights under Section 5.3(b) hereof or which would cause it to exceed its MSV and may request, as part of its nomination, withdrawals in excess of WQ or in excess of its firm rights under Section 5.3(b). Such withdrawals and injections are subject to Section 9 of this Rate Schedule NSS. Rights available under Section 5.5(c) are firm and need not be nominated to the extent of Shipper's no-notice rights hereunder in conjunction with FTS-NB.
- (b) Except as set out in subsection (c), such injections or withdrawals shall be treated as authorized overrun for purposes of scheduling and curtailment under the General Terms and Conditions of this Tariff; provided, however, that injections and withdrawals within MDQ and MSV shall be treated as higher priority than other authorized overrun and interruptible volumes (except corresponding authorized overrun requests under Rate Schedule DSS, which shall be of equal priority), consistent with the priority system set out in the General Terms and Conditions of this Tariff; and provided further that no authorized overrun charges apply unless MDQ and/or MSV is exceeded.
- (c) Forty-eight (48) hours prior to the beginning of each gas day, Natural shall post on the Informational Postings portion of its Interactive Website the availability of additional injection and withdrawal capability in excess of IQ and WQ, determined by taking into account both daily and seasonal storage parameters. Such availability shall be stated as percentage(s) of IQ and WQ and shall be separately stated by pipeline leg. Upon such posting, the storage injection and withdrawal capability stated as available in the posting shall be firm up to MDQ for that day provided neither Shipper's MDQ nor its MSV is exceeded (and shall be available for no-notice rights in conjunction with FTS-NB). If no posting is made for any day, the default volume for firm injection and withdrawal rights on that day shall be five percent (5%) of MDQ in excess of the otherwise applicable IQ and WQ provided neither Shipper's MDQ nor its MSV is exceeded.
- 5.6 In the event Shipper segments its primary point transportation path as a result of a capacity release under the separate transportation Agreement(s) being used to deliver gas to and receive gas from storage, Shipper's injection or withdrawal rights may be affected as provided in Section 19 of the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 200 Original Sheet No. 200: Pending

RATE SCHEDULE NSS

5.7 (a) If Shipper takes action based on inaccurate information provided by Natural regarding Shipper's IQ, WQ or MSV or related storage parameters and such action results in Shipper exceeding its rights, Natural will [except as provided in subsection (b)] waive any charge Shipper could reasonably have avoided had Natural provided accurate information.

(b) Shipper may rely on the most recent information provided by Natural to Shipper regarding any operating parameters, but Shipper shall have an obligation to provide Natural on a timely basis with relevant information within its control to be used in determining such parameters and charges under subsection (a) shall not be waived to the extent inaccurate information provided by Natural resulted from Shipper's failure to provide Natural with accurate information within Shipper's control on a timely basis.

RATE

- 6.1 Shipper shall pay Natural each month for service under this Rate Schedule NSS the sum of the following charges:
 - (a) A Reservation Charge based on Shipper's MDQ;
 - (b) A Storage Fuel Charge;
 - (c) Overrun Charges; and
- (d) Any other applicable charges and penalties (including but not limited to Balancing Service Charges) set out in this Rate Schedule or the General Terms and Conditions of this Tariff.

Separate Maximum and Minimum Reservation Charges are stated in this Tariff for Storage Expansion 2003 service (Storage Expansion 2003 Rates). Storage Expansion 2003 Rates shall apply to Storage Expansion 2003 Agreements. Storage Expansion 2003 Rates shall also apply to any new NSS Agreement which results from the Rollover of a Storage Expansion 2003 Agreement, the exercise of a Right of First Refusal with respect to a Storage Expansion 2003 Agreement or capacity which becomes available due to the expiration or termination of a Storage Expansion 2003 Agreement. Where NSS capacity becomes available due to the expiration or termination of a Storage Expansion 2003 Agreement it shall be separately identified for purposes of allocation in accordance with the provisions of this Tariff.

FERC Docket: RP09-309-000

First Revised Sheet No. 201 First Revised Sheet No. 201

Superseding: Original Sheet No. 201

RATE SCHEDULE NSS

Separate Maximum and Minimum Reservation Charges are stated in this Tariff for Sayre Storage Project service (Sayre Storage Project Rates). Sayre Storage Project Rates shall apply to Sayre Storage Project Agreements. Sayre Storage Project Rates shall also apply to any new NSS Agreement which results from the Rollover of a Sayre Storage Project Agreement, the exercise of a Right of First Refusal with respect to a Sayre Storage Project Agreement or capacity which becomes available due to the expiration or termination of a Sayre Storage Project Agreement. Where NSS capacity becomes available due to the expiration or termination of a Sayre Storage Project Agreement it shall be separately identified for purposes of allocation in accordance with the provisions of this Tariff.

Separate Maximum and Minimum Reservation Charges are stated in this Tariff for North Lansing NSS Project service (North Lansing NSS Project Rates). North Lansing NSS Project Rates shall apply to North Lansing NSS Project Agreements. North Lansing NSS Project Rates shall also apply to any new NSS Agreement which results from the Rollover of a North Lansing NSS Project Agreement, the exercise of a Right of First Refusal with respect to a North Lansing NSS Project Agreement or capacity which becomes available due to the expiration or termination of a North Lansing NSS Project Agreement. Where NSS capacity becomes available due to the expiration or termination of a North Lansing NSS Project Agreement it shall be separately identified for purposes of allocation in accordance with the provisions of this Tariff.

Separate Maximum and Minimum Reservation Charges are stated in this Tariff for Herscher Galesville NSS Project service (Herscher Galesville NSS Project Rates). Herscher Galesville NSS Project Rates shall apply to Herscher Galesville NSS Project Agreements. Herscher Galesville NSS Project Rates shall also apply to any new NSS Agreement which results from the Rollover of a Herscher Galesville NSS Project Agreement, the exercise of a Right of First Refusal with respect to a Herscher Galesville NSS Project Agreement or capacity which becomes available due to the expiration or termination of a Herscher Galesville NSS Project Agreement. Where NSS capacity becomes available due to the expiration or termination of a Herscher Galesville NSS Project Agreement it shall be separately identified for purposes of allocation in accordance with the provisions of this Tariff.

FERC Docket: RP09-309-000

Original Sheet No. 201A Original Sheet No. 201A

RATE SCHEDULE NSS

Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.

- 6.2 The charges identified in subsections (a) through (c) of Section 6.1 shall be assessed as follows:
- (a) The maximum Reservation Charge for service under this Rate Schedule NSS shall be the applicable maximum unit Reservation Rate(s) set out in this Tariff multiplied by Shipper's MDQ. The Reservation Charge payable by any Shipper shall be the applicable maximum Reservation Charge unless a lower Reservation

FERC Docket: RP08-319-000

Original Sheet No. 202 Original Sheet No. 202: Pending

RATE SCHEDULE NSS

Charge is specified in the NSS Agreement or a separate discount agreement or is assessed pursuant to Section 6.5 hereof, which lower charge must be consistent with the minimum rate(s) specified in this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma Agreement set out in this Tariff, discounts may be limited to specific volumes. The Reservation Charge shall constitute the minimum bill under this Rate Schedule NSS. Shipper shall be obligated to pay the Reservation Charge hereunder as of the date NSS service commences. NSS service shall be deemed to commence on the later of the date specified in the NSS Agreement or the date Natural is able to render firm service hereunder to Shipper.

- (b) Shipper shall provide gas in kind for the fuel required to inject gas into storage and withdraw gas from storage under this Rate Schedule NSS. Such charge shall equal the applicable Storage Fuel Rate set out in this Tariff (stated as a percentage fuel rate) multiplied by each Dth of gas injected under the NSS Agreement. Upon the mutual agreement of Natural and Shipper, in lieu of Natural retaining gas in kind, Shipper shall reimburse Natural for fuel at a price established under Section 7.1 of the General Terms and Conditions of this Tariff multiplied by the Storage Fuel Charge.
- (c) Except as otherwise provided herein, Shipper shall pay the Authorized Overrun Rate or the Unauthorized Overrun Rate, as applicable, specified in Section 12 of the General Terms and Conditions of this Tariff on the volume of any overrun of its firm rights under Section 5 of this Rate Schedule NSS.
- 6.3 Shipper shall reimburse Natural within five (5) business days after costs have been incurred by Natural for all fees required by the Commission or any regulatory body including, but not limited to, filing, reporting, and application fees to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees (such as general rate case filing fees). Any costs collected hereunder shall be excluded from Natural's cost of service in determining generally applicable fees.
- 6.4 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in (1) the rates and charges applicable under this Rate Schedule NSS, including both the level and design of such rates and charges; or (2) the terms and conditions of this Rate Schedule NSS. Natural agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.

FERC Docket: RP08-319-000

Original Sheet No. 203 Original Sheet No. 203: Pending

RATE SCHEDULE NSS

- (b) If, at any time and from time to time, the Commission or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the NSS Agreement or any separate discount agreement, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such highest rate, Shipper shall execute or provide such documentation within fifteen (15) calendar days after a written request by Natural. If, at any time and from time to time, the Commission or any other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for service hereunder, the rate shall be decreased to such reduced rate.
- Natural may from time to time and at any time, upon twenty-four (24) hours' verbal or written notice, subject to any provisions on discounting in the NSS Agreement or in a separate discount agreement, charge any individual Shipper for service under this Rate Schedule NSS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate charged may not be less than the applicable minimum rate for service under this Rate Schedule NSS set forth in this Tariff. Natural will confirm any verbal notice of the applicable rate in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Unless otherwise agreed in the NSS Agreement or in a separate discount agreement, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the NSS Agreement and/or any discount agreement) upon twenty-four (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with the Commission any and all reports as required by the Commission's Regulations with respect to the institution or discontinuance of any discount.
- 6.6 All revenues and gas in kind collected by Natural as a result of providing service under this Rate Schedule NSS shall be retained by Natural unless Natural has otherwise explicitly agreed on a different disposition of such amounts. Where crediting or refund mechanisms apply under other provision(s) of this Tariff or pursuant to effective Commission orders or settlements, such mechanisms shall supersede this Section to the extent necessary to carry out such provision(s).

FERC Docket: RP08-319-000

Original Sheet No. 204 Original Sheet No. 204: Pending

RATE SCHEDULE NSS

7. NOMINATIONS, INJECTIONS AND WITHDRAWALS

- (a) Shipper shall provide Natural with nominations of injections and withdrawals in accordance with the General Terms and Conditions of this Tariff; provided, however, that injections and withdrawals hereunder shall be available on a no-notice basis to the limited extent permitted in conjunction with firm transportation if Shipper makes the appropriate election of the NB service option under Rate Schedule FTS or FTS-G, subject to the limitations on such no-notice rights set out in Rate Schedule FTS or FTS-G and the General Terms and Conditions of this Tariff. Except for such no-notice rights, it shall be Shipper's responsibility to cause gas to be delivered to Natural for injection into storage hereunder and to cause gas to be taken from Natural upon withdrawal hereunder in accordance with the information supplied to Natural.
- (b) The Delivery Points for gas injected and the Receipt Points for gas withdrawn hereunder shall be storage point(s) defined in this Tariff. Each NSS Agreement shall have as Delivery and Receipt Points all such storage point(s) and will be subject to the limitations and conditions (including volumes and priorities) specified in the General Terms and Conditions of this Tariff relating to such points. The injection, withdrawal and location of inventory under any NSS Agreement must be consistent with Shipper's right under that NSS Agreement for the Amarillo Line and for the Gulf Coast Line. Volumes injected at any storage point must be withdrawn at the same storage point. Conditions relating to injections and withdrawals at any storage point shall be determined pursuant to Section 5 hereof and the General Terms and Conditions of this Tariff.
- (c) Storage Fuel shall be provided by Shipper upon injection hereunder by reducing the volumes deemed delivered into Shipper's NSS inventory under the separate transportation Agreement used for injection therein and Storage Fuel shall not subsequently be taken into account in determining Shipper's inventory or withdrawal rights under this Rate Schedule NSS.
- (d) Natural will provide all base gas required to support withdrawals under Rate Schedule NSS. Shippers shall provide working gas for service hereunder as provided in Section 5 of this Rate Schedule NSS.

FERC Docket: RP08-319-000

Original Sheet No. 205 Original Sheet No. 205: Pending

RATE SCHEDULE NSS

8. RELATED TRANSPORTATION ARRANGEMENTS/WITHDRAWAL UPON TERMINATION

- Shipper shall arrange separately for the transportation of gas to and from Natural's storage points and shall make all necessary arrangements separately with Natural or other parties upstream of the point where gas is tendered to Natural for injection into storage hereunder and downstream of the point where the gas is withdrawn from storage by Natural for the account of Shipper. Such arrangements must be consistent with this Rate Schedule NSS and must be coordinated with Natural. To be eligible for service hereunder, Shipper or its supplier or purchaser must comply with all applicable provisions of any separate transportation Agreement(s) and any applicable Rate Schedule(s) and the applicable General Terms and Conditions and the injection or withdrawal of gas hereunder shall be subject to confirmation, scheduling and curtailment procedures affecting such transportation. The injection, withdrawal and location of storage inventory under any NSS Agreement must be consistent with Shipper's rights under that NSS Agreement for the Amarillo Line and for the Gulf Coast Line. Storage points hereunder shall be treated as inthe-path, and thus have the same priority as primary points, to the extent provided in Section 5.6 of the General Terms and Conditions of this Tariff. Storage points hereunder may be Receipt and Delivery Points under the separate transportation Agreements used for injection and withdrawal hereunder, with the priority determined pursuant to Section 5.6 of the General Terms and Conditions of this Tariff. Such transportation Agreements may designate NSS storage point(s) as the Delivery Point(s) during a specified injection season, with the applicable priority, and a market area point as the primary Delivery Point(s) for a defined withdrawal season.
- (b) Shipper must have all of its gas withdrawn by the end of the term of a NSS Agreement (or Agreement under a successor service) which is not rolled over or extended. Such withdrawals may be effectuated through an in-field transfer, subject to the then applicable rules governing such transfers. If at any time service pursuant to this Rate Schedule NSS (or successor service) and/or a NSS Agreement (or Agreement under a successor service) is cancelled or terminated prior to the end of the term specified in the NSS Agreement (or successor Agreement), Shipper shall be required to withdraw its gas over a period of time determined by dividing: (1) the volume of gas Shipper is required to withdraw, by (2) fifty percent (50%) of Shipper's MDQ; provided, however, that if service is being terminated pursuant to Section 19.19 of the General Terms and Conditions of this Tariff, the withdrawal

FERC Docket: RP08-319-000

Original Sheet No. 206 original Sheet No. 206: Pending

RATE SCHEDULE NSS

period shall be equal in duration to the prior notice period required under said Section 19.19. If Natural is unable to withdraw any volume requested by Shipper, up to the volume specified in the preceding sentence, the period shall be extended accordingly. If the gas is not withdrawn within the specified time period, the remaining volumes shall be cashed out without any offsets, trading or netting, and the proceeds distributed to Shipper, based on the greater than twenty percent (20%) imbalance tier set out at Section 13.3 of the General Terms and Conditions of this Tariff.

9. OVERRUN SERVICE

Upon request of Shipper, Natural may (but is not obligated to) receive for injection hereunder as authorized overrun on any day quantities of natural gas in excess of Shipper's IQ or which would cause Shipper's storage inventory to exceed its MSV and/or withdraw for delivery hereunder on any day quantities of natural gas in excess of Shipper's WQ, as applicable, when in Natural's reasonable judgment, the capacity and operating capability of its system will permit such injection and/or delivery without impairing the ability of Natural to meet its other service obligations; provided, however, that in no event will Shipper be allowed to withdraw more gas than it has in storage. In granting requests for Authorized Overrun Service, Natural shall act in a manner consistent with the overrun service priorities set out in the General Terms and Conditions of this Tariff, which provisions incorporate a higher priority for injections or withdrawals within MDQ. Charges applicable to authorized or unauthorized overrun volumes shall be as set out in Section 12 of the General Terms and Conditions of this Tariff.

10. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule NSS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Original Sheet No. 207 Original Sheet No. 207: Pending

RATE SCHEDULE PALS PARK AND LOAN SERVICE

1. AVAILABILITY

- (a) This Park and Loan Service (PALS) Rate Schedule is an interruptible service available to any entity (hereinafter called Shipper) which submits to Natural Gas Pipeline Company of America LLC (hereinafter called Natural):
- (1) A valid request for service under this Rate Schedule PALS as defined in Section 4 hereof and executes an agreement for such service (PALS Agreement); and
- (2) A valid PALS Request Order (PALS RO), as defined in Section 4 hereof which, when executed by Natural and Shipper, shall evidence their agreement as to the terms of the particular transaction(s) to park and loan gas pursuant to the PALS Agreement.
- (b) Shipper shall arrange separately with Natural and others as necessary for any transportation attendant to the PALS service provided hereunder and Shipper shall pay separately for such transportation service.
- (c) Shipper shall provide Natural with reasonable assurances that Shipper can satisfactorily perform under an applicable PALS RO.

2. APPLICABILITY AND CHARACTER OF SERVICE

(a) This Rate Schedule PALS shall apply to all PALS services which are rendered by Natural pursuant to an executed PALS Agreement and related PALS RO, and shall permit a Shipper to nominate a quantity of gas to any mutually agreeable Receipt or Delivery Point on Natural's system, to be held or loaned by Natural for a specified period defined in the PALS RO. It is understood that Natural is providing this PALS service through the use of its general system flexibility, including system storage gas; Natural is not providing a supply service under this Rate Schedule PALS. Contracting for and nominating service to and from the designated point(s) shall be the Shipper's sole responsibility.

FERC Docket: RP08-319-000

Original Sheet No. 208 Original Sheet No. 208: Pending

RATE SCHEDULE PALS

(b) Under this Rate Schedule PALS, Natural shall only park gas or loan gas to the extent Natural determines that such actions are not detrimental to its ability to satisfy any of its existing obligations with higher priority or to meet system operational needs. Natural may, based on its reasonable determination of its operational capability and in a non-discriminatory manner, interrupt or decline to schedule any or all of the services hereunder and, if such actions are required to avoid interference with firm service or to protect the integrity of the system, will do so prior to invoking the procedures of Section 23, Operational Control, of the General Terms and Conditions of this Tariff and subject to Section 9 hereof.

- (c) All mutually agreeable points of receipt and delivery on Natural's system applied in a non-discriminatory manner are available for service under this Rate Schedule PALS; provided, however, that the same point must be utilized to complete a specific transaction. The specific Receipt Point and Delivery Point for a park or loan shall be set forth in the applicable PALS RO.
- (d) The Park and Loan Service provided for under this Rate Schedule PALS shall be performed under Part 284 of FERC's Regulations. Shipper shall only tender gas under this Rate Schedule PALS to the extent service hereunder would qualify under the applicable statutes, regulations, FERC orders and the blanket certificate authorizing service by Natural under this Rate Schedule PALS. For service under Subpart B of Part 284, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by FERC's Regulations, Shipper shall cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit any necessary certification.
- (e) Subject to the provisions of Section 2(b) above, Park and Loan Services available under this Rate Schedule PALS include:
- (1) Park Service: shall consist of Natural's receipt of a quantity of natural gas at the designated Receipt Point(s) on the designated date, requested by Shipper under a PALS RO and approved by Natural, Natural's holding of such parked quantity of gas for Shipper's account and Natural's redelivery of the parked quantity of gas to Shipper at the designated Delivery Point(s) and on the designated date(s) set forth in such PALS RO.

FERC Docket: RP08-319-000

Original Sheet No. 209 Original Sheet No. 209: Pending

RATE SCHEDULE PALS

(2) Loan Service: shall consist of Natural lending a specified quantity of natural gas, requested by Shipper and approved by Natural, from designated Delivery Point(s) set forth in Shipper's PALS RO and the Shipper's redelivery of and Natural's acceptance of such volumes for Shipper's account at the designated Receipt Point(s) on the designated date(s) set forth in such PALS RO.

(f) Natural will post on the Informational Postings portion of its Interactive Website by receipt zone on a daily basis whether or not it is accepting requests for parks or loans under this Rate Schedule PALS.

3. PRIORITY OF SERVICE

- (a) Service under Rate Schedule PALS is provided on an interruptible basis. Priorities of service for the purposes of scheduling and curtailment shall be governed by Section 5 of the General Terms and Conditions of this Tariff as follows:
- (1) Service under Rate Schedule PALS shall first be evaluated under the storage service priorities set forth in Sections 5.3(b) and 5.4(b) of the General Terms and Conditions of this Tariff
- (2) If applicable, service under Rate Schedule PALS requiring transportation service or necessitating transportation by Natural shall be evaluated under the transportation service priorities set forth in Section 5.6 or 5.7 of the General Terms and Conditions of this Tariff.
- (b) In the event it is necessary to decline to schedule or to curtail or to suspend service under PALS because of operational conditions or to satisfy obligations with a higher priority, Natural shall provide actual notice to Shipper. In that event, Shipper must comply with the directive(s) contained in Natural's notification within the time specified.
- (c) If Shipper fails to comply with the requirements set out in a notification under Section 3(b) above, then Section 9 of this Rate Schedule PALS shall apply.

FERC Docket: RP08-319-000

Original Sheet No. 210 original Sheet No. 210: Pending

RATE SCHEDULE PALS

VALID REOUESTS FOR PALS AGREEMENT(S) AND FOR PALS RO(S)

- 4.1 A request for service under this Rate Schedule PALS shall be valid as of the date received if it complies with this Section 4 and contains adequate information on all of the items specified in Sections 4.2(a) and 4.2(b), subject to any necessary verification of such information and to the following:
- (a) A request shall not be valid and Natural shall not be required to grant any such request: (1) which could in Natural's judgment interfere with efficient operation of its system or with service to any firm Shipper; (2) which would require the construction, modification, expansion, or acquisition of any facilities; provided, however, that Natural may agree in its reasonable discretion to construct, modify, expand, or acquire facilities to enable it to perform such services; (3) unless and until Shipper has provided Natural with the information required in Section 4.2 hereof and the assurances required under Section 1(c) hereof; (4) if Natural determines, based on the credit analysis referenced in Section 4.2(d), that Shipper does not possess sufficient financial stability to make it reasonably likely the service provided hereunder will be paid for on a timely basis; (5) if the service requested would not comply with this Rate Schedule PALS; or (6) if the service requested is at less than the applicable maximum rate; provided, however, that Natural may agree to provide service hereunder at a discount consistent with this Rate Schedule PALS. Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to Section 4 hereof and while Agreements under this Rate Schedule are in effect. Such matters are governed by Section 3 of this Rate Schedule and the applicable General Terms and Conditions of this Tariff.
- (b) In the event that Natural agrees to construct, modify, expand or acquire facilities to perform service under this Rate Schedule PALS, Natural shall maintain a separate record of the nature and costs of such facilities.

FERC Docket: RP08-319-000

Original Sheet No. 211 Original Sheet No. 211: Pending

RATE SCHEDULE PALS

(c) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request, in whole or in part. Any request shall be null and void unless it is substantially complete and complies with this Rate Schedule. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the PALS Agreement, after which Shipper shall have fifteen (15) days to provide the specified information. In the event such information is not received within fifteen (15) days, Shipper's request shall be null and void.

- (d) Natural shall tender a PALS Agreement to Shipper for execution when Shipper's request for service is accepted. Unless waived by Natural, a request for service shall be invalid if Shipper fails to execute a PALS Agreement hereunder within ten (10) days after the PALS Agreement has been tendered by Natural for execution. An executed PALS RO must be submitted to Natural prior to commencement of service.
- 4.2 Requests for service hereunder shall be deemed valid only after the following information is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department, One Allen Center, 500 Dallas Street, Houston, Texas 77002, or Telecopy Number (713)369-9305:

(1) GAS QUANTITIES

The request shall specify in Dth the maximum aggregate quantity (MAQ) to be parked or loaned under any and all outstanding PALS RO Agreements.

(2) TERM OF SERVICE

The request shall specify the date service is requested to commence and to terminate (primary term), subject to an evergreen provision;

FERC Docket: RP08-319-000

Original Sheet No. 212 Original Sheet No. 212: Pending

RATE SCHEDULE PALS

(3) CREDIT

 $\label{eq:Acceptance} \text{Acceptance of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.}$

(4) COMPLIANCE WITH PALS TARIFF

Submission of a request for service hereunder shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule PALS, including the applicable General Terms and Conditions for any Park and Loan Service provided under the PALS Agreement and any related PALS RO.

(5) COMMISSION-REQUIRED FILING INFORMATION

The following information is to be provided at the time of a request for service hereunder is submitted, if available and shall be updated when any PALS RO is executed:

(i) Affiliation of the Shipper with

Natural;

 $\,$ (ii) The identity of the Shipper, including whether it is a local distribution company, an end user, a producer, or a marketer.

- (6) SHIPPER CONTACT PERSONNEL
- - (1) SERVICE TYPE
 - (2) GAS QUANTITIES

The PALS RO shall specify in Dth the maximum aggregate quantity (MAQ RO) and the daily quantity to be parked and/or loaned under the specific transaction; provided that the sum of all MAQ ROs under all pending PALS ROs cannot exceed the maximum aggregate quantity under the applicable PALS Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 213 Original Sheet No. 213: Pending

RATE SCHEDULE PALS

(3) RELATED PALS AGREEMENT

 $\,$ The PALS RO shall specify the PALS Agreement to which the PALS RO relates.

(4) POINTS

The request shall specify the Receipt Point(s) at which Natural will receive the gas from Shipper and the Delivery Point(s) at which Natural will deliver the gas to Shipper. Any mutually agreeable point(s) on Natural's system may be utilized for service under this Rate Schedule PALS.

(5) TERM OF SERVICE

The request shall specify:

(i) The date service is requested to

commence; and

(ii) The date service is requested to

terminate.

- (6) SHIPPER CONTACT PERSONNEL
- (7) RATE

5. TERM

- (a) The term of service hereunder shall be set forth in the PALS Agreement between Shipper and Natural. The PALS RO shall have a separately stated term applicable to a particular transaction which term may not extend beyond the term of the related PALS Agreement. Upon termination of any PALS Agreement and of any PALS RO, service by Natural to Shipper thereunder shall be terminated and automatically abandoned.
- (b) Natural may terminate any PALS Agreement if Natural is required by the FERC or some other agency or court to provide service for others utilizing the interruptible system capacity or capabilities required for service under such PALS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the type covered by the PALS Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 214 Original Sheet No. 214: Pending

RATE SCHEDULE PALS

6. RATE

6.1 (a) For the Park and Loan Service rendered to Shipper under this Rate Schedule PALS, Shipper shall pay Natural each month a daily usage charge for each unit of gas which is parked or loaned under this Rate Schedule PALS. Such charge shall be calculated on the basis of the daily ending balance for each PALS RO. The maximum and minimum rate(s) applicable to this Rate Schedule PALS are set forth in the Currently Effective Rates section of this Tariff. By mutual agreement between Natural and Shipper which is consistent with the pro forma agreement set out in this Tariff, discounts may be limited to specific volumes.

(b) STORAGE FUEL CHARGE

Shipper shall provide gas in kind for the fuel required to provide injections and withdrawals under this Rate Schedule PALS. Such charge shall be determined by multiplying percentage fuel rates by the volume of gas injected or withdrawn. Specifically, the Storage Fuel Charge shall equal the sum of: (i) the applicable unit Injection Storage Fuel Rate set out in this Tariff multiplied by each Dth of gas injected under the PALS RO; and (ii) and the applicable unit Withdrawal Storage Fuel Rate set out in this Tariff multiplied by each Dth of gas withdrawn under the PALS RO.

- (c) Shipper shall pay any other applicable charges, penalties and fees set out in this Rate Schedule PALS or the General Terms and Conditions of this Tariff. Deviations from the approved PALS RO volumes when compared to the allocated volumes will be assessed Balancing Service Charges under Section 12 of the General Terms and Conditions of this Tariff.
- (d) Charges payable by any Shipper shall be based on the maximum rates set forth in this Tariff applicable to Rate Schedule PALS, which rates are hereby incorporated herein, unless a lower rate for the charge in subsection (a) is specified in the PALS RO.
- (e) The charges referenced herein cover only Park and Loan Services. $% \left\{ 1,2,\ldots,4\right\}$
- (f) Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.

FERC Docket: RP08-319-000

Original Sheet No. 215 Original Sheet No. 215: Pending

RATE SCHEDULE PALS

- 6.2 (a) Shipper shall reimburse Natural within five (5) days after costs have been incurred by Natural for all fees required by the FERC or any regulatory body including, but not limited to, filing, reporting, and application fees to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees (such as general rate case filing fees).
- (b) If Natural constructs, acquires or modifies any facilities to perform service hereunder, then as specified in an agreement related thereto between the parties either:
- (1) Shipper shall reimburse Natural for the cost of such facilities or facility modifications as described in the General Terms and Conditions of this Tariff; or
- $\hbox{\ensuremath{\mbox{(2)}}$ Natural shall assess a monthly charge reflecting such facility costs.}$
- 6.3 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in: (1) the rates and charges applicable under this Rate Schedule PALS, including both the level and design of such rates and charges; and/or (2) the terms and conditions of this Rate Schedule PALS. Natural agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.
- (b) If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the PALS Agreement or PALS RO, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such highest rate, Shipper shall execute or provide such documentation within fifteen (15) days after a written request by Natural. If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for service hereunder, the rate shall be decreased to such reduced rate.

FERC Docket: RP08-319-000

Original Sheet No. 216 original Sheet No. 216: Pending

RATE SCHEDULE PALS

Natural may from time to time and at any time, upon twenty-four (24) hours' verbal or written notice, subject to any provisions on discounting in the PALS Agreement or PALS RO, charge any individual Shipper for service under this Rate Schedule PALS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate charged may not be less than the applicable minimum rate for service under Rate Schedule PALS set forth in this Tariff. Unless otherwise agreed in the PALS Agreement or PALS RO, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the PALS Agreement and/or PALS RO) upon twenty-four (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with FERC any and all reports as required by FERC's Regulations with respect to the institution or discontinuance of any discount.

6.5 All revenues and gas in kind collected by Natural as a result of providing service under this Rate Schedule PALS shall be retained by Natural unless Natural has otherwise explicitly agreed on a different disposition of such amounts. Where crediting or refund mechanisms apply under other provision(s) of this Tariff or pursuant to effective FERC orders or settlements, such mechanisms shall supersede this Section to the extent necessary to carry out such provision(s).

7. QUANTITY

Each PALS RO shall specify in Dth the maximum aggregate quantity (MAQ RO), and a daily quantity to be parked and/or loaned under the specific transaction. The daily schedule of returned volumes by the Shipper or Natural shall also be specified in the PALS RO. The daily quantity shall be the maximum quantity Natural is obligated, on an interruptible basis, to receive from or deliver to Shipper hereunder on the specified day. The MAQ RO shall be the quantity Natural is obligated to hold or loan for the account of Shipper hereunder on an interruptible basis for the specific transaction covered by the PALS RO. The sum total of a Shipper's MAQ ROs shall not exceed the maximum aggregate quantity specified in the PALS Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 217 Original Sheet No. 217: Pending

RATE SCHEDULE PALS

NOMINATIONS

It shall be Shipper's sole responsibility to provide Natural with daily nominations of the quantity of gas to be received or delivered at the Park or Loan point(s) under the applicable PALS RO Agreement. It shall be Shipper's responsibility to cause gas to be delivered to Natural and to cause gas to be received from Natural in accordance with the PALS RO. Nominations shall be subject to confirmation and scheduling in accordance with the General Terms and Conditions of this Tariff.

MANDATORY BALANCING

- 9.1 Mandatory Balancing shall apply in the following instances: (a) at the end of the term specified in any applicable PALS Agreement and PALS RO; (b) where the Shipper fails to comply either with the requirements of Natural's notice referenced in Sections 3 and 9.3(b) hereof; or (c) where Shipper fails to comply with the schedule of activities set forth in the applicable PALS RO and the deviation has not been agreed to and confirmed by Natural.
- 9.2 Natural shall require Mandatory Balancing effective the next day prior to declaring Critical Time or issuing Operational Flow Orders under Section 23, Operational Control, of the General Terms and Conditions of this Tariff, if Natural reasonably determines that doing so would facilitate system operations and minimize the frequency and severity of Critical Time or Operational Flow Orders in the affected region(s).
- (a) In the event that Natural notifies a PALS Shipper under Section 3 hereof, such notice shall specify the parked balance to be removed or the loaned balance to be returned, up to the full MAQ, and the timeframe within which the balance must be effectuated, but in no event shall the specified timeframe be less than three (3) days (in one-third daily increments) from the date of notification. Natural may allow additional time for contract balancing when operational conditions permit. Notification shall first be provided by telephone and then by facsimile, by e-mail or in writing. General notices will be posted on the Informational Postings portion of Natural's Interactive Website. In instances when notification is required during times other than normal $% \left(1\right) =\left(1\right) \left(1\right)$ business hours, Natural will provide such notification by telephone. To the extent Shipper fails to comply with such notice, the PALS RO shall terminate and the provisions of Section 9.4 shall applv.

FERC Docket: RP08-319-000

Original Sheet No. 218 Original Sheet No. 218: Pending

RATE SCHEDULE PALS

(b) In the event that a Shipper fails to comply with the schedule of activities set forth in the applicable PALS RO, Natural shall notify Shipper and the PALS RO shall be subject to termination in accordance with the terms of such notice. The provisions of Sections 9.4(a) and 9.4(b) shall then apply.

- 9.4 (a) In the event that Shipper still has gas parked at the end of its contract term or because of failure to comply with the notice pursuant to Sections 3 and 9.2 above, the remaining balance shall be forfeited to Natural, free and clear of any adverse claims.
- (b) Conversely, if the Shipper has not redelivered gas which was loaned by Natural by the end of its contract term or within the timeframe specified in the notice in Sections 3 and 9.3 above, the Shipper must purchase the unreturned balance at 150% of the average Weekly Index Prices (WIPs) as determined in Section 13.3 of the General Terms and Conditions of this Tariff during the term of the loan. In the event that the information used to determine the WIPs for the term of the loan is not available, then the Shipper must purchase the unreturned balance at 150% of the Average Monthly Index Price (AMIP) as determined in Section 13.3 of the General Terms and Conditions of this Tariff. If Critical Time or Operational Flow Orders are in effect on the date Shipper is required to comply with the notice, Shipper shall be subject to the highest charges set forth in Sections 12.3 and 12.4 of the General Terms and Conditions of this Tariff.
- 9.5 In circumstances where Shipper is unable to eliminate its PALS RO balance because Natural is unable to accept the PALS RO nomination, Shipper shall take any action to reduce the balance which Natural can accommodate and Shipper shall be granted additional time to eliminate its PALS RO balance corresponding to the time Natural was unable to accept the PALS RO nomination. Shipper shall not incur any penalty or daily fees as to that portion of the balance resulting from Natural's inability to accept the PALS RO nomination. Such extension shall only apply in instances where a Shipper's inability to eliminate the PALS RO balance is attributable to Natural's inability to accept and confirm PALS RO nominations.

FERC Docket: RP08-319-000

Original Sheet No. 219 Original Sheet No. 219: Pending

RATE SCHEDULE PALS

Shipper remains responsible to nominate the appropriate level of transportation to meet Natural's notification requirements.

10. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule PALS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Original Sheet No. 220 Original Sheet No. 220: Pending

RATE SCHEDULE LPS LINE PACK SERVICE (PARK AND LOAN)

1. AVAILABILITY

- (a) Line Pack Service (Park and Loan), Rate Schedule LPS, is an interruptible service available to any entity (hereinafter called Shipper) which:
- (1) Submits to Natural Gas Pipeline Company of America LLC (hereinafter called Natural) a valid request for service under this Rate Schedule LPS as defined in Sections 4.1 and 4.2 hereof which results in execution of an agreement for such service (LPS Agreement); and
- (2) Enters into one or more valid LPS Request Orders (LPS RO) as defined in Section 4.3 hereof which, when executed by Natural and Shipper, shall evidence their agreement as to the terms of the particular transaction(s) to park or to loan gas pursuant to the LPS Agreement.
- (b) Shipper shall arrange separately with Natural or others as necessary for any transportation attendant to the service provided hereunder and Shipper shall pay separately for such transportation service.
- (c) Shipper shall provide Natural with reasonable assurances that Shipper can satisfactorily perform under each LPS ${\tt RO}$.

2. APPLICABILITY AND CHARACTER OF SERVICE

(a) This Rate Schedule LPS shall apply to all park and loan services which are rendered by Natural pursuant to an executed LPS Agreement and any related LPS RO. Under Rate Schedule LPS, a Shipper may nominate a quantity of gas to mutually agreeable point(s) on Natural's system; to be held or loaned by Natural for a period defined in the LPS RO. It is understood that Natural is providing service hereunder based on operating conditions on its system and line pack, without relying on storage; Natural is not providing a supply service under this Rate Schedule LPS, nor is Natural providing a storage or transportation service under this Rate Schedule LPS.

FERC Docket: RP08-319-000

Original Sheet No. 221 Original Sheet No. 221: Pending

RATE SCHEDULE LPS

- (b) Service hereunder is subject to the priority provisions set out in Section 3 hereof and to Mandatory Balancing under Section 9 of this Rate Schedule.
- (c) All mutually agreeable points of receipt and delivery on Natural's system (excluding storage points, points on non-contiguous offshore facilities and other offsystem points) are available on a non-discriminatory basis for service under this Rate Schedule LPS. Unless an alternative point is agreed upon by Natural and the Shipper, the same point must be utilized both to initiate and to complete a specific park or loan transaction. Where Natural and a Shipper agree to use a different point or points for completion of a specific park or loan transaction entered into after July 31, 2001, related transportation is required to be separate. The specific point(s) for a park or loan shall be set forth in the applicable LPS RO.
- (d) The Line Pack Service (Park and Loan) provided for under this Rate Schedule LPS shall be performed under Part 284 of FERC's Regulations. Shipper shall only tender gas under this Rate Schedule LPS to the extent service hereunder would qualify under the applicable statutes, regulations, FERC orders and the blanket certificate authorizing service by Natural under this Rate Schedule LPS. For service under Subpart B of Part 284, Shipper shall provide to Natural appropriate certification, including sufficient information in order for Natural to verify that the service qualifies under Subpart B of Part 284 of the Regulations. Where required by FERC's Regulations, Shipper shall cause the intrastate pipeline or local distribution company on whose behalf the service will be provided to submit any necessary certification.
- (e) Line Pack Services (Park and Loan) available under this Rate Schedule LPS include:
- (1) Park Service: shall consist of Natural's receipt of a quantity of natural gas at the designated Receipt Point(s) on the designated date(s), requested by Shipper under a LPS RO and approved by Natural, Natural's holding of such parked quantity of gas for Shipper's account and Natural's redelivery of the parked quantity of gas to Shipper at the designated Delivery Point(s) and on the designated date(s) set forth in such LPS RO.

FERC Docket: RP08-319-000

Original Sheet No. 222 Original Sheet No. 222: Pending

RATE SCHEDULE LPS

(2) Loan Service: shall consist of Natural lending a specified quantity of natural gas, requested by Shipper and approved by Natural, on the designated date(s) from designated Delivery Point(s) set forth in Shipper's LPS RO and the Shipper's redelivery of and Natural's acceptance of such volumes for Shipper's account at the designated Receipt Point(s) on the designated date(s) set forth in such LPS RO.

PRIORITY OF SERVICE

- (a) Service under Rate Schedule LPS is provided on an interruptible basis. Priorities of service for the purposes of scheduling and curtailment shall be governed by Section 5 of the General Terms and Conditions of this Tariff. Under this Rate Schedule LPS, Natural shall only park gas or loan gas to the extent Natural determines that such actions will not adversely affect its ability to satisfy any of its firm service obligations or to meet system operational needs. Natural may, in a non-discriminatory manner, interrupt or decline to schedule any or all of the services hereunder and, if such actions are required to avoid interference with firm service or to protect the integrity of the system, will do so prior to invoking the procedures of Section 23, Operational Control, of the General Terms and Conditions of this Tariff.
- (b) In the event it is necessary to interrupt, curtail or suspend service under any LPS RO because of operational conditions or to satisfy obligations with a higher priority, Natural shall provide actual notice to Shipper. In that event, Shipper must comply with the directive(s) contained in Natural's notification within the time specified in the notice, consistent with the remainder of this subsection (b).
- (1) Without limitation of the foregoing, Natural shall to the extent practicable provide such notice effective no later than the next day prior to declaring Critical Time or issuing Operational Flow Orders under Section 23 of the General Terms and Conditions of this Tariff, Operational Control, if Natural reasonably determines that doing so would facilitate system operations and minimize the frequency and severity of Critical Time or Operational Flow Orders in the affected region(s).

FERC Docket: RP08-319-000

Original Sheet No. 223 Original Sheet No. 223: Pending

RATE SCHEDULE LPS

(2) The notice under this subsection (b) shall specify the parked balance to be removed or the loaned balance to be returned, up to the full volume being parked or loaned, and the time frame within which the balance must be effectuated, but in no event shall the specified time frame be less than three (3) days (in one-third daily increments from the date of notification). Natural may allow additional time for contract balancing when operational conditions permit. Notification shall first be provided by telephone and then by facsimile, by e-mail or in writing. General notices will be posted on the Informational Postings portion of Natural's Interactive Website. In instances when notification is required during times other than normal business hours, Natural will provide such notification by telephone.

- (3) Once service hereunder has been scheduled, that service shall not be interrupted or curtailed to satisfy other interruptible services, except for priority interruptible rights under firm Agreements s identified in Section 5.7(a) of the General Terms and Conditions of this Tariff.
- (c) If the Shipper fails to comply with the requirements set out in a notification under Section 3(b) above, then the LPS RO shall terminate and Section 9 of this Rate Schedule LPS shall apply.
- (d) Natural shall also provide Shipper as much prior notice as practicable if it will be unable to schedule parks or loans pursuant to an LPS $\ensuremath{\mathsf{RO}}$.
- 4. VALID REQUESTS FOR LPS AGREEMENT(S); LPS RO(S)
 - 4.1 A request for service under this Rate Schedule LPS shall be valid as of the date received if it complies with this Section 4.1 and contains adequate information on all of the items specified in Section 4.2, subject to any necessary verification of such information and to the following:
 - (a) A request shall not be valid and Natural shall not be required to grant any such request: (1) which could in Natural's judgment interfere with efficient operation of its system or with service to any firm Shipper; (2) which would require the construction, modification, expansion, or acquisition of any facilities; provided, however, that Natural may agree in its reasonable discretion to construct, modify, expand, or acquire

FERC Docket: RP08-319-000

Original Sheet No. 224 Original Sheet No. 224: Pending

RATE SCHEUDLE LPS

facilities to enable it to perform such services; (3) unless and until Shipper has provided Natural with the information required in Section 4.2 hereof; (4) if Natural determines, based on the credit analysis referenced in Section 4.2(c), that Shipper does not possess sufficient financial stability to make it reasonably likely the service provided hereunder will be paid for on a timely basis; or (5) if the service requested would not comply with this Rate Schedule LPS. Nothing herein is intended to govern the scheduling and curtailment of service once a request for service has been granted pursuant to Section 4 hereof and while Agreements under this Rate Schedule are in effect. Such matters are governed by Section 3 of this Rate Schedule and the applicable General Terms and Conditions of this Tariff.

- (b) In the event that Natural agrees to construct, modify, expand or acquire facilities to perform service under this Rate Schedule LPS, Natural shall maintain a separate record of the nature and costs of such facilities.
- (c) Natural shall promptly notify Shipper if it cannot satisfy an otherwise valid request, in whole or in part. Any request shall be null and void unless it is substantially complete and complies with this Rate Schedule. In the event a request is substantially but not entirely complete, Natural shall inform Shipper in writing of the specific items needed to complete the LPS Agreement, after which Shipper shall have fifteen (15) days to provide the specified information. In the event such information is not received within fifteen (15) days, Shipper's request shall be null and void.
- (d) Natural shall tender an LPS Agreement to Shipper for execution when Shipper's request for service is accepted. Unless waived by Natural, a request for service shall be invalid if Shipper fails to execute an LPS Agreement hereunder within ten (10) days after the LPS Agreement has been tendered by Natural for execution. An executed LPS RO consistent with Section 4.3 hereof must be in place prior to commencement of service under any LPS Agreement.
- 4.2 Requests for service hereunder shall be deemed valid only after the following information is provided by Shipper via Natural's Interactive Website or in writing to Natural's Gas Transportation Department, One Allen Center, 500 Dallas Street, Houston, Texas 77002, or Telecopy Number (713) 369-9305:

FERC Docket: RP08-319-000

Original Sheet No. 225 Original Sheet No. 225: Pending

RATE SCHEDULE LPS

(a) GAS QUANTITIES

The request shall specify in Dth the maximum daily quantity (MDQ) and maximum aggregate quantity (MAQ) to be parked or loaned under any and all LPS ROs outstanding on any day under the LPS Agreement.

(b) TERM OF SERVICE

The request shall specify the date service is requested to commence and to terminate (primary term), subject to an evergreen provision;

(c) CREDIT

 $\label{eq:Acceptance} \text{Acceptance of a request is contingent upon a satisfactory credit appraisal by Natural in accordance with the General Terms and Conditions of this Tariff.}$

(d) COMPLIANCE WITH LPS TARIFF

Submission of a request for service hereunder shall be deemed agreement by Shipper that it will abide by the terms and conditions of this Rate Schedule LPS, including the applicable General Terms and Conditions, for any park or loan provided under the LPS Agreement and any related LPS RO.

(e) COMMISSION-REQUIRED FILING INFORMATION

The following information is to be provided at the time of a request for service hereunder is submitted, if available and shall be updated when any LPS RO is executed: $\frac{1}{2} \frac{1}{2} \frac{1$

- (i) Affiliation of the Shipper with Natural;
- $\,$ (ii) The identify of the Shipper, including whether it is a local distribution company, an end user, a producer, or a marketer.
- 4.3 To implement a specific park and/or loan transaction, Natural and the Shipper with an LPS Agreement in effect shall enter into an LPS RO. The LPS RO shall include the following items:

FERC Docket: RP08-319-000

Original Sheet No. 226 Original Sheet No. 226: Pending

RATE SCHEDULE LPS

(a) SERVICE TYPE/LPS AGREEMENT

 $\,$ The LPS RO must specify that it relates to service under Rate Schedule LPS and must specify the LPS Agreement to which the LPS RO relates.

(b) GAS QUANTITIES

The LPS RO shall specify in Dth the maximum aggregate quantity (MAQ RO) and the maximum daily quantity (MDQ RO) to be parked and/or loaned under the specific transaction; provided that the sum of all MAQ ROs and MDQ ROs under all pending LPS ROs cannot exceed the MAQ and MDQ under the applicable LPS Agreement. The LPS RO shall specify the minimum daily and aggregate volumes and shall set out a park and/or loan schedule containing the quantity and timing information specified in Section 7 of this Rate Schedule.

(c) POINTS

The LPS RO shall specify the point(s) at which gas is to be parked or loaned. Except for storage points, points on noncontiguous offshore facilities, and other offsystem points, any mutually agreeable point(s) on Natural's system may be utilized for service under this Rate Schedule LPS. Unless otherwise mutually agreed, the point for completion of the park or loan must be the same as the point of which the park or loan was initiated.

(d) TERM OF SERVICE

The LPS RO shall specify:

- (i) The date service is to commence; and
- (ii) The date service is to terminate.

The term may include a range of permitted commencement and termination dates for service under the LPS RO, or for any portion of such service. No termination date may extend beyond the term of the LPS Agreement.

(e) SHIPPER CONTACT PERSONNEL

 $\,$ The LPS RO shall specify the persons to be contacted by Natural in connection with the LPS RO.

FERC Docket: RP08-319-000

Original Sheet No. 227 Original Sheet No. 227: Pending

RATE SCHEDULE LPS

(f) RATE

The LPS RO shall specify the rates under Section 6.1(a) at which the park or loan service will be provided. Rates may vary by time period, volumes or other permissible discounting parameters, within the applicable maximum and minimum rates.

(g) SHIPPER ASSURANCES

Shipper shall provide Natural the assurances required by Section $1(\mbox{c})$ hereof in connection with each LPS RO.

5. TERM

- (a) The term of service hereunder shall be set forth in the LPS Agreement between Shipper and Natural. The LPS RO shall have a separately stated term or terms applicable to a particular transaction or any portion of the transaction; provided that no such term may extend beyond the term of the related LPS Agreement. Upon termination of any LPS Agreement and of any LPS RO, service by Natural to Shipper thereunder shall be terminated and automatically abandoned.
- (b) Natural may terminate any LPS Agreement if Natural is required by the FERC or some other agency or court to provide service for others utilizing the interruptible system capacity or capabilities required for service under such LPS Agreement or if Natural ceases (after receipt of any requisite regulatory authorization) to offer service of the type covered by the LPS Agreement.

6. RATE

6.1 (a) (1) For service rendered to Shipper under this Rate Schedule LPS, Shipper shall pay Natural each month the sum of the following charges: (i) an Initial Fee for each unit of gas tendered for park or taken for loan that month; (ii) a Park/Loan Balance Fee for each unit of gas which is parked or loaned under this Rate Schedule LPS that month (such charge shall be calculated on the basis of the daily ending balance for each LPS RO for each day of the month, summed over the month); and (iii) a Completion Fee for each unit of gas returned to Natural on completion (payback) of a loan or received by Shipper on completion (reversal) of a park that month.

FERC Docket: RP08-319-000

Original Sheet No. 228 Original Sheet No. 228: Pending

RATE SCHEDULE LPS

The maximum and minimum rate(s) applicable to this Rate Schedule LPS are set forth in the Currently Effective Rates section of this Tariff. On any day, the sum of the rates for the Initial Fee, the Park/Loan Balance Fee and the Completion Fee assessed for any park or loan may not exceed the sum of the maximum rate for the Initial Fee plus the maximum rate for the Park/Loan Balance Fee.

- (2) If a Shipper has submitted a nomination to reverse a park or loan and that nomination is consistent with the LPS RO but is not confirmed by Natural, the Park/Loan Balance Fee shall only be assessed on the balance which would have resulted if the nomination had been confirmed.
- (3) By mutual agreement between Natural and Shipper which is consistent with the pro forma agreement set out in this Tariff, discounts may be limited to specific volumes or time periods, etc.
- (b) Shipper shall pay any other applicable charges, penalties and fees set out in this Rate Schedule LPS or the General Terms and Conditions of this Tariff, including but not limited to ACA where applicable. Differences between confirmed nominations and allocated volumes will be assessed Balancing Service Charges under Section 12 of the General Terms and Conditions of this Tariff. Overrun charges apply if Shipper exceeds the MDQ and/or MAQ in the LPS Agreement or fails to comply with any quantity or timing parameter in an LPS RO (unless the failure results from Natural not confirming a nomination properly submitted). Authorized overrun charges set out on the rate sheet apply if the overrun reflects confirmed nominations. An unauthorized overrun charge of up to \$10 per Dth applies if the overrun is not nominated and confirmed.
- (c) Charges payable by any Shipper shall be based on the maximum rates set forth in this Tariff applicable to Rate Schedule LPS, which rates are hereby incorporated herein, unless a lower rate for the charge in subsection (a) is specified in the LPS RO.
- (d) The charges referenced herein cover only the park or loan service being provided under this Rate Schedule LPS. Shipper must separately contract for any transportation service required for Shipper to move gas to or away from the point(s) specified in the LPS $_{\rm PO}$

FERC Docket: RP08-319-000

Original Sheet No. 229 Original Sheet No. 229: Pending

RATE SCHEDULE LPS

6.2 Where a Shipper has agreed to pay a Negotiated Rate or a rate under a Negotiated Rate Formula, the rates assessed hereunder shall be governed by Section 49 of the General Terms and Conditions of this Tariff. A request for service at a Negotiated Rate or a rate under a Negotiated Rate Formula shall specify the Negotiated Rate or Negotiated Rate Formula on which the Shipper is willing to agree.

- 6.3 (a) Shipper shall reimburse Natural within five (5) days after costs have been incurred by Natural for all fees required by the FERC or any regulatory body including, but not limited to, filing, reporting, and application fees to the extent such fees are specifically related to service for that Shipper hereunder and are not generally applicable fees (such as general rate case filing fees).
- (b) If Natural constructs, acquires or modifies any facilities to perform service hereunder, then as specified in an agreement related thereto between the parties either:
- (1) Shipper shall reimburse Natural for the cost of such facilities or facility modifications as described in the General Terms and Conditions of this Tariff; or
- $\hbox{(2)} \qquad \hbox{Natural shall assess a monthly charge} \\ \hbox{reflecting such facility costs.}$
- 6.4 (a) Natural shall have the unilateral right to file with any appropriate regulatory authority and make changes effective in: (1) the rates and charges applicable under this Rate Schedule LPS, including both the level and design of such rates and charges; and/or (2) the terms and conditions of this Rate Schedule LPS. Natural agrees that Shipper may protest or contest the aforementioned filings, or may seek authorization from duly constituted regulatory authorities for such adjustment of Natural's existing FERC Gas Tariff as may be found necessary to assure that its provisions are just and reasonable.
- (b) If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises allows or permits Natural to collect, or to negotiate to collect, a higher rate for the service hereunder, the rate shall, subject to any contrary provision of the LPS Agreement or LPS RO or related discount agreement, be increased to the highest such rate. Should additional documentation be required in order for Natural to collect such

FERC Docket: RP08-319-000

Original Sheet No. 230 Original Sheet No. 230: Pending

RATE SCHEDULE LPS

highest rate, Shipper shall execute or provide such documentation within fifteen (15) days after a written request by Natural. If, at any time and from time to time, the FERC or any other governmental authority having jurisdiction in the premises requires Natural to charge a lower rate for service hereunder, the rate shall be decreased to such reduced rate.

- $6.5\,$ Natural may from time to time and at any time, upon twenty-four (24) hours' verbal or written notice, subject to any provisions on discounting in the LPS Agreement or LPS RO or related discount agreement, charge any individual Shipper for service under this Rate Schedule LPS a rate which is lower than the applicable maximum rate set forth in this Tariff; provided, however, that such rate charged may not be less than the applicable minimum rate for service under Rate Schedule LPS set forth in this Tariff. Unless otherwise agreed in the LPS Agreement or LPS RO or related discount agreement, Natural may at any time further change such rate (subject to any restrictions as to maximum or minimum rates set out in this Tariff, the LPS Agreement, the LPS RO, and/or related discount agreement) upon twenty-four (24) hours' verbal notice to Shipper, which notice shall be confirmed in writing. Such notification shall specifically state the effective date of such rate change and the quantity of gas so affected. Natural shall file with FERC any and all reports as required by FERC's Regulations with respect to the institution or discontinuance of any discount.
- 6.6 All revenues and gas in kind collected by Natural as a result of providing service under this Rate Schedule LPS shall be retained by Natural unless Natural has otherwise explicitly agreed on a different disposition of such amounts. Where crediting or refund mechanisms apply under other provision(s) of this Tariff or pursuant to effective FERC orders or settlement, such mechanisms shall supersede this Section to the extent necessary to carry out such provision(s).

7. OUANTITY

Each LPS RO shall specify in Dth the MAQ RO and a daily schedule of the quantities (including the MDQ RO) to be parked and/or loaned under the specific transaction. The daily schedule of returned volumes by the Shipper or Natural shall also be specified in the LPS RO. The quantities may be specified as a range of volumes

FERC Docket: RP08-319-000

Original Sheet No. 231 Original Sheet No. 231: Pending

RATE SCHEDULE LPS

(maximum and minimum aggregate and daily quantities and the related time periods) to be parked and/or loaned and returned, and the schedule may include the flexibility to do either a park or a loan within specified volume and time limits. The schedule may provide for flexibility in total volumes and in the daily volumes parked and/or loaned, in the timing of the park or the loan (or any portion thereof), in the duration of the park and/or loan (or any potion thereof) and/or in the timing of the completion of the park or loan (or any portion thereof) by the return of gas to the Shipper or to Natural, and shall specify the limits of the flexibility allowed. Subject to the flexibility specified in the LPS RO, the MDQ RO shall be the maximum quantity Natural is obligated, on an interruptible basis, to receive from or deliver to Shipper under the RO on the specified day. The MAQ RO shall be the maximum aggregate quantity Natural is obligated to hold or loan for the account of Shipper under the RO. The minimum aggregate and daily volumes to be parked and/or loaned and returned on an interruptible basis shall also be specified in the LPS RO schedule. The sum total of a Shipper's MAQ ROs and MDQ ROs shall not exceed the MAQ and MDQ specified in the LPS Agreement. If a Shipper exceeds the MDQ or MAQ in the LPS Agreement or fails to comply with any quantity or timing parameter in the applicable LPS RO (unless such failure is due to Natural not confirming a nomination properly submitted), it shall be subject to overrun charges consistent with Section 6.1(b) of this Rate Schedule.

8. NOMINATIONS

It shall be solely Shipper's responsibility to provide Natural with daily nominations of the quantity of gas to be received or delivered at the Receipt or Delivery point(s) within the parameters specified in the applicable LPS RO. Nominations for any day or for any nomination cycle must be consistent with the LPS RO. It shall also be Shipper's responsibility to cause gas to be delivered to Natural and to cause gas to be received from Natural in accordance with the LPS RO. Nominations shall be subject to confirmation and scheduling in accordance with the General Terms and Conditions of this Tariff. If a nomination to reverse a park or loan is consistent with the LPS RO but cannot be confirmed by Natural, the Shipper must continue to nominate on subsequent days until Natural can confirm the nomination, unless the parties agree on a revised LPS RO. Differences between confirmed nominations and allocated volumes are subject to Balancing Service Charges under Section 12 of the General Terms and Conditions of this Tariff. Overrun charges apply as specified in Section 6.1(b) of this Rate Schedule.

FERC Docket: RP08-319-000

Original Sheet No. 232 Original Sheet No. 232: Pending

RATE SCHEDULE LPS

9. MANDATORY BALANCING

- 9.1 Mandatory Balancing under Section 9.3 hereof shall apply in the following instances: (a) at the end of the term specified in any applicable LPS Agreement or LPS RO; (b) where the Shipper fails to comply with the requirements of Natural's notice under Section 3 (b) hereof; or (c) where Shipper fails to comply with the schedule of activities set forth in the applicable LPS RO, the deviation has not been agreed to and confirmed by Natural and Natural has provided the notice set out in section 9.2 of this Rate Schedule.
- $9.2\,$ In the event that a Shipper fails to comply with the schedule of activities set forth in the applicable LPS RO, Natural shall notify Shipper and the LPS RO shall be subject to termination in accordance with the terms of such notice.
- 9.3 (a) In the event that Shipper still has gas parked at the time any of the conditions specified in Section 9.1 hereof apply, the remaining balance shall be forfeited to Natural, free and clear of any adverse claims.
- Conversely, if the Shipper has not redelivered gas (b) which was loaned by Natural at the time any of the conditions $% \left(1\right) =\left(1\right) +\left(1$ specified in Section 9.1 hereof apply, the Shipper must purchase the unreturned balance at 150% of the average Weekly Index Prices (WIPs) as determined in Section 13.3 of the General Terms and Conditions of this Tariff during the term of the loan. In the event that the information used to determine the WIPs for the term of the loan is not available, then the Shipper must purchase the unreturned balance at 150% of the Average Monthly Index Price (AMIP) as determined in Section 13.3 of the General Terms and Conditions of this Tariff. This charge is in addition to any overrun charges applicable under other provisions of this Rate Schedule. If Critical Time or Operational Flow Orders are in effect on the date Shipper is required to comply with the notice, Shipper shall be subject to the highest charges set forth in Sections 12.3 and 12.4 of the General Terms and Conditions of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 233 Original Sheet No. 233: Pending

RATE SCHEDULE LPS

9.4 In circumstances where Shipper is unable to eliminate its LPS RO balance because Natural is unable to accept the LPS RO nomination, Shipper shall take any action to reduce the balance which Natural can accommodate and Shipper shall be granted additional time to eliminate its LPS RO balance corresponding to the time Natural was unable to accept the LPS RO nomination. Shipper shall not incur any penalty or daily fees as to that portion of the balance resulting from Natural's inability to accept the LPS RO nomination. Such extension shall only apply in instances where a Shipper's inability to eliminate the LPS RO balance is attributable to Natural's inability to accept and confirm LPS RO nominations.

9.5 Shipper remains responsible to nominate the appropriate level of transportation to meet Natural's notification requirements under Section 3 above, or to meet the requirements of the LPS RO.

10. GENERAL TERMS AND CONDITIONS

The provisions of the General Terms and Conditions of this Tariff, as such provisions may be amended from time to time, are hereby incorporated by reference and made a part of this Rate Schedule LPS and shall apply to service rendered hereunder as though stated herein.

FERC Docket: RP08-319-000

Sheet Nos. 234 - 274 Sheet Nos. 234 - 274 : Pending

Sheet Nos. 234 through 274 have been reserved for future use.

FERC Docket: RP08-319-000

Original Sheet No. 275 Original Sheet No. 275: Pending

GENERAL TERMS AND CONDITIONS

1. DEFINITIONS

The following terms, wherever used in any Rate Schedule or in these General Terms and Conditions, shall be defined as follows:

1.1 AGREEMENT

The term "Agreement" shall mean, as applicable, a DSS Agreement, FTS Agreement, FTS Agreement, FTS-G Agreement, ITS Agreement, IBS Agreement, NSS Agreement, PALS Agreement, FRSS Agreement, BESS Agreement or LPS Agreement, as defined in the respective Rate Schedules, and an Agreement under any X Rate Schedule in Natural's Volume No. 2 Tariff to which these General Terms and Conditions may apply.

1.2 BUSINESS DAY

"Business Day" is defined as Monday through Friday, excluding Federal Banking Holidays for transactions in the U.S., and similar holidays for transactions occurring in Canada and Mexico.

1.3 DAY

The gas Day shall be from 9:00 a.m. to 9:00 a.m. Central Clock Time.

1.4 DELIVERY POINT

The term "Delivery Point" shall mean any point at which Natural delivers to or for the account of Shipper gas which has been transported or stored by Natural under any Rate Schedule in this Tariff, as specified in an Agreement or as applicable to service under such Agreement by operation of this Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 276 Original Sheet No. 276: Pending

GENERAL TERMS AND CONDITIONS

1.5 DTH

The term "Dth" shall mean one million (1,000,000) Btus and is equivalent to one (1) MMBtu.

1.6 ELECTRONIC DATA INTERCHANGE ("EDI")

The term "EDI" shall mean Electronic Data Interchange.

1.7 EOUIVALENT VOLUMES

The term "Equivalent Volumes" shall mean the quantity of gas measured in Dth received by Natural for the account of Shipper at the Receipt Point(s) during any given period of time: (a) reduced by a percentage for Fuel Gas consumed and a percentage for Gas Lost and Unaccounted For in Natural's system; (b) reduced by any plant volume reductions assessed as a result of Shipper's or its supplier's election to process the gas; and (c) adjusted for any variations in Btu content. It is the intent of the parties that the volumes of gas delivered at the Delivery Point(s) shall be equivalent in Dth to the volumes of gas received at the Receipt Point(s) under an Agreement, as adjusted for items (a), (b) and (c) above.

1.8 FUEL GAS

The term "Fuel Gas" shall mean, with respect to any transportation service, the quantities of gas consumed on Natural's system as a result of the transportation service provided and which includes, but is not limited to, Fuel Gas consumed in Natural's compressor stations. The term "Fuel Gas" shall mean, with respect to any storage service, the quantities of gas consumed in connection with the injection and withdrawal of gas.

1.9 GAS LOST AND UNACCOUNTED FOR

The term "Gas Lost and Unaccounted For" shall mean the difference between the sum of all input quantities of gas to Natural's system and the sum of all output quantities of gas (including Fuel Gas) from Natural's system, which difference, after any appropriate adjustments, shall include but shall not be limited to gas vented and gas lost as a result of an event of Force Majeure, the ownership of which cannot be reasonably identified.

FERC Docket: RP08-319-000

Original Sheet No. 277 Original Sheet No. 277: Pending

GENERAL TERMS AND CONDITIONS

1.10 HEATING VALUE

The term "heating value" shall mean the number of Btus per cubic feet of gas at the base condition of 14.73 psia 60 degrees Fahrenheit dry. The Btu value will be determined utilizing the complete actual composition of the gas according to the methods in GPA Standard 2172-96, titled "Calculation of Gross Heating Value, Relative Density and Compressibility Factor for Natural Gas Mixtures from Compositional Analysis," and corrected to the base conditions. For reporting purposes, Btu conversion factors will be reported to not less than three (3) decimal places and Pressure Base conversion factors will be reported to not less than six (6) decimal places. For calculation purposes, not less than six (6) decimal places will be used for both conversion factors.

1.11 INTERACTIVE WEBSITE

The term "Interactive Website" shall mean the interactive internet web site maintained by Natural for communication regarding its transportation and storage service in accordance with applicable Commission Regulations and GISB standards, as more fully described in Section 17 of these General Terms and Conditions.

1.12 MONTH

A period of one calendar month commencing on the first day of such month; provided, however, that the term "month" may be established as some other equivalent period by mutual agreement between Shipper and Natural or between Natural and the operator of any Receipt Point or Delivery Point, as applicable.

1.13 NATURAL GAS

The term "natural gas" shall mean any mixture of hydrocarbons or of hydrocarbons and non-combustible gases, in a gaseous state, consisting essentially of methane.

FERC Docket: RP08-319-000

Original Sheet No. 278 Original Sheet No. 278: Pending

GENERAL TERMS AND CONDITIONS

1.14 NEGOTIATED RATE

The term "Negotiated Rate" shall mean a rate provision under which Natural and Shipper have agreed on the amount to be charged for the service under Rate Schedule FTS, FFTS, FTS-G, ITS, IBS, BESS, DSS, NSS, FRSS, PALS or LPS which results in a rate where, for all or a portion of the contract term, one or more of the individual components of such rate exceeds or may exceed the applicable maximum rate or is less than or may be less than the applicable minimum rate. Any Agreement entered into after the effective date of this subsection which provides for a rate under Rate Schedule FTS, FFTS-G, ITS, IBS, BESS, DSS, NSS, FRSS, PALS or LPS other than the applicable maximum rate shall contain a provision setting out the mutual agreement of the parties, consistent with Commission policy, as to whether the pricing terms represent a discounted rate or a Negotiated Rate.

1.15 NEGOTIATED RATE FORMULA

The term "Negotiated Rate Formula" shall mean a rate formula provision which Natural and Shipper have agreed will be applied to service under Rate Schedule FTS, FFTS, FTS-G, ITS, IBS, BESS, DSS, NSS, FRSS, PALS or LPS which results in a rate where, for all or a portion of the contract term, one or more of the individual components of such rate exceeds or may exceed the applicable maximum rate or is less than or may be less than the applicable minimum rate. Any Agreement entered into after the effective date of this subsection which provides for a rate under Rate Schedule FTS, FFTS, FTS-G, ITS, IBS, BESS, DSS, NSS, FRSS, PALS or LPS other than the applicable maximum rate shall contain a provision setting out the mutual agreement of the parties, consistent with Commission policy, as to whether the pricing terms represent a discounted rate or a rate pursuant to a Negotiated Rate Formula.

1.16 OPERATIONAL BALANCING AGREEMENT ("OBA")

 $\,$ An OBA is a contract between two parties which specifies the procedures to manage operating variances at an interconnect.

1.17 RECEIPT POINT

The term "Receipt Point" shall mean any point at which gas is tendered by or for the account of Shipper to Natural for transportation or storage, as specified in an Agreement or as applicable to service under such Agreement by operation of this Tariff.

FERC Docket: RP09-309-000

First Revised Sheet No. 279 First Revised Sheet No. 279

Superseding: Original Sheet No. 279

GENERAL TERMS AND CONDITIONS

1.18 RECOURSE RATE

The term "Recourse Rate" shall mean the applicable maximum rate(s) which would apply to the service but for the rate flexibility allowed under Section 49 hereof.

1.19 SHIPPER

The term "Shipper" shall mean a Shipper as defined in any of the Rate Schedules governed by these General Terms and Conditions. In addition, in a given context, Shipper may refer to an entity which is seeking to become a Shipper. In addition, "Shipper" may refer to Storage Expansion 2003, Sayre Storage Project, North Lansing NSS Project, and Herscher Galesville NSS Project Shippers.

1.20 STANDARD REPORTING BASIS

Standardize the reporting basis for Btu as 14.73 psia and 60 degrees F (101.325 kPa and 15 degrees C, and dry). Standardize the reporting basis for gigacalorie as 1.035646 Kg/cm squared and 15.6 degrees C, and dry.

Standardize the reporting basis for gas volumes as cubic foot at standard conditions of $14.73~\rm psia$, $60~\rm degrees~F$, and dry. For gas volumes reported in cubic meters, the standard conditions are $101.325~\rm kPa$, $15~\rm degrees~C$, and dry.

1.21 YEAR

A period of three hundred sixty-five (365) consecutive days; provided that any such year which contains the date of February 29 shall consist of three hundred sixty-six (366) consecutive days; and provided further that the term "year" may be established as some other equivalent period by mutual agreement between Shipper and Natural or between Natural and the operator of any Receipt Point or Delivery Point, as applicable.

1.22 STORAGE EXPANSION 2003

The term "Storage Expansion 2003" shall mean the expansion in Natural's Storage capacity resulting from the construction of facilities authorized by the Commission pursuant to orders issued in Docket No. CP02-391-000, 101 FERC 61,125 and 101 FERC 61,361.

FERC Docket: RP08-319-000

Original Sheet No. 280 Original Sheet No. 280: Pending

GENERAL TERMS AND CONDITIONS

1.23 STORAGE EXPANSION 2003 AGREEMENT

The term "Storage Expansion 2003 Agreement" shall mean a NSS Agreement entered into between Natural and a Shipper for capacity resulting from Storage Expansion 2003.

1.24 STORAGE EXPANSION 2003 SHIPPER

The term "Storage Expansion 2003 Shipper" shall mean a Shipper which has entered into a Storage Expansion 2003 Agreement. A Shipper's status as a Storage Expansion 2003 Shipper shall be determined on a contract-by-contract basis. Therefore, a given Shipper may be a Storage Expansion 2003 Shipper with respect to a specific contract and not a Storage Expansion 2003 Shipper with respect to other contracts.

1.25 STORAGE EXPANSION 2003 RATE

Under Rate Schedule NSS, separate rates are stated for Storage Expansion 2003 service. The rates for Storage Expansion 2003 shall apply to Storage Expansion 2003 Agreements.

1.26 SAYRE STORAGE PROJECT

The term "Sayre Storage Project" shall mean the modification to Natural's Storage capacity resulting from the construction of facilities authorized by the Commission pursuant to an Order issued on March 25, 2005, in Docket No. CP05-7-000, 110 FERC Paragraph 61,344.

1.27 SAYRE STORAGE PROJECT AGREEMENT

The term "Sayre Storage Project Agreement" shall mean an Agreement entered into between Natural and a Shipper for capacity resulting from Sayre Storage Project. Initially, all such Agreements are entered into under Rate Schedule NSS.

1.28 SAYRE STORAGE PROJECT SHIPPER

The term "Sayre Storage Project Shipper" shall mean a Shipper which has entered into a Sayre Storage Project Agreement. A Shipper's status as a Sayre Storage Project Shipper shall be determined on a contract-by-contract basis. Therefore, a given Shipper may be a Sayre Storage Project Shipper with respect to a specific contract and not a Sayre Storage Project Shipper with respect to other contracts.

FERC Docket: RP08-319-000

Original Sheet No. 281 Original Sheet No. 281: Pending

GENERAL TERMS AND CONDITIONS

1.29 Sayre Storage Project Rate

Under Rate Schedule NSS, separate rates are stated for Sayre Storage Project service. The rates for Sayre Storage Project shall apply to Sayre Storage Project Agreements.

1.30 NORTH LANSING NSS PROJECT

The term "North Lansing NSS Project" shall mean the modification to Natural's Storage capacity resulting from the construction of facilities authorized by the Commission pursuant to an Order issued on January 23, 2006, in Docket No. CP05-405-000, 114 FERC Paragraph 61,061.

1.31 NORTH LANSING NSS PROJECT AGREEMENT

The term "North Lansing NSS Project Agreement" shall mean an Agreement entered into between Natural and a Shipper for capacity resulting from North Lansing NSS Project. Initially, all such Agreements are entered into under Rate Schedule NSS.

1.32 NORTH LANSING NSS PROJECT SHIPPER

The term "North Lansing NSS Project Shipper" shall mean a Shipper which has entered into a North Lansing NSS Storage Project Agreement. A Shipper's status as a North Lansing NSS Project Shipper shall be determined on a contract-by-contract basis. Therefore, a given Shipper may be a North Lansing NSS Project Shipper with respect to a specific contract and not a North Lansing NSS Project Shipper with respect to other contracts.

1.33 NORTH LANSING NSS PROJECT RATE

Under Rate Schedule NSS, separate rates are stated for North Lansing NSS Project service. The rates for North Lansing NSS Project shall apply to North Lansing NSS Project Agreements.

FERC Docket: RP09-309-000

Original Sheet No. 281A Original Sheet No. 281A

GENERAL TERMS AND CONDITIONS

1.34 HERSCHER GALESVILLE NSS PROJECT

The term "Herscher Galesville NSS Project" shall mean the modification to Natural's Storage capacity resulting from the construction of facilities authorized by the Commission pursuant to an Order issued on August 11, 2008, in Docket No. CP08-32-000, 124 FERC Paragraph 61,154.

1.35 HERSCHER GALESVILLE NSS PROJECT AGREEMENT

The term "Herscher Galesville NSS Project Agreement" shall mean an Agreement entered into between Natural and a Shipper for capacity resulting from Herscher Galesville NSS Project. Initially, all such Agreements are entered into under Rate Schedule NSS.

1.36 HERSCHER GALESVILLE NSS PROJECT SHIPPER

The term "Herscher Galesville NSS Project Shipper" shall mean a Shipper which has entered into a Herscher Galesville NSS Storage Project Agreement. A Shipper's status as a Herscher Galesville NSS Project Shipper shall be determined on a contract-by-contract basis. Therefore, a given Shipper may be a Herscher Galesville NSS Project Shipper with respect to a specific contract and not a Herscher Galesville NSS Project Shipper with respect to other contracts.

1.37 HERSCHER GALESVILLE NSS PROJECT RATE

Under Rate Schedule NSS, separate rates are stated for Herscher Galesville NSS Project service. The rates for Herscher Galesville NSS Project shall apply to Herscher Galesville NSS Project Agreements.

FERC Docket: RP08-319-000

Sheet No. 282 Sheet No. 282 : Pending

GENERAL TERMS AND CONDITIONS

2. RESERVED FOR FUTURE USE

FERC Docket: RP08-319-000

Sheet No. 283 Sheet No. 283 : Pending

GENERAL TERMS AND CONDITIONS

3. RESERVED FOR FUTURE USE

Effective Date: 04/18/2008 Status: Effective FERC Docket: RP08-319-000

Original Sheet No. 284 Original Sheet No. 284 : Pending

GENERAL TERMS AND CONDITIONS

4. RESERVED FOR FUTURE USE

FERC Docket: RP08-319-000

Original Sheet No. 285 Original Sheet No. 285: Pending

GENERAL TERMS AND CONDITIONS

5. PRIORITY OF SERVICE

5.1 ALLOCATION OF FIRM CAPACITY

This Section 5.1 governs the posting, the allocating among services and among entities requesting firm services, and the awarding of firm transmission and storage capacity on Natural's system.

- (a) If at any time Natural has transmission capacity and/or storage capacity which would support additional firm services, Natural shall make such capacity available to perform firm open access services. The allocation of such capacity among firm services shall be determined by Natural, applying objective criteria of the relative engineering suitability and economic value of the various services.
- (b) In assigning priority to otherwise valid bids or requests for any particular firm service, Natural shall allocate firm capacity to valid bids or requests which generate the highest economic value for Natural. The criteria for determining the highest economic value are discussed in Section 5.1(c). In applying such criteria where a Negotiated Rate or Negotiated Rate Formula is involved, the value assigned to a request which includes a Negotiated Rate or Negotiated Rate Formula shall be limited by the Recourse Rate as provided in Section 49 of these General Terms and Conditions. The procedures to award firm capacity are described in Section 5.1(c). Certain posting requirements for firm capacity are set out in Section 5.1(e).

(c) CAPACITY AWARD PROCEDURES

This subsection sets out the procedures to be followed by Natural in awarding all firm forward haul transmission segment capacity and firm storage capacity on its system; provided, however, that these procedures do not apply to the initial allocation of such firm capacity created by the construction of new facilities (including both extensions and expansions of Natural's system).

(1) INITIAL OPEN SEASON Natural shall conduct an Initial Open Season for all Posted Capacity. As used in this Section 5, the term "Posted Capacity" shall mean all firm forward-haul transmission segment capacity and firm storage capacity currently available or becoming available in the future which Natural is obligated to post under Section 5.1(e)(1).

FERC Docket: RP08-319-000

Original Sheet No. 286 Original Sheet No. 286: Pending

GENERAL TERMS AND CONDITIONS

- (i) An Initial Open Season shall be conducted at least once each month for Posted Capacity (as defined in the prior paragraph) which was not previously included in an Initial Open Season. Posted Capacity shall be included in the next Initial Open Season after such firm capacity becomes Posted Capacity.
- (ii) The first Initial Open Season to be held following implementation of this Initial Open Season procedure will cover all Posted Capacity as of that date.
- (iii) In any Initial Open Season, the minimum posting and bidding period applicable to firm capacity available for less than five (5) months is from 9 a.m. to 2 p.m. Central Clock Time on a business day. In any Initial Open Season, the minimum posting and bidding period for firm capacity available for five (5) months or longer is four (4) business days, with the minimum posting periods each day being from 9 a.m. to 2 p.m. Central Clock Time.
- $\mbox{(iv)}$ The posting for an Initial Open Season will include the following items:
- (A) For transmission capacity, the location by pipeline segment of the firm capacity to be covered by the Initial Open Season and any restriction on any segment or on any Receipt or Delivery Point.
- (B) The bidding procedure to be used, including an explanation of how bids are to be submitted, a bid form, instructions for satisfying the request criteria under the applicable rate schedule, and the complete bid evaluation criteria consistent with Section $5.1(\mathrm{d})$.
- (C) The timetable for posting and bidding, which timetable must be consistent with Section $5.1(c)\,(1)\,(iii)$. The timetable will specify any right Natural may have to extend the Initial Open Season but in no event can the Initial Open Season be extended if a valid bid has been submitted.
- (D) The type, duration and amount of firm capacity to be covered by the Initial Open Season, and the date on which such firm capacity will be available.

FERC Docket: RP08-319-000

Original Sheet No. 287 Original Sheet No. 287: Pending

GENERAL TERMS AND CONDITIONS

- $\mbox{\ensuremath{(E)}}$ Contingencies, if any, which will be accepted in open season bids.
- $\mbox{\ensuremath{(F)}}$ Any other bid requirements, conditions, criteria, restrictions or parameters.
- (v) In the Initial Open Season, all bids must be submitted in the basic rate design then in effect on Natural's system and are constrained by the applicable maximum and minimum rates. No bid with rates in the form of a Negotiated Rate or Negotiated Rate Formula will be valid in an Initial Open Season.
- (vi) Natural shall establish a reserve price or reserve price matrix in every Initial Open Season, consistent with Section 5.1(d)(5).
- (vii) Unless otherwise specified in the posting, a bidder may bid on all or any portion of the term for which the firm capacity is available and on all or any portion of the capacity available. Natural may only impose minimum volume or term requirements on bids for operational reasons, such as maintaining pressure or sustaining the minimum level of prudent facility operations on any affected portion(s) of its system.
- (viii) Natural shall have the right to terminate an Initial Open Season through a termination posting if there is a material error in the Initial Open Season posting, which error shall be explained in the termination posting. A bidder may withdraw a bid for a material error by notifying Natural by the deadline for bids that the bid contains a material error, which notification shall explain the material error.
- All bids received during the open season period remain binding on the bidder through the end of the open season unless withdrawn by bidder. A bidder may withdraw its bid at any time during the open season, utilizing the same medium through which a bid can be submitted. bidder which has withdrawn a bid may submit a bid with a higher value during the open season, but neither that bidder nor any of its affiliates may submit a bid with a lower value than the withdrawn bid, except if the withdrawn bid was withdrawn due to a material error, during that open season [value is to be determined applying the criteria in Section 5.1(d), consistent with the posting for that open season]. If an open season is extended, a bidder is free to submit a new bid without restriction in the extended auction even if that bidder withdrew its bid in the original open season. If a bidder withdraws its bid in an extended open season, the same rules as in an original open season apply to that bidder. At the end of any open season, all bids either withdrawn or not accepted shall be deemed null and void.

FERC Docket: RP08-319-000

Original Sheet No. 288 Original Sheet No. 288: Pending

GENERAL TERMS AND CONDITIONS

(x) In an Initial Open Season, Natural may post a prearranged transaction under the basic rate design then in effect on Natural's system (no Negotiated Rate or Negotiated Rate Formula) and shall specify a right to match by the prearranged Shipper.

(xi) If contractual rollover rights are to be available to the winning bidder, or are agreed upon with a prearranged Shipper, the terms of such contractual rollover shall be included in the open season posting and such rollover rights shall be available to all bidders similarly situated.

(xii) Once firm capacity has been subject to an Initial Open Season, Natural is not required to hold any further Initial Open Season for that capacity until it again becomes Posted Capacity; provided, however, that such capacity shall be subject to Shipper-Initiated Open Seasons under Section 5.1(c)(5).

- $\,$ (2) EARLY OPEN SEASON This section applies to firm forward haul transmission segment capacity or firm storage capacity which is not Posted Capacity.
- (i) Natural may conduct an Early Open Season or any Shipper (including the existing capacity holder) may cause Natural to initiate such an Early Open Season by following the Shipper-Initiated Open Season procedure.
- (ii) An Early Open Season shall not commence until the firm capacity to be covered by that Early Open Season has been posted as required pursuant to Section $5.1(e)\,(8)$. Any Early Open Season shall be held at the same time as an Initial Open Season.
- (iii) Except for the timing of the open season, an Early Open Season is subject to the same procedures, limitations, posting requirements, and minimum posting and bidding periods as an Initial Open Season, including the posting of prearranged transactions.
- (iv) To the extent that the existing Shipper holding firm capacity included in an Early Open Season has any right of first refusal, tariff rollover right, contractual rollover right or any other term extension right in Natural's Tariff or the contract (Extension Rights), the award of capacity hereunder shall be conditional in that it is subordinate to and will not impair any Extension Rights of the existing firm capacity holder.

FERC Docket: RP08-319-000

Original Sheet No. 289 Original Sheet No. 289: Pending

GENERAL TERMS AND CONDITIONS

(v) If firm capacity which would satisfy all or part of the conditional capacity award in an Early Open Season is the subject of a subsequent open season or right of first refusal process, the winning bid in the Early Open Season shall be deemed to be a bid in any such open season or right of first refusal process until the winning Early Open Season bidder has fully contracted for the firm capacity it was conditionally awarded in the Early Open Season or until three (3) years have elapsed from the date of the Early Open Season award, (unless Natural and a bidder agree in writing to a shorter period prior to commencement of the Early Open Season) whichever occurs first. Thereafter, the conditional capacity award in the Early Open Season shall have no further effect.

- (vi) The Early Open Season winning bid deemed submitted as a bid in the subsequent open season or right of first refusal process (or a higher value bid submitted by the winning bidder in the Early Open Season) shall establish the reserve price in that open season or process. If that bid is otherwise a valid bid in such subsequent open season or right of first refusal process and is the highest value bid therein under the criteria applicable to that subsequent open season or right of first refusal process, Natural shall be obligated to award such firm capacity to the Early Open Season bidder, subject to any right of the existing capacity holder to match. The winning Early Open Season bidder shall have the right to match the highest bid submitted by any entity other than the existing capacity holder exercising an Extension Right.
- (vii) Nothing herein will prevent the winning bidder in the Early Open Season from submitting an additional bid of higher value [as determined under Section 5.1(d)] in any open season or right of first refusal process in which it is deemed to have submitted a bid under these procedures.
- (viii) Firm capacity offered but not awarded in an Early Open Season must go through an Initial Open Season at the time specified in Section $5.1(c)\,(1)$.
- (3) REQUEST PROCEDURE If Posted Capacity is not awarded in the Initial Open Season, Natural may award that capacity through either the Request Procedure in this Section 5.1(c)(3) or the Additional Open Season Procedure in Section 5.1(c)(4).

FERC Docket: RP08-319-000

Original Sheet No. 290 Original Sheet No. 290: Pending

GENERAL TERMS AND CONDITIONS

(i) Once Posted Capacity has been subject to the Initial Open Season and has not been awarded, Natural may award such firm capacity in response to requests for firm service in writing or on DART which conform to the request requirements in the applicable rate schedule. Requests may be unsolicited or pre-negotiated, but no communication will be considered a request hereunder unless it constitutes a valid request for service consistent with the applicable rate schedule and has been submitted in writing or via Natural's Interactive Website.

- (ii) Requests will be binding on the entity requesting firm capacity for two (2) business days unless the request states a shorter time period or until the request is accepted or rejected by Natural, whichever occurs first. In its request, the customer may specify that the request is binding for a stated period which is less than two (2) business days. Unless granted by Natural, all requests submitted are void effective the earlier of the time when Natural rejects the request or two (2) business days after the request is submitted. Natural cannot grant any request which has become void, but a request becoming void hereunder is without prejudice to any future request by that Shipper or any other Shipper.
- (iii) The rate form under a request may be either the basic rate design then in effect on Natural's system or a Negotiated Rate or Negotiated Rate Formula rate form.
- (iv) Natural is not obligated to award firm capacity based on a request at less than the applicable maximum rate, but any capacity award must be consistent with Section 5.1(d) if there are competing valid requests pending. For purposes of applying the evaluation criteria in Section 5.1(d), only pending valid requests which have not become void under (ii) are considered to be competing.
- (v) Requests shall not be accepted or valid as to any firm capacity which is subject to an Initial Open Season, an Early Open Season, an Additional Open Season or a Shipper-Initiated Open Season during any period between the posting of that open season and the award of capacity (or the decision not to award capacity) under that open season.

(4) ADDITIONAL OPEN SEASON PROCEDURE

(i) If Posted Capacity has not been awarded in the Initial Open Season, Natural may at any time elect to hold an Additional Open Season.

FERC Docket: RP08-319-000

Original Sheet No. 291 Original Sheet No. 291: Pending

GENERAL TERMS AND CONDITIONS

(ii) The posting requirements and minimum posting and bidding periods for an Additional Open Season shall be the same as for an Initial Open Season under Section 5.1(c)(1), except that Natural will post any additional items necessary to comply with this Section 5.1(c)(4).

(iii) In the Additional Open Season, Natural may make firm capacity available only for a specified term or range of terms, which limitation must be included in the posting.

(iv) In the Additional Open Season, bids may be on a Negotiated Rate or Negotiated Rate Formula basis to the extent specified in the posting. Natural may limit the form of Negotiated Rate or Negotiated Rate Formula submitted, in which case that limitation must be reflected in the posting. Bids may always be submitted in the basic rate design then in effect on Natural's system at the applicable maximum rate.

(v) Natural may, but is not required to, establish a reserve price or reserve price matrix in an Additional Open Season, consistent with Section $5.1(d)\ (5)$.

(vi) In an Additional Open Season, Natural may post a prearranged transaction for bid, including a prearranged Negotiated Rate or Negotiated Rate Formula transaction, and shall include a right on the part of the prearranged Shipper to match as specified in the posting.

(5) POINT CHANGES

(i) Natural and a Shipper under any firm service may add or delete primary Delivery or Receipt Points from time to time by mutual agreement. Subject to the availability of firm capacity at the requested point, Natural shall agree to any such change in primary Delivery or Receipt Point to the extent such new point is within the transportation path of the existing primary points. At other points, Natural shall agree to a change to the extent that firm transmission and point capacity is available after taking into account existing capacity commitments under other firm Agreements for the date requested if such capacity is not subject to or affected by any other pending valid request or pending open season. A point change is not subject to or affected by any pending request or open season if the available capacity equals or exceeds the amount necessary to satisfy both the pending request and/or open season and the point change, or if capacity could not be granted to any third party but for the point

FERC Docket: RP08-319-000

Original Sheet No. 292 Original Sheet No. 292: Pending

GENERAL TERMS AND CONDITIONS

change. The rate applicable to the revised point(s) shall be the applicable maximum rate, unless the existing Agreement or related discount contract provides otherwise, or unless Natural and Shipper negotiate a different rate at the time of the point change. Unless otherwise specifically agreed, if the point change results in a shortened path which would reduce the applicable rate, the reservation rate applicable to the longer path shall apply for the remaining term of the Agreement, but the commodity rate applicable to the shortened path shall apply so long as the shortened path is in effect under the Agreement; provided, however, the Fuel Gas and Gas Lost and Unaccounted For for the path utilized shall apply. Natural and Shipper may at any time agree on or modify discounts at secondary points. Notwithstanding the foregoing, a Shipper under Rate Schedule FTS-G may not change primary Delivery Points, except to a different location within Shipper's traditional market area, unless otherwise agreed by Natural.

(ii) From time to time, points are retired or eliminated for various reasons. In such a situation, Natural shall advise any affected Shipper as early as possible of the point retirement or elimination and the reason for the retirement or elimination and shall specify the alternate point(s) where adequate capacity is available in the reasonable geographic vicinity of (and within the same rate zone as) the point being eliminated. Capacity is not available if subject to any open season, pending request or open season award; provided, however, that point changes hereunder shall have priority over any open season which commences or any request for capacity which is submitted after Natural advises Shippers of the point retirement or elimination. An affected Shipper may move to any point initially designated by Natural without any change in rate by notification to Natural. If capacity at any point is not adequate to satisfy all Shippers hereunder, allocation shall be pro rata based on the respective MDQs the various Shippers are seeking to move to that point. The process shall be reiterated at other designated points until each Shipper's MDQ affected by the point retirement or elimination is satisfied and Natural shall be obligated to designate additional points within the same rate zone which are in reasonable proximity and are operationally reasonably equivalent to the point being retired or eliminated as necessary to satisfy point changes desired by all affected Shippers. The rate at these additional points designated by Natural shall be the rate specified in the contract for services at that point or, if the contract is silent, the rate assessed to similarly situated customers for similar service involving that point. If a Shipper elects to change to a point which is not a point designated by Natural, then that Shipper must submit a point change request pursuant to Section 5.1(c)(5)(i) of these General Terms and Conditions; provided, however, that a Shipper shall not be required to submit a point

FERC Docket: RP08-319-000

Original Sheet No. 293 Original Sheet No. 293: Pending

GENERAL TERMS AND CONDITIONS

change request if it desires a change to any other point within the same rate zone which is in reasonable proximity and is operationally reasonably equivalent to the point being retired or eliminated. Nothing herein shall be read to authorize, or to relieve Natural from any obligation to seek and obtain Commission authorization for, abandonment and termination of operations and/or facilities at any receipt or delivery point. Nothing herein shall be read to override or to modify in any way: (A) any obligation of Natural under the Natural Gas Act to treat similarly situated customers the same with respect to rates and discounting; (B) any tariff provision relating to rates and discounting which results from Commission action on Natural's Order No. 637 compliance plan in Docket No. RP00-409; (C) any other provision of Natural's Tariff governing or relating to rates and discounting; or (D) any discount to which Shipper is contractually entitled.

(6) SHIPPER-INITIATED OPEN SEASON If a potential customer submits to Natural a binding offer to accept Posted Capacity at a rate equal to or greater than fifty percent (50%) of the applicable maximum rate for the winter months (November-March) or twenty five percent (25%) of the applicable maximum rate for the summer months (April-October), or a combination which meets the net present value over the term requested of the rate corresponding to those percentages, Natural shall be obligated to commence an open season in the same format and under the same procedures as an Initial Open Season under Section 5.1(c)(1) within five (5) Business Days after receipt by Natural of the request. A Shipper may also cause an Early Open Season to be initiated for firm capacity which is not Posted Capacity using this same procedure. A Shipper-Initiated Early Open Season shall coincide with the next Initial Open Season which commences more than four (4) business days after the

FERC Docket: RP08-319-000

Original Sheet No. 294 Original Sheet No. 294: Pending

GENERAL TERMS AND CONDITIONS

Shipper submits the offer. The reserve price matrix in the resulting open season will include an entry for the specific term in the offer by the Shipper initiating the open season. Otherwise, Natural can establish all parameters of the open season to the same extent as set out in Section 5.1(c) (1) or Section 5.1(c) (2), as applicable. Requests for a Shipper-Initiated Open Season may not include a Negotiated Rate or Negotiated Rate Formula.

(7) SPREADSHEETS Contemporaneous with posting of an open season, Natural shall make available (subject only to reasonable user requirements) electronic spreadsheets setting out the analyses which will be used to determine the winning bid(s), in a format which will allow the user to calculate net present value.

(8) CONDITIONS ON REQUESTS AND BIDS This subsection sets out conditions applicable to all requests and bids for firm capacity. Natural shall reject any request or bid for service which may detrimentally impact the operational integrity of Natural's system (if Natural rejects a bid or request on this basis, it will provide a written explanation of the operational basis for this action); which does not satisfy all the terms of an applicable posting and/or does not provide all the information required by the posting; which contains terms and conditions other than those in Natural's Tariff and/or an applicable posting; which would not constitute a valid request under the applicable rate schedule (it being understood that a bid cannot be rejected for incompleteness if all information required by the bid form and the applicable posting has been provided); or is in any way inconsistent with Natural's Tariff and/or an applicable posting. Any Shipper wishing to bid in an open season or submit a request for capacity must satisfy the creditworthiness requirements in Section 16 of these General Terms and Conditions prior to submitting a bid or request. A Shipper cannot bid for or request services which exceed its prequalified level of creditworthiness. Natural shall process - and encourages - applications from potential bidders or requesters seeking prequalification for bids or requests they may make in the future. Credit applications shall be completed in full with all information required to establish creditworthiness under the credit criteria included in Natural's rate schedule covering the applicable service. Should a potential bidder or requester fail to satisfy such credit criteria, it may still qualify by providing a prepayment, letter of credit, security interest or guarantee satisfactory to Natural as further set forth in Section 16 of these General Terms and Conditions. Based on Natural's continuing review of a Shipper's financial records, Natural shall have the right to amend a Shipper's line of credit and lower or increase the quantity and term.

FERC Docket: RP08-319-000

Original Sheet No. 295 Original Sheet No. 295: Pending

GENERAL TERMS AND CONDITIONS

(9) POSTING OF AWARD Natural will post on the Informational Postings portion of its Interactive Website the term, rate and description of any firm capacity awarded pursuant to any of the procedures in this Section 5.1(c). This posting shall be made no later than ten (10) days after firm capacity is awarded. This posting requirement will be superseded by the requirements for posting of firm capacity awards under Order No. 637 once those requirements become effective. Where applicable, Natural will also comply with Commission filing requirements regarding capacity awards which include a Negotiated Rate and Negotiated Rate Formula, as reflected in Section 49 of these General Terms and Conditions.

(10) SECTION 22 RIGHTS; RESERVED CAPACITY The capacity allocation procedures of this Section 5.1(c) shall not apply where a Shipper is utilizing unilateral rollover rights or the Right of First Refusal procedures provided in Section 22 of these General Terms and Conditions. Instead, the procedures in said Section 22 will govern the award of capacity. With respect to contractual rollover rights permitted under said Section 22, such rollover rights in contracts executed prior to January 1, 1999, shall be honored without regard to the procedures in this Section 5.1(c), but requests for firm capacity pursuant to contractual rollover provisions agreed upon in contracts executed after January 1, 1999, shall be considered on the same basis as any other request.

(11) CAPACITY RESERVED FOR EXPANSION PROJECTS Notwithstanding any other provision of this Section 5, Natural reserves the right, but shall not be obligated, to reserve for expansion projects capacity which is or will become available. Expansion projects can encompass capacity on Natural's existing system which will or may be needed by a Shipper in conjunction with capacity which may be acquired by that Shipper in a future extension project. For purposes of this Section 5.1(c)(11), there are four types of capacity that Natural may reserve subject to the conditions detailed below. The four types of capacity are:

(i) Capacity currently posted on the Informational Postings portion of Natural's Interactive Website which is Posted Capacity;

(ii) Capacity which had previously been posted for bidding pursuant to the Right of First Refusal (Section 22.3) provisions of these General Terms and Conditions and for which no acceptable bids were received and as to which Shippers rights under Section 22.3(f) have expired ("ROFR Capacity");

FERC Docket: RP08-319-000

Original Sheet No. 296 Original Sheet No. 296: Pending

GENERAL TERMS AND CONDITIONS

(iii) Capacity which will be returned to Natural in response to a direct solicitation from Natural to existing Shippers for permanent releases of capacity to serve an expansion project ("Turnback Capacity"); and

(iv) Capacity which will be returned to Natural by an existing Shipper at the termination or expiration of that Shipper's Agreement term(s) ("Terminated Agreement Capacity").

(Hereinafter, any references to the term "capacity" in this Section 5.1(c) (11) shall mean the four types of capacity collectively unless noted otherwise).

Seven days prior to Natural reserving any capacity pursuant to this Section 5.1(c)(11) of these General Terms and Conditions, Natural must first post on the Informational Postings portion of Natural's Interactive Website the capacity which Natural proposes to reserve (Reservation Posting). After seven days, the capacity described in the Reservation Posting shall be deemed to be reserved by Natural and Natural's posting under Section 5.1(e) of these General Terms and Conditions shall be adjusted accordingly. Reservation Posting shall include (but not be limited to): (1) a description of the expansion project for which the capacity will be reserved; (2) the quantity of capacity to be reserved; (3) the location of the proposed reserved capacity on the pipeline system, including all mainline valves and associated quantities at those valves; and (4) the projected in-service date of the expansion project. The Reservation Posting shall also include a non-binding solicitation for Turnback Capacity to serve the expansion project; provided, however, Natural will only solicit Turnback Capacity if it is unable to acquire sufficient capacity from Posted Capacity, ROFR Capacity or Terminated Agreement Capacity (it being understood that solicitation of Turnback Capacity will only occur in instances where redundant construction could be avoided through the use of Turnback Capacity). Any non-binding solicitation for Turnback Capacity shall be posted on the Informational Postings portion of Natural's Interactive Website no later than 90 days after the close of any open season which may be conducted in connection with the expansion project. Natural shall post on the Informational Postings portion of Natural's Interactive Website the results of any

FERC Docket: RP08-319-000

Original Sheet No. 297 Original Sheet No. 297: Pending

GENERAL TERMS AND CONDITIONS

non-binding solicitation for Turnback Capacity within ten (10) days after such results are known. While Natural's solicitation is non-binding, any response to a request for Turnback Capacity accepted by Natural shall be binding upon Shipper in the event Natural proceeds with the expansion project. The Reservation Posting shall remain on the Informational Postings portion of Natural's Interactive Website the entire period the capacity is being reserved and Natural shall make reasonable efforts to update the Reservation Posting up to the in-service date of the expansion project to reflect any material changes in the scope of the expansion project. The updated Reservation Posting shall also indicate whether and how much of the reserved capacity has been sold on a limited-term basis.

FERC Docket: RP08-319-000

Original Sheet No. 298 Original Sheet No. 298: Pending

GENERAL TERMS AND CONDITIONS

Natural may reserve capacity for an expansion project for a period of one (1) year or longer from the date of the Reservation Posting; provided, however, that the reservation of capacity must terminate on the date which is one (1) year from the date of the Reservation Posting in the event that an open season has not commenced (or the public announcement by the project sponsor of its intention to construct has not been issued) by the end of that one (1) year period. After the commencement of the open season (or the public announcement), the reservation of capacity may continue until the in-service date of the expansion project. In addition, prior to reserving Terminated Agreement Capacity, Natural must make Terminated Agreement Capacity generally available through the procedures in this Section 5.1(c), unless such capacity has been previously subjected to the ROFR procedures. When a Terminated Agreement Capacity auction is held prior to any expansion project open season, Natural shall have the right to state in the posting minimum terms and conditions for bids that would be acceptable for consideration that are the same as the minimum terms and conditions anticipated for any open season which may be held in connection with the future expansion project. In the event that any subsequent expansion project open season imposes minimum terms and conditions that are materially different from the terms and conditions imposed in the previous Terminated Agreement Capacity auction, Natural shall hold another open season under Section 5.1(c)(1) for the Terminated Agreement Capacity that uses the same minimum terms and conditions as were imposed for the expansion project open season. If any expansion project open season is held prior to or during the Terminated Agreement Capacity auction, Natural shall use the same minimum terms and conditions as used for the expansion project open season.

Natural will not, absent Commission approval, accept advance payments to reserve capacity under this Section $5.1(c)\ (11)$.

Any capacity reserved under this Section 5.1(c)(11) shall be made available pursuant to these General Terms and Conditions on a limited-term basis up to the in-service date of the expansion project(s). For such limited-term agreements, Natural reserves the right to limit any extension rights provided in the service agreement and pursuant to Section 22 commensurate with the in-service date of the expansion project. Natural will indicate in any auction posting of the capacity the limitations on extension rights that will apply to such limited-term transportation service. The projected in-service date of the expansion project shall be set out in any limited-term agreement (recognizing that the date could slip if construction is delayed).

FERC Docket: RP08-319-000

Original Sheet No. 299 Original Sheet No. 299: Pending

GENERAL TERMS AND CONDITIONS

Reservation of capacity under this Section 5.1(c)(11) shall not in any way modify or limit the rights under Sections 18, 19 and 22 of these General Terms and Conditions of an existing Shipper under a contract entered into prior to September 1, 1999, and shall only modify other Shipper's rights to the extent specified in this Section and in Section 22.1 of these General Terms and Conditions.

- (12) CAPACITY AWARDED FOR INTERIM PERIOD. This provision applies in situations where firm capacity is awarded commencing on a future date and such capacity is not already subject to firm contract(s) for the entire interim period before such future service commencement date. Nothing herein shall affect any Extension Right, as defined in Section 5.1(c)(2) hereof, in effect on the initial effective date of this provision.
- (i) Natural may market such capacity for the interim period until service under the capacity award becomes effective.
- (ii) If firm capacity is available for only an interim period for any reason (e.g., because the firm capacity is already contracted at a future date after such interim period or the firm capacity is only available physically or operationally for that interim period), then Natural may limit the Extension Rights of the Shipper awarded the capacity for that interim period. Natural will indicate in any open season posting for such capacity the limitation on Extension Rights which will apply to such firm capacity awarded for the interim period.
- (d) EVALUATION OF COMPETING BIDS AND REQUESTS In comparing valid bids received in an open season or in comparing two or more valid and competing pending requests for service under the Request Procedure, Natural will award firm capacity based on the highest economic value, as defined in this Section 5.1(d).
- (1) HIGHEST ECONOMIC VALUE The highest economic value is the highest net present value of the stream of incremental revenue produced by a valid bid or request, or combination of valid bids or requests, received by Natural for firm capacity. Incremental revenue is the additional revenue Natural would collect from a Shipper under any bid or request over and above the revenue Natural would otherwise have received after taking into account any revenue lost or affected by the bid or request (i.e., where an existing capacity holder submits a

FERC Docket: RP08-319-000

Original Sheet No. 300 original Sheet No. 300: Pending

GENERAL TERMS AND CONDITIONS

bid or request which is contingent upon turnback by that existing capacity holder of an existing capacity commitment, only the value of such a bid or request net of the revenue which would be lost to Natural due to the turnback of the existing contractual commitment will be considered).

- POSTING OF NET PRESENT VALUE FORMULA AND OTHER CRITERIA Natural will post the criteria to be used in the determination of highest economic value for comparing valid bids in any open season and for comparing pending requests which are valid and competing. The posting will consist of a net present value formula, together with all relevant factors and parameters, such as discount rates. These criteria shall be posted continuously on the Informational Postings portion of Natural's Interactive Website. Natural may change the criteria at any time in a manner not inconsistent with the other provisions of the Section 5.1(d), but the revised criteria may only be applied to an open season the posting of which commenced at least three (3) business days after the change in criteria has been posted. Natural cannot change the criteria for any on-going open season. The revised criteria shall apply immediately to all requests received after the change has been posted. In addition, the posting for each individual open season will include the following elements: the date to which all bids are discounted in the calculation of net present value and the discount rate utilized; how a Shipper's willingness vel non to prorate will affect its bid; the extent to which advance payments will be considered in evaluating bids; and how surcharges and optional service charges will be taken into account.
- (3) ONLY GUARANTEED REVENUE CONSIDERED In the determination of highest economic value, Natural shall consider only reservation charge revenue and any other guaranteed revenue. In the case of a bid or request for firm service involving a Negotiated Rate or Negotiated Rate Formula, the rules on calculating net present value set out in Section 49.2(b) of these General Terms and Conditions shall apply.
- (4) OBLIGATION TO AWARD CAPACITY Natural shall be obligated to award firm capacity if the applicable maximum rate is submitted for the entire term of a valid bid in any open season or is requested for the entire term of a valid request under the Request Procedure, up to the firm capacity covered by the maximum rate bid(s) or request(s). The capacity award will be based on the posted criteria hereunder for evaluating bids or requests. Similarly, Natural shall be obligated to award firm capacity if the reserve price is met in a valid bid for the capacity being posted for open season, up to the firm capacity covered by the bid(s) meeting the reserve price. The

FERC Docket: RP08-319-000

Original Sheet No. 301 original Sheet No. 301: Pending

GENERAL TERMS AND CONDITIONS

capacity awards will be based on application of the posted criteria. Notwithstanding the foregoing, except for conditional capacity awards in an Early Open Season, Natural shall not be obligated to award firm capacity based on the following types of bids or requests: (i) any bid or request for a term of less than one year, under which service is to commence more than sixty (60) days following the close of the open season; and (ii) any bid or request for a term which is not continuous from the commencement of service date to the termination of service date reflected in the bid.

- (5) RESERVE PRICE MATRIX In an Initial Open Season, an Early Open Season, or a Shipper-Initiated Open Season, Natural shall establish a reserve price or reserve price matrix. Natural may, but is not required to, establish a reserve price or reserve price matrix in an Additional Open Season.
- (i) A reserve price defines the minimum price(s) at which Natural will award the firm capacity covered by the open season, which price(s) must be equal to or less than the applicable maximum rate(s). Natural is not obligated to award capacity at less than the applicable maximum rate unless a lower rate is set out in the reserve price matrix. In determining whether a reserve price has been met by any bid, Natural will compare the net present value of the requested prices in the bid with the net present value of the applicable reserve price(s). If the reserve price is met, Natural will award firm capacity to valid bids consistent with the posted bid evaluation criteria and this Section 5.1(d).
- (ii) Reserve prices in a reserve price matrix may vary by relevant elements, including but not limited to term, service type, Receipt Points, Delivery Points and markets. Differences in "markets", as that term is used in the prior sentence, shall refer to differences relating to any of the following: (A) geographical areas (zones, segments, points, etc.), where the value of Natural's services may vary among such geographical areas due to current or projected differences in competitive alternatives, regulation, or operational, supply or capacity factors; (B) customers, where the value of Natural's services may vary by customer due to current or projected differences in competitive alternatives available to them or demand their characteristics (including access to alternative fuels; (C) contract time periods, where the value of Natural's services may vary due to current or projected differences in competitive alternatives or market demand at or over different contract time periods (for example, winter season versus multi-year); (D) products and services, where the value of Natural's product and service offerings may vary due to current or projected differences in competitive alternatives or market

FERC Docket: RP08-319-000

Original Sheet No. 302 Original Sheet No. 302: Pending

GENERAL TERMS AND CONDITIONS

demand (provided that this is not intended to tie separate service together for bidding purposes); or (E) volumes, where the value of Natural's services may vary due to current or projected differences in competitive alternatives, market demand or other factors related to contract volume levels. The reserve price matrix in an Initial Open Season will include multiple terms, from one month or less to multi-year, unless firm capacity is only available for a limited time due to operational or physical constraints or due to the contractual commitment of such capacity in the future.

(iii) Natural shall provide the reserve price or reserve price matrix to an independent third party before the posting of an open season. A bidder in an open season may request the reserve price or reserve price matrix applicable to that open season at any time after the close of the open season and prior to the posting described in the following sentence, and Natural will provide such information to the requesting bidder within one business day after Natural receives the request. Any reserve price or reserve price matrix used in an open season will be posted on the Informational Postings portion of Natural's Interactive Website within thirty (30) days after the close of that open season.

(6) TIE-BREAKER In the event of a tie for highest economic value between two or more competing bids or requests of identical value under the criteria posted consistent with this Section 5.1(d), the procedures in this subsection shall apply. Natural will first notify the Shippers submitting the tying bids or requests that the evaluation resulted in a tie and shall provide each such Shipper an opportunity to improve its bid or request through an entirely new closed bid or request (which must have a higher value than the tying bid or request) submitted via facsimile within 24 hours of such notification. If a tie still exists after the first step, Natural will next allocate the available capacity pro rata to each tying Shipper which has indicated its willingness to prorate. If any capacity remains to be awarded after the second step and a tie still remains, the tie will be broken by coin toss.

(e) POSTING OF FIRM CAPACITY

(1) Natural posts on the Informational Postings portion of its Interactive Website the firm capacity currently available on its system, as provided in Section 17.1(b)(1) of these General Terms and Conditions. Natural also posts consistent with the remainder of this Section 5.1(e), forward-haul firm transmission segment capacity and all Receipt Point, Delivery Point and storage firm capacity becoming available in the future (excluding firm capacity created by the construction of new facilities, including both expansions and extensions, until the initial allocation of such capacity has been made). Postings hereunder will be adjusted for

FERC Docket: RP08-319-000

Original Sheet No. 303 Original Sheet No. 303: Pending

GENERAL TERMS AND CONDITIONS

capacity reserved under Section 5.1(c) (11) of these General Terms and Conditions. Natural is not obligated to post other types of firm capacity. The capacity which must be posted hereunder is Posted Capacity as referenced in Section 5.1(c) (1), except as provided in Section 5.1(e) (8).

- (2) In addition to the firm capacity which must be posted under Section 5.1(e) (1), Natural will also post the availability of firm backhaul capacity prior to awarding such capacity except through the exercise of Extension Rights. Natural will post a statement that firm backhaul capacity is available and will specify the portion of Natural's system on which such capacity is available. Because the extent of backhaul capacity is contingent on the specific path, Natural is not obligated to post a specific volume or term with respect to the firm backhaul available. Natural will award firm backhaul capacity consistent with Section 5.1(c) (3).
- (3) The posting as to any firm capacity in Section 5.1(e)(1) must be made no later than fifteen (15) months prior to the date on which that firm capacity will become available and shall be updated daily to reflect changes; provided, however, that Natural shall post the availability of firm capacity prior to providing any existing or potential Shipper with information on the availability of such capacity. The posting shall not include firm capacity which remains subject to tariff rollover rights, contractual rollover rights, or the right of first refusal.
- (4) The posting pursuant to Section 5.1(e)(1) will contain at a minimum the following information:
- $\hbox{(i)} \qquad \hbox{The date on which the firm capacity will} \\ \hbox{first become available;}$
- (ii) The amount in MDQs of firm capacity becoming available:
- (iii) The type of firm service(s) which will become available: and
- $\,$ (iv) For transmission service, the pipeline segments and the receipt and delivery points at which firm capacity is becoming available and the related MDQs.
- (5) Once posting occurs, the posting will continue until the firm capacity is awarded (if a portion of the firm capacity is awarded, the posting will be revised to reflect the remaining available firm capacity) or is no longer available.

FERC Docket: RP08-319-000

Original Sheet No. 304 Original Sheet No. 304: Pending

GENERAL TERMS AND CONDITIONS

(6) Until the availability of capacity has been posted hereunder, Natural will not award firm capacity which must be posted under Sections 5.1(e)(1) and 5.1(e)(2), except on a conditional basis as to capacity which must be posted under Section 5.1(e)(1)(8) pursuant to an Early Open Season.

- (7) Capacity which may become available subject to a unilateral tariff rollover right, a contractual rollover right or right of first refusal pursuant to Section 22 of these General Terms and Conditions may not be awarded unless and until all of the rights of the existing Shipper under said contingency have terminated or expired, except in a manner consistent with Sections 5.1(c) and 5.1(d).
- (8) In the case of an Early Open Season, the availability of the firm capacity which is to be the subject of the Early Open Season shall be posted at least one (1) day prior to the commencement of the Early Open Season and, where applicable, the posting shall include a notation that the capacity remains subject to rollover rights and/or the Right of First Refusal.

5.2 SCHEDULING OF FIRM SERVICES

(a) While firm services are not ordinarily interrupted or nominations for firm service within MDQ declined due to lack of capacity, capacity constraints may exist from time to time or limitation or interruption of firm service may be necessary for

FERC Docket: RP08-319-000

Original Sheet No. 305 Original Sheet No. 305: Pending

GENERAL TERMS AND CONDITIONS

certain other reasons. Natural may decline to schedule firm service for any of the following reasons and may curtail for any of the same reasons except (iii):

- $\hbox{(i)} \qquad \text{if Shipper tenders gas which does not} \\ \text{conform to the applicable pressure or quality requirements of these} \\ \text{General Terms and Conditions;}$
 - (ii) for reasons of Force Majeure;
- (iii) due to routine repair and maintenance to be reasonably determined by Natural;
- $\qquad \qquad \text{(iv)} \qquad \text{pursuant to Section 5.11 of these General} \\ \text{Terms and Conditions;}$
- (v) to rectify imbalances or to conform physical flows to nominations to the extent consistent with the specific Rate Schedule;
- (vi) with respect to service under Rate Schedule FFTS, if the day is not an Available Day or partial Available Day under the relevant FFTS Agreement;
- (vii) pursuant to the Operational Control provisions of these General Terms and Conditions;
 - (viii) to maintain system integrity; or
- $\mbox{\sc (ix)}$ if there is a dispute over title, ownership or right to tender or to receive gas.

Without limitation to the foregoing, Natural shall have the right to reduce receipts, deliveries, injections or withdrawals of natural gas on any day below Shipper's MDQ and/or below the storage rights applicable to Shipper under the specific Rate Schedule to permit maintenance, repair, overhaul, replacement, or construction of pipelines, compressors, metering, regulating, or other production, gathering, and transmission facilities and equipment, or to maintain system integrity; provided, however, that with respect to routine repair and maintenance, Natural

FERC Docket: RP10-147-005

First Revised Sheet No. 306 First Revised Sheet No. 306

Superseding: Original Sheet No. 306

GENERAL TERMS AND CONDITIONS

will implement restrictions for scheduling purposes only, not for curtailment, and will attempt to schedule such activity during a period when it will not result in limitation of firm services, or when such limitation will be minimized, after consulting with the Shippers which could be affected. If Natural curtails or limits the scheduling of firm volumes pursuant to this Section 5.2(a) for reasons not caused by a Shipper, Natural shall waive applicable tariff restrictions and requirements as to such Shipper (such as seasonal cycling requirements) to the extent that such a waiver is operationally feasible, but such a waiver shall only be granted if and to the extent Shipper could not reasonably be expected to meet such requirements in light of Natural's actions hereunder.

- (b) For Shippers under all firm services, Natural shall provide notice of any curtailment or of any scheduling restriction as far in advance as feasible. Natural shall attempt to provide at least two (2) Days' prior notice, unless more timely action is necessary to respond to a Force Majeure situation, to balance the Agreement to the extent consistent with the applicable Rate Schedule, or to maintain system integrity.
- (c) (1) As used in this Section 5.2(c), Firm Daily Volume shall mean the volume of gas which Natural is obligated to deliver on a firm basis for Shipper at primary Delivery Point(s) under a firm contract on a Day, based on nominations for firm service within firm contract rights at such primary points which could be confirmed plus any applicable firm no-notice rights Shippers seek to exercise on that Day within its firm contract rights under that contract. Except as otherwise provided in this Section 5.2(c) of these General Terms and Conditions, in the event Natural fails to deliver on any Day under any firm contract at least 98% of Shipper's Firm Daily Volume for that Day, then the applicable Reservation Charges and related reservation-based surcharges shall be eliminated for the quantity of gas not delivered by Natural within the Shipper's Firm Daily Volume under the contract.
 - (2) Notwithstanding subsection (1) above, if,
- (i) Natural, for any Day, posts on its interactive website a restriction in firm capacity affecting a segment within the path between Shipper's Primary Receipt Point(s) and Primary Delivery Point(s) that includes a minimum percentage of firm MDQ that will be available to Shipper through the restricted area; and

FERC Docket: RP10-147-005

Original Sheet No. 306A Original Sheet No. 306A

GENERAL TERMS AND CONDITIONS

(ii) Shipper submits a valid nomination (that can be fully confirmed by the upstream and downstream operators) in the timely nomination cycle for such Day for at least such posted minimum percentage of Shipper's MDQ through the restricted area; and

(iii) Shipper's nomination is not reduced in subsequent nomination cycles for such Day,

Shipper's Reservation Charges and related reservation-based surcharges shall be eliminated for the quantity of gas not scheduled by Natural between Shipper's valid nomination (including any increase in Shipper's Timely nomination scheduled in later nomination cycles) and Shipper's MDQ through the restricted area. For purposes of this subsection (2), Shipper's otherwise valid nomination shall not be deemed to be reduced below the minimum percentage posted by Natural, as a result of any incidental minimal reduction in Shipper's nomination during the scheduling process.

(3) With respect to subsection (2) above, Shipper's Reservation Charges and related reservation-based surcharges shall not be eliminated to the extent that the Shipper uses secondary point service outside the affected segment. Moreover, if capacity in excess of all shipper nominations is available through the restricted area after the Timely nomination cycle for any Day, any adjustment to Shipper's Reservation Charges and related reservation-based surcharges that may have resulted from Shipper's nomination for such Day shall be reduced by the amount of any such additional capacity that remains unutilized after the Timely nomination cycle on a revised minimum percentage of firm MDQ that is available to Shipper for nomination in the Evening nomination cycle.

(4) Natural shall not be obligated to adjust the Reservation Charge under any contract pursuant to Section 5.2(c) when Natural's failure to deliver on any Day at least 98% of the Firm Daily Volume:

(i) is the result of the conduct of Shipper or the downstream operator of the facilities at the Delivery Point; or

FERC Docket: RP10-147-005

First Revised Sheet No. 307 First Revised Sheet No. 307

Superseding: Original Sheet No. 307

GENERAL TERMS AND CONDITIONS

(ii) occurs either (a) within ten (10) days following a force majeure event as contemplated by Section 27 of the General Terms and Conditions, or (b) prior to the date Natural has or should have, in the exercise of due diligence, overcome the force majeure event, whichever occurs first.

5.3 GENERAL SCHEDULING PRIORITIES

Natural shall schedule gas properly nominated and confirmed under these General Terms and Conditions in accordance with the following priority sequence, with transportation services scheduled to the extent transportation capacity is available and storage injections and withdrawals scheduled separately to the extent injection, withdrawal and storage capacity are available:

(a) TRANSPORTATION SERVICES (including service on portions of the system where a gathering charge applies and the transportation portion of service under Rate Schedule DSS and, where separately released, under Rate Schedule FRSS):

FERC Docket: RP08-319-000

Original Sheet No. 308 Original Sheet No. 308: Pending

GENERAL TERMS AND CONDITIONS

(1) Natural shall first schedule all firm services (including service under an FFTS Agreement at Shipper's MDQ on an Available Day or at the reduced volume on a partial Available Day) within MDQ at primary points and at secondary points within a primary point path where no point constraint exists. To the extent capacity does not exist to provide for all volumes nominated by Shippers on a firm basis within MDQ at primary points and at secondary points along any path defined by primary points where no point constraint exists under all firm Rate Schedules, plus all firm no-notice rights in excess of such nominations, available capacity shall be allocated pro rata based on MDQ (including MDQ under an FFTS Agreement on an Available Day or the reduced MDQ on a partial Available Day) on any portion of Natural's system affected by a capacity constraint (capacity allocated to any Agreement shall be limited to confirmed nominations except to the extent of nonotice rights available under such Agreement). If Natural cannot schedule all such nominations or if Natural anticipates that it cannot accommodate all such nominations plus the firm no-notice rights it expects may be utilized on that Day, Natural shall limit the scheduling of such nominations and the exercise of firm no-notice rights for that Day on a pro rata basis, based on MDQ and shall post such limitations on the Informational Postings portion of its Interactive Website. Such posting shall also include the following information: the reason(s) for the service limitation; the expected duration of the service limitation; and the actions which Natural is taking to restore firm service. Notice of a service limitation will be provided through Electronic Notice Delivery consistent with NAESB Standards as adopted in Section 48 of these General Terms and Conditions and will be posted on the Informational Postings portion of Natural's Interactive Website. Where feasible, Natural will endeavor to provide and post the notice before 4:00 p.m. Central Time or otherwise will endeavor to notify Shippers by 4:00 p.m. Central Time that they should check the Informational Postings portion of Natural's Interactive Website again at a specified later time to see whether a service limitation will be in effect for the next Day. In addition to the posting, Natural must attempt to give actual notice of a service limitation to affected firm Shippers via telefax or telephone (provided a Shipper has given the numbers to Natural as required in Section 23.1 of these General Terms and Conditions). If feasible, such notice should be given at least four (4) hours prior to the start of the Day during which a service limitation will be effective as to a Shipper.

(2) Natural shall next schedule firm service at other secondary points. Where capacity is not sufficient to grant all such secondary point requests, the secondary point priorities set out in Section 5.5 shall govern.

FERC Docket: RP08-319-000

Original Sheet No. 309 Original Sheet No. 309: Pending

GENERAL TERMS AND CONDITIONS

(3) Third, Natural shall schedule other interruptible and authorized overrun services, applying scheduling priorities set out in Section 5.7.

(b) STORAGE SERVICES (including the storage portion of service under Rate Schedules DSS and FRSS):

(1) Natural shall first schedule all firm injections and withdrawals nominated within MDQ and applicable firm storage rights. If capacity is insufficient to satisfy all such nominations, plus firm no-notice rights in excess of nominations [including the injections or withdrawals under Rate Schedules DSS and NSS within MDQ but in excess of generally applicable firm rights where notice has been given forty-eight (48) hours in advance that capacity is available], allocation of available capacity shall be pro rata based on MDQ (capacity allocated to an NSS or FRSS Agreement shall be limited to confirmed nominations thereunder except to the extent of no-notice service under Rate Schedule NSS in conjunction with FTS-G/NB, FTS-NB and no-notice service under Rate Schedule FRSS on withdrawal). If Natural cannot schedule all such nominations or if Natural anticipates that it cannot accommodate all such nominations plus the firm no-notice rights it expects may be utilized on that Day, Natural shall limit the scheduling of such nominations and the exercise of firm no-notice rights for that Day on a pro rata basis, based on MDQ and shall post such limitations on the Informational Postings portion of its Interactive Website. Such posting shall also include the following information: the reason(s) for the service limitation; the expected duration of the service limitation; and the actions which Natural is taking to restore firm service. Notice of a service limitation will be provided through Electronic Notice Delivery consistent with NAESB Standards as adopted in Section 48 of these General Terms and Conditions and will be posted on the Informational Postings portion of Natural's Interactive Website. Where feasible, Natural will endeavor to provide and post the notice before 4:00 p.m. Central Time or otherwise will endeavor to notify Shippers by 4:00 p.m. Central Time that they should check the Informational Postings portion of Natural's Interactive Website again at a specified later time to see whether a service limitation will be in effect for the next Day. In addition to the posting, Natural must attempt to give actual notice of a service limitation to affected firm Shippers via telefax or telephone (provided a Shipper has given the numbers to Natural as required in Section 23.1 of these General Terms and Conditions). If feasible, such notice should be given at least four (4) hours prior to the start of the Day during which a service limitation will be effective as to a Shipper.

FERC Docket: RP08-319-000

Original Sheet No. 310 original Sheet No. 310: Pending

GENERAL TERMS AND CONDITIONS

(2) Natural shall next schedule authorized overrun injections and withdrawals properly requested under DSS and NSS Agreements in excess of firm storage rights but within MDQ and MSV [including such injections and withdrawals within MDQ and MSV under Sections 5.3(c) of Rate Schedule DSS and 5.3(b) of Rate Schedule NSS or as to which a waiver has been granted under Section 5.1(c) of Rate Schedule DSS]. If Natural cannot provide all such requests, scheduling shall be pro rata based on MDQ.

- (3) Natural will next schedule all other interruptible and authorized overrun injections and withdrawals (including injections and withdrawals under Rate Schedules FRSS, DSS and NSS in excess of MDQ and/or MSV), applying the scheduling priorities set out in Section 5.7.
- (4) Special rules regarding the treatment of storage points for scheduling and curtailment purposes are set out in Section 5.6 of these General Terms and Conditions.
- (c) Firm intra-day nominations are entitled to bump scheduled interruptible volumes, as defined in Section 5.7, only during the Evening and Intra-day 1 Nomination Cycles, as defined in Section 8.2. Firm intra-day nominations are not entitled to bump already scheduled firm volumes.

5.4 GENERAL CURTAILMENT PRIORITIES

In interrupting or curtailing service, Natural shall curtail in accordance with the following priority sequence, with transportation services curtailed to the extent there is insufficient transportation capacity and storage injections and withdrawals curtailed separately to the extent there is insufficient injection, withdrawal or storage capacity (for the purposes of these priorities, service at Shipper's MDQ on an Available Day under any FFTS Agreement or at the reduced volume on a partial Available Day, shall have the same priority as Rate Schedule FTS):

FERC Docket: RP08-319-000

Original Sheet No. 311 Original Sheet No. 311: Pending

GENERAL TERMS AND CONDITIONS

(a) TRANSPORTATION SERVICES (including service on portions of the system where a gathering charge applies and the transportation portion of service under Rate Schedule DSS and, where separately released, under Rate Schedule FRSS):

- $\,$ (1) Natural shall first interrupt service to interruptible and overrun services, applying the curtailment priorities set out in Section 5.7.
- (2) Second, Natural shall interrupt firm services within MDQ (including service under an FFTS Agreement at Shipper's MDQ on an Available Day or at the reduced volume on a partial Available Day) at primary points and at secondary points. Curtailment shall be pro rata based on MDQ (including MDQ under an FFTS Agreement on an Available Day or the reduced MDQ on a partial Available Day) on any portion of Natural's system affected by a capacity constraint (capacity allocated to any Agreement shall be limited to confirmed nominations except to the extent of no-notice rights under such Agreement).
- (b) STORAGE SERVICES (including the storage portion of service under Rate Schedules DSS and FRSS):
- (1) Natural shall first interrupt any interruptible and authorized overrun injections and withdrawals under Rate Schedules BESS and PALS, or under Rate Schedules DSS, FRSS and NSS in excess of MDQ, applying the curtailment priorities set out in Section 5.7.
- (2) Natural shall next interrupt any authorized overrun injections and withdrawals under Rate Schedules DSS and NSS in excess of firm storage rights but within MDQ and MSV [firm rights include the injections and withdrawals under Rate Schedule DSS or NSS within MDQ and MSV but in excess of generally applicable firm rights where notice has been given forty-eight (48) hours in advance that capacity is available]. This priority shall include authorized overrun injections and withdrawals within MDQ and MSV under Section 5.3(c) of Rate Schedule DSS, Section 5.3(b) of Rate Schedule NSS or pursuant to a waiver under Section 5.1(c) of Rate Schedule DSS. If Natural can provide some but not all of such services, curtailment shall be pro rata based on MDQ.
- (3) Natural shall last interrupt firm injections and withdrawals within MDQ and applicable firm injection and withdrawal rights [including the injections or withdrawals under Rate Schedules DSS and NSS within MDQ but in excess of generally applicable firm rights where notice has been given forty-eight (48) hours in advance that capacity is available]. If capacity is sufficient to provide some but not all of such services, curtailment shall be pro rata based on MDQ (the capacity allocated to an NSS or FRSS

FERC Docket: RP08-319-000

Original Sheet No. 312 Original Sheet No. 312: Pending

GENERAL TERMS AND CONDITIONS

Agreement is limited by confirmed nominations thereunder except to the extent of no-notice rights under Rate Schedule NSS in conjunction with Rate Schedule FTS-NB, FFTS-NB, FTS-G/NB and no-notice service under Rate Schedule FRSS on withdrawal).

Special rules regarding the treatment of storage points for scheduling and curtailment purposes are set out in Section 5.6 of these General Terms and Conditions.

5.5 SECONDARY POINTS

(a) (1) To the extent provided in this Tariff, Shippers under Rate Schedules DSS, FRSS, FTS, FFTS and FTS-G shall have the right to use secondary Receipt and Delivery Points. A Shipper which elects the SW service option under Rate Schedule FTS, FFTS or FTS-G (and subject to the restrictions in such Rate Schedules) shall have the right to all Receipt and Delivery Points on Natural's system as secondary points. Otherwise, except as may be specifically provided to the contrary in this Tariff, Shippers under Rate Schedule FTS, FFTS or FTS-G shall be entitled to all Receipt and Delivery Points as secondary points in any rate zone which is within a path created by that Shipper's primary Receipt and Delivery Points; and such secondary points shall include paper pooling and storage points in each receipt zone as provided in these General Terms and Conditions. In addition, a Shipper under Rate Schedule FTS, FFTS or FTS-G shall have the opposite leg rights set out in (2). A Shipper under Rate Schedule DSS using nominated firm transportation rights thereunder shall have as secondary Receipt and Delivery Points all points in the Iowa-Illinois Receipt Zone and the Market Delivery Zone, respectively, with an MDQ at each point equal to its aggregate MDQ. A Shipper under Rate Schedule FRSS shall have as secondary Delivery Points all points in the zone in which the primary Delivery Point(s) are located, with an MDQ at each point and at the aggregate of such points no greater than its aggregate MDQ at primary points in the zone. To the extent a Shipper elects the SW service option under Rate Schedule FTS, FFTS or FTS-G, such Shipper shall have an MDQ at any secondary point equal to the aggregate MDQ. The MDQ for a Shipper under any other type of firm Agreement at any secondary Receipt or Delivery Point shall equal the sum of aggregate MDQs for all primary Receipt Points in the same receipt zone or in any upstream receipt zone, minus the aggregate MDQs for all primary Delivery Points in any upstream receipt zone. Any secondary point service in excess of a Shipper's secondary point MDQ shall be overrun service. The priority of service at secondary points under firm Rate Schedules shall be governed by the remainder of this Section.

(2) A Shipper under an FTS, FFTS or FTS-G Agreement with a primary Delivery Point in the Market Delivery Zone would have, in addition to secondary point rights under (1) above, rights to secondary Receipt Points on an out-of-path basis equal to the

FERC Docket: RP08-319-000

Original Sheet No. 313 Original Sheet No. 313: Pending

GENERAL TERMS AND CONDITIONS

aggregate MDQ of the contract path(s) between its primary Receipt Point(s) in the relevant receipt zone and its primary Delivery Point(s) in the Market Delivery Zone on the opposite leg (Amarillo or Gulf Coast Line) from that on which a primary path lies under its Agreement, as follows:

- (i) Such a Shipper with a primary Receipt Point in the Amarillo Mainline Receipt Zone would have out-of-path secondary Receipt Point rights in the Gulf Coast Mainline Receipt Zone and vice versa for use in connection with deliveries to the Market Delivery Zone or to the storage point in the zone of receipt.
- (ii) Such a Shipper with a Receipt Point in the Midcontinent Zone would have out-of-path secondary Receipt Point rights in the Gulf Coast Mainline and Texok Receipt Zones, while a Shipper with a primary Receipt Point in the Texok Receipt Zone would have out-of-path secondary Receipt Point rights in the Midcontinent and Amarillo Mainline Receipt Zones. These additional secondary Receipt Point rights could only be used in conjunction with deliveries into the Market Delivery Zone or to the storage point in the zone of receipt.
- (iii) Such a Shipper with a primary Receipt point in the Permian Receipt Zone would have out-of-path secondary Receipt Point rights in the Gulf Coast Mainline, Texok, South Texas and Louisiana Receipt Zones. Such a Shipper with a primary Receipt Point in the Louisiana Receipt Zone or the South Texas Receipt Zone would have out-of-path secondary Receipt Point rights in the Permian, Midcontinent and Amarillo Mainline Receipt Zones. These additional secondary Receipt Point rights could be used only in conjunction with deliveries into the Market Delivery Zones or to the storage point in the zone of receipt. The Shipper actually utilizing opposite leg out-of-path secondary Receipt Point rights on any day would pay (in addition to the Reservation Charges under its Agreement) the commodity rate applicable to the actual transportation service provided and the positive reservation rate differential, if any, to the extent the reservation charge, expressed on a 100% load factor basis, is higher for the path of actual flow than for the contract path under the Agreement. Fuel will be assessed based on the actual path utilized. These secondary Receipt Point rights would be available to the original Shipper and any replacement Shipper taking a release of the entire Agreement or a portion of the MDQ for the entire Agreement. These rights would not be available to replacement Shippers under a segmented release. Nothing herein shall affect Delivery Point rights.

FERC Docket: RP08-319-000

Original Sheet No. 314 Original Sheet No. 314: Pending

GENERAL TERMS AND CONDITIONS

- (b) As set out in Sections 5.3 and 5.4, service at the secondary Receipt and Delivery Points under any DSS, FRSS, FTS, FFTS or FTS-G Agreement shall be provided to the extent capacity is available at such points after all nominations for primary point service under all of Natural's firm Agreements have been satisfied; provided, however, that unless a capacity constraint exists at the point (and such exception applies only to the extent of such point constraint), a secondary point nomination at a point within a path created by Shipper's primary points shall be treated the same as a nomination by Shipper at a primary point. Special rules applicable to storage points are set out at Section 5.6 of these General Terms and Conditions.
- (c) In scheduling service at secondary points, the following rules shall apply:
- (1) For a secondary point outside a path created by Shipper's primary points, service at the point and service to or from the point shall be scheduled before interruptible service but shall be scheduled after nominations for primary point service. All types of out-of-path secondary points shall be treated on the same basis, including opposite leg rights and rights under the FTS-SW and FFTS-SW service options. For nominated transportation service under Rate Schedule DSS, all Receipt Points shall be deemed out-of-path except those specifically identified in the DSS Agreement pursuant to Rate Schedule DSS as being in-the-path.
- (2) Service to or from such a secondary point outside the path shall also be scheduled after secondary point service within the path to the extent both services utilize the same capacity. If a capacity constraint exists at the point, subsection (d) shall govern.
- (3) If capacity is available to schedule some but not all service nominated to and from secondary points outside a primary point path, capacity shall be allocated among secondary points under all firm Agreements which are outside the path, pro rata based on MDQ unless there is a point constraint, in which case subsection (d) shall govern.
- (4) Secondary point service shall not be subject to allocation [except as set out in Section 5.2(a)] if no capacity constraint exists at the point or on any segment to or from the point.

FERC Docket: RP08-319-000

Original Sheet No. 315 Original Sheet No. 315: Pending

GENERAL TERMS AND CONDITIONS

- (d) If nominations by all Shippers under all DSS, FRSS, FTS, FFTS and FTS-G Agreements for secondary point service for which such Shippers are eligible exceed Natural's available capacity at any secondary point, available capacity shall be allocated to all DSS, FRSS, FTS, FFTS and FTS-G Shippers within their applicable MDQs at such secondary point(s) based on MDQ.
- $% \left(0\right) =0$ (e) Notwithstanding the foregoing, unless a nomination for service at a secondary point is submitted by the date nominations are due under these General Terms and Conditions for service to commence on the first day of a month, that nomination shall (subject to the remainder of this subsection) be subordinate for that month, for scheduling purposes, to timely first-of-the-month nominations for secondary point service under Rate Schedules DSS, FRSS, FTS, FFTS or FTS-G. Once service under a firm Rate Schedule has commenced during a month at the secondary point, the service will not be interrupted during that month as a result of subsequent nominations for secondary point DSS, FRSS, FTS, FFTS or FTS-G service that would have had higher priority if submitted as a timely first-of-the-month nomination except as follows: (1) properly submitted and confirmed mid-month firm service nominations at primary points will supersede any secondary point service outside the path or where there is a constraint at the point; and (2) properly submitted and confirmed mid-month nominations at secondary points within a path created by primary points will supersede secondary point service outside the path unless the capacity constraint is only at the point. Confirmed mid-month nominations within MDQ at a secondary point by a holder of firm service will interrupt service at that point under any interruptible Rate Schedule.
- (f) The primary Receipt and Delivery Points define the primary path(s), including the direction of flow for the primary path(s). Shippers may nominate service at secondary points so that the direction of flow is the same as or the opposite from the primary path direction of flow. However, if the primary path direction of flow is physically a backhaul on a day, a nomination for service that would require a physical forward haul shall be treated as being outside the primary path on that day.

5.6 POOLING POINTS AND STORAGE POINTS

(a) (1) Service at pooling points shall be governed by Section 9 of these General Terms and Conditions. Except where the Automatic Agency Procedures set forth in subsection (2) below apply, to determine the volumes which can be accommodated at a pooling

FERC Docket: RP08-319-000

Original Sheet No. 316 original Sheet No. 316: Pending

GENERAL TERMS AND CONDITIONS

point, Natural shall evaluate the capacity available both: (i) upstream and at Receipt Points for gas tendered pursuant to the Agreement(s) under which gas is nominated for delivery to the pooling point; and (ii) downstream and at Delivery Points under the Agreement(s) pursuant to which gas is nominated to be received from the pooling point.

- $\hbox{(2)} \quad \hbox{A Shipper under a firm transportation agreement nominating receipt of gas at a pooling point ("Downstream Shipper")}$ will be deemed to have designated the entity nominating the gas to be delivered to the pooling point ("Upstream Shipper") for receipt by the Downstream Shipper as the Downstream Shipper's agent. Under this "Automatic Agency" designation, the nomination by the Upstream Shipper will be given the same priority for scheduling and curtailing the service to the pooling point as though the Downstream Shipper had nominated the same Receipt Point as the Upstream Shipper in lieu of the pooling point. In the event an Upstream Shipper is utilizing more than one Receipt Point and selling gas to two or more ${\tt Downstream}$ Shippers at the pooling point, Natural will request that the Upstream Shipper identify which volumes at each Receipt Point are being sold to which Downstream Shipper when such identification is necessary for scheduling the nominated volume. If such identification is not provided, all of the gas nominated by the Upstream Shipper will be treated as having secondary out-of-path priority. Allocation of capacity at the pooling point among secondary out-of-path volumes shall be based on confirmed nominations. The Automatic Agency procedures do not apply unless the Receipt Point nominated by the Upstream Shipper is in the same zone as the pooling point being utilized or in the event that the Downstream Shipper has no rights within MDQ to utilize the Receipt Point nominated by the Upstream Shipper.
- (3) Gas shall not be confirmed at the pooling point to the extent capacity constraints exist which affect any such receipts or deliveries, applying the priorities set out elsewhere in this Section.
- (b) (1) The zoned storage point approach set out in Section 9 of these General Terms and Conditions is a paper point system predicated upon the system-wide, integrated operation of Natural's storage facilities and not on the storage capacity in any zone. Except as provided to the contrary in this Tariff [and specifically in subsections (2) and (3) below], the storage point in any receipt zone is treated as an out-of-path secondary point under any firm transportation Agreement with point rights in that zone. Given these characteristics, nominations on any day for deliveries

FERC Docket: RP08-319-000

Original Sheet No. 317 Original Sheet No. 317: Pending

GENERAL TERMS AND CONDITIONS

to or receipts from an out-of-path storage point may exceed Natural's ability to accommodate such nominations that day. Natural may reject nominations at any out-of-path storage point on any day to the extent such nominations exceed Natural's capability to deliver gas to or receive gas from storage (as applicable) in relation to that zone, taking into account the operating conditions on Natural's system (including its storage fields).

(2) A storage point under a firm transportation Agreement which has a primary Delivery Point in the Market Delivery Zone will be treated as in-the-path in conjunction with withdrawals where storage rights are separately released or utilized under Rate Schedule FRSS, both injections and withdrawals under Rate Schedules NSS and BESS and for service under Rate Schedule PALS if the following criteria are met. For storage points to be treated as in-the-path for withdrawals under Rate Schedule NSS or separately utilized storage rights under Rate Schedule FRSS, the primary path under the firm transportation Agreement used for withdrawal must extend to the Market Delivery Zone through either Station 103 or 305 from a primary Receipt Point further south and such primary path may not be segmented except in zones upstream of the Midcontinent and Texok Receipt Zones (any Shipper in any release which does not hold the required primary path shall not have in-the-path storage rights). Storage points will be treated as in-the-path on injection under Rate Schedule NSS so long as the primary path under the firm transportation Agreement extends from a primary Receipt Point in a zone other than the Iowa-Illinois Receipt Zone to a primary Delivery Point in the Market Delivery Zone, and such path is not segmented in the Market Delivery Zone. In the case of receipts in the Iowa-Illinois Receipt Zone, such storage points shall be considered as in-the-path so long as the primary path under the firm transportation Agreement extends from Station 109 to a primary Delivery Point in the Market Delivery Zone (including primary paths extending around to points on the Gulf Coast Line), provided that such primary path is not segmented in the Market Delivery Zone (any Shipper in any release which does not hold the required primary path shall not have in-the-path storage rights). Receipts on the Gulf Coast Line north of Station 311 (where no waiver has been posted) shall be considered as in-the-path Amarillo Line receipts under the conditions stated in the prior sentence. Other Gulf Coast Line receipts shall be treated as in-the-path on injection if the Shipper has a primary path from the Receipt Point to a Delivery Point in the Market Delivery Zone and such path is not segmented. A storage point under a firm transportation Agreement which has a primary Delivery Point in any of the production zones (Louisiana, South Texas, Texok,

FERC Docket: RP08-319-000

Original Sheet No. 318 Original Sheet No. 318: Pending

GENERAL TERMS AND CONDITIONS

Midcontinent or Permian) will be treated as in-the-path for transportation of gas withdrawn under Rate Schedule NSS: (i) if the primary path under the transportation Agreement extends through Stations 304 or 103 from a primary Receipt Point which is north of Station 304 or 103 and such primary path is not segmented south of Station 304 or 103; or (ii) if the primary path either includes the entire portion of Natural's Mainline system between Station 304 and the intersection of the A/G Line and the Gulf Coast Mainline or includes a portion of Natural's system which lies between Station 156 and any point on the Amarillo Mainline, and the path is not segmented over that portion of the primary path or between that portion of the primary path and the Delivery Point (any Shipper in any release which does not hold the required primary path shall not have in-the-path storage rights). An Agreement with a primary Delivery Point in a production zone has in-the-path delivery rights in connection with injections under Rate Schedule NSS only to the extent Natural posts such in-the-path rights. Natural shall post on a continuous basis whether and to what extent such in-the-path rights are available and shall provide in the posting at least twenty-four (24) hours' notice before a change in that designation becomes effective. Where a Shipper must have a certain primary path to have inthe-path storage rights, actual receipts may be at any point along that primary path.

(3) The storage points in the Iowa-Illinois Receipt Zone, even though they are not physical points, shall be treated as primary Receipt Points for the purposes of scheduling and curtailing storage withdrawals and the related deliveries under delivered firm storage service pursuant to Rate Schedules DSS and FRSS unless transportation has been separately released. The storage points in the Iowa-Illinois Receipt Zone points shall also be treated as primary points in conjunction with injections pursuant to delivered firm storage service as defined in Rate Schedules DSS and FRSS under a firm transportation Agreement with a primary Delivery Point in the Market Delivery Zone so long as the primary path under the firm transportation Agreement extends from a primary Receipt Point in any zone other than the Iowa-Illinois Receipt Zone to a primary Delivery Point in the Market Delivery Zone, and such path is unsegmented in the Market Delivery Zone (any Shipper in any release which does not hold the required primary path shall not have primary or in-the-path storage rights). In the case of receipts in the Iowa-Illinois Receipt Zone, such storage points shall be treated as primary points for injections under Rate Schedules DSS and FRSS so long as the primary path under the firm transportation Agreement extends from Station 109 to a primary Delivery Point in the Market Delivery Zone (including primary paths extending around to points on the Gulf Coast Line), provided that such primary path is not segmented in the Market Delivery Zone (any Shipper in any release which does not hold the required primary path shall not have primary or in-the-path storage rights). Receipts on the Gulf Coast Line north of Station 311

FERC Docket: RP08-319-000

Original Sheet No. 319 Original Sheet No. 319: Pending

GENERAL TERMS AND CONDITIONS

(where no waiver has been posted) shall be considered primary Amarillo Line receipts under conditions stated in the prior sentence. Other Gulf Coast Line receipts shall be treated as primary on injection if the Shipper has a primary path from the Receipt Point to a Delivery Point in the Market Delivery Zone and such primary path is not segmented in the Market Delivery Zone. Where a Shipper must have a certain primary path to have in-the-path storage rights or injection, actual receipts may be at any point along that primary path.

(4) In the event Natural cannot accept nominations for transportation to or from a storage point or must curtail on any day, Natural shall apply the following priorities when scheduling or curtailing such transportation:

(i) Nominations and no-notice service under firm storage and transportation Agreements within firm injection or withdrawal rights (including the storage rights under Rate Schedules DSS and NSS within MDQ and MSV but in excess of generally applicable firm rights where Natural has given notice that capacity is available) involving a storage point treated as primary or as secondary in-the-path as described in (2) or (3) above will be the first scheduled and the last curtailed. In the event there is sufficient capacity to provide some but not all of the nominated and no-notice service, capacity shall be allocated pro rata based on MDQ under the storage Agreements.

(ii) Nominations and no-notice service under firm transportation Rate Schedules involving a primary or in-the-path secondary storage point for injection or withdrawal under the authorized overrun storage rights within MDQ and MSV properly requested under Rate Schedules NSS, FRSS and DSS will next be scheduled and will be curtailed after category (iii) but before category (i). In the event there is sufficient capacity to provide some but not all of the nominated and interruptible no-notice service, capacity will be allocated pro rata based on MDQ under the storage Agreements.

(iii) Nominations and no-notice service under firm transportation Rate Schedules involving an out-of-path secondary storage point for injection or withdrawal under firm storage rights, including the storage rights under Rate Schedules FRSS and DSS (injection only for DSS and FRSS unless transportation has been separately released, since DSS and FRSS storage points on withdrawal are always treated as primary points) and NSS within MDQ and MSV but in excess of generally applicable firm rights where Natural has given notice that capacity is available, will be the next scheduled and will be curtailed after category (iv) but before category (ii). In the event there is sufficient capacity to provide some but not all of the nominated and no-notice service, capacity will be allocated pro rata based on MDQ under the storage Agreements.

FERC Docket: RP08-319-000

Original Sheet No. 320 Original Sheet No. 320: Pending

GENERAL TERMS AND CONDITIONS

(iv) Nominations and no-notice service under firm transportation Rate Schedules involving an out-of-path secondary storage point for injection or withdrawal under the authorized overrun storage rights within MDQ, MSV and MAC properly requested under Rate Schedules NSS, FRSS and DSS (injection only for DSS and FRSS unless transportation has been separately released, since DSS and FRSS storage points on withdrawal are always treated as primary points) will next be scheduled and will be curtailed after category (v) but before category (iii). In the event there is sufficient capacity to provide some but not all of the nominated and interruptible no-notice service, capacity will be allocated pro rata based on MDQ under the storage Agreements.

(v) Nominations under firm transportation Rate Schedules involving a primary or in-the-path secondary point for injection or withdrawal under a BESS or PALS Agreement and under authorized overrun properly nominated pursuant to Rate Schedules DSS and NSS in excess of MDQ, MSV and/or MAC and authorized overrun properly nominated under Rate Schedule FRSS in excess of MDQ, IQ, WQ or WSWQ will be scheduled next and will be curtailed after category (iv) but before category (vi). In the event there is sufficient capacity to provide some but not all of the nominated service, capacity will be allocated based on the priority of the storage rights as set out in Section 5.7.

(vi) Nominations under firm transportation Rate Schedules involving an out-of-path secondary point for injection or withdrawal under a BESS or PALS Agreement and under authorized overrun properly nominated pursuant to Rate Schedules DSS and NSS in excess of MDQ, MSV and/or MAC and authorized overrun properly nominated under Rate Schedule FRSS in excess of MDQ, IQ, WQ or WSWQ will be scheduled next and will be curtailed after category (vii) and prior to category (v). In the event there is sufficient capacity to provide some but not all of the nominated service, capacity will be allocated based on the priority of the storage rights as set forth in Section 5.7.

(vii) Nominations for interruptible transportation service (including authorized overrun service under firm Rate Schedules) for injection or withdrawal under firm storage rights (including storage rights under Rate Schedules DSS and NSS within MDQ and MSV but in excess of generally applicable firm rights where Natural has given notice that capacity is available) will be scheduled next and curtailed after category (viii) and prior to category (vi). In the event there is sufficient capacity to provide some but not all of the nominated service, capacity will be allocated based on the priority of the transportation Agreement as set forth in Section 5.7.

FERC Docket: RP08-319-000

Original Sheet No. 321 Original Sheet No. 321: Pending

GENERAL TERMS AND CONDITIONS

(viii) Nominations for interruptible transportation service (including authorized overrun service under firm Rate Schedules) for injection or withdrawal under authorized overrun storage rights within MDQ and MSV under Rate Schedules DSS and NSS will be scheduled next and will be curtailed after category (ix) and before category (vii). In the event there is sufficient capacity to provide some but not all of the nominated service, capacity will be allocated based on the priority of the transportation Agreement as set forth in Section 5.7.

(ix) Nominations for interruptible transportation service (including authorized overrun service in excess of MDQ under firm Rate Schedules) for injection or withdrawal under BESS or PALS Agreements and under authorized overrun properly requested under Rate Schedules DSS and NSS in excess of MDQ, MSV and/or MAC and authorized overrun properly requested under Rate Schedule FRSS in excess of MDQ, IQ, WQ or WSWQ will be scheduled last and will be curtailed first. If there is sufficient capacity to provide some but not all of the nominated service, capacity will be allocated based on the priority of the transportation Agreement as set forth in Section 5.7.

Notwithstanding the foregoing, Natural shall exclude any Shipper from scheduling constraints or curtailment to the extent that deliveries/injections or receipts/withdrawals, as applicable, under the applicable transportation or storage Agreement either alleviate a constraint or do not create a constraint in relation to the storage point in light of the specific Receipt or Delivery Point for such gas and the capacity available to Shipper under its Agreement considered in relation to operating conditions that day on Natural's system.

- (c) In applying the sourcing requirements under Rate Schedules DSS and NSS, supplies shall be allocated for sourcing purposes as follows:
- (1) Gas delivered to storage under confirmed nominations in conjunction with an NSS, DSS or FRSS Agreement shall be deemed injected based on the location of the supply source. All supply sources received on the Amarillo Line and north of Station 311 on the Gulf Coast Line are deemed Amarillo receipts and all other receipts are deemed Gulf Coast receipts, the Amarillo and Gulf Coast Lines being otherwise defined as in Section 9.2 of these General Terms and Conditions. Notwithstanding the foregoing, Natural shall post prior to the beginning of the month, at least

FERC Docket: RP08-319-000

Original Sheet No. 322 Original Sheet No. 322: Pending

GENERAL TERMS AND CONDITIONS

twenty-four (24) hours before first-of-the-month nominations are due, the extent (if any) to which receipts between Station 110 and Station 311 will be treated as Gulf Coast receipts. In addition, Natural will review on a daily basis the extent (if any) to which receipts between Station 110 and Station 311 on the next day will be treated as Gulf Coast receipts. Natural will designate such receipts as Gulf Coast receipts to the extent feasible operationally. For daily changes, Natural will post on the Informational Postings portion of its Interactive Website the volumes which can be treated as Gulf Coast receipts no later than the time nominations are due for such day. Such designations may be changed only prospectively through issuance of an Operational Flow Order or in subsequent postings. Except through an Operational Flow Order, the volumes designated as Gulf Coast receipts in the monthly posting may not be reduced by the daily posting during the month.

 $\,$ (2) Gas withdrawn from storage under confirmed nominations pursuant to Rate Schedule NSS shall be allocated based on the location of the storage point from which gas is withdrawn.

(3) All gas withdrawn from storage under Rate Schedule DSS (whether nominated or no-notice) shall be allocated fifty percent (50%) to Amarillo and fifty percent (50%) to Gulf Coast sources without regard to the actual location of injection or withdrawal. Natural shall effectuate such service utilizing the overall capability of its system, including its integrated storage. Notwithstanding the foregoing, the Shipper shall always have the fifty percent (50%) withdrawal rights on one leg up to its remaining inventory on that leg even if the inventory available for withdrawal on the other leg is insufficient to support the remaining fifty percent (50%) of withdrawal rights. For example, if the Shipper's inventory on the Amarillo leg were depleted, the Shipper could still withdraw a volume up to fifty percent (50%) of its withdrawal rights on the Gulf Coast leg on either a nominated or no-notice basis up to its remaining inventory on the Gulf Coast leg.

(4) Gas withdrawn from storage on a no-notice basis shall be deemed withdrawn from the storage point associated with the Delivery Point.

(5) Gas received into or withdrawn from Rate Schedule NSS on a no-notice basis pursuant to the NB option under Rate Schedules FTS, FFTS and FTS-G and gas injected into Rate Schedule DSS on a no-notice basis shall be allocated to leg for sourcing purposes in accordance with this subsection. No-notice injections under Rate

FERC Docket: RP08-319-000

Original Sheet No. 323 Original Sheet No. 323: Pending

GENERAL TERMS AND CONDITIONS

Schedules DSS and NSS shall be allocated between legs as follows: Nonotice injections under Rate Schedule DSS shall initially be allocated fifty percent (50%) to the Gulf Coast Line and fifty percent (50%) to the Amarillo Line and no-notice injections under Rate Schedule NSS shall initially be allocated to the Gulf Coast Line in the Gulf Coast Percentage and to the Amarillo Line in the Amarillo Percentage where the applicable Delivery Point is in the Iowa-Illinois Receipt Zone (Gulf Coast Percentage and Amarillo Percentage are defined in Section 5.3(b)(3) of Rate Schedule NSS); provided, however, that if Natural has posted a waiver of seasonal inventory requirements by leg, the allocation shall reflect such waiver. In the event that the volume so allocated to the Gulf Coast Line for no-notice injections for a Shipper exceeds the volume of gas nominated and flowing from Gulf Coast Line sources for that Shipper on that day, the volume in excess of the nominated and flowing gas shall be reallocated for that day to the Amarillo Line. For no-notice injections under Rate Schedule NSS where the applicable Delivery Point is outside the Iowa-Illinois Receipt Zone, the injections shall be allocated to the storage point where the Delivery Point is located. No-notice withdrawals under Rate Schedule NSS shall be allocated between legs for deliveries to the Market Delivery Zone for any month based on Shipper's inventory as posted by Natural on its Interactive Website twenty-four (24) hours before firstof-the-month nominations are due, with withdrawals coming from the storage point closest to the Delivery Point on each leg, as more fully set out at Section 5.3(e) of Rate Schedules FTS and FFTS. For deliveries to other zones, the gas is deemed to come from that zone. Such allocation could result in Sourcing, Overrun or Balancing Service Charges if Shipper's or point operator's firm transportation or storage rights on either leg are exceeded. No-notice service for injections and/or withdrawals is only available to the extent of firm rights with a primary or in-the-path point as defined in Section 5.6(b)(2) above for injection or withdrawals, as applicable. No-notice rights are subject to the allocation and curtailment provisions set out in the remainder of this Section 5.

(6) (i) Commencing April 1, 1996, any Shipper may trade with any other Shipper having a corresponding Agreement its injection rights under Rate Schedules DSS or NSS and/or its withdrawal rights under Rate Schedule NSS and the corresponding inventory rights. Such trades must be for equal volumes of injection rights or withdrawal rights and the related inventory rights, as set out below.

FERC Docket: RP08-319-000

Original Sheet No. 324 Original Sheet No. 324: Pending

GENERAL TERMS AND CONDITIONS

(ii) Trades must commence on the first day of a month and continue for a minimum period of one (1) calendar month. Trades of rights under Rate Schedule NSS may continue for a period exceeding one (1) year, but any trade under Rate Schedule DSS must conclude no later than the end of the injection season during which the trade is in effect.

(iii) Trades must include both daily and inventory rights and the relative volumes of such rights must be based on the definition of such rights in the applicable tariff provision. Daily rights and inventory rights must revert to the original Shipper simultaneously at the end of the trade.

Once Shippers have agreed upon a conforming (iv) trade, such Shippers shall notify Natural of the trade in writing or, if available, on Natural's Interactive Website. Such notices must specify the volumes of daily injection and/or withdrawal rights involved, the total inventory and/or withdrawal rights involved, the total inventory rights being traded, whether such rights are Amarillo or Gulf Coast, the dates on which the trade is to commence and terminate and the contracts affected. Such notice must also reflect the consent of all Shippers involved in the trade. Any Shipper seeking to participate in a trade may submit an advertisement pursuant to Section 20 of these General Terms and Conditions. For a qualifying trade to be effective for any month, conforming notification of the trade must be received by Natural at least one (1) business day prior to the date nominations are due for the first of the month. Once the trade commences, the rights of the parties to injections and/or $% \left(1\right) =\left(1\right) \left(1\right) \left($ withdrawals and to inventory under their respective Agreements shall be adjusted accordingly. With respect to any trade, each party which is participating in the trade must conform its inventory to the applicable requirements (including cycling requirements) once the trade has terminated.

FERC Docket: RP08-319-000

Original Sheet No. 325 Original Sheet No. 325 : Pending

GENERAL TERMS AND CONDITIONS

5.7 INTERRUPTIBLE AND OVERRUN SERVICES

- (a) This Section governs the priority for scheduling and curtailment of all interruptible and overrun services, including authorized overrun services under firm agreements, other than: (1) secondary point services within MDQ under firm Agreements and (2) the authorized overrun injections and withdrawals within MDQ and MSV pursuant to confirmed nominations under DSS and NSS Agreements. As used in this Section, the term "interruptible" includes all authorized overrun services under both firm and interruptible Agreements, other than those excluded in the prior sentence.
- (b) Natural's interruptible transportation and storage services shall be provided to the extent capacity is available after scheduling services with higher priority as provided in Section 5.3. Natural may decline to schedule and/or may curtail interruptible service for any of the following reasons:
- (i) if Shipper tenders gas which does not conform to the applicable pressure or quality requirements of these General Terms and Conditions;
 - (ii) for reason of Force Majeure;
- $\mbox{(iii)}\mbox{ }\mbox{due}$ to routine repair and maintenance to be reasonably determined by Natural;
- $\hbox{(iv)} \qquad \hbox{pursuant to Section 5.11 of these General} \\ \hbox{Terms and Conditions;}$
- $% \left(v\right) =0$ to rectify imbalances or to conform physical flows to nominations to the extent consistent with the specific Rate Schedule;
- (vi) pursuant to the Operational Control provisions of these General Terms and Conditions;
 - (vii) to maintain system integrity;
- $% \left(0.011\right) =0.011$ (viii) if there is a dispute over title, ownership or right to tender or receive gas; or
- $\mbox{\ensuremath{\mbox{(ix)}}}$ if capacity is required to provide a service with higher priority.

FERC Docket: RP08-319-000

Original Sheet No. 326 original Sheet No. 326: Pending

GENERAL TERMS AND CONDITIONS

(c) (1) SCHEDULING OF INTERRUPTIBLE SERVICES

If capacity remains after all gas properly nominated and confirmed (to the extent nomination is required) under services described in Sections 5.3(a)(1) and (2) and Sections 5.3(b)(1) and (2) has been scheduled, Natural shall schedule the remaining interruptible services (including authorized overrun). To the extent there is insufficient capacity available to schedule all properly nominated and confirmed interruptible services (including authorized overrun), Natural shall identify which nominated services are affected by the lack of capacity and shall schedule all affected interruptible services in accordance with the following sequence:

(i) Natural shall schedule interruptible services (including authorized overrun) based on the rate to be paid, from highest to lowest daily rate, with service for which the highest daily rate being paid is scheduled first. Any Shipper paying the maximum rate applicable to its service (or revenue equal to or greater than the applicable maximum rate pursuant to a Negotiated Rate or Negotiated Rate Formula) shall be afforded highest priority even if a Shipper which has agreed to a Negotiated Rate or Negotiated Rate Formula is paying a higher unit rate.

(ii) In the event there is insufficient capacity to schedule all services for which the same rate is to be paid, Natural shall allocate the available capacity pro rata based on the confirmed nominated volume.

(2) CURTAILMENT OF INTERRUPTIBLE SERVICES

In interrupting or curtailing interruptible services (including authorized overrun) covered by this Section 5.7 due to lack of capacity, Natural shall curtail in accordance with the following priority sequence:

(i) Natural shall interrupt services (including authorized overrun) based on the rate being paid, from lowest to highest daily rate, with service for which the lowest daily rate being paid is interrupted first. Any Shipper paying the maximum rate applicable to its service (or revenue equal to or greater than the applicable maximum rate pursuant to a Negotiated Rate or Negotiated Rate Formula) shall be afforded highest priority even if a Shipper which has agreed to a Negotiated Rate or Negotiated Rate Formula is paying a higher unit rate.

FERC Docket: RP08-319-000

Original Sheet No. 327 Original Sheet No. 327: Pending

GENERAL TERMS AND CONDITIONS

(ii) If there is capacity sufficient to provide some but not all services, Natural shall interrupt such services from the lowest daily rate to the highest daily rate being paid.

- (3) SCHEDULING AND CURTAILMENT OF SERVICE UNDER RATE SCHEDULES IBS AND LPS
- (i) Service under Rate Schedule IBS shall be scheduled and curtailed based on Natural's line pack capability and offsetting Access Requests on the relevant portions of its system. Service shall not be provided under Rate Schedule IBS, service pursuant to a confirmed Access Request under Rate Schedule IBS shall be interrupted, if such service would be detrimental to Natural's ability to provide firm services or to provide any priority interruptible rights as identified in Section 5.7(a) above. Natural shall also interrupt or decline to schedule service under Rate Schedule IBS before taking any actions under the Operational Control Sequence set out in Section 23.4 of these General Terms and Conditions to the extent that the service under Rate Schedule IBS would have any effect adverse to Natural's system in relation to the conditions which give rise to such operational control measures.

(ii) If providing service under any IBS Agreement would have an adverse affect on providing any other interruptible service, the IBS Agreement shall, for scheduling and balancing purposes, be assigned the priority of the transportation Agreement to which the IBS Agreement is linked, with overrun volumes being assigned priority on the same basis as overrun volumes under the linked transportation Agreement.

(iii) In the event Natural can provide some service under Rate Schedule IBS on any portion of its system predicated on the level of line pack, but the line pack capability on any day is insufficient to schedule all Access Requests under Rate Schedule IBS, first priority for scheduling and curtailment purposes shall be granted to Access Requests which are linked to a firm transportation Agreement (if allocation within this class is required, allocation shall be pro rata based on MDQ). If sufficient capability is available to serve some but not all Access Requests under IBS Agreements linked to interruptible transportation Agreements, service under Rate Schedule IBS shall be allocated based on the price paid under the linked interruptible transportation Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 328 Original Sheet No. 328: Pending

GENERAL TERMS AND CONDITIONS

(iv) Service under Rate Schedule LPS shall not be scheduled if doing so would have an adverse effect on any firm service, or on any priority interruptible service associated with firm service identified in Section 5.7(a) of these General Terms and Conditions or in Sections 5.3(c)(1) and 5.5 of Rate Schedule DSS or in Sections 5.3(b)(1) and 5.5 of Rate Schedule NSS, would increase the likelihood of Natural having to issue an Operational Flow Order or declare a Critical Time or would adversely affect storage. Service under Rate Schedule LPS shall be scheduled and curtailed based on Natural's system operational and line pack capability, and shall not adversely affect storage. Service hereunder shall be interrupted and curtailed if continuation of such service would be detrimental to Natural's ability to provide any firm service or any priority interruptible service associated with firm service as identified in Section 5.7(a) of these General Terms and Conditions or if continuation of such service would adversely affect storage. While service hereunder is not ordinarily expected to affect transportation capacity, if providing service hereunder would have an adverse affect on providing interruptible transportation services, an LPS Agreement shall for scheduling and allocation purposes be assigned the same priority as service under Rate Schedule ITS. Any required rate comparison will be based on the level of the Initial Fee, except that upon reversal or completion of the transaction, any rate comparison shall be based on the level of the Completion Fee. The same priority shall be applied for scheduling and allocation in relation to service under Rate Schedule IBS or other LPS Agreements. Once a park or loan has been scheduled on any day under Rate Schedule LPS, that park or loan shall not be interrupted or curtailed to effectuate any other interruptible service, except for priority interruptible service under a firm Agreement as identified in Section 5.7(a) of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 329 Original Sheet No. 329: Pending

GENERAL TERMS AND CONDITIONS

(4) Any Shipper paying a discounted rate may give Natural prior written notice at any time, and from time to time, that Shipper elects to pay a specified higher rate, capped by the applicable maximum rate, during the period the Shipper would otherwise be subject to the curtailment or scheduling restriction. Such election may be rescinded at any time by Shipper again providing Natural with prior written notice. Such prior written notices should be given no later than 8:00 a.m. Central Time of the business day prior to the date of their effectiveness.

(d) An Agreement under Rate Schedule ITS will include all Receipt and all Delivery Points available on Natural's system. Notwithstanding the foregoing, a Shipper may not utilize a point for which there is no regulatory authorization to receive or deliver gas under the Agreement.

5.8 CAPACITY CONSTRAINTS

If Natural experiences a capacity constraint on a portion of its system or at specific points, it shall (to the extent practicable), apply the scheduling and curtailment provisions hereof, for both firm and interruptible services, only to those Shippers with service affected by that portion of the system or at those points. Natural shall endeavor to restrict scheduling limitations and/or curtailment to as limited a geographical area, number of Shippers and services as reasonably feasible, given the operational capabilities of its system.

5.9 UNAUTHORIZED OVERRUN GAS

No Shipper shall have any right to tender unauthorized overrun gas. Unauthorized overruns are subject to the charges set out in Section 12 of these General Terms and Conditions.

5.10 OTHER TRANSPORTERS

Natural's application of the priorities hereunder shall be subject to the actions of other transporters delivering or receiving gas on behalf of Shippers.

FERC Docket: RP08-319-000

Original Sheet No. 330 Original Sheet No. 330: Pending

GENERAL TERMS AND CONDITIONS

5.11 DELINOUENCY IN PAYMENT

- (a) Irrespective of any otherwise applicable priority, Natural may suspend service to any Shipper which is delinquent in payments under any Agreement, subject to the following conditions:
- (1) Natural shall give Shipper written notice of the delinquency and of Natural's intent to suspend service if the deficiency is not cured. If the delinquency is not remedied within fifteen (15) days of such notice, Natural may suspend service. Natural shall simultaneously provide written notice to the Commission of any such suspension.
- (2) If a Shipper which has been deficient in payment hereunder is again deficient in payment within six (6) months after the prior deficiency, then Natural may suspend service to such Shipper within five (5) Business Days after providing notice hereunder unless Shipper remedies the deficiency within that time period. A Shipper shall not be obligated to pay any reservation or demand charges for suspended service attributable to the period when that service is suspended.
- (b) In addition to suspension, Natural may terminate service if the Shipper fails to remedy a delinquency in payment. Any such termination requires thirty (30) days' prior notice to Shipper and to the Commission. To avoid termination, the Shipper must remedy the deficiency within this notice period. Such notice may be given simultaneously with the initial notice under Section 5.11(a)(1) of these General Terms and Conditions or with the notice under Section 5.11(a)(2) of these General Terms and Conditions.
- (c) In the event of a good faith billing dispute, withholding of payment for the amount in dispute by Shipper shall not be considered a delinquency in payment, consistent with Section 15 of these General Terms and Conditions, and will not result in suspension or termination of service.
- (d) Natural may not take any action under this Section 5.11 which conflicts with any order of the U.S. Bankruptcy Court.

5.12 DETERIORATION OF CREDIT

(a) (1) If at any time Natural reasonably determines based on adequate information available to it that a Shipper is not creditworthy under Section 16(a) of these General Terms and Conditions or if Shipper fails to maintain assurance of future performance under Section 16(b) of

FERC Docket: RP08-319-000

Original Sheet No. 331 original Sheet No. 331 : Pending

GENERAL TERMS AND CONDITIONS

these General Terms and Conditions, Natural may notify such Shipper in writing that it has five (5) Business Days to provide Natural with security consistent with Section 16(b) of these General Terms and Conditions which is adequate to cover all charges for one month's advance service. In addition, within thirty (30) days after such notification, the Shipper must fully comply with the means for adequate assurance of future performance, covering three (3) full months of advance service from the end of such 30-day notice period, as provided under Section 16(b) of these General Terms and Conditions. If the Shipper has not satisfied the requirements in either of prior two (2) sentences by the end of the specified prior notice period, Natural may immediately suspend service to Shipper. Natural may terminate service if it has complied with the procedures in Section 5.12(d) of these General Terms and Conditions.

- (2) If Natural does not have sufficient information to determine whether Shipper is creditworthy, it may request additional information in writing from the Shipper consistent with Section 16(a) of these General Terms and Conditions, and Shipper must provide such information within five (5) Business Days.
- (i) If Natural requests additional information to be used for credit evaluation after the initiation of service, Natural, contemporaneous with the request, shall provide its reason(s) for requesting the additional information to the Shipper and designate to whom the response should be sent; provided that Natural and the Shipper may mutually agree to waive this requirement.
- (ii) Upon receipt of a request from Natural for information to be used for creditworthiness evaluation, the Shipper's authorized representative(s) shall acknowledge receipt of Natural's request; provided, however, that Natural and the Shipper may mutually agree to waive this requirement.
- (iii) The Shipper's authorized representative(s) shall respond to Natural's request for credit information, as allowed by this Tariff, on or before the due date specified in the request, which due date must be consistent with the requirement of this Section 5.12(a)(2). The Shipper should provide all the credit information requested by Natural or provide the reason(s) why any of the requested information was not provided.
- (iv) Upon receipt from the Shipper of all credit information provided pursuant hereunder, Natural shall notify the Shipper's authorized representative(s) that it has received such information; provided, however, that Natural and the Shipper may mutually agree to waive this requirement.

FERC Docket: RP08-319-000

Original Sheet No. 332 Original Sheet No. 332: Pending

GENERAL TERMS AND CONDITIONS

(v) Representatives and notices regarding Shipper creditworthiness shall be as provided in Section 16(a) of these General Terms and Conditions.

- (vi) If Shipper fails to provide the requested information or if Natural determines that the Shipper is not creditworthy based on such information, Section 5.12(a)(1) of these General Terms and Conditions shall apply for suspension of service and Section 5.12(d) of these General Terms and Conditions shall apply for termination of service.
- (b) Any suspension of service hereunder shall continue until Natural is reasonably satisfied that Shipper is creditworthy under Section 16(a) of these General Terms and Conditions or until Shipper has provided adequate assurance of future performance under Section 16(b) of these General Terms and Conditions. A Shipper shall not be obligated to pay any reservation or demand charges for suspended service attributable to the period when that service is suspended.
- At any time after a Shipper is determined to be noncreditworthy by Natural, the Shipper may initiate a creditworthiness reevaluation by Natural. As part of the Shipper's re-evaluation request, the Shipper must either update or confirm in writing the prior information provided to Natural related to the Shipper's creditworthiness. Such update should include any event(s) that the Shipper believes could lead to a material change in the Shipper's creditworthiness. Such reevaluation shall be performed consistent with Section 16(a) of these General Terms and Conditions. After Natural's receipt of such a request for credit reevaluation including all required information under this Tariff for evaluation of credit, Natural shall provide a written response to Shipper within five (5) Business Days. Such written response should include either a determination of creditworthiness status, clearly stating the reason(s) for Natural's decision, or an explanation supporting a future date by which a reevaluation determination will be made. In no event should such reevaluation determination exceed twenty (20) Business Days from the date of the receipt of Shipper's Request unless specified in this tariff or if the parties mutually agree to some later date. If Natural determines that Shipper is now creditworthy, any security requirement under Section 16(b) of these General Terms and Conditions shall be terminated and any prepayment amounts (including any applicable interest) released to Shipper from escrow within five (5) Business Days after such determination.
- (d) In addition to suspension, Natural may terminate service if the Shipper fails to provide adequate assurance of future performance consistent with Section 16(b) of these General Terms and Conditions. Any such termination requires thirty (30) days' prior notice to Shipper and to

FERC Docket: RP08-319-000

Original Sheet No. 333 Original Sheet No. 333: Pending

GENERAL TERMS AND CONDITIONS

the Commission. To avoid termination, the Shipper must satisfy Section 16(b) of these General Terms and Conditions within this notice period. Such notice may be given simultaneously with the notice provided for in Section 5.12(a)(1) of these General Terms and Conditions.

- (e) In addition to any prior notice provided for above, Natural shall simultaneously notify the Commission in writing of any suspension or termination of service under this Section 5.12.
- (f) Natural may not take any action under this Section 5.12 which conflicts with any order of the U.S. Bankruptcy Court.

5.13 EMERGENCY CURTAILMENT RELIEF

In the event that Natural is advised by a Shipper under a firm DSS, FTS, FFTS, or FTS-G Agreement that delivery of gas is necessary to avoid irreparable injury to life or property (including environmental emergencies) or to provide for minimum plant protection, Natural may adjust the volumes scheduled or curtailed (including diverting gas from other Shippers as necessary) to deliver the volumes required within such Shipper's MDQ to avoid or mitigate the emergency situation to the extent Natural can do so without creating an emergency situation for the Shippers with reduced volumes; provided, however, that no relief shall be available hereunder to the extent a Shipper has been relying on interruptible gas or transportation to serve the use for which relief is being requested. To qualify for such emergency relief, Shipper must certify within twenty-four (24) hours that it has utilized all other sources of gas supply, that it is providing no interruptible services and that it or its customers could not use alternate fuel. If emergency relief is sought by more than one Shipper, any volumes which Natural can provide shall be prorated, if necessary, based on the sum of MDQs under all of the firm Agreements listed above for each of the requesting Shippers. The Shipper receiving relief volumes shall pay for any volumes it receives in excess of the volumes otherwise available under Natural's scheduling and curtailment procedures a unit rate equal to the highest charge for failure to comply with an Operational Flow Order during a Critical Time as specified in Section 23. If Natural subsequently determines that Shipper was not eligible for the emergency relief received, the charge shall be twice that specified in the prior sentence. Except when emergency curtailment relief is provided during a Critical Time (in which event the disposition of the revenues shall be in accordance with Section 23), any revenues received by Natural from such Shipper in excess of the sum of otherwise applicable charges and any out-of-pocket expenses incurred by Natural as a result of providing relief shall be distributed among all Shippers which received reduced volumes due to such emergency relief in proportion to their volumetric reduction.

FERC Docket: RP08-319-000

Original Sheet No. 334 Original Sheet No. 334: Pending

GENERAL TERMS AND CONDITIONS

5.14 OPERATIONAL CONTROL

The priorities of service set out in this Section 5 shall be subject to the Operational Control provisions of Section 23 of these General Terms and Conditions, which shall override in the event of a conflict for the purpose of determining gas receipts and deliveries on any day, but not with respect to determining a Shipper's capacity entitlement or entitlement to take gas to be used in evaluating whether another Shipper has made unauthorized use of such capacity or gas.

FERC Docket: RP08-319-000

Original Sheet No. 335 Original Sheet No. 335: Pending

GENERAL TERMS AND CONDITIONS

6. RECEIPT POINTS

6.1 FACILITIES AT RECEIPT POINTS

- (a) Unless otherwise agreed, Natural shall own, operate and maintain all pipeline and measurement facilities necessary to receive and measure gas hereunder. Shipper or the interconnecting party shall (in addition to all other applicable charges) reimburse Natural for the actual cost (including income taxes associated with a contribution-in-aid of construction) of any and all facilities installed by Natural pursuant to this Section at Shipper's or the interconnecting party's request in order to provide service for such Shipper or interconnecting party including, but not limited to, the cost of all labor, materials and rights-of-way; provided that Shipper or the interconnecting party shall repay Natural in kind for any gas lost from Natural's pipeline as a result of the installation of such facilities. Natural may submit billings to Shipper or the interconnecting party up to sixty (60) days in advance for the estimated cost of construction to be incurred by Natural. Shipper or the interconnecting party shall make payments within ten (10) days of the date of receipt of any billings submitted by Natural pursuant to this Section. For purposes of this Section, the bill is deemed to be received by Shipper or the interconnecting party three (3) days after the postmark date. Late payments shall be subject to Section 15 of these General Terms and Conditions. Any such estimated billings shall be reconciled to the actual costs of construction, and any payments to reflect such reconciliation shall be made, within a time period and on terms agreed to by the parties. Neither the amounts collected hereunder nor the cost of such facilities shall be recognized in establishing Natural's general system rates.
- (b) Natural may elect, on a nondiscriminatory basis, to pay all or a portion of the costs of the facilities constructed pursuant to subsection (a) above if Natural determines that the construction of such facilities is economically beneficial to Natural. For purposes of determining whether a project is beneficial, Natural will evaluate projects on the basis of various economic criteria, which will include the estimated transportation throughput, cost of the facilities, operating and maintenance as well as administrative and general expenses attributable to the facilities, the revenues Natural estimates will be generated as a result of such construction, and the availability of capital funds on terms and conditions acceptable to Natural. In estimating the revenues to be generated, Natural will base those revenues upon transportation rates it expects to be able to charge, exclusive of any surcharges such as ACA, and the projected incremental volumes which will result from the project.

FERC Docket: RP08-319-000

Original Sheet No. 336 Original Sheet No. 336: Pending

GENERAL TERMS AND CONDITIONS

6.2 OBLIGATION

Natural's maximum obligation to receive gas at the Receipt Point(s) under the Agreement shall never exceed the lesser of: (1) the applicable MDQ under the Agreement in the aggregate or at individual points, as specified in the Agreement or as applicable at such point under this Tariff; or (2) the total daily volume Shipper or its designee is able and willing to tender at the Receipt Point(s).

6.3 LOCATION

Unless otherwise described in the Agreement, the Receipt Point(s) for transportation Agreements shall be located at the interconnection between the facilities of Shipper, or its designee, and the facilities of Natural. For storage Agreements, the Receipt Point(s) shall be the storage points defined in these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 337 Original Sheet No. 337: Pending

GENERAL TERMS AND CONDITIONS

7. DELIVERY OF GAS FOR THE ACCOUNT OF SHIPPER

7.1 DELIVERY VOLUMES

Commencing on the date of first acceptance by Natural of natural gas delivered by or on behalf of Shipper at the Receipt Point(s) pursuant to an Agreement, and continuing thereafter during the term of that Agreement, Natural shall deliver Equivalent Volumes, or cause Equivalent Volumes to be delivered to Shipper, or to a mutually agreeable third party for Shipper's account, at the Delivery Point(s) described in the Agreement or applicable to the Agreement under this Tariff. In determining Equivalent Volumes, Natural shall retain gas in kind for Fuel Gas and Gas Lost and Unaccounted For on its system based on the percentage Rates set out in the Currently Effective Rates section of this Tariff and using the following formula: [1 - (fuel percent/100)] multiplied by the receipt quantity (rounded to the nearest Dth) equals the delivery quantity. Upon the mutual agreement of Natural and Shipper, in lieu of Natural retaining gas in kind, Shipper shall reimburse Natural for Fuel Gas and Gas Lost and Unaccounted For on its system. Such sales shall be provided on a non-discriminatory basis. If Natural intends to make such sales in any month, Natural shall post on the ${\tt Informational\ Postings\ portion\ of\ its\ Interactive\ Website\ at\ least}$ five (5) days prior to the beginning of the month the following information:

- (a) The volume available that month. If the volumes requested exceed the volumes available for such sales, Natural shall allocate the available volumes based on the MDQ of Shippers requesting such sales.
- (b) The point(s) of sale, which point(s) must be at a Receipt Point on the purchasing Shippers' Agreements.
 - (c) The price.

Shipper shall be directly responsible for any Fuel Gas and Gas Lost and Unaccounted For on any upstream pipeline from which Natural receives Shipper's gas.

FERC Docket: RP08-319-000

Original Sheet No. 338 Original Sheet No. 338: Pending

GENERAL TERMS AND CONDITIONS

7.2 DELIVERY FACILITIES

(a) Unless otherwise agreed, Natural shall own, operate and maintain all pipeline and measurement facilities necessary to deliver and measure gas hereunder. Shipper or the interconnecting party shall (in addition to all other applicable charges) reimburse Natural for the actual cost (including income taxes associated with a contribution-in-aid of construction) of any and all facilities installed by Natural pursuant to this Section at Shipper's or the interconnecting party's request in order to provide service for such Shipper or interconnecting party including, but not limited to, the cost of all labor, materials and rights-of-way; provided that Shipper or the interconnecting party shall repay Natural in kind for any gas lost from Natural's pipeline as a result of the installation of such facilities. Natural may submit billings to Shipper or the interconnecting party up to sixty (60) days in advance for the estimated cost of construction to be incurred by Natural. Shipper or the interconnecting party shall make payments within ten (10) days of the date of receipt of any billings submitted by Natural pursuant to this Section. For purposes of this Section, the bill is deemed to be received by Shipper or the interconnecting party three (3) days after the postmark date. Late payments shall be subject to the provisions contained in Section 15 of these General Terms and Conditions. Any such estimated billings shall be reconciled to the actual costs of construction, and any payments to reflect such reconciliation shall be made within a time period and on terms agreed to by the parties. Neither the amounts collected hereunder nor the cost of such facilities shall be recognized in establishing Natural's general system rates.

(b) Natural may elect, on a nondiscriminatory basis, to pay all or a portion of the costs of the facilities constructed pursuant to subsection (a) above if Natural determines that the construction of such facilities is economically beneficial to Natural. For purposes of determining whether a project is beneficial, Natural will evaluate projects on the basis of various economic criteria, which will include the estimated transportation throughput, cost of the facilities, operating and maintenance as well as administrative and general expenses attributable to the facilities, the revenues Natural estimates will be generated as a result of such construction, and the availability of capital funds on terms and conditions acceptable to Natural. In estimating the revenues to be generated, Natural will base those revenues upon transportation rates it expects to be able to charge, exclusive of any surcharges such as ACA, and the projected incremental volumes which will result from the project.

FERC Docket: RP08-319-000

Original Sheet No. 339 Original Sheet No. 339: Pending

GENERAL TERMS AND CONDITIONS

7.3 OBLIGATIONS

Natural's maximum obligation to deliver gas at the Delivery Point(s) under an Agreement shall never exceed the lesser of: (1) the applicable MDQ under the Agreement in the aggregate or at each point as specified in the Agreement or as applicable to any point under this Tariff; or (2) the total daily volume Shipper or its designee is willing and able to receive at the Delivery Point(s).

7.4 HOURLY LIMITATIONS

Natural shall not be obligated, during any hour, to deliver a total volume of gas at any Delivery Point in excess of six and twenty-five hundredths percent (6.25%) of Shipper's aggregate MDQs for firm service at such Delivery Point, and Natural may impose such restriction whenever, in Natural's sole judgment, such restriction is necessary in order to meet its delivery obligations. However, in the event that Shipper's load pattern temporarily requires deliveries at any Delivery Point in excess of such restriction, Natural will, on request of Shipper, use its best efforts to accommodate such load pattern, and will consult with Shipper in an effort to generally accommodate load variation to the maximum extent feasible and compatible with the needs of Shippers as a group. If, after written request by Natural, Shipper fails to restrict its hourly takes as specified herein, Natural may install and operate a load limiting device at any Delivery Point where such failure has occurred. The cost for such device, including the cost of installation, shall be paid by Shipper within 30 days of its receipt of the bill therefor.

7.5 LOCATION

Unless otherwise described in an Agreement, the Delivery Point(s) for transportation Agreements shall be located at the interconnection between the facilities of Shipper or its designee, and the facilities of Natural. Delivery Points for storage Agreements shall be the storage points defined in these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 340 original Sheet No. 340: Pending

GENERAL TERMS AND CONDITIONS

8. NOMINATION/REPORTING AND BALANCING

8.1 GENERAL

(a) Natural provides personnel available to handle nominations seven (7) days a week, twenty-four (24) hours a day. Whenever Shipper desires service (including service on portions of the system where a gathering charge applies), Shipper shall furnish to Natural a separate nomination for each nominated Receipt and Delivery Point under each Agreement with a beginning and end date, or beginning hour if applicable, for flow which can be for any duration within the term of the applicable Agreement; provided, however, any such nomination shall not be binding to the extent Shipper submits subsequent nomination(s) or exercises no-notice rights available under any service. All nominations should be considered original nominations and should be replaced to be changed. When a nomination for a date range is received, each day within that range is considered an original nomination. When a subsequent nomination is received for one or more days within that range, the previous nomination is superseded by the subsequent nomination only to the extent of the days specified. The days of the previous nomination outside the range of the subsequent nomination are unaffected. Nominations have a prospective effect only.

(b) All nominations, including intra-day nominations, should be based on a daily quantity; thus, an intra-day nominator need not submit an hourly nomination. All volumes shall be expressed in Dth per day and shall be stated for each Receipt and Delivery Point. For Rate Schedules DSS, FRSS, FTS-G/NB, FFTS-NB and FTS-NB, nominations for delivered firm storage service and for firm transportation service, respectively, shall not affect a Shipper's no-notice rights thereunder, but it is recognized that Natural will utilize nominations for scheduling under Section 5 and that, depending on the allocation method in effect, nominations may affect allocations at Delivery Points under Section 11 of these General Terms and Conditions. To the extent a Shipper under Rate Schedule DSS is requesting nominated firm transportation service, or a Shipper is requesting transportation service pursuant to a separate release of transportation rights under Rate Schedule FRSS, such Shipper shall so specify in the nomination. Nominated firm transportation service under Rate Schedule DSS is only available and may only be nominated with respect to the period from October 15 through April 15. Nominated firm transportation rights separately released under Rate Schedule FRSS are limited to the Withdrawal Period and are subject to the provisions of Section 2(g) of Rate Schedule FRSS.

FERC Docket: RP08-319-000

Original Sheet No. 341 original Sheet No. 341: Pending

GENERAL TERMS AND CONDITIONS

(c) If an upstream or downstream party requires additional information, if the volumes transported are subject to a discounted rate, or if additional information is otherwise required by Natural, then, upon notification by Natural, Shipper must include in each nomination such additional information as is specified by Natural. Nominations must be submitted to Natural through Natural's Interactive Website, or such other electronic means as are mutually agreed upon by Natural and Shipper. The sending party should adhere to nomination, confirmation, and scheduling deadlines. The receiving party may waive any submittal deadline in this Section 8.

(d) The standard quantity for nominations, confirmation and scheduling is dekatherms per gas day in the United States, gigajoules per gas day in Canada and gigacalories per gas day in Mexico. (For reference, 1 dekatherm = 1,000,000 Btus; 1 gigajoule = 1,000,000,000 joules; and 1 gigacalorie = 1,000,000,000 calories.) For commercial purposes, the standard conversion factor between dekatherms and gigajoules is 1.055056 gigajoules per dekatherm and between dekatherms and gigacalories is 0.251996 gigacalories per dekatherm. The standard Btu is the International Btu, which is also called the Btu(IT); the standard joule is the joule specified in the SI system of units. The International Btu is specified for use in the gas measurement standards of the American Gas Association, the American Petroleum Institute, the Gas Processors Association and the American Society for Testing Materials. For non-commercial purposes, these associations note that the exact conversion factor is 1.05505585262 Gigajoules per Dekatherm.

8.2 STANDARD NOMINATION CYCLES

Natural supports the following standard nomination

cycles:

- (a) The Timely Nomination Cycle: 11:30 a.m. for nominations leaving control of the nomination party; 11:45 a.m. for receipt of nominations by Natural; noon to send Quick Response; 3:30 p.m. for receipt of completed confirmations by Natural from upstream and downstream connected parties; 4:30 p.m. for receipt of scheduled quantities by shipper and point operator (central clock time on the day prior to flow).
- (b) The Evening Nomination Cycle: 6:00 p.m. for nominations leaving control of the nominating party; 6:15 p.m. for receipt of nominations by Natural; 6:30 p.m. to send Quick Response; 9:00 p.m. for receipt of completed confirmations by Natural from upstream and downstream connected parties; 10:00 p.m. for Natural to provide scheduled quantities to affected shippers and point operators, and to provide scheduled quantities to bumped parties (notice to bumped parties), (central clock time on the day prior to flow).

FERC Docket: RP08-319-000

Original Sheet No. 342 Original Sheet No. 342: Pending

GENERAL TERMS AND CONDITIONS

Scheduled quantities resulting from an Evening Nomination that does not cause another Service Requester on Natural to receive notice that it is being bumped should be effective at 9:00 a.m. on the gas Day; and when an Evening Nomination causes another Service Requester on Natural to receive notice that it is being bumped, the scheduled quantities should be effective at 9:00 a.m. on the gas Day.

- (c) The Intra-day 1 Nomination Cycle: 10:00 a.m. for nominations leaving control of the nominating party; 10:15 a.m. for receipt of nominations by Natural; 10:30 a.m. to send Quick Response; 1:00 p.m. for receipt of completed confirmations by Natural from upstream and downstream connected parties; 2:00 p.m. for Natural to provide scheduled quantities to affected shippers and point operators, and to provide scheduled quantities to bumped parties (notice to bumped parties), (central clock time on the gas Day). Scheduled quantities resulting from Intra-day 1 Nominations should be effective at 5:00 p.m. on the gas Day.
- (d) The Intra-day 2 Nomination Cycle: 5:00 p.m. for nominations leaving control of the nominating party; 5:15 p.m. for receipt of nominations by Natural; 5:30 p.m. to send Quick Response; 8:00 p.m. for receipt of completed confirmations by Natural from upstream and downstream connected parties; 9:00 p.m. for Natural to provide scheduled quantities to affected shippers and point operators (central clock time on the gas Day). Scheduled quantities resulting from Intra-day 2 Nominations should be effective at 9:00 p.m. on gas day. Bumping is not allowed during the Intra-day 2 Nomination Cycle.
- (e) For purposes of Section 8.2 (b), (c), and (d), "provide" shall mean, for transmittals pursuant to standards 1.4.x, receipt at the designated site, and for purposes of other forms of transmittal, it shall mean send or post.
- $\,$ (f) The rights of a Releasing Shipper to recall capacity within any nomination cycle shall be governed by Section 19.14 of these General Terms and Conditions.
- (g) A Shipper which has been awarded firm capacity in a capacity release may submit a nomination using such capacity at the next available opportunity for nominations under this Section 8.2 which occurs on or after the time capacity is awarded, including an intraday nomination in either the Intra-day 1 or the Intra-day 2 Nomination Cycle, and which is consistent with Section 19.9(d) of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 343 Original Sheet No. 343: Pending

GENERAL TERMS AND CONDITIONS

8.3 TIMELY NOMINATIONS

- (a) Timely nominations are nominations submitted consistent with the standard nomination cycle set out in Section 8 $2 \, (a)$
- (b) As provided in Section 5 of these General Terms and Conditions, if acceptance of a nomination for service under a specified priority category under this Section 8.3 would cause interruption of another Shipper's gas on Natural's system under a service in the same priority category, then such nomination will not be accepted.
- $\,$ (c) Nominations received after the timely nomination deadline will be scheduled after the nominations received by that deadline.

FERC Docket: RP08-319-000

Original Sheet No. 344 Original Sheet No. 344: Pending

GENERAL TERMS AND CONDITIONS

8.4 REQUIRED NOMINATION CHANGES

If estimated daily flows under a particular Agreement differ from the confirmed nominations, or if an imbalance has occurred due to some other reason, then prospective nomination change(s) (either receipt or delivery adjustments) may be required to bring the receipt and delivery volumes into balance. When a Shipper receives notification of a required change in the nomination, the Shipper shall be responsible for informing upstream and downstream parties of the prospective change and providing Natural with a valid nomination as required in accordance with Section 8.2 hereof.

8.5 CONFIRMATION BY NATURAL

- (a) Nominations made in accordance with Sections 8.2, 8.3, 8.4 and 8.6 hereof shall not become effective until Natural has confirmed both the nominated receipts and deliveries with upstream and downstream parties, subject to Section 8.5(c) and to the other provisions of this Tariff, such confirmations at Receipt Points shall be accepted by Natural unless Natural has a reasonable basis for believing that the confirmed quantity will not be delivered to Natural. Shipper shall designate the appropriate person(s) who has the authority to resolve allocation issues and, if requested by Natural, the appropriate person(s) to confirm nominations. Confirmations must be submitted to Natural through its Interactive Website, or such other electronic means as are mutually agreed upon Natural by and Shipper.
- (b) Subject to Section 8.2 and to the other provisions of this Tariff, Natural shall provide Shippers and point operators via its Interactive Website and by EDI, the quantities that have been scheduled to flow for that Shipper and point operator on the next day.
 - (c) Default confirmation procedures are as follows:

FERC Docket: RP08-319-000

Original Sheet No. 345 original Sheet No. 345 : Pending

GENERAL TERMS AND CONDITIONS

- (i) With respect to the timely nomination/ confirmation process at a receipt or delivery point, in the absence of agreement to the contrary, the lesser of the confirmation quantities will be the confirmed quantity. If there is no response to a Request For Confirmation or an unsolicited Confirmation Response, the lesser of the confirmation quantity or the previously scheduled quantity will be the new confirmed quantity.
- (ii) With respect to the processing of requests for increases during the intraday nomination/confirmation process, in the absence of agreement to the contrary, the lesser of the confirmation quantities will be the new confirmed quantity. If there is no response to a Request For Confirmation or an unsolicited Confirmation Response, the previously scheduled quantity will be the new confirmed quantity.
- (iii) With respect to the processing of requests for decreases during the intraday nomination/confirmation process, in the absence of agreement to the contrary, the lesser of the confirmation quantities will be the new confirmed quantity, but in any event no less than the elapsed-prorated-scheduled quantity. If there is no response to a Request For Confirmation or an unsolicited Confirmation Response, the greater of the confirmation quantity or the elapsed-prorated-scheduled quantity will be the new confirmed quantity. Elapsed-prorated-scheduled quantity means that portion of the scheduled quantity that would have theoretically flowed up to the effective time of the intra-day nomination being confirmed, based upon a cumulative uniform hourly quantity for each nomination period affected.
- (iv) With respect to 8.5 (c) (i), (ii), and (iii), if there is no response to a request for confirmation or an unsolicited confirmation response, the Transportation Service Provider will provide the Service Requester with the following information to explain why the nomination failed, as applicable:
- (1) the Service Requester's Transportation Service Provider did not conduct the confirmation;
- (2) the Service Requester is told by its Transportation Service Provider that the upstream confirming party did not conduct the confirmation;

FERC Docket: RP08-319-000

Original Sheet No. 346 Original Sheet No. 346: Pending

GENERAL TERMS AND CONDITIONS

- (3) the Service Requester is told by its Transportation Service Provider that the upstream Service Requester did not have the gas or submit the nomination;
- (4) the Service Requester is told by its Transportation Service Provider that the downstream confirming party did not conduct the confirmation;
- (5) the Service Requester is told by its Transportation Service Provider that the downstream Service Requester did not have the market or submit the nomination.

 $$\operatorname{\textsc{This}}$ information will be imparted to the Service Requester on the Scheduled Quantity document.

FERC Docket: RP08-319-000

Original Sheet No. 347 Original Sheet No. 347: Pending

GENERAL TERMS AND CONDITIONS

8.6 INTRADAY NOMINATIONS

- (a) An intraday nomination is a nomination submitted after the Timely Nomination Cycle, defined at Section $8.2\,(a)$, whose effective time is no earlier than the beginning of the gas Day and which runs through the end of that Day.
- (b) Natural supports the nomination cycles set forth at Section 8.2 during non-Critical Times. During Critical Times, valid intraday nominations may be submitted at any time.
- (c) Natural will provide notification of bumped volumes through the Scheduled Quantity document, as posted on Natural's Interactive Website, telephone or telefax consistent with Section 23.6(c) of these General Terms and Conditions of this Tariff and through Electronic Notice Delivery consistent with GISB Standards as adopted in Section 48 of the General Terms and Conditions of this Tariff. During non-Critical Times, Natural will waive daily penalties applicable to bumped volumes on the day of the bump. Natural will also waive penalties if it fails to provide appropriate notice of the bump.
- (d) For services that provide for intra-day nominations and scheduling, there is no limitation as to the number of intra-day nominations (line items as per GISB Standard 1.2.1) which a service requester may submit at any one standard nomination cycle or in total across all standard nomination cycles.

FERC Docket: RP08-319-000

Original Sheet No. 348 Original Sheet No. 348: Pending

GENERAL TERMS AND CONDITIONS

(e) Revised predetermined allocations (described in Section 10 hereof) may need to be submitted in conjunction with the Intra-day Nomination in order to properly allocate the gas received at the nominated Receipt Point.

- (f) Unless Natural agrees to the contrary, the revised nomination under an Intra-day Nomination may be limited by Section $8.5\,(c)$. Natural and the interconnecting party will agree on the hourly flows of the Intra-day Nomination.
- (g) An Intra-day Nomination is only effective for a single day. There is no need to re-nominate if the Intra-day Nomination is intended to modify the existing nomination on that Day only. The Shipper should submit a new timely or intraday nomination if the Shipper wants to replace the previously submitted standing nomination or commence service for the next gas Day.

8.7 END-OF-GAS-DAY SCHEDULED QUANTITY DOCUMENT

At the end of each gas day, Transportation Service Provider will provide the final scheduled quantities for the just completed gas day. With respect to the implementation of this process via the 1.4.x scheduled quantity related standards, Transportation Service Provider will send an end of gas day Scheduled Quantity document. Receivers of the end of gas day Scheduled Quantity document can waive the sender's sending of the end of gas day Scheduled Quantity document.

FERC Docket: RP08-319-000

Original Sheet No. 349 Original Sheet No. 349: Pending

GENERAL TERMS AND CONDITIONS

8.8 OVERRUN OUANTITIES

Shippers submitting nominations via Natural's Interactive Website or EDI for transportation, injection, withdrawal or storage of overrun volumes (volumes in excess of the applicable point, leg, zone or Agreement MDQ, MSV or MAC) may either include such overrun volumes in their nominations for volumes within MDQ or MAC, or may submit separate nominations for such overrun volumes. If the Shipper elects to submit a separate nomination, the Shipper should mark that nomination as being for overrun volumes.

8.9 DELEGATION

A Shipper may delegate to any third party responsibility for submitting and receiving notices or nominations or performing other administrative duties under any Agreement, and an entity which controls a point of interconnection with Natural may delegate to any third party responsibility for administering agreements regarding allocation of gas volumes at the point and/or for administering any Point Operator Agreement, subject to the following conditions:

- (a) Any designation of such a representative, and any change in such designation, must be in writing and must be submitted at least two (2) business days prior to the requested effective date.
- (b) The written designation shall specify any limits on the authority of the representative, including any time limit on the designation; provided, however, that Natural may reject any such limited designation if the limitations specified in the designation would result in an undue administrative burden.
- (c) Natural may rely on communications from the designated representative of a Shipper or interconnecting entity for all purposes except to the extent the designation is explicitly limited as specified in the preceding Section 8.9(b). Communications by Natural to such designated representative shall be deemed notice to Shipper or interconnecting entity except to the extent the representative's authority is explicitly limited with respect to the receipt of notice under the procedure set out in said Section 8.9(b).
- (d) Any third party may administer multiple Agreements as the designated representative for one or more Shippers and/or interconnecting entities. However, such representative shall separately administer and account for each such Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 350 original Sheet No. 350 : Pending

GENERAL TERMS AND CONDITIONS

8.10 TRANSFER NOMINATIONS

- (a) Whenever gas is purchased or sold at a Receipt Point (including a storage or pooling point) on Natural's system by an entity that is not going to nominate that gas for receipt by Natural under any Agreement, that entity must submit a transfer nomination to Natural through Natural's Interactive Website (or other mutually agreed electronic means) identifying the quantities (in Dth) and the entities from whom the gas is being bought and the entities to whom the gas is being sold. Such transfer nominations are needed in order to be able to confirm the nominated receipts at that point and thus such transfer nominations are due by the deadlines applicable to Shipper nominations subject to Section 8.2. In addition to the transfer nomination, the purchasing entity should submit a predetermined allocation in accordance with Section 10 of these General Terms and Conditions if there is more than one buyer of the purchasing entity's gas.
- (b) A third party may provide title tracking services on Natural's system as follows:
- (1) The entity seeking to provide such a service (Third Party Account Administrator) shall so notify Natural in writing, in which event Natural shall establish an identification number for nominations involving the Third Party Account Administrator.
- (2) Transfer nominations consistent with this Section 8.10 must be made by the Shipper tendering gas for delivery to the Third Party Account Administrator, where subsequent title to such gas is to be tracked by the Third Party Account Administrator; and
- (3) The Third Party Account Administrator shall maintain records of any title transfers after delivery of gas to it and shall submit a nomination consistent with this Section 8.10 for delivery of gas to the last party in the chain of title, which party shall also submit a nomination for receipt of the gas consistent with this Section 8.10.

8.11 NOMINATION PRIORITIES

As part of the nomination and transfer nomination process, if there is more than one supply source (whether at a single or at multiple Receipt Points) nominated to be delivered to a single Delivery Point or buyer, the nomination or transfer nomination will identify how and which supply sources should be cut in the event all nominated deliveries are not or cannot be made.

FERC Docket: RP08-319-000

Original Sheet No. 351 original Sheet No. 351 : Pending

GENERAL TERMS AND CONDITIONS

Similarly, the nomination or transfer nomination will identify which delivery should be cut in the event gas is not or cannot be received as nominated (i.e., ranking). Ranking should be included in the list of data elements. Transportation service providers should use service requester provided rankings when making reductions during the scheduling process when this does not conflict with tariff-based rules. Nothing in these procedures is intended to affect obligations between any producer or seller of gas and any purchaser of gas under their separate agreements except to the extent consistent with such separate agreements.

8.12 OPERATIONAL BALANCING

Natural agrees that, if requested by a Shipper, it will negotiate with an entity that operates the facilities interconnecting with Natural at a Receipt Point (Balance Operator) in a good faith effort to reach an agreement to deal with imbalances at the Receipt Points specified (which would be a form of a Predetermined Allocation), subject to the following conditions:

FERC Docket: RP08-319-000

Original Sheet No. 352 Original Sheet No. 352: Pending

GENERAL TERMS AND CONDITIONS

(a) Such agreement must set out a mutually agreeable procedure for dealing, as between Natural and Balance Operator, with any difference between confirmed nominations and actual physical gas flow caused by operational conditions, so that any such discrepancy does not affect any Shipper;

- (b) The Balance Operator must meet the same creditworthiness standards as Shipper;
- (c) Natural and Balance Operator must not have previously entered into such agreement which was terminated because of Balance Operator's failure to perform. Nothing herein is intended to restrict Natural's right to terminate in accordance with its terms any agreement entered into hereunder, including without limitation the right to terminate for Balance Operator's failure to perform consistent with its obligations under the agreement.

8.13 CENTRAL DELIVERY POINTS

- (a) Contained in Natural's Catalog of Points are points which Natural has designated as "Central Delivery Points" (CDPs). A CDP is composed of all the interconnections between Natural and an individual local distribution company (LDC) in the Market Delivery Zone that are within close geographical proximity as determined by Natural with regard to the operational constraints of its system. CDPs may also be developed for other entities or groups on a case-bycase basis, subject to any requisite approvals by affected LDCs. Shippers may nominate deliveries under both firm and interruptible Agreements to a CDP in lieu of the individual interconnections that make up the CDP. Except as described in subsection (b) hereof, all deliveries to the individual Delivery Points which make up the CDP will be aggregated together for purposes of allocating deliveries among Shippers and Agreements pursuant to Section 11 and determining whether Balancing Service or Overrun Charges and penalties are applicable. CDPs may not be utilized as the specified Delivery Point in an IBS Agreement, except as permitted under Section 7(c) of Rate Schedule TBS.
- (b) A CDP may not be a primary Delivery Point, but instead the individual Delivery Point(s) to an LDC and corresponding individual MDQ(s) at such point(s) shall be specified in firm transportation Agreements and in DSS Agreements. Unless such deliveries are nominated by Shipper and confirmed by Natural, Natural has no obligation to deliver on any day under any such Agreement any volumes in excess of the individual MDQ specified in that Agreement at a given point. Whenever Natural determines that operating conditions

FERC Docket: RP08-319-000

Original Sheet No. 353 Original Sheet No. 353: Pending

GENERAL TERMS AND CONDITIONS

only permit deliveries at any individual point(s) equal to Natural's obligations, Natural shall notify the affected Shippers under such Agreements. Except to the extent of any authorized overrun, interruptible or secondary point volumes reflected in confirmed nomination(s), such Shippers shall be responsible for having the LDC limit its receipts from Natural at an individual Delivery Point to: (1) the sum of the individual MDQs under any DSS Agreements and under any other firm Agreement containing the point as a primary point for which Natural has confirmed nominations to the CDP; or (2) such other amount as Natural may designate pursuant to an Operational Flow order as provided in Section 23. Any excess takes at that point by the LDC in those circumstances shall be subtracted from the aggregated volumes at the CDP, deemed overrun volumes and allocated consistent with Section 11 among firm Agreements which have the point as a primary Delivery Point.

(c) Segmenting of a primary path within the Market Delivery Zone could result in a physical forward haul which would be in the opposite direction of the primary path under the original Agreement. The use of out-of-path secondary Delivery Points in the Market Delivery Zone may in certain circumstances result in requests to schedule volumes in excess of Natural's system capacity. No segmentation or use of out-of-path secondary points will be permitted in the Market Delivery Zone to the extent such segmentation or use of out-of-path secondary points could result in a physical forward haul in the opposite direction from the primary path under the original Agreement, the scheduling or movement of gas in excess of Natural's system capacity or deliveries to CDPs in excess of the MDQ under the original Agreement. Any Shipper may, however, request that Natural allow segmentation and/or the use of out-of-path secondary points in the Market Delivery Zone on a case-by-case basis, specifying the specific circumstances involved. Natural will promptly notify Shipper in writing as to whether such segmentation or use of out-of-path secondary points can be accommodated based on the specified circumstances without potentially having the type of adverse operational impact described above on Natural, on Natural's system or on other Shippers. If that determination is made, Natural will grant waiver of this Section, limited to the specified circumstances. If Natural determines that a waiver will not be granted, it shall provide the Shipper with a written statement of the reasons for not granting the waiver. Notwithstanding the foregoing, all deliveries actually made to an LDC in the Market Delivery Zone may be aggregated at its CDPs for accounting purposes.

FERC Docket: RP08-319-000

Original Sheet No. 354 Original Sheet No. 354: Pending

GENERAL TERMS AND CONDITIONS

8.14 (a) Commencing December 1, 2003, a Shipper may segment the primary path of its firm capacity to the extent operationally feasible through the nomination process under this Section 8.14. Shipper segments its contract path consistent with this Section 8.14, the single previously agreed-upon demand charge for reservation of capacity shall continue to apply under that contract, as defined in the Agreement and any related discount agreement, regardless of the number of segments. In addition, any Shipper may segment the primary path of its firm capacity by releasing one or more segments of that capacity (the releasing Shipper may retain one or more segments of its primary path), to the extent operationally feasible, by following the procedures set out in Section 19 of these General Terms and Conditions. In the case of segmentation through release, the Original Shipper may segment by nomination any portion of the primary path it retains to the extent operationally feasible and the Replacement Shipper may segment by nomination any portion of the primary path it obtains in the release to the extent operationally feasible.

- (b) For the purposes of this Section 8.14 and subject to the other provisions hereof, whether segmentation is through nomination or results from release of firm capacity on a segmented basis as set out in Section 8.14(a), segmentation must be of the primary path under an Agreement and shall be deemed operationally feasible unless: (i) the segmentation would result in an increase in firm contractual obligation by Natural on any segment or portion of its system (through an overlap of segments); (ii) the segmentation would result in a forward-haul in a direction opposite to the primary path of the Agreement being segmented, as further set out in (e) below [backhauls are also addressed in (e) below]; or (iii) the segmentation is inconsistent with Section 2(h) of Rate Schedule DSS, Section 2(g) of Rate Schedule FRSS, Section 2(e) of Rate Schedule FTS, Sections 4(f) and 6 of Rate Schedule FTS-G, or Section 8.13(c) of these General Terms and Conditions; provided, however, that any separately-contracted transportation capacity in conjunction with service under Rate Schedules DSS and FRSS may be segmented consistent with this Section 8.14.
- (c) (1) In the event a firm capacity path is segmented under this Section 8.14, each segment shall have secondary point rights in the zone(s) traversed by the path of that segment, subject to the remainder of this Section 8.14(c) and Section 5.5 of these General Terms and Conditions (relating to secondary point rights); provided, however, that if nominations result in an overlapping path (including secondary out-of-path overlap resulting from nominations), flows in excess of the original contract MDQ in the area of overlap shall be subject to overrun charges.

FERC Docket: RP08-319-000

Original Sheet No. 355 Original Sheet No. 355: Pending

GENERAL TERMS AND CONDITIONS

- (2) In the case of a segmented release, if nominations result in an overlapping path, through nomination at out-of-path secondary points, overrun charges for volumes in excess of the MDQ under the original Agreement in the area of overlap shall be applied as follows: (i) if a Shipper is nominating or flowing within the primary path under its segment, overrun shall not apply to that segment so long as such Shipper is within its MDQ on that segment and any overrun shall be assigned to the segment on which Shipper is nominating and flowing outside its path; (ii) if a Shipper is nominating or flowing in excess of its MDQ on a segment under the segmented release, the Shipper shall be assessed overrun based on the volume in excess of its MDQ; and (iii) where the Shipper on each segment is within its MDQ, but is nominating on a secondary out-of-path basis so as to create an overlap in nominated paths, and aggregate nomination by the Releasing and Replacement Shipper in the area of overlap are in excess of the original contract MDQ, then except as provided in (i) and (ii), overrun charges will be assessed to the Releasing Shipper unless the release specifies that such overrun charges are to be assessed to the Replacement Shipper.
- (3) If a Shipper segments the primary path under an Agreement which has opposite leg rights pursuant to Section 5.5(a)(2) of these General Terms and Conditions prior to segmentation, whether segmentation is by nomination or in connection with capacity release, opposite leg rights shall be available after segmentation to the downstream segment only and shall be limited to zones on the opposite leg which are downstream of the zone which corresponds to the zone of segmentation, as follows:
- (i) Where the primary path is segmented in the Amarillo Mainline Receipt Zone or the Gulf Coast Mainline Receipt Zone, no opposite leg rights will exist.
- (ii) Where the primary path is segmented in the Midcontinent Zone, the downstream segment would have out-of-path secondary Receipt Point rights in the Gulf Coast Mainline and where the primary path is segmented in the Texok Receipt Zone, the downstream segment would have out-of-path secondary Receipt Point rights in the Amarillo Mainline Receipt Zone. These additional secondary Receipt Point rights could only be used in conjunction with deliveries into the Market Delivery Zone or to the storage point in the zone of receipt.
- (iii) Where the primary path is segmented in the Permian Receipt Zone, the downstream segment would have out-of-path secondary Receipt Point rights in the Gulf Coast Mainline and Texok Receipt Zones. Where the primary path is segmented in the Louisiana Receipt Zone or the South Texas Receipt Zone the downstream segment would

FERC Docket: RP08-319-000

Original Sheet No. 356 original Sheet No. 356: Pending

GENERAL TERMS AND CONDITIONS

have out-of-path secondary Receipt Point rights in the Midcontinent and Amarillo Mainline Receipt Zones. These additional secondary Receipt Point rights could be used only in conjunction with deliveries into the Market Delivery Zones or to the storage point in the zone of receipt.

(d) The segment in the rate zone furthest upstream in relation to direction of flow for the primary path under the Agreement being segmented shall have the pooling and storage points for that rate zone. This restriction on access to pooling and storage points relates only to allowable nominations, not to points under the contract or the definition of any segment. Notwithstanding the first sentence of this subsection (d), Natural shall allow both the upstream and downstream segments to have access to the pooling or storage point within the zone of segmentation, provided that either (i) the pooling point is the only delivery point which may be used during the period of segmentation by the upstream segment with respect to the segmented capacity and the pooling point is the only receipt point which may be used during the period of segmentation for the downstream segment with respect to the segmented capacity; or (ii) the storage point is the only delivery point which may be used during the period of segmentation by the upstream segment with respect to the segmented capacity and the storage point is the only receipt point which may be used during the period of segmentation for the downstream segment with respect to the segmented capacity. These restrictions on the points which may actually be used in the segmentation must be part of the capacity release posting or the nomination provided to Natural, as applicable, in connection with the segmentation. This restriction on points relates solely to the points which may be used on the upstream and downstream segment during a period of segmentation as a condition of access to the pooling point by both upstream and downstream segments in the zone of segmentation and is not predicated on the specific primary point or secondary point rights under the Agreement being segmented so long as the primary path being segmented extends into the relevant zone. In the event a primary firm capacity path is segmented at a point other than a pooling or storage point, the segment which has the pooling and storage points for the rate zone will have interruptible priority for transportation to the storage point in the zone where the path was segmented, but will have out-ofpath secondary point priority from the storage point in that zone. No firm rights on injection will exist unless the path of the segment separately qualifies for primary point rights.

(e) The direction of flow for path segments must be the same direction of flow as for the original path unless the Shipper establishes new primary points which represent a flow reversal, which a Shipper may do if operationally feasible and if the requisite capacity is available, applying the point change procedures in Section

FERC Docket: RP08-319-000

Original Sheet No. 357 Original Sheet No. 357: Pending

GENERAL TERMS AND CONDITIONS

5.1(c)(5) of these General Terms and Conditions. Natural shall provide a written explanation if such a request is denied because the flow reversal is not operationally feasible. A Shipper may segment a backhaul, but such segmentation shall be subject to review by Natural on a case-by-case basis as to whether a backhaul on each resulting segment is operationally feasible. The Shipper (or Replacement Shipper in the case of a release) may nominate service at Receipt and Delivery Points for the path segment that results in a reverse flow from the original path; however, a forward haul will be treated as being outside of the path if the original path is a backhaul, as provided in Section 5.5 of these General Terms and Conditions. Subject to the availability of point capacity and to ordinary nomination procedures, deliveries may be made at the same point for a forward haul on the upstream segment and a backhaul of the downstream segment. The forward haul will have priority at the point if the point capacity is not adequate. At the point of segmentation, the upstream Shipper may nominate deliveries up to MDQ and the downstream Shipper may nominate receipts up to MDQ, subject to constraints on point capacity and ordinary scheduling procedures and priorities.

- (1) Subject to the remainder of this subsection 8.14(f), the Releasing and Replacement Shipper involved in a segmented release may each choose primary Receipt Points and primary Delivery Points within the applicable zones equal to their respective contract MDQs after the release; provided, however, that the resulting segments may not overlap. If the points chosen are within the path of the original Agreement and do not involve changing a primary point thereunder, the point designations shall be accepted, subject only to the availability of firm capacity at those points and agreement on the rate applicable at the new primary points. If one or more points are within the zone but outside the path of the original Agreement, then the point change provision in Section 5.1(c)(5) of these General Terms and Conditions and the provisions of subsection 8.14(f)(2) shall apply. If capacity is not available at the point or to the point, the Shipper may designate such points (not overlapping with other segments) in defining a segment, but the points shall have secondary point priority (either in-path or out-of-path, as applicable).
- (2) A Shipper, a Replacement Shipper or a Subreplacement Shipper may change the primary Receipt or Delivery Points listed in the Agreement to new primary point(s) subject to the point change provisions of Section 5.1(c)(5) of these General Terms and Conditions, if the Shipper (or in the case of a release, the Original Shipper) and Natural agree to amend the Agreement to change the primary Receipt and Delivery Point accordingly; provided, however, that consent of the original Shipper shall not be required in the case of a permanent release by that original Shipper of its firm capacity. Natural shall not be obligated to reserve

FERC Docket: RP08-319-000

Original Sheet No. 358 Original Sheet No. 358: Pending

GENERAL TERMS AND CONDITIONS

firm capacity to reinstate the former primary points upon expiration of the segmentation or the capacity release. Except in cases where the capacity determination is unusually complex, Natural will respond to a point change request hereunder within two (2) Business Days or less.

- Any primary point established under this subsection 8.14(f) which was not a primary point under the original Agreement and is not reflected in a point change which affects the original Agreement, will have the same priority as any other primary point, but shall be a primary point only until the term of the release ends or until that primary point is changed by Shipper pursuant to the provisions of this Tariff, whichever occurs first. Such an additional primary point may not be designated if a Shipper is releasing to itself. Where a Replacement Shipper selects a primary point which is outside the primary path under the Releasing Shipper's contract (and thus creates a new primary path at least partially outside the original primary path), any recall by the Releasing Shipper will be of capacity which contains the changed primary points, not the Releasing Shipper's original primary points; provided that a Releasing Shipper specifying recall rights in a release may include a provision in the release which precludes selection of any primary point by the Replacement Shipper which is outside the original primary path absent the Releasing Shipper's consent.
- (g) The overrun provision of this Tariff shall apply to each segment and the associated secondary point rights.

FERC Docket: RP08-319-000

Original Sheet No. 359 Original Sheet No. 359: Pending

GENERAL TERMS AND CONDITIONS

9. POOLING AND STORAGE POINTS

9.1 GENERAL

One or more paper pooling and storage points have been designated for each receipt zone on Natural's system. These points are not physical points on the system, but are used solely for nomination and transportation rate determination purposes. Subject to the remainder of this Section, any number and type of transportation Agreements may be utilized to deliver gas to or from a storage or pooling point.

9.2 POOLING POINTS

- (a) There will be at least one pooling point in each receipt zone. In Natural's Texok Receipt Zone, there shall be two pooling points: one dedicated to volumes received on the A/G Line (Station 801 and east to the interconnect between the Texok and Gulf Coast Mainline Receipt Zones) and the Amarillo Line ("Texok A/G Pooling Point"); and one dedicated to volumes received elsewhere in the Texok Receipt Zone (from the Louisiana/Texas Border, including contiguous offshore laterals, westward to Station 302 and then northward to the Interconnect with the Gulf Coast Mainline Receipt Zone) or anywhere else on the system other than the A/G Line and the Amarillo Line ("Texok Gulf Coast Pooling Point"). In Natural's Iowa-Illinois Receipt Zone there shall be two pooling points, one dedicated to volumes received on the Amarillo Line ("Amarillo Pooling Point") and one dedicated to volumes received on the Gulf Coast Line ("Gulf Coast Pooling Point"). "Amarillo Line" shall be defined for purposes of this Section 9 as all pipeline in the Permian, Midcontinent and Amarillo Mainline Receipt Zones, and all pipeline west or north of Station 113 in the Iowa-Illinois Receipt Zone. "Gulf Coast Line" shall be defined for purposes of this Section 9 as all pipeline in the South Texas, Texok, Louisiana and Gulf Coast Mainline Receipt Zones, and all pipeline south of Station 113 (including Station 113) in the Iowa-Illinois Receipt Zone. Notwithstanding the definition set out herein, certain supply receipts on the Gulf Coast Line are treated as having been received on the Amarillo Line, as set out in Section 5.6(c) of these General Terms and Conditions, reflecting flow conditions on Natural's system.
- (b) Shippers will be able to nominate gas volumes from one or more Receipt Points for delivery to a pooling point in order to aggregate supplies as long as such gas volumes are nominated for simultaneous receipt and further transportation and delivery by

FERC Docket: RP08-319-000

Original Sheet No. 360 original Sheet No. 360: Pending

GENERAL TERMS AND CONDITIONS

Natural under transportation Agreement(s). Shippers nominating volumes to or from pooling points in the Texok Receipt Zone shall specify the Texok A/G or Texok Gulf Coast Pooling Point, as appropriate, based upon the Receipt Point(s) of the gas delivered to the pooling point. Shippers nominating volumes to or from pooling points in the Iowa-Illinois Receipt Zone shall specify the Amarillo or Gulf Coast Pooling Point, as appropriate, based upon the Receipt Point(s) of the gas delivered to the pooling point.

9.3 STORAGE POINTS

Storage points are imaginary or paper points. There will be one (1) paper storage point in each receipt zone except that two paper (2) storage points will exist in the Iowa-Illinois Receipt Zone to correspond to the two (2) pooling points ("Amarillo Storage Point" and "Gulf Coast Storage Point"). Shippers will be able to nominate gas volumes for delivery to or receipt from a storage point as long as such gas volumes are nominated for simultaneous injection, or withdrawal, as applicable, under a DSS or storage Agreement(s) with Natural. Volumes may only be injected or withdrawn under a DSS Agreement at the Iowa-Illinois storage points. Sourcing of supplies for injection into and withdrawal from storage points shall be governed by Section 5.6(c) of these General Terms and Conditions. Storage injections and withdrawals under Rate Schedule DSS and storage withdrawals under Rate Schedule FRSS may be on a no-notice basis as provided in those Rate Schedules. Volumes injected at a storage point under an NSS, PALS or BESS Agreement may only be withdrawn from that same storage point. Except as otherwise agreed, volumes withdrawn under an FRSS Agreement must be injected at the same storage point. Except for Rate Schedule FRSS, volumes may only be withdrawn or nominated for withdrawal from a storage point if such volumes were previously accounted for as being injected at that storage point. Withdrawals under Rate Schedule DSS shall be governed by the rules set out in Section 5.6. Nominations at a paper storage point may not be evaluated or rejected based on whether storage is physically located in the zone. Instead, such nominations together with no-notice activity shall be evaluated based on Natural's overall systemwide ability to accommodate volumes tendered for injection or requested for withdrawal in the zone. If Natural's capability to accommodate gas tendered for injection into or withdrawals from any storage point is less than the requested volume, available storage capability for such storage point shall be allocated as set out in Section 5 of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 361 original Sheet No. 361 : Pending

GENERAL TERMS AND CONDITIONS

9.4 INCLUSION IN AGREEMENTS

- (a) Except as may be specifically provided to the contrary in this Tariff, Agreements under Rate Schedules FTS-SW, FFTS-SW and ITS automatically have all pooling and storage points as both Receipt and Delivery points with point MDQs equal to the contract MDQ. Except as otherwise provided in Section 5.6(b), such points shall be out-of-path secondary points. Agreements under Rate Schedule DSS shall have the Iowa-Illinois Receipt Zone storage points as Receipt Points. The Iowa-Illinois Receipt Zone storage points may also be Delivery Points under the transportation Agreements used for injection under Rate Schedule DSS. Such storage points shall be treated as primary Receipt Points on withdrawal under Rate Schedule DSS for scheduling, curtailment and similar purposes. Such storage points shall also be treated as primary Delivery Points for scheduling, curtailment and similar purposes on injection under Rate Schedule DSS to the extent provided in Section 5.6(b). FRSS Agreements shall have the storage points which correspond to the primary Delivery Point(s). Such storage points shall be treated as primary Receipt Points on withdrawal under Rate Schedule FRSS for scheduling, curtailment and similar purposes. Such points shall also be treated as primary Delivery Point(s) for scheduling, curtailment and similar purposes to the extent provided in Section 5.6(b).
- (b) Except as may be specifically provided to the contrary in this Tariff, firm transportation Agreements under all other Rate Schedules, except for Rate Schedules in Volume No. 2 of this Tariff, automatically have those pooling and storage points located in the rate zones along the transportation paths created by the primary Receipt and Delivery Points under the applicable firm transportation Agreement. Except as provided in Section 5.6(b), such pooling and storage points will be out-of-path secondary Receipt and Delivery Points with the same receipt and delivery MDQ as other secondary Receipt and Delivery Points in that rate zone.

FERC Docket: RP08-319-000

Original Sheet No. 362 original Sheet No. 362 : Pending

GENERAL TERMS AND CONDITIONS

9.5 USE OF POINTS

Nominations to and from pooling and storage points will be subject to the same nomination and confirmation procedures as all other receipts and deliveries. All volumes nominated for transportation to a pooling point on any day must be matched with nominations of those volumes for transportation from the pooling point on the same day. All volumes nominated for transportation to or from a storage point must be matched with volumes nominated for injection or withdrawal under a DSS, FRSS or storage Agreement the same day. No imbalances will be permitted at the pooling or storage point. Section 5 of these General Terms and Conditions sets out the order of priority for scheduling nominations to and from pooling and storage points.

9.6 CHARGES TO AND FROM NATURAL'S POOLING AND STORAGE POINTS

There will be no transportation commodity charges or Fuel Gas or Gas Lost and Unaccounted For applicable to transportation of gas to a pooling or storage point in the same zone in which the Receipt Point is located, except as otherwise provided under Rate Schedule DSS and FRSS (Storage Fuel is also assessed under Rate Schedule NSS upon injection). If the Receipt Point at which gas is received is located in a different zone than the pooling or storage point, or if gas is moved from a storage or pooling point in the zone of receipt to a storage or pooling point in a zone other than the zone of receipt, the applicable transportation commodity charges (other than ACA) and Fuel Gas and Gas Lost and Unaccounted For rates from the zone of receipt or the zone in which the gas is originally pooled or stored to the new zone will be charged under the Agreement used to transport the gas from the Receipt Point or from the prior pooling or storage point to the pooling or storage point in the new zone. For commodity charges under an ITS Agreement and for Fuel Gas and Gas Lost and Unaccounted For under any Agreement, Natural will reduce such charges by the intrazone charges in the zone to which the gas is moved.

FERC Docket: RP08-319-000

Original Sheet No. 363 Original Sheet No. 363: Pending

GENERAL TERMS AND CONDITIONS

9.7 CHARGES TO AND FROM THIRD PARTY STORAGE POINTS

- (a) All applicable transportation charges (including Fuel Gas and Gas Lost and Unaccounted For) will be charged for delivering gas to Third Party Storage Points, as such term is defined in subsection (d) hereof.
- $\mbox{\ensuremath{\mbox{(b)}}}$ Provided that the conditions of subsection (c) hereof are satisfied,
- (1) Natural shall not assess any commodity charge for transportation of gas from Third Party Storage Points under firm Agreements within MDQ;
- (2) Natural shall not assess any transportation charge otherwise applicable to the rate zone in which the Third Party Storage Point is located, for transportation of gas from Third Party Storage Points under firm Agreements for authorized overrun gas or under an ITS Agreement; and
- (3) Natural shall not assess any Fuel Gas otherwise applicable to the rate zone in which the Third Party Storage Point is located or any Gas Lost and Unaccounted For for transportation of gas from Third Party Storage Points under any Agreement.
- (c) In order for transportation of gas from Third Party Storage Points to be exempt from the charges specified under subsection (b) hereof, the Shipper must request such exemption and must warrant to Natural that the gas (1) has been withdrawn from the third party storage facility and (2) was previously transported by Natural to the Third Party Storage Point under an Agreement between Natural and Shipper. In addition, to be eligible for exemption from the charges, the Shipper must provide or cause the owner of the third party storage facility to provide Natural with sufficient documentation (including, for example, a source and disposition report) to confirm the accuracy of the warranty.

FERC Docket: RP08-319-000

Original Sheet No. 364 Original Sheet No. 364: Pending

GENERAL TERMS AND CONDITIONS

(d) As used in this Section, a "Third Party Storage Point" is a bi-directional (capable of both physically receiving and physically delivering gas) interconnection between Natural and either an underground storage facility owned by a third party, or a pipeline lateral owned by a third party connecting to an underground storage facility that is also owned by that same third party and that is located within twenty-five (25) miles of the interconnection.

9.8 IN-FIELD STORAGE TRANSFERS

- (a) Upon prior notice to Natural, a Shipper ("Transferor") may transfer gas in place in storage from one Agreement to another Agreement under which the Transferor or another entity ("Transferee") is the Shipper, provided that:
- (1) the transfer does not cause the Transferee to exceed its storage rights under the Agreement to which the gas is being transferred;
- (2) all transfers are subject to the applicable priorities set out in Section 5 of these General Terms and Conditions and transfers from a BESS Agreement into any firm Agreement are subject to Natural's prior approval based upon operational conditions on its system;
- (3) the transfer occurs at the same zone storage point (for volumes in a different storage point, Transferor must first transport gas to the transfer storage point) and in the case of transfers from DSS into DSS, from DSS into NSS or from NSS into DSS, the transfer occurs at the storage point in the Iowa-Illinois Receipt Zone where the storage balance to be transferred resides and in the case of transfers into or out of FRSS, the transfer takes place at the storage point under the FRSS Agreement;
- (4) the transfer does not cause Natural's obligations to provide firm service to increase and transfers out of a DSS Agreement cannot be used to satisfy the cycling requirement and, without Natural's prior consent, transfers into an FRSS Agreement cannot be used to satisfy the injection requirement;

FERC Docket: RP08-319-000

Original Sheet No. 365 Original Sheet No. 365: Pending

GENERAL TERMS AND CONDITIONS

(5) the transfer of the gas shall be treated as a withdrawal under the Agreement the gas is being transferred from if such Agreement is subject to a limitation on the number of days of withdrawal, unless the gas is being transferred to an Agreement under the same Rate Schedule; to calculate the number of days of withdrawal utilized as a result of the transfer, the volume transferred shall be divided by the WQ of the Agreement and any resulting fraction shall be rounded to the next higher whole number;

 $\mbox{\ensuremath{\mbox{(6)}}}$ the transfer may not otherwise adversely affect Natural's operations; and

(7) notwithstanding the foregoing, a Shipper under Rate Schedule DSS may transfer each calendar year up to 30% of its MSV in effect on November 1 into its Rate Schedule DSS account from one or more Rate Schedule NSS accounts (held by itself or another Shipper) through an in-ground transfer hereunder; provided that the transfer occurs sometime between November 1 and November 15 of that year, the Transferee and the Transferor provide notice at least two (2) days prior to the date of transfer, the transfer complies with subsections 9.8(a)(1), (3) and (5) above, and the Transferee complies with Section 9.8(b). If that Rate Schedule DSS Shipper has implemented an in-ground transfer for a portion of its MSV prior to November 1 of the same year, Natural will evaluate on a case-by-case basis whether that Shipper's rights hereunder are correspondingly reduced.

(b) All transfers shall be effective as of the start of a day. Charges for storage services shall be the responsibility of the Transferor before the date of transfer and of the Transferee on and after the date of transfer. There will be no charge for a transfer in the same storage point; provided, however, that for transfers into Rate Schedule DSS from a different service, the Transferee shall pay the applicable Fuel Gas and Gas Lost and Unaccounted For (such charges are assessed on injection under Rate Schedule DSS). For transfers into a different storage point, the gas must first be transported to the transfer point and the Transferor must pay any applicable transportation and fuel charges.

9.9 MARKET CENTERS

Pooling and storage points are intended to facilitate aggregation of supplies and the development of market centers. Nothing in this Tariff is intended to inhibit the development of market centers.

FERC Docket: RP08-319-000

Original Sheet No. 366 original Sheet No. 366: Pending

GENERAL TERMS AND CONDITIONS

10. DETERMINATION OF DAILY RECEIPTS

- 10.1 To the extent feasible, all volumes received by Natural at a Receipt Point shall be allocated in accordance with the confirmed nominations for that point. In the event the actual volumes received by Natural do not equal the confirmed nominations for that point, any underage or overage will be allocated as follows:
- (a) First, in accordance with the effective predetermined allocations (PDAs) submitted by those entities (Allocators) owning or controlling the gas being delivered to Natural. An operational balancing agreement ("OBA") is one type of a PDA. Shipper agrees that such an allocation is binding on Shipper.
- (b) Then, if there is no effective PDA, pro rata to the extent applicable based on confirmed nominations or transfer nominations, as applicable. Shipper agrees that such an allocation is binding on Shipper.
- 10.2 Each Shipper shall be responsible for ensuring that its suppliers submit PDAs as provided herein, except that no other PDAs need be submitted if an OBA is in effect at a point. All PDAs must be submitted to Natural through Natural's Interactive Website, EDI, or other agreed upon electronic means, on or before the gas Day the PDA is to be effective. Such PDA shall specify how any underage or overage from the confirmed nominated volumes should be allocated among the entities listed on the PDA. Natural shall acknowledge receipt and acceptance of the PDA through Natural's Interactive Website if received through Natural's Interactive Website or via EDI if received via EDI (or returning via some other mutually agreeable means the PDA to Allocator). Such notification of acknowledgment and acceptance will be within fifteen (15) minutes of receipt. Natural's acceptance is contingent on Natural being able to administer the allocation submitted by the Allocator. Allocation methodology types upon which two parties may agree are: ranked, pro rata, percentages, swing and operator provided value. Other examples of allocation methods which can be used are matching of supply sources with specified customers and combinations of methodology types. Different methods may be submitted for overages and underages. If the parties cannot agree, Section 10.1(b) shall apply.

FERC Docket: RP08-319-000

Original Sheet No. 367 Original Sheet No. 367: Pending

GENERAL TERMS AND CONDITIONS

10.3 A PDA will be effective as of the date specified thereon (which may not be earlier than the date on which the PDA is submitted to Natural unless otherwise agreed) and will continue in effect through the end of the calendar month unless the Allocator submits a new PDA that is accepted by Natural. PDAs may be submitted to Natural on any day or days during the month and should be submitted if necessary to reflect any changes in the Shippers at the point.

- 10.4 Allocators who should submit PDAs include the operator of the upstream facilities, the Shippers or producers/owners of the gas being delivered by the upstream entity, buyers of the gas who are in turn selling the gas at that point, and Shippers who are using more than one Agreement at that point.
- 10.5 After the end of each month, Natural shall provide each Allocator who submits effective PDA(s) with a monthly allocation statement showing the volumes allocated in accordance with such PDA(s).
- 10.6 Natural may rely conclusively on effective PDAs in allocating the gas received at a point. No retroactive changes to the PDA may be made unless Natural and all affected parties agree.

FERC Docket: RP08-319-000

Original Sheet No. 368 Original Sheet No. 368: Pending

GENERAL TERMS AND CONDITIONS

11. DETERMINATION OF DELIVERIES

11.1 PREDETERMINED ALLOCATIONS

In accounting for the volumes delivered by Natural, in circumstances where multiple services are provided at any Delivery Point, the sequence of volumes delivered shall be determined by a predetermined allocation agreement between Natural and the operator of the facilities immediately downstream of the point at which Natural delivers gas. Natural may reject a methodology proposed by a point operator for allocation of volumes if it determines, based on its reasonable judgment, that the proposed methodology is administratively infeasible or unduly burdensome or would have an adverse impact on Natural, on Natural's system or on Shippers on Natural's system not covered by the proposed methodology. Any such rejection shall be in writing and shall specify the reasons for rejection. Such determination shall be made on a non-discriminatory basis, taking into account differences in factual circumstances between various point operators. Allocation methodology types upon which two parties may agree are: ranked, pro rata, percentages, swing and operator provided value. Other examples of allocation methods that can be used are combinations of methodology types. Different methods may be submitted for overages and underages. The predetermined allocation methodology at any Delivery Point may be changed by agreement between Natural and the point operator as frequently as daily. In the absence of such an agreement, Section 11.2 or 11.3, as applicable, shall control. Any new or proposed change to the methodology should be sent to Natural on or before the gas Day on which the methodology is to be effective. Natural shall confirm receipt of the methodology within fifteen (15) minutes through Natural's Interactive Website if received through Natural's Interactive Website, or via EDI if received via EDI.

11.2 DEFAULT ALLOCATION OF DELIVERIES

In the absence of a Point Operator Allocation Agreement under Section 11.3 below, or of any other allocation agreement as referenced in Section 11.1 above, volumes delivered by Natural at any Delivery Point shall be allocated in accordance with this Section 11.2. Volumes delivered by Natural at any Delivery Point shall be allocated among Shippers pro rata based on their respective confirmed nominations. If actual deliveries are either more than or less than confirmed nominations, the discrepancy shall be allocated pro rata based on confirmed nominations. Charges may be applicable under Section 12 of these General Terms and Conditions to the extent the resulting allocated deliveries deviate from applicable delivery rights and from confirmed nominations.

FERC Docket: RP08-319-000

Original Sheet No. 369 Original Sheet No. 369: Pending

GENERAL TERMS AND CONDITIONS

11.3 POINT OPERATOR ALLOCATION AGREEMENT

- (a) The operator of facilities immediately downstream of a Delivery Point which has physical control of the volumes taken away from Natural and is willing to take responsibility for disposition of the volumes it receives from Natural has the option to elect a predetermined allocation pursuant to this Section 11.3. Allocation agreements under this Section 11.3 may apply to physical Delivery Points, to Central Delivery Points or to other point groupings mutually agreeable to Natural and the point operator. To perfect such an election, such point operator must execute a Point Operator Allocation Agreement which adheres to the delivery sequences set out herein, except as such sequence may be modified by mutual agreement between Natural and the point operator, and which covers deliveries under all Agreements pursuant to which gas is delivered by Natural at the point other than those specifically excluded by mutual agreement of Natural and the point operator (Applicable Agreements).
- (b) In the event that volumes delivered by Natural under Applicable Agreements at any Delivery Point equal or exceed the sum of confirmed nominations at such point on any day, the delivery sequence for that day shall be as follows:
- (1) First, confirmed nominations under all Applicable Agreements shall be deemed delivered for that day; provided, however, that volumes deemed delivered for that day are subject to adjustment (as part of the allocation done for that day) under Rate Schedules DSS, FRSS and FTS-NB, FFTS-NB or FTS-G/NB reflecting no-notice changes thereunder pursuant to later steps in this delivery sequence.
- (2) Volumes equal to up to the percentage of the confirmed nominations under step (1) which corresponds to the percentage for the Balancing Service Charge tier under Section 12 of these General Terms and Conditions for which no additional charge is assessed under then prevailing circumstances (or such greater minimum volume as provided under Section 12.6) are the next volumes deemed delivered on any day (which absorbs any allowable balancing service provided without additional charge under Section 12 of these General Terms and Conditions) and are allocated to the point operator's account, but volumes shall be deemed delivered to the point operator's account only to the extent necessary to account for

FERC Docket: RP08-319-000

Original Sheet No. 370 Original Sheet No. 370: Pending

GENERAL TERMS AND CONDITIONS

deliveries on that day. The point operator's account is available for the allocation of volumes hereunder on any day except during a Critical Time or as otherwise provided in an Operational Flow Order; provided that the 1,000 Dth per day exemption under Section 12.6 shall also apply during a Critical Time and when an Operational Flow Order is in effect to the extent provided by said Section 12.6. Moreover, allocations under this step (2) to a point operator's account shall not be made to the extent such an allocation would be inconsistent with an Operational Flow Order nor shall any such allocation be made during a Critical Time; provided that the 1,000 Dth per day exemption under Section 12.6 shall also apply during a Critical Time and when an Operational Flow Order is in effect to the extent provided by said Section 12.6. The cumulative volumes for operator's account hereunder over any month may not exceed two percent (2%) of cumulative confirmed nominations for that month or any lesser amount specified in an Operational Flow Order. The determination related to the prior sentence shall be made at the end of the month and any volumes in excess of the 2% cashed out. Such cashout shall be effectuated without offset, netting or trading, in accordance with Section 13.3 of these General Terms and Conditions, except that imbalances from 2% to 5% shall be cashed out in the "5% to 10%" tier.

- (3) Volumes under Applicable Agreements equal to available rights to withdraw and deliver gas to the point (net of injections) pursuant to DSS Agreements held by the point operator shall be the next volumes deemed delivered that day. If the point operator has more than one DSS Agreement at the point, volumes shall be allocated among such Agreements by a method mutually agreeable to Natural and the point operator.
- (4) Volumes under Applicable Agreements equal to the sum of point MDQs at that point (net of injections) under FTS-NB, FFTS-NB or FTS-G/NB Agreements held by the point operator shall be the next volumes delivered that day up to the MDQ at the point under such Agreements (even if such deliveries result in an overrun of such point operator's NSS rights). If the point operator has more than one FTS-NB, FFTS-NB or FTS-G/NB Agreement at the point, volumes shall be allocated among such Agreements by a method mutually agreeable to Natural and the point operator.
- (5) Volumes under Applicable Agreements equal to the sum of point MDQs at that point (net of injections) under FRSS Agreements held by the point operator shall be the next volumes delivered that day up to the MDQ at the point under such Agreements. If the point operator has more than one FRSS Agreement at the point, volumes shall be allocated among such Agreements by a method mutually agreeable to Natural and the point operator.

FERC Docket: RP08-319-000

Original Sheet No. 371 Original Sheet No. 371: Pending

GENERAL TERMS AND CONDITIONS

- (6) Any additional volumes delivered at the point on any day which are not accounted for by applying steps (1) through (5) shall be allocated for that day as overrun deliveries to the point operator's account and shall be subject to charges under Section 12 of these General Terms and Conditions [it being understood that the allowable balancing service to be provided without additional charge under Section 12 has already been provided under step (2) above].
- (c) In the event that volumes delivered by Natural under Applicable Agreements at any Delivery Point are less than the sum of confirmed nominations at the point on any day, the delivery sequence for that day shall be as follows:
- (1) First, confirmed nominations under all Applicable Agreements shall be deemed delivered on that day; provided, however, that volumes deemed delivered for that day are subject to adjustment (as part of the allocation done for that day) under Rate Schedules DSS, FTS-G/NB, FFTS-NB and FTS-NB reflecting no-notice changes thereunder pursuant to later steps in this delivery sequence.
- (2) Of the volumes nominated but not delivered, volumes equal to up the percentage of the confirmed nominations under step (1) which corresponds to the percentage for the Balancing Service Charge tier under Section 12 of these General Terms and Conditions for which no additional charge is assessed under then prevailing conditions (or such greater minimum volume as provided under Section 12.6) shall be allocated to the point operator's account on any day (such allocation absorbs any allowable balancing service provided without additional charge under Section 12 of these General Terms and Conditions), but only to the extent necessary to account for deliveries on that day. The point operator's account is available for the allocation of volumes hereunder on any day except during a Critical Time or as otherwise provided in an Operational Flow Order; provided that the 1,000 Dth per day exemption under Section 12.6 shall also apply during a Critical Time and when an Operational Flow Order is in effect to the extent provided by said Section 12.6. Moreover, allocations under this step (2) to a point operator's account shall not be made to the extent such an allocation would be inconsistent with an Operational Flow Order nor shall any such allocation be made during a Critical Time; provided that the 1,000 Dth per day exemption under Section 12.6 shall also apply during a Critical Time and when an Operational Flow Order is in effect to the extent provided by said Section 12.6. The cumulative volumes for operator's account hereunder over any month may not exceed two percent (2%) of cumulative confirmed nominations for that month or any lesser amount specified in an Operational Flow Order.

FERC Docket: RP08-319-000

Original Sheet No. 372 Original Sheet No. 372: Pending

GENERAL TERMS AND CONDITIONS

The determination related to the prior sentence shall be made at the end of the month and any volumes in excess of the 2% cashed out. Such cashout shall be implemented without offset, netting, trading, in accordance with section 13.3 of these General Terms and Conditions, except that imbalances from 2% to 5% shall be cashed out in the 5% to 10% tier.

- (3) Volumes under Applicable Agreements equal to available rights to inject gas (net of withdrawals) pursuant to DSS Agreements held by the point operator shall next be deemed injected for that day into the point operator's DSS accounts. If the point operator has more than one DSS Agreement at one point, volumes shall be allocated among such Agreements by a method which is mutually agreeable to Natural and the point operator.
- (4) Volumes under Applicable Agreements equal to the sum of confirmed nominations at the point (net of withdrawals) pursuant to all FTS, FFTS, FTS-G and ITS Agreements held by the point operator shall next be deemed injected under FTS-NB, FFTS-NB and FTS-G/NB Applicable Agreements, if any, held by the point operator up to the MDQ at the point under such Agreements (even if such injections result in an overrun of point operator's NSS rights). If the point operator has more than one FTS-NB, FFTS-NB or FTS-G/NB Agreement at the point, volumes shall be allocated among such Agreements by a method mutually agreeable to Natural and the point operator.
- (5) To the extent less volumes have been delivered on any day than are accounted for by applying steps (1) through (4), such shortfall shall be allocated for that day to the point operator's account and shall be subject to any applicable charges under Section 12 of these General Terms and Conditions [it being understood that the allowable balancing service to be provided without additional charge under Section 12 has already been provided under step (2) above].
- (d) Allocation of volumes between pipeline legs (Amarillo or Gulf Coast) under subsection (b) or (c) shall be in accordance with Section 5.6(c) of these General Terms and Conditions.
- (e) To be eligible for a Point Operator Allocation Agreement, a prospective point operator must meet the same creditworthiness standards as a Shipper. Further, Natural and the point operator must not have previously entered into such an agreement which was terminated because of point operator's failure to perform. Nothing herein is intended to restrict Natural's right to terminate in accordance with its terms any Point Operator Allocation Agreement entered into hereunder, including without limitation the right to terminate because of point operator's failure to perform consistent with its obligations under the Point Operator Allocation Agreement.

FERC Docket: RP08-319-000

Original Sheet No. 373 Original Sheet No. 373: Pending

GENERAL TERMS AND CONDITIONS

(f) Unless otherwise agreed by Natural or otherwise provided herein, cashout procedures shall not apply to volumes in a point operator's account; provided, however, that any volume remaining in operator's account upon termination of a Point Operator Allocation Agreement shall be cashed out based on the cashout prices for the last month such agreement was in effect. The point operator may transfer gas out of its account by submitting nominations for transportation and for storage injections or withdrawals subject to ordinary confirmation and scheduling procedures.

- (g) Natural may cancel any Point Operator Allocation Agreement in the event that point operator's account exceeds a reasonable level in relation to the MDQ at the point(s) covered, if the point operator does not take reasonable steps to rectify imbalances on a timely basis, or if the point operator exceeds with unreasonable frequency the 2% monthly restriction under this Section
- (h) Aggregators of supply at any point which do not control physical operation of the point may request that they be allowed to enter into allocation agreements similar to Point Operator Allocation Agreement(s) as to specified volumes or Agreements delivering gas at the point, but any such request must include a consent by the physical point operator and other entities which could receive an allocation of gas under such agreement and is subject to review by Natural as to its practicability.

11.4 POINT MANAGEMENT AGREEMENT

Natural generally obtains a Point Operator Allocation Agreement for every point where deliveries are being made by Natural on behalf of multiple entities. In some situations, however, the point operator may not be willing to accept responsibility for the disposition of all deliveries at the point. Under those circumstances, Natural shall not be obligated to commence or continue physical gas deliveries to the point or to confirm any nominations for delivery to the point unless there is a Point Management Agreement (PMA) in effect at the point. A PMA shall specify the entity (which may be the point operator or another entity with access to and control of the meter equipment at the point if agreed by the point operator) which is to provide to Natural the allocation of physical gas deliveries to entities behind the point. Any PMA must be agreed to by the point operator and the entity which will be providing allocations to Natural

FERC Docket: RP08-319-000

Original Sheet No. 374 Original Sheet No. 374: Pending

GENERAL TERMS AND CONDITIONS

(if different from the point operator) and by Natural. The PMA will conclusively govern the allocation of physical gas volumes among entities behind the point so long as it remains in effect. The allocation among services of volumes allocated to entities under the PMA shall be based on predetermined allocations as between Natural and the individual entities which are allocated physical gas deliveries under the PMA or, if no valid predetermined allocation is in place, the applicable default provisions of this Section 11 of these General Terms and Conditions. Natural may cancel or terminate a PMA on two (2) days' notice if it is not receiving physical gas allocation information on a timely basis consistent with the terms of the PMA.

FERC Docket: RP08-319-000

Original Sheet No. 375 Original Sheet No. 375: Pending

GENERAL TERMS AND CONDITIONS

12. BALANCING SERVICE AND OVERRUN CHARGES

12.1 GENERAL

This Section applies to transportation services provided under Rate Schedules FTS, FFTS, FTS-G, DSS and ITS, to storage services provided under Rate Schedules PALS, BESS, DSS, FRSS and NSS and to line pack service under Rate Schedules IBS and LPS. Except for the no-notice aspect of service under Rate Schedules DSS, FTS-G/NB, FFTS-NB, FRSS and FTS-NB, each Shipper has the obligation to ensure actual volumes delivered to Natural at Receipt Points and actual volumes taken from Natural at Delivery Points conform to the volumes nominated by the Shipper and confirmed by Natural each day.

12.2 BALANCING SERVICE AND OVERRUN CHARGES DURING NON-CRITICAL TIMES

- (a) During periods when a Critical Time is not in effect, if actual receipts, deliveries, injections or withdrawals under an Agreement at a point on any day do not conform to a Shipper's rights under its Agreements or to the sum of its confirmed nominations and available no-notice rights, after receipts, deliveries, injections and withdrawals are allocated consistent with Section 10 or 11 of these General Terms and Conditions, as applicable, the following Overrun Charges and Balancing Service Charges shall apply (except as provided in Section 12.6):
- (1) In the event that receipts, deliveries, injections or withdrawals allocated to a Shipper under said Section 10 or 11 at any point or under any Agreement exceed Shipper's available firm contractual rights (MDQ, MSV, IQ or WQ), an Overrun Charge shall apply. Specifically, the Overrun Charges set out herein shall apply to the following: volumes taken or tendered in excess of MDQ under Rate Schedule FTS, FFTS or FTS-G; volumes taken or tendered in excess of MDQ, MSV, IQ or WQ or volumes taken in excess of Shipper's maximum withdrawal rights in any fifteen (15) day period under Rate Schedule DSS; and volumes taken or tendered in excess of MDQ, MSV, IQ or WQ under Rate Schedule NSS; volumes taken or tendered in excess of MDQ, IQ, WQ or MSWQ under Rate Schedule FRSS; and volumes taken or tendered in excess of Shipper's maximum daily access volumes or maximum allowed cumulative IBS Balance under Rate Schedule IBS; and volumes taken or tendered in excess of a Shipper's rights under an LPS Agreement or in any way inconsistent with an LPS RO. In addition, if a Shipper fails to satisfy the maximum or minimum monthly withdrawal obligation in Rate Schedule DSS, an Overrun Charge

FERC Docket: RP08-319-000

Original Sheet No. 376 original Sheet No. 376: Pending

GENERAL TERMS AND CONDITIONS

hereunder shall apply to the volume by which Shipper failed to satisfy the obligation. Overrun Charges applicable to Rate Schedules ITS, IBS and BESS shall be separately stated in those Rate Schedules; provided, however, that overrun rates for Rate Schedules ITS, IBS and BESS may be discounted on a non-discriminatory basis to any level between zero and the stated rate. Overrun volumes shall be deemed authorized overrun if tendered or taken pursuant to confirmed nominations except that any non-compliance with monthly maximum or minimum withdrawal rights or maximum withdrawal rights over any fifteen (15) day period shall be unauthorized overrun absent a specific waiver by Natural of the requirements. In the case of authorized overrun, as reflected in overrun requests properly submitted by Shipper and accepted by Natural (i.e., a confirmed nomination), the applicable maximum Overrun Charge for transportation shall be a maximum Authorized Overrun Rate determined on the same basis as the maximum rate for Rate Schedule ITS (i.e., the maximum Authorized Overrun Rate for Rate Schedule FTS and FFTS would equal the maximum rate for Rate Schedule ITS) multiplied by the number of Dth of the authorized overrun gas; provided, however, that the maximum Authorized Overrun Rate for Rate Schedule FTS-G would be the applicable maximum commodity rate for Rate Schedule FTS-G. For storage and line pack services, the Authorized Overrun Rate shall be as stated on the corresponding rate sheet in this Tariff. Such Authorized Overrun Rates may be discounted on a nondiscriminatory basis to any level between zero and the maximum Authorized Overrun Rate. In the case of unauthorized overrun, the Shipper shall pay Natural the applicable authorized overrun rate and, in addition, an unauthorized overrun charge equal to the volume of the overrun multiplied by an Unauthorized Overrun Rate. The maximum Unauthorized Overrun Rate is \$10/Dth, which rate may be discounted to any level between zero and the maximum rate so calculated. Any charges for an unauthorized overrun in excess of the authorized overrun charge shall be waived by Natural if the unauthorized overrun does not cause operational problems. Only a single Overrun Charge shall apply to any volume even though it may represent a overrun of more than one restriction under the applicable Rate Schedule (such as MDQ and MSV). Cashout shall apply to overrun in the event such an overrun would otherwise result in a storage withdrawal which would reduce Shipper's storage account(s) below zero. Notwithstanding the foregoing, no Overrun Charge shall apply to an authorized overrun pursuant to a confirmed nomination under Section 5.5(b) of Rate Schedule DSS or Section 5.5(b) of Rate Schedule NSS.

FERC Docket: RP08-319-000

Original Sheet No. 377 Original Sheet No. 377: Pending

GENERAL TERMS AND CONDITIONS

(2) If the volumes allocated to any FTS, FFTS, FTS-G, LPS or ITS Agreement fail to equal the confirmed nomination under such Agreement or if volumes allocated to any DSS, NSS, FRSS, FTS-G/NB, FFTS-NB, FTS-NB or PALS Agreement are inconsistent with confirmed nominations and/or available no-notice rights, as applicable, the following charges [in addition to any charges applicable under subsection (1)] shall apply, subject to the availability of balancing service under Rate Schedule IBS consistent with its terms, based on the degree of variance between actual, deliveries, injections or withdrawals and Shipper's rights and/or confirmed nominations (no charge hereunder shall apply for variances at Receipt Points):

VARIANCE	CHARGE
0% to 5%	No additional charges
5% to 10%	\$0.10/Dth
10% to 20%	\$0.20/Dth
20% to 50%	\$0.30/Dth
Above 50%	\$0.50/Dth

Such charges shall also apply as specified in Section 6(d) of Rate Schedule IBS.

- (3) Natural may discount or waive charges under subsection (2) on a basis which is not unduly discriminatory.
- (4) Tiered Balancing Service Charges under Sections 12.2, 12.3 and 12.4 shall be applied on a graduated basis, i.e., the specified charge shall apply only to that portion of the variance which is within the corresponding tier between its lower and upper percentage boundaries (variances within a range greater than the lower boundary and equal to or less than the upper boundary) and not to any portion of the variance falling within other tiers.
- (b) A Shipper subject to any Overrun Charges and/or Balancing Service Charge(s) under subsection (a) shall, in addition, pay an applicable transportation commodity charge on takes from Natural in excess of applicable rights and shall be obligated to return any volumes taken from Natural in excess of volumes delivered to Natural which are not accommodated through storage withdrawals from such Shipper's storage account(s). The volume of any imbalance remaining at the end of the month will be cashed out under Section 13 of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 378 Original Sheet No. 378: Pending

GENERAL TERMS AND CONDITIONS

12.3 BALANCING CHARGES WITH OPERATIONAL FLOW ORDERS

(a) On any day when an Operational Flow Order (but not a Critical Time) is in effect, if actual receipts, deliveries, injections or withdrawals allocated to a Shipper at any point or under any Agreement do not conform to the sum of such Shipper's confirmed nominations and no-notice rights applicable to such point and/or Agreement, Balancing Service Charges will be assessed on such variances that are detrimental to Natural's system, based on the conditions described in or giving rise to the Operational Flow Order, except as provided in Section 12.6. Notwithstanding the foregoing, Balancing Service Charges shall not be assessed for variances at Receipt Points unless explicitly stated in the Operational Flow Order. The Operational Flow Order shall also state whether the charges are to be based on variances at individual points, by zone or by other aggregation. Balancing Service Charges during this period will be as follows, based on the variance between actual receipts, deliveries, injections or withdrawals and the applicable confirmed nominations and/or no-notice rights:

VARIAN	ICE	CHARGE						
0% to 3% to				nal charges \$1.00/Dth		50	% of i	AMIP
10% to 20% to	50%	Greater	of	\$2.00/Dth \$4.00/Dth	or	2	times	AMIP
Above	50%	Greater	of	\$8.00/Dth	or	4	times	AMIP

(b) Shipper shall in addition pay all applicable charges specified in Sections 12.2(a)(1) and 12.2(b). Where Balancing Service Charges apply under this Section 12.3, they shall be in lieu of Balancing Service Charges under Section 12.2(a)(2), but the charges identified in the prior sentence shall still apply.

12.4 CHARGES DURING CRITICAL TIMES

(a) On any day when a Critical Time is in effect, if actual receipts, injections, withdrawals and/or actual deliveries allocated to Shipper at any point or under any Agreement for services provided by Natural under the Commission's Regulations at 18 C.F.R. Part 284 do not conform to the sum of confirmed nominated volumes and no-notice rights applicable to such point or Agreement,

FERC Docket: RP08-319-000

Original Sheet No. 379 Original Sheet No. 379: Pending

GENERAL TERMS AND CONDITIONS

Balancing Service Charges will be assessed for the variances that are to the detriment of Natural's system, except as provided in Section 12.6. Balancing Service Charges in effect during a Critical Time will be as follows:

VARIANCE	1		CHARGE
0% to 3%	Greate:	of of	\$4.00/Dth or 2 times AMIP
3% to 10)% Greate:	of	\$12.00/Dth or 6 times AMIP
10% to 20)% Greate:	of	\$40.00/Dth or 20 times AMIP
20% to 50)% Greate:	of	\$80.00/Dth or 40 times AMIP
Above 50)% Greate:	of	\$200.00/Dth or 100 times AMIP

- (b) Shipper shall in addition pay all applicable charges specified in Sections 12.2(a)(1) and 12.2(b). Charges hereunder shall be in lieu of any otherwise applicable Balancing Service Charges under Sections 12.2(a)(2) and 12.3, but the charges identified in the prior sentence shall still apply.
- (c) Any imbalance created during a Critical Time that is not eliminated before the end of the month will be subject to the cashout and offsetting provisions of Section 13; provided, however, that any remaining imbalance created during a Critical Time which helped the system will be cashed out at 100% of the AMIP.

12.5 DECLARATION OF A CRITICAL TIME

(a) A Critical Time may be declared (1) when the total physical receipts to all or a portion of the system are approaching or expected to approach, a level that is in excess of the total physical deliveries plus the maximum volumes of gas Natural is able to inject into Natural's storage without jeopardizing the integrity of Natural's storage facilities; (2) when the total physical deliveries from all or a portion of the system are approaching or expected to approach, a level that is in excess of the total physical receipts and the maximum volume of gas available to be withdrawn from Natural's storage without jeopardizing the integrity of Natural's storage facilities; (3) when system pressure on one or more pipeline segments is falling and approaching a level or is expected to fall and approach a level that is at or below the minimum that Natural considers necessary for system integrity or to fulfill its firm contractual obligations; (4) when system pressure on one or more pipeline

FERC Docket: RP08-319-000

Original Sheet No. 380 Original Sheet No. 380: Pending

GENERAL TERMS AND CONDITIONS

segments is rising and approaching a level or is expected to rise and approach a level that is at or above the maximum pressure at which Natural's facilities may be safely operated; or (5) at other times when Natural is unable or anticipates it may not be able to fulfill its firm contractual obligations or otherwise when necessary to maintain the overall operational integrity of all or a portion of Natural's system. A Critical Time may not be declared on all or a portion of the system for the purpose of maintaining interruptible services on that portion of the system, but interruptible gas may flow at times or on portions of the system when such flow would not violate any operational control restrictions or provisions of this Tariff. Natural shall notify Shippers on the Informational Postings portion of its Interactive Website as to the reason(s) why a Critical Time was declared.

(b) Notice of a Critical Time will be provided through Electronic Notice Delivery consistent with GISB Standards as adopted in Section 48 of these General Terms and Conditions and will be posted on the Informational Postings portion of Natural's Interactive Website. Natural will endeavor to provide and post the notice before 4:00 p.m. Central Time or otherwise will endeavor to notify Shippers by 4:00 p.m. Central Time that they should check the Informational Postings portion of Natural's Interactive Website again at a specified later time to see whether a Critical Time will be in effect for the next day. In addition, Natural must attempt to give actual notice of a Critical Time via telefax or telephone (provided a Shipper has given the numbers to Natural as required in Section 23) at least four (4) hours prior to the start of the day before a Critical Time will be effective as to a Shipper. Such notice shall specify the anticipated duration of the Critical Time and whether the Critical Time Balancing Service Charges will apply to over-receipts and under-deliveries vis-a-vis confirmed nominations or to under-receipts and over-deliveries vis-a-vis confirmed nominations. A Critical Time will be effective at the start of a day and will continue until the end of the day and through the end of successive days until Natural notifies Shippers via Natural's Interactive Website that there is no longer a Critical Time.

12.6 EXEMPTIONS

Balancing Service Charges at a point will not be assessed at a Receipt or Delivery Point where the variance between total confirmed nominations and actual gas flows at that point is less than 1,000 Dth per day or where Natural does not have either daily metering or access to such information. This exemption is applicable during both non-Critical and Critical Times; provided, however, that this exemption shall not apply to the extent specified in Section 6(d) of Rate Schedule IBS. In addition, any Receipt or Delivery Point where the variance between the total confirmed nominations and actual flows is less than 1,000 Dth per day or where Natural does not have either daily metering or access

FERC Docket: RP08-319-000

Original Sheet No. 381 Original Sheet No. 381: Pending

GENERAL TERMS AND CONDITIONS

to such information shall be exempt at all times, including during a Critical Time, from Operational Flow Orders requiring that gas flows be conformed to nominations or to some percentage of nominations. If the variance at a point (including a Central Delivery Point) exceeds 1,000 Dth per day, where Natural does have daily metering or access to such information, the Balancing Service Charges shall apply to the entire variance (in excess of any other applicable tolerance) except as set out in the following sentence. A point operator shall be exempted during both non-Critical and Critical Times from Balancing Service Charges for any variance which is less than 1,000 Dth per day (e.g., the Balancing Service Charges shall apply only to that portion of any variance which is in excess of 1,000 Dth per day); provided, however, that the 1,000 Dth per day shall be included in, and shall not be in addition to the point operator's account under Section 11.3 of these General Terms and Conditions except to the extent the 1,000 Dth per day exceeds the otherwise applicable allowance for a point operator account, in which latter case the point operator's account shall be increased to the level required to accommodate the 1,000 Dth per day, and except that the 1,000 Dth per day exemption applies during Critical Times.

12.7 BILLING

- (a) In the event Natural does not have all actual information by the time the billings are to be sent, Natural may bill Balancing Service Charges applicable to non-Critical Times based on reasonable estimates. Such non-Critical Time Balancing Service Charges will be assessed or revised as a result of the receipt or revision of actual information.
- (b) During a Critical Time, Natural and Shippers shall rely on Natural's operational data posted on Natural's Interactive Website to determine whether a Critical Time Balancing Service Charge is applicable; provided, however, if a Shipper is convinced that Natural's operational data is erroneous and promptly (by the end of the next business day following the posting) communicates the error to Natural, Natural and the Shipper shall work to resolve the discrepancy and determine the appropriate measure of the Shipper's compliance with applicable tolerances.
- (c) For a point operator under Section 11.3 of these General Terms and Conditions, Balancing Service Charges shall be applied to point operator's account as though point operator were a Shipper, with percentage variances measured in relationship to total confirmed nominations and delivery rights under Applicable Agreements at the Delivery Point.

FERC Docket: RP08-319-000

Original Sheet No. 382 Original Sheet No. 382: Pending

GENERAL TERMS AND CONDITIONS

- (d) No additional Balancing Service Charges will be assessed when a prior period adjustment applied to the current month would otherwise cause or increase the Balancing Service Charges to be billed.
- 12.8 All amounts collected by Natural for penalties commencing with December 1, 2003, the initial effective date of Natural's Order No. 637 Compliance Plan, shall be determined for each semi-annual period ending June 30 and December 31 and distributed, through a credit to current billings wherever feasible, within ninety (90) days after each such June 30 and December 31 which occurs after the initial effective date of Natural's Order No. 637 Compliance Plan, with the first such distribution being made within ninety (90) days after June 30, 2004. For purposes of this Section 12.8, penalties shall include: Unauthorized Overrun Charges, Balancing Service Charges, charges for failure to comply with an Operational Flow Order (including during a Critical Time), charges for cashout of imbalances, and gas forfeited to or amounts collected by Natural under Section 9.3 of Rate Schedule LPS. Such distribution shall be made as follows:
- These amounts shall be used first to compensate (a) Natural for any cashout expenses and for any extraordinary out-ofpocket costs it has incurred (including any compensation Natural agreed to provide for voluntary actions and any storage injection or withdrawal costs not collected because of the waiver of the injection and withdrawal charges as described in Section 23.7) to alleviate the conditions which were created by the imbalance or the Unauthorized Overrun or other violation or which resulted in the issuance of an Operational Flow Order or the declaration of Critical Time or other operational action taken by Natural under Section 23 of these General Terms and Conditions. Costs that may be netted against penalty revenue may include only actual, verifiable and prudent incremental costs incurred to resolve the reliability concerns actually caused by the particular party or parties against which the penalty has been assessed and must relate to a circumstance which resulted in the penalty revenue to be credited under this provision. If these amounts are not adequate to reimburse Natural for cashout expenses, the unreimbursed cashout expenses shall be carried forward to future years until recouped.
- (b) Any remaining amounts will be refunded pro rata to all Shippers (both firm and interruptible), except as provided below, through a credit to current billings wherever feasible, based on the ratio of a Shipper's total base reservation and commodity charges paid during the relevant 6-month period to all such charges paid during such period; provided, however, that such calculation shall exclude charges

FERC Docket: RP08-319-000

Original Sheet No. 383 Original Sheet No. 383: Pending

GENERAL TERMS AND CONDITIONS

paid by any Shipper during any month in which such Shipper failed to comply with an Operational Flow Order or had an Unauthorized Overrun. Natural shall also distribute interest on the net penalty revenue distributed from the date such penalty revenue is actually received by Natural to the date of distribution, calculated in accordance with 18 C.F.R. Section 154.501. While distribution is made every six months, the net penalty revenue credit shall be calculated on a monthly basis. A Shipper which failed to comply with an Operational Flow Order or had an Unauthorized Overrun shall be excluded from distribution of net penalty revenues only for the month in which that violation occurred. Where capacity has been released, any amounts distributed hereunder with respect to the released capacity shall be distributed to the Original Shipper and the calculation and amounts distributed to the Original Shipper shall be based on the gross base commodity and reservation charges paid by that Shipper (without considering any credits from Replacement Shippers) during the relevant 6 month period; provided, however, that in the case of a permanent release, any amounts distributed hereunder with respect to the released capacity shall be distributed to the Replacement Shipper and the calculation and the amounts distributed to the Replacement Shipper shall be based on the base commodity and reservation charges paid by that Replacement Shipper during the relevant 6 month period.

FERC Docket: RP08-319-000

Original Sheet No. 384 Original Sheet No. 384: Pending

GENERAL TERMS AND CONDITIONS

13. IMBALANCES

13.1 RESPONSIBILITY FOR BALANCING

In addition to delivering and receiving volumes of gas in conformance with nominations, Shippers are responsible for conforming their takes at Delivery Points with their deliveries to Natural at Receipt Points each day. Except for the no-notice aspect of Rate Schedules DSS, FRSS, FTS-NB, FFTS-NB and FTS-G/NB, and except as otherwise provided under an IBS Agreement, Natural has no obligation to deliver for the account of a Shipper more volumes of gas than Natural has received for the account of the Shipper or to accept for the account of the Shipper more volumes of gas than are being delivered for the account of the Shipper on any day.

13.2 MONTHLY IMBALANCES, NETTING AND OFFSETTING

At the end of each calendar month, to the extent the net receipts (with the appropriate deductions for Fuel Gas and Gas Lost and Unaccounted For and after giving appropriate recognition to storage activity subject to applicable storage rights) do not equal the deliveries under an Agreement on a Dth basis, and except that any IBS Balance under an IBS Agreement is only subject to the provisions of this Section 13 to the extent provided in Section 7(b) of Rate Schedule IBS, the following netting and offsetting procedures will apply:

- (a) Imbalances under a Shipper's different Agreements will be netted together to obtain the Shipper's Total Monthly Imbalance. The Total Monthly Imbalance will be shown with the monthly billings sent to Shippers.
- (b) To assist Shippers in arranging offsets, Natural will post on its Interactive Website the Total Monthly Imbalance of any Shipper which has notified Natural that it has elected to have such information posted. Notification by the Shipper may be in writing or on Natural's Interactive Website and shall be effective by 8:00 a.m. on the next Business Day (Central Clock Time) if the notification is received by 11:45 a.m. on a Business Day. Imbalance information authorized for posting through such notification shall be posted no later than the ninth Business Day of the month after the imbalance occurred. Shippers shall have the ability to post and trade imbalances, and imbalance information shall remain posted, until the seventeenth Business Day of the month after the imbalance occurred.

FERC Docket: RP08-319-000

Original Sheet No. 385 Original Sheet No. 385: Pending

GENERAL TERMS AND CONDITIONS

(c) Natural shall enable the imbalance trading process by:

- (1) Receiving the Request for Imbalance Trade,
- (2) Receiving the Imbalance Trade Confirmation,
- (3) Sending the Imbalance Trade Notification, and
- $\mbox{\ensuremath{(4)}}$ Reflecting the trade prior to or on the next monthly Shipper Imbalance or cashout.
- (d) Imbalance trades can only be withdrawn by the initiating trader and only prior to the confirming trader's confirmation of the trade. Imbalance trades are considered final when confirmed by the confirming trader and effectuated by Natural.
- (e) After receipt of an Imbalance Trade Confirmation, Natural shall send the Imbalance Trade Notification to the initiating trader and the confirming trader no later than noon (Central Clock Time) the next Business Day.
- (f) Shipper imbalances remaining after the imbalance netting and trading procedures set out in subsections (a)-(e) shall be cashed out as described in Section 13.3 of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 386 Original Sheet No. 386: Pending

GENERAL TERMS AND CONDITIONS

13.3 CASHOUT PROCEDURES

(a) Any imbalance remaining will be cashed out on a tiered basis pursuant to the following schedule:

IMBALANCE LEVEL	OVERAGE (Natural pays Shipper)	UNDERAGE (Shipper pays Natural)
0% to 5%	100% x AMIP	100% x AMIP
Greater than 5% to 10%	90% x AMIP	110% x AMIP
Greater than 10% to 15%	80% x AMIP	120% x AMIP
Greater than 15% to 20%	70% x AMIP	130% x AMIP
Greater than 20%	60% x AMIP	140% x AMIP

Notwithstanding the foregoing, any imbalances created during a Critical Time to the benefit of the system that are remaining at the end of the month will be separately cashed out at 100% of the AMIP.

- (b) Following any offsetting with other Shippers, a Shipper's remaining imbalance will be cashed out based on the percentage of that imbalance compared to the total deliveries for that Shipper during the month. For example, if the Total Deliveries were 1,000 Dth and the remaining underage imbalance after offsetting with other Shippers was 100 Dth, the total Imbalance Level would be 10%. The first 5% (50 Dth) would be cashed out at 100% of the AMIP and the remaining 50 Dth would be cashed out at 110% of the AMIP.
- (c) The Average Monthly Index Price (AMIP) is the arithmetic average of the Weekly Index Prices (WIPs). The WIP for any week is the arithmetic average of the following three index prices reported in 'Gas Price Report' issued by "Natural Gas Week":
- $\qquad \qquad \text{(1)} \quad \text{Louisiana, Gulf Coast, Onshore, Spot Delivered} \\ \text{to Pipeline.}$
- $\hspace{1.5cm} \hbox{(2)} \hspace{0.3cm} \hbox{Texas, Gulf Coast, Onshore, Spot Delivered to Pipeline.} \\$

FERC Docket: RP08-319-000

Original Sheet No. 387 Original Sheet No. 387: Pending

GENERAL TERMS AND CONDITIONS

(3) Midcontinent, Spot Delivered to Pipeline.

In calculating the AMIP, the WIPs will be based on the prices reported in the issue of "Natural Gas Week" dated on or after Natural's nomination deadline for first of the month service for that month, and the subsequent issues dated prior to Natural's nomination deadline for the following month's first of the month service

- (d) Following the ten (10) day period for offsetting imbalances, Shippers with remaining imbalances shall pay Natural or will be credited with the appropriate cashout amounts.
- (e) In the event "Natural Gas Week" ceases to publish entirely or fails to publish the index prices listed in subsection (c) above, the following procedures shall apply in determining a month's AMIP:
- (1) Should, in any given week, "Natural Gas Week" fail to publish one of the three index prices used in determining that week's WIP, the WIP will be determined using the average of the remaining two published index prices.
- (2) Should, in any given week, "Natural Gas Week" fail to publish two or more of the three index prices used in determining that week's WIP, there will be no WIP for that week used in determining the month's AMIP.
- (3) Should, in a given month, there be less than two WIP's available for the AMIP calculation, the following alternate AMIP procedures will apply: The AMIP will be defined as the arithmetic average of:
- (i) the closing price for the NYMEX natural gas futures contract applicable to the month in which the imbalance was created (i.e., the price at which that month's contract "went off the board"); and
- (ii) the individual daily closing prices for the following month ("spot month" or "near month") NYMEX natural gas contract during the month in which the imbalance was created, up to and including the day the "spot month" contract "goes off the board."

FERC Docket: RP08-319-000

Original Sheet No. 388 Original Sheet No. 388: Pending

GENERAL TERMS AND CONDITIONS

13.4 OPERATIONAL DATA VS. ACTUALS

In determining the cashout tier applicable under Section 13.3 above, Natural will utilize the operational data posted on its Interactive Website as of the end of the month or the actual flow volumes (or, if actual flow volumes are not available at the time of billing, the reasonable estimates), whichever results in a lower cashout tier.

13.5 PRIOR PERIOD ADJUSTMENTS

Any imbalances for a month that are booked after the transportation for that month has been billed as a result of receiving actual or corrected flow information will be cashed out at 100% of the AMIP in effect during the month the imbalance occurred.

13.6 IMBALANCES THAT OCCURRED PRIOR TO DECEMBER 1, 1993

Notwithstanding anything to the contrary in this Section 13, any imbalances under Shipper's effective or terminated FTS or ITS Agreement(s) applicable to the period prior to December 1, 1993 that have not been made up by that date, will be cashed out as follows:

- (a) All overages (where Natural owes gas to Shipper) will be netted against underages (where Shipper owes gas to Natural) under all of Shipper's transportation agreements with Natural (including transportation and exchange agreements filed in Volume No. 2 of Natural's FERC Gas Tariff). In netting overages with underages, the oldest monthly imbalances will be netted first, so that any remaining imbalance after netting will be attributable to the most recent months in which the overages or underages occurred.
- (b) Any remaining imbalances under the FTS and ITS Agreements will then be cashed out at one hundred percent (100%) of the AMIP applicable to the month in which the imbalance occurred.
- (c) Natural shall credit Shipper's account or shall bill Shipper the appropriate amount for the remaining imbalance by March 15, 1994, except for that portion of the remaining imbalance which is in dispute. For a portion of the remaining imbalance to be considered to be in dispute, the Shipper must have submitted a

FERC Docket: RP08-319-000

Original Sheet No. 389 Original Sheet No. 389: Pending

GENERAL TERMS AND CONDITIONS

written notice prior to March 15, 1994 specifying the volume in dispute along with documentation which supports the Shipper's position, such as allocation statements from the operator of the facilities at which Natural received or delivered the volume being Disputed. Within sixty (60) days Natural and Shipper will meet and/or exchange any other documentation as necessary to resolve the dispute. If the dispute has not been resolved by the end of those sixty (60) days, either party may thereafter submit written notice to the other that any remaining disputed amount shall be resolved utilizing the following arbitration procedures:

(1) Natural and Shipper shall have thirty (30) days from the date of the notice to agree on an arbitrator employed by one of the following accounting firms:

Arthur Andersen & Co. Cooper's & Lybrand Deloitte & Touche Ernst & Young KPMG Peat Marwick Price Waterhouse & Co.

If Natural and Shipper cannot agree, a Federal district court judge from the Northern District of Illinois will be asked to select the arbitrator from one of the above listed accounting firms. The arbitrator selected shall not be employed by a firm that represents either Natural or Shipper as its independent auditor for financial reporting purposes, shall not be financially interested in the outcome of the dispute, and shall not be a current or former employee of Natural or Shipper. The arbitrator shall be qualified by education and/or experience to pass on the particular issues in dispute. Natural and Shipper shall share equally in the compensation and expenses of the arbitrator. Each party shall be responsible for its own expenses in preparing and presenting evidence before the arbitrator.

(2) Natural and Shipper shall furnish the arbitrator with the quantity that each believes represents the appropriate remaining imbalance that is still in dispute, along with all supporting documentation. The arbitrator shall meet and hear the parties with respect to matters relevant to the dispute. The jurisdiction of the arbitrator shall be limited to the selection, based on all relevant information presented, of either the remaining imbalance volume submitted by Shipper or the

FERC Docket: RP08-319-000

Original Sheet No. 390 Original Sheet No. 390 : Pending

GENERAL TERMS AND CONDITIONS

remaining imbalance volume submitted by Natural. The decision of the arbitrator shall be in writing and such decision shall be issued within ninety (90) days of the appointment of such arbitrator. The results of such arbitration shall be final, binding and non-appealable, and may be enforced in any court of law or by the Federal Energy Regulatory Commission.

(d) Any imbalances under a Shipper's FTS or ITS Agreements for periods prior to December 1, 1993 that have not been cashed out by March 15, 1994 because of adjustments to volumes that are booked after March 15, 1994 shall be cashed out as described above and reflected on the next billing after the adjustment is booked.

FERC Docket: RP08-319-000

Original Sheet No. 391 Original Sheet No. 391: Pending

GENERAL TERMS AND CONDITIONS

14. UNAUTHORIZED DELIVERY OF GAS

14.1 DEFINITION

"Unauthorized Gas" shall mean any volume of gas delivered to Natural that Natural cannot identify as being delivered pursuant to a transportation nomination on behalf of an existing Shipper.

14.2 APPLICABILITY NOTICE AND CLAIMS

All Unauthorized Gas shall be subject to this Section. Natural shall give notice in writing to the owner/Shipper of the Unauthorized Gas (if known) and to the operator of the applicable supply point where the Unauthorized Gas was received, that said Unauthorized Gas is subject to these provisions and will be posted on the Informational Postings portion of Natural's Interactive Website. In addition, on the first business day following the fifteenth of each month, Natural shall post on the Informational Postings portion of Natural's Interactive Website the volume, date first received on Natural's system, and point of receipt of any Unauthorized Gas received ("Notice"). Natural will continue to post the Notice on the Informational Postings portion of Natural's Interactive Website until a Valid Claim has been submitted or until one hundred twenty (120) days after the initial posting of the Notice. In order to be a Valid Claim for purposes hereof, a claim must: (a) be provided to Natural in writing; (b) identify the specific Unauthorized Gas delivered; (c) provide independent evidence of ownership of the Unauthorized Gas claimed; and (d) agree to indemnify Natural fully with respect to any adverse claims to ownership of the gas or to the proceeds resulting from the sale thereof. In addition, the Unauthorized Gas must be nominated to an effective transportation Agreement with Natural that contains the Receipt Point at which the Unauthorized Gas was received ("Valid Nomination") in order to prevent the accrual of further penalties under Section 14.3 hereof.

14.3 TREATMENT OF UNAUTHORIZED GAS

(a) If a Valid Claim and Valid Nomination are submitted within three (3) business days of the posting of Notice, no penalty shall be assessed under this Section.

FERC Docket: RP08-319-000

Original Sheet No. 392 Original Sheet No. 392: Pending

GENERAL TERMS AND CONDITIONS

(b) If a Valid Claim and Valid Nomination are submitted after three (3) business days of the posting of Notice but within sixty (60) calendar days of the posting of Notice, Natural shall assess a penalty of fifty cents (\$.50) per Dth upon that Shipper transporting the Unauthorized Gas after nomination to a transportation Agreement with Natural.

- (c) If a Valid Claim and Valid Nomination are submitted after sixty (60) calendar days of the posting of Notice, Natural shall assess a penalty of one dollar (\$1.00) per Dth upon that Shipper transporting the Unauthorized Gas after nomination to a transportation Agreement with Natural.
- (d) If a Valid Claim and Valid Nomination are not submitted within one hundred twenty (120) calendar days of Notice, Natural shall retain the gas without any payment required of Natural.
- (e) For accounting purposes, the gas retained shall be debited to Account 483, Sales for Resale, with a contra entry to Account 232, Refunds Due Customers. The debit shall be valued at a price equal to the lower of the Market Index Price for the month that the Unauthorized Gas was received or the Market Index Price for the month one hundred twenty (120) days after Notice. The Market Index Price shall be the arithmetic average of the prices for spot gas delivered to interstate pipelines, as reported in the first issue published on or after the fourth day of the month, for the following publications and geographical areas:

Publication

Regions

* Inside FERC's Gas Market Report
"Prices of Spot Gas Delivered
Delivered to Pipelines"

- * Oklahoma
- * Kansas
- * Texas
- * Louisiana * Offshore
- * Offshore * Anadarko Basin
- * Permian Basin
- * Those Stations or Zones located in any of the above listed areas.

FERC Docket: RP08-319-000

Original Sheet No. 393 Original Sheet No. 393: Pending

GENERAL TERMS AND CONDITIONS

Publication

Regions

- - * Southern Texas
 - * Texas Panhandle/Anadarko Basin/Oklahoma
 - * South Louisiana
- * Natural Gas Week
 - "Spot Prices on Natural
 Pipeline Systems Delivered-to-Pipeline" Delivered-to-Pipeline"
- * Louisiana
- * Offshore Louisiana
- * Oklahoma * Texas * Kansas

In computing the Market Index Price, the arithmetic average of the prices of spot gas delivered to interstate pipelines shall be first calculated for each publication. A composite price shall then be computed by averaging prices for the three reporting services. In the event that one of the indices described above is not published for the month in question, Natural shall substitute another published index that fairly represents gas prices within the region(s) in question.

Amounts due customers for Unauthorized Gas retained pursuant to Section 14.3(d) above shall be distributed to Natural's jurisdictional transportation customers. The distribution shall be by credit to each Shipper's monthly bill for the month in which the Unauthorized Gas is retained in a pro rata amount equal to each Shipper's volume transported for that month to the total of all volumes transported for that month times the value assigned to retained gas pursuant to Section 14.3(e) less any production tax incurred by reason of such retention.

FERC Docket: RP08-319-000

Original Sheet No. 394 Original Sheet No. 394: Pending

GENERAL TERMS AND CONDITIONS

15. STATEMENTS, BILLING, PAYMENT, PENALTIES AND DISCOUNTING POLICY

15.1 STATEMENT AND INVOICES

Natural shall, on or before the ninth (9th) Business Day of each month, render to Shipper a bill or bills for service under each applicable Rate Schedule during the preceding month. As used in this Section 15, "render" is defined as postmarked, time-stamped and delivered to the designated site. Invoices will be based on actuals (if available) or best available data. Quantities at points where OBAs exist will be invoiced based on scheduled quantities.

15.2 SHIPPER INFORMATION

If information is required from Shipper, or its designee, to actualize volumes or allocations, Shipper shall furnish the required information, or cause it to be furnished, to Natural, on or before the tenth (10th) day of each month.

15.3 IMBALANCE STATEMENT

Imbalance statements will be generated at the same time or prior to the generation of the invoice. Prior to or with the above-required invoice for billing, Natural shall render the gas imbalance statement which details in Dth the gas received and delivered each month at the Receipt and Delivery Point(s) hereunder based on the best information available.

15.4 PAYMENT

Shipper shall pay to Natural at the address indicated on the invoice or, if directed by Natural, by wire transfer to a bank designated by Natural, the amount due Natural for services provided pursuant to an Agreement during the appropriate calendar month as reflected in the billing described above, within ten (10) calendar days after the date of receipt of such billing. For purposes of this Section, the bill is deemed to be received by Shipper three (3) days after the postmark date if sent by mail. The invoice number should be identified on all payments and the Shipper should submit supporting documentation identifying what is being paid. Natural shall apply payment per such supporting documentation. If payment differs from the invoiced amount, remittance detail should be

FERC Docket: RP08-319-000

Original Sheet No. 395 Original Sheet No. 395: Pending

GENERAL TERMS AND CONDITIONS

provided with the payment except when payment is made by electronic funds transfer, in which case the remittance detail is due within two (2) Business Days of the payment due date. Should Shipper fail to pay any undisputed portion of any bill as herein provided when such amount is due, interest on the unpaid portion of the bill shall accrue at the maximum allowable interest permitted under the Commission's Regulations. For any amount to be considered "disputed," Shipper must provide appropriate documentation supporting and identifying the basis for the dispute. If Shipper fails to make payment in accordance with this Section, Natural may, in addition to any other remedy it may have under this Tariff or under commercial law: (1) suspend deliveries as provided in Section 5.11 of these General Terms and Conditions; and (2) offset such deficient payments against any payments, refunds or credits owed by Natural to Shipper.

15.5 ADJUSTMENT OF ERRORS

- (a) The time limitation for disputes of allocations should be six (6) months from the date of the initial month-end allocation with a three (3) month rebuttal period. This standard shall not apply in the case of deliberate omission or misrepresentation or mutual mistake of fact. Parties' other statutory or contractual rights shall not otherwise be diminished by this standard.
- (b) Prior period adjustment time limits should be six (6) months from the date of the initial transportation invoice and seven (7) months from the date of initial sales invoice with a three (3) month rebuttal period, excluding government-required rate changes. This standard shall not apply in the case of deliberate omission or misrepresentation or mutual mistake of fact. Parties' other statutory or contractual rights shall not otherwise be diminished by this standard.
- (c) In no event will any changes be made after twenty-four (24) months from the date of statements, billings or payment, based on actualized volumes, unless the parties mutually agree.
- (d) Any error discovered as a result of a timely claim shall be corrected within thirty (30) days of the determination thereof.

FERC Docket: RP08-319-000

Original Sheet No. 396 original Sheet No. 396: Pending

GENERAL TERMS AND CONDITIONS

15.6 DISCOUNTING POLICY

- (a) In determining whether and the extent to which Natural has discounted various elements of its rates under Agreements for the purposes of its FERC accounts and for determining collections under Sections 21, 38, 39 and 40 of these General Terms and Conditions, the sequence of discounting set out in this Section 15.6 shall apply. Nothing herein is intended to modify as between the parties to any Agreement the provision of such Agreement regarding the payment of various rate elements. Any step in the sequence set out in this Section 15.6 shall not apply to any Agreement to the extent such Agreement is exempt from the relevant charge [see, e.g., Sections 21.3(d)(2) and 38.2(c)(1) of these General Terms and Conditions].
- (b) Where the rate assessed by Natural on any service Agreement is less than the maximum rate applicable to that Agreement plus all surcharges applicable to such Agreement, then for the purposes stated above, Natural shall determine the rate elements covered by the rate actually assessed (and thus the elements which have been discounted) by the following procedure:
- (1) Natural shall first apply all revenue under a firm Agreement to the commodity rate (including the ACA Surcharge under Section 40 of these General Terms and Conditions up to the maximum level of the ACA Surcharge rate applicable to such Agreement), if any, applicable to such Agreement.
- (2) Natural shall next apply all revenue under a firm Agreement in excess of that accounted for under subsection (1) above to demand surcharges for the collection of gas supply realignment costs under Section 38 of these General Terms and Conditions, where applicable, up to the maximum surcharge rate applicable to such Agreement under said Section 38.
- (3) Natural shall next apply all revenue under a firm Agreement in excess of that accounted for under subsections (1) and (2) above to demand surcharges for the recovery of Account No. 858 costs under Section 21 of these General Terms and Conditions, where applicable, up to the maximum surcharge rate applicable to such Agreement under said Section 21.

FERC Docket: RP08-319-000

Original Sheet No. 397 Original Sheet No. 397: Pending

GENERAL TERMS AND CONDITIONS

- (4) Natural shall next apply all revenue under a firm Agreement in excess of that accounted for under subsections (1) through (3) above to base reservation charges up to the maximum reservation charge rate applicable to such Agreement.
- (c) Where the rate assessed by Natural on any interruptible Agreement is less than the maximum rate applicable to that Agreement plus all surcharges applicable to such Agreement, then for the purposes stated above, Natural shall determine the rate elements actually assessed (and thus, the elements which have been discounted) by the following procedure:
- (1) Natural shall first apply all revenue under an interruptible Agreement to the ACA Surcharge under Section 40 of these General Terms and Conditions up to the maximum level of the ACA Surcharge rate, if any, applicable to such Agreement.
- (2) Natural shall next apply all revenue under an interruptible Agreement in excess of that accounted for under subsection (1) above to the component of the interruptible rate for the collection of gas supply realignment costs under Section 38 of these General Terms and Conditions, where applicable, up to the maximum level of the component included in unit rates for that purpose and applicable to such Agreement under said Section 38.
- $\hbox{(3) Natural shall next apply all revenue under an interruptible Agreement in excess of that accounted for under subsections (1) and (2) above to the base commodity rate applicable to such Agreement.}$

FERC Docket: RP08-319-000

Original Sheet No. 398 Original Sheet No. 398: Pending

GENERAL TERMS AND CONDITIONS

16.EVALUATION OF CREDIT

In evaluating requests for service and for certain other purposes under this Tariff, including Section 5.12 of these General Terms and Conditions, Natural will perform a credit appraisal of Shipper.

(a) Acceptance of a Shipper's request for service and the continuation of service are contingent upon the Shipper satisfying creditworthiness requirement on an on-going basis. To determine creditworthiness, a credit appraisal shall be performed in accordance with the following criteria:

Natural shall apply consistent evaluation practices to all similarly situated Shippers to determine the Shipper's financial ability to perform the payment obligations due to Natural over the term of the requested or existing service agreement.

A Shipper will be deemed creditworthy if (i) its longterm unsecured debt securities are rated at least BBB- by Standard & Poor's Corporation ("S&P") and at least Baa3 by Moody's Investor Service ("Moody's") (provided, however, that if the Shipper's rating is at BBBor Baa3 and the short-term or long-term outlook is Negative, Natural may require further analysis as discussed below); and (ii) the sum of reservation fees, commodity fees and any other associated fees and charges for the contract term is less than 15% of Shipper's tangible net worth. For the purposes of this Section 16, the term "tangible net worth" shall mean for a corporation the sum of the capital stock, paidin capital in excess of par or stated value, and other free and clear equity reserve accounts less goodwill, patents, unamortized loan costs or restructuring costs, and other intangible assets. Only actual tangible assets are included in Natural's assessment of creditworthiness. In comparing the overall value of a Shipper's contract to tangible net worth for credit evaluation purposes, Natural will compare the net present value of the demand or reservation charge obligations under such contracts to Shipper's current tangible net worth. If a Shipper has multiple service agreements with Natural, then the total potential fees and charges of all such service agreements shall be considered in determining creditworthiness.

If Shipper does not meet the criteria described above, then Shipper may request that Natural evaluate its creditworthiness based upon the level of service requested relative to the Shipper's current and future ability to meet its obligations. Such credit appraisal shall be based upon Natural's evaluation of the following information and credit criteria:

FERC Docket: RP08-319-000

Original Sheet No. 399 Original Sheet No. 399: Pending

GENERAL TERMS AND CONDITIONS

- (1) S&P and Moody's opinions, watch alerts, and rating actions and reports, rating, opinions and other actions by Dun and Bradstreet and other credit reporting agencies will be considered in determining creditworthiness.
- (2) Consistent financial statement analysis will be applied by Natural to determine the acceptability of Shipper's current and future financial strength. Shipper's balance sheets, income statements, cash flow statements and auditor's notes will be analyzed along with key ratios and trends regarding liquidity, asset management, debt management, debt coverage, capital structure, operational efficiency and profitability.
- $\,$ (3) Results of bank and trade reference checks and credit reports must demonstrate that a Shipper is paying its obligations in a timely manner.
- (4) Shipper must not be operating under any chapter of the bankruptcy laws and must not be subject to liquidation or debt reduction procedures under state laws and there must not be pending any petition for involuntary bankruptcy. An exception may be made for a Shipper who is a debtor in possession operating under Chapter XI of the Federal Bankruptcy Act if Natural is assured that the service billing will be paid promptly as a cost of administration under the federal court's jurisdiction, based on a court order in effect, and if the Shipper is continuing and continues in the future actually to make payment.
- (5) Whether Shipper is subject to any lawsuits or judgments outstanding which could materially impact its ability to remain solvent.
- (6) Whether Shipper has or has had any delinquent balances outstanding for services provided previously by Natural and whether Shipper is paying and has paid its account balances according to the terms established in its service agreements (excluding amounts as to which there is a good faith dispute).
- (7) The nature of the Shipper's business and the effect on that business of general economic conditions and economic conditions specific to it, including Shipper's ability to recover the costs of Natural's services through filings with regulatory agencies or otherwise to pass on such costs to its customers.
- (8) Any other information, including any information provided by Shipper, that is relevant to Shipper's current and future financial strength and Shipper's ability to make full payment over the term of the contract.

FERC Docket: RP08-319-000

Original Sheet No. 400 Original Sheet No. 400: Pending

GENERAL TERMS AND CONDITIONS

Information which Natural may request to be provided by Shipper to Natural in connection with such a credit evaluation includes the following:

- (i) Audited Financial Statements;
- (ii) Annual Report;

(iii) Most recent filed statements with the Securities and Exchange Commission (or an equivalent authority) or other similar publicly available information;

(iv) For public entities, the most recent publicly available interim financial statements, with an attestation by its Chief Financial Officer, Controller, or equivalent (CFO) that such statements constitute a true, correct, and fair representation of the Shipper's financial condition prepared in accordance with Generally Accepted Accounting Principles (GAAP) or equivalent;

(v) For non-public entities, including those that are state-regulated utilities, the most recent available interim financial statements, with an attestation by its CFO that such statements constitute a true, correct, and fair representation of the Shipper's financial condition prepared in accordance with GAAP or equivalent;

(vi) For non-public entities, including those that are state-regulated utilities, an existing sworn filing, including the most recent available interim financial statements and annual financial reports filed with the respective regulatory authority, showing the Shipper's current financial condition;

(vii) For state-regulated utility local distribution companies, documentation from their respective state regulatory commission (or an equivalent authority) of an authorized gas supply cost recovery mechanism;

(viii) List of affiliates, parent companies, and

subsidiaries;

 $\mbox{(ix)} \quad \mbox{Publicly available credit reports from credit} \\ \mbox{and bond rating agencies;} \\$

(x) Private credit ratings, if obtained by the

 ${\tt Shipper;}$

(xi) Bank references;

FERC Docket: RP08-319-000

Original Sheet No. 401 Original Sheet No. 401: Pending

GENERAL TERMS AND CONDITIONS

(xii) Trade references;

(xiii) Statement of legal composition; and

(xiv) Statement of the length of time the business

has been in operation.

Upon receipt of any request from Natural for information to be used for creditworthiness evaluation, the Shipper's authorized representative(s) shall acknowledge receipt of Natural's request; provided, however, that Natural and the Shipper may mutually agree to waive this requirement. The Shipper's authorized representative(s) shall respond to Natural's request for credit information as allowed by this tariff, on or before the due date specified in the request or within five (5) Business Days, whichever is later. The Shipper shall provide all the credit information requested by Natural or provide the reason(s) why any of the requested information was not provided. Upon receipt from the Shipper of all credit information provided pursuant to this Section 16, Natural shall notify the Shipper's authorized representative(s) that it has received such information; provided, however, that Natural and the Shipper may mutually agree to waive this requirement. The Shipper shall designate up to two representatives who are authorized to receive notices regarding the Shipper's creditworthiness, including requests for additional information, pursuant to this Section 16 and provide to Natural the Internet e-mail addresses of such representatives prior to the initiation of service. Written requests and responses shall be provided via Internet E-mail, unless otherwise agreed to by the parties. In complying with the creditworthiness related notifications pursuant to this Section 16, the Shipper and Natural may mutually agree to other forms of communication in lieu of Internet E-mail notification. The obligation of Natural to provide creditworthiness notifications is waived until the above requirement regarding designation of representatives has been met. The Shipper is to manage internal distribution of any creditworthiness notices that are received. Natural shall designate, on its Interactive Website or in written notices to the Shipper, the Internet e-mail addresses of up to two representatives who are authorized to receive notices regarding the Shipper's creditworthiness. The Shipper's obligation to provide confirmation of receipt is met by sending such confirmation to such representatives and Natural is to manage internal distribution of any such confirmations. The provisions of this paragraph relating to representatives and notification also applies to any notice related to creditworthiness under Section 5.11, 5.12 or 19.1 of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 402 Original Sheet No. 402: Pending

GENERAL TERMS AND CONDITIONS

If Natural concludes that a Shipper is non-creditworthy, Natural shall provide written notice to Shipper within ten (10) days after that determination is made. If requested by Shipper, Natural shall provide a written explanation of the reasons for this determination. A Shipper may challenge Natural's determination by providing a written rebuttal to Natural's explanation within ten (10) days after the explanation is provided by Natural. Natural shall respond to such a rebuttal in writing within ten (10) days. Any reevaluation of credit by Natural in response to such a rebuttal by the Shipper shall be based on the credit criteria set out in this Section 16 and shall be performed as provided in Section 5.12(c) of these General Terms and Conditions.

- (b) (1) If a Shipper fails to satisfy the credit criteria, such Shipper may still obtain or continue service hereunder if Shipper, at its sole discretion, provides the security required under one of the following options:
- $\hbox{(i)} \qquad \hbox{payment in advance of all fees and charges for three (3) months' service;}$
- (ii) a standby irrevocable letter of credit covering all fees and charges for three (3) months' service drawn upon a bank acceptable to Natural;
- (iii) security interest covering all fees and charges for three (3) months' service in collateral provided by the Shipper found to be satisfactory to Natural; or
- (iv) guarantee of all fees and charges for three (3) months' service by a person or another entity which does satisfy the credit appraisal.
- (2) Nothing herein shall be read to preclude Natural from requiring, and enforcing for the term of the initial contracts, more than three (3) months of fees and charges as security in agreements supporting an application for a certificate to construct new or expanded facilities, including on any replacement contract entered into upon a permanent release of capacity under such an initial contract.
- (3) Natural may also require security from noncreditworthy Shippers for the value of gas loaned under any loan type of service. The amount of security necessary to collateralize gas loans shall be equal to the outstanding quantities previously loaned to the Shipper as well as any additional quantities which Natural is obligated

FERC Docket: RP08-319-000

Original Sheet No. 403 Original Sheet No. 403: Pending

GENERAL TERMS AND CONDITIONS

to loan Shipper pursuant to an executed RO, times the Index Price. The Index Price as used herein shall mean the Midpoint Price published in Gas Daily on the day the loan is agreed upon (which will generally reflect the previous Business Day's activity) for the Natural Receipt Zone in which the loan is to occur. For example, if the loan is agreed to on March 8, for a loan on March 12 in the Louisiana Zone, the Gas Daily Daily Midpoint Price published on March 8 (reflecting March 7 business) for "NGPL LA" would be used. If such a price is not published by Gas Daily for that day, Natural will use a price based on an appropriate index price in another Gas Daily publication or another publication of common use in the industry as mutually agreed upon by Natural and Shipper.

- Where a Shipper selects the prepayment option under Section 16(b) of these General Terms and Conditions, the prepayment amounts shall be deposited in an interest-bearing escrow account if such an account has been established by Shipper which meets the criteria set out in this paragraph. The costs of establishing and maintaining the escrow account shall be borne by Shipper. The escrow bank must be rated at least AA or better and shall not be affiliated with Shipper. The escrow arrangement shall provide for the prepayment amounts to be applied against the Shipper's obligation under its service agreement(s) with Natural and shall grant Natural a security interest in such amounts as an assurance of future performance. The escrow agreement shall specify the permitted investments of escrowed funds so as to protect principal, and shall include only such investment options as corporations typically use for short-term deposit $\bar{\text{of}}$ their funds. Such escrow account shall at all times maintain the amount of prepayment required under Section 16(b) of these General Terms and Conditions. Natural is required to draw down the funds in escrow, it will notify the Shipper and Shipper must replenish such funds within three (3) Business Days after such notice.
- (d) Natural's credit appraisal procedures involve the establishment of dollar credit limits on a standardized, nondiscriminatory basis. To the extent that a Shipper's accounts with Natural do not exceed such limit, and Shipper has met all creditworthiness requirements as determined in periodic credit reviews by Natural, which reviews may be conducted on at least an annual basis, no new credit appraisals shall be required when an existing Agreement is amended or a request for a new Agreement is made, provided that Shipper's payment history has been satisfactory and there is no bona fide basis for questioning Shipper's creditworthiness, subject to the provisions of Section 5.12 of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 404 Original Sheet No. 404: Pending

GENERAL TERMS AND CONDITIONS

In the event Natural constructs new lateral facilities (e) to accommodate a Shipper, Natural may require from the Shipper security in an amount up to the cost of the facilities. This provision does not apply to mainline expansions. Such security may be in any of the forms available under Section $16\,(b)$ of these General Terms and Conditions, at Shipper's choice. Natural is only permitted to recover the cost of facilities once, either through rates or through this provision. As Natural recovers the cost of these facilities through its rates, the security required shall be reduced accordingly. Specifically, collateral provided by a Shipper related to new facilities shall be returned to that Shipper in equal monthly amounts over the term of its contract for service related to the new facilities or as otherwise mutually agreed by Natural and Shipper. Where facilities are constructed to serve multiple Shippers, an individual Shipper's obligation hereunder shall be for no more than its proportionate share of the cost of the facilities. This requirement is in addition to and shall not supersede or replace any other rights that Natural may have regarding the construction and reimbursement of facilities.

(f) Natural may not take any action under this Section 16 which conflicts with any order of the U.S. Bankruptcy Court.

FERC Docket: RP08-319-000

Original Sheet No. 405 Original Sheet No. 405: Pending

GENERAL TERMS AND CONDITIONS

17. INTERACTIVE WEBSITE

17.1 WEB SITE DESCRIPTION

(a) Natural maintains the Interactive Website, a FERC compliant interactive internet web site which is available for use by Shippers and other interested parties. The web site has both secure and non-secure regions. Information of a general nature is included in the non-secure region while confidential Shipper specific data is accessible only through the secure region, which requires a logon and password. Daily back-up records of information displayed or entered through this web site are archived, and non-secure information is accessible to customers on a non-discriminatory basis. The data is kept for a three (3) year period, inclusive of both current and archived data.

Natural posts gas quality information as follows:

- (1) Natural provides on the Informational Postings portion of its Interactive Website a link to the natural gas quality provisions of this Tariff.
- (2) Natural provides on the Informational Postings portion of its Interactive Website daily average gas quality information for prior gas day(s), to the extent available, for location(s) that are representative of mainline gas flow. The information available for the identified location(s) is provided in a downloadable format. Information is reported in units as specified in this Tariff. In any event, compliance with gas quality requirements is in accordance with Natural's Tariff, including these General Terms and Conditions. The following are examples of gas quality attributes that could be included in the posting for the applicable Gas Day(s) and location(s):

Heating Value
Hydrocarbon Components, % of C1 - Cnn, as used in
determining Heating Value
Specific Gravity
Water
Nitrogen
Carbon Dioxide
Oxygen
Hydrogen
Helium
Total Sulfur
Hydrogen Sulfide
Carbonyl Sulfide

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

First Revised Sheet No. 405 First Revised Sheet No. 405

Superseding: Original Sheet No. 405

GENERAL TERMS AND CONDITIONS

17. INTERACTIVE WEBSITE

17.1 WEB SITE DESCRIPTION

(a) Natural maintains the Interactive Website, a FERC compliant interactive internet web site which is available for use by Shippers and other interested parties. The web site has both secure and non-secure regions. Information of a general nature is included in the non-secure region while confidential Shipper specific data is accessible only through the secure region, which requires a logon and password. Daily back-up records of information displayed or entered through this web site are archived, and non-secure information is accessible to customers on a non-discriminatory basis. The data is kept for a three (3) year period, inclusive of both current and archived data.

Natural posts gas quality information as follows:

- (1) Natural provides on the Informational Postings portion of its Interactive Website a link to the natural gas quality provisions of this Tariff.
- (2) Natural provides on the Informational Postings portion of its Interactive Website daily average gas quality information for prior gas day(s), to the extent routinely collected and readily available, for location(s) that are representative of mainline gas flow. To the extent that Natural monitors tariff-based gas quality provisions for locations representative of mainline gas flow by non-electronic methods (e.g., spot sample) such information will be posted as soon as practicable. The gas quality information posted pursuant to this tariff provision is operational in nature.

For purposes of this tariff provision, "readily available" is that data which is currently available in electronic format or would be available electronically with minor enhancement(s) to existing data collection, processing and reporting capability.

This gas quality information is reported in units as specified in this tariff's General Terms and Conditions. Gas Quality information not specified in these General Terms and Conditions, if posted, is posted using units determined by Natural.

The information available for the identified location(s) is provided in a downloadable format. In any event, compliance with gas quality requirements is in accordance with

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

Original Sheet No. 405A Original Sheet No. 405A

GENERAL TERMS AND CONDITIONS

Natural's Tariff, including these General Terms and Conditions. Listed below are examples of gas quality attributes:

Heating Value
Interchangeability index(ices)/factor(s)
Hydrocarbon liquid drop out control
parameter(s)/factor(s)
Hydrocarbon Components, % of C1 - Cnn, as used in
determining Heating Value
Specific Gravity
Water
Nitrogen
Carbon Dioxide
Oxygen
Hydrogen
Helium
Total Sulfur
Hydrogen Sulfide
Carbonyl Sulfide

FERC Docket: RP09-240-000

First Revised Sheet No. 406 First Revised Sheet No. 406

Superseding: Original Sheet No. 406

GENERAL TERMS AND CONDITIONS

Mercaptans

Mercury and/or any other contaminants being measured Other pertinent gas quality information that is specific in Natural's Tariff, including these General Terms and Conditions

- (3) Data posted pursuant to the prior paragraph, Section 17.1(a)(2), are made available on Natural's Interactive Website for the most recent three-month period. Beyond the initial three-month period, the historical data is made available offline in accordance with regulatory requirements. Such posted data are provided in a tabular downloadable file described by Natural in the posting. The first row of the file contains the column headers. For any location(s), Natural may, at its discretion, elect to provide gas quality information in addition to that specified in the prior paragraph. Natural may choose how to provide the information.
- (b) The non-secure information is primarily comprised of FERC mandated informational postings. Natural, at its sole option, may add informational sections to this web site in order to facilitate timely and complete communications with customers. The secure region provides access to Nominations, Flowing Gas/Volume Inquiry data, Invoicing, Contracting and Capacity Release Processing. Logons and passwords required to enter the secure region of the web site may be obtained per the procedures outlined in Section 17.2.

(1) INFORMATIONAL POSTINGS

The types of information available through the Informational Postings selection of this web site include: (i) all affiliated marketer information, including names and addresses for affiliated marketing companies; (ii) reports on operationally available capacity, design capacity, unsubscribed capacity and released capacity at Receipt/Delivery Points, on the mainline and for storage; (iii) critical notices concerning capacity related issues and non-critical notices, providing relevant contracts and customer information; (iv) the FERC Index of Customers and the FERC Contract Transactional Postings, (v) the tariff, with search, download and print capabilities; (vi) imbalance volumes available for trading among Shippers prior to cashout as provided in Section 13 hereof and (vii) point catalog.

FERC Docket: RP08-319-000

Original Sheet No. 407 Original Sheet No. 407: Pending

GENERAL TERMS AND CONDITIONS

(2) NOMINATIONS

This feature allows for submittal of all transportation and storage nominations, transfer nominations, predetermined allocations and nomination priorities as required in Section 8. Operators can confirm volumes online. Shippers and point operators can review, print or download scheduled quantity reports.

(3) FLOWING GAS/VOLUME INQUIRY

This feature provides volumetric information on total gas flows and allocated flows, at a point and contract level and provides contract level imbalance information. Information on storage balances, as well as storage injections and withdrawals, is available. The timing for reporting daily operational allocations after the gas has flowed is within one (1) Business Day after the end of the gas Day. If the best available data for reporting daily operational allocations is the scheduled quantity, that quantity should be used for the daily operational allocation. Each Shipper and each other entity involved in a transaction at a point will be able to see the total flows at the point and the volumes allocated to or by such Shipper or other entity.

(4) INVOICING

This system component allows Shippers to view and download invoices and a statement of account. Additionally, using this component, Shippers can create and submit a Payment Remittance.

(5) CONTRACT REQUEST PROCESSING

Using this feature, Shippers can review their existing Agreement information, submit new requests for Agreements, submit requests to amend Agreements and execute Service Agreements online.

(6) CAPACITY RELEASE REQUEST AND BID PROCESSING

This interactive feature allows Shippers to submit Capacity Release Requests and Bids, which, in turn, are automatically posted to this web site as provided in Section 19. Additionally, Shipper with recall provisions in a release of capacity can initiate the recall process using this feature.

FERC Docket: RP08-319-000

Original Sheet No. 408 Original Sheet No. 408: Pending

GENERAL TERMS AND CONDITIONS

17.2 ACCESS TO WEB SITE

Shippers and other interested parties may obtain access to the interactive transactional web pages by contacting a representative of Natural's Electronic Customer Services Department in Houston. Logons, passwords and access instructions will be supplied upon request under the terms and conditions set forth in Sections 17.3 through 17.13. The internet address for this web site is: http://pipeline.kindermorgan.com. This web site replaces DART and Subscribers to DART can continue to use their DART logon and password to access these web pages. By continuing to use such DART logons, Subscribers agree to the terms and conditions set forth in Sections 17.3 through 17.13.

17.3 AUTHORITY OF EMPLOYEE

Users of this web site shall be deemed to have agreed and admitted that any employee permitted by Subscriber to access this web site shall have the legal authority to act on behalf of Subscriber in performing any functions, including those functions which are available presently and those functions which become available at a later date.

17.4 INSTALLATION OF SOFTWARE

Each Subscriber shall purchase and ensure that lawful installation of Internet browser software occurs for each personal computer (PC) from which this web site is accessed.

17.5 CONFIDENTIALITY

Certain information contained in this web site is confidential. A Subscriber shall not reproduce, disclose or otherwise make available confidential information contained therein to any other company, corporation, individual, or partnership.

17.6 RELIANCE BY NATURAL

Natural may act, and shall be fully protected by a Subscriber in acting, in reliance upon any acts or things done or performed by Subscriber's employees or designated agents on behalf of Subscriber and in respect to all matters conducted through this web site. Natural may correct errors in information entered into this web site by a Subscriber promptly after receiving notice of the corrections or may require Subscribers to enter the corrections directly into this web site.

FERC Docket: RP08-319-000

Original Sheet No. 409 Original Sheet No. 409: Pending

GENERAL TERMS AND CONDITIONS

17.7 ACCESS TO CONFIDENTIAL INFORMATION

Should a Subscriber require access to confidential information (such as Agreement, point, nomination, volume, or other customer-specific information deemed to be of a confidential nature requiring controlled access), Natural will require the Subscriber to provide a written request and officer level approval for issuance of a company, departmental or individual level computer access (logon) identification code and password. Upon receipt of such request, Natural will ensure return of a confidential logon code and password within one (1) business day.

17.8 LOGON

A Subscriber's logon and password are confidential and are used to identify that Subscriber. A Subscriber shall keep its logon(s) and password(s) confidential. A Subscriber will ensure that only authorized employees and agents of Subscriber will be given logon(s) and password(s) and only these authorized persons will be permitted to access this web site on Subscriber's behalf. A Subscriber and its employees and agents will not disclose the Subscriber's logon(s) and password(s) to anyone without authority to access this web site on behalf of the Subscriber. To ensure such confidentiality is not breached, requests from Subscriber employees or agents for information regarding Subscriber logon(s) and password(s) made subsequent to issuance of the original logon(s) and password(s) may not be honored without receipt by Natural of additional authorization from Subscriber. Subscriber shall be responsible for and accepts liability for any security breach that is traced to Subscriber's logon(s) and password(s).

17.9 BREACH OF SECURITY

A Subscriber shall promptly notify Natural if there is any indication that a security breach has occurred with regard to Subscriber's logon(s) and password(s). This includes, but is not limited to: (1) loss of confidentiality of logon(s) and password(s); (2) termination of employment of any authorized employee; or (3) loss of authority to access this web site by any authorized employee. Such notification shall be made to Natural's Electronic Customer Services Department.

FERC Docket: RP08-319-000

Original Sheet No. 410 original Sheet No. 410: Pending

GENERAL TERMS AND CONDITIONS

17.10 LIMITATION TO ACCESS

A Subscriber may attempt to access only that data for which Subscriber has authorization. A Subscriber shall provide supporting legal documentation prior to being given access to data of other subsidiaries, affiliates, or companies for whom it has an agency relationship. See Section 8 of these General Terms and Conditions for information on delegation.

17.11 INDEMNITY

Each Subscriber shall indemnify Natural and hold Natural harmless for all damages, losses, and liabilities arising out of:

- (a) Subscriber's or its employees' or agents' breach of any of Subscriber's obligations under this Section 17, including any breach of confidentiality with respect to the assignment of logon(s) and password(s) to Subscriber's authorized employees and agents and any unauthorized use by a formerly authorized person or by any unauthorized person who gained knowledge of Subscriber's logon(s) and password(s) through no fault of Natural;
- (b) any omission or failure by Subscriber's employees or agents to act or perform any duty required by a web site function; and
- (c) any action taken by Subscriber, its employees or agents, its former authorized employees and agents or unauthorized persons who gained knowledge of Subscriber's logon(s) and password(s) through no fault of Natural, which interferes with the proper operation of this web site.

Notwithstanding the foregoing, neither Natural nor Subscriber shall be liable to the other if an unauthorized user gains access to this web site through no fault of either Natural or Subscriber.

17.12 LIMITS OF RESPONSIBILITY

Natural shall not be responsible for an omission or failure by Natural to act or perform any duty requested by a function accessed via this web site if such omission or failure to act is caused by or related to data lost in the communication of such data from Subscriber's to Natural's computer system, power failures, failure of backup systems, or any other event beyond the reasonable control of Natural.

FERC Docket: RP08-319-000

Original Sheet No. 411 Original Sheet No. 411: Pending

GENERAL TERMS AND CONDITIONS

17.13 RESERVATION

Natural reserves the right to add, modify or terminate web site functions at any time, subject to compliance with applicable Commission Regulations.

17.14 AGREEMENT BY NON-SHIPPER

Any Subscriber who is not a Shipper under one of the Rate Schedules in this Volume of Natural's FERC Gas Tariff will be required, as a precondition of access to this web site, to sign an agreement with Natural pursuant to which the Subscriber agrees to be bound by the provisions of this Section.

FERC Docket: RP08-319-000

Sheet No. 412 Sheet No. 412: Pending

GENERAL TERMS AND CONDITIONS

18. RESERVED FOR FUTURE USE

FERC Docket: RP08-319-000

Original Sheet No. 413 Original Sheet No. 413: Pending

GENERAL TERMS AND CONDITIONS

19. CAPACITY RELEASE BY FIRM SHIPPERS

19.1 GENERAL

- (a) Subject to the terms, conditions and limitations set forth in this Section 19, a Shipper holding capacity rights under an Eligible Firm Transportation Agreement shall have the right to release all or a portion of such capacity rights and, if a capacity release is effectuated under this Section 19, to receive a credit for reservation charge revenues received by Natural from that other Shipper for such released capacity.
- (b) The deadlines set forth in this Section 19 are applicable to all parties involved in the capacity release process; however, they are only applicable if all information provided by the parties to the transaction is valid and the Replacement Shipper (or Subreplacement Shipper, if applicable) has been determined to be creditworthy before the Qualified Bid is tendered, and there are no special terms or conditions of the release.
- (c) Following is a summary of the capacity release process and deadlines set forth in greater detail in the remainder of this Section $19\colon$
- $\hspace{1.5cm} \hbox{(1)} \hspace{0.5cm} \hbox{For biddable releases (less than one (1)} \\ \hbox{year):} \\$
- (i) The Capacity Release Request should be tendered by no later than 12:00 p.m. Central Clock Time on a Business Day;
- (ii) The open season ends no later than 1:00 p.m. Central Clock Time on a Business Day (evaluation period begins at 1:00 p.m. Central Clock Time during which contingency is eliminated, determination of winning Qualified Bid(s) is made, and ties are broken);
- (iii) Evaluation period ends and award posting if no match required at 2:00 p.m. Central Clock Time;
- (iv) Match or award is communicated by 2:00 p.m. Central Clock Time;
- $\qquad \qquad \text{(v)} \qquad \text{Match response by 2:30 p.m. Central } \\ \text{Clock Time;} \\$

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

First Revised Sheet No. 413 First Revised Sheet No. 413

Superseding: Original Sheet No. 413

GENERAL TERMS AND CONDITIONS

19. CAPACITY RELEASE BY FIRM SHIPPERS

19.1 GENERAL

- (a) Subject to the terms, conditions and limitations set forth in this Section 19, a Shipper holding capacity rights under an Eligible Firm Transportation Agreement shall have the right to release all or a portion of such capacity rights and, if a capacity release is effectuated under this Section 19, to receive a credit for reservation charge revenues received by Natural from that other Shipper for such released capacity.
- (b) The capacity release timeline set forth in this Section 19 applies to all parties involved in the capacity release process provided that: 1) all information provided by the parties to the transaction is valid and the Replacement Shipper (or Subreplacement Shipper, if applicable) has been determined to be creditworthy before the Qualified Bid is tendered, 2) for indexbased capacity release transactions, the Releasing Shipper has provided Natural with sufficient instructions to evaluate the corresponding bids according to the timeline, and 3) there are no special terms or conditions of the release. Further, Natural may complete the capacity release process on a different timeline if the offer includes unfamiliar or unclear terms and conditions (e.g., designation of an index not supported by Natural).
- (c) Following is a summary of the capacity release process and deadlines set forth in greater detail in the remainder of this Section $19\colon$
 - (1) For biddable releases (one (1) Year or less):
- (i) The Capacity Release Request should be tendered by no later than 12:00 p.m. Central Clock Time on a Business Day;
- (ii) The open season ends no later than $1:00~\rm p.m.$ Central Clock Time on a Business Day (evaluation period begins at $1:00~\rm p.m.$ Central Clock Time during which contingency is eliminated, determination of winning Qualified Bid(s) is made, and ties are broken);
- (iii) Evaluation period ends and award posting if no match required at 2:00 p.m. Central Clock Time;
- (iv) Match or award is communicated by 2:00 p.m. Central Clock Time;

FERC Docket: RP08-319-000

Original Sheet No. 414 Original Sheet No. 414: Pending

GENERAL TERMS AND CONDITIONS

(vi) Award posting where match required by $3:00 \ \mathrm{p.m.}$ Central Clock Time;

(vii) Contract issued within one (1) hour of award posting (with new contract number when applicable), nomination possible beginning at the next available nomination cycle for the effective date of the contract, consistent with Section 19.9(d) hereof (nomination is not contingent on a contract being issued or executed so long as the Replacement Shipper has preapproved credit).

- (2) For biddable releases (one (1) year or more):
- (i) The Capacity Release Request should be tendered by no later than $12:00~\rm p.m.$ Central Clock Time four (4) Business Days before the award;
- (ii) The open season ends no later than
 1:00 p.m. Central Clock Time on the Business Day before timely
 nominations are due (open season is three (3) Business Days);
- (iii) Evaluation period begins at 1:00 p.m. Central Clock Time during which contingency is eliminated, determination of best bid is made, and ties are broken;
- (iv) Evaluation period ends and award posting if no match required at $2:00\ p.m.$ Central Clock Time;
- (v) Match or award is communicated by 2:00 p.m. Central Clock Time;
- (vi) Match response by 2:30 p.m. Central

Clock Time;

(vii) Award posting where match required by 3:00 p.m. Central Clock Time;

(viii) Contract issued within one (1) hour of award posting (with new contract number when applicable), nomination possible beginning at the next available nomination cycle for the effective date of the contract, consistent with Section 19.9(d) hereof (nomination is not contingent on a contract being issued or executed so long as the Replacement Shipper has preapproved credit).

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

First Revised Sheet No. 414 First Revised Sheet No. 414

Superseding: Original Sheet No. 414

GENERAL TERMS AND CONDITIONS

(v) Match response by 2:30 p.m. Central

Clock Time;

(vi) Award posting where match required by 3:00 p.m. Central Clock Time;

(vii) Contract issued within one (1) hour of award posting (with new contract number when applicable), nomination possible beginning at the next available nomination cycle for the effective date of the contract, consistent with Section 19.9(d) hereof (nomination is not contingent on a contract being issued or executed so long as the Replacement Shipper has preapproved credit).

 $\hspace{1cm} \hbox{(2)} \hspace{0.2cm} \hbox{For biddable releases (more than one (1)} \\ \hbox{Year):} \\$

(i) The Capacity Release Request should be tendered by no later than $12:00~\rm p.m.$ Central Clock Time four (4) Business Days before the award;

(ii) The open season ends no later than
1:00 p.m. Central Clock Time on the Business Day before timely
nominations are due (open season is three (3) Business Days);

(iii) Evaluation period begins at 1:00~p.m. Central Clock Time during which contingency is eliminated, determination of best bid is made, and ties are broken;

(iv) Evaluation period ends and award posting if no match required at 2:00 p.m. Central Clock Time;

(v) Match or award is communicated by $2:00 \ \text{p.m.}$ Central Clock Time;

(vi) Match response by 2:30 p.m. Central

Clock Time;

(vii) Award posting where match required by 3:00 p.m. Central Clock Time;

(viii) Contract issued within one (1) hour of award posting (with new contract number when applicable), nomination possible beginning at the next available nomination cycle for the effective date of the contract, consistent with Section 19.9(d) hereof (nomination is not contingent on a contract being issued or executed so long as the Replacement Shipper has preapproved credit).

FERC Docket: RP08-319-000

Original Sheet No. 415 Original Sheet No. 415: Pending

GENERAL TERMS AND CONDITIONS

(3) For prearranged releases not requiring bidding under this Section 19:

(i) For the Timely Cycle, postings are due by 10:30 a.m.; contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract (Central Clock Time).

(ii) For the Evening Cycle, postings are due by 5:00 p.m.; contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract (Central Clock Time).

(iii) For the Intraday 1 Cycle, postings are due by 9:00 a.m.; contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract (Central Clock Time).

(iv) For the Intraday 2 Cycle, postings are due by 4:00 p.m.; contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract (Central Clock Time).

(d) (1) Natural will not award capacity release offers to a Shipper until and unless the Shipper meets Natural's creditworthiness requirements applicable to all services that it receives from Natural, including the service represented by the capacity release.

(2) Natural shall provide the original Releasing Shipper with Internet E-mail notification reasonably proximate in time with any of the following formal notices given by Natural to the Releasing Shipper's Replacement Shipper(s), of the following:

(i) Notice to the Replacement Shipper regarding the Replacement Shipper's past due, deficiency, or default status pursuant to Natural's tariff;

(ii) Notice to the Replacement Shipper regarding the Replacement Shipper's suspension of service notice;

(iii) Notice to the Replacement Shipper regarding the Replacement Shipper's contract termination notice due to default or credit-related issues; and

FERC Docket: RP08-319-000

Original Sheet No. 416 Original Sheet No. 416: Pending

GENERAL TERMS AND CONDITIONS

(iv) Notice to the Replacement Shipper that the Replacement Shipper(s) is no longer creditworthy and has not provided credit alternative(s) pursuant to Natural's tariff.

19.2 DEFINITIONS

(a) BID VALUE

The value assigned to a Qualified Bid or a Prearranged Release according to the bid evaluation procedures set forth in Section 19.10 or, if applicable, the bid evaluation procedures set forth in the Capacity Release Request.

(b) CAPACITY RELEASE REQUEST

 $$\operatorname{\textsc{The}}$ request that a Releasing Shipper submits to initiate the capacity release procedure under this Section 19.

(c) ELIGIBLE FIRM TRANSPORTATION AGREEMENT

A transportation agreement under one of the following Rate Schedules, to the extent it provides for firm transportation [which includes storage, subject to the limitations stated in Section 19.5(f)] by Natural: Rate Schedules DSS, NSS, FRSS, FTS and FFTS.

(d) MAXIMUM BID VOLUME

 $$\operatorname{\textsc{The}}$ maximum amount of capacity the Qualified Bidder agreed to accept in its Qualified Bid.

(e) MINIMUM BID VOLUME

 $$\operatorname{\textsc{The}}$ minimum amount of capacity the Qualified Bidder agreed to accept in its Qualified Bid.

FERC Docket: RP09-240-000

First Revised Sheet No. 417 First Revised Sheet No. 417

Superseding: Original Sheet No. 417

GENERAL TERMS AND CONDITIONS

(f) ORIGINAL SHIPPER

 $$\operatorname{\textsc{The}}$ entity who is the Shipper under an Eligible Firm Transportation Agreement (other than through a capacity release).

(q) PREARRANGED RELEASE

The binding written release agreement between a Releasing Shipper and a Prearranged Shipper covering Eligible Firm Transportation Agreement capacity rights, the effectiveness of which is subject only to: (1) the prequalification of the Prearranged Shipper under Section 19.15; and (2) the release of such capacity rights to the Prearranged Shipper as provided by this Section 19.

- (i) A Prearranged Release between a Releasing Shipper and an Asset Manager as that term is defined in 18 C.F.R. Section 284.8(h)(3), shall be defined for purposes of this Section 19, as a "Prearranged Asset Manager Release".
- (ii) A Prearranged Release between a Releasing Shipper and a Marketer Participating in a State-Regulated Retail Access Program, as that term is defined in 18 C.F.R. Section 284.8(h)(4), shall be defined for purposes of this Section 19, as a "Prearranged Release to a Marketer Participating in a State-Regulated Retail Access Program".

(iii) A Prearranged Asset Manager Release and a Prearranged Release to a Marketer Participating in a State-Regulated Retail Access Program are exempt from the Open Season Requirements set forth in this Section 19.

(h) PREARRANGED SHIPPER

A person or entity prequalified under Section 19.15 who has entered into a Prearranged Release with a Releasing Shipper for Eligible Firm Transportation Agreement capacity rights, including a Replacement Shipper under either a Prearranged Asset Manager Release, or a Prearranged Release to a Marketer Participating in a State-Regulated Retail Access Program.

FERC Docket: RP09-240-000

Original Sheet No. 417A Original Sheet No. 417A

GENERAL TERMS AND CONDITIONS

(i) QUALIFIED BID

A binding bid prequalified under Section 19.15 by a Qualified Bidder for capacity rights subject to a Capacity Release Request under this Section 19, including a Replacement Shipper under either a Prearranged Asset Manager Release, or a Prearranged Release to a Marketer Participating in a State-Regulated Retail Access Program.

(j) QUALIFIED BIDDER

Any person or entity prequalified under Section 19.15 who bids for capacity rights being released under this Section 19, including a Replacement Shipper under either a Prearranged Asset Manager Release, or a Prearranged Release to a Marketer Participating in a State-Regulated Retail Access Program.

(k) RELEASED FIRM TRANSPORTATION AGREEMENT

The agreement between Natural and a Replacement Shipper or a Subreplacement Shipper by which the Replacement Shipper or Subreplacement Shipper confirms the receipt of capacity rights under an Eligible Firm Transportation Agreement released by a Releasing Shipper under this Section 19.

(1) RELEASING SHIPPER

Any Shipper holding capacity rights under an Eligible Firm Transportation Service Agreement who has released or seeks to release such capacity rights pursuant to this Section 19.

FERC Docket: RP09-240-000

First Revised Sheet No. 418 First Revised Sheet No. 418

Superseding: Original Sheet No. 418

GENERAL TERMS AND CONDITIONS

(m) REPLACEMENT SHIPPER

A Shipper receiving capacity rights under an Eligible Firm Transportation Agreement pursuant to a direct release from an Original Shipper under this Section 19.

(n) SHORT-TERM PREARRANGED RELEASE

 $\,$ A Prearranged Release with a term of thirty-one (31) days or less.

(o) SUBREPLACEMENT SHIPPER

A Shipper receiving capacity rights released from an Eligible Firm Transportation Agreement by a Replacement Shipper or a Subreplacement Shipper under this Section 19.

(p) UNIT BID VALUE

 $\,$ The unit value per Dth assigned to a Qualified Bid or a Prearranged Release according to the bid evaluation procedures set forth in Section 19.10.

(q) WINNING BID VALUE

The highest possible total Bid Value achievable under Section 19.10 for the Capacity Release Request from the Qualified Bids consistent with the Capacity Release Request and this Section 19.

19.3 RELEASE WITHOUT A PREARRANGED SHIPPER

A Shipper seeking to release its Eligible Firm Transportation Agreement capacity rights without a Prearranged Shipper shall deliver a Capacity Release Request to Natural's Interactive Website (or in writing for posting on Natural's Interactive Website if Natural's Interactive Website is unavailable for receiving Capacity Release Requests) which sets forth:

(a) The Releasing Shipper's legal name, address and phone number, the Eligible Firm Transportation Agreement number, the date of the Eligible Firm Transportation Agreement and the name and title of the individual responsible for authorizing the capacity release;

FERC Docket: RP09-240-000

First Revised Sheet No. 419 First Revised Sheet No. 419

Superseding: Original Sheet No. 419

GENERAL TERMS AND CONDITIONS

- (b) The quantity of the capacity (in Dth/day) and the transportation path(s) (or segment(s) thereof) being released, including identification by Natural's PIN Number (or Common Code) of the Receipt Points and Delivery Points defining the release path/segment and the firm capacity to be released at each such point; and the extent to which storage capacity in which the Releasing Shipper has gas is being released;
- (c) Whether the capacity being released is subject to recall and/or reput, and if so, the exact conditions for such recall and/or reput (which conditions must conform to Sections 19.5 and 19.14):
- (d) The proposed effective date and proposed term of the release;
- (e) Whether the Releasing Shipper wants Natural to actively market the Releasing Shipper's capacity rights pursuant to Section 20 of these General Terms and Conditions;
- (f) Whether the Releasing Shipper will accept Qualified Bids which are contingent on subsequent events (such as the subsequent purchase of upstream or downstream capacity), and if so, what events and the last date by which such contingency must be fulfilled;
- (g) The starting date for the open season and the length of time for the open season (which must conform to Section 19.7);
- (h) Whether the Releasing Shipper will accept Qualified Bids whose revenues will vary by the volume transported, and if so, any minimum amount to be billed as a reservation charge even if there is no flow (or insufficient flow);
- (i) Which of the bid evaluation procedures set forth in Section 19.10 the Shipper wishes to use, if any;
- (j) Whether the Qualified Bids are to specify dollars and cents and/or percentage of the maximum tariff rate or an index based formula as detailed in the Capacity Release Request;
- (k) Under a release of storage capacity, whether the capacity being released is subject to certain conditions on the sale and/or repurchase of gas in storage inventory and on there being a certain amount of gas left in storage at the end of the release and if so, any such conditions; and

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

Second Revised Sheet No. 419 Second Revised Sheet No. 419

Superseding: First Revised Sheet No. 419

GENERAL TERMS AND CONDITIONS

- (b) The quantity of the capacity (in Dth/day) and the transportation path(s) (or segment(s) thereof) being released, including identification by Natural's PIN Number (or Common Code) of the Receipt Points and Delivery Points defining the release path/segment and the firm capacity to be released at each such point; and the extent to which storage capacity in which the Releasing Shipper has gas is being released;
- (c) Whether the capacity being released is subject to recall and/or reput, and if so, the exact conditions for such recall and/or reput (which conditions must conform to Sections 19.5 and
- (d) The proposed effective date and proposed term of the release;
- (e) Whether the Releasing Shipper wants Natural to actively market the Releasing Shipper's capacity rights pursuant to Section 20 of these General Terms and Conditions;
- (f) Whether the Releasing Shipper will accept Qualified Bids which are contingent on subsequent events (such as the subsequent purchase of upstream or downstream capacity), and if so, what events and the last date by which such contingency must be fulfilled;
- (g) The starting date for the open season and the length of time for the open season (which must conform to Section 19.7);
- (h) Whether the Releasing Shipper will accept Qualified Bids whose revenues will vary by the volume transported, and if so, any minimum amount to be billed as a reservation charge even if there is no flow (or insufficient flow);
- (i) Which of the bid evaluation procedures set forth in Section 19.10 the Shipper wishes to use, if any;
- Which one of the following methods is acceptable for bidding on the given capacity release offer:
 - (1) Non-Index-based release dollars and cents,(2) Non-Index-based release percentage of
 - maximum rate, or
 - (3) Index-based formula as detailed in the capacity release offer

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

Original Sheet No. 419A Original Sheet No. 419A

GENERAL TERMS AND CONDITIONS

The bids for the given offer should adhere to the method specified by the Releasing Shipper. The bidder may bid the maximum reservation rate, in Natural's tariff, as an alternative to the method specified by the Releasing Shipper except when the release is index-based for a term of one (1) Year or less or utilizes market-based rates.

(k) Under a release of storage capacity, whether the capacity being released is subject to certain conditions on the sale and/or repurchase of gas in storage inventory and on there being a certain amount of gas left in storage at the end of the release and if so, any such conditions; and

FERC Docket: RP09-240-000

First Revised Sheet No. 420 First Revised Sheet No. 420

Superseding: Original Sheet No. 420

GENERAL TERMS AND CONDITIONS

(1) Any other applicable conditions (which must conform to Section 19.5), including any minimum price condition and whether the Releasing Shipper wishes to apply a bid evaluation procedure different than the bid evaluation procedure set forth in Section 19.10 for evaluating Qualified Bids for its capacity rights, and if so, all the factors to be used in evaluating Qualified Bids, including how its capacity rights are to be awarded in the event of a tie for the highest valued Qualified Bid and whether the Releasing Shipper has presubmitted a computer diskette for such bid evaluation procedure pursuant to Section 19.5(a).

19.4 PREARRANGED RELEASE

Subject to Section 19.6, a Shipper seeking to release its Eligible Firm Transportation Agreement capacity rights to a Prearranged Shipper shall deliver a Capacity Release Request via Natural's Interactive Website or via EDI at Natural's designated site for an open season. The Capacity Release Request shall set forth:

- (a) The Releasing Shipper's legal name, address and phone number, the Prearranged Shipper's legal name, and where applicable, identification of the Prearranged Replacement Shipper as an "Asset Manager" as that term is defined in 18 C.F.R. 284.8(h)(3) or a "Marketer Participating in a State-Regulated Retail Access Program" (as that term is defined in 18 C.F.R. 284.8(h)(4)), address, phone number, and telefax number, the Eligible Firm Transportation Agreement number, the date of the Eligible Firm Transportation Agreement and the name and title of the individuals at the Releasing Shipper and the Prearranged Shipper responsible for authorizing the capacity release;
- (b) A statement that the Prearranged Shipper has agreed to be bound by a capacity award to the Prearranged Shipper under this Section 19 by Natural and to execute a Released Firm Transportation Agreement which consists of Natural's standard form of Agreement covering the Rate Schedule applicable to the released capacity and the terms and conditions of the Prearranged Release, in accordance with Natural's Tariff. Such statement shall also set forth:
- (1) The quantity of the capacity (in Dth/day) and the transportation path(s) (or segment(s) thereof) being released, including identification by Natural's PIN Number (or Common Code) of the Receipt Points and Delivery Points defining the released path/segment and the firm capacity to be released at each such point; and the extent to which any storage capacity in which the Releasing Shipper has gas is being released;

FERC Docket: RP08-319-000

Original Sheet No. 421 Original Sheet No. 421: Pending

GENERAL TERMS AND CONDITIONS

(2) The fixed reservation charge and/or volumetric charge the Prearranged Shipper has agreed to pay for the released capacity;

- (3) Whether the capacity being released is subject to recall and/or reput in the Prearranged Release, and if so, the exact conditions of such recall and/or reput (which conditions must conform with Sections 19.5 and 19.14); and
- $\mbox{(4)}$ The proposed effective date of the Prearranged Release and the proposed term of the Prearranged Release.
- (c) Whether the Releasing Shipper will accept Qualified Bids which are contingent on subsequent events (such as the subsequent purchase of upstream or downstream capacity), and if so, what events and the last date by which such contingency must be fulfilled;
- (d) Whether the Releasing Shipper will accept Qualified Bids with longer terms or larger volumes, and if so, what is the maximum volume and the longest term the Releasing Shipper will accept:
- (e) Whether the Releasing Shipper wants Natural to actively market its capacity rights subject to the Prearranged Release pursuant to Section 20 of these General Terms and Conditions:
- (f) The starting date for and the length of time for the open season (which must conform to Section 19.7) and the length of time (consistent with Section 19.9(b)) for the Prearranged Shipper to be able to match a winning Qualified Bid;
- (g) Whether the Releasing Shipper will accept Qualified Bids whose revenues will vary by the volume transported, and if so, any minimum amount to be billed as a reservation charge even if there is no flow (or insufficient flow);
- (h) Which of the bid evaluation procedures set forth in Section 19.10 the Shipper wishes to use, if any;

FERC Docket: RP09-240-001

Substitute First Revised Sheet No. 422 Substitute First Revised Sheet No. 422

Superseding: First Revised Sheet No. 422

GENERAL TERMS AND CONDITIONS

(i) Whether the Qualified Bids are to specify dollars and cents and/or percentage of the maximum tariff rate or an index-based formula as detailed in the Capacity Release Request;

- (j) Under a release of storage capacity, whether the capacity being released is subject to certain conditions on the sale and/or repurchase of gas in storage inventory and on there being a certain amount of gas left in storage at the end of the release and if so, any such conditions;
- (k) Whether the release is a Prearranged Asset Manager Release as defined in Section 19.2(g) (i) hereof, and the Asset Manager's obligation to deliver gas to, or purchase gas from, the Releasing Shipper;
- (1) Whether the release is a Prearranged Release to a Marketer Participating in a State-Regulated Retail Access Program, as defined in Section 19.2(g)(ii) hereof; and
- (m) Any other applicable conditions (which must conform with Section 19.5), including any minimum price condition and whether the Releasing Shipper wishes to apply a bid evaluation procedure different than the bid evaluation procedure set forth in Section 19.10 for evaluating Qualified Bids for its capacity rights, and if so, all the factors to be used in evaluating Qualified Bids, including how its capacity rights are to be awarded in the event of a tie for the highest valued Qualified Bid and whether the Releasing Shipper has presubmitted a computer diskette for such bid evaluation procedure pursuant to Section 19.5(a).

19.5 CAPACITY RELEASE REQUIREMENTS

(a) All terms and conditions relating to a release which is the subject of a Capacity Release Request:
(1) must be nondiscriminatory and applicable to all potential bidders; (2) must be made available to Natural for posting;
(3) must relate to the details of acquiring or maintaining the transportation capacity rights on Natural consistent with this Tariff and Order No. 712, which are the subject of the release; and (4) must not place any obligations or burdens on Natural in addition to the terms and conditions applicable to a capacity release under this Section 19 which are specified in Natural's Tariff. Any bid evaluation procedure elected by a Releasing Shipper different from Natural's bid evaluation

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

Second Revised Sheet No. 422 Second Revised Sheet No. 422

Superseding: Substitute First Revised Sheet No. 422

GENERAL TERMS AND CONDITIONS

(i) Which one of the following methods is acceptable for bidding on the given capacity release offer:

- (1) Non-Index-based release dollars and cents,
- (2) Non-Index-based release percentage of maximum rate, or
- (3) Index-based formula as detailed in the capacity release offer

The bids for the given offer should adhere to the method specified by the Releasing Shipper. The bidder may bid the maximum reservation rate, in Natural's tariff, as an alternative to the method specified by the Releasing Shipper except when the release is index-based for a term of one (1) Year or less or utilizes market-based rates.

- (j) Under a release of storage capacity, whether the capacity being released is subject to certain conditions on the sale and/or repurchase of gas in storage inventory and on there being a certain amount of gas left in storage at the end of the release and if so, any such conditions;
- (k) Whether the release is a Prearranged Asset Manager Release as defined in Section 19.2(g) (i) hereof, and the Asset Manager's obligation to deliver gas to, or purchase gas from, the Releasing Shipper;
- (1) Whether the release is a Prearranged Release to a Marketer Participating in a State-Regulated Retail Access Program, as defined in Section 19.2(q) (ii) hereof; and
- (m) Any other applicable conditions (which must conform with Section 19.5), including any minimum price condition and whether the Releasing Shipper wishes to apply a bid evaluation procedure different than the bid evaluation procedure set forth in Section 19.10 for evaluating Qualified Bids for its capacity rights, and if so, all the factors to be used in evaluating Qualified Bids, including how its capacity rights are to be awarded in the event of a tie for the highest valued Qualified Bid and whether the Releasing Shipper has presubmitted a computer diskette for such bid evaluation procedure pursuant to Section 19.5(a).

FERC Docket: RP09-240-000

Original Sheet No. 422A Original Sheet No. 422A

GENERAL TERMS AND CONDITIONS

procedure set forth in Sections 19.10(b) through 19.10(d) must be objective, nondiscriminatory in all circumstances and contain a complete description of the bid evaluation procedure for posting on the Informational Postings portion of Natural's Interactive Website. Natural may require the Releasing Shipper to submit a working computer program to Natural in diskette form which is compatible with Natural's Interactive Website computer and which will enable Natural to make such alternative bid evaluation entirely through Natural's Interactive Website, including prior to the time any alternative bid evaluation procedure is requested, if such bid evaluation procedure is not based on (1) highest rate; (2) net revenue; or (3) present value as determined in Sections 19.10(b)(1) through 19.10(b)(4) (collectively referred to as "Acceptable Alternative Bid Evaluation Procedure") and the remaining procedures set forth in Sections 19.10(c) and 19.10(d). If the Releasing Shipper elects a bid evaluation procedure

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

First Revised Sheet No. 422A First Revised Sheet No. 422A

Superseding: Original Sheet No. 422A

GENERAL TERMS AND CONDITIONS

19.5 CAPACITY RELEASE REQUIREMENTS

(a) All terms and conditions relating to a release which is the subject of a Capacity Release Request: (1) must be nondiscriminatory and applicable to all potential bidders; (2) must be made available to Natural for posting; (3) must relate to the details of acquiring or maintaining the transportation capacity rights on Natural consistent with this Tariff and Order No. 712, which are the subject of the release; and (4) must not place any obligations or burdens on Natural in addition to the terms and conditions applicable to a capacity release under this Section 19 which are specified in Natural's Tariff. Any bid evaluation procedure elected by a Releasing Shipper different from Natural's bid evaluation procedure set forth in Sections 19.10(b) through 19.10(d) must be objective, nondiscriminatory in all circumstances and contain a complete description of the bid evaluation procedure for posting on the Informational Postings portion of Natural's Interactive Website. Natural may require the Releasing Shipper to submit a working computer program to Natural in diskette form which is compatible with Natural's Interactive Website computer and which will enable Natural to make such alternative bid evaluation entirely through Natural's Interactive Website.

FERC Docket: RP09-240-000

First Revised Sheet No. 423 First Revised Sheet No. 423

Superseding: Original Sheet No. 423

GENERAL TERMS AND CONDITIONS

that differs from Natural's bid evaluation procedure or the Acceptable Alternative Bid Evaluation Procedure and the remaining procedures set forth in Sections 19.10(c) and 19.10(d), Natural shall not be held to the subsequent deadlines set forth in this Section 19, but Natural shall make a reasonable attempt to adhere to such deadlines. Natural shall publish standards relative to such computer diskettes, but it is the responsibility of the Releasing Shipper to develop and provide the working computer diskette. The Releasing Shipper shall warrant that the computer diskette conforms to the bid evaluation procedure in the Capacity Release Request.

- (b) The quantity sought to be released under a Capacity Release Request shall not be less than the minimum quantity required for the Eligible Firm Transportation Agreement under Natural's Tariff.
- (c) The term of any release of capacity sought under this Section 19 shall be at least one full day and shall not exceed the remaining term of the Eligible Firm Transportation Agreement.
- (d) (l) Segmented releases are subject to Section 8.14 of these General Terms and Conditions.
- (2) (i) The maximum rates that may be bid and charged for a Released Firm Transportation Agreement that is for a term greater than one (1) year are the maximum lawful rates applicable to the Eligible Firm Transportation Agreement held by the Original Shipper. If the Original Shipper is paying a Negotiated Rate or a rate under a Negotiated Rate Formula pursuant to Section 49 of these General Terms and Conditions, a Qualified Bidder may not bid a rate which exceeds the applicable Recourse Rate or is less than the applicable minimum rate a qualified Bidder may bid a rate form which would be a Negotiated Rate or Negotiated Rate Formula if and only if the rate form is one explicitly recognized in Natural's Tariff as available for capacity releases (such as volumetric rates); provided the resulting charges must be within the range set by the applicable maximum and minimum rates.
- (ii) There is no maximum rate limitation applicable to bids for capacity release for a term of one (1) year or less, if the release is to take effect on or before one (1) year from the date on which Natural is notified of the release.

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

Second Revised Sheet No. 423 Second Revised Sheet No. 423

Superseding: First Revised Sheet No. 423

GENERAL TERMS AND CONDITIONS

The Releasing Shipper shall warrant that the computer diskette conforms to the bid evaluation procedure in the Capacity Release Request.

For the capacity release business process timing model, only the following methodologies are supported by Natural and provided to Releasing Shippers as choices from which they may select and, once chosen, will be used in determining the awards from the bid(s) submitted. They are: 1) highest rate, 2) net revenues and 3) present value. For index-based capacity release transactions, the Releasing Shipper should provide the necessary information and instructions to support the chosen methodology. Other choices of bid evaluation methodology (including other Releasing Shipper defined evaluation methodologies) can be accorded similar timeline evaluation treatment at the discretion of Natural. However, Natural is not required to offer other choices or similar timeline treatment for other choices, nor, is Natural held to the timeline should the Releasing Shipper elect another method of evaluation.

- (b) The quantity sought to be released under a Capacity Release Request shall not be less than the minimum quantity required for the Eligible Firm Transportation Agreement under Natural's Tariff.
- (c) The term of any release of capacity sought under this Section 19 shall be at least one full day and shall not exceed the remaining term of the Eliqible Firm Transportation Agreement.
- (d) (l) Segmented releases are subject to Section 8.14 of these General Terms and Conditions.
- (2) (i) The maximum rates that may be bid and charged for a Released Firm Transportation Agreement that is for a term greater than one (1) year are the maximum lawful rates applicable to the Eligible Firm Transportation Agreement held by the Original Shipper. If the Original Shipper is paying a Negotiated Rate or a rate under a Negotiated Rate Formula pursuant to Section 49 of these General Terms and Conditions, a Qualified Bidder may not bid a rate which exceeds the applicable Recourse Rate or is less than the applicable minimum rate a qualified Bidder may bid a rate form which would be a Negotiated Rate or Negotiated Rate Formula if and only if the rate form is one explicitly recognized in Natural's Tariff as available for capacity releases (such as volumetric rates); provided the resulting charges must be within the range set by the applicable maximum and minimum rates.
- (ii) There is no maximum rate limitation applicable to bids for capacity release for a term of one (1) year or less, if the release is to take effect on or before one (1) year from the date on which Natural is notified of the release.

FERC Docket: RP08-319-000

Original Sheet No. 424 Original Sheet No. 424: Pending

GENERAL TERMS AND CONDITIONS

(3) No Replacement Shipper or Subreplacement Shipper shall have the right to change the primary Receipt or Delivery Points listed in the Eligible Firm Transportation Agreement, unless the Original Shipper and Natural agree to amend the Eligible Firm Transportation Agreement to accordingly change the primary Receipt and Delivery Points.

- (e) Release of storage rights under Rate Schedule NSS or DSS may include storage capacity in which the Releasing Shipper has gas. Releases of either storage or transportation rights under Rate Schedule DSS shall be subject to the rules set out in Section 2(h) of Rate Schedule DSS. Release of service under Rate Schedule FRSS shall be subject to the rules set out in Section 2(g) of Rate Schedule FRSS.
- (f) The value of storage gas may not be used in evaluating Qualified Bid(s) relating to the release of storage capacity. The Releasing Shipper shall be solely responsible for arranging the transfer of title of any gas in storage. In-field transfers of gas are subject to Section 9 of these General Terms and Conditions. Natural shall be held harmless by the Releasing Shipper in any title disputes involving deliveries of storage gas.
- (g) A Capacity Release Request may include the right by a Releasing Shipper to recall all or part of the capacity, and/or to reput all or part of the recalled capacity, at any time. All recalls and reputs must be made in accordance with the other provisions of Natural's Tariff.
- (h) The Releasing Shipper may withdraw its posted Capacity Release Request during an open season under this Section 19 where unanticipated circumstances justify and no minimum bid has been received; following the close of the open season, a Releasing Shipper may not reject a winning Qualified Bid. Notice of a withdrawal of a Capacity Release Request must be delivered via Natural's Interactive Website or via EDI or in writing no later than the end of the open season for the Capacity Release Request.
- (i) Releases of capacity may affect primary or inthe-path rights in conjunction with storage points under Rate Schedules DSS and NSS to the extent provided in Section $5.6\,(b)$ of these General Terms and Conditions.

FERC Docket: RP09-240-000

First Revised Sheet No. 425 First Revised Sheet No. 425

Superseding: Original Sheet No. 425

GENERAL TERMS AND CONDITIONS

- (j) A Replacement Shipper or Subreplacement Shipper may release the capacity under the provisions of this Section 19 (except as prohibited by the Federal Energy Regulatory Commission Regulations).
- $\mbox{(k)}$ Any Capacity Release Request not in compliance with this Section 19.5 and the other provisions of Natural's Tariff shall be null and void and, even if posted, may be removed from Natural's Interactive Website by Natural at any time.

19.6 OPEN SEASON EXCEPTIONS

An open season is not required for: (a) a Prearranged Release for more than (1) year at the maximum reservation charge applicable to the capacity being released, (b) a Short-term Prearranged Release, (c) a Prearranged Asset Manager Release, as defined in Section 19.2(g)(i) hereof or (d) a Prearranged Release to a Marketer Participating in a State-Regulated Retail Access Program, as defined in Section 19.2(g)(ii) hereof. A Capacity Release Request which is not subject to an open season need only contain the information required in Sections 19.4(a) and (b). Such Capacity Release Request must be delivered via Natural's Interactive Website (or in writing for posting on Natural's Interactive Website if Natural's Interactive Website system is unavailable for receiving Capacity Release Requests) sufficiently in advance so that the release may become effective under Section 19.9 before the release transaction is to commence. A Releasing Shipper may not rollover, extend or in any way continue a Shortterm Prearranged Release exempt from bidding under subsection (b) hereof with the same Replacement or Subreplacement Shipper until twenty-eight (28) days after the Short-term Prearranged Release has ended unless the Releasing Shipper complies with the Capacity Release Request provisions in Sections 19.3 or 19.4, or the rerelease qualifies for any of the other exemptions from bidding, referenced in subsections (a), (c) or (d) hereof.

19.7 POSTINGS; OPEN SEASON

(a) A Capacity Release Request received by Natural via EDI (which is applicable only for Prearranged Capacity Release Request) or through Natural's Interactive Website prior to the starting time of the open season requested by the Releasing

FERC Docket: RP09-240-000

Original Sheet No. 425A Original Sheet No. 425A

GENERAL TERMS AND CONDITIONS

Shipper in its Capacity Release Request in conformance with this Section 19 shall be posted on the Informational Postings portion of Natural's Interactive Website as requested. The posting shall contain the information contained in the Capacity Release Request, except that the minimum price in any minimum price condition requested to be held confidential by the Releasing Shipper (but not the existence of the minimum bid condition), shall be kept confidential and shall not be posted. The posting shall also

FERC Docket: RP09-240-000

First Revised Sheet No. 426 First Revised Sheet No. 426

Superseding: Original Sheet No. 426

GENERAL TERMS AND CONDITIONS

include the maximum reservation charge, if applicable, (including all reservation surcharges) applicable to the capacity subject to the Capacity Release Request, the beginning and ending time for the open season and the time the notice was posted. Natural shall post the Capacity Release Request upon receipt, unless the Releasing Shipper requests otherwise. If the Releasing Shipper requests a posting time, Natural will comply with that request as long as it comports with the deadlines set forth in this Section 19.

- (b) An open season shall consist of: (1) a one (1) hour period on a Business Day between 12:00 p.m. and 1:00 p.m. Central Clock Time or (2) any number (no fractions) of Business Days running from 12:00 p.m. Central Clock Time on a Business Day to 1:00 p.m. Central Clock Time on the following Business Day, as requested by the Releasing Shipper in its Capacity Release Request; provided, however, that any Capacity Release Request for a period of one (1) year or longer must have an open season of at least three (3) Business Days, for a period at least running from 12:00 p.m. Central Clock Time on a Business Day to 1:00 p.m. Central Clock Time two (2) Business Days later.
- (c) A Releasing Shipper may not specify an extension of an open season or the match period for a Prearranged Release. Rather, the Releasing Shipper must submit a new Capacity Release Request.

19.8 QUALIFIED BIDS FOR RELEASED CAPACITY RIGHTS

- (a) At any time during an open season, a Qualified Bidder may submit a Qualified Bid via Natural's Interactive Website (or in writing for posting on Natural's Interactive Website if Natural's Interactive Website is unavailable for receiving Qualified Bids) seeking released capacity rights under a Capacity Release Request. In addition to being prequalified for credit pursuant to Section 19.15, each Qualified Bid must include the following:
- (1) The Qualified Bidder's legal name, address, phone number, telefax number, the name and title of the individual responsible for authorizing the Qualified Bid and identification of the capacity rights for which the Qualified Bid is made;

FERC Docket: RP08-319-000

Original Sheet No. 427 Original Sheet No. 427: Pending

GENERAL TERMS AND CONDITIONS

- (2) The term for the purchase;
- (3) A Minimum Bid Volume and a Maximum Bid Volume (in Dth per day);
- (4) The fixed reservation charge and/or volumetric charge that the Qualified Bidder agrees to pay for the capacity (and if a volumetric charge, any minimum amount to be billed as a reservation charge, which must be equal to or greater than any such amount designated by the Releasing Shipper);
- (5) A statement that the Qualified Bidder agrees to all the terms and conditions of the Capacity Release Request, with only the modifications as expressly provided in its Qualified Bid, which modifications must be permitted by the Capacity Release Request and must conform with the requirements in Section 19. In the event that the Releasing Shipper has stated that Qualified Bid(s) may be contingent upon subsequent events and the Qualified Bidder submits such a contingent Qualified Bid, then the Qualified Bidder must state in full the nature of the condition and the last date by which the Qualified Bid is null and void if the contingency does not occur; and
- (6) Agreement that the Qualified Bidder is bound by the terms and conditions of the capacity award by Natural pursuant to this Section 19 to the Qualified Bidder, including Natural's standard form of Agreement covering the Rate Schedule applicable to the released capacity and the terms and conditions of the Qualified Bid and the Capacity Release Request, in accordance with Natural's Tariff.
- (b) The volume in a Qualified Bid may not be less than the minimum volume required for an Eligible Firm Transportation Agreement under Natural's Tariff. Neither the volume nor the release term specified in a Qualified Bid may exceed the maximum volume or term specified in a Capacity Release Request, unless the Capacity Release Request specifically allows otherwise. A Qualified Bidder must accept all the terms and conditions of a Capacity Release Request submitted under Section 19.4 (involving a Prearranged Release) except for the level of the reservation charge and the MDQ, unless the Capacity Release Request specifically allows otherwise.

FERC Docket: RP09-240-000

First Revised Sheet No. 428 First Revised Sheet No. 428

Superseding: Original Sheet No. 428

GENERAL TERMS AND CONDITIONS

- (c) For releases for a term of more than one (1) year, a Qualified Bidder may not bid rates which would exceed Natural's maximum reservation charge applicable to the Eligible Firm Transportation Agreement capacity. A Qualified Bid may state a dollar amount or a percentage of Natural's maximum reservation rate, unless the Releasing Shipper has specified that the bid must specify one or the other, in which case, the Qualified Bid must comport with the Releasing Shipper's choice. A bid at a rate equal to Natural's maximum reservation rate shall be a bid for Natural's maximum reservation rate as such rate may increase or decrease from time to time in the future. The maximum reservation charge includes all non-commodity charges and add-ons permitted by Natural's Tariff direct-billed charges and Order No. 636 transition costs which are applicable to the Eligible Firm Transportation Agreement capacity rights.
- (d) All Qualified Bids shall provide for payment of maximum commodity charges under Natural's Tariff for the capacity bid, as well as all other applicable add-on charges and surcharges under Natural's Tariff, such as, but not limited to, ACA, Fuel Gas and Gas Lost and Unaccounted For and any FERC Order No. 636 commodity-based transition cost recovery surcharge, unless Natural agrees in writing otherwise.
- (e) A Qualified Bid received by Natural during an open season shall be posted by Natural on the Informational Postings portion of Natural's Interactive Website upon receipt, without the name of the Qualified Bidder. A Qualified Bid may be withdrawn by the Qualified Bidder prior to the close of the open season, but may not be withdrawn thereafter. Following such withdrawal, the Qualified Bidder cannot bid for the same capacity during the open season at a lower rate.
- (f) All Qualified Bids must be consistent with all provisions of Natural's Tariff. Any Qualified Bid inconsistent with Natural's Tariff or the applicable Capacity Release Request shall be null and void.

FERC Docket: RP08-319-000

Original Sheet No. 429 Original Sheet No. 429: Pending

GENERAL TERMS AND CONDITIONS

19.9 AWARDING OF RELEASED CAPACITY; EFFECTIVE DATE; GAS NOMINATIONS

- (a) For a Prearranged Release for which no open season is required under Section 19.6 and which is received at least one (1) hour prior to a nomination deadline on a Day, Natural shall tender the contract to the Prearranged Shipper within one (1) hour after release notification, provided that all applicable provisions of this Section 19 have been complied with.
- (b) As to any other Prearranged Release, in the event there was no winning Qualified Bid(s) with a higher total Bid Value than the Prearranged Shipper's Bid Value, Natural shall notify the Prearranged Shipper. If, during an open season, the winning Qualified Bid(s) have a higher total Bid Value than the Bid Value of the Prearranged Release under the bid evaluation procedure selected by the Releasing Shipper, Natural shall notify the Prearranged Shipper of the terms and conditions of the winning Qualified Bid(s), except for any identification of the Qualified $\operatorname{Bidder}(s)$. The Prearranged Shipper may elect to match any or all of such winning Qualified Bid(s), but may not elect to match only a portion of a winning Qualified Bid. Such election shall consist of the Prearranged Shipper submitting notice to Natural of its unconditional agreement to the terms and conditions of one or more of such winning Qualified Bid(s) in writing or electronic means. In the event of a timely match, then the Prearranged Shipper shall be awarded the released capacity. To the extent that the Prearranged Shipper fails to timely match (within the required time frame) the winning Qualified Bid(s) with a higher Bid Value, then the Qualified Bidder(s) who made the winning Qualified Bid shall be awarded the capacity. The timelines for the above actions shall be as provided in Section 19.1.
- (c) For any other Capacity Release Request, the capacity rights shall be automatically awarded to the winning Qualified Bidder(s) when Natural has identified the entity(s) to receive the released capacity under this Section 19.
- (d) A capacity release shall become effective upon the awarding of capacity consistent with this Section 19. Nominations for gas service utilizing the released capacity shall be accepted at the next available nomination opportunity which occurs on or after the time the release becomes effective hereunder, consistent with 18 C.F.R. Section 284.12(c)(1)(ii); provided that nominations

FERC Docket: RP08-319-000

Original Sheet No. 430 Original Sheet No. 430: Pending

GENERAL TERMS AND CONDITIONS

cannot be effectuated prior to the beginning time specified in the release. Natural shall issue a contract to the winning Qualified Bidder within one (1) hour after the capacity has been awarded. So long as the winning bidder has pre-approved credit, that bidder can submit a nomination consistent with the above regardless of whether a contract with Natural covering the capacity awarded has been issued or executed; provided, however, that a contract must be executed under the provisions of the relevant rate schedule regarding timely execution of a contract tendered by Natural in order for a Shipper to have continued service beyond the maximum time specified for timely contract execution. If the original Shipper has already submitted a nomination on a Day under the Agreement being released, and if the Replacement Agreement covering the released capacity is effective that same Day, the original Shipper may incur overrun charges if his nomination exceeds the reduced contractual parameters under the original Agreement resulting from the release (i.e., if the original Shipper fails to reduce its nomination, or does not adequately reduce its nomination, at the first opportunity the Replacement Shipper has to nominate).

- (e) Gas nominations for transportation pursuant to released capacity are subject to the provisions of Section 8 of these General Terms and Conditions. Gas nominations by a Shipper utilizing released capacity awarded by Natural shall constitute Shipper's binding acceptance of the terms and conditions of the capacity award by Natural pursuant to this Section 19, including Natural's standard form of Agreement covering the Rate Schedule applicable to the released capacity and the terms and conditions of the Qualified Bid and the Capacity Release Request, in accordance with Natural's Tariff.
- (f) Subject to the other provisions in this Section 19, in the event that there is no Qualified Bidder or Prearranged Shipper for posted Eligible Firm Transportation Agreement capacity during an open season, no capacity release will be awarded and the Releasing Shipper shall retain the capacity sought to be released.

19.10 BID EVALUATION PROCEDURE

(a) Unless specifically requested otherwise by a Releasing Shipper in its Capacity Release Request, Qualified Bids for released capacity shall be evaluated pursuant to Sections below. Any Qualified Bid which does not meet a minimum price condition stated in the Capacity Release Request shall be rejected

FERC Docket: RP08-319-000

Original Sheet No. 431 Original Sheet No. 431: Pending

GENERAL TERMS AND CONDITIONS

outright. Any Qualified Bid with a contingency must have such contingency eliminated before 3:00 p.m. Central Clock Time following the close of the open season, unless the Releasing Shipper's offer has specified a later time; otherwise, such Qualified Bid will be rejected.

- (b) Natural shall calculate a Bid Value for each Qualified Bid and Prearranged Release (if any), and shall calculate the Winning Bid Value, as follows:
- (1) For each month, the volume and reservation charge per Dth stated in the Qualified Bid shall be multiplied together to derive a gross monthly revenue figure. If the Qualified Bids contain volumetric-based charges permitted by the Capacity Release Request, then the gross monthly revenue figure shall be equal to any minimum amount designated by the bidder to be billed as a reservation charge even if there is no (or insufficient) flow.
- (2) Each gross monthly revenue figure shall be discounted to a net present value figure as of the first day of the capacity release as sought in the Capacity Release Request, using the current Federal Energy Regulatory Commission interest rate as defined in 18 C.F.R. Section 154.501(d).
- (3) The net present value figures for the proposed release shall be summed, and such sum shall be the Bid Value.
- (c) Capacity shall be awarded among the bids, best bid first (highest Bid Value or other evaluation criteria as specified by the Shipper in the Capacity Release Offer), until all offered capacity is awarded. The best qualified bid will be awarded its Maximum Bid volume and any subsequent bids will be awarded up to their Maximum Bid Volume but no less than their Minimum Bid Volume.
- (d) In the event ties exist among Qualified Bids, the Qualified Bid submitted and received earliest by Natural's Interactive Web Site (or if Natural's Interactive Web Site is not available and the Qualified Bid is submitted in writing, the time Natural received the Qualified Bid) shall be the winning bid.
- (e) The following are examples of how (c) and (d) are applied:

FERC Docket: RP08-319-000

Original Sheet No. 432 Original Sheet No. 432: Pending

GENERAL TERMS AND CONDITIONS

Example (1) - Awarding by Best Bid

Assume: Capacity Release = 100,000/day for 5 years

Qualified Bids:

	Maximum Bid Volume	Term	Bid Price	Minimum Bid Volume
Bid (a) Bid (b) Bid (c)	40,000/day 40,000/day 40,000/day	5 years 5 years 5 years	\$.18 \$.17 \$.15	0 0 0

Winning Qualified Bids: Bid (a) has the highest Bid Value and would be awarded 40,000; Bid (b) has the next highest bid value and would be awarded 40,000; Bid (c) would be awarded the remainder of the capacity (20,000).

Example (2) - Awarding with Minimum Bid Volumes

The assumptions remain the same as in Example 1, except that we assume that Bid (c) has a Minimum Bid Volume of 40,000.

Winning Qualified Bids: Bids (a) and (b) would be allocated their Maximum Bid Volume of 40,000. Bid (c) would not be awarded any capacity since their Minimum Bid Volume is 40,000. 20,000 would remain with the Releasing Shipper.

FERC Docket: RP08-319-000

Original Sheet No. 433 Original Sheet No. 433: Pending

GENERAL TERMS AND CONDITIONS

Example (3) - Awarding using the tie-breaker

Assume: Capacity Release = 100,000/day for 5 years

Qualified Bids:

		Maximum Bid Volume	Term	Bid Price	Minimum Bid Volume	Time Bid Received
Bid Bid Bid Bid	(b) (c)	60,000/day 50,000/day 35,000/day 35,000/day	5 years 5 years 5 years 5 years	\$.18 \$.17 \$.17 \$.17	0 50,000 0 0	13:57:40 13:55:05 13:56:40 13:56:30

Winning Qualified Bid (a) receives 60,000 since it has the highest Bid Value; Bid (b) receives 0 because of its Minimum Bid Volumes; Bid (c) receives 5,000 because the bid was submitted after Bid (d); Bid (d) receives 35,000.

- (f) In no event shall this Section 19.10 result in winning Qualified Bids with a total volume in excess of the capacity specified in the Capacity Release Request.
- (g) The bid evaluation procedure set forth in this Section 19.10 shall only consider Qualified Bids to the extent they provide for an objectively quantifiable payment by the Qualified Bidder. A Qualified Bid based on a percentage of Natural's reservation charge shall be evaluated by Natural based solely on the maximum reservation charge being charged by Natural for such service as of the end of the open season.

FERC Docket: RP09-240-000

First Revised Sheet No. 434 First Revised Sheet No. 434

Superseding: Original Sheet No. 434

GENERAL TERMS AND CONDITIONS

(h) If the Releasing Shipper selected a bid evaluation procedure which is different from the procedure set forth in this Section 19.10, which procedure must comply with Section 19.5, Natural shall determine the winning Qualified Bid(s) pursuant to the Releasing Shipper's bid evaluation procedure in its Capacity Release Request and computer diskette (if any) submitted by the Releasing Shipper pursuant to Section 19.5(a).

19.11 CONFIRMATIONS; RELEASED FIRM TRANSPORTATION AGREEMENT

At the time the award of capacity under this Section 19 is posted, Natural shall send the winning Qualified Bidder or the Prearranged Shipper confirmation of the capacity release awarded to such Qualified Bidder or Prearranged Shipper. Prior to Natural awarding capacity on a Prearranged Release, the Prearranged Shipper shall confirm electronically the terms of the Prearranged Release.

19.12 COMPLETED TRANSACTIONS

By 5:00 p.m. Central Clock Time after capacity has been awarded, Natural shall post on the Informational Postings portion of its Interactive Website the name(s) of the winning Qualified $\label{eq:bidder} \mbox{Bidder}(s)\,,\,\,\mbox{identification of the winning Qualified Bid}(s)\,\,\mbox{and any}$ minimum bid conditions held confidential during the open season. The Releasing Shipper is responsible for reviewing the Qualified Bids to ensure that the released capacity was correctly awarded. The Releasing Shipper shall notify Natural of any error in the award of capacity within one business day after such posting on the Interactive Website. In the event of an error, the capacity shall be reawarded by Natural. As between Natural and the Releasing Shipper, the Releasing Shipper shall indemnify and hold Natural harmless as to any costs, damages or expenses relating to the bid evaluation procedure for which timely notice of an error was not provided to Natural by the Releasing Shipper hereunder. Natural shall correct an error in a timely fashion after receiving notice of such error from the Releasing Shipper or another person.

19.13 BILLING

(a) Natural shall bill the Replacement Shippers and the Subreplacement Shippers the rate(s) specified in the Released Firm Transportation Agreements and any other applicable charges and each such Replacement Shipper and Subreplacement Shipper shall pay the billed amounts directly to Natural. Natural shall not be responsible for billing the Replacement Shipper for any amounts attributable to

FERC Docket: RP09-240-000

First Revised Sheet No. 435 First Revised Sheet No. 435

Superseding: Original Sheet No. 435

GENERAL TERMS AND CONDITIONS

gas purchase or gas inventory volumes tied to a transportation or storage capacity release. Such charges shall be between the Releasing Shipper and Replacement Shipper. Natural shall have the right to discount the commodity rates under the Released Firm Transportation Agreement. Natural will support volumetric releases with volumetric commitments by fully accounting for volumetric and reservation components, consistent with the rules and regulations enunciated by the Federal Energy Regulatory Commission.

- (b) A Releasing Shipper shall be billed the reservation charge associated with the entire amount of released capacity pursuant to its contract rate, which includes all noncommodity based charges under Natural's Tariff for such released capacity including but not limited to additional direct-bill charges and FERC Order No. 636 transition costs, with a concurrent conditional credit for payment of the reservation charge due from the Replacement or Subreplacement Shipper(s), as applicable, which received the released capacity. A Releasing Shipper shall also be billed a marketing fee, if applicable, pursuant to the provisions of Section 20 of these General Terms and Conditions. As to any capacity released by a Releasing Shipper, the Releasing Shipper shall not be billed or be responsible for: (1) commodity charges; (2) scheduling or imbalance charges or penalties; and (3) add-on charges and surcharges applicable to Natural's commodity rates under Natural's Tariff such as ACA, Fuel Gas and Gas Lost and Unaccounted For, which are incurred by a Replacement Shipper or Subreplacement Shipper which received the released capacity.
- (c) If a Replacement Shipper or Subreplacement Shipper does not make payment to Natural of the reservation portion of the charges due as set forth in its Released Firm Transportation Agreement, Natural shall bill the Releasing Shipper(s) from whom such Replacement or Subreplacement Shipper received the capacity for the amount(s) due, including all applicable late charges authorized by Natural's Tariff, and such amount shall be paid by such Releasing Shipper within ten (10) days of the receipt of such billing, or interest shall continue to accrue. In the event that the Replacement or Subreplacement Shipper has not paid such amount(s) due by the end of such ten (10) day period, then: the Releasing Shipper has the right to recall the capacity; (2) Natural's rights against the delinquent Replacement/Subreplacement Shipper shall be subrogated to the related rights of the Releasing Shipper; and (3) Natural shall make a reasonable effort to collect from the Replacement/Subreplacement Shipper the amount(s) due. Such reasonable effort shall not include incurring costs from outside attorneys, collection agents or other third parties.

FERC Docket: RP09-240-000

First Revised Sheet No. 436 First Revised Sheet No. 436

Superseding: Original Sheet No. 436

GENERAL TERMS AND CONDITIONS

(d) All payments received from a Replacement or Subreplacement Shipper shall first be applied to reservation charges, then to late charges on reservation charges, then to penalties, then to late charges not on the reservation charges, and then last to commodity-based charges. Except as may be otherwise provided pursuant to Section 49.4 of these General Terms and Conditions, payments by Replacement or Subreplacement Shippers in excess of the total amount(s) due for the Released Firm Transportation Agreement capacity shall be a credit applied to any outstanding balance owed under any contract with Natural, or a refund if requested in writing and no such outstanding balance exists.

(e) REFUNDS DUE UNDER RELEASES GREATER THAN ONE (1) YEAR

If Natural is obligated to refund any amounts attributable to reservation charges for capacity which has been released, Natural shall make the applicable refund to the Replacement Shipper to the extent that Natural has actually received reservation charge amounts from the Replacement Shipper in excess of the amounts assessable under the revised maximum reservation charge rates (which amounts are credited to the account of the Releasing Shipper under Section 19.13(d) above). Except as may be otherwise provided pursuant to Section 49.4 of these General Terms and Conditions, Natural shall make a corresponding adjustment to the capacity release credit provided to the Releasing Shipper, and may reflect the reduced capacity release credit in a lower refund or adjusted billings to the Releasing Shipper, with any other applicable reservation charge refunds going to the Releasing Shipper. Except as may be otherwise provided pursuant to Section 49.4 of these General Terms and Conditions, commodity rate refunds shall go to the party which paid the commodity charge.

(f) REFUNDS DUE UNDER RELEASES ONE (1) YEAR OR LESS

If Natural is obligated to refund any amounts attributable to reservation charges for capacity which has been released above the maximum rate for one year or less, Natural shall make the applicable refund to the Releasing Shipper notwithstanding the amount that Natural has actually received from the Replacement Shipper (which amounts are credited to the account of the Releasing Shipper under Section 19.13(d) above). Natural shall reflect the same credit from the Replacement Shipper but shall adjust its billing to the Releasing Shipper to reflect the revised maximum reservation rates. Except as may otherwise be provided pursuant to Section 49.4 of these General Terms and Conditions, commodity rate refunds shall go to the party which paid the commodity charge.

FERC Docket: RP09-240-000

Original Sheet No. 436A Original Sheet No. 436A

GENERAL TERMS AND CONDITIONS

19.14 NOMINATIONS/SCHEDULING; RECALLS AND REPUTS

(a) RECALLING CAPACITY - GENERAL

Natural supports the ability of a Releasing Shipper to specify as a condition of a capacity release offer which recall notification periods as set out below will be available for use by the parties to the release and whether recall notices must be provided on a Business Day. All Replacement and Subreplacement Shippers shall nominate and schedule natural gas for service hereunder directly with Natural in accordance with the applicable procedures set forth in Section 8 of these General Terms and Conditions. In order for any capacity recall or capacity reput to be effective for a day, a Releasing Shipper must give prior notice of such recall or reput, and any allocation of a partial recall or reput, to Natural.

(b) RECALL NOMINATION TIMELINE

Releasing Shippers may, to the extent permitted as a condition of the capacity release, recall released capacity (scheduled or

FERC Docket: RP08-319-000

Original Sheet No. 437 Original Sheet No. 437: Pending

GENERAL TERMS AND CONDITIONS

unscheduled) at any of the daily nomination cycles consistent with the following (notice of the allocation of capacity between the Releasing Shipper and the Replacement Shipper hereunder is intended to be provided in a manner that will permit affected parties sufficient time to place nominations or take other corrective actions and thereby avoid penalties):

(1) TIMELY RECALL NOTIFICATION

(i) A Releasing Shipper recalling capacity must provide notice of such recall to Natural and to the first Replacement Shipper no later than 8:00 a.m. on the day that Timely Nominations are

(ii) Natural shall provide notification of such recall to all affected Replacement Shippers no later than 9:00 a.m. on the day that Timely Nominations are due (Central Clock Time);

(2) EARLY EVENING RECALL NOTIFICATION

(i) A Releasing Shipper recalling capacity must provide notice of such recall to Natural and to the first Replacement Shipper no later than 3:00 p.m. on the day that Evening Nominations are due;

(ii) Natural shall provide notification of such recall to all affected Replacement Shippers no later than 4:00 p.m. on the day that Evening Nominations are due (Central Clock Time);

(3) EVENING RECALL NOTIFICATION

(i) Releasing Shipper recalling capacity must provide notice of such recall to Natural and to the first Replacement Shipper no later than 5:00 p.m. on the day that Evening Nominations are due;

(ii) Natural shall provide notification of such recall to all affected Replacement Shippers no later than 6:00~p.m. on the day that Evening Nominations are due (Central Clock time);

(4) INTRADAY 1 RECALL NOTIFICATION

(i) A Releasing Shipper recalling capacity must provide notice of such recall to Natural and to the first Replacement Shipper no later than 7:00 a.m. on the day that Intraday 1 Nominations are due;

FERC Docket: RP08-319-000

Original Sheet No. 438 Original Sheet No. 438: Pending

GENERAL TERMS AND CONDITIONS

(ii) Natural shall provide notification of such recall to all affected Replacement Shippers no later than 8:00 a.m. on the day that Intraday 1 Nominations are due (Central Clock time); and

(5) INTRADAY 2 RECALL NOTIFICATION

(i) A Releasing Shipper recalling capacity must provide notice of such recall to Natural and to the first Replacement Shipper no later than 2:30 p.m. on the day that Intraday 2 Nominations are due;

(ii) Natural shall provide notification of such recall to all affected Replacement Shippers no later than 3:30 p.m. on the day that Intraday 2 Nominations are due (Central Clock time).

(6) OTHER

For recall notifications provided to Natural prior to the recall notification deadline specified in (1)-(5) of this Section 19.14(b) and received by Natural between 7:00 a.m. and 5:00 p.m., Natural shall provide notification to all affected Replacement Shippers no later than one hour after receipt of such recall notification. For recall notifications provided to Natural after 5:00 p.m. and prior to 7:00 a.m., Natural should provide notification to all affected Replacement Shippers no later than 8:00 a.m. after receipt of such recall notification. (Central Clock Time)

(c) METHODS OF NOTIFICATION

(1) The Replacement Shipper is to provide Natural with no more than two Internet E-mail addresses to be used for recall notification under Section 19.14(b) of these General Terms and Conditions. The obligation of Natural to provide notification is waived until at least one of the addresses has been provided. When Natural sends Internet E-mail notification for recalling of capacity to each affected Replacement Shipper, the subject line of the E-mail should include the following information separated by commas in the following order: (1) "Recall", (2) the recall notification period, (3) the Effected Date in YYYYMMDD format, (4) Natural's name or abbreviation (excluding commas), and (5) Natural's D-U-N-S number. The body of such E-mail notification is to contain at least the affected Replacement Shipper's Contract Number, the quantity of capacity being recalled, and the Offer Number or Award Number, if

FERC Docket: RP08-319-000

Original Sheet No. 439 Original Sheet No. 439: Pending

GENERAL TERMS AND CONDITIONS

necessary to uniquely identify the capacity being recalled. For recalls that are effective at non-standard times, the appropriate recall notification period is to be included in the subject line and the effective time of the recall is to be in the body of the E-mail. If Natural allows capacity recall notification mechanisms in addition to Internet E-mail, the notification is to include at least the same level of information. Affected Replacement Shippers are to manage internal distribution of notifications of recall received from Natural.

- (2) The Releasing Shipper shall provide capacity recall notification to Natural through Natural's Interactive Website. The Releasing Shipper shall provide notice to its affected Replacement Shipper at the same time it provides notification to Natural. The recall notification must specify the recall notification period for the specified effective gas Day, as well as any other information needed to uniquely identify the capacity being recalled. The mode of notification is to be mutually agreed upon between the Releasing and the Replacement Shipper.
- $\,$ (3) All recalled capacity notices must indicate whether penalties will apply for the gas day for which quantities are reduced due to a capacity recall.

(d) QUANTITY ALLOCATION

In the event of an intra-day capacity recall, Natural shall determine the allocation of capacity between the Releasing Shipper and the Replacement Shipper(s) based upon the Elapsed Prorata Capacity (EPC). In any recall notification provided to Natural, the quantity should be expressed in terms of the adjusted total released capacity entitlements based upon EPC. EPC means that portion of the capacity that would have theoretically been available for use prior to the effective time of the intraday recall based upon a cumulative uniform hourly use of the capacity. The amount of the capacity allocated to the Replacement Shippers(s) should equal the original released capacity less the recalled capacity that is adjusted based upon the EPC. Natural shall not be obligated to deliver in excess of the total daily contract quantity of the release as a result of any recall. The service flexibility available to either the Releasing Shipper or the Replacement Shipper for the subject capacity shall not be less as a result of the recall.

FERC Docket: RP08-319-000

Original Sheet No. 440 Original Sheet No. 440: Pending

GENERAL TERMS AND CONDITIONS

(e) REPUTS

 $\label{eq:when capacity is recalled, it may not be reput for the same gas day. \\$

(f) DISPUTES

In the event of a dispute between the Releasing Shipper and any other person as to the validity of any recall or reput, or the status of the holder of the capacity rights, Natural shall be entitled to conclusively rely on any notice provided by the Releasing Shipper. The Original Shipper, Replacement Shipper and/or Subreplacement Shipper involved in any such dispute shall indemnify and hold Natural harmless from any costs, damages or expenses relating to Natural's reliance on such notice.

19.15 QUALIFICATION FOR PARTICIPATION IN THE CAPACITY RELEASE PROGRAM

- (a) Any person wishing to become a Qualified Bidder and make a Qualified Bid must satisfy the creditworthiness requirements in Section 16 of these General Terms and Conditions prior to submitting a Qualified Bid under this Section 19. A person cannot bid for services which exceed its pre-qualified level of creditworthiness. Natural shall process—and encourages—applications from potential Qualified Bidders seeking prequalification for bids they may make in the future.
- (b) Credit applications shall be completed in full with all information required to establish creditworthiness under the credit criteria included in Natural's rate schedule covering the applicable transportation on Natural's system. Should a potential bidder fail to satisfy such credit criteria, the potential bidder may still become a Qualified Bidder by providing a prepayment, letter of credit, security interest or guarantee satisfactory to Natural as further set forth in Section 16 of these General Terms and Conditions.
- (c) Based on Natural's continuing review of a Shipper's financial records, Natural shall have the right to amend a Shipper's line of credit and lower or increase the quantity and term.

FERC Docket: RP08-319-000

Original Sheet No. 441 Original Sheet No. 441: Pending

GENERAL TERMS AND CONDITIONS

(d) Natural's determination of a Shipper's creditworthiness is solely for Natural's purposes under Natural's Tariff and such determination is neither a representation nor a guarantee to a Releasing Shipper or any other entity as to the ability of a Replacement or Subreplacement Shipper to pay any outstanding amount under a Released Firm Transportation Agreement.

19.16 COMPLIANCE BY SHIPPER

By acquiring released capacity, a Shipper agrees that it will comply with all provisions of Natural's Tariff and all applicable Commission orders, rules and regulations. Such Shipper also agrees to be responsible to Natural for compliance with all applicable terms and conditions of Natural's Tariff, as well as the terms and conditions of the Released Firm Transportation Agreement.

19.17 OBLIGATIONS OF RELEASING SHIPPER

- (a) Subject to Section 19.17(c), the Releasing Shipper shall continue to be liable and responsible for all reservation charges associated with the released capacity up to the reservation charge specified in such Releasing Shipper's Agreement with Natural. The Releasing Shipper agrees that the award of capacity to a Replacement Shipper or Subreplacement Shipper shall automatically reduce the Releasing Shipper's firm capacity rights under the Agreement with Natural effective on the effective date of the release for the period of the release, except for any period that the firm capacity is recalled by the Releasing Shipper (if the successful bid so permits) until such capacity is reput to the Replacement or Subreplacement Shipper, in accordance with this Section 19.
- (b) A release by a Replacement Shipper shall not relieve the Original Shipper or the Replacement Shipper of their obligations under this Section 19.
- (c) In the event that a Released Firm Transportation Agreement covers the remaining term of the Eligible Firm Transportation Agreement at maximum rates, then the Original Shipper may request in writing that the Original Shippers' rights and obligations under the Eligible Firm Transportation Agreement shall be prospectively assigned to, and be assumed by, the Replacement Shipper. Following such request, Natural shall send the Original Shipper and Replacement Shipper an assignment agreement to so

FERC Docket: RP08-319-000

Original Sheet No. 442 Original Sheet No. 442: Pending

GENERAL TERMS AND CONDITIONS

provide. In the event that the Original Shipper and the Replacement Shipper execute such assignment agreement, the Original Shipper shall be released from all liability under the Eligible Firm Transportation Agreement arising after such execution date.

19.18 CONVERSIONS BETWEEN MONTHLY AND DAILY RESERVATION RATES

For less than maximum rate transactions only, to convert a daily rate to a monthly rate, multiply the daily rate by the number of days in the rate period, and then divide the result by the number of months in the rate period, taking the remainder out to five (5) decimal places and rounding up or down to the decimal place used in Natural's applicable schedule of rates. To convert a monthly rate to a daily rate, multiply the monthly rate by the number of months in the rate period, and then divide the result by the number of days in the rate period, taking the remainder out to five (5) decimal places and rounding up or down to the decimal place used in Natural's applicable schedule of rates.

19.19 NATURAL'S RIGHT TO TERMINATE A CAPACITY RELEASE

Natural may elect to terminate a Replacement Shipper's Agreement with Natural upon at least thirty (30) days' prior written notice of termination to the Replacement Shipper, under the following conditions

- (a) The Releasing Shipper has failed to make timely payment or maintain credit (or provide adequate assurance of payment) in accordance in Sections 5.11, 5.12 and/or 16 of these General Terms and Conditions and Natural has suspended or terminated service to the Releasing Shipper or has provided notice under Section 5.11 or Section 5.12 which ultimately results in suspension or termination of service; and
- (b) The rate stated in the Replacement Shipper's Agreement is less than the rate for service under Natural's contract with the Original Shipper; provided, however, that a Replacement Shipper which is creditworthy can continue an existing capacity release by notifying Natural that it agrees to pay a rate which it specifies that equals or exceeds the lower of: (i) the applicable maximum rate; or (ii) the same rate as is in the original Agreement between Natural and the Releasing Shipper. Alternatively, notwithstanding Sections 19.5(d)(2) and 19.8(c) of these General Terms and conditions, Natural and the Replacement Shipper may agree upon other pricing terms, in which case the release shall continue. Such notification or agreement must be effectuated prior to the end of the notice period.

FERC Docket: RP08-319-000

Original Sheet No. 443 Original Sheet No. 443: Pending

GENERAL TERMS AND CONDITIONS

20. ADVERTISEMENT AND MARKETING FEES

20.1 ADVERTISEMENTS

Any person may advertise for the purchase of capacity on Natural's system on its Interactive Website by submitting the desired advertisement (up to one page) to Natural. Natural shall post such advertisement on the Informational Postings portion of its Interactive Website no later than the business day following receipt thereof if so requested, so long as the advertisement is not unlawful or inconsistent with Natural's Tariff. The posted period requested may be for a period of time not to exceed one month. A response in and of itself to an advertisement seeking to purchase capacity never constitutes a capacity release; to release capacity, the Shipper holding the capacity rights must utilize the release procedures set forth in Section 19 of these General Terms and Conditions.

20.2 FEE FOR ACTIVE MARKETING

When a Releasing Shipper under Section 19 of these General Terms and Conditions requests that Natural actively market capacity to be released, the Releasing Shipper and Natural shall negotiate the terms of the marketing service to be provided by Natural and the marketing fee to be charged therefor.

FERC Docket: RP08-319-000

Original Sheet No. 444 Original Sheet No. 444: Pending

GENERAL TERMS AND CONDITIONS

21. TRACKING OF CERTAIN ACCOUNT NO. 858 COSTS

21.1 PURPOSE

This Section governs the recoverability by Natural of costs reflected in Account No. 858, Transmission and Compression of Gas by Others, unless excluded from this Section, and sets out a tracking mechanism to recover certain Account No. 858 costs which Natural has been required by Commission order in Docket No. RS92-45 to remove from its initial base rates filed in compliance with Order No. 636 to be effective December 1, 1993.

21.2 ESTABLISHMENT OF INITIAL TRACKING LEVEL

- (a) Except for contracts identified in subsection (c)(1), Natural shall endeavor to reduce its Account No. 858 costs by assignment of Account No. 858 contracts pursuant to Section 3 of these General Terms and Conditions.
- (b) The costs reflected in Natural's Account No. 858 shall be recovered by a tracking mechanism as set out in Section 21.3 of these General Terms and Conditions. The level of costs to be tracked shall be based on Account No. 858 agreements in effect, adjusted for known and measurable changes, as of the date of the tracking filing, less the exclusions identified in subsection (c), as adjusted under Section 21.3(c). The initial level of collections under this tracker shall be determined in a separate filing under Section 4 of the Natural Gas Act which Natural must submit to initiate collection under this Section. Natural may submit such a filing to be effective December 1, 1993.
- (c) The Account No. 858 costs recoverable in the tracking mechanism shall exclude: (1) the costs of any Account No. 858 contracts assigned by Natural under subsection (a) of this Section; and (2) costs under any new Account No. 858 contracts entered into after December 1, 1993 or costs under any extension of an existing Account No. 858 contract where such extension was pursuant to an agreement executed after December 1, 1993. In addition, this tracking mechanism shall not reflect any out-of-period adjustments attributable to Account No. 858 activity prior to December 1, 1993.

FERC Docket: RP08-319-000

Original Sheet No. 445 Original Sheet No. 445: Pending

GENERAL TERMS AND CONDITIONS

21.3 TRACKING RATE ADJUSTMENTS

- (a) The initial tracking amount determined under Section 21.2 of these General Terms and Conditions shall become effective December 1, 1993. Effective June 1, 1994 and effective the beginning of each subsequent December 1 and June 1 thereafter while this Section 21 is in effect, Natural shall adjust its rates to reflect: (1) changes in the level of annual costs in Account No. 858 as determined under subsection (b); and (2) buyout or buydown costs as determined pursuant to Section 21.4 of these General Terms and Conditions. Semi-annual tracking filings submitted in accordance with this Section shall become effective, subject to refund, on the designated effective dates. Any changes from the prior tracking level shall be subject to review in the semi-annual tracking filing proceeding. Such review may encompass issues of the eligibility and prudence of such costs.
- (b) Changes from the prior tracking level for Account No. 858 costs shall be based on a comparison between the known and measurable annual costs of reservation, demand and other fixed charges under Account No. 858 contracts and the level of such costs reflected in the most recent prior tracking filing. The base cost level shall be the annual fixed cost level reflected in the initial tracking level in the case of the filing to be effective June 1, 1994 and the cost level reflected in the last semi-annual tracking filing in the case of any subsequent semi-annual tracking filing. Buyout or buydown costs may be reflected by an amortization consistent with Section 21.4 of these General Terms and Conditions. Such cost changes shall be reflected in a tracking filing made by Natural thirty (30) days prior to any of the effective dates specified in subsection (a).
- (c) Beginning December 1, 1993, Natural shall maintain a deferred account to reflect any differences between amounts recovered under this Section 21 and actual costs; provided, however, that for any new contracts entered into after November 30, 1993, Natural shall be deemed (notwithstanding actual collections) to have recovered an amount equal to the surcharge hereunder multiplied by the nominal contract demand, except where adjusted for discounting under subsection d(2). The net deferred balance shall be an adjustment to each semi-annual tracking filing. Account No. 858 activity associated with any period prior to December 1, 1993 shall be excluded from this deferred account. Natural shall also reflect any refunds accrued under Account No. 858

FERC Docket: RP08-319-000

Original Sheet No. 446 original Sheet No. 446: Pending

GENERAL TERMS AND CONDITIONS

arrangements covered by this Section attributable to the period this tracker is in effect, including any refund(s) credited to Natural for such period after this tracker has otherwise expired. A rate adjustment (which may be positive or negative) to eliminate the balance of this deferred account over a 6-month period commencing with the next effective date shall be included in each semi-annual tracking filing. Any over or under-recoveries in this deferred account shall be reflected in future deferred account adjustments.

- (d) Natural shall establish separate rates hereunder for the Midwest and Production Zones, as defined in Section 21.7 of these General Terms and Conditions; provided, however, that once Natural has no remaining costs to be recovered in the Production Zone due to the expiration of its underlying contracts, the last of which expired on November 30, 1997, all future rate adjustments associated with the Production Zone with a cumulative net cost effect of \$25,000 or less shall be reflected in the Midwest Zone only. If the cumulative net adjustments reallocated from the Production Zone to the Midwest Zone ever exceed a net cost of \$25,000, Natural shall reinstate the Production Zone surcharge, with all Production Zone amounts in excess of such net cumulative cost of \$25,000 being allocated to the Production Zone. Rates hereunder shall be developed and billings hereunder shall be assessed based on the demand or reservation billing units under all Part 284 firm transportation Agreements in effect sixty (60) days prior to the date the semi-annual tracking filing will go into effect [storage services other than DSS and individually certificated transportation arrangements in Volume No. 2 of this Tariff are excluded, and Shippers under Rate Schedule FTS-G ("G Customers") shall be reflected at an imputed daily contract demand equal to G Customers' annual commodity volumes divided by the product of three hundred sixty-five (365) times fifty percent (50%)], adjusted as follows:
- (1) Known and measurable changes that will be effective on the effective date of the semi-annual adjustment.
- (2) No Account No. 858 costs shall be allocated to contracts executed prior to December 1, 1993 to the extent Natural is unable contractually to collect any additional amounts attributable to Account No. 858 surcharges under such contracts nor, to the extent such contracts are so excluded, shall the billing units under such contracts be utilized in developing rates under this Section 21. No Account No. 858 costs shall be allocated to, nor shall Natural assess any charges to, FRSS Agreements.

FERC Docket: RP08-319-000

Original Sheet No. 447 Original Sheet No. 447: Pending

GENERAL TERMS AND CONDITIONS

- (3) For every other contract, Natural shall include the full MDQ in the billing determinants hereunder except to the extent Natural has discounted the Account No. 858 surcharge, in which case Natural shall include the full rate equivalent of that contract's MDQ, based on the ratio of the discounted Account No. 858 surcharge rate to the maximum Account No. 858 surcharge. In determining whether and the extent to which Natural has discounted the Account No. 858 surcharge, the discounting sequence set out in Section 15.6 of these General Terms and Conditions shall be applied.
- (e) (1) Natural's last obligation to pay Account No. 858 costs eligible for recovery hereunder expires as of December 31, 2002. In the tracking filing hereunder to be effective June 1, 2002, Natural has included estimated eligible contract costs, net of any offsets for capacity release revenue and interruptible service revenue crediting, together with estimated interest, through the expiration of its obligations. Natural will notify the Commission by November 15, 2002, whether in its judgment based on its best estimates any further tracking filing is necessary. Surcharges hereunder shall terminate effective December 2002, subject to the remainder of this Subsection (e).
- (2) In the case of underrecovery as of November 30, 2002, Natural may, and in the case of an overrecovery as of November 30, 2002, Natural must, file to continue surcharges hereunder on and after December 1, 2002, recalculated to either passback or recover, as applicable, the entire remaining balance over the new tracking period. Any such filing of surcharge extensions hereunder to be effective on and after December 1, 2002, shall amortize the remaining balance over a period of three (3) months, otherwise using procedures consistent with prior semi-annual tracking filings under this Section 21.
- (3) Natural will notify the Commission of the final Account No. 858 balance no later than January 31, 2003, or within thirty (30) days after the end of the collection period for subsequent tracking filing(s), whichever is later.
- (4) In the event of an overrecovery pursuant to a tracking filing under subsection (2) above, Natural must, and in the event of an underrecovery, Natural may, make further tracking filings consistent with subsection (2) above. Any overrecovery by Natural hereunder shall be passed back by Natural through a negative surcharge in a subsequent tracking period until the Account No. 858 balance is zero or greater. Any underrecovery by Natural hereunder may be recovered by Natural through a positive surcharge in a subsequent tracking period until the Account No. 858 balance is fully recovered. Any such tracking filing shall be made within thirty (30) days prior to the proposed effective date.

FERC Docket: RP08-319-000

Original Sheet No. 448 Original Sheet No. 448: Pending

GENERAL TERMS AND CONDITIONS

(5) Natural reserves the right to reinstitute this tracking mechanism based on Commission action regarding its rehearing request filed December 29, 1995, in Docket No. RP95-326.

21.4 BUYOUT AND BUYDOWN COSTS

If, in Natural's judgment, the costs borne by its customers will be reduced by the buyout or buydown of Account No. 858 contracts reflected in rates pursuant to this Section, Natural may enter into settlement agreements for the buyout and buydown of such contracts and may reflect the resulting costs in the semi-annual tracking filings under Section 21.3 of these General Terms and Conditions. Subject to the following sentence, the costs of any buyout or buydown shall be recovered by amortizing such cost on a levelized basis over a period of two (2) years and reflecting the amount so amortized in such semi-annual tracking filings. In the case of any buyout or buydown entered into prior to December 1, 1994 and reflected in the initial tracking filing or the tracking filings to be effective June 1, 1994 or December 1, 1994, the buyout or buydown costs shall be amortized over the remainder of the period ended November 30, 1995.

21.5 CARRYING CHARGES

Carrying costs at the interest rate allowed under the Commission's Regulations, 18 C.F.R. 154.501(d), shall be applicable to any adjustments under Section 21.3 of these General Terms and Conditions and to any amount amortized under Section 21.4 of these General Terms and Conditions. Carrying costs shall apply from the first of the month after Natural begins to incur Account No. 858 costs or makes payment of buyout or buydown costs which exceed the level of costs being recovered hereunder and shall continue until the costs are recovered. If, in developing tracking rates hereunder, Natural reflects any anticipated costs which are not in fact incurred, it shall pay interest on any funds collected until returned to the customer. Adjustments for such carrying costs shall be made in the semi-annual tracking filings under Section 21.3 of these General Terms and Conditions.

21.6 RELEASE OF ACCOUNT NO. 858 AGREEMENTS

(a) This provision shall apply to Upstream Capacity, which shall mean Account No. 858 contracts which Natural was unable to assign and is unable to buyout or buydown consistent with Section 21.4 of these General Terms and Conditions; provided, however, that (i) Upstream Capacity shall exclude the arrangements identified in Section 21.2(c)(1) hereof, and (ii) Upstream Capacity shall include

FERC Docket: RP08-319-000

Original Sheet No. 449 Original Sheet No. 449: Pending

GENERAL TERMS AND CONDITIONS

858 contracts only to the extent the costs of such contracts are reflected in Natural's jurisdictional rates. Upstream Capacity must be held by Natural's Transportation Division, not by Natural's Merchant Division, if any.

- (b) Natural shall release all or any portion of Upstream Capacity, in accordance with the capacity release program of the upstream pipeline, provided that such a release reduces the cost to Natural of such arrangements. The entity to which Natural releases its capacity shall be responsible for providing Fuel Gas or Gas Lost and Unaccounted For to the upstream pipeline. All revenue received by Natural for releases (net of any variable costs incurred by Natural from the upstream pipeline in connection with such releases) shall be credited by Natural to Account No. 858. Such a credit shall reduce Account No. 858 costs in the next semi-annual tracking filing hereunder by Natural. Natural shall first apply such revenue to offset any amounts allocated to Agreements which Natural cannot collect due to contractual constraints in such Agreements (such as offsetting underrecoveries prior to institution of the discounting adjustment on April 1, 1994); provided, however, that such revenue may not be applied to offset any revenue Natural is deemed to have collected but did not actually collect for new contracts entered into after the end of the rate case test period as $% \left(1\right) =\left(1\right) \left(1\right) +\left(1\right) \left(1\right) \left(1\right) +\left(1\right) \left(1\right) \left($ set out in Section 21.3(c). With respect to any remaining revenue, Natural shall allocate such net revenue credit in the same manner that the costs of Upstream Capacity are allocated in developing Natural's tracking rates under Section 21.3(d) of these General Terms and Conditions.
- 21.7 As utilized in this Section 21 and in Section 38 of these General Terms and Conditions, the terms "Midwest Zone" and "Production Zone" shall have the following meanings:
- (a) The Midwest Zone shall consist of all facilities north of Station 103 on the Amarillo Line and north of Station 302 on the Gulf Coast Line.
- (b) The Production Zone shall consist of all other facilities on Natural's system.

FERC Docket: RP08-319-000

Original Sheet No. 450 original Sheet No. 450: Pending

GENERAL TERMS AND CONDITIONS

22. PRE-GRANTED ABANDONMENT AND ROLLOVER RIGHTS (INCLUDING RIGHT OF FIRST REFUSAL)

22.1 GENERAL

Subject to Section 22.4, service performed by Natural under Part 284 of the Commission's Regulations shall expire, and shall be automatically abandoned, upon contract termination under: (i) any firm transportation or storage Agreement with a primary term of less than one (1) year; and (ii) any interruptible transportation or storage Agreement regardless of term. Service under any firm transportation or storage Agreement with a term of one (1) year or greater shall expire, and shall be automatically abandoned, on contract termination unless service is continued pursuant to Sections 22.2, 22.3 or 22.4. Unless Natural and Shipper expressly agree otherwise in the Agreement(s), a Shipper who has entered into a limited-term firm service Agreement(s) pursuant to Section 5.1(c)(11) of these General Terms and Conditions may not elect to extend such limited-term agreement pursuant to the provisions of this Section 22; provided, however, that if the term of the limited-term agreement is for a period of one year or more, the limited-term agreement shall have Right of First Refusal rights extended up to the in-service date of the expansion project. Rollover rights and the Right of First Refusal may also be restricted pursuant to Section 5.1(c)(12) of these General Terms and Conditions.

22.2 ROLLOVERS

- (a) A Shipper under Rate Schedule DSS, FTS, FFTS, FTS-G, FRSS or NSS which has entered into an Agreement thereunder with a term of three (3) years or longer shall have a unilateral right to continue receiving service on a firm basis beyond the term of the existing Agreement and to rollover such Agreement subject to the following conditions:
- (1) Shipper must give notice at least twelve (12) months prior to the termination of its existing Agreement of its desire to continue receiving service beyond the expiration date of the existing Agreement; provided, however, that for Agreements expiring November 30, 1995 or December 1, 1995, notice may be given no later than May 5, 1995. Such notice must be in a form specified by Natural and must indicate a desired term of service [which term must be at least three (3) years] and the desired MDQ, in total and at each primary point. Such MDQ must be equal to or less than the existing MDQ in total and at each primary point and the primary

FERC Docket: RP08-319-000

Original Sheet No. 451 Original Sheet No. 451: Pending

GENERAL TERMS AND CONDITIONS

point path under the request must not result in a rate reduction vis a vis the path in the existing Agreement; provided, however, that a Shipper shall have the one-time right to adjust its path to the extent provided in Appendix C to the "Stipulation and Agreement on Reserved Issues" filed February 8, 1995, in Docket No. RP93-36. Any request for an increase in MDQ in total or at any primary point shall be treated as a request for new service, but only to the extent of the increase. Any notice hereunder specifying a decrease in MDQ in total or at any primary point shall not affect the existing Agreement during the remainder of its term.

(2) Within ninety (90) days after receipt of the notice described in (1) above, Natural will have evaluated the creditworthiness of Shipper. If Shipper meets the requirements of Natural's credit appraisal procedures and its request conforms to the requirements of (1), Natural shall tender to Shipper a rollover Agreement which conforms to the requirements of this Tariff. Shipper and Natural shall execute such rollover Agreement, or any modified Agreement upon which Natural and Shipper may mutually agree which is not inconsistent with this Tariff, within ninety (90) days thereafter. Failure to execute a conforming rollover Agreement shall constitute a waiver by Shipper of its rights under this Section 22.2.

(3) Except as expressly agreed to the contrary between Natural and Shipper, no discount or other special terms previously in effect shall apply to a rollover Agreement.

(b) A former sales customer of Natural under Rate Schedule DMQ-1 or G-1 which converted to other service(s) effective December 1, 1993, pursuant to the restructuring procedures of Natural's Tariff then in effect shall have the unilateral right to rollover any Agreement initially entered into on conversion regardless of its term, but such unilateral rollover right shall not extend to the rollover Agreement, which Agreement must be evaluated on a stand-alone basis as to any further rollover. In addition, a converting sales customer which elected service under Rate Schedule S-2 or S-2/G pursuant to the restructuring provisions of Natural's Tariff shall have a one-time right, upon termination of the S-2 or S-2/G Agreement entered into upon conversion, to elect to rollover all or part of such Agreement into service under Rate Schedule FTS or (in the case of a "G" Shipper) FTS and/or FTS-G, in any combination which results in an aggregate rollover

FERC Docket: RP08-319-000

Original Sheet No. 452 Original Sheet No. 452: Pending

GENERAL TERMS AND CONDITIONS

contract MDQ under FTS or (where applicable) FTS-G equal to or less than the original contract MDQ. Such Shipper may, but is not required to, also rollover the storage portion of such Agreement (whether or not all or any part of such contract was rolled over into FTS or FTS-G service) into the corresponding level of service under Rate Schedule NSS. Finally, any Shipper with an Agreement which is eligible for rollover hereunder which entered into a new firm Agreement as a result of transitional procedures which were in effect prior to October 1, 1999 shall have both unilateral rollover rights and the right of first refusal under the new Agreement regardless of its term, but such right shall not extend to a subsequent rollover Agreement which must be evaluated on a standalone basis as to further rollover. Any rollover right under this subsection (b) must be exercised pursuant to a notice which conforms to Section 22.2(a) or Section 22.3, as applicable.

- (c) If a Shipper submits a written notice which conforms to Section 22.2(a), such notice shall be binding on Shipper. Shipper and Natural shall be obligated to sign an Agreement consistent with such notice; provided, however, that nothing herein shall be deemed to preclude the parties from negotiating a more acceptable Agreement by mutual consent which is not inconsistent with this Tariff.
- (d) If an Agreement is rolled over pursuant to this Section 22.2, the Right of First Refusal under Section 22.3 shall not apply.

22.3 RIGHT OF FIRST REFUSAL

(a) Any Shipper under a DSS, FTS, FFTS, FTS-G, FRSS or NSS Agreement with a term of one (1) year or greater shall have the right to continue receiving service after the expiration of its existing Agreement if, pursuant to the Right of First Refusal procedures set forth in this Section 22.3, it matches the price and term offered for such service by any other bidder; provided, however, that (irrespective of the price offered by the existing Shipper or any bidder) Natural shall not be required to provide service at a discount from its applicable maximum rate unless it otherwise agrees; and, provided further that if a bid is submitted for a Negotiated Rate or Negotiated Rate Formula under Section 49 of these General Terms and Conditions, the existing Shipper need match only the lesser of the value of the bid at the Negotiated

FERC Docket: RP08-319-000

Original Sheet No. 453 Original Sheet No. 453: Pending

GENERAL TERMS AND CONDITIONS

Rate or Negotiated Rate Formula or the value of that bid utilizing the Recourse Rate in lieu of the Negotiated Rate or Negotiated Rate Formula consistent with said Section 49. A bidder may only submit a Negotiated Rate or Negotiated Rate Formula bid with the prior written agreement of Natural or if the existing Agreement contains a Negotiated Rate or Negotiated Rate Formula provision in the same rate form.

(b) To exercise the Right of First Refusal, Shipper must provide Natural with notice of its intent to do so in a form specified by Natural and must submit such notice on or before the later of: (i) six (6) months prior to the expiration of the existing Agreement; or (ii) one (1) month after receiving written notice from Natural that the Shipper may utilize these Right of First Refusal procedures; provided, however, that a Right of First Refusal may not be exercised later than three (3) months prior to the expiration of the Agreement unless mutually agreed between Natural and the Shipper. The Shipper's notice must specify a desired term of service and the desired MDQ in total and at each primary point. If the requested MDQ is greater than the existing MDQ in total and at each primary point, any such increase shall be treated as a request for new service under the applicable Rate Schedule and only the original MDQ shall be subject to the Right of First Refusal under this Section. The Right of First Refusal may apply to a portion of the original Shipper's then effective service. Any notice specifying a decrease in MDQ in total or at any point shall not affect the existing Agreement during its remaining term. Natural is not obligated to accept any notice which specifies a primary point path which results in a reduced rate as compared with the primary point path in the existing Agreement; provided, however, that a Shipper shall have the one-time right to adjust its path to the extent provided in Appendix C to the "Stipulation and Agreement on Reserved Issues" filed February 8, 1995 in Docket No. RP93-36.

(c) Within fifteen (15) days after receipt of a notice under Section 22.3(b), Natural shall post on the Informational Postings portion of its Interactive Website an Announcement of Capacity Availability Subject to Right of First Refusal (Capacity Announcement) which shall: (i) specify the original Shipper's service rights; (ii) indicate the availability of such service as of the date the existing Agreement expires, subject to the Right of First Refusal; (iii) state the maximum rate applicable to such service; (iv) set out any other information required by this Section; and (v) solicit bids for such service. Such Capacity Announcement shall be maintained, and bids accepted via Natural's Interactive Website or other mutually agreed means, for a period of one (1) month from the initial posting.

FERC Docket: RP08-319-000

Original Sheet No. 454 Original Sheet No. 454: Pending

GENERAL TERMS AND CONDITIONS

(d) (1) Within one (1) week after the end of the one month period during which the Capacity Announcement is posted, Natural shall convey to the original Shipper a term sheet for the best bid (based on price and term) which would qualify for such service in all respects (including meeting applicable credit criteria), which is a bona fide bid and which Natural is willing to accept. Natural may, but is not required to, accept any bid which reflects a discount from the applicable maximum rate. In assessing which is the best bid if more than one bid is received, Natural shall apply the same capacity evaluation criteria as are currently posted for that capacity (except that contract terms in excess of five years shall not increase the value of any bid). If a bid is received for a Negotiated Rate or Negotiated Rate Formula pursuant to Section 49 of these General Terms and Conditions which would result in a higher value for that bid than if the Recourse Rate was used, the value of the bid shall be assessed utilizing the Recourse Rate in lieu of the Negotiated Rate of Negotiated Rate Formula consistent with said Section 49. In comparing bids hereunder or in assessing rollovers under Section 22.2, Natural shall not take into account as increasing the value of any bid any separate amount paid, or which a Shipper agrees to pay, for the LN and/or SW service option under Rate Schedule FTS, FFTS or FTS-G. In addition, Natural shall not take into account as increasing the value of any bid any amount paid, or which a Shipper agrees to pay, for any surcharge from which the original Shipper is exempt under Section 2.7 of these General Terms and Conditions or for Supply Realignment Costs to the extent the Agreement for the original Shipper is within the level of contract covered by settlements as set out in Section 38.4 or 38.5 of these General Terms and Conditions.

(2) Where firm capacity which is subject to the Right of First Refusal was conditionally awarded pursuant to an Early Open Season under Section 5.1(c)(2) of these General Terms and Conditions, then the following conditions apply: (1) the winning bid in the Early Open Season establishes a reserve price for the Right of First Refusal, and (ii) the Shipper exercising the Right of First Refusal must, to retain the capacity, match the higher of the winning bid in the Early Open Season or the highest bid received in the Right of First Refusal process. The fact that capacity has been conditionally awarded in an Early Open Season shall not impose on the existing Shipper any obligation to exercise its Right of First Refusal at any earlier time than is provided for in Section 22.3(b) of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 455 Original Sheet No. 455: Pending

GENERAL TERMS AND CONDITIONS

(3) Natural's term sheet shall contain any and all terms of the bid but shall not identify the bidder; provided, however, such bid sheet shall indicate if the best bid was submitted by an affiliate of Natural. Except for the providing of such term sheet to the original Shipper, all terms and conditions of any bid and the identity of the bidder shall remain confidential; provided that the Commission may on request have access to such information on a confidential basis.

- (4) The original Shipper shall have two (2) weeks to notify Natural whether or not it desires to match the best bid. To match the best bid, the original Shipper must agree to a price (up to the applicable maximum rate or Recourse Rate) and a term (up to five years) which at least equals the bid on all or any portion of the service the original Shipper desires to retain; provided, however, that if the original Shipper seeks to retain only a portion of its MDQ, the analysis of whether the original Shipper has matched the best bid may take into account the MDQ requested under the best bid relative to the MDQ which the original Shipper seeks to retain. The original Shipper may provide a counteroffer which contains either a higher price than the best bid or a longer term than the best bid to offset a shorter term or a lower price than that offered in the best bid. Natural shall determine whether such a counteroffer constitutes a match, utilizing the same criteria as were applied to determine the best bid.
- (e) (1) If the original Shipper matches the best bid, it shall be entitled to continuation of service and shall be obligated to sign an Agreement tendered by Natural which reflects the best bid or any counteroffer by the original Shipper which matches such best bid.
- (2) If the existing Shipper fails to match the best bid, the existing Agreement shall terminate at the later of (i) the end of its term or (ii) the date Natural executes a contract with the winning bidder. Service to the existing Shipper shall be automatically abandoned upon such termination.
- (3) Submission of a bid shall be binding on the bidder. The bidder submitting the best bid shall be obligated to sign an Agreement reflecting its bid if the original Shipper fails to match.

Effective Date: 11/09/2009 Status: Effective

FERC Docket: RP09-855-002

First Revised Sheet No. 456 First Revised Sheet No. 456

Superseding: Original Sheet No. 456

GENERAL TERMS AND CONDITIONS

(f) In the absence of a qualified bid, the rate (within applicable maximums and minimums) and the term shall be negotiated between Natural and the Shipper. Capacity must be awarded consistent with the posted bid criteria and for a value no lower than previously submitted qualifying bids. No discount or other special terms shall apply to a rollover Agreement unless Natural and Shipper mutually agree. Shipper may require that Natural enter into an Agreement to provide service at the applicable maximum rate for a term specified by Shipper and running from the date the existing Agreement expires. Shipper shall have ten (10) days from date on which Natural notifies it that no acceptable bids were received to exercise the right set out in the prior sentence or to negotiate a rollover Agreement with Natural. If the Shipper fails to exercise such right on a timely basis, the right shall expire, in which case Natural shall post the capacity as available and the negotiation procedures hereunder shall terminate. Unless Shipper so elects, service hereunder shall be terminated and automatically abandoned.

(g) If the Shipper is eligible to receive continued service under this Section 22.3, Natural shall tender a rollover Agreement which conforms to the requirements of this Tariff prior to the expiration of the existing Agreement. Shipper and Natural shall execute such rollover Agreement, or any modified Agreement upon which Natural and Shipper may mutually agree which is not inconsistent with this Tariff, within two (2) weeks. If Shipper fails to execute the rollover Agreement on a timely basis, Shipper shall (in addition to all other remedies available to Natural for Shipper's failure to fulfill its obligation to execute such Agreement) forfeit any right to continuation of service after the expiration of the existing Agreement.

22.4 CONTRACTUAL EXTENSION RIGHTS

The term of service under any firm or interruptible transportation or storage Agreement may be extended pursuant to a rollover, evergreen or right of first refusal provision in such Agreement, which provision supersedes any otherwise applicable rollover or right of first refusal pursuant to this Section. The parties may negotiate rollover, evergreen or right of first refusal provisions which differ from this Section. Natural is not obligated to offer or agree to any such rollover, evergreen or right of first refusal provisions; provided, however, that to the extent it offers or agrees to any such provision, it must do so on a non-discriminatory basis for similarly situated Shippers. Natural posts such contractual rollover, evergreen or right of first refusal provision on its Transactional Report in accordance with Section 284.13 of the Commission's regulations.

FERC Docket: RP08-319-000

Original Sheet No. 457 Original Sheet No. 457: Pending

GENERAL TERMS AND CONDITIONS

22.5 VALID REQUEST CRITERIA

Unless waived by Natural, the requirements for a valid request under the applicable Rate Schedule (including the applicable credit analysis) apply to any Agreement entered into pursuant to this Section 22.

22.6 FURTHER ROLLOVER

Any Agreement entered into pursuant to this Section 22 shall be evaluated on a stand-alone basis hereunder for purposes of determining whether it, in turn, is eligible for rollover or the Right of First Refusal under this Section.

FERC Docket: RP08-319-000

Original Sheet No. 458 Original Sheet No. 458: Pending

GENERAL TERMS AND CONDITIONS

23. OPERATIONAL CONTROL

23.1 GENERAL

(a) Natural shall endeavor to maintain adequate pressure throughout its system and to preserve the overall operational integrity of its system; provided, however, that Natural shall not be obligated to buy or sell gas or to install additional compression or otherwise modify its system for these purposes. Operating personnel for Shippers and other entities which are physically taking delivery of gas from Natural or tendering gas to Natural shall cooperate with Natural in furtherance of this Section. Each Shipper shall designate the telephone and telefax numbers of one or more persons [but not more than two (2) primary and two (2) backup persons] for Natural to contact on operating matters (including the receipt of Operational Flow Orders and notices of a Critical Time) at any time, on a 24-hour a day, 365-day a year basis. Such contact persons must have adequate authority and expertise to deal with such operating matters.

(b) For the purpose of these General Terms and Conditions, the overall operational integrity of Natural's system shall encompass the integrity of the physical system and the preservation of physical assets and their performance (including the capability and deliverability of storage reservoirs), the overall operating performance of the entire physical system as an entity (or any portion thereof), and the maintenance (on a reliable and operationally sound basis) of total system deliverability and the quality of gas delivered.

23.2 FACILITY CONTROL

Natural shall maintain actual physical and operational control of all storage, transmission and other facilities on its system.

23.3 OPERATING PLAN

(a) Natural shall, on an annual or such more frequent basis as Natural deems necessary for proper operation of its system, prepare and circulate to all Shippers an Operating Plan. Such Operating Plan shall set out a plan for operation of Natural's system on a basis which approaches an optimum level, given ordinary

FERC Docket: RP08-319-000

Original Sheet No. 459 Original Sheet No. 459: Pending

GENERAL TERMS AND CONDITIONS

engineering and operating constraints, and for maintenance of service at projected levels to all Shippers, consistent with applicable priorities of service. The Operating Plan may include, inter alia, a specification of:

- $\,$ (i) $\,$ Desired levels of flow gas for each category of Shipper during various periods of the year.
- $\,$ (ii) A schedule of desired storage injections and withdrawals to maintain storage deliverability and protect the integrity of Natural's reservoirs.
- $\,$ (iii) Plans to deal with specific contingencies Natural anticipates may occur during the period covered by the Operating Plan.
- (b) The Operating Plan shall be for planning and informational purposes only and shall not be binding on Natural or on any Shipper. It is further understood that, since all elements of an Operating Plan are subject to continual change and are contingent on the actions of numerous Shippers, the Operating Plan may not necessarily be an accurate depiction of Natural's system at any point in time.
- (c) Natural may request that Shippers periodically provide non-binding estimates of flow patterns, injections and withdrawals and other operating parameters. Such information may be used by Natural in preparation of the Operating Plan or in testing whether an Operating Plan previously circulated requires revision.

23.4 OPERATIONAL CONTROL SEQUENCE

In the event Natural's observations or projections indicate that a situation is or may be developing in which adequate pressures may not be maintained or the overall operational integrity of its system (or any portion thereof) could be threatened, or in the event that such a situation actually occurs, Natural is empowered by this Section to take action to alleviate this situation. In responding to such a situation, Natural shall first apply the Advisory Action procedures of Section 23.5. If such measures are not sufficient in Natural's judgment to address the situation fully, Natural shall next employ Operational Flow Orders as provided in Section 23.6. In the event Operational Flow Orders alone are not adequate, Natural may invoke the Critical Time

FERC Docket: RP08-319-000

Original Sheet No. 460 Original Sheet No. 460: Pending

GENERAL TERMS AND CONDITIONS

procedures set out in Section 23.7. Finally, Natural may take unilateral action as provided in Section 23.8. The procedures set out in such provisions, and their sequencing, are intended to be applied only to the extent any of the specific actions indicated, or such sequencing, would tend to alleviate the situation to be addressed. In issuing Advisory Actions, Operational Flow Orders or a Critical Time, Natural shall describe the conditions and the specific responses required from the affected parties. To the extent practicable, Natural shall direct its actions hereunder to Shippers creating or anticipated to create the situation to be addressed and shall act consistent with Section 23.9. Nothing herein shall preclude Natural from bypassing any of the above procedures if, in its judgment, the situation so requires. Natural will keep Shippers advised through posting on the Informational Postings portion of Natural's Interactive Website on the status of the situation.

23.5 ADVISORY ACTIONS

In the event Natural determines that action is required to avoid a situation in which system pressure is not maintained or in which the overall operational integrity of the system or any portion of the system is jeopardized, Natural may take the Advisory Actions set out herein to forestall the development of such a situation.

- (a) Natural may request Shippers or other entities affecting its system to take any of the following actions, or other similar actions, to the extent such actions would tend to alleviate the situation, on a voluntary basis:
- $\hbox{(i)} \qquad \hbox{Increase or decrease the storage/supply mix} \\$ of deliveries;
- (ii) Shift receipts to obtain better capacity balance between pipeline systems;
 - (iii) Change Receipt or Delivery Points;
- (iv) Change usage patterns (e.g., end users switch to alternate fuels);
- $\hbox{(v)} \qquad \hbox{Provide assistance from market area} \\ \hbox{resources (such as customer storage);} \\$
- (vi) Activate pre-negotiated voluntary
 arrangements under which gas is diverted from one Shipper to another
 or from a non-Shipper to a Shipper (which arrangements may specify
 appropriate compensation);

FERC Docket: RP08-319-000

Original Sheet No. 461 original Sheet No. 461: Pending

GENERAL TERMS AND CONDITIONS

(vii) Reconcile transportation imbalances;

and/or

 $% \left(v_{1},v_{2}\right) =0$ (viii) Such other voluntary action as would tend to alleviate or forestall the situation.

- (b) Natural may also take actions within its control which might tend to alleviate or forestall the situation. Such actions may include the following:
- (i) Advise any Shipper which is not maintaining receipts and deliveries in balance (after taking into account such Shipper's storage rights) that such imbalances must not continue;
- $\hbox{(ii)} \qquad \hbox{Curtail or require adjustments or supply shifts in ITS service;} \\$
- (iii) Curtail or adjust injections or withdrawals under Rate Schedules PALS and/or BESS;
- (iv) Curtail or adjust interruptible injections or withdrawals under Rate Schedules DSS and/or NSS; and/or
- (v) Take such other actions as are within Natural's control and discretion to alleviate or forestall the situation.

23.6 OPERATIONAL FLOW ORDERS

(a) (1) In the event that, in Natural's judgment, the Advisory Actions under Section 23.5 are not sufficient to alleviate conditions, inter alia, which threaten or could threaten the safe operations or system integrity of Natural's system or to maintain operations required to provide efficient and reliable firm service, Natural is authorized to issue Operational Flow Orders. Notwithstanding the foregoing, Natural shall take reasonable actions to minimize the issuance and the adverse impact of Operational Flow Orders, or of any other measure taken under this Section 23 in response to adverse operational events on Natural's system, including issuance of Operational Flow Orders which, to the extent practicable and where appropriate under the circumstances, are directed only to the specific Shipper or Shippers whose actions caused or contributed significantly to the conditions which required Natural to issue the Operational Flow

FERC Docket: RP08-319-000

Original Sheet No. 462 Original Sheet No. 462: Pending

GENERAL TERMS AND CONDITIONS

Order. Natural will issue Operational Flow Orders only if necessary to maintain the pressure of its system within the range of normal operating parameters, respond to changes (including anticipated changes) in weather conditions or respond to or prevent facility outages or other conditions which could have a detrimental impact on system reliability or service integrity on its system. Natural may extend an Operational Flow Order to allow replenishment of market storage after a period of significant withdrawals if necessary to restore market storage fields to normal operating parameters and protect the deliverability of market storage fields, but only to support and protect Natural's firm service obligations. Operational Flow Orders shall be lifted as soon as practicable once such conditions (including the necessary replenishment of storage described in the prior sentence) no longer prevail.

- (2) The Operational Flow Order shall identify with specificity the situation to be addressed and shall (in addition to mandating specific actions) indicate voluntary actions by Shippers (withdrawals/injections/increased takes or receipts/decreased takes or receipts, etc.) which would alleviate such situation. Operational Flow Orders shall also specify the time when compliance must be achieved. Such Operational Flow Orders may, subject to Section 23.11, require a Shipper to take any of the following actions, or similar actions, to the extent such actions would tend to alleviate the situation to be addressed:
- (i) In the case of a storage Shipper, inject gas into storage in accordance with a schedule ordered by Natural. Injections may be required at any time during the year.
- (ii) In the case of a storage Shipper, withdraw gas in accordance with a schedule ordered by Natural. Withdrawals may be required at any time during the year.
- (iii) Commence or increase supply inputs into Natural's system or at specific points, or shift such supply inputs (in whole or in part) to different points.
- $% \left(\frac{1}{2}\right) =0$ (iv) Cease or reduce supply inputs into Natural's system or at specific points.
- $$\rm (v)$$ Commence or increase takes of gas from Natural's system or from specific points, or shift takes to different points.

FERC Docket: RP08-319-000

Original Sheet No. 463 Original Sheet No. 463: Pending

GENERAL TERMS AND CONDITIONS

 $% \left(vi\right) =\left(vi\right) =\left(vi\right) -\left(vi\right) =\left(vi\right) +\left(vii\right) +\left(vi$

- (vii) Reconcile transportation imbalances.
- (viii) Change the storage/supply delivery mix.
- $\mbox{(ix)}$ Such other actions as are within Shipper's control which would tend to alleviate the situation to be addressed.
- (3) No Shipper will be required under an Operational Flow Order to exceed its total firm MDQ (less firm DSS, FRSS and/or NSS storage rights, if applicable) under its Agreements with Natural under Part 284 of the Commission's Regulations. Nor will a Shipper be required to accept delivery of gas which the Shipper cannot use in its plant or service territory. Further, a Shipper subject to an Operational Flow Order issued to increase deliveries at Receipt Points will have the option to decrease takes at Delivery Points by a like amount instead, and vice versa. Similarly, a Shipper subject to an Operational Flow Order issued to decrease deliveries at Receipt Points will have the option to increase takes at Delivery Points by a like amount instead, and vice versa.

An entity other than the Shipper may agree to take responsibility for Operational Flow Orders on behalf of a Shipper or a point operator or on behalf of the operator of a specified end-use facility. To accept such responsibility, the entity must provide Natural with written notification which includes: written consent by the affected Shipper or operator; (ii) specification of any limits on the entity's responsibility; (iii) the term over which the notification is effective; and (iv) an identification of adequate resources (such as no-notice services at a point) which will be used to support the assumption of responsibility. Any notification or change in notification is effective on the second Business Day after delivery to Natural. Notwithstanding the foregoing, Natural may reject a notification upon written notice to such entity and any affected Shipper if such notification fails to comply with this Section or contains any provision which could adversely affect implementation of an Operational Flow Order or if the entity is not in a position to accept responsibility either operationally or financially for

FERC Docket: RP08-319-000

Original Sheet No. 464 Original Sheet No. 464: Pending

GENERAL TERMS AND CONDITIONS

Operational Flow Orders. Such a rejection must be made by Natural within two (2) Business Days after the notification is tendered to Natural. If a notification is already in effect and Natural reasonably determines that the conditions for such a notification are no longer met, Natural may provide notice of rejection, which notice shall be effective within five (5) Days or such later time as is specified in the notice. As part of any rejection of a notification hereunder, Natural shall provide to the Shipper and to the entity which submitted the notification a written explanation of the rejection. Such explanation must be consistent with this paragraph. If such a notification is in effect, Natural will direct any Operational Flow Order within the scope of the notification to the entity agreeing to take responsibility.

- (b) In issuing Operational Flow Orders to correct problems with either too much gas or insufficient gas being received visa-vis deliveries, Natural will generally follow the following sequence, to the extent there is sufficient time:
- (1) Natural will require all Shippers out of balance to the detriment of the system to balance their Agreements.
- $$\rm (2)$$ Natural will seek voluntary action from Shippers, subject to the Shipper and Natural negotiating adequate compensation.
- (3) Natural will interrupt interruptible services if that will restore system flexibility prior to issuance of generally applicable Operational Flow Orders or curtailment of firm services. This step will be taken when Shippers are failing to comply with previously issued Operational Flow Orders or when Natural cannot identify which Shippers are creating the problem.
- (c) Notice of an Operational Flow Order will be provided through Electronic Notice Delivery consistent with GISB Standards as adopted in Section 48 of these General Terms and Conditions and will be posted on the Informational Postings portion of Natural's Interactive Website. Natural shall also post, as soon as available, information about operational parameters which affect when an Operational Flow Order will begin and end (e.g., significant changes in pressure on any pipeline segment, status of facility repairs, etc.). Natural shall provide as much advance warning as possible of: (i) conditions which may create the

FERC Docket: RP08-319-000

Original Sheet No. 465 Original Sheet No. 465: Pending

GENERAL TERMS AND CONDITIONS

need to issue an Operational Flow Order; and (ii) the issuance, termination or modification of an Operational Flow Order. Natural will endeavor to provide and post the notice that it will issue an Operational Flow Order before 4:00 p.m. Central Time or otherwise will endeavor to notify Shippers by $4:00\ \mathrm{p.m.}$ Central Time that they should check the Informational Postings portion of Natural's Interactive Website again at a specified later time to see whether an Operational Flow Order will be in effect for the next day. In addition, Natural must attempt to give actual notice of an Operational Flow Order via telefax or telephone (provided a Shipper has given the numbers to Natural as required in Section 23.1) at least four (4) hours prior to the start of the day before an Operational Flow Order will be effective as to a Shipper. Such notice shall specify the anticipated duration of the Operational Flow Order. An Operational Flow Order will be effective at the start of a day and will continue until the end of the day and through the end of successive days until Natural notifies Shippers that the Operational Flow Order has been lifted.

(d) If any Shipper fails to comply with an Operational Flow Order [other than those described in subsection (e) below] during any period which is not a Critical Time, it shall be subject to a penalty of \$5.00 per Dth times any volume of gas by which it deviated from the requirements of the Operational Flow Order. A Shipper shall be exempt from such a penalty under this Section 23.6 to the extent the Operational Flow Order requires action beyond Shipper's contract limits under its Agreement with Natural or if Shipper has complied within a reasonable range, which range will be specified in the Operational Flow Order.

FERC Docket: RP08-319-000

Original Sheet No. 466 Original Sheet No. 466: Pending

GENERAL TERMS AND CONDITIONS

(e) In addition to the Operational Flow Orders described in subsection (a) above, Natural may issue Operational Flow Orders as follows:

(1) In order to improve system operations, Natural may require a Shipper which has a variance of twenty percent (20%) or more between actual deliveries to Natural at a Receipt Point and the confirmed nomination at that Receipt Point to conform the deliveries to the confirmed nomination as of the day that commences with the effectiveness of the Operational Flow Order.

(2) Natural may also require Shippers to keep any variances between actual flows and confirmed nominations at all of the Receipt Points or Delivery Points in a Rate Zone under all of a Shipper's Agreements, which are to the detriment of Natural's system, within a ten percent (10%) tolerance. If a Shipper ("Swing Shipper") agrees to keep all other Shippers at a Delivery Point (or, if applicable, a Central Delivery Point) whole with their nominations, such other Shippers' confirmed nominations and actual takes at that point will be included in the total confirmed nominations and the total actual deliveries of the Swing Shipper at all Delivery Points in the Rate Zone for purposes of determining the Swing Shipper's variance.

Shippers that fail to comply with such Operational Flow Orders described in this subsection (e) shall be subject to a penalty of \$10.00 per Dth outside of a ten percent (10%) tolerance.

- (f) A Shipper shall not be subject to overrun penalties and charges from Natural or Balancing Service Charges with respect to any action taken in reasonable conformance with an Operational Flow Order issued by Natural during a Critical or non-Critical Time.
- (g) Except as provided in Section 6(d) of Rate Schedule IBS, any Receipt or Delivery Point where the variance between the total confirmed nominations and actual flows is less than 1,000 Dth per day or where Natural does not have either daily metering or access to such information shall be exempt at all times, including during a Critical Time, from Operational Flow Orders requiring that gas flows be conformed to nominations or to some percentage of nominations.
- (h) Within thirty (30) days after an Operational Flow Order has been lifted, Natural will post on its EBB a report which describes the specific operational factors which caused the Operational Flow Order to be issued and to be lifted.

FERC Docket: RP08-319-000

Original Sheet No. 467 Original Sheet No. 467: Pending

GENERAL TERMS AND CONDITIONS

23.7 CRITICAL TIME

- (a) Natural shall advise Shippers on its system if it is declaring a Critical Time, as described in Section 12.5 of these General Terms and Conditions, and shall specify the nature of the situation creating the Critical Time.
- (b) Natural may issue Operational Flow Orders as described in Section 23.6 during a Critical Time. Natural will waive (i) injection storage charges under Rate Schedules DSS and NSS for Shippers that provide extra gas to the system during a Critical Time when there is a need for more gas, and (ii) withdrawal storage charges under Rate Schedules DSS and NSS for Shippers that increase takes from the system during a Critical Time when there is a need to reduce the amount of gas on the system.
- (c) The penalty for failure to abide by an Operational Flow Order issued during a Critical Time shall be equal to the volume (in Dth) by which Shipper deviated from the requirement of the Operational Flow Order multiplied by the greater of (i) \$15.00 or (ii) the sum of three hundred percent (300%) of the highest authorized overrun rate which can be applicable to an FTS Agreement on Natural's system pursuant to Section 12 of these General Terms and Conditions plus three hundred percent (300%) of the highest Weekly Index Price used to determine the Average Monthly Index Price (as determined in accordance with Section 13) for the month in which the violation occurred. Additional amounts may be billed pursuant to Section 23.10.

23.8 UNILATERAL ACTION

In the event that the actions under Sections 23.4 through 23.7 are inadequate or there is insufficient time to carry out such procedures, Natural may periodically have to take unilateral action to maintain system pressure and preserve the overall operational integrity of Natural's system (or any portion thereof). Natural is authorized to use all the resources of its system to such ends, through the integrated operation of storage, line pack, and supply received into Natural's system, even though gas may be owned by a person other than the entity receiving delivery. Natural shall not, however, be responsible as a supplier of gas to any Shipper.

FERC Docket: RP08-319-000

Original Sheet No. 468 Original Sheet No. 468: Pending

GENERAL TERMS AND CONDITIONS

23.9 APPLICABILITY OF ACTIONS

- (a) In exercising its authority pursuant to Sections 23.4 through 23.8, Natural shall generally direct its actions to Shippers in the following sequence, to the extent such actions and/or sequencing will tend to alleviate the situation to be addressed:
- (1) First, to any Shipper which is causing disruption due to its failure to maintain receipts and deliveries in balance or to match physical flows with nominated receipts or deliveries (after taking into account any storage rights of such Shipper);
- (2) Second, to any Shipper which has failed or is failing to take action to anticipate a change in demand (i.e., a temperature sensitive LDC or end user failing to respond to changes in weather);
- $\mbox{(3)}$ Third, to any Shipper which is operating in a manner which conflicts with sound operational practices in relation to Natural's system; and
 - (4) Lastly, to all other Shippers.
- (b) Notwithstanding subsection (a), any Shipper which has taken voluntary action to alleviate a situation shall be exempted from further action hereunder to the extent of its voluntary action until Natural has applied equivalent measures to other Shippers.
- (c) Operational Flow Orders shall require action within any Shipper class proportionate to appropriate and relevant parameters, such as applicable contract quantities or current or recent flowing gas volumes.

23.10 USE OF PENALTY FUNDS

(a) Funds collected during each calendar year from Shippers billed for violations of Operational Flow Orders issued during Critical and non-Critical Times will be subject to the provisions of Section 12.8 of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 469 Original Sheet No. 469: Pending

GENERAL TERMS AND CONDITIONS

23.11 STANDARDS

(a) In issuing Operational Flow Orders or taking other operational control action under this Section, Natural shall apply consistent and objective engineering and operational criteria to define the overall operational integrity of the system and acceptable pressure levels to be protected, to evaluate the imminent nature of any threat to these factors, and to determine what steps are necessary to preserve such factors. Such criteria may be changed from time to time as operating experience indicates.

(b) In applying this Section, Natural shall operate its system on a non-discriminatory manner, without regard to the source of supply, the identity or nature of any Shipper or the identity of any entity tendering or receiving gas except as otherwise explicitly provided herein.

23.12 LIABILITY

Natural shall not be liable to any person for the manner in which it operates its system, or for any diversion of gas or capacity rights or any other adverse consequences to such person which may result from its actions, provided that Natural's actions were undertaken in furtherance of and in accordance with this Section and provided further that such adverse consequences are not attributable to Natural's negligence or misfeasance.

FERC Docket: RP08-319-000

Original Sheet No. 470 original Sheet No. 470: Pending

GENERAL TERMS AND CONDITIONS

24. METHOD OF MEASUREMENT AND MEASUREMENT EQUIPMENT

24.1 UNIT OF VOLUME/FUNDAMENTAL STANDARDS

All quantities referenced in this Tariff, unless otherwise expressly stated, are in terms of Dth. Where measurement is by orifice meter, all fundamental constants, observations, records, and procedures involved in the determination and/or verification of the quantity and other characteristics of gas delivered hereunder shall, unless otherwise specified herein, be in accordance with the standards prescribed in the 1985 edition of ANSI/API 2530 on "Orifice Metering of Natural Gas" or (at Natural's discretion) any revisions, amendments, or supplements to such standards. Where measurement is by other than an orifice meter, standards commonly accepted in the industry and adopted by Natural shall be used in the determination of all factors involved in the computation of gas volumes. Except as otherwise expressly stated herein, all fundamental constants, observations, records and procedures involved in determining and/or verifying the quantity and other characteristics of natural gas delivered hereunder shall (at Natural's discretion) be in accordance with such standards therefor as are or may hereafter from time to time be approved and recommended by the American Gas Association or such other standards as are commonly accepted in the industry and adopted by Natural.

24.2 BASIS

The measurement hereunder shall be corrected for deviation from Boyle's Law at the pressures and temperatures under which gas is measured hereunder by use of the method specified by AGA Report #3, Third Edition, October 1990, titled "Orifice Metering of Natural Gas and other Related Hydrocarbon Fluids," Part 1 through Part 4, or (at Natural's discretion) any revisions, amendments or supplements to such method.

24.3 DETERMINATION OF HEATING VALUE

(a) At Natural's sole option, the heating value of the gas may be determined by the use of a chromatograph, a continuous gas sampler, or by the taking of gas samples.

FERC Docket: RP08-319-000

Original Sheet No. 471 Original Sheet No. 471: Pending

GENERAL TERMS AND CONDITIONS

(b) If a continuous sampler is installed, then the heating value of the gas so taken shall be applied prospectively to the gas delivered beginning no later than the first of the following month. If gas samples are taken, then the heating value of the sample so taken shall be applied prospectively to the gas delivered beginning no later than the first of the following month and used until a new sample is taken. When a chromatograph is used to update heating values into an electronic flow computer, either locally or remotely, the average heating value shall be considered as the average heating value recorded by the electronic flow computer. heating value determinations made with a chromatograph shall use physical constants for gas compounds as outlined in the AGA Report #3, Third Edition, October, 1990 titled "Orifice Metering of Natural Gas" or (at Natural's discretion) any subsequent amendment or revision to such report. If tested against a certified standard sample, the heating value determining device shall be considered correct if it shall check in such test within ten (10) Btu per cubic foot.

24.4 DETERMINATION OF FLOWING TEMPERATURE

The temperature of the gas flowing through the meter or meters shall be determined by the continuous use of a recording thermometer installed so that it will properly record the temperature of the gas flowing through the meter or meters. The temperatures recorded during periods of flow each day shall be used in computing the volumes of gas for that day.

24.5 DETERMINATION OF SPECIFIC GRAVITY

- (a) At Natural's sole option, the specific gravity value of the gas may be determined by the use of a chromatograph, by the use of a continuous sampler, or by the taking of gas samples.
- (b) If a continuous sampler is installed, the specific gravity value of the gas so taken shall be applied prospectively to the gas delivered beginning no later than the first of the following month. If gas samples are taken, then the specific gravity value of the sample so taken shall be applied prospectively to the gas delivered beginning no later than the first of the following month and used until a new sample is taken. When a chromatograph is used to update specific gravity values into an electronic flow computer, either locally or remotely, the average specific gravity value shall be considered as the average specific gravity value recorded by the

FERC Docket: RP08-319-000

Original Sheet No. 472 Original Sheet No. 472: Pending

GENERAL TERMS AND CONDITIONS

electronic flow computer. All specific gravity value determinations made with a chromatograph shall use physical constants for gas compounds as outlined in the AGA Report #3, Third Edition, October, 1990 titled "Orifice Metering of Natural Gas" or (at Natural's discretion) any subsequent amendments or revisions to such report. If tested against a certified standard sample, the specific gravity determining device shall be considered correct if it shall check in such test within ten points in the third decimal place (0.010).

24.6 MEASUREMENT EQUIPMENT

(a) Unless otherwise agreed, Natural shall be the measuring party and Natural shall operate and maintain in satisfactory working order and condition at each Receipt Point and Delivery Point a measuring and regulating station with such meters, gauges, and other equipment as will make possible the determination of the daily quantities of natural gas delivered hereunder. The measuring stations shall be equipped with orifice meter runs, orifice meter gauges, recording gauges, or other types of meter or meters of standard make and design, commonly acceptable in the industry, so as to accomplish the accurate measurement of gas delivered hereunder. At Natural's sole option, a computer, transducers and other associated sensing devices may be installed to accomplish the accurate measurement of gas delivered hereunder in place of orifice meter gauges.

(b) If Natural is not the measuring party, Shipper agrees to comply, or to cause a third party which is the measuring party to comply, with all applicable provisions of this Section 24.

24.7 CALIBRATION AND TESTS OF METERS

Chromatographs, if used, shall be calibrated by the measuring party at least weekly against a certified standard gas sample. Other measuring equipment shall be calibrated quarterly unless experience dictates otherwise. The other party may, at its option, be present for such calibration and adjustment. For the purpose of measurement and meter calibration, the atmospheric pressure shall be derived from the Smithsonian Meteorological Tables based on the elevation of the meter at the place of measurement, irrespective of variations in natural atmospheric pressure from time to time. The measuring party shall give the other party notice of the time of all tests sufficiently in advance of conducting the same so that both parties may conveniently have their representatives

FERC Docket: RP08-319-000

Original Sheet No. 473 Original Sheet No. 473: Pending

GENERAL TERMS AND CONDITIONS

present. The other party at its sole expense, and subject to the work requirements of the measuring party, may request interim calibration or testing. Following any test, any measuring equipment found to be inaccurate to any degree shall be adjusted immediately to measure accurately.

24.8 ACCESS TO METERS, CHARTS AND RECORDS

The other party shall have access at all reasonable times to the measuring equipment and all other instruments used by the measuring party in determining the measurement and quality of the gas delivered under the Agreement, but the reading, calibrating, and adjusting thereof shall be done only by employees, agents or representatives of the measuring party. The charts and records shall be kept on file by the measuring party, for mutual use of the parties hereto, for a minimum period of the greater of: (1) the period during which billings are subject to correction under these General Terms and Conditions; or (2) such retention period as is required by applicable governmental regulation. The measuring party shall, upon request, submit to the other party records and charts from such equipment, subject to return by that party within thirty (30) days after receipt thereof. If Natural is not the measuring party, Shipper shall provide, or arrange for Natural to have, access to such data.

24.9 CORRECTION OF ERRORS

If, in the aggregate, measurement errors for a meter are found to be inaccurate by more than the greater of two percent (2%) or 500 Dth per day, registration thereof and any payments based upon such registration shall be corrected at the rate of such inaccuracy for any period of inaccuracy which is definitely known or agreed upon, but not to exceed the twelve (12) months prior to the current gas month, but in case the period of inaccuracy is not definitely known or agreed upon, then for a period extending back one-half (1/2) of the time elapsed since the last day of calibration, but not to exceed forty-five (45) days. Except where there is a deliberate omission or misrepresentation or mutual mistake of fact, measurement data corrections will be processed within six (6) months of the production month with a three (3) month rebuttal period (recognizing that the parties' other statutory or contractual rights shall not otherwise be diminished by this sentence).

FERC Docket: RP08-319-000

Original Sheet No. 474 Original Sheet No. 474: Pending

GENERAL TERMS AND CONDITIONS

24.10 FAILURE OF METERS

If for any reason the measuring equipment is out of service or out of repair so that the quantity of gas delivered through such measuring equipment cannot be ascertained or computed from the readings thereof, the quantity of gas so delivered during the period such equipment is out of service or out of repair shall be estimated and agreed upon by the parties hereto upon the basis of the best available data, using the first of the following methods which is feasible:

- (a) By using the registration of any duplicate measuring equipment installed by either party if installed and registering correctly;
- (b) By correcting the error if the percentage of error is ascertainable by calibration, test, or mathematical calculation;
- (c) By estimating the quantity of deliveries by deliveries during preceding periods under similar conditions when the measuring equipment was registering accurately.

24.11 CHECK MEASURING EQUIPMENT

- (a) Either party may install, maintain and operate at its own expense, at or near each Receipt Point or Delivery Point, such check measuring equipment as desired, provided that such equipment is installed so as not to interfere with the operation of any other measuring equipment.
- (b) Whenever any point is on premises of the delivering party, the receiving party shall have the right of free use and ingress and egress at all reasonable times for the purpose of installation, operation, repair, or removal of such check measuring equipment.
- (c) In the event check measuring equipment is installed by either party, the other party shall have access to the same at all reasonable times, but the reading, calibration and adjusting thereof and the changing of charts shall be done only by the party installing the checking equipment.

FERC Docket: RP08-319-000

Original Sheet No. 475 Original Sheet No. 475: Pending

GENERAL TERMS AND CONDITIONS

25. PRESSURE AND DELIVERY CONDITIONS

25.1 RECEIPT POINT PRESSURE

Shipper shall deliver gas to Natural at the pressure prevailing in Natural's system at the Receipt Point, as such pressure may vary from time to time.

25.2 DELIVERY POINT PRESSURE

Unless otherwise agreed in writing by Natural and Shipper, Natural shall deliver natural gas to Shipper at the Delivery Point at the pressure available in Natural's pipeline at such point. Such pressure shall not be less than three hundred (300) pounds per square inch gauge unless Natural and Shipper agree on a different minimum pressure.

25.3 OPERATIONAL CONDITIONS

Natural shall not be required to install additional compression or alter the ordinary operation of its compressors to receive gas into its pipeline under any Agreement or to transport and deliver such gas, or to lower its system operating pressure, alter the direction of gas flow, the gas load, or other operation or utilization of its facilities or otherwise change its normal pipeline operations in order to receive, transport or deliver gas under any Agreement. At each Receipt Point, Shipper shall provide, or cause to be provided, equipment acceptable to Natural which will prevent overpressuring Natural's pipeline.

FERC Docket: RP08-319-000

Original Sheet No. 476 Original Sheet No. 476: Pending

GENERAL TERMS AND CONDITIONS

26. OUALITY

26.1 SPECIFICATIONS

Unless otherwise stated in an Agreement, natural gas delivered by or on behalf of Shipper to Natural at any Receipt Point, and natural gas delivered by Natural to or for the account of Shipper at any Delivery Point, shall be of pipeline quality and shall conform to the following specifications:

(a) OXYGEN

The oxygen content shall not exceed ten parts per million (10 ppm) by volume of uncombined oxygen, and the parties shall make reasonable efforts to maintain the gas free from oxygen.

(b) HYDROGEN SULFIDE

. The hydrogen sulfide content shall not exceed one-quarter (1/4) grain per hundred (100) cubic feet of gas.

(c) MERCAPTANS

The mercaptan content shall not exceed one-quarter (1/4) grain per one hundred (100) cubic feet of gas.

(d) TOTAL SULFUR

The total sulfur content, including mercaptans and hydrogen sulfide, shall not exceed one-half (1/2) grain per one hundred (100) cubic feet of gas.

(e) CARBON DIOXIDE

. The carbon dioxide content shall not exceed two percent (2%) by volume.

(f) LIQUIDS

The gas shall be free of water and hydrocarbons in liquid form at the temperature and pressure at which the gas is delivered. In addition, the gas shall not contain any hydrocarbons which might condense to free liquids in the pipeline under normal pipeline conditions and shall in no event contain water vapor in excess of seven (7) pounds per one million (1,000,000) cubic feet.

FERC Docket: SYS-GEN-ORIG

First Revised Sheet No. 477 First Revised Sheet No. 477

Superseding: Original Sheet No. 477

GENERAL TERMS AND CONDITIONS

(a) DUST, GUM AND SOLID MATTER

 $$\operatorname{\textsc{The}}$$ gas shall be commercially free of dust, gum, gum-forming constituents and other solid matter.

(h) HYDROCARBON DEWPOINT

(1) Natural may, from time to time, as operationally necessary, establish and post on the Informational Postings portion of its Interactive Website an upper limit on the dewpoint for receipts on specified segments or other specified locations on its system to prevent hydrocarbon fallout, consistent with Section 26.1(f) hereof. Natural will include in such posting the anticipated duration of the limitation. Points with an actual flow of 100 Dth/d or less will be exempt from the limitation in any such posting. Natural will provide as much prior notice of any such limitation as reasonably practicable and will attempt to provide this prior notice in the posting at least ten (10) Days before the beginning of the month in which the limitation is to be effective. If prior notice by posting at least ten (10) Days before the beginning of the month is not practicable, Natural will explain the reason in the posting why it was unable to give such prior notice.

(2) Notwithstanding the foregoing, Natural may not decline to accept gas based on its dewpoint if the dewpoint of that gas is equal to or less than fifteen degrees Fahrenheit (15 F.) In addition, Natural shall continuously post "safe harbor" dewpoint values based on operational and engineering considerations and shall make any changes in these "safe harbor" values effective no sooner than 30 days after the revised "safe harbor" values have been posted. Natural may not decline to accept gas which conforms to such posted "safe harbor" values if the gas meets the other quality standards set out in this Section 26. No such posting may set out a dewpoint "safe harbor" of less than fifteen degrees Fahrenheit (15 F).

(3) Where any posting of a limitation under Section 26.1(h)(1) includes a dewpoint limitation, Natural will calculate the dewpoint at any point of receipt into Natural's system affected by the posting on request of the operator of that point or any producer, purchaser, supply aggregator or Shipper with gas being tendered at that point.

(4) Natural shall post on its Interactive Website, within twenty-four (24) hours after making the calculations, the following: every receipt point dewpoint value Natural calculates, along with the method by which the dewpoint was calculated; and every blended dewpoint Natural calculates for a line segment of its system.

FERC Docket: RP08-319-000

Original Sheet No. 477 Original Sheet No. 477: Pending

GENERAL TERMS AND CONDITIONS

(g) DUST, GUM AND SOLID MATTER

The gas shall be commercially free of dust, $\operatorname{\mathsf{gum}}$, $\operatorname{\mathsf{gum}}$ -forming constituents and other solid matter.

(h) HEATING VALUE

The gas delivered shall contain a daily, monthly and yearly average heating content of not less than nine hundred fifty (950) Btus per cubic foot. In addition, Natural may, from time to time, as operationally necessary, establish and post on the Informational Postings portion of its Interactive Website an upper Btu/cf limit and/or a limit on the dewpoint for receipts on specified segments or other specified locations on its system to prevent hydrocarbon fallout, consistent with Section 26.1(f) hereof, or to assure that gas will be accepted for delivery into interconnects with interstate pipelines, intrastate pipelines, end-users or directly connected local distribution companies. Natural will include in such posting the anticipated duration of the limitation. Points with an actual flow of 100 Dth/d or less will be exempt from the limitation in any such posting. Natural will provide as much prior notice as reasonably practicable and will attempt to provide in the posting at least two (2) Business Days' prior notice before the limitation becomes effective. If prior notice of at least two (2) Business Days is not practicable, Natural will explain the reason in the posting why it was unable to give such prior notice. Where the posting includes a dewpoint limitation, Natural will, upon request, provide current information regarding the dewpoint at any point of receipt into Natural's system affected by the posting to the operator of that point or any producer, purchaser, supply aggregator or Shipper with gas being tendered at that point.

(i) TEMPERATURE

The gas shall not be delivered at a temperature of less than forty degrees Fahrenheit (40 F.) or more than one hundred twenty degrees Fahrenheit (120 F.).

(j) NITROGEN

The nitrogen content shall not exceed three percent

(3%) by volume.

(k) HYDROGEN

The gas shall contain no carbon monoxide, halogens, or unsaturated hydrocarbons, and no more than four hundred parts per million (400 ppm) of hydrogen.

FERC Docket: SYS-GEN-ORIG

Original Sheet No. 477A Original Sheet No. 477A

GENERAL TERMS AND CONDITIONS

(i) BTU VALUE

(1) Gas delivered to Natural or by Natural shall contain a daily, monthly and yearly average Btu Value of not less than nine hundred fifty (950) Btus per cubic foot. Natural may, from time to time, establish and post on the Informational Postings portion of its Interactive Website an upper Btu value limit for receipts on specified segments or other specified locations on its system to assure that gas will be accepted for delivery into interconnects with interstate and intrastate pipelines, plants, end-users or directly connected local distribution companies, consistent with the delivery requirement set out in subsection (3) below. Natural will include in any such posting the anticipated duration of the limitation. Points with an actual flow of 100 Dth/d or less will be exempt from the limitation in any such posting. Natural will provide as much prior notice of any such limitation as reasonably practicable and will attempt to provide this prior notice in the posting at least ten (10) Days before the beginning of the month in which the limitation is to be effective. If prior notice by posting at least ten (10) Days before the beginning of the month is not practicable, Natural will explain the reason in the posting why it was unable to give such prior notice.

- (2) Notwithstanding the foregoing, except under an Operational Flow Order, Natural may not decline to accept gas based on its Btu Value if the Btu Value of the gas is equal to or less than: (i) 1,110 Btus per cubic foot, for gas receipts in the Permian, Midcontinent, South Texas, TexOk, and Louisiana Receipt Zones; and (ii) 1,065 Btus per cubic foot for gas receipts in the Amarillo Mainline, Gulf Coast Mainline and the Iowa-Illinois Market Receipt Zones.
- (3) Except as otherwise agreed with the downstream entity receiving the gas, Natural will deliver gas with a Btu Value of not greater than 1,065 Btus per cubic foot in the Iowa-Illinois Receipt Zone, and not greater than 1,110 Btus per cubic foot in all other Receipt Zones.
- (4) If Natural posts a Btu Value limitation under subsection (1) hereof, Natural will calculate the Btu Value at any point of receipt into Natural's system affected by the posting on request of the operator of that point or any producer, purchaser, supply aggregator or Shipper with gas being tendered at that point.

FERC Docket: RP08-319-000

Original Sheet No. 478 Original Sheet No. 478: Pending

GENERAL TERMS AND CONDITIONS

26.2 TOXIC HAZARDOUS SUBSTANCES

Shipper agrees to supply or cause its designee to supply to Natural upon demand, at any time and from time to time, a sample of liquids removed from the gas stream at any Receipt Point, whether removed by a coalescer or otherwise, for analysis at a laboratory of Natural's choosing. If at any time PCBs or any other toxic substances or chemicals that Natural deems hazardous and/or in any way unsafe for transportation are found in the liquid samples supplied to Natural by Shipper, Natural may in its sole discretion immediately cease the receipt of such gas and any associated liquids through its facilities. Upon proof that such toxic or hazardous substances are no longer present at levels deemed unsafe by Natural, Natural shall restore service to Shipper at the affected Receipt Point.

26.3 NON-CONFORMING GAS

- (a) If at any time, gas tendered under an Agreement shall fail to conform to any of the quality specifications set forth above the receiving party may, at its option exercised on a non-discriminatory basis, refuse to accept delivery pending correction of the deficiency by the delivering party.
- (b) If gas tendered under an Agreement fails to conform to any of the quality specifications set forth above, the party tendering the gas shall be responsible for any damages to the facilities of Natural caused by such tender of non-conforming gas; provided, however, that if Natural has consented to receive the non-conforming gas, the tendering party shall not be liable to the extent expressly stated in the consent.

26.4 POSTING OF GAS QUALITY INFORMATION

Natural posts information on gas quality on its system as described in Section 17.1(a) of these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 479 Original Sheet No. 479: Pending

GENERAL TERMS AND CONDITIONS

27. FORCE MAJEURE

27.1 EFFECT OF FORCE MAJEURE

In the event of either Natural or Shipper being rendered unable by Force Majeure (on its part or that of a necessary third party) to carry out, wholly or in part, its obligations under the provisions of an Agreement, it is agreed that the obligations of the party affected by such force majeure, other than the obligation to make payments thereunder, shall be suspended during the continuance of any inability so caused but for no longer period, and such cause shall, so far as possible, be remedied with all reasonable dispatch.

27.2 DEFINITION OF FORCE MAJEURE

- (a) The term "Force Majeure" as employed herein shall mean acts and events not within the control of the party claiming suspension and shall include acts of God, strikes, lockouts or other industrial disturbances, inability to obtain pipe or other material or equipment or labor, wars, riots, insurrections, epidemics, landslides, lightning, earthquakes, fires, storms, floods, washouts, arrests and restraint of rulers and people, interruptions by government or court orders, present or future orders of any regulatory body having proper jurisdiction, civil disturbances, explosions, breakage or accident to machinery or lines of pipe, freezing of wells or pipelines, and any other cause whether of the kind herein enumerated or otherwise, not within the control of the party claiming suspension and which, by the exercise of due diligence, such party is unable to overcome.
- (b) Nothing contained herein, however, shall be construed to require either party to settle a strike against its will. Such causes or contingencies affecting the performance by either party, however, shall not relieve it of liability in the event of its concurring negligence or in the event of its failure to use reasonable diligence to remedy the situation and remove the cause in an adequate manner and with all reasonable dispatch, nor shall such causes or contingencies relieve either party of liability otherwise unless such party shall give notice and full particulars of the same in writing or by electronic means to the other party as soon as possible after the occurrence relied on.

FERC Docket: RP08-319-000

Original Sheet No. 480 Original Sheet No. 480: Pending

GENERAL TERMS AND CONDITIONS

28. POSSESSION OF GAS, TITLE AND RESPONSIBILITY

Shipper warrants that it will at the time of delivery to Natural have good title to all gas so delivered free and clear of all liens, encumbrances and claims whatsoever. As between Shipper and Natural, Shipper shall be deemed to be in control and possession of gas and responsible therefor and shall hold Natural harmless of and from any damage or injury caused thereby until the gas shall have been delivered to Natural at the Receipt Point(s), after which Natural shall be deemed to be in control and possession of such gas until its delivery to Shipper or for Shipper's account at the Delivery Point(s) and while in such possession Natural shall be responsible therefor and hold Shipper harmless of and from any damage or injury caused thereby. Natural shall have no responsibility with respect to any gas on account of anything which may be done, happen or arise with respect to said gas until it is received by Natural. Shipper shall have no responsibility with respect to said gas after its receipt by Natural on account of anything which may be done, happen or arise with respect to said gas after such receipt until its delivery to Shipper, or for Shipper's account, at the Delivery Point(s). The point of the division of responsibility shall be the point of interconnection between the facilities of Natural and Shipper, or their respective agents, at the Receipt or Delivery Point(s), as applicable. The foregoing provisions of this Section shall not relieve either party from responsibility for acts of gross negligence or willful misconduct of such party, its agents or emplovees.

FERC Docket: RP08-319-000

Original Sheet No. 481 Original Sheet No. 481: Pending

GENERAL TERMS AND CONDITIONS

29. NOTIFICATION

29.1 GENERAL

Except as provided otherwise in this Tariff or the Agreement, operational communications may be made by telephone or other mutually agreeable means without subsequent written confirmation, unless written confirmation is requested by either party hereto. Any notice, request, demand, statement or other formal communication shall only be deemed given when delivered by first class, certified or registered U.S. mail, overnight delivery, courier, telefax or Electronic Notice Delivery consistent with GISB Standards as adopted in Section 48 of these General Terms and Conditions. Such delivery shall: (1) be sent to Natural at the address specified in the Agreement, or through such electronic means as are available and authorized by Natural, or at an address otherwise stated in a notice by Natural to Shipper; and (2) be sent to Shipper at the address otherwise stated in a notice by Shipper to Natural.

29.2 NOTIFICATION PROCEDURES

(a) PRICING

- (1) The availability and pricing of services on Natural's system is governed by this Tariff. From time to time, Natural changes or updates its Tariff by filings with the FERC. Each Shipper is notified by Natural of such filings and is provided a copy of each filing.
- (2) Telephone inquiries related to the availability or pricing of services are answered by representatives of Natural and upon request, potential Shippers are provided copies of Natural's Tariff filings.
- (3) Shippers desiring a rate under any Agreement other than the maximum rate on file with the FERC are required to submit such requests in writing or by electronic medium to the Director, Transportation/Storage Services, in Houston. Any lower rate agreed to by Natural is evidenced in writing to such Shipper, and such rate is considered confidential until it is reported to the FERC as required by the Regulations. In order to attract or determine interest in the use of any particular service, representatives of Natural may from time to time contact Shippers by telephone, to the extent consistent with Section 5.1(e) of these General Terms and Conditions. Such conversations are confidential and may or may not result in Shipper submitting a request for a discounted rate for a particular service.

FERC Docket: RP08-319-000

Original Sheet No. 482 Original Sheet No. 482: Pending

GENERAL TERMS AND CONDITIONS

(b) CAPACITY

- (1) Capacity available for firm service is communicated to requestors of that service under the provisions of Section 5.1(e) of these General Terms and Conditions and of the applicable firm Rate Schedule.
- (2) Capacity available for interruptible services is communicated to holders of interruptible Agreements by representatives of Natural in response to the Shippers' nominations for service. The general availability of interruptible capacity is communicated by the Informational Postings portion of Natural's Interactive Website system, which is described in Section 17 of these General Terms and Conditions.
- (3) When available capacity is affected by construction projects or unforeseen conditions, Natural communicates such information primarily via its Interactive Website (and primarily by posting on the Informational Postings portion thereof) to its Shippers. Natural also uses letters or telephone calls to communicate capacity information when such means are appropriate.

FERC Docket: RP08-319-000

Original Sheet No. 483 Original Sheet No. 483: Pending

GENERAL TERMS AND CONDITIONS

30. FACILITIES/OBLIGATION TO CARRY OUT AGREEMENT/FILINGS

30.1 FACILITIES

The nature of, and responsibility for, any facilities which must be acquired, modified or constructed to effectuate an Agreement are to be set out in a separate agreement between Natural and Shipper or the operator of a point. To the extent that Shipper builds facilities to interconnect with Natural's system, such facilities shall be in conformance with Department of Transportation regulations, and any other applicable governmental regulations, and shall be subject to inspection and prior approval by Natural.

30.2 OBLIGATIONS TO CARRY OUT AGREEMENT

Other provisions of an Agreement notwithstanding, Natural shall be under no obligation to commence service thereunder unless and until: (1) all facilities, of whatever nature, as are required to permit (as applicable) the receipt, measurement, transportation, storage, and delivery of natural gas under the Agreement have been installed and are in operating condition; (2) any payments due Natural thereunder have been received; and (3) Natural has, in its reasonable discretion, determined that such service is authorized under all applicable Regulations.

30.3 REGULATORY FILINGS

After the execution of an Agreement, each party shall make and diligently prosecute any and all necessary filings with Federal or other governmental bodies, or both, as may be required for the initiation and continuation of the service which is the subject of an Agreement. Upon either party's request, the other party shall timely provide or cause to be provided to the requesting party such information and material not within the requesting party's control and/or possession that may be required for such filings. Each party shall promptly inform the other party of any changes in the representations made by such party hereunder and/or in the information provided pursuant to this Section. Each party shall promptly provide the other party with a copy of all filings, notices, approvals, and authorizations in the course of the prosecution of its filings.

FERC Docket: RP08-319-000

Original Sheet No. 484 Original Sheet No. 484: Pending

GENERAL TERMS AND CONDITIONS

31. INDEMNIFICATION

31.1 GENERAL

Shipper will indemnify and hold Natural harmless from and against any and all suits, actions, causes of action, claims and demands arising from or out of any adverse claims by third parties claiming ownership of or an interest in the gas tendered under an Agreement. Natural will indemnify and save Shipper harmless from and against any and all suits, actions, causes of action, claims and demands arising from or out of any adverse claims by third parties claiming ownership of or an interest in the gas delivered to Shipper, or for Shipper's account, under an Agreement.

31.2 ELIGIBILITY FOR SERVICE

Shipper warrants that its requested service meets all requirements for service under the applicable Rate Schedule and these General Terms and Conditions and conforms to applicable Regulations of the FERC. Shipper further agrees to abide by the terms of the applicable Rate Schedule and these General Terms and Conditions. Shipper will indemnify Natural and hold Natural harmless from all suits, actions, damages, costs, losses, expenses (including reasonable attorney fees) and regulatory proceedings arising from its breach of this warranty. Shipper further agrees to indemnify Natural and save Natural harmless from any claims asserted by any person because of any curtailment or interruption of service which is consistent with the applicable Rate Schedule and these General Terms and Conditions. Shipper, however, shall have no obligation to indemnify Natural for the results of any intentional or unintentional acts by Natural that contravene the applicable Rate Schedule or these General Terms and Conditions.

FERC Docket: RP08-319-000

Original Sheet No. 485 Original Sheet No. 485: Pending

GENERAL TERMS AND CONDITIONS

32. SUCCESSORS AND ASSIGNS

Any company which shall succeed by purchase, merger or consolidation to the properties, substantially as an entirety, of Shipper or Natural shall be entitled to the rights and shall be subject to the obligations of its predecessor in title under the Agreement; provided, however, that Natural reserves the right to evaluate and approve the creditworthiness of the new entity in accordance with the Credit Evaluation section of these General Terms and Conditions. No other assignment of an Agreement or any of the rights or obligations thereunder shall be made by Shipper unless there first shall have been obtained the written consent thereto of Natural. Shipper or Natural may pledge or assign their respective right, title and interest in and to and under the Agreement to a trustee or trustees, individual or corporate, as security for bonds or other obligations or securities without the necessity of such trustee or trustees becoming in any respect obligated to perform the obligations of the assignor under the Agreement and, if any such trustee be a corporation, without its being required to qualify to do business in any State in which performance of the Agreement may occur.

FERC Docket: RP08-319-000

Original Sheet No. 486 Original Sheet No. 486: Pending

GENERAL TERMS AND CONDITIONS

33. REGULATION

The operation of the provisions of this Tariff shall be subject to any and all governmental statutes and all lawful orders, rules, and regulations affecting the receipt, storage, transportation or delivery of gas hereunder or the equipment required in connection with such receipt, storage, transportation or delivery. It is understood that performance under any Agreement shall be subject to all valid laws, orders, rules and regulations of duly constituted governmental authorities having jurisdiction or control of the matter related hereto. Should either of the parties, by force of any such law, order, rule or regulation, at any time during the term of the Agreement be ordered or required to do any act inconsistent with the provisions thereof, then for that period only during which the requirements of such law, order, rule or regulation are applicable, the Agreement shall be deemed modified to conform with the requirement of such law, order, rule or regulation; provided, however, nothing herein shall alter, modify or otherwise affect the respective rights of the parties to cancel or terminate the Agreement under its terms and conditions.

FERC Docket: RP08-319-000

Original Sheet No. 487 Original Sheet No. 487: Pending

GENERAL TERMS AND CONDITIONS

34. SALES BY NATURAL'S MERCHANT DIVISION

34.1 GENERAL

Effective December 1, 1993, sales service by Natural under Rate Schedules DMQ-1, G-1, WS-1, WS-2, E-1, AOR-1 and IS-1 shall terminate. Thereafter, any sale made by Natural shall be pursuant to this Section and to a blanket certificate issued by the Commission. Sales by Natural shall be subject to any conditions imposed by the Commission in authorizing, and to any Commission Rules and Regulations applicable to, such blanket certificate. This Section shall not apply to any gas sales made by any affiliate of Natural.

34.2 SEPARATION OF FUNCTIONS

(a) In the event that Natural makes any sale after December 1, 1993 other than a sale by its Transportation Division identified in subsection (c) of this Section 34.2, Natural shall establish a Merchant Division to make such sale. Natural does not anticipate that any Merchant Division will be formed, but if it is, the Merchant Division shall be separate from Natural's Transportation Division, which markets and implements transportation and storage services. The Merchant Division, if any, shall be treated as a marketing affiliate for the purposes of the requirements of Order Nos. 497, et seq., as such requirements may be in effect from time to time. Natural's Transportation Division shall not afford Natural's Merchant Division, if any, any preference or favorable treatment vis a vis any other Shipper similarly situated in any dealings between the two divisions. Natural's Transportation Division shall provide services of equal quality to all Shippers, consistent with the applicable Rate Schedule, without regard to whether gas was purchased from Natural's Merchant Division or from another entity. Natural's Transportation Division shall disclose to non-affiliated suppliers and Shippers non-public operating data available to personnel in Natural's Merchant Division (if any). Unless otherwise indicated or required by the context, references in this Tariff to "Natural" shall encompass only Natural's Transportation Division.

FERC Docket: RP08-319-000

Original Sheet No. 488 Original Sheet No. 488: Pending

GENERAL TERMS AND CONDITIONS

(b) Natural's Merchant Division, if any, may not hold capacity on Natural's system or on any upstream pipeline which interconnects with Natural; provided, however, that Natural's Merchant Division, if any, is not precluded from holding capacity on such an upstream system for the movement of gas which does not enter Natural's system.

(c) Natural's Transportation Division is not providing a supply service under any Rate Schedule of this Tariff but may buy and sell gas in connection with the providing of storage and transportation services. Without limitation of the foregoing, Natural may buy and sell gas to the extent necessary to maintain system pressure, to manage system storage, to replenish any storage retained by Natural's Transportation Division for system operations, to provide adequate storage inventory to support storage services except to the extent the Shipper is responsible for such inventory, to maintain line pack and provide additional line pack for new facilities, to implement the cashout procedures $% \left(1\right) =\left(1\right) \left(1\right) +\left(1\right) \left(1\right) \left(1\right) +\left(1\right) \left(1\right)$ under Section 13 of these General Terms and Conditions, and to perform other functions of Natural's Transportation Division in connection with transportation and storage services. Such sales are authorized pursuant to Natural's blanket sales certificate. Nothing herein shall impose on Natural's Transportation Division any obligation to provide a supply function to any of its transportation or storage Shippers; provided, however, that Natural may rely on system storage and line pack to provide instantaneous receipt and delivery and tolerances as applicable under the particular Rate Schedule. Natural's Transportation Division may retain capacity and make sales of gas to the limited extent necessary to carry out Section 38.8 of these General Terms and Conditions relating to Coal Gas.

34.3 MERCHANT DIVISION SALES

(a) Except as provided in subsection (b) of this Section 34.3, all terms and conditions of any sales service provided by Natural's Merchant Division, if any, including the level, form and billing of rates and any quantity entitlements, delivery obligations and take obligations, shall be negotiated between Natural and the gas purchaser, and shall be and remain confidential as between Natural's Merchant Division and its purchasers.

FERC Docket: RP08-319-000

Original Sheet No. 489 Original Sheet No. 489: Pending

GENERAL TERMS AND CONDITIONS

(b) Curtailment of gas sales by Natural's Merchant Division, if any, shall be consistent with Section 34.5 and the point of any sale by Natural's Merchant Division shall be consistent with Section $34.4\,(a)$.

34.4 POINT OF SALE

- (a) Gas sales by Natural's Merchant Division (if any) may be made at any point not on Natural's system (i.e., at points on other pipeline systems), or at any point at which gas first enters any facilities owned by Natural. Points where gas first enters Natural's facilities will be listed in the Catalog of Points on the Informational Postings portion of Natural's Interactive Website. Subject to the foregoing, the specific sales point(s) for any particular sale by Natural's Merchant Division and the allocation of purchase volumes among multiple points shall be negotiated between Natural's Merchant Division and its purchasers.
- (b) Gas sales by Natural's Transportation Division for operational purposes only may be made at any point at which gas first enters facilities owned by Natural, at any pooling point and at any zone storage point. Such sales may include in-field storage transfers in accordance with Section 9 of these General Terms and Conditions. Unless the point of sale is otherwise restricted under another provision of this Section, the point of sale for any other sale by Natural must be at a point not on Natural's system (i.e., at points on other pipeline systems) or at any point where gas first enters any facilities owned by Natural.

34.5 CURTAILMENT

(a) Sales by Natural's Merchant Division, if any, may be curtailed for supply-related shortages or due to conditions of Force Majeure or routine maintenance on facilities upstream of the sales points or for other reasons as set out in the sales agreement between Natural's Merchant Division and the purchaser. As used in this Section 34, Force Majeure shall have the same meaning as is set out in Section 27 of these General Terms and Conditions. Without limitation of the foregoing, Force Majeure shall include Force Majeure on the part of any necessary entity upstream of the point of sale.

FERC Docket: RP08-319-000

Original Sheet No. 490 Original Sheet No. 490: Pending

GENERAL TERMS AND CONDITIONS

(b) Capacity curtailment associated with transportation or storage of gas sold by Natural's Merchant Division, if any, shall be determined under the applicable Rate Schedule. Natural's Transportation Division shall not afford any preference or favorable treatment to any arrangement for the movement or storage of gas purchased from Natural's Merchant Division, if any, regardless of the identity of the Shipper.

34.6 AGENCY

- (a) Natural's Merchant Division (if any exists) may, but is not obligated to, act as agent for any Shipper with respect to any transportation or storage Agreement between such Shipper and Natural's Transportation Division. Natural's Merchant Division (if any) must comply with the provisions of Natural's Tariff governing delegation and agency in the same manner as any other entity similarly situated. Natural's Transportation Division may not afford the Merchant Division (if any), acting in such agency capacity, any preference or favorable treatment vis a vis any other entity which is eligible to act, and has been requested to act, as agent under Natural's Tariff.
- (b) The terms and conditions of any such agency relationship, including any fee assessed for such service, shall be negotiated between Natural's Merchant Division (if any) and the entity designating it as agent.

34.7 ABANDONMENT

Natural's Merchant Division, if any, is authorized to abandon any sale upon termination of the sales agreement. Natural's Transportation Division is authorized to abandon any sale identified in Section 34.2(c) upon termination of the sales agreement.

FERC Docket: RP08-385-000

First Revised Sheet No. 491 First Revised Sheet No. 491

Superseding: Original Sheet No. 491

FERC GAS TARIFF

SEVENTH REVISED VOLUME NO. 1

(Supersedes Sixth Revised Volume No. 1)

of

NATURAL GAS PIPELINE COMPANY OF AMERICA LLC

Filed with the

FEDERAL ENERGY REGULATORY COMMISSION

Communications Concerning this Tariff Should be Addressed to:

Bruce H. Newsome, Vice President
Natural Gas Pipeline Company of America LLC
3250 Lacey Road, 7th Floor
Downers Grove, Illinois 60515-7918
Telephone: (630) 725-3070
Facsimile: (630) 725-3108
GENERAL TERMS AND CONDITIONS

35. SHIPPER COMPLAINT PROCEDURE

35.1 COMPLAINT PROCEDURE

It is the goal of Natural, as a provider of transportation and storage services for others, that each of its existing and potential Shippers receive service that is in accordance with this Tariff and is fully satisfactory to the customer. To that end, it is the policy of Natural that customer concerns and problems, communicated in any form to any representative of Natural, be satisfactorily resolved as informally, as rapidly and at as low a level as is possible. If attempts to resolve problems and concerns through such normal communication channels are unsuccessful, the procedures set forth in subsections (a) through (e) should be followed.

- (a) Formal complaints by Shippers and potential Shippers shall be addressed to the Vice President, Business Management, located in Downers Grove, Illinois. A complaint should contain as much specific information as is possible in order to facilitate the appropriate resolution of the matter. Anyone making a verbal complaint should specifically identify the communication as a complaint.
- (b) The Vice President, or his designee, shall acknowledge the receipt of the complaint within forty-eight (48) hours of receipt. If appropriate, Natural's resolution of the matter will be communicated tentatively to the complainant at that time.
- (c) The Vice President, or his designee, shall communicate, as necessary, with others concerning the complaint and the formation of an appropriate response to it.

- (d) The timing and nature of subsequent communications with the complainant, including final resolution of the matter, shall be at the discretion of the Vice President. Every effort shall be made to resolve finally each complaint in writing within thirty (30) days after the complaint was originally received. At a minimum, Natural shall notify Shipper in writing of the status of the complaint within thirty (30) days of its receipt.
- (e) The foregoing recognizes that individual complaints may vary greatly as to complexity and seriousness. For this reason, the informed judgment of the Vice President shall be relied upon in each instance for the necessary determinations concerning such things as: (1) the exact steps to be taken in addressing the complaint; (2) the need to involve more senior officers in the matter; and (3) the appropriate final resolution of the complaint.

FERC Docket: RP08-319-000

Sheet No. 492 Sheet No. 492: Pending

GENERAL TERMS AND CONDITIONS

36. RESERVED FOR FUTURE USE

FERC Docket: RP08-319-000

Original Sheet No. 493 Original Sheet No. 493: Pending

GENERAL TERMS AND CONDITIONS

37. LATERAL LINE CHARGE

Any Buyer under Rate Schedule DMQ-1 or G-1 which was paying a charge under Rate Schedule F-1 as of November 30, 1993 in connection with construction by Natural of sales laterals shall continue to pay such a charge in accordance with the provisions of Rate Schedule F-1 as they existed on November 30, 1993. Such charges shall be revised and shall terminate in the same manner as if Rate Schedule F-1 had remained in effect. The applicable charges in effect from time to time shall be set out on the Currently Effective Rates section of Natural's Tariff.

FERC Docket: RP08-319-000

Original Sheet No. 494 Original Sheet No. 494: Pending

GENERAL TERMS AND CONDITIONS

38. RECOVERY OF ORDER NO. 636 GAS SUPPLY REALIGNMENT COSTS AND UPSTREAM SUPPLIER TRANSITION COSTS

38.1 APPLICABILITY

This Section sets forth procedures pursuant to Commission Order Nos. 636, et seq., for Natural to assess charges to compensate it for Supply Realignment Costs and transition costs assessed by upstream pipelines. Assessments under this Section 38, ceased as of December 1, 2001 except for any remaining amounts under Section 38.5.

38.2 DEFINITIONS

(a) SUPPLY REALIGNMENT COSTS

Supply Realignment Costs shall mean Contract Reformation Costs, Auction Costs or Pricing Differential Costs. The method of establishing Supply Realignment Costs shall be determined by Natural as to each contract in its reasonable discretion exercised to minimize costs. Natural may not include as Supply Realignment Costs any punitive or exemplary damages, any costs or category of costs already reflected in Natural's base rates or any assessment of the costs of other parties.

(1) Contract Reformation Costs shall mean amounts in cash or other consideration prudently incurred by Natural to terminate, buydown or modify contracts or settle claims by suppliers related to elimination or modification of supply commitments in effect on November 30, 1993, to the extent that such amounts are associated with a period subsequent to November 30, 1993. Such amounts shall include payments to suppliers to buyout or buydown contract volumes or prices under gas supply contracts; to terminate such contracts; to reform volume, take-or-pay, minimum take, price or any other pertinent terms of any gas purchase contract; to satisfy damage awards to suppliers arising out of litigation related to gas supply contracts to the extent such awards relate to periods subsequent to November 30, 1993; or to cover the costs incurred by such suppliers to third parties related to such contracts to the extent recoverable from Natural under the terms of such contracts.

 $(2) \quad \text{Auction Costs apply where a contract is} \\ \text{assigned pursuant to an auction as provided by Section 4 of these} \\ \text{General Terms and Conditions, in which case the best bid shall} \\ \text{establish the level of Supply Realignment Costs.}$

FERC Docket: RP08-319-000

Original Sheet No. 495 Original Sheet No. 495: Pending

GENERAL TERMS AND CONDITIONS

(3) Pricing Differential Costs shall apply in instances where Natural is unable to reach a negotiated resolution with a supplier at reasonable cost and Natural concludes that auction of the contract may not reduce costs. In such cases, Supply Realignment Costs shall be determined pursuant to a "pricing differential" method and shall reflect the difference (which may be positive or negative) between the price at which Natural sells the gas at auction or, if an auction is not utilized, the greater of the price at which Natural actually sells the gas or the market price of gas for the period in question, as determined by reference to published spot price indices in general use in the industry and applicable to Natural's system, and the price Natural is obligated to pay the supplier, times the volume of gas Natural takes from such supplier over the period in question. Such costs shall also reflect any out-ofperiod adjustments relating to contracts treated on a pricing differential basis if such adjustments relate to periods subsequent to November 30, 1993. Natural has provided the Commission as part of its first tracking filing under this Section 38: (i) a List A of all contracts to be handled initially under the pricing differential method; (ii) an initial List B of all below market contracts; and (iii) a specification of the applicable indices and other factors used in determining the price differential. Such sales are to be made consistent with Section 34 of these General Terms and Conditions. Natural may continue to use the pricing differential method to determine Supply Realignment Costs for auctions held or sales made during the period through November 30, 1997; provided, however, that Natural may continue to use the pricing differential method to determine Supply Realignment Costs associated with one remaining gas supply contract, as reflected in its filing with the Commission to revise this Section 38 made on August 26, 1997, until December 31, 2000, or any earlier date on which such contract is terminated. Natural will advise the Commission within thirty (30) days in the event that the parties to this contract agree to any substantive change in its terms and conditions. The pricing differential method shall also apply to Coal Gas as provided in Section 38.8 of these General Terms and Conditions.

(b) ASSESSABLE SHIPPER

For purposes of this Section, Assessable Shipper shall mean any Shipper receiving Assessable Transportation Services.

(c) ASSESSABLE TRANSPORTATION SERVICES

Assessable Transportation Services shall mean all firm transportation services under Rate Schedule FTS or FFTS (and Rate

FERC Docket: RP08-319-000

Original Sheet No. 496 Original Sheet No. 496: Pending

GENERAL TERMS AND CONDITIONS

Schedules S-2, S-2/G, FTS-E or FTS-E/G prior to such Rate Schedules being superseded effective December 1, 1995, and DSS and FTS-G for periods prior to December 1, 1997) pursuant to Agreements in effect from and after December 1, 1993, excluding (1) any such Agreements entered into prior to December 1, 1993 to the extent Natural is unable contractually to collect any additional amounts attributable to Supply Realignment Costs; (2) any Agreements excludible pursuant to a filing under Section 38.10(c); and (3) any Agreement for such service between Natural and any member of the Natural Customer Group, any of the G Customers, or any of the Other Consenting Customers, except as otherwise provided under other terms of this Section 38. No costs shall be allocated or charges assessed hereunder to any FRSS Agreement.

(d) AMORTIZATION PERIOD

Amortization Period shall mean the period commencing December 1, 1993 and continuing through November 30, 1997.

(e) EXTENDED AMORTIZATION PERIOD

Extended Amortization Period shall mean the period from December 1, 1997 through November 30, 2001; provided, however, that the Extended Amortization Period shall end on such earlier date as collections under Section 38.11 cease.

(f) NATURAL CUSTOMER GROUP

Natural Customer Group shall include the following local distribution utilities served by Natural: Northern Illinois Gas Company; The Peoples Gas Light and Coke Company; Northern Indiana Public Service Company; MidAmerican Energy Company; North Shore Gas Company; Wisconsin Natural Gas Company (limited to contracts of its predecessor, Wisconsin Southern Gas Company); Interstate Power Company; and IES Utilities, Inc. (limited to contracts of its predecessors, Iowa Electric Light and Power Company and Iowa Southern Utilities Company).

(g) NCG SETTLEMENT

NCG Settlement shall mean the settlement filed jointly by Natural and the Natural Customer Group on February 1, 1994 in Docket No. RS92-45, relating to recovery by Natural of Supply Realignment Costs from the Natural Customer Group, which settlement was approved by the Commission on May 12, 1994, 67 FERC Paragraph 61,174.

FERC Docket: RP08-319-000

Original Sheet No. 497 Original Sheet No. 497: Pending

GENERAL TERMS AND CONDITIONS

(h) OTHER CONSENTING CUSTOMERS

Other Consenting Customers shall mean Associated Natural Gas Company and City of Nebraska City, Nebraska, which are former DMQ-1 sales customers which are not part of the Natural Customer Group, and Industrial Energy Applications, Inc.

(i) OTHER SETTLEMENTS

Other Settlements shall mean the two settlements filed on April 8, 1994 in Docket No. RS92-45 relating to recovery by Natural of Supply Realignment Costs from Associated Natural Gas Company and City of Nebraska City, Nebraska, which settlements were authorized by the Commission along with the NCG Settlement, and the settlement filed July 12, 1994 relating to recovery by Natural of Supply Realignment Costs from Industrial Energy Applications, Inc., which settlement was authorized by the Commission in Docket No. RP94-346 on September 29, 1994, 68 FERC Paragraph 61,388.

(j) G CUSTOMERS

The entities formerly served by Natural under Rate Schedule G-1 which are listed in Section 38.5(a)(1) of these General Terms and Conditions and which are covered by the G Customer Settlement.

(k) G CUSTOMER SETTLEMENT

The settlement filed on August 11, 1994 relating to recovery by Natural of Supply Realignment Costs from G Customers and other matters, which settlement was authorized by the Commission in Docket No. RP94-346 on September 29, 1994, 68 FERC Paragraph 61,388.

(1) INITIAL EFFECTIVE DATE

Initial Effective Date shall mean December 1, 1993.

38.3 APPLICATION OF CHARGES

(a) The charges calculated under Section 38.4 hereof shall be assessed to members of the Natural Customer Group and Other Consenting Customers as separate monthly billings of Demand Surcharges based on fixed transportation billing determinants.

FERC Docket: RP08-319-000

Original Sheet No. 498 Original Sheet No. 498: Pending

GENERAL TERMS AND CONDITIONS

(b) The charges under Section 38.5 shall be assessed to ${\tt G}$ Customers as volumetric charges or as Demand Surcharges in accordance with the options set out therein.

- (c) The charges under Section 38.6 shall be assessed as Demand Surcharges for Assessable Transportation Services.
- $\,$ (d) $\,$ The charges under Section 38.7 shall be assessed on a volumetric basis.
- (e) The charges under Section 38.8 shall be assessed as Demand Surcharges on Assessable Transportation Services, as set out in Sections 38.4, 38.5 and 38.6.
- (f) The charges under Section 38.11 shall be assessed as Demand Surcharges or on a volumetric basis, as applicable.
- $\,$ (g) The charges determined under Section 38.12 shall be assessed as Demand Surcharges.
 - 38.4 SUPPLY REALIGNMENT COST CHARGES TO NATURAL CUSTOMER GROUP AND OTHER CONSENTING CUSTOMERS

This Section 38.4 sets out charges to members of the Natural Customer Group for Supply Realignment Costs as provided in the NCG Settlement. This Section also sets out charges on a consistent basis for Other Consenting Customers. Such charges ceased effective December 1, 2001.

- (a) (1) Members of the Natural Customer Group shall pay a Demand Surcharge for recovery of Supply Realignment Costs (GSR Demand Surcharge) each month on the following transportation billing units: Interstate Power Company -- 9,781; IES Utilities, Inc. (relating only to contracts formerly held by Iowa Southern Utilities Company and Iowa Electric Light and Power Company) -- 39,805; MidAmerican Energy Company -- 136,713; Northern Illinois Gas Company -- 1,143,989; Northern Indiana Public Service Company -- 511,780; North Shore Gas Company -- 164,180; Peoples Gas Light and Coke Company -- 669,026; and Wisconsin Natural Gas Company (relating only to contracts formerly held by Wisconsin Southern Gas Company) -- 46,992. These transportation billing units for recovery of Supply Realignment Costs are fixed for a period of forty-eight (48) months commencing December 1, 1993.
- (2) The average rate for the GSR Demand Surcharge to members of the Natural Customer Group for the forty-eight (48) month billing period commencing December 1, 1993 applicable to the transportation billings units set out in subsection (a)(1) shall not

FERC Docket: RP08-319-000

Original Sheet No. 499 Original Sheet No. 499: Pending

GENERAL TERMS AND CONDITIONS

exceed \$3.12, except as provided in subsection (a)(5). The rate applicable in any quarter shall be a rate filed by Natural, which may not exceed \$3.42 [except as provided in subsection (a)(5)] and such rate may only exceed \$3.12 in any quarter [except as provided in subsection (a)(5)] to the extent Natural has previously charged less than \$3.12 [i.e., the cumulative average of the rates which have been charged by Natural may not exceed \$3.12 at any point during the forty-eight (48) month period except as provided in subsection (a)(5)]. If Natural charges a rate less than \$2.81, then for the purpose of calculating the average rate Natural is allowed to charge over the forty-eight (48) month recovery period, the rate for any period such lower rate is in effect shall be reflected at These rates are not specifically related to, and need not be supported by, costs incurred at any point in time but are subject to reconciliation and refund under Section 38.10 of these General Terms and Conditions.

(3) Once Natural has sufficient information to determine the total amount of its Supply Realignment Costs, Natural shall file to adjust the GSR Demand Surcharge applicable hereunder. Such filing shall set out the rate (including zero if applicable) which is necessary to recover the remaining unrecovered balance of such costs allocated to the Natural Customer Group members from them over the remainder of the forty-eight (48) month recovery period, which rate must be consistent with the ceiling rates set forth in subsection (a)(2). Natural may at any time reduce this rate further or suspend billings, but may not increase any rate charged under this subsection (a)(3). Any reduction in rate under this subsection (a)(3) cannot be offset by an increase in rate under subsection (a)(2).

(4) Natural may change the GSR Demand Surcharge applicable to members of the Natural Customer Group once each quarter, consistent with the parameters set out herein, by filing a revised rate at least thirty (30) days prior to the beginning of such quarter. For the purposes of the NCG Settlement, quarters shall mean the three (3) month periods commencing December 1, 1993 and on the first day of each succeeding third month over the ensuing forty-eight (48) month period; provided, however, that for billing purposes the initial quarter shall consist of two months (January and February, 1994) but the rate level for this initial quarter shall be subject to the same limitations in subsection (a) (1) as for any other quarter; and provided, further, that the rates set out in filings by Natural in Docket Nos. RP94-87, RP94-122, RP94-169, RP94-195, RP94-249 and RP94-260 satisfy this subsection for the periods covered by such filings.

FERC Docket: RP08-319-000

Original Sheet No. 500 Original Sheet No. 500: Pending

GENERAL TERMS AND CONDITIONS

(5) The NCG Settlement is predicated upon a forty-eight (48) month billing period commencing December 1, 1993. Natural shall adjust the rate hereunder applicable to the remainder of the recovery period after the Effective Date of the NCG Settlement such that total collections are consistent with the collections had the rates provided for hereunder been in effect commencing December 1, 1993. Natural shall reflect applicable interest in such adjusted rate consistent with Section 38.10(a)(3) of these General Terms and Conditions to compensate Natural or the Natural Customer Group for any resulting timing difference in collection. Collections of Supply Realignment Costs from Natural Customer Group members under any other applicable recovery mechanism rate prior to the Effective Date of the NCG Settlement shall be taken into account in calculating such rate. The cap on Natural's collections under subsection (b) shall be adjusted for any interest which is attributable solely to the fact that the NCG Settlement is not implemented in accordance with the schedule set out in this subsection (a).

(b) (1) Excluding the flowthrough of transition costs from upstream pipelines (it being recognized, however, that all upstream transition costs associated with Coal Gas are encompassed by the NCG Settlement), and except as provided in Section $38.10\,(a)\,(5)$, the total amount of Supply Realignment Costs recoverable from the Natural Customer Group, inclusive of interest on principal amounts allocated to the Natural Customer Group under subsection (b)(2) shall not exceed \$408 million, which amount represents the collections under the transportation billing units and maximum average surcharge set out in subsection (a). Notwithstanding the prior sentence, Natural may make the adjustment provided for in subsection (a)(5). In addition, in the event the total aggregate interest owed by Natural to Natural Customer Group members at the time of the reconciliation, as calculated under Section 38.10(a)(3)(ii) of these General Terms and Conditions, exceeds the total aggregate interest owed to Natural by the Natural Customer Group at the time of the reconciliation, as calculated under Section 38.10(a)(3)(i) of these General Terms and Conditions, the cap shall be adjusted downward by the net amount of such interest.

(2) The total principal amount of Supply Realignment Costs recoverable from Natural Customer Group members (exclusive of transition costs flowed through from upstream pipelines) shall, subject to the limitation in subsection (b)(1) equal the total principal amount of Natural's Supply Realignment Costs less the sum of the following exclusions:

FERC Docket: RP08-319-000

Original Sheet No. 501 Original Sheet No. 501: Pending

GENERAL TERMS AND CONDITIONS

(i) twenty percent (20%) of the first \$300 million in Supply Realignment Costs;

(ii) twenty-eight percent (28%) of the Supply Realignment Costs in excess of \$300 million, but less than \$500\$ million; and

(iii) thirty-five percent (35%) of Supply Realignment Costs in excess of \$500 million. Interest shall be assessed on the principal amounts allocated above in accordance with Section 38.10(a)(3) of these General Terms and Conditions. If this calculation results in an amount of Supply Realignment Costs plus interest in excess of the cap of \$408 million, as adjusted under subsections (a)(5) and (b)(1), Natural's recovery of Supply Realignment Costs from the Natural Customer Group shall be limited to \$408 million consistent with subsection (b)(1) except for any adjustment under subsections (a)(5) and (b)(1). If the calculation results in an amount less than the cap of \$408 million, as adjusted under subsections (a)(5) and (b)(1), Natural shall make refunds to the Natural Customer Group in accordance with the reconciliation procedures in Section 38.10(a) of these General Terms and Conditions.

- (c) All matters related to recovery of Supply Realignment Costs from Natural Customer Group members shall be resolved consistent with the NCG Settlement. In particular, issues of eligibility and prudence are subject to Article IV of the NCG Settlement.
- (d) (1) This Section 38.4 sets out a comprehensive procedure with respect to the allocation of Supply Realignment Costs to, and collection of Supply Realignment Costs from, members of the Natural Customer Group, including Coal Gas costs under Section 38.8 of these General Terms and Conditions. For purposes of this mechanism, any actual or imputed collection of Supply Realignment Costs by Natural from other customers shall not reduce such recoveries from the Natural Customer Group members nor [except as provided in subsection (d)(2)] shall the Natural Customer Group members be entitled to any credit or refund as a result of Natural having collected or being deemed to have collected any Supply Realignment Costs from any other Shipper; provided, however, nothing herein shall be read as inconsistent with the crediting provisions of Section 38.7 or with the additional collections under Section 38.10(a)(5).

FERC Docket: RP08-319-000

Original Sheet No. 502 Original Sheet No. 502: Pending

GENERAL TERMS AND CONDITIONS

(2) Supply Realignment Costs shall not be assessed against any specific Agreement held by any member of the Natural Customer Group; provided, however, with respect to Wisconsin Natural Gas Company, this statement is limited to Agreements formerly held by Wisconsin Southern Gas Company or successor Agreements arising out of such Agreements, and with respect to IES Utilities, Inc., this statement is limited to Agreements formerly held by Iowa Southern Utilities Company and Iowa Electric Light and Power Company or successor Agreements arising out of such Agreements; and provided, further, that nothing herein shall preclude Natural from seeking to assess, or shall prejudice any Natural Customer Group member in objecting to an assessment of, the following:

- (i) generally applicable charges, including any amounts for recovery of Supply Realignment Costs, assessed on any ITS Agreements, whether or not held by a Natural Customer Group member; and
- (ii) generally applicable charges to recover Supply Realignment Costs assessed on new firm transportation Agreement(s) of a Natural Customer Group member to the extent such Agreement(s) would increase the total firm contract demand of such Natural Customer Group member for transportation services on Natural above the level of its aggregate firm contract demand for firm transportation services on Natural in effect on December 1, 1993.
- (e) (1) The Other Consenting Customers shall be assessed the same Demand Surcharge as Natural Customer Group members in any month, based on the following transportation billing determinants which are fixed for a period of forty-eight (48) months from December 1, 1993:

Associated Natural Gas	Company	5 , 477
City of Nebraska City,	Nebraska	4,100
Industrial Energy Appli	ications, Inc.	2,744

The rate paid by the Other Consenting Customers shall be adjusted to equal the effective rate paid by members of the Natural Customer Group based on the reconciliation in Section $38.10\,(a)$. Natural shall pay interest on any resulting refunds at the rate prescribed in the Commission's Regulations at $18~\mathrm{C.F.R.}$ Section $154.501\,(d)$.

FERC Docket: RP08-319-000

Original Sheet No. 503 Original Sheet No. 503: Pending

GENERAL TERMS AND CONDITIONS

(2) All matters related to recovery of Supply Realignment Costs from Other Consenting Customers shall be resolved consistent with the Other Settlements. In particular, issues of eligibility and prudence are subject to Article IV of the Other Settlements.

38.5 SUPPLY REALIGNMENT COST CHARGE TO G CUSTOMERS

 $\hbox{This Section 38.5 sets out charges for Supply Realignment Costs to G Customers consistent with the G Customer Settlement. }$

(a) (1) The G Customers and their responsibility for Supply Realignment Costs in their traditional market areas are as follows:

Effective Date: 04/18/2008 Status: Effective FERC Docket: RP08-319-000

Original Sheet No. 504 Original Sheet No. 504: Pending

GENERAL TERMS AND CONDITIONS

G Customers	Billing Determinants	Allocated Cost
Agenda, Kansas, City of Arkansas Louisiana Gas Company Bethany, Illinois, Village of Brighton, Iowa, Town of Central Illinois Public Service Company Corn, Oklahoma, Town of Corning Municipal Utilities Entex Findlay, Illinois, Village of Frohna, Missouri, City of Grand Tower, Illinois, City of Lenox, Iowa, Town of Lorimor, Iowa, City of Marietta, Texas, City of Monarch Gas Company Montezuma, Iowa, Town of Nashville, Illinois, City of Pawnee Rock, Kansas, City of Peoples Natural Gas Company Perryville, Missouri, City of Salem, Illinois, City of Spearville, Kansas, City of Sullivan, Illinois, City of Twin County Gas Company United Cities Gas Company Wellman, Iowa, Town of Western Resources (KPL Gas)	125 174 1,500 490 3,993 389 2,000 1,100 750 500 430 4,000 319 579 8,500 1,743 5,025 308 1,402 4,975 3,000 7,000 600 5,500 807 4,000 6,800 828 619	\$ 14,639 20,378 175,670 57,385 467,634 45,557 234,227 128,825 87,835 58,557 50,359 468,453 37,359 67,809 995,464 204,129 588,495 36,071 164,193 582,639 351,340 819,794 70,268 644,123 94,510 468,453 796,371 96,970 72,493
Total		\$7,900,000 ======

FERC Docket: RP08-319-000

Original Sheet No. 505 Original Sheet No. 505: Pending

GENERAL TERMS AND CONDITIONS

(2) Each of the amounts in (1) is a fixed obligation for the designated customer which is not affected by termination of any Agreement, but shall be subject to adjustment pursuant to Section $38.5\,(\mathrm{d})$.

- (3) (i) This Section 38.5 shall be a comprehensive treatment of the assessment of Supply Realignment Costs to G Customers and the charges provided for herein are in lieu of all otherwise applicable charges for Supply Realignment Costs. Natural's recovery of Supply Realignment Costs from G Customers relating to service in their traditional market areas shall be limited to the provisions of this Section 38.5 and Supply Realignment Costs recoverable hereunder shall not [except as expressly provided in Section 38.5(d)] be subject to adjustment. No demand charges or volumetric charges other than provided herein for recovery of such charges shall apply to any Agreement of any of the G Customers for delivery in such customer's traditional market area. For the purposes hereof, the term "traditional market area" shall refer to markets of any of the G Customers into which Natural had at any time since 1985 provided G-1 service for such customer.
- (ii) Any actual or imputed collection of Supply Realignment Costs by Natural from other customers shall not reduce recoveries from any of the G Customers nor, except as otherwise provided in Section 38.7, shall any of the G Customers be entitled to any credit or refund as a result of Natural having collected or being deemed to have collected any Supply Realignment Costs from any other Shipper. Underrecoveries of Supply Realignment Costs from other customers (whether firm or interruptible) shall not be shifted to any of the G Customers nor shall overrecoveries of Supply Realignment Costs from any such customers reduce the allocation of Supply Realignment Costs to any of the G Customers.
- (iii) Notwithstanding the remainder of this subsection (3), nothing herein shall prejudice or preclude Natural from filing to collect pursuant to generally applicable tariff provisions any transition costs under Order Nos. 636, et seq., flowed through from upstream pipelines (it being understood, however, that the G Customer Settlement encompasses all upstream transition costs associated with Coal Gas). Each of the G Customers reserves all rights to protest and object to any filing by Natural to institute such recovery. Further, to the extent Supply Realignment Costs are allocated to and embedded in base rates for Rate Schedule ITS, nothing herein shall be read to preclude collection of such base rates by Natural from any of the G Customers

FERC Docket: RP08-319-000

Original Sheet No. 506 original Sheet No. 506: Pending

GENERAL TERMS AND CONDITIONS

nor shall collection of any such amounts in ITS base rates reduce or be counted toward the cost responsibility of any of the G Customers hereunder.

- (b) (1) Each of the G Customers shall be billed monthly an amount for Supply Realignment Costs equal to the product of: (i) the volumes delivered to or for the account of such customer in its traditional market area that month under any type of transportation agreement with Natural multiplied by (ii) a one-part volumetric rate which is the derivative at an imputed fifty percent (50%) load factor of a Demand Surcharge of \$3.12. Such monthly billings shall continue for each of the G Customers until that customer has fully paid its Supply Realignment Cost obligation set out in Section 38.5(a)(1), as adjusted under Section 38.5(d). Alternatively, any of the G Customers may choose to make a lump sum payment under (2) in lieu of such charges. Any of the G Customers electing to pay the volumetric one-part rate hereunder shall pay interest on the cumulative amount by which collections from December 1, 1993 are less than if it had paid monthly Demand Surcharges commencing December 1, 1993 of \$2.44 per unit of billing determinants underlying the G Customer Settlement [as shown in Section $38.5(a)\,(1)$], as adjusted under Section 38.5(d). Interest shall be at the Commission approved rate in 18.C.F.R. Section 154.501(d).
- (2) Any of the G Customers may at any time prepay all or any portion of its Supply Realignment Cost responsibility set out in Section 38.5(a)(1). Such cost responsibility shall be reduced to present value for such a prepayment by application of a discount factor equal to the Commission approved interest rate under 18 C.F.R. Section 154.501 on the assumption the customer paid the amount allocable to it in Section 38.5(a)(1) in equal monthly installments over that portion of the forty-eight (48) months commencing December 1, 1993 which remains as of the date payment is made.
- (3) Billings under this Section $38.5\,(\mathrm{b})$, and the associated interest, are subject to adjustment pursuant to Section $38.5\,(\mathrm{d})$.
- (4) Any amounts paid by any of the G Customers pursuant to Natural's Supply Realignment Cost recovery provisions in effect prior to the date this Section 38.5 becomes effective shall be applied to reduce the remaining balance of the amount due from that customer set out in Section 38.5(a)(1). Natural shall pay

FERC Docket: RP08-319-000

Original Sheet No. 507 Original Sheet No. 507: Pending

GENERAL TERMS AND CONDITIONS

interest to the extent such payments exceeded monthly demand charges commencing December 1, 1993 of \$2.44 per unit of billing determinants underlying the G Customer Settlement [as shown in Section 38.5(a) (1)], as adjusted under Section 38.5(d).

- (5) In the event that any of the G Customers no longer has any transportation agreement for delivery into its traditional market area in effect on Natural but still has an unpaid balance remaining of the amount set out in Section 38.5(a)(1), as adjusted under Section 38.5(d), such customers shall pay Natural such remaining unpaid balance, plus applicable interest at Commission approved rates, in equal monthly installments over the period during which such customer would have fully funded its cost responsibility at the average monthly Supply Realignment Cost revenue collected from it throughout the interval after December 1, 1993 when such customer did have such transportation agreement(s) on Natural.
- (c) The G Customer Settlement resolves all issues of prudence and eligibility as between Natural and the G Customers with respect to Order No. 636 transition costs; provided, however, that any of the G Customers may raise eligibility issues to the extent consistent with Article IV of the NCG Settlement and shall have the same rights with respect to such issues as a member of the Natural Customer Group under said Article IV.
- (d) (1) On or before September 1, 1998, Natural shall prepare and file with the Commission and provide to the G Customers a Reconciliation Report setting out by customer total collections hereunder from the G Customers compared with the amounts recoverable from each of the G Customers under Section 38.5(a)(1). Natural will be preparing concurrently a Reconciliation Report for the Natural Customer Group under Section 38.10(a). In the event that the average Demand Surcharge rate for the Natural Customer Group for the forty-eight (48) month period commencing December 1, 1993, recalculated based on such Natural Customer Group reconciliation, is less than \$3.12, then the amount payable by each of the G Customers shall be reduced to an amount equal to the figure shown in Section 38.5(a)(1) multiplied by the ratio of the recalculated Demand Surcharge rate to \$3.12. Any such lower amount shall be incorporated in the reconciliation under this subsection.

FERC Docket: RP08-319-000

Original Sheet No. 508 Original Sheet No. 508: Pending

GENERAL TERMS AND CONDITIONS

(2) If collections from any of the G Customers equal that customer's cost responsibility under Section 38.5(a)(1), as adjusted under this subsection (d), collections hereunder from that customer shall cease and that customer shall have no further responsibility for Supply Realignment Costs. If collections from any of the G Customers exceed that customer's cost responsibility under Section 38.5(a)(1), as adjusted under this Section 38.5(d), then Natural shall refund the difference between the amount collected hereunder from that customer and its cost responsibility, plus applicable interest at the Commission approved rate under 18 C.F.R. Section 154.501(d). In the event that total collections from any of the G Customers hereunder are less than that customer's total cost responsibility in Section 38.5(a)(1), as adjusted under this Section 38.5(d), billing hereunder shall continue at rates provided for in Section 38.5(b), and Natural shall repeat the reconciliation procedure at twelve (12) month intervals, until such customer's cost responsibility has been fully satisfied.

 $\hbox{(3)} \quad \hbox{The Reconciliation Report shall be made} \\ \hbox{available only to G Customers and interested state commissions.}$

38.6 DEMAND SURCHARGES FOR NON-SETTLING CUSTOMERS

This Section governs the Demand Surcharges to be assessed to Assessable Shippers. Such Assessable Shippers are non-settling firm transportation customers of Natural, the Supply Realignment Cost charges to which are not governed by either Section 38.4 or Section 38.5 of these General Terms and Conditions. Such Demand Surcharges ceased effective December 1, 2001.

(a) Each Assessable Shipper shall be subject to a monthly Demand Surcharge. Commencing June 1, 1995, the Demand Surcharge over the remainder of the Amortization Period shall equal \$3.12 per unit of contract demand per month for deliveries in Natural's Midwest Zone, as defined in Section 21.7 of these General Terms and Conditions. For deliveries in any other zone, the rate shall be twenty percent (20%) of \$3.12 or \$.624. Assessments of Demand Surcharges shall be based on the contract demands for Assessable Shippers in effect from time to time. Such rates shall be filed consistent with subsection (b). Demand Surcharges collected from any Assessable Shipper shall be subject to adjustment as provided in subsection (e) below.

FERC Docket: RP08-319-000

Original Sheet No. 509 Original Sheet No. 509: Pending

GENERAL TERMS AND CONDITIONS

(b) The monthly Demand Surcharge rate shall only be subject to adjustment during the Amortization Period on a quarterly basis to the extent the rate actually being assessed to members of the Natural Customer Group under Section 38.4 is less than \$3.12. For any Extended Amortization Period, rates shall be calculated consistent with Section 38.10(b), but subject to the limitation in subsection (d) of this Section 38.6 as to collections under any individual Agreement. Changes in the level of the Demand Surcharge to track such revised rates are to be made by the filing of a revised tariff sheet to be effective the ensuing March 1, June 1, September 1, or December 1, subject to refund, upon thirty (30) days' notice. Filings hereunder need not be supported by specific cost data but are subject to the reconciliation and refund provisions of subsection (e) of this Section 38.6 and of Section 38.10 of these General Terms and Conditions. No quarterly filing need be made for any quarter if no change in rate is required to maintain rates consistent with subsection (a).

- (c) Natural shall maintain an Adjustment Account which shall reflect the difference for the period commencing December 1, 1993, between total revenues collected under this Section 38.6 or precessor provisions of this Section 38 and the Supply Realignment Costs allocable to Assessable Shippers. The amount of Supply Realignment Costs allocable to Assessable Shippers shall be determined as follows:
- (1) Natural shall first determine the total Supply Realignment Costs in a manner consistent with Section 38.2(a) of these General Terms and Conditions;
- (2) Ten percent (10%) of the Supply Realignment Costs determined under (1) will be allocated for recovery pursuant to Section 38.7 of these General Terms and Conditions;
- (3) Next, Natural shall exclude all costs allocable to the Natural Customer Group, G Customers and Other Consenting Customers under Sections 38.4 and 38.5 of these General Terms and Conditions; and
- (4) The remaining Supply Realignment Costs shall be allocated among Assessable Shippers (which allocation may reflect appropriate customer classes) based on their respective contract demands for Assessable Transportation Services in effect at that time.

FERC Docket: RP08-319-000

Original Sheet No. 510 original Sheet No. 510: Pending

GENERAL TERMS AND CONDITIONS

Natural shall be deemed to have collected Supply Realignment Costs from all Shippers with Assessable Transportation Services based on MDQs in effect from time to time, irrespective of whether Natural had discounted such Agreements, except as provided in filings pursuant to Section 38.10(c); provided, however, that no Supply Realignment Costs shall be allocated to, nor shall Natural be deemed to have collected Supply Realignment Costs from all Shippers with Assessable Transportation Services based on MDQs in effect from time to time, irrespective of whether Natural had discounted such Agreements, except as provided in filings pursuant to Section 38.10(c); provided, however, that no Supply Realignment Costs shall be allocated to, nor shall Natural be deemed to have collected Supply Realignment Costs on, any Agreement which replaces an Agreement currently held by a settling customer and is within the contract demand for such customer utilized as billing determinants for calculating Supply Realignment Costs for such customers in Section 38.4 or 38.5. If, at the end of the Amortization Period, a positive balance remains in the Adjustment Account, Natural may establish rates to be assessed to Assessable Shippers for an Extended Amortization Period, which rates shall equal the rates specified in subsection (a) or such lower rate as was last assessed during the Amortization Period under subsection (b), but subject to the limitations of subsection (d).

- (d) Notwithstanding any other provision of this Section 38, no Assessable Shipper shall be billed or obligated to pay Supply Realignment Costs for any firm Agreement in excess of forty-eight (48) times the applicable contract MDQ times \$3.12 for Midwest Zone deliveries or \$.624 for deliveries in the Production Zones, as Midwest and Production Zones are defined in Section 21.7 of these General Terms and Conditions [or such lesser monthly Demand Surcharge rate as is determined in a reconciliation under subsection (e)]. Nothing herein shall be read or implied to override any contractual arrangement between Natural and any Shipper resulting in a lesser payment of Supply Realignment Costs under any Agreement.
- (e) Billings to and collections from Assessable Shippers shall be subject to a reconciliation consistent with Section 38.10 of these General Terms and Conditions at the end of the Amortization Period and of any Extended Amortization Period. In the event that the reconciliation for the Natural Customer Group under Section 38.10(a) of these General Terms and Conditions results in a lower average monthly demand surcharge rate than \$3.12 for Midwest Zone deliveries and \$.624 for deliveries in the Production Zones, as Midwest and Production Zones are defined in Section 21.7 of these General Terms and Conditions, Natural shall make refunds with applicable interest to any Shipper which has

FERC Docket: RP08-319-000

Original Sheet No. 511 Original Sheet No. 511: Pending

GENERAL TERMS AND CONDITIONS

actually paid more than such redetermined rate per unit of contract demand under any Agreement times the number of months [not to exceed forty-eight (48)] that surcharges hereunder were applicable to its Agreement(s) and the limitation set out in subsection (d) shall be adjusted downward accordingly. In the event Natural has collected more from any Assessable Shipper than is properly allocable to or recoverable from it, Natural shall make refunds with applicable interest consistent with Section 38.10(b) of these General Terms and Conditions.

38.7 INTERRUPTIBLE RATES AND CREDITS

(a) Subject to the provisions of Section 38.11, the ten percent (10%) of Supply Realignment Costs which were excluded under Sections 38.6 and 38.8 of these General Terms and Conditions from the calculation of Demand Surcharges shall be allocated by Natural to Rate Schedule ITS. Natural shall also allocate a representative level of non-gas fixed costs to Rate Schedule ITS. Such allocation ceased effective December 1, 2001. Such representative level shall be determined in Natural's general rate case(s). The level of the ITS rate shall be determined in Natural's general rate case(s) and may be a load factor rate without regard to the costs allocated to Rate Schedule ITS; provided, however, that Natural shall exclude, in developing the ITS rate, any component attributable to Demand Surcharges.

FERC Docket: RP08-319-000

Original Sheet No. 512 original Sheet No. 512: Pending

GENERAL TERMS AND CONDITIONS

(b) Natural shall provide an aggregate credit to all Shippers under Rate Schedules DSS and NSS equal to ninety percent (90%) of any revenue collected under Rate Schedule BESS in excess of the sum of (i) the non-gas fixed costs allocated to BESS service in the design of Natural's underlying base storage rates and (ii) all variable costs reflected in BESS rates. The determination of any such revenue credit shall be made on an annual basis, commencing with the period ending November 30, 1994 and annually thereafter until the Commission allows termination of this crediting mechanism; provided, however, that Natural shall not be required to begin applying previously accrued credits to customer bills until the day after the Commission has issued a final order in Docket No. RP93-36, it being understood that Natural shall pay interest at the Commission authorized rate if credits are deferred as a result of this proviso. Any excess revenue BESS credit shall be reflected as a credit on the bills of individual Shippers. percentage of excess revenues to be allocated to each Shipper will be determined pursuant to the following ratio: the annual reservation revenues collected from each Shipper under the Rate Schedule designated above attributable to storage to the total of all such revenues collected from all such Shippers and attributable to storage during the same time period. Any excess revenues so determined will be passed back to individual Shippers as a credit against their subsequent transportation or storage billings until completely refunded. In no event shall a Shipper's Reservation Charge be reduced below zero.

38.8 COAL GASIFICATION CHARGES

- (a) Except as provided in subsection (d), Natural shall continue to purchase Coal Gas from Dakota pursuant to the provisions of the existing contract, or any modified contract mutually agreed upon by Natural and Dakota (Coal Gas Contract), and shall recover the attendant costs, including all costs of transportation to the point(s) of sale in accordance with this Section.
- (b) Natural will conduct an auction of the Coal Gas for each month commencing with December, 1993 for delivery at Harper, Iowa, at the interconnection between the facilities of Dakota and Northern Border Pipeline Company (Northern Border) (or other valid

FERC Docket: RP08-319-000

Original Sheet No. 513 Original Sheet No. 513: Pending

GENERAL TERMS AND CONDITIONS

point under Natural's agreement with Dakota) or at the interconnection between Northern Border and Northern Natural Gas Company; provided, however, that Natural may auction all or any portion of the Coal Gas for a period exceeding one month if, in Natural's reasonable judgment, such an auction would likely reduce costs. The auction will be conducted pursuant to the procedures set out in subsection (e) of this Section. If the auction price for Coal Gas is less than the sum of the cost of the Coal Gas delivered by Dakota and all transportation costs to Harper, Iowa, and (if the sale is at a point other than Harper, Iowa) all transportation costs to such other point of sale (in addition to the unavoidable fixed costs of transportation to Harper, Iowa), Natural shall recover this difference (Premium) pursuant to subsection (c). Unless the sale is made at a point other than Harper, Iowa, Natural will reserve for itself upstream capacity sufficient for Natural to move the Coal Gas to Harper, Iowa.

- (c) Natural shall recover that portion of the ninety percent (90%) of the Premium allocable to the Natural Customer Group, G Customers and Other Consenting Customers pursuant to the procedures set out in Sections 38.4 and 38.5. Natural shall recover the remainder of the Premium, or that portion thereof allocable to Assessable Shippers consistent with Section 38.6, as part of the Demand Surcharge(s) determined under Section 38.6. Ten percent (10%) of the Premium shall be recoverable under Section 38.7 of these General Terms and Conditions.
- (d) Notwithstanding the foregoing, at any time Natural and Dakota may reach a mutual agreement for the buyout or buydown of the Coal Gas Contract. Any such buyout costs are to be encompassed in the procedures under Sections 38.4 and 38.5 of these General Terms and Conditions for the Natural Customer Group, G Customers and Other Consenting Customers, respectively, under Section 38.6 of these General Terms and Conditions for nonsettling customers and under Section 38.7 of these General Terms and Conditions for costs allocated to interruptible services. The procedures under this Section 38.8 shall continue in effect for determining that portion of Supply Realignment Costs other than buyout/buydown costs or demand charges associated with Coal Gas until the Coal Gas Contract terminates or is assigned by Natural.

FERC Docket: RP08-319-000

Original Sheet No. 514 Original Sheet No. 514: Pending

GENERAL TERMS AND CONDITIONS

38.9 ACCOUNTING

A separate account by customer of collections and applicable interest under Sections $38.4,\ 38.5,\ 38.6,\ 38.7,\ 38.11$ and 38.12 shall be maintained by Natural.

38.10 RECONCILIATION

- Natural shall prepare and file with the (a) (1) Commission and provide to Natural Customer Group members a Reconciliation Report relating to collections from Natural Customer Group members only within nine (9) months after the earlier of: (i) the end of the forty-eight (48) month collection period defined in Section 38.4(a); or (ii) the date billings by Natural cease under Section 38.4(a). Such report shall compare total collections hereunder from Natural Customer Group members with the amounts allocated to the Natural Customer Group under Section 38.4, reflecting Supply Realignment Costs (including Coal Gas costs) determined as of the date of the reconciliation. To the extent such Supply Realignment Costs have not actually been paid by Natural at the time of the reconciliation, Natural shall reflect an amount for such remaining Supply Realignment Costs determined by discounting the outstanding obligation to its present value utilizing the interest rate in subsection (a)(3)(ii) and employing estimates to the extent necessary.
- (2) If the amount allocated to Natural Customer Group members exceeds \$408 million (as adjusted under Section 38.4), Natural shall retain all collections from the Natural Customer Group. If the amount so allocated is less than \$408 million (as adjusted under Section 38.4), Natural shall refund the difference between the amount actually collected from Natural Customer Group members under Section 38.4 and the allocated amount (as adjusted) plus applicable interest. Allocation of such refunds among Natural Customer Group members shall be based on the actual collections hereunder from each Natural Customer Group member.

FERC Docket: RP08-319-000

Original Sheet No. 515 Original Sheet No. 515: Pending

GENERAL TERMS AND CONDITIONS

(3) For the purposes of this reconciliation, applicable interest as to Natural Customer Group members shall be determined as follows:

(i) to the extent the cumulative amount Natural has actually paid for Supply Realignment Costs allocable to Natural Customer Group members under the formula in Section 38.4 exceeds, as of the end of any quarter, the cumulative collections from Natural Customer Group members, Natural shall be entitled to add interest for the quarter on such excess at the rate then in effect under the Commission's Regulations at the Commission-approved rate under 18 C.F.R. Section 154.501(d); and

(ii) to the extent the cumulative collections Natural has received hereunder from Natural Customer Group members exceeds, as of the end of any quarter, the cumulative actual payments of Supply Realignment Costs allocable to Natural Customer Group members under the formula in Section 38.4, Natural shall reduce Supply Realignment Costs by an interest factor applied to such excess equal to its overall after-tax rate of return last approved in a final Commission order no longer subject to appeal.

For the purposes of this subsection (a)(3), the term "cumulative" shall refer to cumulative costs and cumulative collections from the Initial Effective Date, including interest accruals for all prior quarters. Interest utilizing the rates set out above shall be calculated for each quarter based on the average of the ending cumulative balance of the quarter and the ending cumulative balance of the prior quarter.

(4) In the event that Natural has based its determination of any part of the amount allocated to Natural Customer Group members on estimates in the reconciliation hereunder, Natural shall prepare a further reconciliation once the outstanding obligation has been fully discharged, Natural shall make refunds, including applicable interest, consistent with subsection (a) (3) if the actual costs incurred were less than such estimates, to the extent that use of the actual amounts incurred in lieu of the estimates would have resulted in a refund in the original reconciliation. If more than one item of Supply Realignment Costs has been estimated in this manner, Natural shall perform a separate reconciliation from time to time as each such item is discharged.

FERC Docket: RP08-319-000

Original Sheet No. 516 original Sheet No. 516: Pending

- (5) Reconciliation Report(s) under this subsection (a) shall be made available only to the Commission, Natural Customer Group members and interested state commissions.
- (6) Natural shall concurrently prepare Reconciliation Reports for Other Consenting Customers consistent with Section 38.4(e) (1) and for G Customers consistent with Section 38.5(d).
- (7) Recovery of Supply Realignment Costs under Rate Schedule ITS and the crediting of ITS revenue shall be governed by Sections 38.7 and 38.11.
- (b) (1) Natural shall, no later than September 1, 1998, submit a Reconciliation Report consistent with Section 38.6 and shall revise the rates assessed under Section 38.6 to correspond to the average rate paid by members of the Natural Customer Group as adjusted by the reconciliation under subsection (a). Natural shall also prepare a final reconciliation within nine (9) months after the end of the Extended Amortization Period.
- (2) The Reconciliation Report to be prepared by September 1, 1998, shall compare: (i) total revenues collected (or deemed to be collected under the following sentence) pursuant to Section 38.6 with the collections which would have been made using the rate paid by members of the Natural Customer Group, as adjusted by the reconciliation under subsection (a) of this Section 38.10. For the purposes of this Reconciliation, Natural shall be deemed to have collected Supply Realignment Costs from all Shippers with Assessable Transportation Services based on the MDQs in effect from time to time and the applicable rates under Section 38.6, irrespective of whether Natural discounted under such Agreements, except as provided in Section 38.6(c) and in subsection (c) of this Section 38.10. To the extent Supply Realignment Costs are allocated on different bases, a separate reconciliation shall be made of costs and revenues for each such separate allocation. If the $% \left(1\right) =\left(1\right) \left(1\right)$ collections which would have been made at the adjusted Natural Customer Group rates are less than the amount actually collected, Natural shall make a refund. No individual Shipper shall pay more with respect to any Agreement than is provided in Section 38.6(d). Refunds shall be calculated separately for each separate reconciliation.

FERC Docket: RP08-319-000

Original Sheet No. 517 Original Sheet No. 517: Pending

- (3) The final reconciliation due within nine (9) months after the end of the Extended Amortization Period shall compare total Supply Realignment Costs to total collections under this Section 38. For the purposes of this final reconciliation, Natural shall utilize the attribution rules employed in subsection (2) above. If costs are less than the amount collected, Natural shall make a corresponding refund.
- (4) Any refund hereunder shall reflect interest calculated pursuant to Section 154.501(d) of the Commission's Regulations. Refunds shall be determined separately for each customer class subject to a different allocation of costs. Such refunds shall be allocated among such customer classes and within each such class in the ratio that the amount properly billed to and collected from each entity which is subject to payments under Section 38.6 bears to the total of such billings to and collections from all such entities during the period covered by the Reconciliation Report.
- (5) Reconciliation Reports submitted under this Section shall be available on a confidential basis to the Commission and to customers paying any charge hereunder directly or indirectly.
- Natural in any way with respect to any filing, including filings under Section 38.11, it may make to recover any Supply Realignment Costs allocated to interruptible service which Natural has been unable to recover through rates to interruptible customers, the Commission having provided assurance that it did not intend that such an allocation of costs to interruptible service would result in underrecovery; provided, however, that Natural may not seek to recover such costs from the Natural Customer Group, G Customers or Other Consenting Customers to the extent such recovery would be inconsistent with the NCG Settlement or the G Customer Settlement.
- (c) Natural may make a filing to roll-forward any costs allocated to Assessable Shippers under Section 38.6 but not actually recovered from such Shippers. Such filing shall define the limits or such roll-forward and the individual cost responsibility of any Assessable Shipper. Any person other than a member of the Natural Customer Group or Other Consenting Customers is free to object to such a filing.

FERC Docket: RP08-319-000

Original Sheet No. 518 original Sheet No. 518: Pending

GENERAL TERMS AND CONDITIONS

38.11 COLLECTION OF UNRECOVERED SUPPLY REALIGNMENT COSTS

- (a) Natural has net underrecoveries of approximately \$22.9 million in Supply Realignment Costs, resulting from costs allocated to Rate Schedule ITS but not recovered from ITS Shippers netted against recoveries from firm Shippers, as of December 1, 1998. To collect this underrecovery, the following charges will be instituted during the Extended Amortization Period:
- (1) A maximum Demand Surcharge of up to \$0.3600 for the Market Zone and up to \$0.0700 for the Production Zone, to be assessed where contractually permitted on each firm contract under Rate Schedules FTS and FFTS; provided, however, that no such charge shall apply if on such contract or its predecessor, the Shipper has paid on the aggregate the maximum amount of Supply Realignment Costs assessable under Sections 38.6(c) and 38.6(d) hereof, which is equal to \$3.12 times forty-eight (48) months times the contract MDQ; and, provided further, that such a surcharge shall only apply to the contracts of the Natural Customer Group or Other Consenting Parties to the extent such contracts represent contract demand in excess of the aggregate firm contract demand on Natural in effect for such customer as of December 1, 1993, and no charge hereunder shall apply to any G Customer if inconsistent with Section 38.5(a)(3). The charge will become effective on December 1, 1998.
- (2) Natural shall incorporate in its maximum rate under Rate Schedule ITS an amount attributable to Supply Realignment Costs equal to \$0.0012 per Dth for Midwest Zone deliveries and \$0.0002 for Production Zone deliveries. The adjustment to ITS rates hereunder will become effective on December 1, 1998.
- (b) The charges under this Section 38.11 are discountable to zero. The charges under this Section 38.11 shall terminate on the earliest of: (1) the full collection hereunder by Natural of all Supply Realignment Costs, applying the attribution rules specified in Section 38.10(b); (2) the full collection by Natural of the amount of net underrecoveries identified in subsection (a) hereof, applying the attribution rule specified in Section 38.10(b)(2); or (3) November 30, 2001.

FERC Docket: RP08-319-000

Original Sheet No. 519 original Sheet No. 519: Pending

GENERAL TERMS AND CONDITIONS

(c) Within nine (9) months after the assessments under this Section 38.11 cease, which occurred effective December 1, 2001, Natural will file with the Commission a reconciliation of all collections thereunder with total Supply Realignment Costs and with the actual underrecovery of Supply Realignment Costs allocated to Rate Schedule ITS. Natural shall not assess any interest on the underrecovery of Supply Realignment Costs initially allocated to Rate Schedule ITS subsequent to the earliest date on which any of the charges hereunder commence.

38.12 UPSTREAM PIPELINE COSTS

Natural may flow-through any Order No. 636 transition costs paid by it to any upstream pipeline, including any such amounts flowed through by such upstream pipeline from other upstream pipeline(s). Such amounts shall be flowed through by Natural to Shippers through a demand charge based on the sum of the MDQs of Assessable Transportation Services in effect at the time Natural seeks to recover such costs and the billing units of the Natural Customer Group, G Customers and Other Consenting Customers; provided, however, that any non-settling customer shall have the right to protest the resulting allocation of costs to it. Billings under this Section shall reflect amortization of costs over the longer of: (1) where applicable, the remainder of the Amortization Period, or an Extended Amortization Period; or (2) one year; provided that any Shipper shall also have the option of a lump sum payment or of amortization over a one-year period.

38.13 TERM

This Section shall remain in effect until all Supply Realignment Costs have been fully recovered as provided herein and Reconciliation Reports have been finalized, including the distribution of any refunds associated with such Reconciliation; provided, however, that Section 38.12 hereof shall remain in effect for so long as Natural is subject to billing of transition costs by any upstream pipeline, and Section 38.8 shall remain in effect as provided by its terms.

FERC Docket: RP08-319-000

Sheet No. 520 Sheet No. 520: Pending

GENERAL TERMS AND CONDITIONS

39. RESERVED FOR FUTURE USE

FERC Docket: RP08-319-000

Original Sheet No. 521 Original Sheet No. 521: Pending

GENERAL TERMS AND CONDITIONS

40. ANNUAL CHARGES ADJUSTMENT CHARGE

40.1 PURPOSE

This Section of the General Terms and Conditions is filed pursuant to Section 154.402 and Part 382 (Subpart B) of the Federal Energy Regulatory Commission's Regulations under the Natural Gas Act and the Natural Gas Policy Act of 1978. The intent and purpose of this Section is to establish an Annual Charges Adjustment (ACA) provision under which Natural can recover from its customers annual charges assessed to it by the Commission pursuant to Part 382 of the Commission's Regulations (ACA Cost). All amounts assessed pursuant to Part 382 of the Commission's Regulations shall be recorded in Account 928. Natural will not seek to recover annual charges assessed to it pursuant to Part 382 of the Commission's Regulations in an NGA Section 4 rate case. For the purpose of recovering annual charges assessed to Natural pursuant to Part 382 of the Commission's Regulations, this Section establishes an ACA charge as set forth in the Currently Effective Rates section of this Tariff.

40.2 APPLICABILITY

The ACA charge shall be applicable to all transportation transactions performed by Natural, including all transportation performed pursuant to X Schedules in Natural's FERC Gas Tariff, Volume No. 2.

40.3 BASIS OF THE ACA CHARGE

The rates for all transactions specified in Section 40.2 hereof shall be adjusted by a unit charge to recover ACA Cost. Such unit charge shall be that increment, adjusted to Natural's pressure base and heating value, if required, which has been established by the Commission. The ACA unit charge shall be applied to the commodity component of rates.

40.4 FILING PROCEDURE

The ACA charge shall be filed annually by Natural at least thirty (30) days prior to the Effective Date of Charge. Any such filing shall become effective on the effective date of charges hereunder without suspension or refund obligation.

40.5 EFFECTIVE DATE OF CHARGE

 $$\operatorname{\textsc{The}}$ effective date of charges filed pursuant to this Section shall be October 1.

FERC Docket: RP08-319-000

Original Sheet No. 522 Original Sheet No. 522: Pending

GENERAL TERMS AND CONDITIONS

41. RESEARCH, DEVELOPMENT AND DEMONSTRATION ADJUSTMENT

Periodically, rates and charges under Natural's Tariff shall be adjusted to reflect changes in Natural's expenditures for research, development and demonstration (RD&D); provided, however, such adjustment shall not include funding payments to Gas Research Institute (GRI) that are to be recovered by Natural under the provisions of Section 39 of these General Terms and Conditions. The expenditures recoverable under this Section shall include both expenditures chargeable to operations as well as expenditures includible in rate base.

41.1 PROCEDURES FOR COMPUTATION OF ADJUSTMENT

The Research, Development and Demonstration Adjustment (RD&D Adjustment) shall be reflected in rates only when it amounts to at least \$0.0001 per Dth of annual jurisdictional services. The RD&D Adjustment shall be applied to the commodity component of all applicable rates and shall maintain the then existing rate design relationships among the rates and charges under affected Rate Schedules.

(a) RESEARCH, DEVELOPMENT AND DEMONSTRATION CHARGEABLE TO OPERATIONS

Except for expenditures as to which approval has been granted pursuant to Section 154.401 of the Commission's Regulations under the Natural Gas Act, Natural's RD&D Adjustment for expenditures chargeable to operations shall reflect the amount by which actual RD&D expenditures chargeable to operations during the twelve (12) month period ending three (3) months prior to the effective date of the RD&D Adjustment exceed or are less than the actual RD&D expenditures in Natural's last RD&D Adjustment under this Section.

(b) RESEARCH, DEVELOPMENT AND DEMONSTRATION INCLUDIBLE IN RATE BASE

The RD&D Adjustment shall reflect the amount by which the actual balances in Account 188 during the twelve (12) month period ending three (3) months prior to the effective date of the RD&D Adjustment exceed or are less than the balances in Account 188 included in Natural's last prior RD&D Adjustment under this Section. For the purpose of the computation under this

FERC Docket: RP08-319-000

Original Sheet No. 523 Original Sheet No. 523: Pending

GENERAL TERMS AND CONDITIONS

subsection (b), Natural shall reduce the balance in Account 188 by all monies recorded in Account 495 applicable to the RD&D expenditures, and shall increase or reduce such account balance, as appropriate, by the deferred income taxes in Account 283 applicable to RD&D expenditures. The rate of return used to determine the rate effect of the rate base treatment of the balance in Account 188 shall be the rate of return last allowed by the Commission during the previous three year period. If there has been no such rate of return allowed during the previous three-year period, then, in the absence of evidence submitted to the contrary, the return utilized shall be the present interest rate used for computing refunds as specified in Section 154.501(d) of the Commission's Regulations under the Natural Gas Act.

41.2 PROCEDURE FOR FILING

- (a) Rate changes under this Section shall be computed and filed not more frequently than semiannually. Any filing made hereunder to increase Natural's rates shall meet the notice requirements of Section 154.207 of the Commission's Regulations and shall be made at least forty-five (45) days prior to the date on which any change(s) in Natural's existing rates is to become effective. Simultaneously with such filing, Natural shall furnish to the Commission, jurisdictional customers, and interested State Commissions a report containing detailed computations which clearly show the derivation of the proposed RD&D Adjustment. The effect upon jurisdictional rates shall be determined by computing the unit change (either an increase or a decrease) based upon volumes of jurisdictional services for the twelve (12) month period ending three (3) months prior to the effective date of such RD&D Adjustment.
- (b) Filings made under this Section shall include a statement as to the anticipated scope and objective of the RD&D and the relationship of such objective to the service for which the RD&D Adjustment is to apply.
- (c) Each RD&D Adjustment shall become effective on the proposed effective date without suspension; provided that, except in the case of rate adjustments based on expenditures approval pursuant to Section 154.401 of the Regulations, any rate increase shall be subject to reduction and to refund of any portion found after hearing to be unjustified by a final and non-appealable Commission order.
- $\,$ (d) Natural shall not be required to reduce its rates under this Section by an increment exceeding the aggregate increases allowed hereunder.

FERC Docket: RP08-319-000

Sheet No. 524 Sheet No. 524 : Pending

GENERAL TERMS AND CONDITIONS

42. RESERVED FOR FUTURE USE.

FERC Docket: RP08-319-000

Original Sheet No. 525 Original Sheet No. 525: Pending

GENERAL TERMS AND CONDITIONS

43. REPORTS

Natural's Merchant Division (if any) or Transportation Division, as applicable, shall file such reports of jurisdictional gas sales, storage, transportation or other activity as may be required by the FERC or other governmental agencies. Shippers shall cooperate by providing Natural on a timely basis with such information as may be required by Natural from Shipper to prepare such reports.

FERC Docket: RP08-319-000

Original Sheet No. 526 Original Sheet No. 526: Pending

GENERAL TERMS AND CONDITIONS

44. WAIVERS

Natural may waive any of its rights or any obligations of Shipper under this Tariff on a basis which is not unduly discriminatory; provided that no waiver shall operate or be construed as a waiver of other or future rights or obligations, whether of a like of different character.

FERC Docket: RP08-319-000

Sheet No. 527 Sheet No. 527 : Pending

GENERAL TERMS AND CONDITIONS

45. RESERVED FOR FUTURE USE

FERC Docket: RP08-319-000

Original Sheet No. 528 Original Sheet No. 528: Pending

GENERAL TERMS AND CONDITIONS

46. GATHERING AFFILIATES

46.1 Natural will provide nondiscriminatory access to all sources of supply in accordance with Part 284 of the Commission's Regulations and the provisions of this Tariff, and will not give Shippers who also are receiving service from any of Natural's gathering affiliates undue preference over Shippers receiving services from nonaffiliated gatherers or over other Shippers in scheduling, transportation, storage or curtailment priority.

46.2 Natural will not condition or tie its agreement to provide transportation service to an agreement by a producer, customer, end-user, or Shipper relating to (a) any service by any gathering affiliate of Natural, (b) any services by Natural on behalf of any of its gathering affiliates, or (c) any services in which any of Natural's gathering affiliates are involved.

FERC Docket: RP08-319-000

Sheet No. 529 Sheet No. 529 : Pending

GENERAL TERMS AND CONDITIONS

47. RESERVED FOR FUTURE USE

FERC Docket: RP09-650-000

First Revised Sheet No. 530 First Revised Sheet No. 530

Superseding: Original Sheet No. 530

GENERAL TERMS AND CONDITIONS

48. COMPLIANCE WITH 18 C.F.R., SECTION 284.12

48.1 Natural shall comply with the following business practice and electronic communication standards incorporated by reference in Section 284.12 of the Commission's Regulations (18 C.F.R. Section 284.12):

(a) ADDITIONAL STANDARDS (Version 1.8): 0.1.1, 0.1.2, 0.1.3, 0.2.1, 0.2.2, 0.2.3, 0.3.1, 0.3.2, 0.3.11, 0.3.12, 0.3.13, 0.3.14, and 0.3.15.

(b) NOMINATIONS, CONFIRMATIONS AND SCHEDULING (Version 1.8): 1.1.9, 1.1.12, 1.1.13, 1.1.14, 1.1.16, 1.1.17, 1.1.20, 1.1.21, 1.1.22, 1.2.1, 1.2.2, 1.2.3, 1.2.5, 1.2.6, 1.2.8, 1.2.9, 1.2.10, 1.2.11, 1.2.13, 1.2.14, 1.2.15, 1.2.16, 1.2.17, 1.2.18, 1.2.19, 1.3.2 (vi), 1.3.7, 1.3.15, 1.3.16, 1.3.17, 1.3.18, 1.3.20, 1.3.24, 1.3.25, 1.3.27, 1.3.28, 1.3.29, 1.3.30, 1.3.31, 1.3.33, 1.3.34, 1.3.35, 1.3.36, 1.3.37, 1.3.38, 1.3.39, 1.3.40, 1.3.41, 1.3.42, 1.3.43, 1.3.44, 1.3.45, 1.3.46, 1.3.47, 1.3.48, 1.3.49, 1.3.50, 1.3.51, 1.3.52, 1.3.53, 1.3.54, 1.3.55, 1.3.56, 1.3.57, 1.3.58, 1.3.59, 1.3.60, 1.3.61, 1.3.62, 1.3.63, 1.3.64, 1.3.65, 1.3.66, 1.3.67, 1.3.68, 1.3.69, 1.3.70, 1.3.71, 1.3.72, 1.3.73, 1.3.74, 1.3.75, 1.3.76, 1.3.77, and 1.3.79.

(c) FLOWING GAS (Version 1.8): 2.1.2, 2.1.3, 2.1.5, 2.1.6, 2.2.2, 2.2.3, 2.2.4, 2.2.5, 2.3.1, 2.3.2, 2.3.4, 2.3.7, 2.3.8, 2.3.11, 2.3.12, 2.3.13, 2.3.15, 2.3.19, 2.3.21, 2.3.22, 2.3.23, 2.3.25, 2.3.27, 2.3.29, 2.3.30, 2.3.31, 2.3.32, 2.3.33, 2.3.34, 2.3.35, 2.3.40, 2.3.41, 2.3.42, 2.3.43, 2.3.44, 2.3.45, 2.3.46, 2.3.47, 2.3.48, 2.3.49, 2.3.50, 2.3.51, 2.3.52, 2.3.53, 2.3.54, 2.3.55, 2.3.56, 2.3.57, 2.3.58, 2.3.59, 2.3.60, 2.3.61, 2.3.62, 2.3.63, and 2.3.64.

(d) INVOICING (Version 1.8): 3.3.1, 3.3.2, 3.3.3, 3.3.4, 3.3.5, 3.3.6, 3.3.7, 3.3.8, 3.3.10, 3.3.11, 3.3.12, 3.3.13, 3.3.16, 3.3.20, 3.3.21, 3.3.22, 3.3.23, 3.3.24, 3.3.25, and 3.3.26.

(e) QUADRANT ELECTRONIC DELIVERY MECHANISMS (Version 1.8): 4.1.2, 4.1.3, 4.1.4, 4.1.6, 4.1.7, 4.1.10, 4.1.12, 4.1.13, 4.1.15, 4.1.16, 4.1.17, 4.1.18, 4.1.19, 4.1.20, 4.1.21, 4.1.22, 4.1.23, 4.1.24, 4.1.26, 4.1.27, 4.1.28, 4.1.29, 4.1.30, 4.1.31, 4.1.32, 4.1.33, 4.1.34, 4.1.35, 4.1.36, 4.1.37, 4.1.38, 4.1.39, 4.1.40, 4.2.1, 4.2.2, 4.2.3, 4.2.4, 4.2.5, 4.2.6, 4.2.7, 4.2.8, 4.2.9, 4.2.10, 4.2.11, 4.2.12, 4.2.13, 4.2.14, 4.2.15, 4.2.16, 4.2.17, 4.2.18, 4.2.19, 4.2.20, 4.3.1, 4.3.2, 4.3.3, 4.3.5,

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

Second Revised Sheet No. 530 Second Revised Sheet No. 530

Superseding: First Revised Sheet No. 530

GENERAL TERMS AND CONDITIONS

48. COMPLIANCE WITH 18 C.F.R., SECTION 284.12

48.1 Natural has adopted the Business Practices and Electronic Communications Standards which are required by the Commission in 18 CFR, Section 284.12(a), as amended from time to time, in accordance with Order Nos. 587, et al. In addition to the NAESB WGQ Standards referenced elsewhere in the Tariff, Natural specifically incorporates by reference the following NAESB WGQ Version 1.9 Standards, Definitions, and Data Sets, by reference:

- (a) ADDITIONAL STANDARDS
 - (1) General:
 - (A) Standards: 0.3.1, 0.3.2, 0.3.16, 0.3.17
 - (2) Creditworthiness:
 - (A) Standards:
 - (3) Gas/Electric Operational Communications:
 - (A) Definitions: 0.2.1, 0.2.2, 0.2.3
 - (B) Standards: 0.3.11, 0.3.12, 0.3.13,

0.3.14, 0.3.15

- (4) Storage Information:
 - (A) Data Sets: 0.4.1
- (5) Nominations Related Standards:

```
(A) Definitions: 1.2.1, 1.2.2, 1.2.3, 1.2.5, 1.2.6, 1.2.8, 1.2.9, 1.2.10, 1.2.11, 1.2.13, 1.2.14, 1.2.15, 1.2.16, 1.2.17, 1.2.18, 1.2.19
```

```
(B) Standards: 1.3.2(vi), 1.3.9, 1.3.11, 1.3.15, 1.3.16, 1.3.17, 1.3.18, 1.3.20, 1.3.24, 1.3.25, 1.3.27, 1.3.28, 1.3.29, 1.3.30, 1.3.31, 1.3.33, 1.3.34, 1.3.35, 1.3.36, 1.3.37, 1.3.38, 1.3.39, 1.3.40, 1.3.41, 1.3.42, 1.3.43, 1.3.44, 1.3.45, 1.3.46, 1.3.47, 1.3.48, 1.3.49, 1.3.50, 1.3.51, 1.3.52, 1.3.53, 1.3.54, 1.3.55, 1.3.56, 1.3.57, 1.3.58, 1.3.59, 1.3.60, 1.3.61, 1.3.62, 1.3.63, 1.3.64, 1.3.65, 1.3.66, 1.3.67, 1.3.68, 1.3.69, 1.3.70, 1.3.71, 1.3.72, 1.3.73, 1.3.74, 1.3.75, 1.3.76, 1.3.77, 1.3.79, 1.3.80
```

FERC Docket: RP09-865-000

Second Revised Sheet No. 531 Second Revised Sheet No. 531

Superseding: First Revised Sheet No. 531

```
GENERAL TERMS AND CONDITIONS
```

```
4.3.16, 4.3.17, 4.3.18, 4.3.20, 4.3.22, 4.3.23, 4.3.24, 4.3.25, 4.3.26, 4.3.27, 4.3.28, 4.3.29, 4.3.30, 4.3.31, 4.3.32, 4.3.33, 4.3.34, 4.3.35, 4.3.36, 4.3.38, 4.3.39, 4.3.40, 4.3.41, 4.3.42, 4.3.43, 4.3.44, 4.3.45, 4.3.46, 4.3.47, 4.3.48, 4.3.49, 4.3.50, 4.3.51, 4.3.52, 4.3.53, 4.3.54, 4.3.55, 4.3.56, 4.3.57, 4.3.58, 4.3.59, 4.3.60, 4.3.61, 4.3.62, 4.3.65, 4.3.66, 4.3.67, 4.3.68, 4.3.69, 4.3.72, 4.3.73, 4.3.74, 4.3.75, 4.3.76, 4.3.78, 4.3.79, 4.3.80, 4.3.81, 4.3.82, 4.3.83, 4.3.84, 4.3.85, 4.3.86, 4.3.87, and 4.3.93.

(f) CAPACITY RELEASE (Version 1.8): 5.1.2, 5.1.3, 5.1.4, 5.2.1, 5.2.2, 5.2.3, 5.3.7, 5.3.9, 5.3.10, 5.3.12, 5.3.18, 5.3.20, 5.3.21, 5.3.22, 5.3.23, 5.3.24, 5.3.29, 5.3.30, 5.3.31, 5.3.32, 5.3.33, 5.3.34, 5.3.35, 5.3.36, 5.3.37, 5.3.38, 5.3.39, 5.3.40, 5.3.41, 5.3.42, 5.3.43, 5.3.54, and 5.3.60.

(g) INTERNET ELECTRONIC TRANSPORT (Version 1.8): 10.1.1, 10.1.2, 10.1.3, 10.1.4, 10.1.5, 10.1.6, 10.1.7, 10.1.8, 10.1.9, 10.2.1, 10.2.2, 10.2.3, 10.2.4, 10.2.5, 10.2.6, 10.2.7, 10.2.8, 10.2.9, 10.2.10, 10.2.11, 10.2.12, 10.2.13, 10.2.14, 10.2.15, 10.2.16, 10.2.17, 10.2.18, 10.2.19, 10.2.20, 10.2.21, 10.2.22, 10.2.23, 10.2.24, 10.2.25, 10.2.26, 10.2.27, 20.2.28, 10.2.29, 10.2.30, 10.2.31, 10.2.32, 10.2.33, 10.2.34, 10.3.5, 10.3.6, 10.3.7, 10.3.8, 10.3.9, 10.3.10, 10.3.11, 10.3.12, 10.3.13, 10.3.14, 10.3.15, 10.3.16, 10.3.17, 10.3.18, 10.3.19, 10.3.20, 10.3.21, 10.3.22, 10.3.23, 10.3.24, and 10.3.25.
```

48.2 (a) Natural shall utilize the standardized datasets provided by the standards incorporated by reference in 18 C.F.R. Section 284.12. Natural's implementation guide for the standardized data sets specifies Natural's intended use, if any, of the data elements that are coded as "business conditional" (BC) and "mutually agreeable" (MA) for purposes of EDI. The implementation guide may be obtained by contacting the Website Security Help Line and asking for the EDI Coordinator. The Website Security Help Line phone number can be found at http://pipeline.kindermorgan.com.

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

Third Revised Sheet No. 531 Third Revised Sheet No. 531

Superseding: Second Revised Sheet No. 531

GENERAL TERMS AND CONDITIONS

```
(C) Data Sets: 1.4.1, 1.4.2, 1.4.3, 1.4.4, 1.4.5, 1.4.6, 1.4.7
```

(6) Flowing Gas Related Standards:

```
(A) Definitions: 2.2.2, 2.2.3, 2.2.4, 2.2.5
```

```
(B) Standards: 2.3.1, 2.3.2, 2.3.4, 2.3.7, 2.3.8, 2.3.11, 2.3.12, 2.3.13, 2.3.15, 2.3.19, 2.3.20, 2.3.21, 2.3.22, 2.3.23, 2.3.25, 2.3.27, 2.3.29, 2.3.30, 2.3.31, 2.3.32, 2.3.33, 2.3.34, 2.3.35, 2.3.40, 2.3.41, 2.3.42, 2.3.43, 2.3.44, 2.3.45, 2.3.46, 2.3.47, 2.3.48, 2.3.49, 2.3.50, 2.3.51, 2.3.52, 2.3.53, 2.3.54, 2.3.55, 2.3.56, 2.3.57, 2.3.58, 2.3.59, 2.3.60, 2.3.61, 2.3.62, 2.3.63, 2.3.64
```

(C) Data Sets: 2.4.1, 2.4.2, 2.4.3, 2.4.4, 2.4.5, 2.4.6, 2.4.7, 2.4.8, 2.4.9, 2.4.10, 2.4.11, 2.4.12, 2.4.13, 2.4.14, 2.4.15, 2.4.16

(7) Invoicing Related Standards:

(A) Definitions:

```
(B) Standards: 3.3.1, 3.3.2, 3.3.3, 3.3.4, 3.3.5, 3.3.6, 3.3.7, 3.3.8, 3.3.10, 3.3.11, 3.3.12, 3.3.13, 3.3.16, 3.3.20, 3.3.21, 3.3.22, 3.3.23, 3.3.24, 3.3.25, 3.3.26
```

(C) Data Sets: 3.4.1, 3.4.2, 3.4.3, 3.4.4

(8) Quadrant Electronic Delivery Mechanism Related Standards:

```
(A) Definitions: 4.2.1, 4.2.2, 4.2.3, 4.2.4, 4.2.5, 4.2.6, 4.2.7, 4.2.8, 4.2.9, 4.2.10, 4.2.11, 4.2.12, 4.2.13, 4.2.14, 4.2.15, 4.2.16, 4.2.17, 4.2.18, 4.2.19, 4.2.20
```

```
(B) Standards: 4.3.1, 4.3.2, 4.3.3, 4.3.5, 4.3.16, 4.3.17, 4.3.18, 4.3.20, 4.3.22, 4.3.23, 4.3.24, 4.3.25, 4.3.26, 4.3.27, 4.3.28, 4.3.29, 4.3.30, 4.3.31, 4.3.32, 4.3.33, 4.3.34, 4.3.35, 4.3.36, 4.3.38, 4.3.39, 4.3.40, 4.3.41, 4.3.42, 4.3.43, 4.3.44, 4.3.45, 4.3.46, 4.3.47, 4.3.48, 4.3.49, 4.3.50, 4.3.51, 4.3.52, 4.3.53, 4.3.54, 4.3.55, 4.3.56, 4.3.57, 4.3.58, 4.3.59, 4.3.60, 4.3.61, 4.3.62, 4.3.65, 4.3.66, 4.3.67, 4.3.68, 4.3.69, 4.3.72, 4.3.73, 4.3.74, 4.3.75, 4.3.76, 4.3.78, 4.3.79, 4.3.80, 4.3.81, 4.3.82, 4.3.83, 4.3.84, 4.3.85, 4.3.86, 4.3.87, 4.3.93, 4.3.94, 4.3.95, 4.3.96, 4.3.97, 4.3.98, 4.3.99
```

FERC Docket: RP09-650-000

First Revised Sheet No. 532 First Revised Sheet No. 532

Superseding: Original Sheet No. 532

GENERAL TERMS AND CONDITIONS

- $\begin{tabular}{ll} \begin{tabular}{ll} \beg$
- (1) ADDITIONAL RELATED STANDARDS (Version 1.8):

0.4.1.

- (2) NOMINATIONS RELATED STANDARDS (Version 1.8): 1.4.1, 1.4.2, 1.4.3, 1.4.4, 1.4.5, 1.4.6, and 1.4.7.
- (3) FLOWING GAS RELATED STANDARDS (Version 1.8): 2.4.1, 2.4.2, 2.4.3, 2.4.4, 2.4.5, 2.4.6, 2.4.7, 2.4.8, 2.4.9, 2.4.10, 2.4.11, 2.4.12, 2.4.13, 2.4.14, 2.4.15, and 2.4.16.
- (4) INVOICING RELATED STANDARDS (Version 1.8): 3.4.1, 3.4.2, 3.4.3, and 3.4.4.
- (5) CAPACITY RELEASE RELATED STANDARDS (Version 1.8): 5.4.1, 5.4.2, 5.4.3, 5.4.4, 5.4.5, 5.4.6, 5.4.7, 5.4.8, 5.4.9, 5.4.10, 5.4.11, 5.4.12, 5.4.13, 5.4.14, 5.4.15, 5.4.16, 5.4.17, 5.4.18, 5.4.19, 5.4.20, 5.4.21, and 5.4.22.
- (c) Natural shall utilize the North American Energy Standards Board Trading Partner Agreement (Version 1.8, Standard 6.3.3).
- $48.3\,$ Natural's HTML page(s) is accessible via the Internet's World Wide Web at the following address:

http://pipeline.kindermorgan.com

Effective Date: 11/01/2010 Status: Pending

FERC Docket: RP10-1223-000

5.2.4, 5.2.5

Second Revised Sheet No. 532 Second Revised Sheet No. 532

(A)

Superseding: First Revised Sheet No. 532

GENERAL TERMS AND CONDITIONS

(9) Capacity Release Standards:

```
(B) Standards: 5.3.7, 5.3.9, 5.3.10, 5.3.12, 5.3.13, 5.3.17, 5.3.18, 5.3.20, 5.3.21, 5.3.22, 5.3.23, 5.3.24, 5.3.27, 5.3.29, 5.3.30, 5.3.31, 5.3.32, 5.3.33, 5.3.34, 5.3.35, 5.3.36, 5.3.37, 5.3.38, 5.3.39, 5.3.40, 5.3.41, 5.3.42, 5.3.43, 5.3.54, 5.3.60, 5.3.61, 5.3.62, 5.3.62a, 5.3.63, 5.3.64, 5.3.65, 5.3.66, 5.3.67, 5.3.68, 5.3.69
```

(C) Data Sets: 5.4.1, 5.4.2, 5.4.3, 5.4.4, 5.4.5, 5.4.6, 5.4.7, 5.4.8, 5.4.9, 5.4.10, 5.4.11, 5.4.12, 5.4.13, 5.4.14, 5.4.15, 5.4.16, 5.4.17, 5.4.18, 5.4.19, 5.4.20, 5.4.21, 5.4.22

(10) Internet Electronic Transport Related Standards:

Definitions: 5.2.1, 5.2.2, 5.2.3,

```
(A) Definitions: 10.2.1, 10.2.2, 10.2.3, 10.2.4, 10.2.5, 10.2.6, 10.2.7, 10.2.8, 10.2.9, 10.2.10, 10.2.11, 10.2.12, 10.2.13, 10.2.14, 10.2.15, 10.2.16, 10.2.17, 10.2.18, 10.2.19, 10.2.20, 10.2.21, 10.2.22, 10.2.23, 10.2.24, 10.2.25, 10.2.26, 10.2.27, 10.2.28, 10.2.29, 10.2.30, 10.2.31, 10.2.32, 10.2.33, 10.2.34, 10.2.35, 10.2.36, 10.2.37, 10.2.38
```

```
(B) Standards: 10.3.1, 10.3.3, 10.3.4, 10.3.5, 10.3.6, 10.3.7, 10.3.8, 10.3.9, 10.3.10, 10.3.11, 10.3.12, 10.3.14, 10.3.15, 10.3.16, 10.3.17, 10.3.18, 10.3.19, 10.3.20, 10.3.21, 10.3.22, 10.3.23, 10.3.24, 10.3.25, 10.3.26, 10.3.27
```

- 48.2 (a) Natural shall utilize the standard data sets provided by the standards incorporated by reference in 18 CFR Section 284.12. Natural's implementation guide for the standardized data sets specifies Natural's intended use, if any, of the data elements that are coded as "business conditional" (BC) and "mutually agreeable" (MA) for purposes of EDI. The implementation guide may be obtained by contacting the Website Security Help Line and asking for the EDI Coordinator. The Website Security Help Line phone number can be found at http://pipeline.kindermorgan.com.
- (b) Natural shall utilize the North American Energy Standards Board Electronic Data Interchange Trading Partner Agreement Dated: June 15, 2009 (Version 1.9, Standard 6.3.3).
- 48.3 Natural's HTML page(s) is accessible via the Internet's World Wide Web at the following address: http://pipeline.kindermorgan.com

FERC Docket: RP08-319-000

Original Sheet No. 533 Original Sheet No. 533: Pending

GENERAL TERMS AND CONDITIONS

49. NEGOTTATED RATES

49.1 PRECONDITIONS TO NEGOTIATED RATES

Rates to be charged by Natural for service to any Shipper under Rate Schedule FTS, FFTS, FTS-G, ITS, IBS, BESS, DSS, NSS, FRSS, PALS or LPS may deviate in either form or level or both from the applicable maximum and/or minimum rate level in this Tariff, subject to the following provisions:

- (a) Natural and Shipper have executed a valid Agreement containing therein or in a related agreement a specific mutual understanding that Negotiated Rate(s) or a Negotiated Rate Formula will apply to service for that Shipper;
- (b) At the time of execution of the Agreement (or the amendment to an Agreement), which first provides for the applicability to Shipper of the Negotiated Rate(s) or Negotiated Rate Formula, service was available pursuant to the terms and conditions (not modified by this Section 49) of Rate Schedule FTS, FTSTS, FTS-G, ITS, IBS, BESS, DSS, NSS, FRSS, PALS OR LPS of this Tariff, as applicable;
- (c) Natural may only offer or accept offers for service under this Section 49 if and to the extent service is available at the Recourse Rate;
- (d) Natural will not negotiate terms and conditions of service under this Section $49\ensuremath{\,;}$ and
- No later than the Business Day on which Natural commences service at such Negotiated Rate(s) or Negotiated Rate Formula (or if the day on which Natural commences service is not a Business Day, then no later than the next Business Day after Natural commences service), Natural will file a tariff sheet advising the Commission of such Negotiated Rate or Negotiated Rate Formula, stating the name of Shipper, the type of service, the primary Receipt and Delivery Point(s) applicable to the service, the volume of the gas to be transported, any other charges, and specifying either: (i) the specific Negotiated Rate included in such Agreement; or (ii) the Negotiated Rate Formula included in such Agreement with sufficient specificity such that the rate in effect from time to time can be readily calculated. Either the tariff sheet must also incorporate a statement that the Agreement does not deviate from the form of Service Agreement in any material respect, or Natural must file the contract containing a material deviation.

FERC Docket: RP08-319-000

Original Sheet No. 534 Original Sheet No. 534: Pending

GENERAL TERMS AND CONDITIONS

49.2 CAPACITY ALLOCATION

(a) To the extent the revenue level pursuant to the Negotiated Rate(s) or Negotiated Rate Formula provided for in Section 49.1 above, as calculated under Subsection (b) of this Section 49.2, exceeds the comparable revenue level at the Recourse Rate, the Shipper bidding or paying such Negotiated Rate(s) or rate(s) under a Negotiated Rate Formula shall be treated, for all capacity allocation purposes, as if the rate(s) bid or paid had been equal to the Recourse Rate. Any Shipper, existing or new, paying the Recourse Rate(s) has the same right to capacity as a Shipper willing to pay a higher Negotiated Rate(s) or rate(s) under a Negotiated Rate Formula. If the Negotiated Rate or the rate under a Negotiated Rate Formula is higher than the corresponding Recourse Rate, the Recourse Rate rather than the Negotiated Rate will be used as the price cap for release capacity pursuant to Section 19.5 of these General Terms and Conditions and for the Right of First Refusal pursuant to Section 22.3 of these General Terms and Conditions. Where the Negotiated Rate(s) or rate(s) under a Negotiated Rate Formula results in revenue which is greater than the Recourse Rate during certain portions of the relevant evaluation period but less than the revenue at the Recourse Rate during other portions of the relevant evaluation period (but the revenue pursuant to the Negotiated Rate(s) or rate(s) under a Negotiated Rate Formula equals or exceeds that which would be generated at the Recourse Rate for the entire evaluation period), the value of bids and requests at the Negotiated Rate(s) or rate(s) under a Negotiated Rate Formula shall be evaluated as though the Recourse Rate applied under such bid or request for the entire evaluation period. Where the Negotiated Rate(s) or rate(s) under the Negotiated Rate Formula result in revenue which is less than revenue at the Recourse Rate over the relevant evaluation period, the value of the bids or requests at the Negotiated Rate(s) or rate(s) under the Negotiated Rate Formula shall be evaluated based on such lower revenue and shall be afforded a correspondingly lower priority than bids or requests at the Recourse Rate.

(b) In evaluating bids for firm service, in any capacity auction or in otherwise allocating capacity among competing requests for firm service where one or more bid is at a Negotiated Rate or Negotiated Rate Formula, Natural will consider, in assigning value to such bid(s), only reservation or demand charge revenue or other revenue which is guaranteed to be received by Natural (i.e., a

FERC Docket: RP08-319-000

Original Sheet No. 535 Original Sheet No. 535: Pending

GENERAL TERMS AND CONDITIONS

minimum throughput condition or minimum bill). For capacity evaluation purposes, the net present value of any such bid for firm service shall be capped by the net present value of the maximum applicable reservation rate for such service over the contract term bid.

49.3 ACCOUNTING FOR COSTS AND REVENUES

The allocation of costs to, and the recording of revenues from, service at Negotiated Rate(s) will follow Natural's normal practices associated with all of its services under this Tariff. Natural will maintain separate records of Negotiated Rate and Negotiated Rate Formula transactions for each billing period. These records shall include the volumes transported, the billing determinants (contract MDQ), the rates charged and the revenue received associated with such transactions. Natural will separately identify such transactions in Statements G, I and J (or their equivalent) filed in any general rate proceeding.

49.4 CAPACITY RELEASE REVENUE

Natural and Shipper may agree hereunder to a Negotiated Rate or Negotiated Rate Formula which includes payment obligations or crediting mechanisms in the event of a capacity release which vary from those set out in Section 19.13 of these General Terms and Conditions. Nothing in the forgoing sentence, however, shall authorize Natural or Shipper to violate the Commission's policy with respect to the negotiation of terms and conditions of service.

FERC Docket: RP10-909-000

Sixth Revised Sheet No. 536 Sixth Revised Sheet No. 536

Superseding: Fifth Revised Sheet No. 536

GENERAL TERMS AND CONDITIONS

50. Non-Conforming Agreements

The Commission has directed that the following Agreements be filed with the Commission because they contain provisions which do not conform to Natural's pro forma service agreements:

- 50.1 Wise County Power Company, LP Firm Transportation Negotiated Rate Agreement For Prearranged Capacity Dated September 19, 2000.
- 50.2 Green Valley Chemical Corporation Transportation Rate Schedule FTS Agreement Dated July 2, 2009 (Agreement No. 137095).
- 50.3 Black Hills Utility Holdings, Inc. (formerly Aquila, Inc. d/b/a Aquila Networks), Firm Transportation Rate Discount Agreement Dated June 28, 2002.
- 50.4 The Board of Trustees of University of Illinois, Firm Transportation Rate Discount Agreement, Dated August 13, 2004.
- $50.5\,$ OXY USA WTP, LP, Transportation Rate Schedule FTS Agreement, Dated April 27, 2005.
- 50.6 Black Hills Utility Holdings, Inc. (formerly Aquila, Inc., d/b/a Aquila Networks), Transportation Rate Schedule FTS Agreement, Dated August 29, 2002 (Agreement No. 127721), as amended by Amendment No. 4, Dated April 28, 2008.
- $50.7\,$ Nicor Gas Company, Transportation Rate Schedule FTS Agreement, Dated June 10, 2009 (Agreement No. 137030).
- $50.8\,$ Wisconsin Electric Power Company, Transportation Rate Schedule FTS Agreement, Dated June 12, 2003 (Agreement No. 129979), as amended by Amendment No. 4, Dated June 6, 2008.
- 50.9 Wisconsin Electric Power Company, Transportation Rate Schedule FTS Agreement, Dated June 12, 2003 (Agreement No. 129985), as amended by Amendment No. 6, Dated January 13, 2009.
- 50.10 Wisconsin Electric Power Company, Transportation Rate Schedule FTS Agreement, Dated June 12, 2003 (Agreement No. 129986), as amended by Amendment No. 8, Dated January 13, 2009.
- 50.11 BP Canada Energy Marketing Corporation, Storage Rate Schedule NSS Agreement, Dated April 17, 2003 (Agreement No. 129824), as amended by Amendment No. 7, Dated April 28, 2008.
- 50.12 Brazos Electric Power Cooperative, Inc., Transportation Rate Schedule FTS Agreement, Dated May 19, 2005 (Contract No. 131934).

FERC Docket: RP99-176-195

First Revised Sheet No. 537 First Revised Sheet No. 537

Superseding: Original Sheet No. 537

- 50. Non-Conforming Agreements (con't)
 - $50.13\,$ Brazos Electric Power Cooperative, Inc., Firm Transportation Rate Discount Agreement, Dated May 19, 2005 (Agreement No. 131937).
 - 50.14 Brazos Electric Power Cooperative, Inc., Transportation Rate Schedule IBS, Dated May 19, 2005 (Contract No. 131935).
 - 50.15 Brazos Electric Power Cooperative, Inc., Interruptible Balancing Service Rate Discount Agreement, Dated May 19, 2005 (Agreement No. 131953).
 - 50.16 Nicor Gas Company, Firm Transportation and Storage Negotiated Rate Agreement, Dated September 26, 2005 (Agreement No. 116184A2).
 - $50.17\,$ Suez Energy Marketing NA, Inc., Storage Rate Schedule NSS Agreement, Dated October 1, 2004 (Agreement No. 131191).
 - 50.18 Occidental Energy Marketing, Inc., Storage Rate Schedule NSS Agreement, Dated September 24, 2004 (Agreement No. 131170).
 - $50.19\,$ Oneok Energy Services Co., L.P., Storage Rate Schedule NSS Agreement, Dated October 4, 2004 (Agreement No. 131197).
 - 50.20 The Board of Trustees of University of Illinois, Firm Transportation Rate Discount Agreement, Dated February 13, 2006 (Agreement No. 132903).
 - $50.21\,$ Nicor Gas Company, Transportation Rate Schedule FTS Agreement, dated September 26, 2005 (Agreement No. 132373).
 - 50.22 Crosstex Gulf Coast Marketing Ltd., Transportation Rate Schedule FTS Agreement, Dated June 7, 2006 (Agreement No. 133345).
 - 50.23~ J-W Gathering Company, Transportation Rate Schedule FTS Agreement, Dated June 7, 2006 (Agreement No. 133344).
 - 50.24~ J-W Gathering Company, Transportation Rate Schedule FTS Agreement, Dated August 24, 2006 (Agreement No. 133509).
 - 50.25 Devon Gas Services, L.P., Transportation Rate Schedule FTS Agreement, Dated June 12, 2006 (Agreement No. 133348).
 - $50.26\,$ Devon Gas Services, L.P., Transportation Rate Schedule FTS Agreement, Dated September 11, 2006 (Agreement No. 133553).

FERC Docket: RP10-656-000

Fourth Revised Sheet No. 538 Fourth Revised Sheet No. 538

Superseding: Third Revised Sheet No. 538

- 50. Non-Conforming Agreements (con't)
 - 50.27 Occidental Energy Marketing, Inc., Storage Rate Schedule NSS Agreement, Dated April 5, 2005 (Agreement No. 131775).
 - 50.28 Kinder Morgan Tejas Pipeline, L.P., Storage Rate Schedule NSS Agreement, Dated April 19, 2005 (Agreement No. 131823).
 - $50.29\,$ Kinder Morgan Tejas Pipeline, L.P., Storage Rate Schedule NSS Agreement, Dated April 5, 2005 (Agreement No. 131776).
 - $50.30\,$ Enserco Energy Inc., Storage Rate Schedule NSS Agreement, Dated April 5, 2005 (Agreement No. 131777).
 - $50.31\,$ Enbridge Marketing (U.S.) L.P., Storage Rate Schedule NSS Agreement, Dated April 25, 2006 (Agreement No. 133209).
 - 50.32 Enbridge Marketing (U.S.) L.P., Storage Rate Schedule NSS Agreement, Dated June 21, 2006 (Agreement No. 133364).
 - $50.33\,$ Occidental Energy Marketing, Inc., Storage Rate Schedule NSS Agreement, Dated June 21, 2006 (Agreement No. 133363).
 - 50.34 Northern Indiana Public Service Company, Firm Transportation Rate Schedule FTS Agreement, Dated October 1, 2004 (Agreement No. 131188).
 - 50.35 Tenaska Marketing Ventures, Transportation Rate Schedule FTS Agreement, dated March 9, 2004 (Agreement No. 130687), as amended by Amendment No. 5, dated March 25, 2009.
 - 50.36 Wisconsin Electric Power Company, Transportation Rate Schedule FTS Agreement, Dated July 11, 2006 (Agreement No. 133396).
 - 50.37 Brazos Electric Power Cooperative Inc., Line Pack Service (Park and Loan) Agreement (LPS), Dated January 17, 2007 (Agreement No. 133867).
 - $50.38\,$ Occidental Energy Marketing, Inc., Storage Rate Schedule NSS Agreement, Dated June 21, 2006 (Agreement No. 133362), as amended by Amendment No. 1, Dated April 1, 2009.
 - 50.39 Laclede Energy Resources, Inc., Firm Transportation Negotiated Rate Agreement, Dated December 13, 2006 (Agreement No. 133779).

Effective Date: 05/01/2009 Status: Pending

FERC Docket: RP99-176-197

Second Revised Sheet No. 539 Second Revised Sheet No. 539

Superseding: First Revised Sheet No. 539

- 50. Non-Conforming Agreements (con't)
 - 50.40 Interstate Power and Light Company, Firm Transportation Negotiated Rate Agreement, Dated October 20, 2006 (Agreement No. 133589).
 - 50.41 Interstate Power and Light Company, DSS Storage Service Negotiated Rate Agreement, Dated October 23, 2006 (Agreement No. 133644).
 - 50.42 CenterPoint Energy Minnesota Gas, Firm Transportation and Storage Negotiated Rate Agreement, Dated July 24, 2006 (Agreement No. 133424)
 - 50.43 CenterPoint Energy Minnesota Gas, Transportation Rate Schedule FTS Agreement, Dated November 9, 1995 (Agreement No. 110813).
 - 50.44 Laclede Energy Resources, Inc., Firm Transportation Negotiated Rate Agreement, Dated December 13, 2006 (Agreement No. 133780).
 - 50.45 Chesapeake Energy Marketing, Inc., Transportation Rate Schedule FTS Agreement, Dated December 21, 2006 (Agreement No. 133817).
 - $50.46\,$ MidAmerican Energy Company, Amendment No. 1 Dated January 31, 2007, to Transportation Rate Schedule FTS Agreement Dated July 29, 2004 (Agreement No. 131024).
 - $50.47\,$ J.P. Morgan Ventures Energy Corporation, Transportation Rate Schedule FTS Agreement, Dated November 14, 2007 (Agreement No. 134926).
 - 50.48 Reserved For Future Use.
 - $50.49\,$ MidAmerican Energy Company, Storage Rate Schedule DSS Agreement, Dated October 30, 1995 (Agreement No. 110754).
 - 50.50 Encana Marketing (USA) Inc., Transportation Rate Schedule FTS Agreement, Dated February 23, 2006 (Agreement No. 132914).
 - 50.51 Reserved For Future Use.

FERC Docket: RP10-787-000

Fourth Revised Sheet No. 539 Fourth Revised Sheet No. 539

Superseding: Third Revised Sheet No. 539

- 50. Non-Conforming Agreements (con't)
 - 50.40 Interstate Power and Light Company, Firm Transportation Negotiated Rate Agreement, Dated October 20, 2006 (Agreement No. 133589).
 - 50.41 Interstate Power and Light Company, DSS Storage Service Negotiated Rate Agreement, Dated October 23, 2006 (Agreement No. 133644).
 - 50.42 CenterPoint Energy Minnesota Gas, Firm Transportation and Storage Negotiated Rate Agreement, Dated July 24, 2006 (Agreement No. 133424)
 - 50.43 CenterPoint Energy Minnesota Gas, Transportation Rate Schedule FTS Agreement, Dated November 9, 1995 (Agreement No. 110813).
 - $50.44\,$ Laclede Energy Resources, Inc., Firm Transportation Negotiated Rate Agreement, Dated December 13, 2006 (Agreement No. 133780).
 - 50.45 Chesapeake Energy Marketing, Inc., Transportation Rate Schedule FTS Agreement, Dated December 21, 2006 (Agreement No. 133817).
 - $50.46\,$ MidAmerican Energy Company, Amendment No. 1 Dated January 31, 2007, to Transportation Rate Schedule FTS Agreement Dated July 29, 2004 (Agreement No. 131024).
 - $50.47\,$ J.P. Morgan Ventures Energy Corporation, Transportation Rate Schedule FTS Agreement, Dated November 14, 2007 (Agreement No. 134926).
 - $50.48\,$ Occidental Energy Marketing, Inc., Storage Rate Schedule NSS, Dated July 23, 2004 (Agreement No. 131004), as amended by Amendment No. 1, Dated April 1, 2009.
 - 50.49 MidAmerican Energy Company, Storage Rate Schedule DSS Agreement, Dated October 30, 1995 (Agreement No. 110754).
 - $50.50\,$ Encana Marketing (USA) Inc., Transportation Rate Schedule FTS Agreement, Dated February 23, 2006 (Agreement No. 132914).
 - $50.51\,$ The Board of Trustees of University of Illinois, Transportation Rate Schedule ITS Agreement, Dated May 7, 2010 (Agreement No. 138179).

FERC Docket: RP10-970-000

Fifth Revised Sheet No. 540 Fifth Revised Sheet No. 540

Superseding: Fourth Revised Sheet No. 540

- 50. Non-Conforming Agreements (con't)
 - 50.52 Reliant Energy Wholesale Generation, LLC, Storage Rate Schedule NSS Agreement, Dated March 30, 2001 (Agreement No. 118949), as amended by Amendment No. 2, Dated October 11, 2006.
 - 50.53 Reliant Energy Wholesale Generation, LLC, Storage Rate Schedule NSS Agreement, Dated December 12, 2002 (Agreement No. 129101), as amended by Amendment No. 1, Dated October 11, 2006.
 - 50.54 Reliant Energy Wholesale Generation, LLC, Storage Rate Schedule NSS Agreement, Dated February 21, 2003 (Agreement No. 129651), as amended by Amendment No. 1, Dated October 11, 2006.
 - $50.55\,$ Occidental Energy Marketing, Inc., Transportation Rate Schedule FTS Agreement, Dated January 7, 2005 (Agreement No. 131457), as amended by Amendment No. 1, Dated March 21, 2007.
 - $50.56\,$ Tylex, Inc., Transportation Rate Schedule FTS Amendment No. 3 Dated February 16, 2007 to Agreement Dated June 30, 2003 (Contract No. 130017).
 - 50.57 Nicor Gas, Firm Transportation and Storage Negotiated Rate Agreement, Dated September 12, 2008 (Agreement No. 116184A3).
 - 50.58 United Energy Trading, LLC, Transportation Rate Schedule FTS Agreement, Dated November 17, 2004 (Agreement No. 131351), as amended by Amendment No. 2, Dated December 23, 2008.
 - 50.59 Shell Energy North America (US), L.P., Storage Rate Schedule NSS Agreement, Dated December 6, 2004 (Agreement No. 131391), as amended by Amendment No. 1, Dated February 10, 2009.
 - 50.60 Oneok Energy Services Company, L.P., Storage Rate Schedule NSS Agreement, Dated March 4, 1999 (Agreement No, 114829), as amended by Amendment No. 5, Dated April 2, 2009.
 - $50.61~\rm BP$ Energy Company, Storage Rate Schedule NSS Agreement, Dated December 6, 2001 (Agreement No. 119945), as amended by Amendment No. 2, Dated September 30, 2008.
 - $50.62~\rm BP~Energy$ Company, Storage Rate Schedule NSS Agreement, Dated February 25, 2003 (Agreement No. 129664), as amended by Amendment No. 2, Dated September 30, 2008.

FERC Docket: RP10-787-000

First Revised Sheet No. 540A First Revised Sheet No. 540A

Superseding: Original Sheet No. 540A

- 50. Non-Conforming Agreements (con't)
 - $50.63\,$ Wisconsin Electric Power Company, Storage Rate Schedule DSS Agreement, Dated June 12, 2009 (Agreement No. 137038).
 - $50.64\,$ Wisconsin Electric Power Company, Storage Rate Schedule NSS Agreement, Dated June 12, 2009 (Ageement No. 137043).
 - $50.65\,$ J.P. Morgan Ventures Energy Corporation, Storage Rate Schedule NSS Agreement, Dated November 14, 2007 (Agreement No. 134932), as amended by Amendment No. 1, Dated February 3, 2009.

FERC Docket: RP10-787-000

First Revised Sheet No. 540B First Revised Sheet No. 540B Superseding: Sheet Nos. 540B Through 540F

Sheet Nos. 540B through 540F are Reserved for Future Use.

FERC Docket: RP10-530-000

Original Sheet No. 540D Original Sheet No. 540D

GENERAL TERMS AND CONDITIONS

50. Non-Conforming Agreements (con't)

 $50.62\,$ BP Energy Company, Storage Rate Schedule NSS Agreement, Dated February 25, 2003 (Agreement No. 129664), as amended by Amendment No. 2, Dated September 30, 2008.

FERC Docket: RP10-537-000

Original Sheet No. 540E Original Sheet No. 540E

GENERAL TERMS AND CONDITIONS

50. Non-Conforming Agreements (con't)

50.63 Wisconsin Electric Power Company, Storage Rate Schedule DSS Agreement, Dated June 12, 2009 (Agreement No. 137038).

 $50.64\,$ Wisconsin Electric Power Company, Storage Rate Schedule NSS Agreement, Dated June 12, 2009 (Agreement No. 137043).

FERC Docket: RP10-532-000

Original Sheet No. 540F Original Sheet No. 540F

GENERAL TERMS AND CONDITIONS

50. Non-Conforming Agreements (con't)

 $50.65\,$ J.P. Morgan Ventures Energy Corporation, Storage Rate Schedule NSS Agreement, Dated November 14, 2007 (Agreement No. 134932), as amended by Amendment No. 1, Dated February 3, 2009.

Effective Date: 04/18/2008 Status: Effective

FERC Docket: RP08-319-000

Original Sheet No. 541 Original Sheet No. 541: Pending

GENERAL TERMS AND CONDITIONS

51. ACQUIRED CAPACITY

- (a) Natural may from time to time enter into transportation or storage agreements with upstream or downstream entities, including other interstate pipelines, intrastate pipelines, or local distribution companies (Acquired Capacity). Natural may use Acquired Capacity for its system operational needs and to render service to its customers. Except as provided in subsection (b), Natural states that, if it transports or stores gas for others using Acquired Capacity, it will apply to such services the same rates and tariffs as are applicable to onsystem customers, as such rates and tariffs may change from time to time. For purposes of any use of Acquired Capacity covered by this Section 51(a), the "shipper must hold title" requirement is waived.
- (b) Nothing herein shall be read to preclude Natural from filing with the Commission for different tariff provisions applicable to any service which Natural provides using Acquired Capacity; provided, however, that the waiver of the "shipper must hold title" requirement hereunder shall not apply in such a circumstance and Natural will be required to seek a case-specific waiver of that requirement from the Commission.

Effective Date: 04/18/2008 Status: Effective FERC Docket: RP08-319-000

Sheet Nos. 542 - 599 Sheet Nos. 542 - 599 : Pending

Sheet Nos. 542 through 599 are reserved for future use.

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 600 Substitute First Revised Sheet No. 600 Superseding: Original Sheet No. 600

			[FOR RATE SCHEDULES FTS, FFTS, FTS-G, ITS] Contract No
			NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL) TRANSPORTATION RATE SCHEDULE
1	1.	SHIPPE	ER is:, a
2	2.	(a) 1	MDQ totals: Dth per day.
		(b) S	Service option selected (FTS and FFTS only) (check any or all):
			LN SW NB
		(C)	Available Days (FFTS only)
			Minimum Available Days Maximum Available Days Applicable Time Period(s)
		V	Will Shipper accept partial Available Days? Yes No (Specify any limitations on partial Available Days)
• • • • •		• • • • • •	
			Are any days precluded from being Available Days? Yes No If so, specify the days so precluded
	3.		through
Terms			onditions of Natural's FERC Gas Tariff.]
4	4.		ce will be ON BEHALF OF: . Shipper or . Other: a
	5.		LTIMATE END USERS are (check one):
			customers of the following LDC/pipeline company(ies):; customers in these states:; or customers within any state in the continental U.S.
(6.		This Agreement supersedes and cancels a Agreement dated
			(a), and (b) the date capacity to provide the service hereunder is available on Natural's system. Other:
-	7.	SHIPPE	ER'S ADDRESSES NATURAL'S ADDRESSES
			all of the following provisions may be included (where applicable) in maximum rate
agreer			negotiated rate or discount contracts, if any:
		a. (I	DISCOUNTED RATE AGREEMENTS ONLY) Applicable Maximum and Minimum Rates. Notwithstanding
any		ot	ther provision of this Agreement, in no event shall a discounted rate billed by Natural be
less			nan the applicable minimum rate or more than the applicable maximum rate set forth in
Natura	a⊥'		ERC Gas Tariff, as may be revised from time to time.

b. expressly to minimum

(NEGOTIATED RATE AGREEMENTS ONLY) Maximum and Minimum Tariff Rates. Unless otherwise provided in this Agreement, the negotiated rates shall apply to service provided by Natural Shipper for the term of the Agreement notwithstanding any otherwise applicable maximum or rates set forth in Natural's FERC Gas Tariff as revised from time to time.

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 601 Substitute First Revised Sheet No. 601

Superseding: Original Sheet No. 601

[FOR RATE SCHEDULES FTS, FFTS, FTS-G, ITS]

Contract No.

NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL)
TRANSPORTATION RATE SCHEDULE
AGREEMENT DATED OF PART 284
OF THE FERC'S REGULATIONS
(CON'T)

c. (DISCOUNTED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund

to

Shipper any amounts collected for service to which the discounted rate(s) apply, unless the relevant discounted rate billed to Shipper exceeds the corresponding applicable effective maximum rates set forth in Natural's FERC Gas Tariff, as approved by the FERC from time to time.

d. (NEGOTIATED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund to

Shipper any amounts collected for service to which the negotiated rates apply, notwithstanding

any otherwise applicable maximum or minimum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

e. Notifications. Except as otherwise may be expressly provided herein, any notice or communication contemplated or required by this Agreement shall be in writing unless oral notification is expressly authorized herein, and shall be sent to the appropriate party at

the

relevant address set forth in the Transportation Agreement, as may be revised from time to time.

f. Nonwaiver of Rights. No delay or failure to exercise any right or remedy accruing to either Natural or Shipper upon breach or default by the other will impair any right or remedy or be construed to be a waiver of any such breach or default, not will a waiver of any single

breach

be deemed a waiver of any other breach or default.

g. Succession and Assignment. Any entity which shall succeed by purchase, merger or consolidation $% \left(1\right) =\left(1\right) +\left(1\right) +$

to title to the properties, substantially as an entirety, of Natural or Shipper as the case may be, shall be entitled to the rights and shall be subject to the obligations of its predecessor in title under this Agreement. No other assignment of this Agreement nor of any of the individual rights or obligations hereunder by Shipper shall be effective as to

Natural

without the prior express written consent of Natural.

h. No Third Party Beneficiaries. This Agreement shall not create any rights in any third parties,

and no provision of this Agreement shall be construed as creating any obligations for the benefit of, or rights in favor of, any person or entity other than Natural or Shipper.

i. Conformance to Law. It is understood that performance hereunder shall be subject to all valid

laws, orders, rules and regulations of duly constituted governmental authorities having jurisdiction or control of the matters related hereto, including without limitation the Federal Energy Regulatory Commission.

- j. Effect of Tariff. This Agreement shall at all times be subject to all applicable provisions of Natural's FERC Gas Tariff.
- k. GOVERNING LAW. THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT OF THIS AGREEMENT SHALL BE GOVERNED BY THE LAWS OF THE STATE OF EXCLUDING ANY CONFLICT OF LAW RULE WHICH WOULD REFER ANY MATTER TO THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF
- Entire Agreement. This Agreement contains the entire agreement between Natural and Shipper with respect to the subject matter hereof, and supersedes any and all prior understandings and

agreements, whether oral or written, concerning the subject matter hereof, and any and all such prior understandings and agreements are hereby deemed to be void and of no effect. No

amendments to or modifications of this Agreement shall be effective unless agreed upon in a written instrument executed by Natural and Shipper which expressly refers to this Agreement.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002 Substitute First Revised Sheet No. 602 Substitute First Revised Sheet No. 602 Superseding: Original Sheet No. 602 [FOR RATE SCHEDULES FTS, FFTS, FTS-G, ITS] Contract No. NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL) TRANSPORTATION RATE SCHEDULE AGREEMENT DATED OF PART 284
OF THE FERC'S REGULATIONS (CON'T) The above-stated Rate Schedule, as revised from time to time, controls this Agreement and is incorporated herein. The attached Exhibits A, B and C (for firm service only), and D and E (if applicable), are a part of this Agreement. Shipper shall provide the actual end user purchaser 9. name(s) to Natural if Natural must provide them to the FERC. Agreed to by: NATURAL SHIPPER /s/: /s/: NAME: TITLE: NAME:

TITLE:

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 603 Substitute First Revised Sheet No. 603 Superseding: Original Sheet No. 603

EXHIBIT A DATED

	BIIBOIIVE				
Shipper:					
Contract No.:					
Natural Receipt Point(s):					
Name / Location	County/Parish/Area	State	PIN No.	Zone	MDQ (Dth)
PRIMARY RECEIPT POINT(S):					
SECONDARY RECEIPT POINT(S):					
			•••••		
Receipt Pressure, Assumed Atmos	spheric Pressure				
Shipper shall not deliver gas at a pres for each Receipt Point in Natural's an assumed atmospheric pressure co Rates	s Catalog of Points. Th	e measurin	g party shal	l use or cause to	
Except as otherwise prov	ided below or in any wri	tten agree	ement(s) betweent	een the parties i	n effect
during the term hereof, Shipper shall as specified in Natural's applical rate					_
other than the applicable maximinimum	num rate so long as such	rate is b	etween the a	oplicable maximum	and
rates specified for such service discounted rate will apply only to certain discounted rate will apply only that a	n volumes under the agre	ement. The	parties may	agree that a spe	cified
specified discounted rate will not	apply only if specified	l volumes a	re achieved	or only if the vo	lumes do
exceed a specified level; that the $\ $	a specified discounted	rate will	apply only d	uring specified p	eriods of
year or for a specifically define specified	ined period; that a spec	ified disc	counted rate	will apply only t	0
points, zones, mainline segment geographical	ts, supply areas, transp	ortation p	eaths, market	s or other define	d
area(s); that a specified disconactually	ounted rate(s) will appl	y in a spe	cified relat	ionship to the vo	lumes
transported (i.e., that the resactually transported); and/or to Natural's system. Notwithstanding the form	that the discount will a	pply only	to reserves	dedicated by Ship	per to
	J J	1		2	

certain volume level will be invalidated if the Shipper transports an incremental volume above that agreed

level. In addition, the discount agreement may include a provision that if one rate component which was at or

below the applicable maximum rate at the time the discount agreement was executed subsequently exceeds the

applicable maximum rate due to a change in Natural's maximum rates so that such rate component must be adjusted downward to equal the new applicable maximum rate, then other rate components may be adjusted upward

to achieve the agreed overall rate, so long as none of the resulting rate components exceed the maximum rate

applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein

shall be construed to alter a refund obligation under applicable law for any period during which rates which

had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. If the parties agree upon a rate other than the applicable maximum rate, such written Agreement

shall specify that the parties mutually agree either: (1) that the agreed rate is a discount rate; or (2)

that the agreed rate is a Negotiated Rate (or Negotiated Rate Formula). In the event that the parties agree

upon a Negotiated Rate or Negotiated Rate Formula, this Agreement shall be subject to Section 49 of the General Terms and Conditions of Natural's Tariff.

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 604 Substitute First Revised Sheet No. 604

Superseding: Original Sheet No. 604

EXHIBIT A (CON'T)
DATED
EFFECTIVE

Fuel Gas and Gas Lost and Unaccounted For Percentage (%)

Shipper will be assessed the applicable percentage for Fuel Gas and Gas Lost and Unaccounted For unless Natural and Shipper mutually agree on monetary reimbursement.

Transportation of Liquids

Transportation of liquids may occur at permitted points identified in Natural's current Catalog of Points, but only if the parties execute a separate liquids agreement.

Delivery Pressure, Assumed Atmospheric Pressure

Effective Date: 11/09/2009 Status: Effective

FERC Docket: RP09-855-002

Natural gas to be delivered by Natural to Shipper, or for Shipper's account, at the Delivery Point(s) shall be at the pressures available in Natural's pipeline facilities from time to time; provided, however, that the delivery pressure shall not be less than party shall use or cause to be used an assumed atmospheric pressure corresponding to the elevation at such Delivery Point(s).

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 606 Substitute First Revised Sheet No. 606

Superseding: Original Sheet No. 606

EXHIBIT C
DATED
EFFECTIVE

PRIMARY TRANSPORTATION PATH SEGMENT MDQs

Pursuant to Natural's Tariff, an MDQ exists for each primary transportation path segment and direction under the Agreement. Such MDQ is the maximum daily quantity of gas which Natural is obligated to transport on a firm basis along a primary transportation path segment.

A primary transportation path segment is the path between a primary receipt, delivery or node point and the next primary receipt, delivery or node point. A node point is the point of interconnection between two or more of Natural's pipeline facilities.

A segment is a section of Natural's pipeline system designated by a segment number whereby the Shipper under the terms of their agreement based on the points within the segment identified on Exhibit C have throughput capacity rights.

The segment numbers listed on Exhibit C reflect this Agreement's path corresponding to Natural's most recent Pipeline System Map which identifies segments and their corresponding numbers. All information provided in this Exhibit C is subject to the actual terms and conditions of Natural's Tariff.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002 Substitute First Revised Sheet No. 607 Substitute First Revised Sheet No. 607 Superseding: Original Sheet No. 607
EXHIBIT C DATED EFFECTIVE
Shipper:
Contract No.:
PRIMARY TRANSPORTATION PATH SEGMENT MDOS

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Substitute Original Sheet No. 607A

Substit	ute Origin	1al Sneet No. 60/A Substitute Or				
		EXHIBIT D	[Negotiated Rate Agreement - FT: DATED	S]		
		EFFE	CTIVE			
COILCE	acc No		ARTICLE 1 GOTIATED RATE PARAMETERS			
1.1	Negotiat	ed Rate Term: FromThr	ough			
1.2	Negotiat	ed Rates:				
rate or	(a)	Negotiated Monthly Base Reserv	ation Rate: [\$/Dth.	of MDQ, or the negotiated		
race or		negotiated rate formula]				
rate	(b)	Negotiated Base Commodity Rate formula]	: [\$ /Dth., or the new	gotiated rate or negotiated		
1.3	Eligible	Firm Transportation Quantity:	Dth./day.			
1.4	_	-	t forth in Exhibit A, unless ot	herwise listed herein]		
		NAME	PIN 	ELIGIBLE POINT MDQ (Dth/d.)		
1.5	basis fr	Secondary Receipt Points: The om the following secondary rece	Negotiated Rates shall apply to	service provided on a firm		
	nereinj	NAME	PIN 	ELIGIBLE POINT MDQ (Dth/d.)		
1.7 firm	_	Secondary Delivery Points: The the following secondary delive	Negotiated Rates shall apply to			
1.8	1.8 Incremental Rate Secondary Receipt/Delivery Points: [Specify Shipper's obligation to pay an incremental volumetric reservation rate at the secondary receipt and delivery points and the					
	associat		at the secondary receipt and de			
1.9 Service	Authoriz	tal volumetric reservation rate ed rates set forth below, if an	at the secondary receipt and de	elivery points and the		
	Authoriz	tal volumetric reservation rate ed rates set forth below, if an	at the secondary receipt and degrees.	elivery points and the		
Service	Authoriz 'if any.] Other Ra	tal volumetric reservation rate ed rates set forth below, if an ed Overrun Rate(s): [Describe te Related Provisions: [Descri	at the secondary receipt and degrees.	elivery points and the for Authorized Overrun sion authorized by the		
Service	Authoriz , if any.] Other Ra applicab	tal volumetric reservation rate ed rates set forth below, if an ed Overrun Rate(s): [Describe te Related Provisions: [Descri	at the secondary receipt and do y.] the applicable negotiated rate be any other rate-related provi- 's list of permissible discount	elivery points and the for Authorized Overrun sion authorized by the		
Service	Authoriz , if any.] Other Ra applicab	tal volumetric reservation rate ed rates set forth below, if an ed Overrun Rate(s): [Describe te Related Provisions: [Describe Rate Schedule and/or Natural e covered by the Negotiated Rat	at the secondary receipt and do y.] the applicable negotiated rate be any other rate-related provi- 's list of permissible discount	elivery points and the for Authorized Overrun sion authorized by the provisions and which are		
Service	Authoriz if any.] Other Ra applicab otherwis General d to	tal volumetric reservation rate ed rates set forth below, if an ed Overrun Rate(s): [Describe te Related Provisions: [Describe Rate Schedule and/or Natural e covered by the Negotiated Rat NEGOTIATED RATE Negotiated Rate Limitations: Th	at the secondary receipt and dey.] the applicable negotiated rate be any other rate-related provi. 's list of permissible discount e Agreement.] ARTICLE 2 TERMS AND CONDITIONS; ADDITION.	elivery points and the for Authorized Overrun sion authorized by the provisions and which are AL CHARGES only to: (i) service		
1.10 not	Authoriz if any.] Other Ra applicab otherwis General d to Shipper e	tal volumetric reservation rate ed rates set forth below, if an ed Overrun Rate(s): [Describe te Related Provisions: [Describe Rate Schedule and/or Natural e covered by the Negotiated Rat NEGOTIATED RATE Negotiated Rate Limitations: The Description of the Natural under the Transporta	at the secondary receipt and dry.] the applicable negotiated rate the applicable negotiated rate to be any other rate-related proving list of permissible discount e Agreement.] ARTICLE 2 TERMS AND CONDITIONS; ADDITIONS e Negotiated Rates shall apply the agreement from the Eligible	elivery points and the for Authorized Overrun sion authorized by the provisions and which are AL CHARGES only to: (i) service e Receipt Points to the		
1.10 not 2.1 provide Eligibl	Authoriz if any.] Other Ra applicab otherwis General d to Shipper e Delivery rtation	tal volumetric reservation rate ed rates set forth below, if an ed Overrun Rate(s): [Describe te Related Provisions: [Describe te Rate Schedule and/or Natural e covered by the Negotiated Rat NEGOTIATED RATE Negotiated Rate Limitations: The by Natural under the Transporta Points during the Negotiated R	at the secondary receipt and dry.] the applicable negotiated rate be any other rate-related provious list of permissible discount e Agreement.] ARTICLE 2 TERMS AND CONDITIONS; ADDITIONS e Negotiated Rates shall apply the state of the secondary and the second	elivery points and the for Authorized Overrun sion authorized by the provisions and which are AL CHARGES only to: (i) service e Receipt Points to the maximum daily firm		
1.10 not 2.1 provide Eligibl	Authoriz if any.] Other Ra applicab otherwis General d to Shipper e Delivery rtation quantity	tal volumetric reservation rate ed rates set forth below, if an ed Overrun Rate(s): [Describe te Related Provisions: [Describe Rate Schedule and/or Natural e covered by the Negotiated Rat NEGOTIATED RATE Negotiated Rate Limitations: The by Natural under the Transporta Points during the Negotiated R equal to the Eligible Firm Tra	at the secondary receipt and dry.] the applicable negotiated rate the applicable negotiated rate to be any other rate-related proving list of permissible discount e Agreement.] ARTICLE 2 TERMS AND CONDITIONS; ADDITIONS e Negotiated Rates shall apply the agreement from the Eligible	for Authorized Overrun sion authorized by the provisions and which are AL CHARGES only to: (i) service e Receipt Points to the maximum daily firm in Section 1.3 above, for		

release replacement agreements. For any aggregate quantities transported on a firm basis on any day for Shipper and any associated capacity release replacement shippers which: (i) are in excess of the applicable Eligible Firm Transportation Quantity; or (ii) involve any receipt or delivery points

FERC Docket: RP09-855-002

Original Sheet No. 607B Original Sheet No. 607B

EXHIBIT D [Negotiated Rate Agreement - FTS]

DATED

EFFECTIVE

(CON'T)

which are not Eligible Receipt or Delivery Points, Shipper shall be charged the greater of: (a)

the

100% load factor daily rate equivalent of the Negotiated Rates; and (b) the 100% load factor v

daily

rate equivalent of the applicable maximum base reservation and base commodity rates set forth in Natural's FERC Gas Tariff, as may be revised from time to time, as well as all applicable additional

rates, charges and surcharges described in Section 2.3 below.

- 2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural is:
 - (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges as part of the negotiated rates.]
- 2.3 Additional Rates, Charges, and Surcharges: In addition to the Negotiated Rates, and unless otherwise expressly provided in this Agreement or agreed to in writing by Natural, Shipper shall also pay Natural all other applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to time, including without limitation all applicable maximum: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation charges and surcharges; (iv) commodity charges and surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges.
 - and lateral line charges. Notwithstanding the foregoing sentence, to the extent that any such additional maximum rate, charge, surcharge, or penalty is derived from the applicable base reservation rate or applicable base commodity rate set forth in Natural's FERC Gas Tariff, then Natural shall derive such additional rate, charge, surcharge, or penalty utilizing the greater
- of:
 (i) the applicable maximum base reservation rate or applicable maximum base commodity rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time; and (ii) the Negotiated Rates.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 607C Original Sh

Origina	Al Sheet No. 607C Original Sheet No. 607	7C	
		IT D [Discount Agreement - FTS] DATED FECTIVE	
	FE	EECTIVE	
	per: cact No.:		
		ARTICLE 1 DISCOUNT PARAMETERS	
1.1	Discount Term: FromThrough		
1.2	Discounted Rates:		
	(a) Discounted Monthly Base Reser (b) Discounted Base Commodity Rat	rvation Rate: \$ /Dth. of te: \$ /Dth.	MDQ.
1.3	Discounted Firm Transportation Quantit	ty: Dth./day.	
1.4	Discounted Primary Receipt Points: [as	s set forth in Exhibit A, unless o	therwise listed herein]
	NAME 	PIN 	DISCOUNT POINT MDQ (Dth/d.)
1.5 firm	Discounted Secondary Receipt Points: 1	The Discounted Rates shall apply t	o service provided on a
	basis from the following secondary rec	= =	
1.6	Discounted Primary Delivery Points: [a		otherwise listed herein]
	NAME 	PIN 	DISCOUNT POINT MDQ (Dth/d.)
1.7 firm	Discounted Secondary Delivery Points:	The Discounted Rates shall apply	to service provided on a
	basis to the following secondary deliv		
1.8	Incremental Rate Secondary Receipt/Delincremental volumetric reservation rat associated rates set forth below, if a	te at the secondary receipt and de	
1.9 Service	Authorized Overrun Rate(s): [Describe e, if any.]	e the applicable discount rate for	Authorized Overrun
	Other Rate Related Provisions: [Descrapplicable Rate Schedule and/or Natura		
not	otherwise covered by the Discount Agre	eement.	
	DISCOUNT TE	ARTICLE 2 ERMS AND CONDITIONS; APPLICABLE CH	ARGES
2.1	General Discount Limitations: The Disc Shipper by Natural under the Transport Discounted Delivery Points during the transportation quantity equal to the D	tation Agreement from the Discount Discount Term; and (ii) an aggreg	ed Receipt Points to the ate maximum daily firm
	above, for all quantities transported associated capacity release replacement charged all applicable maximum rates,	nt agreements. Unless otherwise a	greed, Shipper shall be
Tariff,	as may be revised from time to time, f Shipper and any associated capacity re applicable Discounted Firm Transportat	elease replacement shippers which:	(i) are in excess of the
points			

which are not Discounted Receipt or Delivery Points.

FERC Docket: RP09-855-002

Original Sheet No. 607D Original Sheet No. 607D

EXHIBIT D [Discount Agreement - FTS]
DATED
EFFECTIVE
(CON'T)

2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural

is:

- (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges to Shipper.]
- 2.3 Applicable Maximum Rates, Charges, and Surcharges: Unless otherwise expressly provided in this Agreement or agreed to in writing by Natural, all applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time

to

time, shall apply to service provided to Shipper under the Transportation Agreement, including without limitation all applicable: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 607E Original Sheet No. 607E

		EXHIBI	T D [Negotiated Rate Agre	eement - FF	TS]			
			EFFECTIVE					
Contr	act No.:		ARTICLE 1 NEGOTIATED RATE PARAME	ETERS				
1.1	Negotiat	ed Rate Term: From	Through					
1.2	Negotiat	ed Rates:						
rate or	(a)	Negotiated Daily Base Res	ervation Rate: [\$	/Dth. o	of MDQ, or the negoti	lated		
	(b)	negotiated rate formula] Negotiated Base Commodity	Rate: [\$/Dth.	., or the n	negotiated rate or ne	egotiated		
rate		formula]						
1.3	Eligible	Firm Transportation Quant	ity: Dth./day.					
1.4	Eligible	Primary Receipt Points: [as set forth in Exhibit A	A, unless o	therwise listed her	rein]		
		NAME 		PIN	ELIGIBLE POI MDQ (Dth/c	d.)		
1.5		e Secondary Receipt Points: com the following secondary						
1.6	Eligible herein]	Primary Delivery Points:	[as set forth in Exhibit	B, unless	otherwise listed			
		NAME 		PIN 	ELIGIBLE POI MDQ (Dth/c			
1.7	Eligible	Secondary Delivery Points	: The Negotiated Rates sh	nall apply	to service provided	on a		
firm	basis to	basis to the following secondary delivery points:						
1.8	incremen	ital Rate Secondary Receipt ital volumetric reservation ed rates set forth below,	rate at the secondary re					
1.9		ed Overrun Rate(s): [Descif any.]	ribe the applicable negot	iated rate	e for Authorized Over	run		
1.10	applicab	tte Related Provisions: [Dele Rate Schedule and/or Natural Research	tural's list of permissib					
1.11	determin	e Days: [Describe the min led, any restrictions that litted under the FFTS Agree	certain days are not Avai					

FERC Docket: RP09-855-002

Original Sheet No. 607F Original Sheet No. 607F

ARTICLE 2

NEGOTIATED RATE TERMS AND CONDITIONS; ADDITIONAL CHARGES

2.1 General Negotiated Rate Limitations: The Negotiated Rates shall apply only to: (i) service provided

to Shipper by Natural under the Transportation Agreement from the Eligible Receipt Points to the Eligible Delivery Points during the Negotiated Rate Term; and (ii) an aggregate maximum daily

firm

transportation quantity equal to the Eligible Firm Transportation Quantity set forth in Section

1.3

above, for all quantities transported on a firm basis under the Transportation Agreement and any associated capacity release replacement agreements. For any aggregate quantities transported on

а

firm basis on any day for Shipper and any associated capacity release replacement shippers which: (i) are in excess of the applicable Eligible Firm Transportation Quantity; or (ii) involve any receipt or delivery points which are not Eligible Receipt or Delivery Points, Shipper shall be charged the greater of: (a) the 100% load factor daily rate equivalent of the Negotiated Rates; and (b) the 100% load factor daily rate equivalent of the applicable maximum base reservation and base commodity rates set forth in Natural's FERC Gas Tariff, as may be revised from time to time, as well as all applicable additional rates, charges and surcharges described in Section 2.3

below.

2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural

is:

- (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges as part of the negotiated rates.]
- 2.3 Additional Rates, Charges, and Surcharges: In addition to the Negotiated Rates, and unless otherwise expressly provided in this Agreement or agreed to in writing by Natural, Shipper shall also pay Natural all other applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to time, including without limitation all applicable maximum: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation charges and surcharges; (iv) commodity charges and surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore

charges,

and lateral line charges. Notwithstanding the foregoing sentence, to the extent that any such additional maximum rate, charge, surcharge, or penalty is derived from the applicable base reservation rate or applicable base commodity rate set forth in Natural's FERC Gas Tariff, then Natural shall derive such additional rate, charge, surcharge, or penalty utilizing the greater

of:

(i) the applicable maximum base reservation rate or applicable maximum base commodity rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time; and (ii) the Negotiated Rates.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 607G Original Sheet No. 607G

	DATED	int Agreement - FFTS]				
		CICLE 1 PARAMETERS				
1.1	Discount Term: FromThrough					
1.2	Discounted Rates:					
	(a) Discounted Daily Base Reservation Rate: (b) Discounted Base Commodity Rate: \$		Q.			
1.3	Discounted Firm Transportation Quantity:	Dth./day.				
1.4	Discounted Primary Receipt Points: [as set forth	ı in Exhibit A, unless ot	herwise listed herein]			
	NAME 	PIN 	DISCOUNT POINT MDQ (Dth/d.)			
1.5	Discounted Secondary Receipt Points: The Discounfirm basis from the following secondary receipt					
1.6	Discounted Primary Delivery Points: [as set fortherein]	th in Exhibit B, unless o	therwise listed			
	NAME 	PIN 	DISCOUNT POINT MDQ (Dth/d.)			
1.7 Eirm	Discounted Secondary Delivery Points: The Discounted Rates shall apply to service provided on a					
	basis to the following secondary delivery points	:				
1.8	Incremental Rate Secondary Receipt/Delivery Points: [Specify Shipper's obligation to pay an incremental volumetric reservation rate at the secondary receipt and delivery points and the associated rates set forth below, if any.]					
1.9	Authorized Overrun Rate(s): [Describe the appli Overrun Service, if any.]	cable discount or negoti	ated rate for Authorized			
1.10	Other Rate Related Provisions: [Describe any otapplicable Rate Schedule and/or Natural's list of					
101	otherwise covered by the Discount Agreement.]					
1.11 are	Available Days: [Describe the minimum and maxim determined, any restrictions that certain days a					
	permitted under the FFTS Agreement.]					

FERC Docket: RP09-855-002

Original Sheet No. 607H Original Sheet No. 607H

EXHIBIT D [Discount Agreement - FFTS]

DATED

EFFECTIVE

(CON'T)

ARTICLE 2 DISCOUNT TERMS AND CONDITIONS; APPLICABLE CHARGES

- 2.1 General Discount Limitations: The Discounted Rates shall apply only to: (i) service provided to Shipper by Natural under the Transportation Agreement from the Discounted Receipt Points to the Discounted Delivery Points during the Discount Term; and (ii) an aggregate maximum daily firm transportation quantity equal to the Discounted Firm Transportation Quantity set forth in Section 1.3
- above, for all quantities transported on a firm basis under the Transportation Agreement and any associated capacity release replacement agreements. Unless otherwise agreed, Shipper shall be charged all applicable maximum rates, charges, and surcharges set forth in Natural's FERC Gas
- Tariff,
 as may be revised from time to time, for any aggregate quantities transported on a firm basis for Shipper and any associated capacity release replacement shippers which: (i) are in excess of the applicable Discounted Firm Transportation Quantity; or (ii) involve any receipt or delivery points
- which are not Discounted Receipt or Delivery Points.
- 2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural is:
- (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges to Shipper.]
- 2.3 Applicable Maximum Rates, Charges, and Surcharges: Unless otherwise expressly provided in this Agreement or agreed to in writing by Natural, all applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time
- time, shall apply to service provided to Shipper under the Transportation Agreement, including without limitation all applicable: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 6071 Original Sheet No. 6071

Originio	in Onlock No.	0071		
	EXHIBI	T D [Negotiated Rate Agree DATED	ement - FTS-G]	
		EFFECTIVE	• • • • •	
	er:act No.:	ARTICLE 1 NEGOTIATED RATE PARAME:	TERS	
1.1	Negotiated Rate Term: From	Through		
1.2	Negotiated Base Commodity Rate: [\$	-		~ ~~~~+: ~+~d ~~+~
formula		/ Deli, or the ne	egotiated rate o	i negotiated rate
1.3	Eligible Firm Transportation Quant	ity: Dth./day.		
1.4	Eligible Primary Receipt Points: [as set forth in Exhibit A	, unless otherwi	se listed herein]
	NAME 		PIN	ELIGIBLE POINT MDQ (Dth/d.)
1.5	Eligible Secondary Receipt Points: provided on a firm basis from the			
1.6	Eligible Primary Delivery Points: herein]	[as set forth in Exhibit]	B, unless otherw	rise listed
	NAME		PIN 	ELIGIBLE POINT MDQ (Dth/d.)
1.7	Eligible Secondary Delivery Points provided on a firm basis to the fo			
1.8	Incremental Rate Secondary Receipt incremental volumetric reservation associated rates set forth below,	rate at the secondary red		
1.9	Authorized Overrun Rate(s): [Desc Service, if any.]	ribe the applicable negot:	iated rate for A	uthorized Overrun
1.10	Other Rate Related Provisions: [Dapplicable Rate Schedule and/or Na not otherwise covered by the Negot	tural's list of permissib		
	NEGOTIATED	ARTICLE 2 RATE TERMS AND CONDITIONS	S; ADDITIONAL CH	ARGES
2.1	General Negotiated Rate Limitation service provided to Shipper by Nat			
Receipt	Points to the Eligible Delivery Po	ints during the Negotiated	d Rate Term; and	(ii) an aggregate
maximum	daily firm transportation quantity	equal to the Eligible Fig	rm Transportatio	n Quantity set forth
in	Section 1.3 above, for all quantit	ies transported on a firm	basis under the	Transportation
Agreeme	For any aggregate quantities trans excess of the applicable Eligible delivery points which are not Elig greater of: (a) the Negotiated Ba	Firm Transportation Quant: ible Receipt or Delivery 1	ity; or (ii) inv Points, Shipper	olve any receipt or shall be charged the
	rates set forth in Natural's FERC applicable additional rates, charg			

FERC Docket: RP09-855-002

Original Sheet No. 607J Original Sheet No. 607J

EXHIBIT D [Negotiated Rate Agreement - FTS-G]
DATED
EFFECTIVE
(CON'T)

- 2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural is:
 - (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges as part of the negotiated rates.]
 - 2.3 Additional Rates, Charges, and Surcharges: In addition to the Negotiated Base Commodity Rate, and unless otherwise expressly provided in this Agreement or agreed to in writing by Natural, Shipper shall also pay Natural all other applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to time,
- including without limitation all applicable maximum: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation charges and surcharges; (iv) commodity charges and surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore
- charges,
 and lateral line charges. Notwithstanding the foregoing sentence, to the extent that any such
 additional maximum rate, charge, surcharge, or penalty is derived from the applicable base
 reservation rate or applicable base commodity rate set forth in Natural's FERC Gas Tariff, then
 Natural shall derive such additional rate, charge, surcharge, or penalty utilizing the greater
- of:
 (i) the applicable maximum base reservation rate or applicable maximum base commodity rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time; and (ii) the Negotiated Base Commodity Rate.

FERC Docket: RP09-855-002

Origina	Sheet No. 607K Original Sheet No. 607K					
	DAT	Discount Agreement - FTS-G] TED VE				
	er:act No.:	ARTICLE 1 SCOUNT PARAMETERS				
1.1	Discount Term: FromThrough					
1.2	Discounted Base Commodity Rate: \$	/Dth.				
1.3	Discounted Firm Transportation Quantity:	Dth./day.				
1.4	Discounted Primary Receipt Points: [as set	forth in Exhibit A, unless otherw	ise listed herein]			
	NAME 	PIN 	DISCOUNT POINT MDQ (Dth/d.)			
1.5	Discounted Secondary Receipt Points: The Diprovided on a firm basis from the following					
1.6	Discounted Primary Delivery Points: [as set herein]	forth in Exhibit B, unless other	wise listed			
	NAME 	PIN 	DISCOUNT POINT MDQ (Dth/d.)			
1.7	Discounted Secondary Delivery Points: The I provided on a firm basis to the following s		ll apply to service			
	• • • • • • • • • • • • • • • • • • • •					
1.8	Incremental Rate Secondary Receipt/Delivery incremental volumetric reservation rate at associated rates set forth below, if any.]					
1.9	Authorized Overrun Rate(s): [Describe the	applicable discounted rates for A	uthorized Overrun			
Service	if any.]					
1.10	Other Rate Related Provisions: [Describe a applicable Rate Schedule and/or Natural's 1					
1100	otherwise covered by the Discount Agreement.]					
	DISCOUNT TERMS A	ARTICLE 2 AND CONDITIONS; APPLICABLE CHARGES				
2.1	General Discount Limitations: The Discounter provided to Shipper by Natural under the Tr					
Points	to the Discounted Delivery Points during the Discount Term; and (ii) an aggregate maximum daily					
firm	transportation quantity equal to the Discou	unted Firm Transportation Quantity	set forth in Section			
1.3	above, for all quantities transported on a otherwise agreed, Shipper shall be charged					
set .	forth in Natural's FERC Gas Tariff, as may	be revised from time to time, for	any aggregate			
quantit	ies transported on a firm basis for Shipper whi Transportation Quantity; or (ii) involve ar Receipt or Delivery Points.	.ch: (i) are in excess of the Disc ny receipt or delivery points whic	ounted Firm h are not Discounted			

FERC Docket: RP09-855-002

Original Sheet No. 607L Original Sheet No. 607L

EXHIBIT D [Discounted Agreement - FTS-G]
DATED
EFFECTIVE
(CON'T)

- 2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural
- is:
 (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges to Shipper.]
- 2.3 Applicable Maximum Rates, Charges, and Surcharges: Unless otherwise expressly provided in this Agreement or agreed to in writing by Natural, all applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time
- time, shall apply to service provided to Shipper under the Transportation Agreement, including without limitation all applicable: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 607M Original Sheet No. 607M

_						
		iated Rate Agreement - ITS]				
		•••••				
		ARTICLE 1 ED RATE PARAMETERS				
1.1	Negotiated Rate Term: FromThrough					
1.2 formula	Negotiated Base Commodity Rate: [\$		or negotiated rate			
1.3	Eligible Transportation Quantity:	Dth./day.				
1.4	Eligible Receipt Points:	-				
	NAME 	PIN 	ELIGIBLE POINT MDQ (Dth/d.)			
1.5	Eligible Delivery Points:					
	NAME 	PIN 	ELIGIBLE POINT MDQ (Dth/d.)			
1.6	Authorized Overrun Rate(s): [Describe the ap Service, if any.]	plicable negotiated rate for	Authorized Overrun			
1.7	Other Rate Related Provisions: [Describe any applicable Rate Schedule and/or Natural's lis					
not	otherwise covered by the Negotiated Rate Agre	ement.]				
		ARTICLE 2 AND CONDITIONS; ADDITIONAL	CHARGES			
2.1	General Negotiated Rate Limitations: The Nego service provided to Shipper by Natural under					
Receipt	Points to the Eligible Delivery Points during	the Negotiated Rate Term; a	nd (ii) an aggregate			
1.3	daily transportation quantity equal to the El	igible Transportation Quanti	ty set forth in Section			
	above, for all quantities transported under the Transportation Agreement. For any aggregate quantities transported on any day for Shipper which: (i) are in excess of the applicable Eligible Transportation Quantity; or (ii) involve any receipt or delivery points which are not Eligible					
Receipt	or Delivery Points, Shipper shall be charged	the greater of: (a) the Neg	otiated Base Commodity			
Rate;	and (b) the base commodity rates set forth in	Natural's FERC Gas Tariff,	as may be revised from			
time	to time, as well as all applicable additional	rates, charges and surcharg	es described in Section			
2.3	below.					
2.2	Discountable Third Party Surcharges: From time approved by the FERC for inclusion in Natural collect from Shipper and remit to the FERC or discount the amount of such surcharge from the	's FERC Gas Tariff which Nat to another third party; and	ural is: (i) required to (ii) permitted to			
FERC	Gas Tariff ("Discountable Third Party Surchar	ges"). [Describe any applic	able surcharges covered			
by	this Section and the applicability of such su	rcharges as part of the nego	tiated rates.]			
2.3	Additional Rates, Charges, and Surcharges: In unless otherwise expressly provided in this A shall also pay Natural all other applicable m any nature set forth in Natural's FERC Gas Ta	greement or agreed to in wri aximum rates, charges, surch	ting by Natural, Shipper arges, and penalties of			

including

without limitation all applicable maximum: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized overrun charges; (iii) commodity charges and surcharges; (iv)

ACA

surcharges; and (v) gathering charges, offshore charges, and lateral line charges. Notwithstanding

the foregoing sentence, to the extent that any such additional maximum rate, charge, surcharge,

or

penalty is derived from the applicable base reservation rate or applicable base commodity rate

forth in Natural's FERC Gas Tariff, then Natural shall derive such additional rate, charge, surcharge, or penalty utilizing the greater of: (i) the applicable maximum base reservation rate or applicable maximum base commodity rate set forth in Natural's FERC Gas Tariff, as may be revised

from time to time; and (ii) the Negotiated Base Commodity Rate.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 607N Original Sheet No. 607N

		D [Discount Agreement - ITS] DATED					
	er:act No.:	ARTICLE 1 DISCOUNT PARAMETERS					
1.1	Discount Term: FromThrough						
1.2	Discounted Base Commodity Rate: \$	/Dth.					
1.3	Discounted Transportation Quantity:	Dth./day.					
1.4	Discounted Receipt Points:						
	NAME 	PIN 	DISCOUNT POINT MDQ (Dth/d.)				
1.5	Discounted Delivery Points:						
	NAME 	PIN 	DISCOUNT POINT MDQ (Dth/d.)				
1.6 Service	,	the applicable discount rate for A	uthorized Overrun				
	if any.]						
1.7	Other Rate Related Provisions: [Describe any other rate-related provision authorized by the applicable Rate Schedule and/or Natural's list of permissible discount provisions and which are						
not	otherwise covered by the Discount Agreement.]						
	DISCOUNT TER	ARTICLE 2 RMS AND CONDITIONS; ADDITIONAL CHAR	GES				
2.1 Points	General Discount Limitations: The Discounted Base Commodity Rate shall apply only to: (i) service provided to Shipper by Natural under the Transportation Agreement from the Discounted Receipt						
	to the Discounted Delivery Points during the Discount Term; and (ii) an aggregate maximum daily transportation quantity equal to the Discounted Transportation Quantity set forth in Section 1.3 above, for all quantities transported on any day under the Transportation Agreement. Unless otherwise agreed, Shipper shall be charged all applicable maximum rates, charges, and surcharges						
set	forth in Natural's FERC Gas Tariff, as	may be revised from time to time,	for any aggregate				
quantit	ies transported on a daily basis for Shippe	er which: (i) are in excess of the	Discounted				
Transpo	rtation Quantity; or (ii) involve any receipt o	or delivery points which are not Di	scounted Receipt or				
Deliver	Y Points.						
2.2	Discountable Third Party Surcharges: Fr approved by the FERC for inclusion in N collect from Shipper and remit to the F discount the amount of such surcharge f	Natural's FERC Gas Tariff which Nat TERC or to another third party; and	ural is: (i) required to (ii) permitted to				
FERC	Gas Tariff ("Discountable Third Party S	Surcharges"). [Describe any applic	able surcharges covered				
by	this Section and the applicability of s	such surcharges to Shipper.]					
2.3	Applicable Maximum Rates, Charges, and Agreement or agreed to in writing by Na and penalties of any nature set forth i	tural, all applicable maximum rate	s, charges, surcharges,				
to	-	to Shipper under the Transportation					

without limitation all applicable: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized overrun charges; (iii) commodity rates, charges and surcharges; (iv) ACA surcharges; and (v) gathering charges, offshore charges, and lateral line charges.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 607O Original Sheet No. 6070

	T E [NB Service Op				
EFFECT	CIVE	• • • • •			
Shipper:					
Contract No.:					
FTS-NB Delivery Point(s)					
	County/				NB Service MDQ
Name / Location	Parish/Area	State	PIN No.	Zone	(Dth)
NGPL Storage Agreements Dedicated to FTS-NB Ser	rvice: 				
	• •				
Third Party Storage Provider/Point No.:					

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 608 Substitute First Revised Sheet No. 608 Superseding: Original Sheet No. 608

		Contract No
		NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL) RATE SCHEDULE IBS
		UNDER SUBPART OF PART 284 OF THE FERC'S REGULATIONS
	1.	SHIPPER is:, a
		(a) If Shipper is affiliated with Natural, check here
:	2.	Linked Transportation Agreement:
	3.	IBS MDQ: Dth per day.
		Identification of End-Use Facility or Facilities
		TERM: through
		and month to month thereafter, if applicable to coincide with the linked transportation
agree	men	:; provided, however, that this Agreement shall terminate if the linked transportation Agreement is
no		longer in effect for any reason and the parties have not entered into or designated a substitute
linke		transportation Agreement. Further, service hereunder shall be suspended if the linked
•	-	ation Agreement is released and the parties have not entered into or designated a substitute linked transportation Agreement; provided, however, that if the release is partial, service hereunder
shall	be	suspended only partially, to the extent of the partial release.
	6.	Service will be ON BEHALF OF (where applicable):
		Shipper or Other: , a
	7.	The ULTIMATE END USERS are (check one):
		customers of the following LDC/pipeline company(ies): ; customers in these states: ; or customers within any state in the continental U.S.
	8.	This Agreement supersedes and cancels a Agreement dated
	9.	SHIPPER'S ADDRESSES NATURAL'S ADDRESSES
	10.	Any or all of the following provisions may be included (where applicable) in maximum rate agreements or in negotiated rate or discount contracts, if any:
	ā	a. (DISCOUNTED RATE AGREEMENTS ONLY) Applicable Maximum and Minimum Rates. Notwithstanding
any		other provision of this Agreement, in no event shall a discounted rate billed by Natural be less than the applicable minimum rate or more than the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.
1.	ŀ	O. (NEGOTIATED RATE AGREEMENTS ONLY) Maximum and Minimum Tariff Rates. Unless otherwise expressly provided in this Agreement, the negotiated rates shall apply to service provided
рÀ		Natural to Shipper for the term of the Agreement notwithstanding any otherwise applicable maximum or minimum rates set forth in Natural's FERC Gas Tariff as revised from time to

time.

c. (DISCOUNTED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund

to

Shipper any amounts collected for service to which the discounted rate(s) apply, unless the relevant discounted rate billed to Shipper exceeds the corresponding applicable effective maximum rates set forth in Natural's FERC Gas Tariff, as approved by the FERC from time to time.

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 609 Substitute First Revised Sheet No. 609

Superseding: Original Sheet No. 609

[FOR RATE SCHEDULES IBS]

Contract No.

.

NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL)
RATE SCHEDULE IBS
AGREEMENT DATED OF PART 284
OF THE FERC'S REGULATIONS
(CON'T)

d. (NEGOTIATED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund

Shipper any amounts collected for service to which the negotiated rates apply, notwithstanding

any otherwise applicable maximum or minimum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

e. Notifications. Except as otherwise may be expressly provided herein, any notice or communication contemplated or required by this Agreement shall be in writing unless oral notification is expressly authorized herein, and shall be sent to the appropriate party at

relevant address set forth in this Agreement, as may be revised from time to time.

f. Nonwaiver of Rights. No delay or failure to exercise any right or remedy accruing to either Natural or Shipper upon breach or default by the other will impair any right or remedy or be construed to be a waiver of any such breach or default, nor will a waiver of any single

breach

the

be deemed a waiver of any other breach or default.

g. Succession and Assignment. Any entity which shall succeed by purchase, merger or consolidation ${\sf Succession}$

to title to the properties, substantially as an entirety, of Natural or Shipper as the case may be, shall be entitled to the rights and shall be subject to the obligations of its predecessor in title under this Agreement. No other assignment of this Agreement nor of any of the individual rights or obligations hereunder by Shipper shall be effective as to

Natural

without the prior express written consent of Natural.

h. No Third Party Beneficiaries. This Agreement shall not create any rights in any third parties,
and no provision of this Agreement shall be construed as creating any obligations for the benefit of, or rights in favor of, any person or entity other than Natural or Shipper.

i. Conformance to Law. It is understood that performance hereunder shall be subject to all valid
laws, orders, rules and regulations of duly constituted governmental authorities having jurisdiction or control of the matters related hereto, including without limitation the Federal Energy Regulatory Commission.

- j. Effect of Tariff. This Agreement shall at all times be subject to all applicable provisions of Natural's FERC Gas Tariff.
- k. GOVERNING LAW. THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT OF THIS AGREEMENT SHALL BE GOVERNED BY THE LAWS OF THE STATE OF, EXCLUDING ANY CONFLICT OF LAW RULE WHICH WOULD REFER ANY MATTER TO THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF
- Entire Agreement. This Agreement contains the entire agreement between Natural and Shipper with respect to the subject matter hereof, and supersedes any and all prior understandings

and

agreements, whether oral or written, concerning the subject matter hereof, and any and all such prior understandings and agreements are hereby deemed to be void and of no effect. No amendments to or modifications of this Agreement shall be effective unless agreed upon in a written instrument executed by Natural and Shipper which expressly refers to this Agreement.

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 610 Substitute First Revised Sheet No. 610 Superseding: Original Sheet No. 610

[FOR RATE SCHEDULES IBS]

			Contract No
		RATE SCHEDULE AGREEMENT DATED UNDER SUBPART OF THE FERC	MPANY OF AMERICA LLC (NATURAL) IBS OF PART 284 'S REGULATIONS N'T)
this	11.	The above-stated Rate Schedule, as revised from t incorporated herein. The attached Exhibit A, and	•
must.		user purchaser name(s) to Natural if Natural	
musc		provide them to the FERC.	
		Agreed to by:	
		NATURAL /s/:	SHIPPER /s/: NAME: TITLE:

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 611 Substitute First Revised Sheet No. 611 Superseding: Original Sheet No. 611

		EXHIB				
		DATED				
	Shipper:	EFFECTIVE		• • • • • • • • • • • • • • • • • • • •		
	Contract No.:					
	Delivery Point(s):					
	Delivery Pressure:					
		0 1 10 1 10			-	IBS MDQ
	Name / Location		State 	PIN No.	Zone 	(Dth)
moni	If any end-use facility is toring Agreement among Natural, Ship consistent with Section 3.2(c) place as to such an end-use,	per, Facility Operator of Rate Schedule IBS	and the F	Point Operator an IBS Monit	must be executed coring Agreement :	d ,
lawf	charges as specified in Natura	er shall pay Natural t	he applica	ble maximum 1	rate(s) and all of	cher
Ship	shall pay a rate other than th	ne applicable maximum	rate so lo	ong as such ra	ate is between the	e
appı	icable maximum and minimum rates specthat a specific discounted ratmay agree that a specified divolumes) under the agreement; are achieved or only if the volumes.	te will apply only to scounted rate will app that a specified disc	certain vo ly only to ounted rat	olumes under to specified vo se will apply	the agreement. The plumes (MDQ or cor only if specified	parties nmodity d volumes
rate	will apply only during specific discounted rate shall not apply cycle; that a specified discount agraphical area(s); and/or relationship to the volumes as in a specified relationship to discount agreement may provide invalidated if the Shipper tra	ly to Access Requests inted rate will apply that a specified discoctually transported (in volumes actually trate that an agreed disco	received a only to spunted rate .e., that nsported).	after a special ecified point (s) will apply the reservation a certain vol	Fied time in the ris, zones or other y in a specified on charge will be ling the foregoing the level will be	nomination defined adjusted g, no
the	discount agreement may include applicable maximum rate at the applicable maximum rate due to	e time the discount ag	reement wa	ıs executed sı	bsequently exceed	ds the
be	adjusted downward to equal the	e new applicable maxim	um rate, t	hen other rat	te components may	be
adju	sted upward to achieve the agreed of	overall rate, so long	as none of	the resultir	ng rate components	s exceed
the	maximum rate applicable to the prospectively, commencing with nothing contained herein shall	at rate component. Su n the date a Commissio	ch changes n order ac	to rate comp ccepts revised	conents shall be a	applied However,
any	period during which rates which ultimately are found to be just applicable maximum rate, such	st and reasonable. If	the parti	es agrée upor	a rate other tha	an the

either:

(1) that the agreed rate is a discount rate; or (2) that the agreed rate is a Negotiated Rate (or Negotiated Rate Formula). In the event that the parties agree upon a Negotiated Rate or Negotiated Rate Formula, this Agreement shall be subject to Section 49 of the General Terms and Conditions of Natural's Tariff.

FERC Docket: RP09-855-002

Substitute Original Sheet No. 611A Substitute Original Sheet No. 611A

Cabotit	ato original oriot to or in bubblicate originar t	JIICCC NO. OIIII				
		ted Rate Agreement - IBS]				
-1.1						
	er:et No.:					
		ICLE 1 RATE PARAMETERS				
1.1	Negotiated Rate Term: FromThrough					
1.2	Swing Up Negotiated Daily Access Rate: \$/Dth	Swing Down \$/Dth.				
1.3	Eligible Daily Access Quantity: Dth.	/day.				
1.4	Eligible Delivery Points: [as set forth in Exhib	bit A, unless otherwise la	isted herein]			
	NAME	PIN 	ELIGIBLE POINT MDQ (Dth/d.)			
1.5	Linked Transportation Agreement: [Describe the	Linked Agreement, MDQ and	d term.]			
1.6	Other Rate Related Provisions: [Describe any of applicable Rate Schedule and/or Natural's list of					
not	otherwise covered by the Negotiated Rate Agreement.]					
		ICLE 2 D CONDITIONS; ADDITIONAL (CHARGES			
2.1 aggrega	General Negotiated Rate Limitations: The Negotia	ated Daily Access Rate sha	all apply only to an			
1.3	maximum daily access quantity equal to the Elig	ible Daily Access Quantity	y set forth in Section			
reduced	above, based upon the absolute values of confirm Requests will not be netted against swing down in provided, however, that to the extent that the	Access Requests) at the E	ligible Delivery Points;			
to	under the provisions of Natural's FERC Gas Tariff, then effective as of the date of any such reduction, the Eligible Daily Access Quantity shall be reduced pro rata based on the ratio of t Eligible Daily Access Quantity to the IBS Agreement MDQ, as such ratio existed immediately prio					
Eligibl	the IBS Agreement MDQ reduction. Any balancing ${}^{\circ}$	service provided on any o	day in excess of the			
set	Daily Access Quantity shall be billed at all app	plicable maximum rates, cl	harges, and surcharges			
Access	forth in Natural's FERC Gas Tariff, as may be re	evised from time to time.	The Eligible Daily			
	Quantity shall only apply to balancing service provided by Natural to Shipper to the Designated Agreement.					
2.2 all	Additional Rates, Charges, and Surcharges: Unles	ss otherwise expressly pro	ovided in this Agreement			
FERC	applicable maximum rates, charges, surcharges,	and penalties of any natu	re set forth in Natural's			
the	Gas Tariff, as may be revised from time to time					
unautho	IBS Agreement, including without limitation all prized	applicable authorized over	errun charges,			

overrun charges, and balancing service charges.

FERC Docket: RP09-855-002

Original Sheet No. 611B Original Sheet No. 611B

EXHIBIT B [Negotiated Rate Agreement - IBS)
DATED
EFFECTIVE

2.3 Linked Transportation Agreement Charges: In addition to charges under the IBS Agreement, Shipper shall also be responsible for payment of any and all applicable rates, charges, surcharges, and Fuel Gas and Gas Lost and Unaccounted For under the Linked Transportation Agreement, in accordance

with Section 5.3 of Natural's Rate Schedule IBS and all other applicable provisions of Natural's FERC Gas Tariff, as may be revised from time to time.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 611C Original Sheet No. 611C

	EXHIBIT B [Discount DATED EFFECTIVE				
	r:ct No.:				
	ARTICLI DISCOUNT PAR				
1.1	Discount Term: FromThrough				
1.2	Swing Up Discounted Daily Access Rate: \$/Dth.	Swing Down \$/Dth.			
1.3	Discounted Daily Access Quantity: Dth./o	day.			
1.4	Discounted Delivery Points: [as set forth in Exhib.	it A, unless otherwise	listed herein]		
	NAME	PIN 	ELIGIBLE POINT MDQ (Dth/d.)		
1.5	Linked Transportation Agreement: [Described the L:	inked Agreement, MDQ a	nd term.]		
1.6	Other Rate Related Provisions: [Describe any other applicable Rate Schedule and/or Natural's list of p				
not	otherwise covered by the Discount Agreement.]				
	ARTICLE DISCOUNT TERMS AND CONDI	= = =	GES		
2.1	General Discount Limitations: The Discounted Daily maximum daily access quantity equal to the Discount 1.3 above, based upon the absolute values of confid	ted Daily Access Quant:	ity set forth in Section		
Access	Requests will not be netted against swing down Acce	ess Requests) at the D:	iscounted Delivery		
Points;	provided, however, that to the extent that the IBS Agreement MDQ is permanently or temporarily reduced under the provisions of Natural's FERC Gas Tariff, then effective as of the date of any such reduction, the Discounted Daily Access Quantity shall be reduced pro rata based on the ratio of the Discounted Daily Access Quantity to the IBS Agreement MDQ, as such ratio existed				
	prior to the IBS Agreement MDQ reduction. Any bala the Discounted Daily Access Quantity shall be bille				
and	surcharges set forth in Natural's FERC Gas Tariff, Discounted Daily Access Quantity shall only apply				
with	transportation service provided by Natural to Shipp	per to the Designated I	End Use Facility under		
the	Linked Transportation Agreement.				
2.2	Applicable Rates, Charges, and Surcharges: Unless all applicable maximum rates, charges, surcharges, Natural's FERC Gas Tariff, as may be revised from the	and penalties of any m	nature set forth in		
to	Shipper under the IBS Agreement, including without charges, unauthorized overrun charges, and balancing		able authorized overrun		

FERC Docket: RP09-855-002

Original Sheet No. 611D Original Sheet No. 611D

2.3 Linked Transportation Agreement Charges: In addition to charges under the IBS Agreement, Shipper shall also be responsible for payment of any and all applicable rates, charges, surcharges, and Fuel Gas and Gas Lost and Unaccounted For under the Linked Transportation Agreement, in accordance

with Section 5.3 of Natural's Rate Schedule IBS and all other applicable provisions of Natural's FERC Gas Tariff, as may be revised from time to time.

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 612 Substitute First Revised Sheet No. 612 Superseding: Original Sheet No. 612

		[FOR RATE SCHEDULES DSS, NSS, FRSS, BESS]
		Contract No NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL) STORAGE RATE SCHEDULE AGREEMENT DATED UNDER SUBPART OF PART 284 OF THE FERC'S REGULATIONS
1.	SHIP	PER is:, a
2.	(a)	MDQ totals: Dth per day.
	(b)	MAC totals (BESS only): Dth.
	(c)	MSWQ equals MDQ times: 10 or 20 (check one) [FRSS only]
	(d)	(NSS only) Leg Rights:% Amarillo;% Gulf Coast (sum must equal 100%).
ontion	(e)	(FRSS only) Service will be: (i) delivered storage; or (ii) storage-only
option under		(check one). If the storage-only option is selected, does Shipper desire no-notice rights
firm		Section 7.5(b) of Rate Schedule FRSS: Yes No (Check One). If yes, designate a
TIIII		transportation Agreement:
hereto	(f)	The primary Delivery Point(s) and associated MDQ(s) are contained in Exhibit A attached
the		and are a part of this Agreement (DSS only). The primary Delivery and Receipt Point(s) and
Exhibit		associated MDQs (defining a primary path for capacity release purposes) are contained on
		A attached hereto and are a part of this Agreement (delivered FRSS only).
3. Section		:
		of the General Terms and Conditions of Natural's FERC Gas Tariff.]
4.	Serv	ice will be ON BEHALF OF:
		Shipper or, a, a
5.		This Agreement supersedes and cancels a Agreement dated
		(a), and (b) the date capacity to provide the service hereunder is available on Natural's system. Other:
6.	SHIP	PER'S ADDRESSES NATURAL'S ADDRESSES
7.		or all of the following provisions may be included (where applicable) in maximum rate ements or in negotiated rate or discount contracts, if any:
any	a.	(DISCOUNTED RATE AGREEMENTS ONLY) Applicable Maximum and Minimum Rates. Notwithstanding
-		other provision of this Agreement, in no event shall a discounted rate billed by Natural be less than the applicable minimum rate or more than the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

(NEGOTIATED RATE AGREEMENTS ONLY) Maximum and Minimum Tariff Rates. Unless otherwise

expressly provided in this Agreement, the negotiated rates shall apply to service provided by

Natural to Shipper for the term of the Agreement notwithstanding any otherwise applicable maximum or minimum rates set forth in Natural's FERC Gas Tariff as revised from time to time.

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 613 Substitute First Revised Sheet No. 613

Superseding: Original Sheet No. 613

[FOR RATE SCHEDULES DSS, NSS, FRSS, BESS]

Contract No.

.

NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL)
STORAGE RATE SCHEDULE
AGREEMENT DATED OF PART 284
OF THE FERC'S REGULATIONS
(CON'T)

c. (DISCOUNTED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund

to Shipper any amounts collected for service to which the discounted rate(s) apply, unless the

relevant discounted rate billed to Shipper exceeds the corresponding applicable effective maximum rates set forth in Natural's FERC Gas Tariff, as approved by the FERC from time to time.

d. (NEGOTIATED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund

Shipper any amounts collected for service to which the negotiated rates apply, notwithstanding $% \left(\frac{1}{2}\right) =\frac{1}{2}\left(\frac{1}{2}\right) +\frac{1}{2}\left(\frac{1}{2}\right) +\frac$

any otherwise applicable maximum or minimum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

e. Notifications. Except as otherwise may be expressly provided herein, any notice or communication contemplated or required by this Agreement shall be in writing unless oral notification is expressly authorized herein, and shall be sent to the appropriate party at the

relevant address set forth in this Agreement, as may be revised from time to time.

f. Nonwaiver of Rights. No delay or failure to exercise any right or remedy accruing to either Natural or Shipper upon breach or default by the other will impair any right or remedy or be construed to be a waiver of any such breach or default, nor will a waiver of any single

breach

be deemed a waiver of any other breach or default.

 $\,$ g. Succession and Assignment. Any entity which shall succeed by purchase, merger or consolidation

to title to the properties, substantially as an entirety, of Natural or Shipper as the case may be, shall be entitled to the rights and shall be subject to the obligations of its predecessor in title under this Agreement. No other assignment of this Agreement nor of any of the individual rights or obligations hereunder by Shipper shall be effective as to

Natural

without the prior express written consent of Natural.

h. No Third Party Beneficiaries. This Agreement shall not create any rights in any third parties,
and no provision of this Agreement shall be construed as creating any obligations for the benefit of, or rights in favor of, any person or entity other than Natural or Shipper.

i. Conformance to Law. It is understood that performance hereunder shall be subject to all valid laws, orders, rules and regulations of duly constituted governmental authorities having jurisdiction or control of the matters related hereto, including without limitation the Federal Energy Regulatory Commission.

- j. Effect of Tariff. This Agreement shall at all times be subject to all applicable provisions of Natural's FERC Gas Tariff.
- k. GOVERNING LAW. THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT OF THIS AGREEMENT SHALL BE GOVERNED BY THE LAWS OF THE STATE OF, EXCLUDING ANY CONFLICT OF LAW RULE WHICH WOULD REFER ANY MATTER TO THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF
- 1. Entire Agreement. This Agreement contains the entire agreement between Natural and Shipper with respect to the subject matter hereof, and supersedes any and all prior understandings and

agreements, whether oral or written, concerning the subject matter hereof, and any and all

such prior understandings and agreements are hereby deemed to be void and of no effect. No amendments to or modifications of this Agreement shall be effective unless agreed upon in a written instrument executed by Natural and Shipper which expressly refers to this Agreement.

Substitu	cket: RP09-855-002 ute First Revised Sheet No. 614 Substitute First Reding: Original Sheet No. 614	vised Sheet No. 614			
	[FOR RATE SCHEDUL:	ES DSS, NSS, FRSS, BESS]			
		Con	tract No.		
	STORAGE RATE AGREEMENT DAT: UNDER SUBPART OF THE FER	OMPANY OF AMERICA LLC (NATURAL) SCHEDULE EDOF PART 284 C'S REGULATIONS ON'T)			
8.	The above-stated Rate Schedule, as revised from time to time, controls this Agreement and is incorporated herein. The attached Exhibit A, and Exhibit B (if applicable), is (are) a part of				
this	Agreement.				
	Agreed to by:				
	NATURAL	SHIPPER			
	/s/: NAME: TITLE:	/s/: NAME: TITLE:			

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 615 Substitute First Revised Sheet No. 615

Superseding: Original Sheet No. 615

			EXHIBI DATED EFFECTIVE				
Sh	ippe	r:					
Со	ntra	ct No.:					
Α.	Pri	mary Delivery Point(s):					
		Name / Location	County/Parish/Area	State	PIN No.	Delivery Point Pressure (DSS Only)	MDQ (Dth)
	1.						
В.	Pri:	mary Receipt Point(s) (De		y Relea	ıse):		
		Name / Location	County/Parish/Area	State	PIN No.	MDQ Zone (Dth)	
	1.						

C. Rates

Except as otherwise provided below or in any written agreement(s) between the parties in effect during the term hereof, Shipper shall pay Natural the applicable maximum rate(s) and all other lawful charges specified in Natural's applicable rate schedule. Shipper and Natural may agree that Shipper shall pay a rate other than the applicable maximum rate so long as such rate is between the applicable maximum and minimum rates specified for such service in the Tariff. Natural and Shipper may agree that a specific discounted rate will apply only to certain volumes under the agreement. The parties may agree that a specified discounted rate will apply only to specified volumes (MDQ, WQ, IQ or commodity volumes) under the agreement; that a specified discounted rate will apply only if specified volumes are achieved or the volumes do not exceed a specified level; that a specified discounted rate will apply only during specified periods of the year or for a specifically defined period; that a specified discounted rate will apply only to specified points, zones, mainline segments, supply areas,

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 616 Substitute First Revised Sheet No. 616

Superseding: Original Sheet No. 616

EXHIBIT A
DATED

EFFECTIVE

(CON'T)

transportation paths, markets or other defined geographical area(s); that a specified discounted rate(s) will apply in a specified relationship to the volumes actually tendered (i.e., that the reservation charge will be adjusted in a specified relationship to volumes actually tendered); and/or that the discount will apply only to reserves dedicated by Shippers to Natural's system. Notwithstanding the foregoing, no discount agreement may provide that an agreed discount as to a certain volume level will be invalidated if the Shipper transports an incremental volume above that agreed level. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable maximum rate at the time the discount agreement was executed subsequently exceeds the applicable maximum rate due to a change in Natural's maximum rates so that such rate component must be adjusted downward to equal the new applicable maximum rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the maximum rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. If the parties agree upon a rate other than the applicable maximum rate, such written Agreement(s) shall specify that the parties mutually agree either: (1) that the agreed rate is a discount rate; or (2) that the agreed rate is a Negotiated Rate (or Negotiated Rate Formula). In the event that the parties agree upon a Negotiated Rate or Negotiated Rate Formula, this Agreement shall be subject to Section 49 of the General Terms and Conditions of Natural's Tariff.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Substitute Original Sheet No. 616A Substitute Original Sheet No. 616A

		ΕΣ	DATED .	ted Rate Agreement - DSS]	
				FICLE 1 RATE PARAMETERS	
1 1	Manatiat	- d D- t			
1.1	_	ed Rate Term: From	Through	• • • • •	
1.2	_	ed Rates:		5A (D.)	
	(a)	or negotiated monthly Bas		te: [\$ /Dth. of	MDQ, or negotiated rate
rate	(b)	Negotiated Base Commod	dity Rate: [\$	/Dth., or negot	iated rate or negotiated
race		formula]			
1.3	Eligible	Firm Quantities:			
	(a) (b)	Eligible Contract MDQ: Eligible Contract MSV:		-	
1.4	Eligible	Primary Receipt Points	s: [as set forth :	in Section 7.5 of the DSS	Rate Schedule]
1.5		Secondary Receipt Poir		ed Rates shall apply to s	ervice provided on a firm
1.6	Eligible	Primary Delivery Point	ts: [as set forth	in Exhibit A, unless oth	erwise listed herein]
		NAME 		PIN 	ELIGIBLE POINT MDQ (Dth/d.)
1.7	Eligible	Secondary Delivery Pos	ints: The Negotia [,]	ted Rates shall apply to	service provided on a
firm	basis to	the following secondar	ry delivery point:	s:	
1.8		ed Overrun Rate(s): [I if any.]	Describe the appl:	icable negotiated rate fo	r Authorized Overrun
1.9	applicab		r Natural's list o	ther rate-related provision permissible discount preement.]	
		NEGOTIA		FICLE 2 ND CONDITIONS; ADDITIONAL	CHARGES
2.1 provide	d	-	_	ated Rates shall apply on	_
Eligibl	е	_		nt from the Eligible Rece	_
	quantity transpor	equal to the Eligible ted on a firm basis und	Firm Quantities s der the Storage Ag	and (ii) an aggregate met forth in Section 1.3 greement and any associat tities stored on a firm b	above, for all quantitie ed capacity release
Shipper		associated capacity re	lease replacement	shippers which: (i) are	in excess of the
applica	Eligible Receipt daily ra	or Delivery Points, Shi te equivalent of the Ne	ipper shall be cha egotiated Rates; a	eceipt or delivery points arged the greater of: (a and (b) the 100% load fac or applicable maximum ba	the 100% load factor tor daily rate equivalent
forth	in Natura	al's FERC Gas Tariff, a	as mav be revised	from time to time, as we	ell as all applicable

FERC Docket: RP09-855-002

Original Sheet No. 616B Original Sheet No. 616B

EXHIBIT B [Negotiated Rate Agreement - DSS]

DATED

EFFECTIVE

(CON'T)

2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural is:

- (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges as part of the negotiated rates.]
- 2.3 Additional Rates, Charges, and Surcharges: In addition to the Negotiated Rates and unless otherwise

expressly provided in this Agreement or agreed to in writing by Natural, Shipper shall also pay Natural all other applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to time, including without limitation all applicable maximum: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation charges and surcharges; (iv) commodity charges and surcharges; (v) ACA surcharges; and (vi) gathering charges, offshore

charges,

and lateral line charges. Notwithstanding the foregoing sentence, to the extent that any such additional maximum rate, charge, surcharge, or penalty is derived from the applicable base reservation rate set forth in Natural's FERC Gas Tariff, then Natural shall derive such additional

rate, charge, surcharge, or penalty utilizing the greater of: (i) the applicable maximum base reservation rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time; and (ii) the Negotiated Rates.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 616C Original Sheet No. 616C

		Γ	B [Discount Agreement - DSS] DATED	
	per:			
		Γ	ARTICLE 1 DISCOUNT PARAMETERS	
1.1	Discount Ter	rm: FromThrough		
1.2	Discounted F	Rates:		
		scounted Monthly Base Reservat scounted Base Commodity Rate:		f MDQ.
1.3	Discounted E	Firm Quantities:		
		scounted Contract MDQ: scounted Contract MSV:		
1.4	Discounted E	Primary Receipt Points: [as se	et froth in Section 7.5 of the	DSS Rate Schedule]
1.5	Discounted S	Secondary Receipt Points: The	Discounted Rates shall apply	to service provided on a
firm		the following secondary receip	· · · · · · · · · · · · · · · · · · ·	
1.6	Discounted E	Primary Delivery Points: [as s	set forth in Exhibit A, unless	otherwise listed herein]
		NAME	PIN	DISCOUNTED POINT MDQ (Dth/d.)
1 7				
1.7 firm	basis to the	Secondary Delivery Points: The e following secondary delivery	points:	to service provided on a
• • • • • • •				
1.8	Authorized (Service, if	Overrun Rate(s): [Describe the any.]	ne applicable discount rate fo	r Authorized Overrun
1.9	applicable F	Related Provisions: [Describe Rate Schedule and/or Natural's se covered by the Discount Agr	s list of permissible discount	
		DISCOUNT TERMS	ARTICLE 2 AND CONDITIONS; APPLICABLE CH	ARGES
2.1	Shipper by N	count Limitations: The Discour Natural under the Storage Agre		
Discour		ints during the Discount Term;	and (ii) an aggregate maximu	m daily firm storage
quantit		e Discounted Firm Quantities s	set forth in Section 1.3 above	, for all quantities stored
on	a firm basis	s under the Storage Agreement	and any associated capacity r	elease replacement
agreeme	ents. Unless other	3	-	-
any	aggregate qu	rwise agreed, Shipper shall be set forth in Natural's FERC Ga uantities stored on a firm bas shippers which: (i) are in ex	as Tariff, as may be revised f sis for Shipper and any associ	rom time to time, for any ated capacity release

FERC Docket: RP09-855-002

Original Sheet No. 616D Original Sheet No. 616D

EXHIBIT B [Discount Agreement - DSS]
DATED
EFFECTIVE

2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural

is:

- (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges to Shipper.]
- 2.3 Applicable Maximum Rates, Charges, and Surcharges: Unless otherwise expressly provided in this Agreement or agreed to in writing by Natural, all applicable maximum rates, charges, surcharges, and

penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to time,

shall apply to service provided to Shipper under the Storage Agreement, including without limitation

all applicable: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized

overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and surcharges; (v) ACA

surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 616E Original Sheet No. 616E

J		EXHIBIT B [Negotiated Rate Agreement - NSS] DATED					
		EFFECTIVE					
		ARTICLE 1 NEGOTIATED RATE PARAMETERS					
1.1	Negotiated Rate Term: FromThrough						
1.2	Negotiate	ed Rates:					
	(a)	Negotiated Monthly Base Reservation Rate: [\$ /Dth. of MDQ, or the negotiated					
rate or		negotiated rate formula] Negotiated Base Commodity Rate: [\$/Dth, or the negotiated rate or					
negotia	ted rate	formula]					
1.3	Eligible	Firm Quantities:					
	(a) (b)	Eligible Contract MDQ: \$ /Dth. of MDQ. Eligible Contract MSV: \$ /Dth.					
1.4	Authorize Service,	ed Overrun Rate(s): [Describe the applicable negotiated rate for Authorized Overrun if any.]					
1.5	applicabl	ce Related Provisions: [Describe any other rate-related provision authorized by the le Rate Schedule and/or Natural's list of permissible discount provisions and which are twise covered by the Negotiated Rate Agreement.]					
		ARTICLE 2 NEGOTIATED RATE TERMS AND CONDITIONS; ADDITIONAL CHARGES					
2.1	Shipper baggregate	Regotiated Rate Limitations: The Negotiated Rates shall apply only to service provided to by Natural under the NSS Agreement during the Negotiated Rate Term, up to a maximum are firm quantity stored on a firm basis under the NSS Agreement and any associated					
capacit	release r and any a	replacement agreements. For any aggregate quantities stored on a firm basis for Shipper associated capacity release replacement shippers which are in excess of the Eligible MDQ or the Eligible Contract MSV, as applicable, Shipper shall be charged the greater					
of:		(a) the 100% load factor equivalent of the Negotiated Rates; and (b) the 100% load factor equivalent of the applicable maximum monthly base reservation rate set forth in Natural's FERC					
Gas	Tariff, as may be revised from time to time.						
2.2		able Third Party Surcharges: From time to time, certain reservation and/or commodity es may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural					
is:	permitted Natural's	red to collect from Shipper and remit to the FERC or to another third party; and (ii) to discount the amount of such surcharge from the applicable maximum rate set forth in FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable as covered by this Section and the applicability of such surcharges as part of the					
negotia							
2.3		al Maximum Rates, Charges, and Surcharges: Unless otherwise expressly provided in this or agreed to in writing by Natural, all applicable maximum rates, charges, surcharges,					
and	penalties	s of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to					
time,		ply to service provided to Shipper under the Storage Agreement, including without					
unautho	all appli	cable: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and					
ACA		charges; (iii) reservation surcharges; (iv) commodity rates, charges and surcharges; (v)					

surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 616F Original Sheet No. 616F

	EXHIBIT B [Discount Agreement - NSS] DATED
	EFFECTIVE
	er: act No.:
	ARTICLE 1 DISCOUNT PARAMETERS
1.1	Discount Term: FromThrough
1.2	Discounted Rates:
	(a) Discounted Monthly Base Reservation Rate: \$ /Dth. of MDQ. (b) Discounted Base Commodity Rate: \$ /Dth. of MDQ.
1.3	Discounted Firm Quantities:
	(a) Discounted Contract MDQ: Dth./day. (b) Discounted Contract MSV: Dth.
1.4	Authorized Overrun Rate(s): [Describe the applicable discount rate for Authorized Overrun
Service,	if any.]
1.5	Other Rate Related Provisions: [Describe any other rate-related provision authorized by the applicable Rate Schedule and/or Natural's list of permissible discount provisions and which are not otherwise covered by the Discount Agreement.]
	ARTICLE 2 DISCOUNT TERMS AND CONDITIONS; APPLICABLE CHARGES
2.1	General Discount Rate Limitations: The Discounted Rates shall apply only to service provided to Shipper by Natural under the NSS Agreement during the Discount Term, up to a maximum aggregate
firm	quantity equal to the respective Discounted Contract MDQ and the Discounted Contract MSV set
forth in	n Section 1.3 above, for all quantities stored on a firm basis under the NSS Agreement and any associated capacity release replacement agreements. For any aggregate quantities stored on a
firm	basis for Shipper and any associated capacity release replacement shippers which are in excess of
the	Discounted Contract MDQ or the Discounted Contract MSV, as applicable, Shipper shall be charged
the	greater of: (a) the 100% load factor equivalent of the Discounted Rates; and (b) the 100% load factor equivalent of the applicable maximum monthly base reservation rate set forth in Natural's
FERC	Gas Tariff, as may be revised from time to time.
2.2	Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural
LS:	(i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges to Shipper.]
2.3	Applicable Maximum Rates, Charges, and Surcharges: Unless otherwise expressly provided in this Agreement or agreed to in writing by Natural, all applicable maximum rates, charges, surcharges,
and	penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to
cime,	shall apply to service provided to Shipper under the Storage Agreement, including without
Limitati	all applicable: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and
inauthoi ACA	rized overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and surcharges; (v)

surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 616G Original Sheet No. 616G

			EXHIBIT	B [Negotiat	ted Rate Agr	eement - 1	FRSS]		
				DATED .					
Shinne	er.								
					RTICLE 1 D RATE PARAM	ETERS			
1.1	Negotiate	ed Rate Term: Fi	om	.Through					
1.2	_	ed Monthly Base ed rate formula		Rate: [\$	/Dt	h. of MDQ	, or negoti	iated rate or	
1.3	Eligible	Firm Quantities	s:						
	(a) (b)	Eligible Contra Eligible Contra							
	If Shippe	er elects delive	ered option,	Sections 1.	.4 through 1	.7 apply.			
1.4	Eligible	Primary Receipt	Points: [a	s set forth	in Exhibit	A]			
1.5		Secondary Recestorovided on a fi							
1.6	Eligible	Primary Deliver	ry Points: [as set forth	h in Exhibit	A, unles	s otherwise	e listed here	in]
		NAME 				PIN		ELIGIBLE POIN MDQ (Dth/d.)
1.7		Secondary Delivorovided on a fi							
1.8	Authorize,	ed Overrun Rate if any.]	(s): [Descr	ibe the appl	licable nego	tiated ra	te for Auth	norized Overr	un
1.9	applicabl	te Related Provi Le Rate Schedule rwise covered by	and/or Nat	ural's list	of permissi				
			NEGOTIATED		RTICLE 2 AND CONDITIO	NS; ADDIT	IONAL CHARG	GES	
2.1	General N	Negotiated Rate	Limitations	: The Negoti	iated Monthl	y Base Re	servation F	Rate shall ap	ply
-	to: (i) service provided to Shipper by Natural under the Storage Agreement from the Eligible Receipt Points to the Eligible Delivery Points during the Negotiated Rate Term; and (ii) a								
naximum all	Eligible transport	e negotiated rat Contract MSWQ s Led, if the Stor and any assoc	set forth in Tage Agreeme	Section 1.3 nt is for De	3 above for elivered Ser	all quanta vice) on a	ities store a firm basi	ed (and/or is under the	Storag
oe oe	applicabl	le maximum rates	s, charges a	nd surcharge	es set forth	in Natura	al's FERC G	Gas Tariff, a	s may
Storage	revised f	from time to tir	ne, for any	aggregate qı	uantities st	ored (and,	or transpo	orted, if the	
2	Agreement	is for Deliver	red Service)	on a firm k	basis for Sh	ipper and	any associ	iated capacit	У
release		ent shippers whi applicable, or							
	or Doline	ary Points							

FERC Docket: RP09-855-002

Original Sheet No. 616H Original Sheet No. 616H

- 2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural is:
 - (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges as part of the negotiated rates.]
- 2.3 Additional Rates, Charges, and Surcharges: In addition to the Negotiated Monthly Base Reservation Rate and unless otherwise expressly provided in this Agreement or agreed to in writing by Natural,
- Shipper shall also pay Natural all other applicable maximum rates, charges, surcharges, and penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to time, including without limitation all applicable maximum: (i) Fuel and Gas Lost and Unaccountable
- For charges; (ii) authorized and unauthorized overrun charges; (iii) reservation charges and surcharges; (iv) commodity charges and surcharges; (v) ACA surcharges; and (vi) gathering
- $\,$ offshore charges, and lateral line charges. Notwithstanding the foregoing sentence, to the extent
- that any such additional maximum rate, charge, surcharge, or penalty is derived from the applicable
 - base reservation rate set forth in Natural's FERC Gas Tariff, then Natural shall derive such additional rate, charge, surcharge, or penalty utilizing the greater of: (i) the applicable maximum base reservation rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time; and (ii) the Negotiated Monthly Base Reservation Rate.

Effective Date: 11/09/2009 Status: Effective FERC Docket: RP09-855-002

Original Sheet No. 6161 Original Sheet No. 6161

	EXF	HIBIT B [Discount Agreeme		
		DATED EFFECTIVE		
	er:act No.:			
		ARTICLE 1 DISCOUNT PARAMETER	S	
1.1	Discount Term: FromThrou	ıgh		
1.2	Discounted Monthly Base Reservation	on Rate: \$/Dth	. of MDQ.	
1.3	Discounted Firm Quantities:			
	(a) Discounted Contract MDQ: (b) Discounted Contract MSWQV			
	If Shipper elects delivered option	n, Section 1.4 through 1.	7 apply.	
1.4	Discounted Primary Receipt Points:	: [as set forth in Exhibi	t A]	
1.5	Discounted Secondary Receipt Point service provided on a firm basis i			
1.6	Discounted Primary Delivery Points	s: [as set forth in Exhib	it A, unless	otherwise listed herein]
	NAME		PIN	DISCOUNTED POINT MDQ (Dth/d.)
1.7	Discounted Secondary Delivery Poir to service provided on a firm basi			
1.8	Authorized Overrun Rate(s): [Desc Service, if any.]	cribe the applicable disc	ount rate for	Authorized Overrun
1.9	Other Rate Related Provisions: [I applicable Rate Schedule and/or Na not otherwise covered by the Disco	atural's list of permissi		
	DISCOUNT	ARTICLE 2 T TERMS AND CONDITIONS; A	PPLICABLE CHAI	RGES
2.1	General Discount Limitations: The	Discounted Monthly Base	Reservation Ra	ate shall apply only to:
(i)	service provided to Shipper by Nat	tural under the FRSS Agre	ement from the	e Discounted Receipt
Points	to the Discounted Delivery Points dur	ring the Discount Term; a	nd (ii) a max:	imum aggregate discount
firm	quantity equal to the respective I	Discounted Contract MDQ a	nd the Discour	nted Contract MSWQ set
forth	in Section 1.3 above, for all quar Delivered Service) on a firm basis replacement agreements. Shipper s surcharges set forth in Natural's aggregate quantities stored (and/o	s under the FRSS Agreemen shall be charged all appl FERC Gas Tariff, as may	t and any asso icable maximur be revised fro	ociated capacity release m rates, charges, and om time to time, for any
on a	firm basis for Shipper and any ass	sociated capacity release	replacement s	shippers which: (i) are
in	excess of the respective Discounter	ed Contract MDQ and the D	iscounted Cont	tract MSWQ, as applicable
or	(ii) involve any receipt or delive	ery points which are not	Discounted Red	ceipt or Delivery Points.

FERC Docket: RP09-855-002

Original Sheet No. 616J Original Sheet No. 616J

EXHIBIT B [Discount Agreement - FRSS]
DATED
EFFECTIVE

- 2.2 Discountable Third Party Surcharges: From time to time, certain reservation and/or commodity surcharges may be approved by the FERC for inclusion in Natural's FERC Gas Tariff which Natural is:
- (i) required to collect from Shipper and remit to the FERC or to another third party; and (ii) permitted to discount the amount of such surcharge from the applicable maximum rate set forth in Natural's FERC Gas Tariff ("Discountable Third Party Surcharges"). [Describe any applicable surcharges covered by this Section and the applicability of such surcharges to Shipper.]
- 2.3 Applicable Maximum Rates, Charges, and Surcharges: Unless otherwise expressly provided in this Agreement or agreed to in writing by Natural, all applicable maximum rates, charges, surcharges, and

penalties of any nature set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

shall apply to service provided to Shipper under the Storage Agreement, including without limitation

all applicable: (i) Fuel and Gas Lost and Unaccountable For charges; (ii) authorized and unauthorized

overrun charges; (iii) reservation surcharges; (iv) commodity rates, charges and surcharges; (v) ACA

surcharges; and (vi) gathering charges, offshore charges, and lateral line charges.

Effective Date: 04/18/2008 Status: Effective

FERC Docket: RP08-319-000

 $\textbf{Original Sheet No. 617} \; \texttt{Original Sheet No. 617} \; : \; \; \texttt{Pending}$

[FOR RATE SCHEDULE PALS]

	Con	ntract No
	NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL) PARK AND LOAN SERVICE AGREEMENT DATED UNDER SUBPART OF PART 284 OF THE FERC'S REGULATIONS	
1.	SHIPPER is:, a	
2.	Maximum Aggregate Quantity (Dth) (On any day, the total can never exceed this quantity.)	al of all MAQ ROs
3.	Daily Quantity, Receipt and Delivery Point: As set forth in the applicable PAL	3 RO.
4. hereunde	written notice by either party. Termination shall not discharge any obligations	
5.	Service will be ON BEHALF OF: Shipper or Other:	
6.	SHIPPER'S ADDRESSES NATURAL'S ADDRESSES	
7.	This Agreement supersedes and cancels a Agreement dated Other:	
8. form,	PALS Request Order: The form of the PALS RO attached hereto or such other mutua	ally agreeable
•	when executed by the parties shall evidence their agreement as to the terms of transaction for the Park and Loan Service pursuant to this Agreement, including	
rate,	Receipt and Delivery Points for parking and lending and the term.	
9.	PARK AND LOAN QUANTITY:	
	(i) Park Service: shall consist of Natural's receipt of a quantity of nature designated Receipt Point(s) on the designated date, requested by Shipper	
and account	approved by Natural, Natural's holding of such parked quantity of gas for	c Shipper's
Deliver	and Natural's redelivery of the parked quantity of gas to Shipper at the	designated
Deliver;	Point(s) and on the designated date(s) set forth in such PALS RO.	
in	(ii) Loan Service: shall consist of Natural lending a specified quantity of requested by Shipper and approved by Natural, from designated Delivery Po	
	Shipper's PALS RO and the Shipper's redelivery of and Natural's acceptant for Shipper's account at the designated Receipt Point(s) on the designated	
forth	in such PALS RO.	
10.	RATES: Except as otherwise provided below or in any written agreement(s) betwee effect during the term hereof, Shipper shall pay Natural the applicable maximum other lawful charges as specified in Natural's applicable rate schedule. Shipper agree that Shipper shall pay a rate other than the applicable maximum rate so le is between the applicable maximum and minimum rates specified for such service. Natural and Shipper may agree that a specific discounted rate will apply only to under the agreement. The parties may agree that a specified discounted rate will specified volumes (MDQ, MAQ or commodity volumes) under the agreement; that a specified rate will apply only if specified volumes are achieved or the volumes specified level; that a specified discounted rate will apply only during specific the year or for a specifically defined period; that a specified discounted rate to specified points, zones or other defined geographical area(s); and/or that a	rate(s) and all r and Natural may ong as such rate in the Tariff. to certain volumes a apply only to pecified a do not exceed a ded periods of will apply only

discounted rate(s) will apply in a specified relationship to the volumes actually tendered. Notwithstanding the foregoing, no discount agreement may provide that an agreed discount as to a certain volume level will be invalidated if the Shipper transports an incremental volume

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 618 Substitute First Revised Sheet No. 618

Superseding: Original Sheet No. 618

[FOR RATE SCHEDULE PALS]

Contract No.

NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL)
PARK AND LOAN SERVICE AGREEMENT DATED
UNDER SUBPART OF PART 284 OF THE FERC'S REGULATIONS
(CON'T)

above that agreed level. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable maximum rate at the time the discount agreement

was executed subsequently exceeds the applicable maximum rate due to a change in Natural's $\max_{i=1}^{n} \sum_{j=1}^{n} a_{ij} \sum_{$

rates so that such rate component must be adjusted downward to equal the new applicable maximum rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long

as none of the resulting rate components exceed the maximum rate applicable to that rate component.

Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates

which

had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. If the parties agree upon a rate other than the applicable maximum rate, such written Agreement(s) shall specify that the parties mutually agree either: (1) that the agreed rate is a discount rate; or (2) that the agreed rate is a Negotiated Rate (or Negotiated Rate Formula). In the event that the parties agree upon a Negotiated Rate or Negotiated Rate Formula, this Agreement shall be subject to Section 49 of the General Terms and Conditions of Natural's Tariff.

- 11. Any or all of the following provisions may be included (where applicable) in maximum rate agreements or in negotiated rate or discount contracts, if any:
- a. (DISCOUNTED RATE AGREEMENTS ONLY) Applicable Maximum and Minimum Rates. Notwithstanding any other provision of this Agreement, in no event shall a discounted rate billed by Natural be less than the applicable minimum rate or more than the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.
- b. (NEGOTIATED RATE AGREEMENTS ONLY) Maximum and Minimum Tariff Rates. Unless otherwise expressly provided in this Agreement, the negotiated rates shall apply to service provided by

 Natural to Shipper for the term of the Agreement notwithstanding any otherwise applicable

maximum or minimum rates set forth in Natural's FERC Gas Tariff as revised from time to time.

- c. (DISCOUNTED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund to

 Shipper any amounts collected for service to which the discounted rate(s) apply, unless the relevant discounted rate billed to Shipper exceeds the corresponding applicable effective maximum rates set forth in Natural's FERC Gas Tariff, as approved by the FERC from time to
- d. (NEGOTIATED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund to

 Shipper any amounts collected for service to which the negotiated rates apply,

notwithstanding

any otherwise applicable maximum or minimum rate set forth in Natural's FERC Gas Tariff, as

may be revised from time to time.

e. Notifications. Except as otherwise may be expressly provided herein, any notice or communication contemplated or required by this Agreement shall be in writing unless oral notification is expressly authorized herein, and shall be sent to the appropriate party at

the relevant address set forth in this Agreement, as may be revised from time to time.

f. Nonwaiver of Rights. No delay or failure to exercise any right or remedy accruing to either Natural or Shipper upon breach or default by the other will impair any right or remedy or be

construed to be a waiver of any such breach or default, nor will a waiver of any single breach

be deemed a waiver of any other breach or default.

Succession and Assignment. Any entity which shall succeed by purchase, merger or consolidation

to title to the properties, substantially as an entirety, of Natural or Shipper as the case may be, shall be entitled to the rights and shall be subject to the obligations of its predecessor in title under this Agreement. No other assignment of this Agreement nor of any

of the individual rights or obligations hereunder by Shipper shall be effective as to

Natural

without the prior express written consent of Natural.

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 619 Substitute First Revised Sheet No. 619

Superseding: Original Sheet No. 619

[FOR RATE SCHEDULE PALS]

Contract No.

.

NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL)
PARK AND LOAN SERVICE AGREEMENT DATED
UNDER SUBPART OF PART 284 OF THE FERC'S REGULATIONS
(CON'T)

- h. No Third Party Beneficiaries. This Agreement shall not create any rights in any third parties,
 and no provision of this Agreement shall be construed as creating any obligations for the
- benefit of, or rights in favor of, any person or entity other than Natural or Shipper.

 i. Conformance to Law. It is understood that performance hereunder shall be subject to all
- valid

 laws, orders, rules and regulations of duly constituted governmental authorities having jurisdiction or control of the matters related hereto, including without limitation the Federal Energy Regulatory Commission.
 - j. Effect of Tariff. This Agreement shall at all times be subject to all applicable provisions of Natural's FERC Gas Tariff.
 - k. GOVERNING LAW. THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT OF THIS AGREEMENT SHALL BE GOVERNED BY THE LAWS OF THE STATE OF, EXCLUDING ANY CONFLICT OF LAW RULE WHICH WOULD REFER ANY MATTER TO THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF

written instrument executed by Natural and Shipper which expressly refers to this Agreement.

- 1. Entire Agreement. This Agreement contains the entire agreement between Natural and Shipper with respect to the subject matter hereof, and supersedes any and all prior understandings and agreements, whether oral or written, concerning the subject matter hereof, and any and all such prior understandings and agreements are hereby deemed to be void and of no effect. No amendments to or modifications of this Agreement shall be effective unless agreed upon in a
 - 12. The above-stated Rate Schedule, as revised from time to time, controls this Agreement and is incorporated herein. The attached PALS Request Order Form is a part of this Agreement. Shipper shall provide the actual end user purchaser name(s) to Natural if Natural must provide them to

the

Agreed to by:

FERC.

NATURAL SHIPPER

 /s/:
 /s/:

 NAME:
 NAME:

 TITLE:
 TITLE:

Effective Date: 04/18/2008 Status: Effective FERC Docket: RP08-319-000

Original Sheet No. 620 Original Sheet No. 620: Pending

			RO No.:
• • • • • • • • • • • • • • • • • • • •			
	PALS REQUEST DATED		
Shipper:		Type of Service:	Park Loan
Point Name:		MAO RO:	
(Dth)			
Delivery Pressure:		PALS Agreement #:	
PIN:		Rate:	
Term: Start E			
Schedule:			
Date(s) Service to be Provided		Daily Quantity	(Dth)
The same the			Tana an Dank Withdraw
From Through		ayback 	Loan or Park Withdrawal
		• • • • • • • • • •	
• • • • • • • • • • • • • • • • • • • •			
• • • • • • • • • • • • • • • • • • • •			
		• • • • • • • • • • • • • • • • • • • •	
	• • • • • • • • • • • • • • • • • • • •	• • • • • • • • • • • • • • • • • • • •	
		• • • • • • • • • • • • • • • • • • • •	
MAQ RO (total of daily volumes):		=	
Park or Loan Payback = volumes wh Loan or Park Withdrawal = volumes			
Service will be ON BEHALF OF:			
Shipper or			
Other:		, a	
SHIPPER'S CONTACT AND ADDRESS			
Phone:		Fax:	
	• • • • • •		
Other Conditions (if any):			
	• • • • • • • • • • • • • • • • • • • •	• • • • • • • • • • • • • • • • • • • •	
Agreed to by: NATURAL	S	SHIPPER	

/s/:	•	•	•	•	•	•	 	 •	•	•	٠	•	•	•	•	•	 	 •	•	•	•	•	•	•	•	•	•	•	 	•	•	•	•	•	•		/:	s /	:	
NAME:																																					N	MA	Ε	:
TITLE																													 								Т	ΙΤ	L	Ε

Effective Date: 04/18/2008 Status: Effective

FERC Docket: RP08-319-000

Original Sheet No. 621 original Sheet No. 621 : Pending

[FOR RATE SCHEDULE LPS]

				Contract No
		NATURAL GAS PIPELINE COMPANY LINE PACK SERVICE (PARK AND LOAN) AGRI UNDER SUBPART OF PART	EEMENT DATED	
1.	SHIPPE	R is:	, a	
2.		m Aggregate Quantity (Dth)ver exceed this quantity.)	(On any day, the	total of all MAQ ROs
3.	Maximu: quanti	nm Daily Quantity (Dth) (the	e total of all MDQ RO can ne	ver exceed this
4.	writte			
	Servic	ee will be ON BEHALF OF: Shipper or Other:		
6.	SHIPPE	R'S ADDRESSES	NATURAL'S ADDRESSES	
7.		This Agreement supersedes and cancels a		
8. rate,	when e transa Receip	equest Order: The form of the LPS RO attack executed by the parties shall evidence their action for the Park and Loan Service pursuant and Delivery Points for parking and lending equantity and term of a Park and Loan. A limits specified.	r agreement as to the terms nt to this Agreement, includ ing and the term. The LPS R	of the particular ing the quantity, O may specify a range
9.		ND LOAN QUANTITY:		
	(i)	Park Service: shall consist of Natural's designated Receipt Point(s) on the designated		
and		approved by Natural, Natural's holding of	such parked quantity of gas	for Shipper's
account		and Natural's redelivery of the parked qua	antity of gas to Shipper at	the designated
Deliver	У	Point(s) and on the designated date(s) set	forth in such LPS RO.	
	(ii)	Loan Service: shall consist of Natural le requested by Shipper and approved by Natur		
in		Shipper's LPS RO and the Shipper's redeling for Shipper's account at the designated Re		
forth		in such LPS RO.		
10.	effect other agree is bet Natura under specif discou specif	Except as otherwise provided below or in a during the term hereof, Shipper shall pay lawful charges as specified in Natural's at that Shipper shall pay a rate other than the ween the applicable maximum and minimum rate. I and Shipper may agree that a specific distribute agreement. The parties may agree that a field volumes (MDQ, MAQ or commodity volumes) inted rate will apply only if specified volumed rate will apply only if specified volumed rate will apply only if specified volumed rate or for a specifically defined period; the	Natural the applicable maxi- policable rate schedule. Shi- ne applicable maximum rate s ces specified for such servi- scounted rate will apply onla a specified discounted rate under the agreement; that times are achieved or the vol- e will apply only during spe	mum rate(s) and all pper and Natural may o long as such rate ce in the Tariff. y to certain volumes will apply only to a specified umes do not exceed a cified periods of

to specified points, zones or other defined geographical area(s); and/or that a specified discounted rate(s) will apply in a specified relationship to the volumes actually tendered. Notwithstanding the foregoing, no discount agreement may provide that an agreed discount as to a certain volume level will be invalidated if the Shipper transports an incremental volume

FERC Docket: RP09-855-002

which

time.

the

Substitute First Revised Sheet No. 622 Substitute First Revised Sheet No. 622

Superseding: Original Sheet No. 622

[FOR RATE SCHEDULE LPS]

Contract No.

NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL)
LINE PACK SERVICE (PARK AND LOAN) AGREEMENT DATED
UNDER SUBPART OF PART 284 OF THE FERC'S REGULATIONS (CON'T)

above that agreed level. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable maximum rate at the time the discount agreement

was executed subsequently exceeds the applicable maximum rate due to a change in Natural's $\max_{x \in \mathcal{X}} \max_{x \in \mathcal{X}} \min_{x \in \mathcal{X}} \max_{x \in \mathcal{X}} \min_{x \in \mathcal{X}} \min_{$

rates so that such rate component must be adjusted downward to equal the new applicable maximum rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long

as none of the resulting rate components exceed the maximum rate applicable to that rate component.

Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates

had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. If the parties agree upon a rate other than the applicable maximum rate, such written Agreement(s) shall specify that the parties mutually agree either: (1) that the agreed rate is a discount rate; or (2) that the agreed rate is a Negotiated Rate (or Negotiated Rate Formula). In the event that the parties agree upon a Negotiated Rate or Negotiated Rate Formula, this Agreement shall be subject to Section 49 of the General Terms and Conditions of Natural's Tariff.

- 11. Any or all of the following provisions may be included (where applicable) in maximum rate agreements or in negotiated rate or discount contracts, if any:
- a. (DISCOUNTED RATE AGREEMENTS ONLY) Applicable Maximum and Minimum Rates. Notwithstanding any other provision of this Agreement, in no event shall a discounted rate billed by Natural be less than the applicable minimum rate or more than the applicable maximum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.
- b. (NEGOTIATED RATE AGREEMENTS ONLY) Maximum and Minimum Tariff Rates. Unless otherwise expressly provided in this Agreement, the negotiated rates shall apply to service provided by

 Natural to Shipper for the term of the Agreement notwithstanding any otherwise applicable maximum or minimum rates set forth in Natural's FERC Gas Tariff as revised from time to
- c. (DISCOUNTED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund to

 Shipper any amounts collected for service to which the discounted rate(s) apply, unless the relevant discounted rate billed to Shipper exceeds the corresponding applicable effective maximum rates set forth in Natural's FERC Gas Tariff, as approved by the FERC from time to
- time.

 d. (NEGOTIATED RATE AGREEMENTS ONLY) Refunds. In no event shall Natural be required to refund to

Shipper any amounts collected for service to which the negotiated rates apply, notwithstanding

any otherwise applicable maximum or minimum rate set forth in Natural's FERC Gas Tariff, as may be revised from time to time.

- e. Notifications. Except as otherwise may be expressly provided herein, any notice or communication contemplated or required by this Agreement shall be in writing unless oral notification is expressly authorized herein, and shall be sent to the appropriate party at relevant address set forth in this Agreement, as may be revised from time to time.
- f. Nonwaiver of Rights. No delay or failure to exercise any right or remedy accruing to either Natural or Shipper upon breach or default by the other will impair any right or remedy or be

construed to be a waiver of any such breach or default, nor will a waiver of any single breach

be deemed a waiver of any other breach or default.

 $\ensuremath{\mathtt{g}}.$ Succession and Assignment. Any entity which shall succeed by purchase, merger or consolidation

to title to the properties, substantially as an entirety, of Natural or Shipper as the case may be, shall be entitled to the rights and shall be subject to the obligations of its predecessor in title under this Agreement. No other assignment of this Agreement nor of any of the individual rights or obligations hereunder by Shipper shall be effective as to

Natural

without the prior express written consent of Natural.

FERC Docket: RP09-855-002

Substitute First Revised Sheet No. 623 Substitute First Revised Sheet No. 623

Superseding: Original Sheet No. 623

[FOR RATE SCHEDULE LPS]

Contract No.

.

NATURAL GAS PIPELINE COMPANY OF AMERICA LLC (NATURAL)
LINE PACK SERVICE (PARK AND LOAN) AGREEMENT DATED
UNDER SUBPART OF PART 284 OF THE FERC'S REGULATIONS
(CON'T)

- h. No Third Party Beneficiaries. This Agreement shall not create any rights in any third parties,
 and no provision of this Agreement shall be construed as creating any obligations for the benefit of, or rights in favor of, any person or entity other than Natural or Shipper.
- i. Conformance to Law. It is understood that performance hereunder shall be subject to all valid
 laws, orders, rules and regulations of duly constituted governmental authorities having jurisdiction or control of the matters related hereto, including without limitation the Federal Energy Regulatory Commission.
 - j. Effect of Tariff. This Agreement shall at all times be subject to all applicable provisions of Natural's FERC Gas Tariff.
 - k. GOVERNING LAW. THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT OF THIS AGREEMENT SHALL BE GOVERNED BY THE LAWS OF THE STATE OF, EXCLUDING ANY CONFLICT OF LAW RULE WHICH WOULD REFER ANY MATTER TO THE LAWS OF A JURISDICTION OTHER THAN THE STATE OF

Entire Agreement. This Agreement contains the entire agreement between Natural and Shipper

- with respect to the subject matter hereof, and supersedes any and all prior understandings and agreements, whether oral or written, concerning the subject matter hereof, and any and all such prior understandings and agreements are hereby deemed to be void and of no effect. No amendments to or modifications of this Agreement shall be effective unless agreed upon in a written instrument executed by Natural and Shipper which expressly refers to this Agreement.
 - 12. The above-stated Rate Schedule, as revised from time to time, controls this Agreement and is incorporated herein. The attached LPS Request Order Form is a part of this Agreement. Shipper shall provide the actual end user purchaser name(s) to Natural if Natural must provide them to

the FERC.

Agreed to by:

NATURAL

 /s/:
 /s/:

 NAME:
 NAME:

 TITLE:
 TITLE:

SHIPPER

Effective Date: 04/18/2008 Status: Effective FERC Docket: RP08-319-000

Original Sheet No. 624 Original Sheet No. 624 : Pending

					RO No.:
			EST ORDER FORM		
Shipper:			Type of Serv	ice: Park	Loan
 Initiation Poin	t Name(s) and PIN	(s):	MAQ RO:		
	t Name(s) and PIN	th) (s):	Minimum Aggre	egate	
Quantity: Term: Start		End	LPS Agreemen	t #:	
	• • • • • • • • • • • • • • • • • • • •				
Schedule:					
	to be Provided Range of Dates)		Daily Qua (May Reflect a Ra	ntity (Dth) ange of Volumes)	
From	Through	Park or Lo Minimum	an Payback Maximum	Loan or Park Minimum	Withdrawal Maximum
• • • • • • • • • • • • • • • • • • • •					
• • • • • • • • • • • • • • • • • • • •					
• • • • • • • • • • • • • • • • • • • •	• • • • • • • • • • • • • • • • • • • •		• • • • • • • • • • • • • • • • • • • •	• • • • • • • • • • • • • • • • • • • •	
Agreement Initial Fee:				in the Pro Forma So	ervice
Completion Fee:					
Park/Loan Balan	ce Fee:				
Park or Loan Pa	yback = volumes w	hich Natural RECEI	VES from Shipper. LIVERS to Shipper.	• • • • • • •	
Service will be Shipper	or				
Other:			• • • • • • • • • • • • • • • • • • • •	, a	
SHIPPER'S CONTA	CT AND ADDRESS				
Other Condition		• • • • • • • • • • • • • • • • • • • •			

Agreed to by:	
NATURAL	SHIPPER
/s/:	/s/:
NAME:	NAME:
NAME:	NAME:
TITLE:	TITLE:

Effective Date: 04/18/2008 Status: Effective

FERC Docket: RP08-319-000

Sheet Nos. 625 - 699 Sheet Nos. 625 - 699 : Pending

Sheet Nos. 625 through 699 has been reserved for future use.